AD 636 392

# FLIGHT TEST ENCINEERING HANDBOOK AFTR NO. 6273

United States Air Force Edwards Air Force Base, CA

Jan 66

## AF TECHNICAL REPORT NO. 5273

.

MAY 1951 Corrected and revised June 1964 - January 1966

1636392

-

ł

# FLIGHT TEST ENGINEERING HANDBOOK

RUSSEL M. HERRINGTON Najor, USAF

PAUL E. SHOEMACHER Captain, USAF

EUGENE P. BARTLETT 1st Lieutenant, USAF

EVERETT W. DUNLAP

REPRODUCED BY NATIONAL TECHNICAL INFORMATION SERVICE US. DEPARIMENT OF COMMERCE SPRINGFIELD, VA 22161

. .

54

UNITED STATES AIR FORCE AIR FORCE SYSTEMS COMMAND AIR FORCE FLIGHT TEST CENTER HOWARDS AIR FORCE BASE, CALIFORNIA

UNCLASSIFIED Security Classification			
المحجوب بيواباه الالمغر يسعون بعدك المحب ويقرن وجم بسينينين فأكلك الفلاش تيفيري بوفاعت فيعود			
DOCUMENT	CONTRUL DATA - RE		
(Security classification of title, body of abstract and in		intered when	
QRIGINATIN & ACTIVITY (Corporate author)		1	DRT SECURITY C LASSIFICATION
mited States Air Force, Air Force S		20 680	Inclassified
Air Force Flight Test Center, Edward	s AFB, Callf		N/A
REPORT TITLE Flight Test Engineering Handbook AF	TR No. 6273		1
DESCRIPTIVE NOTES (Type of report and inclusive detee Orrected and revised January 1966	)		
AUTHOR(S) (List name. (Int name. Initial) lerrington, Russel M., Major, USAF Shoemacher, Paul E., Capt, USAF Bartlett, Eugene P., 1st Lt, USAF	Bunlap, Eve	erett W.	
REPORT DATE	7. TOTAL NO. OF	PAGES	75 NO OF REFS
May 1951	687		30
CONTRACT OR GRANT NO.	S. ORIGINATOR'S	EPORT NU	IMBER(S)
N/A P Project no N/A	AFTR No. 627	3	
Ϋ́Ν/Λ	Sb OTHER REPORT this report)	NO(3) (Ar	y other numbers that may be assigned
A N/A	N/A		
1 SUPPLEMENTARY NOTES	Air Force F1	ight Te	st Center
	Elucate Arn		. I. Anna
N/A B ABSTRACT We thous of obtaining flight test data	Edwards AFB,		
	a for reciprocati re presented toge Correction of a led derivations o aphs and charts c incor are present	ng engi ther wi ircraft f corre ontaini	ne aircraft (including th various methods of performance to standa. ction factors and ng information required
ABSTRACT wethods of obtaining flight test data elicopters) and turbojet aircraft and ata analysis and data presentation. onditions is included, as are detail erformance parameters. Numerous gra- y and useful to the flight test engi- eduction forms and sample flight test	a for reciprocati re presented toge Correction of a led derivations o aphs and charts c ineer are present st programs.	ng engi ther wi ircraft f corre ontaini	ne aircraft (including th various methods of performance to standa. ction factors and ng information required
ABSTRACT ethods of obtaining flight test data elicopters) and turbojet aircraft and ata analysis and data presentation. onditions is included, as are detail erformance parameters. Numerous gray y and useful to the flight test engi eduction forms and sample flight test eduction forms and sample flight test manual sample flight test eduction forms and sample flight test satisfies a satisfies a satisfi	a for reciprocati re presented toge Correction of a led derivations o aphs and charts c ineer are present st programs.	ng engi ther wi ircraft f corre ontaini	ne aircraft (including th various methods of performance to standa. ction factors and ng information required
ABSTRACT ethods of obtaining flight test data elicopters) and turbojet aircraft and ata analysis and data presentation. onditions is included, as are detail erformance parameters. Numerous gray y and useful to the flight test engi eduction forms and sample flight test	a for reciprocati re presented toge Correction of a led derivations o aphs and charts c ineer are present st programs.	ng engi ther wi ircraft f corre ontaini ed, tog	ne aircraft (including th various methods of performance to standa. ction factors and ng information required

.





COPY





#### CLASSIFIED Security Classification

KEY WORDS		LI	LINK A		LINK		LINKC	
		ROLE	WT	ROLE	WT.	ROLE	W	
ght Test, Aircraft Performance, Reciprod bojet, Level Flight Performance, Climb H e, Descent Performance, Takeoff, Landing icopter Performance	Perform							
INSTR	UCTIONS							
RIGINATING ACTIVITY: Enter the name and address	imposed such as:	by securit	y clessifi	cetion, u	ing eten	dard state	mente	
.e sciusisy or other organization (corporate author) issuing report.		"Qualified report from		ma may ot	otein cop	ies of thi	•	
REPORT SECURITY CLASSIFICATION: Enter the over- ecurity classification of the report. Indicate whether atricted Data" is included. Marking is to be in accord-		"Foreign report by I	DDC is no	ot authoria	red."			
with appropriate security regulations. GROUP: Automatic downgrading is specified in DoD Di- ve 5200.10 and Armed Forces Industrial Nanual. Enter group number. Also, when applicable, show that optional	(3)	"U. S. Go this report users shal	directly	from DDC				
TPORT TITLE. Enter the complete report title in all the second se	(4)	"U. S. mįl report dire shall requ	ctly from	DDC. O				
igful title cannot be aelected without classifica- now title classification in all capitals in parenthesis mately following the title.	(5)	"All distr ified DDC					Qual	
ESCRIPTIVE NOTES: If appropriate, enter the type of e.g., interim, progress, summary, annual, or final.	If th	e report he	s been fu	rnished to	the Off	ice of Tea	.'' :hnice	

1.1. be inclusive dates when a specific reporting period is

THOR(S): Enter the name(s) of suthor(s) as shown on in the report. Enter last name, first name, middle initial, cilitary, show rank and branch of service. The name of principal withor is an absolute minimum requirement.

REPORT DATL. Enter the date of the report we day, ith, year, or month, year. If more than one date appears the report, use date of publication.

TOTAL NUMBER OF PAGES: The total page count of follow normal pagination procedures, i.e., enter the of pages containing information.

NUMBER OF REFERENCES. Enter the total number of grences cited in the report.

CONTRACT OR GRANT NUMBER: If appropriate, enter applicable number of the contract or grant under which report was written.

W: & 8d PROJECT NUMBER: Enter the appropriate stary department identification, such as project number, project number, system numbers, task number, etc.

ORIGINATOR'S REPORT NUMBER(S): Enter the offireport number by which the document will be identified controlled by the originating activity. This number must prugue to this report.

OTHER REPORT NUMBER(S): If the report has been i, ned any other report numbers (either by the originator b) the sponnor), also enter this number(s).

AVAILABILITY/LIMITATION NOTICES: Enter any limtions on further dissemination of the report, other than those

If the report has been furnished to the Office of Technical Services, Department of Commerce, for sale to the public, indicate this fact and enter the price, if known

1L SUPPLEMENTARY NOTES: Use for additional explanatory notes.

12. SPONSO: TWG MILITARY ACTIVITY: Enter the name of the departmental project office or laboratory sponsoring (payring for) the research and development. Include address.

13. ABSTRACT: Enter an abstract giving a brief and factual summary of the document indicative of the report, even though it may also uppear elsewhere in the body of the technical report. If additional space is required, a continuation sheet shall be attached

It is highly desirable that the abstract of classified reports be unclassified. Each paragraph of the abstract shall end with an indication of the military security classification of the information in the paragraph, represented as (TS). (S), (C), or (U)

There is no limitation on the length of the ubstract. However, the suggested length is from 150 to 225 words.

14 KEY WORDS: Key words are tschnically meaningful terms or short phrases that characterize a report and may be used us index entries for cataloging the report. Key words must be selected so that no security classification is required. Identiflers, such as equipment model designation, trade name, military project code name, geographic location, may be used as key words but will be followed by an indication of technical context. The assignment of links, rules, and weights is optional.



UNCLASSIFIED

Security Classification

## NOTICE

THIS DOCUMENT HAS BEEN REPRODUCED FROM THE BEST COPY FURNISHED US BY THE SPONSORING AGENCY. ALTHOUGH IT IS RECOGNIZED THAT CERTAIN PORTIONS ARE ILLEGIBLE, IT IS BEING RELEASED IN THE INTEREST OF MAKING AVAILABLE AS MUCH INFORMATION AS POSSIBLE.

#### FOREWORD

The publication of a series of handbooks on the performance testing and evaluation of all types of Air Force aircraft is planned by the Flight Research Division, Air Force Flight Test Center. This handbook has been issued as an interim measure to provide assistance to flight test personnel pending publication of the new series of handbooks. Chapters I and III of the original Technical Report Number 6273 have been replaced by AFFTC Technical Notes 59-22 and 59-47. These technical notes are on airspeed, altitude and temperature measurement, and turbojet engine performance. They represent updated and improved versions of the original contents of TR Number 6273.

As a matter of expediency, the numbering of charts, figures, and equations in the technical notes has been retained. This has led to inconsistency in the numbering system, but, since appropriate references in the text have been changed, it is felt that no confusion will result.

The United States Standard Atmosphere used as the basis for charts and tables in Chapter 1 is equivalent to the International Civil Aviation Organization (ICAO) Standard Atmosphere adopted by NACA on November 20, 1952 and contained in NACA Report 1235 "Standard Atmosphere - Tables and Data for Altitudes to 65,800 Feet", 1955 (Reference 6). The equations of this report were used to extend the tables to 80,000 feet. The properties tabulated in Chapter 1 are identical with those in the ARDC Model Atmosphere, 1956, the U. S. Extension to the ICAO Standard Atmosphere, 1958 (Reference 7) and the ARDC Model Atmosphere, 1959 (Reference 8). One exception should be noted: the sea-level speed of sound is taken as 1116.45 ft sec<sup>-1</sup> in Chapter 1, whereas it was 1116.89 ft sec<sup>-1</sup> in NACA Report 1235, since the ratio of specific heats,  $\gamma$ , was taken as 1.4 exactly for Chapter 1 and implied as 1.4011 in NACA Report 1235, on the basis of experimental values of sound speed.

The constants and conversion factors used in Chapters 2 through 7 and Appendixes I and II are based on the earlier "Standard Atmosphere Tables and Data", NACA Report 218, 1948. The gas constant, R used in Chapter I, e.g. in the perfect gas law  $P_a = \rho RT_a$ , has the dimensions ft<sup>2</sup>/sec<sup>2</sup>°K. It is equal to the product of the gas constant used in the remaining charters and the acceleration due to gravity. The dimensions of the latter is are ft/°K.

While this handbook continues to provide, in general, adequate instruction for conducting performance tests on turbojet and reciprocating engine powered conventional aircraft, Chapter VII, "Helicopter Flight Test Performance and Analysis", is in need of updating. Also, analysis is lacking in regard to high performance aircraft. Caution should be exercized in applying correction procedures to flight data obtained with this type of aircraft. For example, significant errors may be incurred in making corrections to climb data for wind gradients and for weight because of the simplifying assumptions which have been made.

The addition of a list of references has been made (reference TABLE OF CONTENTS). Contained in these references is considerable supplementary information including data on standard atmospheres  $^{6}$ ,  $^{7}$ ,  $^{8}$ , a review of aerodynamics prepared by the USAF Experimental Test Pilot School  $^{2}$ , and a comprehensive NATO flight test manual prepared under the auspices of the Advisory Group for Aeronautical Research and Development  $^{3}$ .

#### ABSTRACT

Methods of obtaining flight test data for reciprocating engine aircraft (including helicopters) and turbojet aircraft are presented together with various methods of data analysis and data presentation. Correction of airoraft performance to standard conditions is included, as are detailed derivations of correction factors and performance parameters. Numerous graphs and charts containing information required by and useful to the flight test engineer are presented, together with sample data reduction forms and sample flight test programs.

## DISTRIBUTION OF THIS DOCUMENT IS UNLIMITED

#### PUBLICATION REVIEW

Manuscript copy of this report has been reviewed and found satisfactory for publication.

FOR THE COMMANDING GENERAL:

6 BollishF

Colonel, USAF Chief, Flight Test Division



iii

# TABLE OF CONTENTS

LIST OF REFERE	ENCE AND COMPUTATIONAL CHARTS	ix
INTRODUCTION		xiii
	Speed, Altitude, and Temperature Instruments - and Calibration	
	Symbols Used in Chapter One	x٧
Section 1	The Standard Atmosphere	1
Section 2	Theory of Altitude, Air-Speed, Mach Number and Air Temperature Measurement	7
Section 3	Instrument Error - Theory and Calibration	24
Section 4	Pressure Lag Error - Theory and Calibration	28
Section 5	Position Error - Theory and Calibration	48
Section 6	Calibration of the Free Air Temperature Instrumentation	93
Section 7	Data Reduction Outlines	100
Section 8	Charts	118
Section 9	U. S. Standard Atmosphere Tables	265
CHAPTER TWO	Reciprocating Engine Performance	2-1
Section 2.1	Horsepower Determination for Test Conditions	2-1
Section 2.2	Power Correction for Temperature Variation at Constant Manifold Pressure	2-5
Section 2.3	Power Correction for Manifold Pressure Variation Resulting from Temperature Variation and Flight Mach Number Variation	2-6
Section 2.4	Power Correction for Turbosupercharger RPM and Back Pressure Variation at Constant Manifold Pressure	[ 2-13
Section 2.5	Critical Altitude	2-17
Section 2.6	Engine Data Plotting, Prop Load, BMEP Data, Supercharger Operation	2-18
Section 2.7		2-23
Section 2.8	•	2-25

Preceding page blank

\* . ..<del>.</del> .

**11 1** 

AFTR 6273

İ

v

CHAPTER THREE	Turbojet Engine	Performance
OIMPIER INKEE	Turbojet Engine	Feriormance

	Symbols Used in Chapter Three	3-i
Section 1	Introduction to Thrust Measurement	3-1
Section 2	Turbojet Engine Performance Parameters	3 - 8
Section 3	Air Induction System Performance	3-15
Section 4	Stancardization of Test Data with Engine Parameters	3-29
Section 5	Air Flow Measurement	3 - 33
Section 6	In-Flight Thrust Measurement	3 - 40
Section 7	Water Injection	3 - 62
Section 8	Data Reduction Cutlines	3 - 66
Section 9	Charts	3 - 68
CHAPTER FOUL	R Level Flight Performance	4-1
Section 4.1	Densicy Altitude and Pressure Altitude Flight Test Methods	4-1
Section 4.2	Aerodynamic Forces and Their Relation to Engine Power and Propulsive Thrust	4-2
Section 4.3	Speel-Power Curves - Reciprocating Engine Aircraft	4-6
Section 4.4	Weight Corrections for Speed Power Data - Reciprocating Engine Aircraft	4-9
Section 4.5	Configuration Change Corrections for Speed Power Data	4-11
Section 4.6	<ul> <li>The Generalized Power Parameter (PIW) and Speed Parameter (VIW) - Reciprocating Engine Aircraft</li> </ul>	4-13
Section 4.7	Fuel Consumption - Range and Endurance - Reciprocating Engine Aircraft	4-16
Section 4.8	S Speed Power Curves - Turbojet Aircraft	4-21
Section 4.9	<ul> <li>Weight Change Corrections for Speed Power data - Turbojet Aircraft</li> </ul>	4-28
Section 4.2	0 Fuel Consumption - Endurance and Range - Turbojet Aircraft	4-33
Section 4.1	1 Flight Thrust Measurement Applications to Drag and Lift Coefficient and Aircraft Efficiency Determination	4-42

•

CHAPTER FIVE	Climb and Descent Performance	5-1
Section 5.1	Rate of Climb Parameters - Derivation	5-1
Section 5.2	Temperature Variation Correction to Rate of Climb Data	5-6
Section 5.3	Weight Variation Correction to Rate of Climb Data	5-9
Section 5.4	Vertical Wind Gradient Correction to Rate of Climb Data	5-11
Section 5.5	Climb Path Acceleration Correction to Rate of Climb Data	5-13
Section 5.6	Temperature Effects on Fuel Comsumption and Weight During Climb	5-16
Section 5.7	Determination of Best Rate of Climb and Best Climbing Speed	5-18
Section 5.8	Dimensionless Rate of Climb Plotting	5-22
Section 5.9	General Climb Test Information	5-27
Section 5.10	Rate of Descent Data	5-35

CHAPTER SIX T	ake-off and Landing Performance	6-1
Section 6.1	Techniques and Configurations for Take- off Tests - JATO Operation	6-1
Section 6.2	Distance and Height Measurements and Equipment	6-2
Section 6.3	Take-off Data Corrections for Wind, Weight, and Density	6-6
Section 6.4	Landing Performance Tests and Corrections	6-15
Section 6.5	Dimensionless Parameters for Take-off and Landing Performance Data	6-18
CHAPTER SEVEN	Helicopter Flight Test Performance and Analysis	7 - 1
Section 7.1	Introduction	7 - 1
Section 7.2	Level Flight Performance	7-3
Section 7.3	Rotor Thrust, Power, and Efficiency in Hovering Flight	7-11

AFTR 6273

vii

Section 7.4	Climbs and Descents (Autorotation)	7-19
Section 7.5	Fuel Consumption, Endurance and Range	7-26
Section 7.6	Airspeed, Altimeter, and Temperature System Calibrations	7-31

8-1

# REFERENCES

.

•

APPENDIX I Density Altitude Charts	8 <b>-</b> 5
Differential -Static Pressure Ratio versus Mach Number (Supersonic-Normal Shock Condition)	8-17
Reynolds Number-Mach Number Ratio versus Pressure Altitude and Temperature	8-18
Supersonic Mach Number Functions	8-20
Psychrometric Chart	8-33

# APPENDIX II

Nomenclature	835
Physical Information and Systems of Units	8-46
Conversion Tables	8-47

viii

# LIST OF REFERENCE AND COMPUTATIONAL CHARTS

CHAPTER ONE		Page
CHART 8.1	H(G/g <sub>SL</sub> ) versus h-H(G/g <sub>SL</sub> )	118
CHART 8.2	M versus T <sub>a</sub> for constant t <sub>ic</sub> and constant Temperature Probe Recovery Factor, K	120
CHART 8.21	Compressibility Correction to Calibrated Airspeed	126
CHART 8.3	H <sub>c</sub> versus V <sub>e</sub> /M	128
CHART 8.4	V <sub>tt</sub> versus t <sub>c*</sub> for constant M	130
CHART 8.5	M versus V <sub>C</sub> for constant H <sub>C</sub> and M versus V <sub>C</sub> for constant V <sub>ts</sub> also M <sub>ic</sub> versus V <sub>ic</sub> for constant H <sub>ic</sub>	139
CHART 8.61	$^{\lambda}$ H <sub>ic</sub> / $^{\lambda}$ SL versus H <sub>ic</sub> for constant V <sub>ic</sub>	203
CHART 8.62	$\lambda/\lambda_{Hic}$ versus $H_{ic}$ for constant $t_{at}$	204
CHART 8.63	F1(Hic, Vic) versus Vic for constant Hic	205
CHART 8.7	$H_{ic}$ versus $\Delta P_{p} / \Delta H_{pc}$ for constant $\Delta H_{pc}$	207
CHART 8.8	$H_{ic}$ versus $\Delta P_p / \Delta H_{pc}$ for constant $\Delta P_p$	211
CHART 8.9	$V_{ic}$ versus $\Delta P_p / \Delta V_{pc}$ for constant $\Delta V_{pc}$	214
CHART 8.10	$V_{ic}$ versus $\Delta P_{p} / \Delta V_{pc}$ for constant $\Delta P_{p}$	216
CHART 8.11	$\Delta V_{pc}$ versus $V_{ic}$ for constant $\Delta P_{p}/q_{cic}$	218
CHART 8.12	$\Delta H_{pc} / \Delta V_{pc}$ versus $V_{ic}$ for constant $H_{ic}$	222
CHART 8.13	$\begin{array}{c} \Delta V_{pc}  \text{versus}  \Delta P_{p}  \text{for constant}  V_{ic} \\ \text{and}  \Delta H_{pc}  \text{versus}  \Delta P_{p}  \text{for constant}  H_{ic} \\ \text{also}  \Delta V_{pc}  \text{versus}  \Delta H_{pc}  \text{for constant}  V_{ic}  \text{and}  H_{ic} \end{array}$	228
CHART 8.14	$M_{ic}$ versus $\Delta M_{pc} / \Delta H_{pc}$ for constant $H_{ic}$	235
CHART 8.15	$\Delta M_{pc}$ versus $\Delta P_p/P_s$ for constant $M_{ic}$ and $\Delta H_{pc}$ versus $\Delta P_p/P_s$ for constant $H_{ic}$ also $\Delta M_{pc}$ versus $\Delta H_{pc}$ for constant $M_{ic}$ and $H_{ic}$	
CHART 8.16		240
CHART 8.17	$\Delta M_{pc} / \Delta V_{pc}$ versus $M_{ic}$ for constant $H_{ic}$ $M_{ic}$ versus $\Delta M_{pc} / (\Delta P_{p} / q_{cic})$	251
CHART 8.18	$\Delta P_{\rm p}/q_{\rm cic}  \text{versus}  \Delta M_{\rm pc}  \text{for constant}  M_{\rm ic}$	256
	9.3 United States Standard Atmosphere	260
TABLE 9.4		265
	Mach Number for various values of a <sub>c</sub> /P <sub>a</sub> a <sub>c</sub> /P <sub>a</sub> ≤ 0.893 (M ≤ 1.0)	309
TABLE 9.5	Mach Number for various values of $q_c/P_a$ for $M \ge 1.0$	313
TABLE 9.6	Differential Pressure, q <sub>c</sub> for various values of V <sub>c</sub>	321
TABLE 9.7	Centigrade - Fahrenheit Conversion Table	337

CHAPTER TWO			Page
CHART	2.31	Manifold Pressure Correction. (For change in Supercharger Inlet Air Temperature) P <sub>2</sub> /P <sub>1</sub> , CAT	2-31
CHART	2.32	Manifold Pressure Correction (For change in Carburetor Inlet Air Temperature) P <sub>2</sub> /P <sub>1</sub> , CAT	2-32
CHART	2.33	Ram Pressure Ratio - Ram Efficiency - Mach Number M, P <sub>t2</sub> /P <sub>a</sub> , η <sub>R</sub>	2-33

# CHAPTER THREE

CHART 9.1	Relation between Total Pressure Recovery and Ram Efficiency	3-68
CHART 9.2	Total Pressure Recovery for Inlets with Sharp Lips	3-69
CHART 9.3	Turbulent Boundary Layer Thickness for Flat-Plates at Zero Angle of Attack as a Function of Flight Speed and Altitude	3-70
CHART 9.4	Pressure Recovery of Boundary Layer Air Admitted into Side-Inlet Installation - Turbulent Flow	3.71
CHART 9.5	Total Momentum Ratio for Various Scoop Height to Boundary Layer Ratios	3-73
CHART 9.6	Total Pressure Recovery for Straight Subsonic Diffusers	3-74
CHART 9.7	Total Pressure Loss in Compound Subsonic Diffuser Bends	3-76
CHART 9.8	M <sub>o</sub> versus Pt1/Pto - Normal Shock Conditions	3-77
CHART 9.9	Total Pressure Ratios for 2-Dimensional 2-Shock Compression	3-78
CHART 9.10	Total Pressure Ratios for 2-Dimensional 3-Shock Compression	3-79
CHART 9.11	Total Pressure Ratius for Conical 2-Shock Compression	<b>3-8</b> 0
CHART 9.12	Mach Number Change Through an Oblique Shock for a Two-Dimensional Wedge	3-81
CHART 9.13	Total Pressure Ratio Across an Oblique Shock for a Two-Dimensional Wedge	3-82
CHART 9.14	Theoretical Additive-Drag Coefficients for Open-Nosed Inlets	3-83
CHART 9.15	Theoretical Additive-Drag Coefficients for Annular Nose Inlets with Conical Flow at the Inlet	3-84
CHART 9.16	Change in Cowl Drag Coefficient with a Change in Mass Flow Ratio as a Function of Mucn Number	3-85
CHART 9.17	Idealized Gas Flow with Subcritical Operation	3-86
CHART 9.18	Idealized Gas Flow with Supercritical Operation - Convergi <sub>ng</sub> Nozzle	3-87
CHART 9.19	Gross Thrust Parameter versus Nozzle Pressure Ratio with Subcritical Operation	<b>3-9</b> 0
CHART 9.20	Gross Thrust Parameter versus Nozzle Pressure Ratio with Supercritical Operation - Converging Nozzle.	3-91

CHART 9.21	1 Gross Thrust Parameter versus Nozzle Pressure Ratio with Supercritical Operation Converging - Diverging Nozzle	
CHART 9.22	P'/P versus Mach Number (Rayleigh Supersonic Pitot Formula)	3-97

# CHAPTER FOUR

CHART 4.41	Power Correction for Weight Change W <sub>t</sub> , △W, b, M, H <sub>c</sub>	4-49
CHART 4.71	Natural Log of Initial to Final Gross Weight Ratio for Range and Endurance Computations $W_1/W_2$ , $\log_e W_1/W_2$	4-55
CHART 4.72	Gross Weight Factor for Range and Endurance Computations W, 2/VW	4-56

# CHAPTER FIVE

CHART 5.21	Rate of Climb Power Correction for Temperature Variation $MP_t/P_a, \sqrt{T_{at}/T_{as}}, W_t$	5-35
CHART 5.22	Turbojet Rate of Climb Power Correction Factor H <sub>c</sub> , M, W <sub>t</sub>	5-36
CHART 5.23	Turbojet Rate of Climb Power Correction Factor W/δ <sub>a</sub> , H <sub>c</sub>	5-38
CHART 5.31	Rate of Climb Induced Drag Correction Factor H <sub>c</sub> , M, b	5-40
CHART 5.41	Rate of Climb Vertical Wind Gradient Correction Factor $H_c$ , V, dV, /dh	5-42
CHART 5.51	Rate of Climb Acceleration Correction Factor H <sub>c</sub> , V <sub>c</sub>	5-43
CHART 5.52	Rate of Climb Acceleration Correction Factor $H_c$ , $V_c$ , $dV_c/dH_c$	5-44

...

## APPENDIX I

CHART I-1	Density Altitude Charts	8-5
CHART I-2	Differential-Static Pressure Ratio versus Mach Number (Supersonic-Normal Shock Condition)	8-17
CHART I-3	Reynolds Number-Mach Number Ratio versus Pressure Altitude	8-18
CHART I-4	Supersonic (Isentropic) Mach Number Functions $M$ versus $P_t/P_s$ M versus $q/P$	8-20
	M versus q/P <sub>s</sub>	8-21
	M versus q'Pt	8-22
	M versus ρ <sub>t</sub> /ρ <sub>s</sub>	8-23
	M versus $T_t/T_s$	8-24
CHART I-5	Supersonic (Normal Shock Conditions) Mach Number Functions	
	M versus $P_t^{\prime}/P_{m{s}}$	8-25
	M versus P'/P s s	8-26
	M versus Pt'/Pt	<b>B-</b> 27
	M versus $\rho_s'/\rho_s$	8-28
	M versus $T_{s}^{i}/T_{s}$	8-29
	M versus M' M versus V'/V M versus a'/a	8-30 8-31 8-32
CHART I-6	Psychrometric Chart	8-33

## APPENDIX II

CHART II-1	Nomenclature	8-35
CHART II-2	Physical Information and Systems of Units	8-46
CHART II-3	Conversion Tables	8-47

#### INTRODUCTION

No single or rigid method of data analysis and presentation has been set down in this report. Rather, an attempt has been made to show various methods of data standardization and plotting. The flight testing agency can best determine the procedures most suited to the particular test, type of aircraft, or type of report desired.

Considerable detail concerning the derivation of correction factors and performance parameters has been included. The function of these derivations is not to prove the results, but to show the methods. When performance analysis problems result from new type of aircraft, engines, or flight conditions, these methods of deriving corrections and parameters may be useful as a starting point.

Aircraft stability and control tests and methods are not included, but will be the subject of a separate report.

Although extreme care was taken in the preparation of this report, there is a possibility that errors are present. Please address correspondence to,

COMMANDER

AIR FORCE FLIGHT TEST CENTER EDWARDS AIR FORCE BASE EDWARDS, CALIFORNIA ATTN: Flight Research Division, FTTER

# SYMBOLS USED IN CHAPTER ONE

Term	Definition	Units
a	Speed of sound, 38.967 $\sqrt{T_a(^{\circ}K)}$	knots
a <sub>s</sub>	Standard day speed of sound, 38.967 $\sqrt{T_{as}}(^{\circ}K)$	knots
a <sub>t</sub>	Test day speed of sound, 38.967 $\sqrt{T_{at}(^{\circ}K)}$	knots
asl	Speed of sound at standard sea level, 661.48	knots
°C	Degrees centigrade	
CL	Airplane lift coefficient, $nW/(\rho V_t^2 S/2)$	
C <sub>Lic</sub>	"Indicated" lift coefficient, nW/\$P <sub>B</sub> Mic <sup>2</sup> S/2)	
đ	Differential Example: dH <sub>ic</sub> = differential indicated pressure altitude corrected for instrumer error	ıt
d/dt	Time rate of change Example: dH <sub>ic</sub> /dt = time rate of change indicated pressure altitude corrected for instrument error	of
fn	Function of ( ) Example: $P_8 = f_2(H_{ic})$ . This means that is a function of $H_{ic}$ . In other words, $P_8$ be determined if $H_{ic}$ is known.	
g	Acceleration due to gravity at a point	feet/second <sup>2</sup>
<sup>g</sup> sl	Acceleration due to gravity at standard sea level	32.17405 feet/ second <sup>2</sup>
G	Gravitational constant	32.17405 feet <sup>2</sup> / second <sup>2</sup> ~ geopotential feet
h	Tapeline altitude	feet
H	Geopotential at a point (this is a measure of the gravitational potential energy of a unit mass at this point relative to mean sea level)	geopotential feet
н <sub>с</sub>	Pressure altitude, $H_i + \Delta H_{ic} + \Delta H_{ic} + \Delta H_{pc}$	feet
"Hg	Inches of mercury	
Hi	Indicated pressure altitude	feet
Hic	In jicated pressure altitude corrected for instrument error, $H_i + \Delta H_{ic}$	feet

ΔH <sub>ic</sub>	Altimeter instrument correction	feet
H <sub>icl</sub>	Indicated pressure altitude corrected for instrument and lag errors, $H_i + \Delta H_{ic} + \Delta H_{ic\hat{k}}$	feet
$\Delta H_{icl}$	Altimeter lag correction	feet
∆Hp	Altimeter position error corresponding to $\Delta P_p$	feet
$\Delta H_{pc}$	Altimeter position error correction	feet
Kn	A constant Example: K <sub>5</sub> = 52.86784	
К	Temperature probe recovery factor	
°K	Degrees Kelvin	
m	The slope of a line at a point	
М	Flight or free stream Mach number	
M <sub>i</sub>	Indicated Mach number	
M <sub>ic</sub>	Indicated Mach number corrected for instrume error, $M_i + \Delta M_{ic}$	ent
$\Delta M_{ic}$	Machmeter instrument correction	
∆Mp	Machmeter position error corresponding to $\Delta R$	p
∆M <sub>pc</sub>	Machmeter position error correction	
n	Load factor	
Npr	Prandtl number, M/pd where d is the therma diffusivity	1
NR	Reynolds number, $\rho L V / \mu$ where L is a characteristic length and V is axial velocity	
P	The applied pressure at time t	''Hg
Pa	Atmospheric pressure corresponding to $H_c$	"Hg
P <sub>aSL</sub>	Atmospheric pressure at standard sea level	29.92126"Hg
Pi	The indicated pressure at time t	''Hg
$\Delta P_{p}$	Static pressure error (or position error)	''Hg
Ps	Pressure corresponding to H <sub>ic</sub>	"Hg
Pt	Free stream total pressure	"Hg

P <sub>t</sub> '	Total pressure at total pressure source (for subsonic speeds, $P_t'$ is equal to the free stream total pressure, $P_t$ . For supersonic speeds, $P_t'$ is equal to the total pressure behind the shock which forms in front of the probe and is therefore not equal to $P_t$ ).	''Hg
<b>9</b> .	Dynamic pressure, $q = \rho V_t^2/2 = 0.7 P_a M^2$	''Hg
q <sub>c</sub>	Differential pressure, P <sub>t</sub> ' - P <sub>a</sub> (q <sub>c</sub> is also called impact pressure or compressible dynamic pressure)	''Hg
<b>9</b> cic	Differential pressure corresponding to $V_{ic}$ , $P_t' - P_s$	''Hg
r	Radius of the earth	feet
R	Gas constant for dry air	3089.67 feet <sup>2</sup> / •K second <sup>2</sup>
S	Total wing area	feet <sup>2</sup>
t	Time	seconds
ta	Atmospheric temperature	°C
tas	Standard day atmospheric temperature corresponding to H <sub>c</sub>	°C
taSL	Standard sea level atmospheri, iemperature	15°C
tat	Test day atmospheric temperature	°C
ti	Indicated temperature	°C
t <sub>ic</sub>	Indicated temperature corrected for instrument error, $t_i + \Delta t_{ic}$	۰C
$\Delta t_{ic}$	Air temperature instrument correction	°C
Ta	Atmospheric temperature	°K
Tas	Standard day atmospheric temperature corresponding to H <sub>C</sub>	۰ĸ
TaSL	Standard sea level atmospheric temperature	288.16°K
Tat	Test day atmospheric temperature	°K
Τ <sub>i</sub>	Indicated Temperature	۰K
T <sub>ic</sub>	Indicated temperature corrected for instrument error, $T_i + \Delta T_{ic}$	°K

xvii

ΔT <sub>ic</sub>	Air temperature instrument correction	•K
Ŧ		۰K
Tt	Total temperature	N
v <sub>c</sub>	Calibrated airspeed, $V_i + \Delta V_{ic} + \Delta V_{ic \ell} + \Delta V_{pc}$	knots
V <sub>e</sub>	Equivalent airspeed, $V_{c} + \Delta V_{c}$ or $V_{t} \sqrt{\sigma}$	knots
Vi	Indicated airspeed	knots
Vic	Indicated airspeed corrected for instrument error, $V_i + \Delta V_{ic}$	knots
$\Delta V_{ic}$	Airspeed indicator instrument correction	knots
Vicl	Indicated airspeed corrected for instrument and lag errors, $V_i + \Delta V_{ic} + \Delta V_{ic} \ell$	knots
$\Delta V_{icl}$	Airspeed indicator lag correction	knots
$\Delta V_p$	Airspeed indicator position error corresponding to $\Delta P_p$	knots
∆V <sub>pc</sub>	Airspeed indicator position error correction	knots
V <sub>t</sub>	True airspeed	knots
V <sub>ts</sub>	Standard day true airspeed	knots
V <sub>tt</sub>	Test day true airspeed	knots
w	Aircraft gross weight	pounds
×	Angle of attack	
β	Angle of sideslip	
r	Ratio of specific heats, 1.40 for air	
8	$P_a/P_{aSL}$	
8 ic	P <sub>s</sub> /P <sub>aSL</sub>	
θ	$T_a/T_{aSL}$	
θs	Tas/TaSL	
θt	$T_{at}/T_{aSL}$	
λ	Lag constant	seconds
$\lambda_{\rm Hic}$	Lag constant corresponding to H <sub>ic</sub>	seconds
$\lambda_{s}$	Static pressure lag constant	seconds
$\lambda_{SL}$	Lag constant at standard sea level	seconds
λ <sub>sSL</sub>	Static pressure lag constant at standard sea level	seconds

-----

×.,

-:=

xviii



λ <sub>t</sub>	Total pressure lag constant	seconds
<sup>λ</sup> tSL	Total pressure lag constant at standard sea level	seconds
μ	Viscosity at temperature Ta	pounds - second/feet <sup>2</sup>
Hic	Viscosity corresponding to H <sub>ic</sub>	pounds- second/feet <sup>2</sup>
JL SL	Viscosity at standard sea level	3.7452 x 10 <sup>-7</sup> pounds-second/ feet <sup>2</sup>
P	Air density	slugs/feet <sup>3</sup>
ρ <sub>s</sub>	Standard day air density corresponding to H <sub>C</sub>	slugs/feet <sup>3</sup>
PSL	Air density at standard sea level	0.0023769 slugs/feet <sup>3</sup>
Pt	Test day air density	slugs/feet <sup>3</sup>
σ	p/psl	
σ <sub>s</sub>	Ps/PSL	
σ <sub>t</sub>	Pt'PSL	
τ	Acoustic lag	seconds

#### CHAPTER ONE

## SEGTION 1

## THE STANDARD ATMOSPHERE

The performance of an aircraft is influenced by the pressure and temperature of the air through which the aircraft is flying. Studies of the earth's atmosphere have shown that these quantities depend primarily on altitude, and vary relatively little from day to day. Consequently, a "standard" atmosphere can be usefully established by definition of a pressure and temperature for each altitude. This standard will approximate the atmospheric conditions for any day fairly closely. By applying small corrections to data acquired on a non-standard day, the data may be reduced to the standard day. This makes possible comparison of results obtained on other days with the same aircraft and with other aircraft.

## 1.1 THE UNITED STATES STANDARD ATMOSPHERE

For many years a standard atmosphere based on NACA Report No. 218, "Standard Atmosphere - Tables and Data," has been used in the United States. Recently many organizations including the Air Research and Development Command of the United States Air Force have adopted a new standard, the United States Standard Atmosphere, which is consistent with that of the International Civil Aviation Organization. This new standard atmosphere is discussed in NACA Report No. 1235, "Standard Atmosphere - Tables and Data for Altitudes to 65,800 Feet." All charts and tables in this manual are based on the US Standard Atmosphere.

1.1.1 Basic Assumptions:

The United States Standard Atmosphere is derived from the following assumptions which closely approximate true atmospheric conditions:

(1) The air is dry

(2) The atmosphere is a perfect diatomic gas:

$$\mathbf{P}_{\mathbf{a}} = \int \mathbf{R} \mathbf{T}_{\mathbf{a}}$$
 1.1

In specific units

$$\rho = 0.022891 \frac{P_a}{T_a}$$
 1.2

$$\sigma = 9.6306 \frac{P_a}{T_a}$$
 1.3

where

P<sub>a</sub> = atmospheric pressure, "Hg T<sub>a</sub> = atmospheric temperature, °K
ρ = atmospheric density, slugs/ft<sup>3</sup>
R = the gas constant for dry air, 3089.7 ft<sup>2</sup>/sec<sup>2</sup> °K
σ- = density ratio, ρ/ρ<sub>SL</sub>
(3) Hydrostatic equilibrium exists:

$$dP_a = -\rho g dh 1.4$$

This equation is derived from a consideration of the forces acting in the vertical direction on a small column of air of unit area. (See Figure 1.1)





Forces Acting on a Small Column of Air of Unit Area

(4) The measure of vertical displacement is geopotential. Geopotential is a measure of the gravitational potential energy of a unit mass at a point relative to mean sea level. It is defined in differential form by the equation

2

#### G dH = g dh

where

- h = tapeline altitude; i.e., the actual distance from mean sea level to a point in the atmosphere, feet
- g = acceleration due to gravity at the same point, feet/sec<sup>2</sup>
- H = geopotential at the point, geopotential feet
- $G = a \text{ dimensional constant, } 32.17405 \text{ ft}^2/(\sec^2 \text{geopotential feet})$  for the above system of units

Each point in the atmosphere has a definite geopotential as "g" is a function of latitude and altitude.

- (5) Sea level pressure is 760 mm Hg or 29.92126 inches Hg
- (6) Sea level temperature is 15°C or 288.16°K

(7) Temperature variation with geopotential is expressed as a series of straight line segments:

- (a) The temperature lapse rate in the troposphere (sea level to 36,089 geopotential feet) is 0.00198120°C/geopotential feet.
- (b) The temperature above 36,089 geopotential feet and below 82,021 geopotential feet is constant -56.50°C. (The latest issue of "The ARDC Model Atmosphere" should be consulted for data above 82,021 geopotential feet.)

#### 1.1.2 Relationship Between Variables:

From the basic assumptions listed above it is possible to express the atmospheric pressure, temperature, and density as functions of geopotential.

Introducing the definition of geopotential (Equation 1.5) into the equilibrium equation 1.4,

$$dP_a = -\beta G dH \qquad 1.6$$

Eliminating  $\rho$  by means of the perfect gas equation 1.1,

$$\frac{dP_a}{P_a} = -\frac{G}{R} \frac{dH}{T_a}$$
 1.7

Assumption(7) above expresses

 $T_a = f_1(H)$  only.

1.5

Hence, integration of equation 1.7 is possible with the result

 $P_a = f_2(H)$  only.

Finally, from the perfect gas equation,

$$\rho = f_3$$
 (H) only.

For geopotentials below 36,089 geopotential feet

$$\theta = \frac{T_a}{T_a SL} = (1 - K_1 H)$$
 1.8

$$\delta = \frac{P_a}{P_{aSL}} = (1 - K_1 H)^{5.2561}$$
 1.9

$$\sigma = \frac{\rho}{\rho_{SL}} = (1 - K_1 H)^{4.2561}$$
 1.10

where

$$K_1 = 6.87535 \times 10^{-6}$$
/geopotential feet

For geopotentials above 36,089 geopotential feet and below 82,021 geopotential feet

$$T_a = -56.50$$
°C = 216.66°K 1.11

$$\delta = \frac{P_a}{P_{aSL}} = 0.223358 e^{-K_2(H - K_3)}$$
 1.12

$$\tau = \frac{\rho}{\rho SL} = 0.29707 e^{-K_2(H - K_3)}$$
 1.13

where

 $K_2 = 4.80634 \times 10^{-5}$ /geopotential feet  $K_3 = 36,089.24$  geopotential feet

From the above equations, pressure, temperature, and density, plus several other parameters useful in flight test are tabulated in Table 9.2 for incremental geopotentials of 100 geopotential feet. In addition,  $P_a$  in inches Hg and  $\delta$  are tabulated in Table 9.3 for every 10 geopotential feet. A summary of basic data is given in Table 9.1.

1.1.3 Determination of Tapeline Altitude:

In flight testing, the exact position in space is usually not important; altitude is important only as a means of describing the properties of the air through which the test aircraft is flying. Hence, it is seldom necessary to determine tapeline altitude. It is sufficient to express the atmospheric properties in terms of geopotential.

If one finds it necessary to determine the tapeline altitude, the acceleration of gravity as a function of tapeline altitude must be defined to allow integration of equation 1.5. An approximate expression is obtained by assuming that the altitude variation of the acceleration of gravity from its sea level value is given by the Newtonian inverse square law\*

$$g = g_{SL} \left(\frac{r}{r+h}\right)^2 \qquad 1.14$$

where

 $g_{SL}$  = the sea level value of the acceleration of gravity, 32.17405 ft/sec<sup>2</sup>

- r = radius of the earth, 20,930,000 feet
- h = tapeline altitude, feet.

Introducing this expression into equation 1.5 and integrating yields

$$H = \frac{gSL}{G} \left(\frac{rh}{r+h}\right)$$
 1.15

where

 $G = 32.17405 \text{ ft}^2/\text{sec}^2$  - geopotential feet

Solving for h - H(G/gSL)

h - H 
$$\left(\frac{G}{g_{SL}}\right) = \frac{H^2 \left(\frac{G}{g_{SL}}\right)^2}{(r - H \frac{G}{g_{SL}})}$$
 1.16

where

 $G/g_{SL} = 1$  ft/geopotential feet

A plot of altitude correction factor,  $h - H(G/g_{SL})$ , versus  $H(G/g_{SL})$  is given in Chart 8.1. This factor, when added to the geopotential,

<sup>\*</sup>Use of the Newtonian into rise square law is based on the assumption that the earth is a nonrotating sphere composed of spherical shells of equal density. This assumption is very good at altitudes attained in routine flight test work (H  $\leq$  100,000 geopotential feet). For higher altitudes, a more sophisticated analysis may be necessary. A method which is good to several million feet is given in AFCRC TN-56-204, "The ARDC Model Atmosphere, 1956," by R.A. Minzner and W. S. Ripley.

 $H(G/g_{SL})$ , will give the corresponding tapeline altitude.

## 1.2 THE NON-STANDARD ATMOSPHERE

Flight test data is always reduced to a standard day so that comparison may be made among data obtained on different days. The usual technique is to present the data in terms of pressure altitude. (Pressure altitude is defined as the geopotential at which a given pressure is found in the standard atmosphere.) Whether found in a standard atmosphere or nonstandard atmosphere, any given pressure indicates one and only one corresponding pressure altitude. Therefore, reduction to a standard day consists of making corrections for temperature to the value given in the standard atmosphere corresponding to the test day pressure altitude (or pressure).

The pressure altitude and geopotential are not simply related on a nonstandard day. If the geopotential is desired, it is necessary to make a survey of the atmosphere to determine  $T_a$  as a function of  $P_a$  to allow integration of equation 1.7. Fortunately, this operation is seldom required. However, the computation is outlined in NACA Report No. 538, "Altitude - Pressure Tables Based on the United States Standard Atmosphere".

#### SECTION 2

# THEORY OF ALTITUDE, AIRSPEED, MACH NUMBER AND AIR TEMPERATURE MEASUREMENT

Pressure altitude, airspeed, Mach number and free air temperature are basic parameters in the performance of aircraft. The instruments used to measure these quantities are the altimeter, the airspeed indicator, the machmeter, and the free air temperature probe. The basic theory of the construction and calibration of these instruments is given in this section. The actual methods employed in their calibration will be given in subsequent sections.

#### 2.1 THE ALTIMETER

Most altitude measurements are made with a sensitive absolute pressure gage, called an altimeter, scaled so that a pressure decrease indicates an altitude increase in accordance with the U.S. Standard Atmosphere. If the altimeter setting\* is 29.92, the altimeter will read pressure altitude whether in a standard or non-standard atmosphere.

$$\frac{P_a}{P_{aSL}} = (1 - 6.87535 \times 10^{-6} H_c)^{5.2561}$$
  
for H<sub>c</sub> < 36,089 ft 2.1

$$\frac{P_a}{P_{aSL}} = 0.223358 e^{-4.80634 \times 10^{-5} (H_c - 36,089.24)}$$
  
for 36,089 < H\_c < 82,021 ft 2.2

where

P<sub>a</sub> = atmospheric pressure, inches Hg
H<sub>c</sub> = pressure altitude, feet
P<sub>aSL</sub> = 29.92126 inches Hg

7

<sup>\*</sup>The altimeter setting is an adjustment that allows the scale to be moved so that the altimeter can be made to read field elevation when the aircraft touches the ground. In flight testing, the altimeter setting should be 29.92 in order that the altimeter reading will be pressure altitude.

The altimeter is constructed and calibrated according to this relationship.

The heart of the altimeter is an evacuated metal bellows which expands or contracts with changes in outside pressure. The bellows is connected to a series of gears and levers which cause a pointer to move as the bellows expands or contracts. The whole mechanism is placed in an airtight case which is vented to a static pressure source; the indicator then reads the pressure supplied to the case. The dial is calibrated to indicate pressure altitude. The altimeter construction is shown in Figure 2.1



Altimeter Schematic

The static pressure measured at the static source of the altimeter  $(P_{a})$  may differ slightly from the atmospheric pressure  $(P_{a})$ . For any

8

 $P_s$ , the altimeter, when corrected for instrument error,<sup>\*</sup> will indicate the corresponding indicated pressure altitude corrected for instrument error  $(H_{ic})$ .

$$\frac{P_s}{P_{aSL}} = (1 - 6.87535 \times 10^{-6} H_{ic})$$
  
for  $H_{ic} < 36,089$  ft 2.3

$$\frac{P_s}{P_{aSL}} = 0.223358_e^{-4.80634 \times 10^{-5}(H_{ic} - 36,089.24)}$$
  
for 36,089 <  $H_{ic} < 82,021$  2.4

The quantity  $P_s - P_a$  is called the static pressure error or position error. The value which is added to  $H_{ic}$  to determine  $H_c$  is termed the altimeter position error correction. The position error corrections for the altimeter and the other instruments will be considered in later sections.

The altimeters available and their expected characteristics are:

Type	Range - ft	Readability - ft	Repeatability
C-12	0 to 50,000	5	Determined by calibration
C-19	0 to 80,000	5	

#### 2.2 THE AIRSPEED INDICATOR

True airspeed  $(V_t)$  is the velocity of an aircraft with respect to the air through which it is flying. It is difficult to measure  $V_t$  directly. Instead, it is usually determined from calibrated airspeed  $(V_c)$ , atmospheric pressure  $(P_a)$ , and atmospheric temperature  $(T_a)$ .  $V_c$  is obtained from a conventional airspeed indicator,  $P_a$  is measured with an altimeter, and  $T_a$  is measured with a free air temperature probe.

<sup>\*</sup>The instrument error is an error built into the instrument consisting of such things as scale error and hysteresis. This error is discussed in Section 3.

The airspeed indicator operates on the principle of Bernoulli's compressible equation for frictionless adiabatic (isentropic) flow in which airspeed is expressed as the difference between **tot**al and static pressures. Therefore, the airspeed indicator consists of a pitot-static pressure system which is used to measure the difference between total and static pressures.

At subsonic speeds Bernoulli's equation expressed as follows is applicable

$$\frac{q_{c}}{P_{a}} = \left[1 + \frac{r-1}{2} \left(\frac{V_{1}}{a}\right)^{2}\right] \frac{r}{r-1} - 1 \qquad 2.5$$

where:

q<sub>c</sub> = P<sub>t</sub> · P<sub>a</sub> = differential pressure. (This is equal to the free stream impact pressure or compressible dynamic pressure (P<sub>t</sub> - P<sub>a</sub>) for subsonic flow )
 P<sub>t</sub> = free stream total pressure
 P<sub>a</sub> = free stream static pressure (or atmospheric pressure)
 # = ratio of specific heats
 V<sub>t</sub> = true airspeed
 a = local atmospheric speed of sound

For air, f = 1.40. Equation 2.5 becomes

$$\frac{q_c}{P_a} = \left[1 + 0.2 \left(\frac{V_t}{a}\right)^2\right]^{3.5} - 1 \qquad 2.6$$

For supersonic flight, a shock wave will form in front of the total pressure probe. Therefore equation (2.5, 2.6) is no longer valid. The solution for supersonic flight is derived by considering a normal shock compression in front of the total pressure tube and an isentropic compression in the subsonic region aft of the shock. The normal shock assumption is good as the pitot tube has a small frontal area so that the radius of the shock in front of the hole may be considered infinite. The resulting equation, known as the Rayleigh supersonic pitot equation, relates the total pressure behind the shock to the free stream ambient pressure.

$$\frac{q_c}{P_a} = \left[\frac{t+1}{2}\left(\frac{V_t}{a}\right)^2\right] \xrightarrow{\frac{v}{v-1}} \left[\frac{v+1}{1-v+2v\left(\frac{V_t}{a}\right)^2}\right] \xrightarrow{\frac{1}{v-1}} - 1$$
2.7

where

- $q_c = P_t' P_a = differential pressure.$  (This is not equal to the free stream impact pressure or compressible dynamic pressure,  $P_t - P_a$ , for supersonic flow as  $P_t' \neq P_t$ .)
- Pt' = total pressure at total pressure pickup behind the shock. (This is not equal to the free stream total pressure (Pt) for supersonic flight.)

For air, x = 1.40. Equation 2.7 becomes

$$\frac{q_{c}}{P_{a}} = \frac{7\left(\frac{V_{t}}{a}\right)^{2} - 1}{6} \left\{ \frac{36\left(\frac{V_{t}}{a}\right)^{2}}{5\left[7\left(\frac{V_{t}}{a}\right)^{2} - 1\right]} \right\}^{3.5} - 1 \qquad 2.8$$

This may be written more conveniently in the form

$$\frac{q_{c}}{P_{a}} = \frac{K_{3} \left(\frac{V_{t}}{a}\right)^{2}}{\left[7 \left(\frac{V_{t}}{a}\right)^{2} - 1\right]^{2.5}} - 1$$
2.9

where

 $K_3 = \frac{(7.2)^{3.5}}{6} = 166.921$ 

Examination of equations (2.5, 2.6) and (2.7, 2.8, 2.9) shows that the true velocity  $(V_t)$  is dependent on the local atmospheric properties, speed of sound (a), and static pressure  $(P_a)$ , as well as the differential pressure  $(q_c)$ . Therefore, an airspeed indicator measuring differential pressure can be made to read true airspeed at one and only one above precise condition. Standard sea level is taken for this condition. Therefore, the dial of the airspeed indicator is scaled so that a given differential pressure will indicate a speed in accordance with equations 2.6 and (2.8, 2.9) in which sea level standard a and  $P_a$  are inserted. This sea level standard value of  $V_t$  is defined as calibrated airspeed  $(V_c)$ .

$$\frac{q_{c}}{P_{aSL}} = \left[ 1 + 0.2 \left( \frac{V_{c}}{a_{SL}} \right) \right]^{3.5} - 1 \qquad 2.10$$

for  $V_c \leq a_{SL}$ , and

$$\frac{q_{c}}{P_{aSL}} = \frac{166.921 \left(\frac{V_{c}}{a_{SL}}\right)}{\left[\frac{7 \left(\frac{V_{c}}{a_{SL}}\right)^{2} - 1}{2.5}\right]^{2} \cdot 5} - 1 \qquad 2.11$$

for  $V_c \ge a_{SL}$ .

where

q<sub>c</sub> = differential pressure, inches Hg
 V<sub>c</sub> = calibrated airspeed, knots
 a<sub>SL</sub> = 661.48 knots
 P<sub>aSL</sub> = 29.92126 inches Hg

Airspeed indicators are constructed and calibrated according to these equations.

In operation, the airspeed indicator is similar to the altimeter, but, instead of being evacuated, the inside of the capsule is connected to a total pressure source and the case to the static pressure source. The instrument then senses total pressure  $(P_t)$  within the capsule and static pressure  $(P_s)$  outside it as shown in Figure 2.2.



Figure 2.2 Airspeed Indicator Schematic

For any indicated differential pressure  $(q_{cic})$  felt by the instrument, the airspeed indicator, when corrected for instrument error, will indicate the corresponding indicated airspeed corrected for instrument error  $(V_{ic})$ , or

$$\frac{q_{cic}}{P_{a}_{SL}} = \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{3.5} - 1 \qquad 2.12$$

for  $V_{ic} \leq a_{SL}$  and  $\frac{q_{cic}}{P_{aSL}} = \frac{\frac{166.921 \left(\frac{V_{ic}}{a_{SL}}\right)^{7}}{\left[7 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]} 2.5 - 1$ 2.13
for  $V_{ic} \ge a_{SL}$ .

In the general case,  $q_{cic}$  will differ from  $q_c$  as a result of static pressure error. As a result, an airspeed position error correction must be added to  $V_{ic}$  to obtain  $V_c$ , the desired result. This correction will be discussed in later sections.

 $q_{cic}$  in inches Hg is given for various values of  $V_{ic}$  in knots in Table 9.5. This table is also good as  $q_c$  in inches Hg versus  $V_c$  in knots.

At the present time, the following airspeed indicators are commonly used in flight test work.

Type	Range	Readability	Repeatability
F - 1	50 to 650 knots	0.5 knots	Determined by calibration
059	50 to 850 knots	0,5 knots	
0153	l0 to 150 miles per hour	0.5 miles per hour	

Calibrated airspeed  $(V_c)$  represents the true velocity of the aircraft  $(V_t)$  at standard sea level conditions only.  $V_t$  may be determined at altitude by a knowledge of atmospheric pressure and density (or temperature).

The equivalent airspeed  $(V_e)$  is defined as

$$V_e = V_t \int e^{-2.14}$$

where  $\sigma$  is the density ratio,  $\rho/\rho_{SL}$ . Solving the subsonic equation 2.6 for  $V_t^2$ .

$$V_{t}^{2} = 5a^{2} \left[ \left( \frac{q_{c}}{P_{a}} + 1 \right)^{2/7} - 1 \right]$$
2.15

The speed of sound in a perfect gas may be expressed as

$$a = \sqrt{\frac{f P_a}{\rho}}$$
 2.16

Introducing 
$$V_e^2/\sigma$$
 for  $V_e^2$  and replacing  $a^c\sigma$  by  $\gamma P_a/S_{SL}$ :  

$$V_e = \sqrt{\frac{7P_a}{f_{SL}} \left[ \left( \frac{q_c}{P_a} + 1 \right) - 1 \right]} \qquad 2.17$$

Introducing equation 2.10, the following result is obtained:

$$\frac{V_{a}}{V_{c}} = \left\{ \begin{bmatrix} \frac{q_{c}}{P_{a}} + 1 \\ \frac{q_{c}}{P_{a}} + 1 \\ \frac{q_{c}}{P_{a}} + 1 \\ \frac{q_{c}}{P_{a}} + 1 \end{bmatrix} \begin{bmatrix} \frac{P_{a}}{P_{a}} \\ \frac{P_{a}}{P_{a}} \end{bmatrix} \right\}^{\frac{1}{2}}$$
2.18

From equation 2.14,  $V_t$  is simply  $V_e$  corrected for the difference between sea level standard density and actual ambient density. This has been shown for subsonic flight only. It could similarly be shown to be true for supersonic flight as well.

This relationship between  $V_c$  and  $V_t$  is presented for explanation only; a shorter method of obtaining  $V_t$  from the same required variables is given later in Section 2.5.

### 2.3 MACH NUMBER AND THE MACHMETER

#### 2.3.1 Mach Number:

Mach number (M) is defined as the ratio of the true airspeed to the local atmospheric speed of sound

$$M = \frac{V_t}{a}$$
 2.19

With the advent of high speed aircraft, Mach number has become a very important parameter in flight testing.

For isentropic flow of a perfect gas, Bernoulli's equation states

$$\frac{P_{t}}{P_{a}} = \left(1 + \frac{t-1}{2}M^{2}\right) \frac{t}{t-1} 2.20$$

where

 $P_t =$ free stream total pressure  $P_a =$ free stream static pressure  $\gamma =$ ratio of specific heats

For air,  $\gamma = 1.40$ . Equation 2.20 becomes

$$\frac{P_{t}}{P_{a}} = (1 + 0.2M^{2})^{3.5} \qquad 2.21$$

This equation which relates Mach number to the free stream total and static pressures is good for supersonic as well as subsonic flight. It

must be remembered however, that  $P_t$  rather than  $P_t$  is measured in supersonic flight.

## 2.3.2 The Machmeter:

The Machmeter equation for subsonic flight is formed by inserting the definition for M into equation 2.6.

$$\frac{q_c}{P_a} = (1 + 0.2M^2)^{3.5} - 1 \qquad 2.22$$

Solving for M

$$M = \sqrt{5 \left[ \left( \frac{q_c}{P_a} + 1 \right)^2 - 1 \right]}$$
 2.23

For supersonic flight, from equation 2.9

$$\frac{q_c}{P_a} = \frac{166.921M^7}{(7M^2 - 1)^{2.5}} - 1 \qquad 2.24$$

Equation 2.24 cannot be solved explicitly for M. It can, however, be put in the following form which is convenient for rapid iteration:

$$M = \sqrt{\frac{q_{c}}{P_{a}} + 1} \frac{1}{K_{4}} \left(1 - \frac{1}{7M^{2}}\right)^{\frac{5}{2}} 2.25$$

where

 $K_4 = 1.287560$ 

The machmeter is essentially a combination altimeter and airspeed indicator designed to solve these equations for Mach number. An altimeter capsule and an airspeed capsule simultaneously supply signals to a series of gears and levers to produce the Mach number indication. A machmeter schematic is given in Figure 2.3.



Figure 2.3 Machmeter Schematic

For any static pressure  $(P_g)$  and differential pressure  $(q_{cic} = P_t^* - P_g)$ felt by the instrument, the Mach meter, when corrected for instrument error, will indicate the corresponding indicated Mach number corrected for instrument error  $(M_{ic})$ , or

$$\frac{q_{cic}}{P_{g}} = (1 + 0.2M_{ic}^{2})^{3.5} - 1 \qquad 2.26$$

for  $M_{ic} \leq 1.00$ , and

$$\frac{q_{cic}}{P_{g}} = \frac{166.921 M_{ic}^{7}}{(7M_{ic}^{2} - 1)^{2.5}} - 1 \qquad 2.27$$

for  $M_{ic} \ge 1.00$ .

The true Mach number (M) is determined from  $M_{ic}$  and the Mach meter position error correction which is a result of the static pressure error,  $P_{g} - P_{a}$ .

These equations relating M to  $q_c/P_a$  and  $M_{ic}$  to  $q_{cic}/P_s$  are useful not only as machmeter equations, but as a means for relating calibrated airspeed and pressure altitude to Mach number.  $M_{ic}$  is given for values of  $q_{cic}/P_s$  for  $M_{ic} \leq 1.00$  in Table 9.4.  $q_{cic}/P_s$  is given for values of  $M_{ic}$  from 1,00 to 3.00 in Table 9.5. These tables can also be used to find M as a function of  $q_c/P_a$ ,

At present the accuracy of these meters is **poor** so that they are not suitable for precision work, but as flight-safety indicators only. The machmeters in general use are:

Туре	Range	Readability	Repeatability
Al	0.3 to 1.0; 0 to 50,000 feet	0.01	Determined by calibration
A2	0.5 to 1.5; 0 to 50,000 feet	0.01	
G09501	0.7 to 2.5; 0 to 60,000 feet	0.01	

#### 2.4 FREE AIR TEMPERATURE PROBE

The atmospheric temperature is a measurement of the internal thermal energy of the air. Therefore, it is a very important parameter in aircraft and engine performance. Unfortunately, it is difficult to measure accurately in flight. If the air surrounding the probe is brought to a complete stop adiabatically and the probe correctly senses the resulting temperature then

$$\frac{T_{ic}}{T_{a}} = \frac{T_{t}}{T_{a}} = 1 + \frac{\gamma - 1}{2} M^{2}$$
 2.28

where

- $T_{ic}$  = indicated temperature corrected for instrument error,  ${}^{\circ}K$  $T_t$  = total temperature,  ${}^{\circ}K$
- $T_a$  = free stream static temperature,  ${}^{o}K$
- M = free stream Mach number

For various reasons, such as radiation or heat leakage, most probes

do not register the full adiabatic temperature rise. It is, however, acceptable to write

$$\frac{T_{1C}}{T_{a}} = 1 + K \frac{r-1}{2} M^{2} \qquad 2.29$$

For air with  $\mathbf{a} = 1, 40$ , this becomes

$$\frac{T_{ic}}{T_{a}} = 1 + K \frac{M^{2}}{5}$$
 2.30

The value of K represents the percentage of the a labatic temperature rise detected by the probe and is called the probe recovery factor. For many installations it may be considered a constant, but it may vary with altitude and Mach number, particularly at supersonic speeds. K seldom is less than 0.90 for test installations and is usually between 0.95 and 1.00. Methods for determining K for a given installation are discussed in Section 6.2.

Equation 2.30 is plotted in Chart 8.2 as  $T_{ic}/T_a$  versus M for constant K and as  $T_a$  versus M for constant  $t_{ic}$  and K.

The free air thermometers now in use are all of the electrical resistance type. Their operation is based on the fact that the resistances of the sensing elements change with temperature. To obtain a signal from such a temperature sensing unit the element is placed in a bridge circuit. The circuit is designed so that the indicator registers the ratio of the current flow in two legs which makes the indication independent of the source voltage supply. (See Figure 2.4.)



Resistance Temperature Bulb Bridge Circuit

The indicator consists of an ammeter whose armature contains both indicator coils wound so that the indication is proportional to the two currents. (See Figure 2.5)



Construction of Resistance Temperature Bulb Indicator

The following instrument is in general use:

Type	Range	Readability	Accuracy
C-10	-60 to +60 degrees C (or other range as desired)	0.5 degrees C	±0.5 degrees C

### 2.5 THE CALCULATION OF EQUIVALENT AIRSPEED, MACH NUMBER, AND STANDARD DAY TRUE AIRSPEED

2.5.1 Equivalent Airspeed:

The equivalent airspeed  $(V_e)$  is frequently used as a basis for reducing flight test data for piston-engined airplanes as it is a direct measure of the free stream dynamic pressure (q),

$$q = \frac{1}{2} \rho V_t^2 = \frac{1}{2} \rho_{SL} V_e^2 = K V_e^2$$
 2.31

 $V_e$  may be expressed in terms of pressure altitude  $(H_c)$  and Mach number (M) as

$$\frac{V_e}{M} = a \sqrt{e}$$

This equation is plotted in Chart 8.3 as  $V_e/M$  versus  $H_c$ .

## 2.5.2 Mach Number:

The machmeter in its present form should not be used in precision flight test work as it is not sufficiently accurate. Therefore. Mach number must be determined by other means

If the true airspeed and ambient temperature are known, Mach number is defined by the relation

$$M = \frac{V_{tt}}{a_t}$$
 2.33

where

 $V_{tt}$  = test day true airspeed

 $a_t$  = test day speed of sound

The velocity of sound in a perfect gas is proportional to the square root of the temperature, or

$$\frac{a_{t}}{a_{SL}} = \sqrt{\frac{T_{at}}{T_{aSL}}}$$
 2.34

where

 $T_{at}$  = test day ambient temperature, °K  $a_{SL}$  = 661.48 knots  $T_{aSL}$  = 288.16°K

Hence

$$a_t = 38.967 T_{at}$$
 knots 2.35

and

$$M = \frac{V_{11} (knots)}{38.967 \sqrt{T_{a1} (^{\circ}K)}}$$
2.36

This equation is plotted in Chart 8.4 as  $V_t$  versus  $T_a$  for constant Mach number lines.

Inasmuch as the true velocity is seldom available directly. Mach number is more conveniently obtained through the compressible flow equation (2.23, 2.25). M is given as a function of  $q_c/P_a$  in Tables 9.4 and 9.5.  $P_a$  is obtained from pressure altitude ( $H_c$ ) in the standard atmosphere, Table 9.2 or Table 9.3, and  $q_c$  is found from  $V_c$  and Table 9.6. This information is plotted in Chart 8.5 as Mach number (M) versus calibrated airspeed  $(V_c)$  for constant pressure altitude  $(H_c)$ . Given any two of these variables the third may be found directly from this chart. Chart 8.5 is also applicable for indicated quantities corrected for instrument error. In this case, the chart may be interpreted as  $M_{ic}$  versus  $V_{ic}$  for constant  $H_{ic}$ .

### 2.5.3 Standard Day True Airspeed:

In the previous section, Mach number is expressed as a function of pressure altitude and calibrated airspeed; therefore, at a given  $H_c$  and  $V_c$ , the test day Mach number is equal to the standard day Mach number.

$$M_{test} = \frac{V_{tt}}{a_t} = M_{std} = \frac{V_{ts}}{a_s} = M \qquad 2.37$$

where

 $V_{ts}$  = standard day true airspeed  $a_s$  = standard day speed of sound

The verity of this statement is evidenced by the fact that Mach number is a function of  $P_a$  and  $P_t$  (equation 2.20, 2.21) and therefore can be expressed independent of the ambient temperature. The standard day speed of sound can be expressed as:

$$a_{s} = 38.967 \sqrt{T_{as}}, knots$$
 2.38

where

T<sub>as</sub> = standard day ambient temperature (corresponding to H<sub>c</sub> in the standard atmosphere), °K

Hence,

 $V_{ts} = 38.967 \, M \sqrt{T_{as}}, \text{ knots}$  2.39

where  $T_{as}$  is in  ${}^{\circ}K_{s}$ . This equation is plotted in Chart 8.5. This chart can be used to find  $V_{ts}$  from M and  $H_{c}$ , M and  $V_{c}$ , or  $H_{c}$  and  $V_{c}$ .

#### SECTION 3

### INSTRUMENT ERROR - THEORY AND CALIBRATION

Several corrections must be applied to the indicated altimeter and airspeed indicator readings before pressure altitude and calibrated airspeed can be determined. The indicated readings must be corrected for instrument error, pressure lag error and position error, in that order. In level unaccelerated flight there will be no pressure lag, in which case the position error correction can be applied directly following the instrument correction. The instrument error is the subject of this section. The pressure lag error and position error are discussed in sections 4 and 5.

### 3.1 INSTRUMENT ERROR

The altimeter and airspeed indicator are sensitive to pressure and pressure differential respectively, but the dials are calibrated to read altitude and calibrated airspeed according to equations (2, 3, 2, 4) and (2, 12, 2, 13). It is not possible to perfect an instrument which con-represent such nonlinear equations exactly under all flight conditions. As a result an error exists called instrument error. Instrument error is the result of several things:

- (1) Scale error and manufacturing discrepancies
- (2) Hysteresis
- (3) Temperature changes
- (4) Coulomb and viscous friction
- (5, Inertia of moving parts

The calibration of an altimeter or airspeed indicator for instrument error is usually conducted in an instrument laboratory. A known pressure or pressure differential is applied to the instrument to be tested. The instrument error is determined as the difference between this known pressure and the instrument indicated reading. Such things as friction and temperature errors are considered as tolerances since they are not dependent on the instrument readings.

24

An instrument with excessive friction or temperature errors should be rejected.

Data should be taken in both directions so that the hysteresis can be determined. Hysteresis is then the difference between the "up" and "down" corrections. An instrument with large hysteresis must be rejected as it is difficult to account for this effect in flight.

As an instrument wears, its calibration changes. Therefore, each instrument should be recalibrated periodically. The repeatability of the instrument is determined from the instrument calibration history. The repeatability of the instrument must be good for the instrument calibration to be meaningful.

## 3 2 THE ALTIMETER

The altimeter is calibrated by placing it in a vacuum chamber where pressure is measured by a mercury barometer. The chamber pressure is varied up and down throughout the range for which the altimeter is intended to be used. Simultaneous readings of the barometer and altimeter are taken. The instrument correction  $(\Delta H_{ic})$  is determined as the difference between the instrument corrected and indicated altitudes.

 $\Delta H_{ic} = H_{ic} - H_i$ where  $H_{ic}$  corresponds to the applied pressure according to equations 2.3 and, 2.4. The results are usually plotted as shown in Figure 3.1.





To use this instrument correction chart, the instrument correction  $(\Delta H_{ic})$  is added to the indicated altitude  $(H_i)$  to obtain the indicated altitude corrected for instrument error  $(H_{ic})$ 

$$H_{ic} = H_i + \Delta H_{ic} \qquad 3.2$$

In general, at the Air Force Flight Test Center, the altimeter is calibrated every 1,000 feet to 20,000 feet and every 2,000 feet for higher altitudes.

### 3.3 THE AIRSPEED INDICATOR

The airspeed indicator is calibrated by applying a known differential pressure to the instrument to be calibrated. The pressure is varied up and down throughout the range for which the instrument is intended to be used The instrument correction  $(\Delta V_{ic})$ is determined as the difference between the instrument corrected and indicated airspeeds.

en <del>de alt</del> de les en altre d

$$\Delta V_{ic} = V_{ic} - V_i \qquad 3.3$$

where  $V_{ic}$  corresponds to the applied differential pressure according to equations 2.12 and 2.13 The results are plotted in the same general form as the altimeter instrument correction the correction is plotted versus the indicated airspeed To use this instrument correction chart, the instrument correction ( $\Delta V_{ic}$ ) is added to the indicated airspeed ( $V_i$ ) to obtain the indicated airspeed corrected for instrument error ( $V_{ic}$ )

 $V_{ic} = V_i + \Delta V_{ic} \qquad 3.4$ 

At the AFFTC, the airspeed indicator is calibrated every 10 knots throughout the intended speed range

#### SECTION 4

## PRESSURE LAG ERROR - THEORY AND CALIBRATION

### 4.1 PRESSURE LAG ERROR AND THE LAG CONSTANT

The altimeter and airspeed indicator are subject to an error called pressure lag error. This error exists only when the aircraft in which the instruments are installed is changing airspeed or altitude, as during an acceleration or climb. In this case, there is a time lag between such time as the pressure change occurs and when it is indicated on the instrument dial. The effect on the altimeter is obvious; as the aircraft climbs, the instrument will indicate an altitude less than the actual altitude. In the airspeed indicator, the lag may cause a reading too large or too small depending on the proportion of the lag in the total and static pressure systems. Converted to "feet" or "knots", this error is often insignificant. However, it may be significant and should be considered in certain maneuvers such as high speed dives and zoom climbs in which the instrument diaphragms must undergo large pressure rates. Pressure lag is discussed in detail in NACA Report No. 919, "Accuracy of Airspeed Measurements and Flight Calibration Procedures," by Wilbur B. Huston.

Pressure lag is basically a result of:

- (1) Pressure drop in the tubing due to viscous friction.
- (2) Inertia of the air mass in the tubing,
- (3) Instrument inertia and viscous and kinetic friction.

(4) The finite speed of pressure propogation; i.e., acoustic lag A detailed mathematical treatment of the response of such a system would be difficult. Fortunately, a very simple approach is possible which will supply adequate lag corrections over a large range of flight conditions encompassing those presently encountered in the performance testing of aircraft In this approach, it is assumed that the pressure system can be adequately represented by a linear first order equation:

$$\frac{\mathrm{d}\mathbf{F}_{1}(t)}{\mathrm{d}t} + \frac{1}{\lambda} \mathbf{P}_{1}(t) = \frac{1}{\lambda} \mathbf{P}(t) \qquad 4.1$$

28

where

- P(t) = the applied pressure at time (t). This is  $P_s$  in the case of the altimeter and either  $P_s$ or  $P_t'$  in the case of the airspeed indicator.
- $F_1(t)$  = the indicated pressure at time (t).

This equation is derived by means of dimensional analysis.

The lag constant for laminar flow of air in tubing can be expressed as:

$$\lambda = \frac{32 \mathcal{A} L^2}{D^2 rP} \left(1 + \frac{Q}{LA}\right)$$
 4.2

where

 $\mu$   $\sim$  coefficient of viscosity of air, slugs/ft-sec

L : length of tubing, feet

D = diameter of tubing, feet

8 ratio of specific heats, 1.4 for air

 $P = applied pressure, lbs/feet^2$ 

Q = instrument volume, feet<sup>3</sup>

A = cross-sectional area of tubing, feet<sup>2</sup>

Many assumptions are made in the formulation of the differential equation and in its solution. The most important of these are:

(1) The rate of change of the applied pressure is nearly constant.

P(t) = Kt 4.3 where  $K = \frac{dP}{dt} = a \text{ constant}$ . This is a good assumption.

(2) Laminar flow exists. For this to be true, it is necessary that the Reynolds number  $(N_R)$  be less than 2000, where

$$N_R \propto \frac{\rho}{\mu} \frac{1}{P} \frac{dP}{dt}$$
 4.4

for a given installation. In typical altimeter and airspeed systems, a  $N_R$  of 500 is seldom exceeded in flight. Therefore, in laboratory calibrations, pressure rates greater than those encountered in flight should not be applied or erroneous results may be obtained.

(3) The pressure lag is small compared with the applied pressure.

This is generally the case; however, at very high altitudes this assumption becomes critical.

(4) The air and instrument inertias are negligible.

(5) The acoustic lag ( $\gamma$ ) is negligible.  $\gamma$  is defined as the time for a pressure disturbance to travel the length of the tubing.

$$\tau = \frac{L}{c}$$
 4.5

where

L = length of tubing, feet

c = speed of pressure propogation in the tubing,
 1000 feet per second for small diameter tubing.

In flight test application, the acoustic lag contribution is usually small. However, if  $\mathcal{T}$  is not small compared to  $\lambda$ , this assumption is not valid and a more detailed analysis such as that outlined in NACA Report No. 919 is necessary.

(6) The pressure drop across orifices and restrictions is negligible. This is true only if a minimum of such restrictions exist so that the tubing is nearly a smooth, straight "pipe" of uniform diameter.

(7) The lag constant  $(\lambda)$  is a constant. This is not strictly true as

$$\lambda \propto \vec{p}$$
 4.6

for a given installation. However, over a small pressure range,  $\lambda$  is nearly constant so that it may be treated as such in the solution of equation 4.1.

The particular solution to the differential equation with these assumptions is:

$$P_{i} = \frac{dP}{dt} (t - \lambda) \qquad 4.7$$

From equation 4.3 and 4.7.

$$P_i = P - \lambda \frac{dP}{dt}$$
 4.8

Solving for  $\lambda$ , the definition of the lag constant is

$$\lambda = \frac{P - P_i}{dP/dt}$$
 4.9

The lag constant for a given static or total pressure system can be determined experimentally by comparing the indicated and applied pressures for a given pressure rate. This can be done in flight or in the laboratory. In either case, the test should be conducted over a small range of pressures so that the assumption that  $\lambda$  is a constant is not violated.

When the lag constant at one value of  $\mathcal{M}/P$  is obtained, it may be extrapolated to other conditions by the expression

$$\frac{\lambda_1}{\lambda_2} = \frac{\mu_1 P_2}{\mu_2 P_1}$$
4.10

which is obtained from equation 4.6. Usually, the test results are reduced to sea level standard static conditions. Then the lag constant at any value of  $\mathcal{M}$  and P can be obtained from the expression

$$\lambda = \lambda_{\rm SL} \frac{\mu}{\mu_{\rm SL}} \frac{P_{\rm SL}}{P}$$
 4.11

With the lag constants for the static and total pressure systems known, the error in altimeter and airspeed indicator readings due to pressure lag can be calculated for any test point from the basic indicator readings.

Due to the nature of the approximations made in this analysis. is generally not possible to assume that the overall lag error correction can be made with a precision of more than 80 percent. Reduction of instrument and line volumes, however, can usually reduce the system lag errors under any set of conditions to a small percentage of the quantity being measured; in which case, more precise corrections are not required for practical work.

## 4.2 CORRECTION OF FLIGHT TEST DATA FOR LAG

## 4.2.1 The Altimeter:

The indicated pressure altitude corrected for instrument error  $(H_{ic})$  is related to the static pressure  $(P_s)$  by the differential

31

equation:

$$dP_s = -G\rho_s dH_{1c} \qquad 4.12$$

where  $\rho_5 =$  the standard day air density. For small increments, the differentials of equation 4.12 may be assumed to be finite differences.

$$\Delta \mathbf{P}_{sl} = -G\rho_s \Delta H_{icl} \qquad 4.13$$

where

$$\Delta P_{sl} = \text{the static pressure lag:}$$
  
$$\Delta P_{sl} = (P_{sl} - P_s) \qquad 4.14$$

where

 $P_s = static pressure corresponding to H_{ic}$   $P_{sf} = static pressure at static pressure source$  $\Delta H_{icf} = altimeter lag error correction:$ 

and

$$\Delta H_{1c} \ell = (H_{1c} \ell - H_{ic}) \qquad 4.15$$

where

H<sub>ic</sub> = indicated pressure altitude corrected for instrument error

H<sub>icl</sub> = indicated pressure altitude corrected for instrument and lag error

The lag constant for the static pressure system ( $\chi_s$ ) can be defined from equation 4.9 as:

$$\lambda_{s} = \frac{P_{s\ell} - P_{s}}{dP_{s\ell} / dt} = \frac{\Delta P_{s\ell}}{dP_{s\ell} / dt}$$
4.16

With the approximation that

$$\frac{dP_{ss}}{dt} = \frac{dP_{s}}{dt}$$
 4.17

equation 4.16 can be written as

$$\lambda_{s} = \frac{\Delta P_{s} 2}{d P_{s} / dt}$$
4.18

Substituting for  $\Delta P_{s}$  and  $dP_{s}/dt$ 

$$\Delta H_{ic} l = \lambda_s \frac{dH_{ic}}{dt}$$
 4.19

From equation 4.11

$$\lambda_{\rm s} = \lambda_{\rm sSL} \frac{\mu}{\mu_{\rm SL}} \frac{P_{\rm aSL}}{P_{\rm s}}$$
 4.20

where  $\lambda_{sSL}$  is the ig constant for the static pressure system at standard sea level conditions.

For convenience in plotting, equation 4.20 is rewritten as

$$\lambda_{s} = \lambda_{sSL} \frac{\lambda_{sH_{1C}}}{\lambda_{sSL}} \frac{\lambda_{s}}{\lambda_{sH_{1C}}} 4.21$$

where

$$\frac{\lambda_{sHic}}{\lambda_{sSL}} = \frac{\mu_{H_{1C}}}{\mu_{SL}} + \frac{P_{aSL}}{P_{s}}$$
4.22

$$\frac{\lambda_s}{\lambda_{sH_{ic}}} = \frac{\mu}{\mu_{H_{ic}}} \simeq \frac{T_{at}}{T_{as}}$$
 4.23

The approximation of equation 4.23 is very good for the usual case where the difference between the test and standard day temperatures is small.

Equation 4.22 is plotted as the STATIC LINE of Chart 8.61 in the form

$$\frac{\Lambda H_{ic}}{\Lambda SL}$$
 versus  $H_{ic}$  for  $V_{ic} = 0$  (STATIC)

(The parameter  $V_{ic}$  included on this chart is used in the determination of the total pressure lag constant.)

Equation 4.23 is plotted in Chart 8.62 as

$$\frac{\lambda}{\lambda H_{ic}}$$
 versus  $H_{ic}$  for  $t_{at}(^{\circ}C)$ 

In summary, the calculation for altimeter lag error correction  $(\Delta H_{ic} \not )$  at any test point  $(H_{ic}, t_{at}, dH_{ic}/dt)$  is then:

$$\Delta H_{ic} l = \frac{\lambda_s}{60} \frac{dH_{ic}}{dt} \qquad 4.24$$

where

where  

$$\lambda_{s} = \lambda_{sSL} \frac{\lambda_{sHic}}{\lambda_{sSL}} \frac{\lambda_{s}}{\lambda_{sHic}}$$
4.21  
with  $dH_{ic}/dt = indicated rate-of-climb corrected for instrument error, feet/minute
 $\lambda_{sSL} = sea \ level \ static \ pressure \ lag \ constant, \ from previous \ calibration, \ sec$ 

$$\frac{\lambda_{sHic}}{\lambda_{sSL}} from \ Chart \ 8.61 \ for \ H_{ic}, \ V_{ic} = \ STATIC$$

$$\frac{\lambda_{s}}{\lambda_{sHic}} from \ Chart \ 8.62 \ for \ H_{ic}, \ t_{at} \ (^{\circ}C)$$$ 

The indicated altitude corrected for instrument and lag error is then

$$H_{ic} = H_{ic} + \Delta H_{ic} \qquad 4.25$$

- -

An example of the calculation of  $H_{icl}$  is given with Chart 8.6.

#### 4.2.2 The Airspeed Indicator:

The differential pressure corresponding to the indicated airspeed corrected for instrument error may be given as,

$$q_{cic} = P_t' - P_s \qquad 4.26$$

where

Pt' = total pressure felt by the total pressure diaphragm of the airspeed indicator

P<sub>s</sub> = static pressure felt by the static pressure diaphragm of the airspeed indicator

With any lag in the total and static pressure systems accounted for,

$$q_{cic} l = P_t l - P_s l \qquad 4.27$$

where

- $P_t'_{\ell}$  = total pressure applied to total pressure source of pitot static system
- P<sub>sf</sub> = static pressure applied to static pressure source of pitot static system

Defining the differential pressure error due to lag  $(\Delta q_{cicl})$  as

$$\Delta q_{cic\ell} = q_{cic\ell} - q_{cic} \qquad 4.28$$

it follows from equation 4.9 that

$$\Delta q_{cicl} = (P'_t - P'_t) - (P_{sl} - P_s)$$
$$= \lambda_l \frac{dP'_t}{dt} - \lambda_s \frac{dP_{sl}}{dt} \qquad 4.29$$

Differentiating 4.27 and dividing by dt,

$$\frac{dq_{cic\ell}}{dt} = \frac{dP_{t\ell}}{dt} - \frac{dP_{s\ell}}{dt} - 4.30$$

Therefore,

$$\Delta q_{cicl} = \lambda_{t} \frac{dq_{cicl}}{dt} - (\lambda_{s} - \lambda_{t}) \frac{dP_{sl}}{dt} \qquad 4.31$$

With the approximations that:

$$\frac{\Delta q_{cic\ell}}{dt} = \frac{dq_{cic}}{dt}, \frac{dP_{s\ell}}{dt} = \frac{dP_s}{dt} \qquad 4.32$$

equation 4.32 becomes

$$\Delta q_{cicl} = \lambda_t \frac{dq_{cic}}{dt} - (\lambda_s - \lambda_t) \frac{dP_s}{dt} \qquad 4.33$$

With the use of the altimeter equation 4.12 which relates  $dP_s$  to  $dH_{ic}$ , and the airspeed indicator equation (2.12, 2.13) which may be differentiated to give  $dq_{cic}$  as a function of  $dV_{ic}$  and  $V_{ic}$ , equation 4.33 may be modified to give the airspeed indicator lag correction factor ( $\Delta V_{ic}$ ) in terms of  $dV_{ic}/dt$ in knots/sec and  $dH_{ic}/dt$  in feet/min as dH

$$\Delta V_{ict} = \lambda_t \frac{dV_{ic}}{dt} + \frac{(\lambda_s - \lambda_t)G\sigma_s - \frac{ic}{dt}}{170.921V_{ic} \left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^2\right]^{2.5}}$$
4.34

for  $V_{ic} = a_{SL}$ , and

SL  

$$\Delta V_{icl} = \lambda_{t} \frac{dV_{ic}}{dt} + \frac{(\lambda_{s} - \lambda_{t})G\rho_{s}}{224,287(\frac{V_{ic}}{a_{SL}})^{6}} \frac{V_{ic}}{\left[2(\frac{ic}{a_{SL}})^{2} - 1\right]}$$
4.35

for V<sub>ic</sub> a<sub>SL</sub>, where

$$\Delta V_{icl} = V_{icl} - V_{ic} \qquad 4.36$$

and V = indicated airspeed corrected for instrument error ic

 $V_{icl}$  = indicated airspeed corrected for instrument and lag error This may be written as:

$$\Delta V_{ic\ell} = \lambda_t \frac{dV_{ic}}{dt} + \frac{(\lambda_s - \lambda_t)}{60} \frac{dH_{ic}}{dt} \times F_1(H_{ic}, V_{ic}) \qquad 4.37$$

where

$$F_{1}(H_{ic}, V_{ic}) = \frac{G \sigma_{s}}{2.84869 V_{ic} \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5}}$$
4.38

for 
$$V_{ic} \leq a_{SL}$$
, and  
 $F_{1}(H_{ic}, V_{ic}) = \frac{G \rho_{s}}{3738.11} \frac{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}}{\left(\frac{V_{ic}}{a_{SL}}\right)^{6}\left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}$ 
4.39

for  $V_{ic} \ge a_{SL}$ , where  $\sigma_s$  and  $\rho_s$  are measured at  $H_{ic}$ .  $F_1(H_{ic}, V_{ic})$  has been plotted versus  $V_{ic}$  with  $H_{ic}$  as the parameter in Chart 8,63.

As in the case of the altimeter,

$$\lambda_{s} = \lambda_{sSL} \frac{\lambda_{sH_{ic}}}{\lambda_{sSL}} \frac{\lambda_{s}}{\lambda_{sH_{ic}}} 4.21$$

Similarly,

$$\lambda_{t} = \lambda_{tSL} \frac{\lambda_{tH_{ic}}}{\lambda_{tSL}} \frac{\lambda_{t}}{\lambda_{tH_{ic}}}$$
4.40

where

$$\frac{\lambda_{tH_{ic}}}{\lambda_{tSL}} = \frac{\mu_{H_{ic}}}{\mu_{SL}} \frac{P_{aSL}}{P_{s}+q_{cic}}$$
4.41

$$\frac{\lambda_{t}}{\lambda_{tH_{ic}}} = \frac{\mu}{\mu_{H_{ic}}} \approx \frac{T_{at}}{T_{as}}$$
 4.42

Equation 4.41 has been plotted in Chart 8.61 as

$$\frac{\lambda_{H_{ic}}}{\lambda_{SL}} \text{ versus } H_{ic} \text{ for } V_{ic}$$

Equation 4.42 has been plotted in Chart 8.62 as

$$\frac{\lambda}{\lambda H_{ic}}$$
 versus  $H_{ic}$  for  $t_{at}(°C)$ 

In summary, the calculation for airspeed indicator lag error correction  $(\Delta V_{ic})$  at any test point (H<sub>ic</sub>, t<sub>at</sub>, V<sub>ic</sub>, dV<sub>ic</sub>/dt, dH<sub>ic</sub>/dt) is then:

$$\Delta V_{ic} = \lambda_t \frac{dV_{ic}}{dt} + \frac{(\lambda_s - \lambda_t)}{60} \frac{dH_{ic}}{dt} \times F_1 (H_{ic}, V_{ic}) \qquad 4.43$$

where

$$\lambda_{s} = \lambda_{sSL} \frac{\lambda_{sHic}}{\lambda_{sSL}} \frac{\lambda_{s}}{\lambda_{sHic}} 4.21$$

$$\lambda_{t} = \lambda_{tSL} \frac{\lambda_{tH_{ic}}}{\lambda_{tSL}} \frac{\lambda_{t}}{\lambda_{tH_{ic}}}$$
4.40

with

dV <sub>ic</sub> /dt		indicated acceleration corrected for instrument error. knots/second
dH <sub>ic</sub> /dt	=	indicated rate-of-climb corrected for instrument error, feet/minute

$$\lambda_{sSL}$$
 = sea level static pressure lag constant, from previous calibration, seconds

$$\lambda_{tSL}$$
 = sea level total pressure lag constant, from previous calibration, seconds

$$\frac{\lambda_{sHic}}{\lambda_{sSL}}$$
 from Chart 8 61 for  $H_{ic}$ ,  $V_{ic}$  = STATIC

$$\frac{\lambda_{t}}{\lambda_{tH_{ic}}} = \frac{\lambda_{s}}{\lambda_{sH_{ic}}} \quad \text{from Chart 8, 62 for } H_{ic}, \ t_{a}(^{\circ}C)$$

$$\frac{\lambda_{tH_{1c}}}{\lambda_{tSL}}$$
 from Chart 8.61 for H<sub>ic</sub>, V<sub>ic</sub>

Then the indicated airspeed corrected for instrument and lag error is

$$V_{ic} g = V_{ic} + \Delta V_{ic} g \qquad 4.44$$

Data reduction outline 7.1 is included in Section 7 as a guide in performing this calculation. A numerical example is given with Chart 8.6.

## 4.3 DETERMINATION OF THE LAG CONSTANT

With the aid of equation 4.2 it is possible to compute theoretically the lag constants for an aircraft pitot-static system. The lag constants used for flight corrections, however, should not be computed, but should be determined experimentally, either in flight or in the laboratory. The computed lag constant is useful only as a rough check of the approximate magnitude of the lag error that may be expected under certain flight conditions.

# 4.3.1 Laboratory Calibration:

The lag constant (  $\lambda$  ) has been defined in a previous section as

$$\lambda = \frac{P - P_i}{dP/dt}$$
 4.9

where

P = the applied pressure at time t

 $P_i$  = the indicated pressure at time t

This equation suggests the use of a laboratory procedure to determine  $\lambda$  in which a steady rate of change of pressure is applied to the aircraft pitot-static system with P, P<sub>i</sub> and dP/dt all determined as a function of time.

4.3.1.1 The Static Pressure Lag Constant

The static pressure lag constant can be determined by the use of an experimental apparatus similar to that shown in Figure 4.1



Schematic of Equipment for Determination of Altimeter Lag Constant

The static pressure vent on the probe is sealed in a close fitting enclosure. An altimeter (or pressure gage) is mounted on a photo-panel as close as possible to the enclosure. Another altimeter (or pressure gage) is connected to the static pressure system. Timing counters operating at a one-per-second rate from an intervalometer are installed as shown in the figure. The pressure in the enclosure is lowered to the limit of the adjacent altimeter by means of a vacuum pump. The cameras and intervalometer are started and the needle value is opened gradually to maintain a dh/dt of about 5000 feet per minute.

If pressure gages are used, the data from the two camera films is most conveniently plotted in accordance with equation 4.16 as shown in Figure 4.2.

$$\lambda_{g} = \frac{P_{g} \ell - P_{g}}{dP_{g}/dt}$$

Here,  $P_{s}$  is the pressure in the enclosure and  $P_s$  is the pressure indicated by the aircraft static pressure system. At a given  $F_s$ ,  $\lambda_s$  is equal to the time increment between  $P_s$  and  $P_s \rho$ .







If altimeters are used rather than pressure gages, it is convenient to plot the data in accordance with equation 4.19 (see Figure 4.3).

$$\lambda_{\mathbf{s}} = \frac{H_{ic} \ell - H_{ic}}{dH_{ic}/dt}$$
 4.45

Again, the lag constant for a given  $H_{ic}$  is given by the interval between the two lines representing the indicated and actual simulated altitudes.







The value for  $\lambda_{g}$  obtained at any altitude, is the lag constant for the static pressure corresponding to that altitude and the temperature of the room in which the test was conducted. The sea level static pressure lag constant (  $\lambda_{SL}$ ) can be determined from the relation

$$\lambda_{sSL} = \lambda_{s} \frac{T_{aSL}}{T_{a}} \frac{P_{s}}{P_{aSL}}$$

where

 $T_a = room temperature$  $P_s = pressure in the enclosure$ 

This equation is applied to a number of pressures of Figure 4.2 or altitudes of Figure 4.3. From this information, a final value for  $\lambda_{\rm sSL}$ is selected. In general, the values obtained for high altitudes will be the most reliable, as  $\lambda_{\rm s}$ , the quantity with the most uncertainty, is larger. A sample format for the determination of  $\lambda_{\rm sSL}$  is included as data reduction outline 7.2.

#### 4.3.1.2 The Total Pressure Lag Constant

The total pressure lag constant can be determined by the use of a somewhat modified apparatus. In this case, a pressure is applied to the total pressure source and the static pressure source is left open to pick up atmospheric pressure. Either pressure gages or airspeed indicators may be used. (If airspeed indicators are used the pressure applied to the total pressure source should not exceed ambient pressure by an amount greater than the  $q_{cic}$  corresponding to the maximum  $V_{ic}$ for which the airspeed indicator was designed.) The applied pressure is bled off slowly to give the change in pressure (or airspeed) as a function of time.

If pressure gages are used, the data may be plotted in accordance with the definition of the total pressure lag constant (  $\lambda_t$ ). From equation 4.9

$$\lambda_{t} = \frac{P_{t}' 2 - P_{t}'}{dP_{t} \ell' / dt} \qquad 4.47$$

where  $P_t / 2$  is the pressure in the enclosure and  $P_t$  is the pressure indicated by the aircraft total pressure system. Such a plot is shown in Figure 4.4.

42



Figure 4.4 Plot Used to Determine Total Pressure Lag Constant, λ.

Here, as before, the total pressure lag constant at a given  $P_t^{\dagger}$  is equal to the time increment between  $P_t^{\dagger}$  and  $P_t^{\dagger}$ .

It is usually more convenient to use airspeed indicators. With the applied static pressure held constant,  $dH_{ic}/dt = 0$ ; therefore, from equations (4.34, 4.35)

$$\lambda_{t} = \frac{\Delta V_{ic} \ell}{dV_{ic}/dt} = \frac{V_{ic} - V_{ic}}{dV_{ic}/dt} \qquad 4.48$$

Hence, the data can be plotted as in Figure 4.5. Then, the total pressure lag constant at a given  $V_{ic}$  is equal to the time increment between  $V_{ic}$  and  $V_{ic} \ell^{*}$ .



Plot Used to Determine Total Pressure Lag Constant,  $\lambda_{t}$ 

The value for  $\lambda_t$  obtained at any airspeed is the lag constant for the total pressure corresponding to that airspeed and the temperature of the room in which the test was conducted. The sea level total pressure lag constant is then obtained from the relation

$$\lambda_{tSL} = \lambda_{t} \frac{T_{aSL}}{T_{a}} \frac{P_{t}'}{P_{aSL}}$$
$$= \lambda_{t} \frac{T_{aSL}}{T_{a}} \frac{(q_{cic} + P_{g})}{P_{aSL}}$$
4.49

where  $q_{cic} = f(V_{ic})$  and is given in Table 9.6. A sample format for the determination of the sea level lag constant is included as data reduction outline 7.3.

4.3.2 In-Flight Calibration:

Little experience has been obtained with in-flight methods for determining lag constants. However, since ground calibrations must be extrapolated to altitude where lag constants are much greater, in-flight calibrations do have an obvious advantage in that they can be determined more accurately provided suitable measurements can be made. Special equipment which is not generally available is necessary, however. Using in-flight methods, tapeline altitude is measured while the aircraft is changing flight conditions rapidly, as during a maximum power climb. (These measurements are perhaps best made with Askania cameras.) Tapeline altitudes are then converted to pressure altitudes by means of radiosonde data. A special installation must be made in the aircraft to provide correlation of altitudes recorded on the ground to those recorded in the aircraft.

4.3.2.1 The Static Pressure Lag Constant

The static pressure lag constant can be determined in flight as the aircraft climbs or dives. The indicated altitude  $(H_i)$  is compared to the pressure altitude  $(H_c)$  where

$$H_{c} = H_{i} + \Delta H_{ic} + \Delta H_{ic} \ell + \Delta H_{pc}$$
 4.50

where

 $\Delta H_{ic} = altimeter instrument error correction corresponding$  $to H_i$  $\Delta H_{ic} = altimeter lag error correction corresponding$  $to H_{ic}$  $\Delta H_{pc} = altimeter position error correction corresponding$  $to H_{ic} \\$ 

The altimeter lag error correction is determined as

$$\Delta H_{icl} = H_c - H_{ic} - \Delta H_{vc} \qquad 4.51$$

With  $\Delta H_{iCl}$  known, the static pressure lag constant can be determined from

$$\lambda_{s} = \frac{\Delta H_{ic} l}{dH_{ic} / dt}$$
4.45

 $dH_{ic}/dt$  can be determined from a time history of the test aircraft altimeter. The rate-of-climb indicator can be used but it may introduce considerable error as it is subject to lag error. The pressure altitude at which the test aircraft is operating  $(H_c)$  can be determined either by the use of a pacer aircraft or by radar tracking. These methods are discussed in Section 5.6. The altimeter position error correction  $(\Delta H_{DC})$  must be known from a previous calibration.

The use of this method is limited by the accuracy with which  $\Delta H_{ic} \mathcal{L}$  can be determined. This requires that the position error correction and the pressure altitude must be known with considerable accuracy, for it is quite possible that the error due to pressure lag can be completely hidden by errors in these quantities. Therefore, lag constants determined by this method should not be accepted without some reservation.

# 4.3, 2.2 The Total Pressure Lag Constant

Inflight methods for determining the total pressure lag constant are not present'y used due to difficulty encountered in the measurement of the calibrated airspeed with sufficient accuracy. The airspeed indicator lag error correction has been expressed as

$$\Delta V_{ic\ell} = \lambda_t \frac{dV_{ic}}{dt} + (\lambda_s - \lambda_t) \frac{dH_{ic}}{dt} \times F_1(H_{ic}, V_{ic}) \quad 4.37$$

Several flight procedures are theoretically possible by which  $-\lambda_t$  can be determined

(1) Level acceleration  $(dH_{ic}/dt = 0)$ 

$$\lambda_{t} = \frac{\Delta V_{ic} \ell}{dV_{ic} / dt}$$
4.52

(2) Climb or dive at constant  $V_{ic}$  ( $dV_{ic}/dt = 0$ )

$$\lambda_{t} = \lambda_{s} - \frac{\Delta V_{ic} \ell}{dH_{ic}/dt \times F_{1}(H_{ic}, V_{ic})}$$
4.53

(3) Climb or dive at a constant acceleration or deceleration

$$\lambda_{t} = \frac{\Delta V_{ic} \ell - \lambda_{s} \frac{dH_{ic}}{dt} \times F_{1}(H_{ic}, V_{ic})}{\frac{dV_{ic}}{dt} - \frac{dH_{ic}}{dt} F_{1}(H_{ic}, V_{ic})}$$
4.54

In all of these procedures, it is necessary to determine  $\Delta V_{ic} \mathcal{L}$  where

.

Ì

$$\Delta V_{ic} \ell = V_c - V_{ic} - \Delta V_{pc} \qquad 4.55$$

Tracking methods are not reliable to give velocities accurately and the pacers are not calibrated for lag; therefore, it is not possible to obtain  $V_c$  with sufficient accuracy to give a reliable  $\Delta V_{ic} \ell$  and hence  $\lambda_t$ .

# SECTION 5 POSITION ERROR - THEORY AND CALIBRATION

In addition to instrument error and pressure lag error, the altimeter and airspeed indicator a subject to another error called position error. Once corrections for instrument and pressure lag error have been made, position error may be accounted for and suitable corrections made. Under steady level flight conditions there is no lag error, in which case position error corrections can be made directly following the instrument error correction.

### 5.1 ORIGIN OF POSITION ERROR

Determination of the pressure altitude and airspeed at which an aircraft is operating is dependent on the measurement of free stream impact pressure and free stream static pressure by the aircraft pitotstatic system as evidenced by equations (2.1, 2.2) and (2.10, 2.11). Generally, the pressures registered by the pitot-static system differ from free stream pressures as a result of:

- (1) The existence of other than free stream pressures at the pressure source.
- (2) Error in the local pressure at the source caused by the pressure sensors.

The resulting error is called position error. In the general case, position error may result from error at both the static and total pressure sources, For most flight test work it may be presumed that all of the position error originates at the static pressure source. The possibility of a total pressure error must; however, always be considered.

5.1.1 Total Pressure Error:

As an aircraft moves through the air, a static pressure disturbance is generated in the air producing a static pressure field around the aircraft. At subsonic speeds, the flow perturbations due to the aircraft static pressure field are very nearly isentropic in nature and hence do not affect the total pressure. Therefore, as long as the total pressure source is not located behind a propeller, in the

48

wing wake, in a boundary layer, or in a region of localized supersonic flow, the total pressure error due to the position of the total pressure head in the aircraft pressure field will usually be negligible. Normally, it is possible to locate the total pressure pickup properly and thus avoid any difficulty. This is most desirable as such things as localized supersonic flow regions produce rather erratic readings.

An aircraft capable of supersonic speeds should be supplied with a nose boom pitot-static system so that the total pressure pickup will be located ahead of any shock waves formed by the aircraft. This condition is essential for it is difficult to correct for total pressure errors which result when oblique shock waves exist ahead of the pickup. The shock wave due to the pickup itself is considered in the calibration equation (2.10, 2.11) discussed in Section 2.2.

Failure of the total pressure sensor to register the local pressure may result from the shape of the pitot-static head, inclination to flow, or a combination of both. Pitot-static tubes have been designed in many varied shapes. These tubes are tested in wind tunnels before installation to assure good design. Some are suitable only for relatively low speeds while others are designed to operate in supersonic flight as well. Therefore, if a proper design is selected and the pitot lips are not burred or dirty, there should be no error in total pressure due to the shape of the probe. Errors in total pressure caused by the **angle** of incidence of a probe to the relative wind are negligible for most flight conditions. Commonly used probes produce no significant errors at angles of attack or sideslip up to approximately 20 degrees. This range of insensitivity can be increased by using either a shielded or a swivel head probe.

5.1.2 Static Pressure Error:

The static pressure field surrounding an aircraft in flight is a function of speed and altitude as well as the secondary parameters, angle of attack, Mach number, and Reynolds number. Hence, it is seldom

49
possible to find a location for the static pressure source where the free stream pressure will be sensed under all flight conditions. Therefore, an error in the measurement of the static pressure due to the position of the static pressure orifice in the aircraft pressure field will generally exist.

At subsonic speeds, it is often possible to find some position on the aircraft fuselage where the static pressure error is small under all flight conditions. Therefore, aircraft limited to subsonic flight are best instrumented by the use of a flush static pressure port in such a position. The problem of the selection of an optimum static pressure orifice location is discussed in NACA Report 919, "Accuracy of Airspeed Measurements and Flight Calibration Procedures".

Aircraft capable of supersonic flight should be provided with a nose boom installation to minimize the possibility of total pressure error. This position is also advantageous for the measurement of static pressure as the effects of the aircraft pressure field will not be felt ahead of the aircraft bow wave. Therefore, at supersonic speeds when the bow wave is located downstream of the static pressure orifices, there will be no error due to the aircraft pressure field (See Figure 5.1).



Figure 5.1 Bow Wave of Supersonic Aircraft That Has Passed Behind Static Pressure Ports

Any error which will exist is a result of the probe itself. Hence, the calibration at supersonic speeds may be derived from wind tunnel tests on the probe, or flight tests of the probe on another aircraft. Assuming the head registers the local static pressure without error, any error which exists is a result of interference from shoulder on the boom installation, or of influence on the static pressure from the shock wave in front of the boom. Available evidence suggests that free stream static pressure will exist if the static ports are located more than 8-10 tube diameters behind the nose of the pitot-static tube and 4-6 diameters in front of the shoulder. (See Figure 5.2).



Pitot-Static Probe

In addition to the static pressure error introduced by the position of the static pressure orifices in the pressure field of the aircraft, there may be error in the registration of the local static pressure due primarily to inclination of flow. Error oue to sideslip is often minimized in the case of flush static ports by the location of holes on opposite sides of the fuselage manifolded together. In the case of boom installations, circumferential location of the static pressure ports will reduce the adverse effect of sideslip and angle of attack. The use of a swivel head also reduces this form of error.

## 5.2 DEFINITION OF POSITION ERROR

.

From the previous discussion it is seen that position error is created at the static pressure source by the pressure field around the aircraft. It should be borne in mind that position error in the total source may exist, resulting, for instance, from imperfections in the pitot tube. Sufficient airspeed calibrations should always be made on test aircraft to determine the possible existance of position error in the total pressure. Since in nearly all installations this does not occur

the following derivations consider pressure error in the static source only.

The relation of static pressure at any point within the pressure field of an aircraft to the free stream static pressure depends on Mach number (M), angle of attack ( $\propto$ ), sideslip angle ( $\beta$ ), Reynolds number (N<sub>R</sub>) and Prandtl number (N<sub>P</sub>).

$$\frac{P_s}{P_a} = f_1 (M, \alpha, \beta, N_R, N_{P_r})$$
 5.1

(The symbol f denotes a functional relationship which is usually different each time it appears). Defining the position error,  $\Delta P_{p}$ , as

$$\Delta P_{\rm D} = P_{\rm s} - P_{\rm a} \qquad 5.2$$

equation 5.1 can be written as

$$\frac{\Delta P_p}{P_a} = f_2 (M, \alpha, \beta, N_R, N_{P_r})$$
 5.3

Sideslip angles can be kept small;  $Np_r$  is approximately constant; and NR effects are negligible as long as the static pressure source is not located in a thick boundary layer. Hence, equation 5.3 can be simplified to

$$\frac{\Delta P_p}{P_a} = f_3(M, \alpha) \qquad 5.4$$

With no loss in generality, this equation can be changed to read:

$$\frac{\Delta P_p}{q_{cic}} = f_4(M_{ic}, C_{Lic})$$
 5.5

with

$$M_{ic} = f_{5} \left(\frac{q_{cic}}{P_{s}}\right) \qquad 2.26, 2.27$$

$$C_{Lic} = \frac{nW}{\delta P_{s}M_{ic}^{2} S/2} = \frac{nW}{\delta_{ic}} \frac{1}{M_{ic}^{2}} \frac{2}{\gamma SP_{aSL}} \qquad 5.6$$

where

q<sub>cic</sub> = indicated differential pressure, P<sub>t</sub>' - P<sub>s</sub>
n = load factor
W = airplane gross weight

 $s^{-}$  = ratio of specific heats, 1.4 for air

S = wing area, constant for a given airplane

 $\delta_{ic}$  = pressure ratio corresponding to  $H_{ic}$ ,  $P_s/P_{aSL}$ The term  $\Delta P_p/q_{cic}$  is termed the position error pressure coefficient, and is very useful in the reduction of position error data From the definition of  $C_{Lic}$ 

$$\frac{\Delta P_p}{q_{cic}} = f_6 \left( \frac{nW}{\delta_{ic}}, M_{ic} \right)$$
 5.7

Frequently weight and load factor effects may be neglected when presenting position error data; however, for aircraft carrying large fuel loads and whose weight accordingly may change markedly during the course of a flight or for aircraft in windup turns, the "nW" effects should be taken into account.

Consequently, when the relationship between the variables in equation 5.7 has been determined by means of a calibration, the following chart can be prepared for all weights and all load factors for the given aircraft in a given configuration.





Non-Dimensional Plot of Position Error Data to Include Weight and Load Factor Variation

## 5.3 RELATIONSHIP BETWEEN VARIOUS FORMS OF THE POSITION ERROR

The static pressure position error  $(\Delta P_p)$  causes error in the altimeter and airspeed indicator readings and in the Mach number calculated from these quantities. The resulting errors are designated  $\Delta H_p$ ,  $\Delta V_p$  and  $\Delta M_p$  respectively:

$$\Delta H_{\rm p} = H_{\rm ic} - H_{\rm c} \qquad 5.8$$

where

H<sub>c</sub> = pressure altitude

'H<sub>ic</sub> = indicated pressure altitude corrected for instrument error

$$\Delta V_{p} = V_{ic} - V_{c} \qquad 5.9$$

where

V<sub>c</sub> = calibrated airspeed V<sub>ic</sub> = indicated airspeed corrected for instrument error

$$\Delta M_{\rm p} = M_{\rm ic} - M$$
 5.10

where

M = Mach number

(In these definitions it is assumed that there is no lag error.) In general, it is more convenient to work with position error corrections rather than with the error itself, or

$$\Delta H_{pc} = H_c - H_{ic} = -\Delta H_p$$
 5.11

$$\Delta V_{pc} = V_c - V_{ic} = -\Delta V_p \qquad 5.12$$

$$\Delta M_{pc} = M - M_{ic} = -\Delta M_{p} \qquad 5.13$$

It can be seen that the corrections are <u>added</u> to the indicated quantities to obtain the actual quantities.

When the position error is produced entirely by pressure coefficient variation at the static source, it is possible to relate altimeter position error directly to airspeed and machmeter position errors (since in most installations the altimeter and airspeed indicator utilize the same static source). It is possible to develop equations relating  $\Delta P_p$ ,  $\Delta H_{pc}$ ,  $\Delta V_{pc}$ ,  $\Delta M_{pc}$ , and  $\Delta P_p/q_{cic}$ . This is the subject of the following section.

5.3.1  $\Delta P_p$  and  $\Delta H_{pc}$ :

The differential pressure equation for the altimeter can be written as

$$dP_s = -G_{\rho_s} dH_{1c}$$
 5.14

$$\frac{dP_s}{dH_{ic}} = -0 \ 0010813 \sigma_s$$
 5.15

where

dP<sub>s</sub> = differential static pressure, "Hg
 dH<sub>ic</sub> = differential indicated pressure altitude corrected for instrument error, feet
 ρ<sub>s</sub> = standard day air density at H<sub>ic</sub>, slugs/feet<sup>3</sup>

G = gravitational constant, 32.17405 feet/second<sup>2</sup>  $\sigma_s$  = standard day air density ratio at H<sub>ic</sub>,  $\rho_s/\rho_{SL}$ 

In the case of small errors, these differential quantities may be treated as finite differences. In this case,

$$dP_s = P_s - P_a = \Delta P_p, "Hg \qquad 5.16$$

$$dH_{ic} = H_{ic} - H_c = \Delta H_p = -\Delta H_{pc}$$
 5.17

With this approximation

AND WE WILLING TO THE COURSE

$$\frac{\Delta P_p}{\Delta H_{pc}} = 0.0010813\alpha_s, \frac{"Hg}{feet} 5.18$$

where  $\sigma_{e}$  is the standard day air density ratio at H<sub>ic</sub>. Then

$$\Delta P_{p} = \left(\frac{\Delta P_{p}}{\Delta H_{pc}}\right) \quad \Delta H_{pc} \qquad 5.19$$

This approximation is good for small errors, say  $\Delta H_{pc} \leq 1000$  feet, but cannot be used for large errors without introducing some error.

The exact relationship between  $\Delta P_p$  and  $\Delta H_{pc}$  can be obtained by insertion of  $\varphi_s = f(H)$ , equation 1.10, 1.13 into the altimeter equation 1.6 and integrating.

$$\int_{1}^{2} dP_{a} = P_{2} - P_{1} = -0.0010813 \int_{1}^{2} \sigma dH \qquad 5.20$$

where 1 represents the actual quantity and 2 represents the indicated quantity. With this nomenclature

$$P_2 - P_1 = P_s - P_a = \Delta P_p \qquad 5.21$$

Hence

$$\Delta P_{p} = -0.0010813 \int_{H_{c}}^{H_{1c}} \sigma \, dH$$
 5.22

where

$$\sigma = (1 - 6.87535 \times 10^{-6} H)^{4.2561}$$
 1.10

for 
$$H \leq 36,089$$
 feet, and  
 $\sigma = 0.29707_e^{-4.80634 \times 10^{-5}} (H - 36,089.24)$  1.13

for  $H \ge 36,089$  feet. Performing this integration and expanding in terms of  $\Delta H_{pc}$  by use of the Binomial theorem, an infinite series is obtained. Fortunately, only the first two terms are significant. With this simplification, the result can be expressed as

$$\frac{\Delta P_{p}}{\Delta H_{pc}} = 0.0010813\sigma_{s}, \frac{"Hg}{feet} 5.23$$

where  $\sigma_s$  is measured at  $(H_{ic} + \frac{\Delta H_{DC}}{2})$ .

$$\sigma_s = 1 - 6.87535 \times 10^{-6} (H_{ic} + \frac{\Delta H_{pc}}{2})$$
 5.24

1 25/1

for  $(H_{ic} + \frac{\Delta H_{pc}}{2}) \leq 36\ 089\ feet, and$   $\sigma_s = 0.29707e^{-4.80634} \times 10^{-5} \left[ (H_{ic} + \frac{\Delta H_{pc}}{2}) - 36, 089.24 \right]$ 5.25

for  $(H_{ic} + \frac{\Delta H_{DC}}{2}) \ge 36,089$  feet. Equation 5.23 is plotted in several forms:

$$\frac{\Delta P_{p}}{\Delta H_{pc}} \text{ versus } H_{ic} \text{ for } \Delta H_{pc} \qquad 8.7$$

$$\frac{\Delta P_{p}}{\Delta H_{pc}} \quad \text{versus } H_{ic} \text{ for } \Delta P_{p} \qquad \qquad 8.8$$

 $\Delta H_{pc}$  versus  $\Delta P_{p}$  for  $H_{ic}$  8.13

Another way to determine  $\Delta P_p$  from  $\Delta H_{pc}$  is to find the values for  $P_s$  and  $P_a$  in the Standard Atmosphere, Table 9.2 or 9.3, corresponding to  $H_{ic}$  and  $H_c$  and subtract.

#### Example:

Given:  $H_{ic} = 17,140$  feet  $\Delta H_{pc} = 550$  feet Required:  $\Delta P_p$  in "Hg Solution:

 $H_c = H_{ic} + \Delta H_{pc} = 17,690$  feet

From Table 9.3:

$$P_s = 15.480$$
 "Hg;  $P_a = 15.134$  "Hg  
 $\Delta P_p = P_s - P_a = 0.346$  "Hg

# 5.3.2 $\Delta P_p$ and $\Delta V_{pc}$ :

An approximate expression for the relationship between  $\Delta P_p$ and  $\Delta V_{pc}$  can be obtained by taking the first derivative of the standard airspeed indicator equation, and considering the derivative to be a finite difference. The resulting equation is good for small errors, say  $\Delta V_{pc} < 10$  knots. The standard airspeed indicator equation is given in Section 2 as

$$\frac{q_{cic}}{P_{aSL}} = \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{3.5} - 1 \qquad 2.12$$

7

for  $V_{ic} \leq a_{SL}$ , and

$$\frac{q_{cic}}{P_{aSL}} = \frac{166.921 \left(\frac{V_{ic}}{a_{SL}}\right)}{\left[7 \left(\frac{V_{ic}}{a_{SL}}\right)^2 - 1\right]^{2.5}} = 1$$
2.13

for  $V_{ic} \ge a_{SL}$ . The definition of qcic is

$$q_{civ} = P_t' - P_s \qquad 5.26$$

With no error an the total pressure, equation 5.26 can be differentiated to give

$$d(q_{cic}) = -d(P_s)$$
 5.27

Differentiating equation 2.12 and replacing  $dq_{cic}$  by its equivalent,  $-dP_s$ , gives the result:

$$\frac{dP_{6}}{dV_{ic}} = -\frac{1.4P_{aSL}}{aSL} \frac{V_{ic}}{aSL} \left[1 + 0.2\left(\frac{V_{ic}}{aSL}\right)^{2}\right]^{2.5} 5.28$$

for  $V_{ic} \leq a_{SL}$ 

where

 $dP_s$  = differential static pressure

 $dV_{ic}$  = differential airspeed

Assuming the derivatives to be finite differences

$$dP_8 = P_8 - P_a \qquad 5.29$$

$$dV_{ic} = V_{ic} - V_c = \Delta V_p = -\Delta V_{pc}$$
 5.30

With this approximation

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = \frac{1.4 P_{aSL}}{a_{SL}} \frac{V_{ic}}{a_{SL}} \left[ 1 + 0.2 \left( \frac{V_{ic}}{a_{SL}} \right)^{2} \right]^{2.5} 5.31$$

for  $V_{ic} \leq a_{SL}$ . Similarly, for the case that  $V_{ic} \geq a_{SL}$ 

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = 52.854 \left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}} 5.32$$

Then

$$\Delta \mathbf{P}_{\mathbf{p}} = \left(\frac{\Delta \mathbf{P}_{\mathbf{p}}}{\Delta \mathbf{V}_{\mathbf{pc}}}\right) \Delta V_{\mathbf{pc}}$$
 5.33

The exact expression relating  $\Delta P_p$  and  $\Delta V_{pc}$  is derived in the following manner for the case when no error in the total pressure source exists.

$$q_{cic} = P_t' - P_s = f(V_{ic})$$
 2.12, 2.13

For the case of no position error

$$q_c = P_t' - P_a = f(V_c)$$
 2.10, 2.11

as  $q_{cic} = q_c$  and  $V_{ic} = V_c$  with no position error. Now

$$\Delta \mathbf{P}_{\mathbf{p}} = \mathbf{P}_{\mathbf{s}} - \mathbf{P}_{\mathbf{a}} = (\mathbf{P}_{\mathbf{t}}' - \mathbf{P}_{\mathbf{a}}) - (\mathbf{P}_{\mathbf{t}}' - \mathbf{P}_{\mathbf{s}}) = q_{\mathbf{c}} - q_{\mathbf{cic}}$$

Therefore,

$$\Delta P_p = q_c - q_{cic} = f(V_c) - f(V_{ic}) \qquad 5.34$$

Since  $V_c = V_{ic} + \Delta V_{pc}$ , it is possible to expand the right hand side of this equation into a series for  $\Delta V_{pc}$  by use of the Binomial Theorem. The resulting series may be discontinued after the second term with no loss in accuracy for  $\Delta V_{pc} \leq 50$  knots. The resulting equation takes the form

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = \frac{1.4P_{aSL}}{a_{SL}} \frac{V_{ic}}{a_{SL}} \left[ 1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} \right]^{2.5} 5.35$$

$$+ \frac{0.7P_{aSL}}{a_{SL}} \left[ 1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} \right]^{1.5} \left[ 1 + 1.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} \right] \frac{\Delta V_{pc}}{a_{SL}}$$

for  $V_{ic} \leq a_{SL}$ , and

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = 52.854 \left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3} \cdot 5}$$

$$+ 52.854 \left(\frac{V_{ic}}{a_{SL}}\right)^{5} \frac{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{4} - 4 \cdot 5\left(\frac{V_{ic}}{a_{SL}}\right)^{2} + 3\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{4} \cdot 5} \frac{\Delta V_{pc}}{a_{SL}}$$

for  $V_{ic} \ge a_{SL}$ . Note that the first term is identical with that obtained by the approximate method. The second term may be considered as a correction to the first term that must be applied for large  $\Delta V_{pc}$ .

Equation (5.35, 5.36) has been plotted in several forms for the convenience of the reader

$$\frac{\Delta P_{p}}{\Delta V_{pc}} \text{ versus } V_{ic} \text{ for } \Delta V_{pc} \qquad \text{Chart} \qquad 8.9$$

$$\frac{\Delta P_p}{\Delta V_{pc}} \text{ versus } V_{ic} \text{ for } \Delta P_p \qquad \qquad \text{Chart} \qquad 8.10$$

$$\Delta V_{pc}$$
 versus  $\Delta P_p$  for  $V_{ic}$  Chart 8.13

## 5.3.3 $\Delta V_{pc}$ and $\Delta P_p/q_{cic}$ :

ŝ

The position error pressure coefficient is very useful as a parameter in high speed flight ( $M_{ic} \ge 0.6$ ). To facilitate obtaining  $\Delta P_p/q_{cic}$  from  $\Delta V_{pc}$ , a graph of  $\Delta V_{pc}$  versus  $V_{ic}$  for  $\Delta P_p/q_{cic}$  is plotted as Chart 8.11. This chart is determined from the following considerations.

$$\frac{\Delta P_{P}}{q_{cic}} = \frac{(\Delta P_{P}/P_{aSL})}{(q_{cic}/P_{aSL})}$$
5.37

From equations (5,35, 5.36, 2.12, 2.13, and 5.37)

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{\left\{1.4 \frac{V_{ic}}{\sigma_{SL}} \left[1 + 0.2 \left(\frac{V_{ic}}{\sigma_{SL}}\right)^{2}\right]^{2.5} \frac{\Delta V_{pc}}{\sigma_{SL}}\right\} + \left\{0.7 \left[1 + 0.2 \left(\frac{V_{ic}}{\sigma_{SL}}\right)^{2}\right]^{1.5} \left[1 + 1.2 \left(\frac{V_{ic}}{\sigma_{SL}}\right)^{2}\right] \left(\frac{\Delta V_{pc}}{\sigma_{SL}}\right)^{2}\right\}}{\left[1 + 0.2 \left(\frac{V_{ic}}{\sigma_{SL}}\right)^{2}\right]^{3.5} - 1}$$
for  $V_{ic} \leq a_{SL}$ , and

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{7 \frac{V_{ic}}{\sigma_{SL}} \left[ \frac{2 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} - 1}{7 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} - 1} + \frac{7 \left[ \frac{7 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} - 4.5 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} + 3 \right]}{\left[ \frac{7 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} - 1 \right]^{2}} \left( \frac{\Delta V_{pc}}{\sigma_{SL}} \right)^{2}} \right]} \left( \frac{\frac{\Delta V_{pc}}{\sigma_{SL}}}{\left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} \left[ \frac{1}{1 - \frac{\left[ 7 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{2} - 1 \right]}{166.921 \left( \frac{V_{ic}}{\sigma_{SL}} \right)^{7}} \right]} \right]}$$

5.39

5.41

for  $V_{ic} \ge a_{SL}$ .

5.3.4  $\Delta H_{pc}$  and  $\Delta V_{pc}$ :

When working with small errors, say  $\Delta H_{pc} = 1000$  feet or  $\Delta V_{pc}$  knots, the value of  $\Delta H_{pc}$  can be determined from  $\Delta V_{pc}$ , or vice versa, by the following relation which is obtained by dividing the approximate equations (5.35 and 5.36) by 5.22.

$$\frac{\Delta H_{pc}}{\Delta V_{pc}} = \frac{58.566}{\sigma_s} \left(\frac{V_{ic}}{\sigma_{SL}}\right) \left[1 + 0.2\left(\frac{V_{ic}}{\sigma_{SL}}\right)^2\right]^{2.5}$$
5.40

for  $V_{ic} \leq a_{SL}$ , and

$$\frac{\Delta H_{pc}}{\Delta V_{pc}} = \frac{48,880}{\sigma s} \left(\frac{V_{ic}}{\sigma SL}\right)^{6} \frac{2\left(\frac{V_{ic}}{\sigma SL}\right)^{2} - 1}{\left[\frac{7}{\sigma SL}\right]^{2} - 1}$$

for  $V_{ic} \ge a_{SL}$ , where  $\sigma_s$  is measured at  $H_{ic}$ . This equation has been plotted in Chart 8.12 as  $\Delta H_{pc} / \Delta V_{pc}$  versus  $V_{ic}$  for  $H_{ic}$ . For the case of larger errors where equations 5.18 and 5.31, 5.32 are not valid, the resulting equation 5.40, 5.41 and Chart 8.12 are of course not valid. In this case, one should use Chart 8.13 which is developed from the following relation which in turn is obtained by consideration of equations 5.35, 5.36 and 5.23.

$$\Delta P_{p} = 0.0010813 \sigma_{s} \Delta H_{pc} \qquad 5.42$$

$$= 1.4 P_{aSL} \frac{V_{ic}}{a_{SL}} \left[ 1 + 0.2 \left( \frac{V_{ic}}{a_{SL}} \right)^{2} \right]^{2.5} \frac{\Delta V_{pc}}{a_{SL}}$$

$$+ 0.7 P_{aSL} \left[ 1 + 0.2 \left( \frac{V_{ic}}{a_{SL}} \right)^{2} \right]^{1.5} \left[ 1 + 1.2 \left( \frac{V_{ic}}{a_{SL}} \right)^{2} \right] \left( \frac{\Delta V_{pc}}{a_{SL}} \right)^{2}$$

for  $V_{ic} \leq a_{SL}$ , and

÷

1168.45 P<sub>aSL</sub> 
$$\left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3} \cdot 5}$$
  
+ 1168.45 P<sub>aSL</sub>  $\left(\frac{V_{ic}}{a_{SL}}\right)^{5} \frac{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{4} - 4 \cdot 5\left(\frac{V_{ic}}{a_{SL}}\right)^{2} + 3\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{4} \cdot 5} \left(\frac{\Delta V_{pc}}{a_{SL}}\right)^{2}$ 

for  $V_{ic} \ge a_{SL}$ 

where  $\sigma_s$  is measured at  $(H_{ic} + \frac{\Delta H_{pc}}{2})$ . Chart 8.13 is in the form of  $\Delta V_{pc}$  versus  $\Delta P_p$  for  $V_{ic}$  and  $\Delta P_p$  versus  $\Delta H_{pc}$  for  $H_{ic}$ , or simply  $\Delta V_{pc}$  versus  $\Delta H_{pc}$  for  $V_{ic}$  and  $H_{ic}$ .

# 5.3.5 $\Delta M_{pc}$ and $\Delta H_{pc}$ :

The Mach number equation may be written as

$$\frac{P_t'}{P_s} = (1 + 0.2M_{ic}^2)^{3.5}$$
 5.44

for  $M_{ic} \leq 1.00$ . Differentiating, with  $P_t$ ' constant,

$$\frac{dP_{s}}{dM_{ic}} = \frac{1.4P_{s} M_{ic}}{(1+0.2M_{ic}^{2})}$$
 5.45

Making the approximations

$$dP_s = P_s - P_a = \Delta P_p \qquad 5.46$$

$$dM_{ic} = M_{ic} - M = \Delta M_{p} = -\Delta M_{oc} \qquad 5.47$$

the relation between the static pressure error and the Mach number position correction is obtained.

$$\frac{\Delta P_{p}}{\Delta M_{pc}} = \frac{1.4 P_{s} M_{ic}}{(1+0.2 M_{ic}^{2})}$$
 5.48

This approximate equation is valid for small errors. The Mach number position correction can be related to the altimeter position correction by dividing equation 5.22 by equation 5.52 and introducing the perfect gas equation 1.3.

$$\frac{\Delta M_{pc}}{\Delta H_{pc}} = 0.007438 \frac{(1+0.2M_{ic}^2)}{T_{as} M_{ic}} 5.49$$

for  $M_{ic} \leq 1.00$ , where  $T_{BS}$  is the standard day temperature corresponding to  $H_{ic}$ .

In the supersonic case,  $M_{ic} \ge 1.00$ 

$$\frac{\mathbf{P_t}'}{\mathbf{P_s}} = \frac{166.921 M_{1c}^{-7}}{(7M_{1c}^{-2} - 1)^{2.5}}$$
5.50

Proceeding as in the subsonic case

$$\frac{\Delta P_{pc}}{\Delta M_{pc}} = \frac{7P_{s}(2M_{ic}^{2} - 1)}{M_{ic}(7M_{ic}^{2} - 1)}$$
5.51

and

$$\frac{\Delta M_{pc}}{\Delta H_{pc}} = 0.001488 \frac{M_{ic} (7M_{ic}^2 - 1)}{T_{as}(2M_{ic}^2 - 1)}$$
 5.52

for  $M_{ic} \ge 1.00$ , where  $T_{as}$  corresponds to  $H_{ic}$ .

Equation 5.49, 5.52 has been plotted in Chart 8.14 in the form:

 $\Delta M_{pc}/\Delta H_{pc} \text{ versus } M_{ic} \text{ for } H_{ic} \qquad \text{Chart} \qquad 8.14$ Chart 8.14 and the above equations on which it is based are valid only for small errors, say  $\Delta M_{pc} < 0.04$  or  $\Delta H_{pc} < 1000$  feet.

For larger errors, a better approximation is necessary. The exact result can be obtained from the following analysis. In general, for  $M_{ic} \leq 1.00$ .

$$\frac{P_t'}{P_g} = (1 + 0.2M_{ic}^2)^{3.5}$$
 5.53

For the case of no position error,

$$\frac{\mathbf{P}_{t}}{\mathbf{P}_{a}} = (1 + 0.2M^{2})^{3.5}$$
 5.54

With

$$\Delta P_p = P_s - P_a \qquad 5.2$$

and

$$\Delta M_{pc} = M - M_{ic} \qquad 5.17$$

it is possible to express the exact relationship

$$\frac{\Delta F_{p}}{P_{s}} = f(M_{ic}, \Delta M_{pc})$$
 5.55

Expanding by the Binomial Theorem and retaining the first two terms yields the result

$$\frac{\Delta P}{P_{s}} = \frac{1.4M_{ic}\Delta M_{pc}}{(1+0.2M_{ic}^{2})} + \frac{0.7(1-1.6M_{ic}^{2})\Delta M_{pc}^{2}}{(1+0.2M_{ic}^{2})^{2}} 5.56$$

for  $M_{ic} \leq 1.00$ . Similarly for the supersonic case  $(M_{ic} \geq 1.00)$ 

$$\frac{\Delta P_{p}}{P_{s}} = \frac{7(2M_{1c}^{2} - 1)\Delta M_{pc}}{M_{ic} (7M_{ic}^{2} - 1)} - \frac{7(21M_{ic}^{4} - 23.5M_{ic}^{2} + 4)\Delta M_{pc}^{2}}{M_{ic}^{2} (7M_{ic}^{2} - 1)^{2}}$$
5.57



The final result is obtained by dividing equation 5.27 by equations 5.60, 5.61

$$\frac{\Delta P_{p}}{P_{g}} = 0.0010813 \frac{\sigma_{s}}{P_{g}} \Delta H_{pc} \qquad 5.58$$
$$= \frac{1.4M_{ic} \Delta M_{pc}}{(1+0.2M_{ic}^{2})} + \frac{0.7(1-1.6M_{ic}^{2}) \Delta M_{pc}^{2}}{(1+0.2M_{ic}^{2})^{2}}$$

for  $M_{ic} \leq 1.00$ , and

$$\frac{\Delta P}{P_s} P = 0.0010813 \frac{\sigma_s}{P_s} \Delta H_{pc} \qquad 5.59$$

$$= \frac{7(2M_{ic}^2 - 1)}{M_{ic}(7M_{ic}^2 - 1)} \Delta M_{pc} - \frac{7(21M_{ic}^4 - 23.5M_{ic}^2 + 4)\Delta M_{pc}^2}{M_{ic}^2(7M_{ic}^2 - 1)^2}$$

for  $M_{ic} \ge 1.00$ , where  $P_s$  is measured at  $H_{ic}$  and  $\sigma_s$  is measured at  $(H_{ic} + \frac{\Delta H_{pc}}{2})$ . This equation is plotted in Chart 8.15 in the form of  $\Delta H_{pc}$  versus  $\Delta P_p/P_s$  for  $H_{ic}$ , and  $\Delta M_{pc}$  versus  $\Delta P_p/P_s$  for  $M_{ic}$ , or simply  $\Delta M_{pc}$  versus  $\Delta H_{pc}$  for  $M_{ic}$  and  $H_{ic}$ .

## 5.3.6 $\Delta M_{pc}$ and $\Delta V_{pc}$ :

For small errors, say  $\Delta M_{pc} < 0.04$  or  $\Delta V_{pc} < 10$  knots, the ratio  $\Delta M_{pc}/\Delta V_{pc}$  can be obtained by multiplying equations 5.49, 5.52, 5.40, 5.41 with the result

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{P_{aSL}}{a_{SL}} \frac{1}{P_{s}} \frac{V_{1c}}{a_{SL}} \left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5} \frac{(1 + 0.2M_{ic}^{2})}{M_{ic}}$$
5.60

for  $V_{ic} \leq a_{SL}$ ,  $M_{ic} \leq 1.00$ ;

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{P_{aSL}}{5_{aSL}} \frac{1}{P_{s}} \frac{V_{ic}}{a_{SL}} \left[ 1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} \right]^{2.5} \frac{M_{ic}(7M_{ic}^{2}-1)}{(2M_{ic}^{2}-1)}$$
5.61

for  $V_{ic} \leq a_{SL}$ ,  $M_{ic} \geq 1.00$ ; and

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{166.921P_{aSL}}{aSL} \frac{1}{P_{s}} \frac{\left(\frac{V_{ic}}{a_{SL}}\right)^{6} \left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}} \frac{M_{ic}(7M_{ic}^{2} - 1)}{(2M_{ic}^{2} - 1)} 5.62$$

for  $V_{ic} \ge a_{SL}$ , Equations (5.60, 5.61, 5.62) are plotted in Chart 8.16 in the form

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} \text{ versus } M_{ic} \text{ for } H_{ic}$$
 Chart 8.16

No chart has been prepared in which one can directly relate  $\Delta M_{pc}$  to  $\Delta V_{pc}$  for the case of very large position error where Chart 8.16 is not valid. In the case of large error, it is possible to determine  $\Delta M_{pc}$  from  $\Delta V_{pc}$ , or vice versa, by the following indirect method:

- (1) Given  $\Delta V_{pc}$  and  $V_{ic}$ , determine  $\Delta P_p/q_{cic}$  from Chart 8.11
- (2) Determine  $M_{ic}$  from  $V_{ic}$  and  $H_{ic}$  and Chart 8.5
- (3) Determine  $\Delta M_{pc}$  from  $M_{ic}$  and  $\Delta P_p/q_{cic}$  and Chart 8.18

## 5.3.7 $\Delta M_{pc}$ and $\Delta P_p/q_{cic}$ :

For small errors, say  $\Delta M_{pc} < 0.04$ , the ratio  $\Delta M_{pc}/(\Delta P_p/q_{cic})$  may be formed by dividing equation (2.26, 2.27) by equation (5.48, 5.51) with the result

$$\frac{\Delta M_{pc}}{(\Delta P_p/q_{cic})} = \frac{(1+0.2M_{ic}^2)}{1.4M_{ic}} \left[ (1+0.2M_{ic}^2)^{3.5} - 1 \right] 5.65$$

for  $M_{ic} \leq 1.00$  and

$$\frac{\Delta M_{\rm BC}}{(\Delta P_{\rm p}/q_{\rm cic})} = \frac{M_{\rm ic} \left[ 166.921 M_{\rm ic}^{7} - (7M_{\rm ic}^{2} - 1)^{2.5} \right]}{7(7M_{\rm ic}^{2} - 1)^{1.5}(2M_{\rm ic}^{2} - 1)} 5.64$$

for  $M_{ic} \ge 1.00$ . This equation is plotted in Chart 8.17 in the form

$$\frac{\Delta M_{pc}}{(\Delta P_p/q_{cic})}$$
 versus  $M_{ic}$  Chart 8.17

The expression for large errors is obtained by dividing equations (5.56) and (5.57) by equations (2.26) and (2.27).

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{\frac{1.4M_{ic} \Delta M_{pc}}{(1+0.2M_{ic}^{2})} + \frac{0.7(1-1.6M_{ic}^{2})\Delta M_{pc}^{2}}{(1+0.2M_{ic}^{2})^{2}}}{\left[(1+0.2M_{ic}^{2})^{3.5} - 1\right]} 5.65$$

for  $M_{ic} \leq 1.00$ , and

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{\frac{7(2M_{ic}^{2} - 1)\Delta M_{pc}}{M_{ic}(7M_{ic}^{2} - 1)} - \frac{7(21M_{ic}^{4} - 23.5M_{ic}^{2} + 4)\Delta M_{pc}^{2}}{M_{ic}^{2}(7M_{ic}^{2} - 1)^{2}}}{\left[\frac{166.921 M_{ic}^{7}}{(7M_{ic}^{2} - 1)^{2.5}} - 1\right]} 5.66$$

for  $M_{ic} \ge 1.00$ . Equations (5.65 and 5.66) is plotted in Chart 8.18 in the form

$$\frac{\Delta P_{p}}{q_{cic}} \text{ versus } \Delta M_{pc} \text{ for } M_{ic}$$
 Chart 8.18

## 5.4 EXTRAPOLATION OF RESULTS

In general, the position error corrections must be established by a flight calibration made under all flight conditions. In some cases, however, it is possible to extrapolate over a wide range of conditions from a calibration over the speed range at one altitude. It has been shown that

$$\frac{\Delta P_p}{q_{cic}} = f_1 (M_{ic}, C_{Lic})$$
 5.5

$$= f_2(M_{ic}, \frac{nW}{\delta_{ic}})$$
 5.6

To derive this relation experimentally for direct application to any flight condition would thus require calibrations at several weights and load factors over the full altitude and speed range of the aircraft. The appropriate assumptions on which predictions to other conditions can be made from tests at one altitude depend on the Mach number and are considered in this section for several ranges of that parameter.

## 5.4.1 Low Mach Number Range (Mic 40.6):

For low Mach numbers, the effects of compressibility on pressure error may be considered negligible. Without introducing serious error, it may be said that the pressure coefficient is a function only of lift coefficient ( $C_L$ ) as shown in Figure 5.4.



This plot will represent the flow field around the probe for all flight conditions in the low Mach number range.

The position error calibrations for a low speed aircraft are often presented in another manner.

$$\frac{\Delta P}{q_{cic}} = f_3 (C_{Lic}) \qquad 5.67$$

Since  $C_L = n W / (\rho V_e^2 S/2)$  and in the low Mach number range  $V_c \cong V_e$ , it can be assumed that

$$C_{L} = \frac{nW}{\rho V^{2}S/2}$$
 5.68

or

$$C_{Lic} = \frac{nW}{\rho V_{sLic}^2 S/2}$$
5.69

Substituting equation 5.69 in equation 5.67

$$\frac{\Delta P_p}{q_{cic}} = f_4(\frac{nW}{V_{ic}}) = f_5(\frac{nW}{V_{ic}}) \qquad 5.70$$

It is possible to obtain a curve of  $\Delta P_p/q_{cic}$  versus  $\sqrt{nW/V_{ic}}$ from the results of a position error calibration over the  $C_L$  range at one altitude. From this plot, the position error pressure coefficient at any relevant altitude, weight and normal acceleration can be obtained.

A typical plot of  $\Delta P_p/q_{cic}$  versus  $V_{ic}$  showing nW variation and Mach number effects at the higher speeds is given in Figure 5.5. It may be seen from this figure that a change in nW at low speed can cause a substantial change in position error.





# Figure 5.5 Plot of $\Delta P_p/q_{cic}$ versus $V_{ic}$ for Low Speed Aircraft

The altimeter position error correction for low speed aircraft can be extrapolated from one altitude to another altitude at the same indicated airspeed corrected for instrument error  $(V_{ic})$  as long as there are no appreciable changes in weight or load factor. It has been shown for low speed aircraft in which there are no Mach effects that

$$\frac{\Delta \mathbf{P} \mathbf{p}}{q_{cic}} = f(V_{ic}) \text{ only} \qquad 5.71$$

for constant nW. Therefore, at a given  $V_{ic}$ , and hence  $q_{cic}$ , and constant weight and load factor, the static pressure error  $(\Delta P_p)$  is constant during altitude changes. Hence, for a given  $V_{ic}$  and constant nW

$$\Delta H_{pc2} = \Delta H_{pc1} \frac{\left(\frac{\Delta P_p}{\Delta H_{pc}}\right)}{\left(\frac{\Delta P_p}{\Delta H_{pc}}\right)} 5.72$$

In the case of small errors, equation 5.22 yields the result,

$$\Delta H_{pc2} = \Delta H_{pc1} \left(\frac{\sigma_{s1}}{\sigma_{s2}}\right)$$
 5.73

where  $\sigma_{s1}$  is the density ratio at  $H_{icl}$ , and  $\sigma_{s2}$  is the density ratio at  $H_{ic2}$ . In that the position error in the low speed range is always small, the problem of large error does not need to be considered here. Equation 5.77 states that the altimeter position error correction can be extrapolated to another altitude at the same  $V_{ic}$  by multiplication of  $\Delta H_{pc}$  by the ratio of the standard day air densities. This procedure is good only in the low speed range when there are no Mach number effects and when the variation in nW is not of significance.

# 5.4.2 Medium Subsonic and Transonic Mach Number Range $(0.6 < M_{ic} < 1.0)$ :

In this Mach number range, the position error pressure coefficient will in general depend on both  $M_{\rm ic}$  and  $C_{\rm Lic}$  so the

general equation must be considered.

$$\frac{\Delta P_{p}}{q_{cic}} = f_{1} (M_{ic}, C_{L_{ic}}) \qquad 5.5$$

$$= f_2 \left( M_{ic}, \frac{nW}{\delta_{ic}} \right)$$
 5.6

Therefore, in the general case a position error calibration must be conducted at several altitude and weight combinations. In many installations however, the effect of the  $C_{L_{ic}}$  parameter is negligible in this Mach number range. In this case, a calibration at one altitude can be extrapolated to other altitudes. The existence of any  $C_{L_{ic}}$  effect should be investigated by performing tests at two widely different altitudes and plotting curves of  $\Delta P_p/q_{cic}$  versus  $M_{ic}$  for the values of  $nW/\delta_{ic}$ . The result for a typical nose boom installation is shown in Figure 5.6. This curve shows that  $nW/\delta_{ic}$  effects exist in the system tested. This curve would be a single line if there were no  $C_{L_{ic}}$  effects.



Plot of  $\Delta P_p / q_{cic}$  versus  $M_{ic}$  for a Typical Nose Boom Installation Showing  $nW/q_{ic}$  Effects at Low Speed End

When there are no appreciable  $C_{\rm Lic}$  effects as indicated by a single curve of  $\Delta P_p/q_{\rm cic}$  (versus  $M_{\rm ic}$  for all  $nW/\delta_{\rm ic}$ ) the altimeter position error correction at one altitude can be extrapolated to any other altitude at the same  $M_{\rm ic}$ . With no  $C_{\rm Lic}$  effect.

$$\frac{\Delta P_{\rm p}}{\rm 9cic} = f(M_{\rm ic}) \text{ only} \qquad 5.74$$

Equations 5.63, 5.64 and 5.65 state

$$\Delta M_{pc} = f(M_{1c}, \frac{\Delta P_p}{q_{cic}})$$
5.63, 5.64  
5.65

From these equations, it follows that  $\Delta M_{pc}$  is a function of  $M_{ic}$  only and hence independent of altitude when there are no  $C_{Lic}$  effects.

$$\Delta M_{\rm DC} = f(M_{\rm 1C}) \text{ only} \qquad 5.75$$

Therefore, one may write

$$\Delta H_{pc2} = \Delta H_{pc1} \frac{\left(\frac{\Delta M_{pc}}{\Delta H_{pc}}\right)}{\left(\frac{\Delta M_{pc}}{\Delta H_{pc}}\right)} 5.76$$

for  $M_{icl} = M_{ic2}$ . In the case of small errors, equation 5.49, 5.52 yields the result

$$\Delta H_{pc2} = \Delta H_{pc1} \left( \frac{T_{as2}}{T_{as1}} \right)$$
 5.77

for  $M_{icl} = M_{ic2}$ , where  $T_{asl}$  and  $T_{as2}$  are the standard day air temperatures corresponding to  $H_{icl}$  and  $H_{ic2}$  respectively. In the case of large errors, it would appear that the above method of extrapolation would no longer be valid as equations (5.53 and 5.56) from which it is derived are no longer valid. Fortunately this is not the case and equation 5.81 can be used for very large errors, say  $\Delta H_{pc} \leq$ 3000 feet, with no appreciable loss of accuracy.

## 5.4.3 Supersonic Mach Number Range (Mic > 1.0):

An aircraft capable of supersonic flight should be equipped with a nose boom installation. In this case, the aircraft bow wave will pass behind the static pressure holes at a  $M_{ic}$  of 1.03 or so. At higher Mach numbers, the effect of the lift coefficient on the position error pressure coefficient will be zero as the pressure field of the aircraft will not be felt in front of the bow wave. Therefore, any pressure error that does exist will be a function of Mach number only so that a plot of  $\Delta P_p/q_{cic}$  versus  $M_{ic}$  will be valid for all altitudes. In the usual case, this error is quite small and may be zero.

#### 5.5 CORRELATION OF RESULTS OF POSITION ERROR CALIBRATIONS

In the position error calibration methods discussed in the next section, data is usually obtained in the form of  $\Delta H_{pc}$  or  $\Delta V_{pc}$  for the altitude at which the test was conducted. In this section, methods by which data from different calibrations can best be correlated is given. The final report presentation is usually given as  $\Delta H_{pc}$  and  $\Delta V_{pc}$  versus  $V_{ic}$ with  $H_{ic}$  as the parameter. This can be done for both light weight and heavy weight configurations if weight is an important parameter.

For low speeds in which there are no Mach number effects, the position error obtained from several calibrations is best correlated by the use of a plot of  $\Delta V_{pc}$  versus  $V_{ic}$ . Such a plot will be a single line which is good for all altitudes for a constant nW with the absence of  $M_{ic}$  effects.

It has been shown that in the low Mach number range

$$\frac{\Delta P_{p}}{9cic} = f_{1} (V_{ic}) \text{ only} \qquad 5.70$$

for constant nW. From Section 5.3.3

$$\Delta V_{pc} = f_2(\frac{\Delta P_p}{q_{cic}}, V_{ic}) \qquad 5.38, 5.39$$

Therefore, in the absence of Mach number effects

$$\Delta V_{pc} = f_3(V_{ic}) \text{ only} \qquad 5.78$$

for constant nW

At higher speeds, when there is the possibility of both  $M_{ic}$  and  $C_{Lic}$  effects, the results of calibrations are best correlated by a plot of  $\Delta P_p/q_{cic}$  or  $\Delta M_{pc}$  versus  $M_{ic}$ . It has been shown in the previous section that this will usually be a single line for  $M_{ic} > 0.6$  except for possible low speed  $nW/\delta_{ic}$  breakoffs.

## 5.6 CALIBRATION METHODS

The static pressure error can be determined by any method in which the indicated static pressure and the free stream static pressure are obtained at the same time. The indicated pressure is obtained by installing a sensitive aneroid such as an altimeter in the static pressure system to be calibrated. The free stream static pressure can be obtained directly from a measurement of the atmospheric pressure or indirectly from a measurement of airspeed, in which case the total pressure error must be known or assumed to be zero. The direct method is called an altimeter calibration. Some of the more common methods are; the tower fly-by, the pacer and aircraft fly-by, the altitude pressure comparison methods, and the trailing bomb method. The indirect method is called an airspeed calibration. Airspeed calibrations can be obtained by the speed course method and the pacer and radar methods when airspeeds are compared. In general, the accuracy of the altimeter calibration is far superior to the airspeed calibration as the altimeter is a relatively accurate instrument compared to the airspeed indicator concerning such things as hysteresis and repeatability. It will be shown, however, that at very low speeds an airspeed calibration may be superior.

The choice of a method will, in general, depend on the instrumentation available, the degree of accuracy required, and the speed and altitude range for which a calibration is desired. The most desirable method or combination of methods is one which requires a minimum of time, equipment and manpower to arrive at an accurate calibration over the entire speed and altitude range of the aircraft; it must be quick and inexpensive, yet reliable and complete. Several methods are discussed in this section with this in mind. Each method is described in detail. Then the advantages and disadvantages of each are discussed so that the reader may choose the method or combination of methods which best fulfills his need.

## 5.6.1 The Tower Fly-By Method(See Data Reduction Outline 7.5):

The tower fly-by is a low altitude method in which the altitude indicated by the aircraft pitot-static system is compared to the actual pressure altitude to determine the static pressure error. A theodolite is set up in a control tower or a tall building at a known distance from a line marked on a runway. The aircraft to be calibrated is flown at constant speed over this course as close to theodolite level as possible but at least one full wing span off the ground to be out of ground effect. As the aircraft passes the theodolite position, the pilot records altitude  $(H_i)$  and airspeed  $(V_i)$ ; the theodolite operator measures the vertical angle to the aircraft. The atmospheric pressure at the theodolite station is measured with an absolute pressure gage or altimeter, or static pressure and temperature are measured at the ground and reference level static pressure is computed on the basis of the standard temperature lapse rate. The true pressure altitude of the aircraft is determined by adding the physical difference in height between the theodolite and the aircraft to the pressure altitude at theodolite level.

 $H_c = H_c$  at theodolite level +  $\Delta h$  5.79 where  $\Delta h$  is determined from the theodolite reading. This operation is valid, even during extreme atmospheric conditions, as the pressure gradient will not vary from standard enough to cause appreciable error in the small height difference between the aircraft and the theodolite. This method is illustrated in the following figure.



Figure 5.7 Tower Fly-by Method

The tower fly-by method is limited to level flight speeds above stalling speed by a safe margin. The upper speed limit may be set by local restrictions prohibiting supersonic flight at or near ground level in a congested area.

The static pressure error can be determined with very good accuracy by the use of this method. At low speeds, however, any error in the measurement of the static pressure error becomes very important when converted to airspeed position error ( $\Delta V_p$ ) as evidenced by equation (5.35, 5.36). This effect is illustrated in Figure 5.8.



Indicated Airspeed Corrected for Instrument Error, V<sub>ic</sub> (knots)

# Figure 5.8 Plot of $\Delta P_p$ vs $\forall_{ic}$ Determined from Tower Fly-By Calibrations

At the Air Force Flight Test Center, with the use of conventional aircraft instrumentation and a "visual theodolite", this method is not used for speeds below 200 knots or so for this reason.

The tower fly-by method is very quick, requiring only a few minutes per point for the flight and manual data reduction. It is relatively inexpensive as 1 hour of flight time will cover adequately the speed range of the aircraft and no extensive equipment is necessary.

In an improvement of this technique, two ground stations may be used, one on each side of the lined course. This allows the aircraft to deviate from the runway without introducing error.

Que disadvantage of the tower fly-by method, as discussed above, is the hazard of flying at high speed near the ground. This hazard can be eliminated by the use of a modified system. In this method, a photograph is taken as the aircraft passes over a camera which is directed vertically upward from a position on the marked course. The tapeline altitude of the aircraft is then determined from the focal

length of the camera and the proportion of the size of the image on film to the true dimensions of the object. The static pressure at this altitude can be computed or determined by flying the test aircraft at a speed for which the pressure error is known. Good results have been obtained with the use of a conventional 35mm camera up to altitudes of 1000 feet. This method is discussed in the report, "Position Error Determination by Stadiametric Ranging with a 35mm Movie Camera," Technical Report No. 2-55, Test Pilot Training Division, U.S. Naval Air Test Center (Patuxent River, Maryland), June 24, 1955 by W.J. Hesse.

## 5.6.2 The Ground Speed Course Method (See Data Reduction Outline 7.6):

The ground speed course is another low altitude method which is especially good at low speeds. It is best used in conjunction with the tower fly-by method to obtain a low altitude position error calibration over the entire speed range. This is an "airspeed calibration" in that the error in airspeed is measured directly from which the static pressure error may be determined - providing the error in total pressure is known or can be assumed to be zero.

The aircraft to be calibrated is flown over a course of known length at a uniform speed and at constant altitude. True airspeed is obtained from time and distance data. Calibrated airspeed, calculated from true airspeed, is compared to the airspeed indicated by the aircraft pitot-static system to obtain the error in airspeed due to static pressure error. The conversion of  $V_t$  to  $V_c$  requires that both pressure altitude and free air temperature be known. The pressure altitude can be obtained by adding the pressure altitude corresponding to the ground atmospheric pressure to the estimated height of the aircraft above the ground. Instead of estimating the height of the aircraft above the ground, an iterative process can be used where the instrument corrected altimeter reading is first used to find position error. This position error can then be used to correct the altimeter reading and the process repeated. Ambient temperature is determined from indicated readings recorded in the aircraft. This method is described in Figure 5.9.



#### Figure 5.9

Ground Speed Course Method

The aircraft should be flown on reciprocal headings at each speed so that the effect of head and tail wind can be averaged out. The averaged ground speed is assumed to be true speed. The aircraft should be allowed to drift with the wind so that the adverse effect of cross wind can be eliminated. The test should not be conducted on a windy day for any shifting winds introduce error in true speed. The aircraft should be fairly well stabilized as the timing gives average speed. However, the holding of an exact speed is not critical. The speed course should be flown at least one wing span above the ground to be out of ground effect. This distance should be kept to a safe minimum, however, because of the need for an estimation of the aircraft height.

Theoretically, this method is good for all level flight speeds above the stalling speed of the aircraft. The accuracy obtained, however, is a function of the timing method and the length of the course and diminishes as speed increases. At high speeds, errors in time measurement may cause the error in airspeed to be obscured by errors in the measurement of true speed. Therefore, this method gives best results at low speeds and can be used at high speeds only if adequate timing equipment is used and the course is relatively long. The Air Force Flight Test Center maintains a ground speed course approximately 4 miles long. Time is kept with a stop watch operated by the pilot or by an aircraft observer. This course is not used for speeds above 250 knots.

The accuracy of the ground speed course is poor even in the low speed range. There is always a scatter of points due to timing errors, shifting winds and the estimate of temperature at aircraft height which is needed for calculation of true speed. However, the results obtained at low speeds are in general better than those obtained by the tower flyby method.

The ground speed course is inexpensive and very simple to maintain and operate. Each double point takes approximately 10 minutes for the flight and 10 minutes for manual data reduction.

A variation of the ground speed course is the photogrid method. The test is conducted in the same manner except that true speed is determined by means of a camera, a timer, and a calibrated grid installed in a control tower or other vantage point by a runway. As the aircraft passes the camera station, photographs are taken through the grid. The film record gives accurate speed and altitude of the aircraft. (See Figure 5.10.) This method can be used only when a low wind condition exists or when the wind direction is approximately parallel to the runway or the same errors will be introduced as when crabbing on a speed course.



Photogrid Method.

#### 5.6.3 The Pacer Method (See Data Reduction Outline 7.7);

The tower fly-by and ground speed course methods which have been discussed are good for low altitude calibrations. These calibrations may be extrapolated to higher altitudes as discussed in Section 5.4. However, such extrapolations are not always possible. Furthermore, any extrapolations that are made should be checked at altitude. Therefore, calibration methods are necessary by which an aircraft can be calibrated at altitude. One such method is the pacer method in which the test aircraft is calibrated against another previously calibrated aircraft called a pacer. This method is very useful when frequent routine calibrations of aircraft are required.

In the basic form of this method the test aircraft and pacer are flown side by side approximately one wing span apart to prevent aircraft pressure field interaction. When the aircraft are stabilized at the desired speed and altitude, the pilots read the airspeed and altitude simultaneously, or record the data on a photopanel. (See Figure 5.1L) A static pressure calibration can be obtained directly from a comparison of the altitudes or indirectly from a comparison of airspeeds. By comparing both altitude and airspeed readings a check can be made for error in the total pressure system. (An error in total pressure should be suspected if  $\Delta V_{pc}$  determined by a comparison of airspeeds is consistently greater than  $\Delta V_{pc}$  determined by a comparison of altimeter readings.) This procedure is followed for a series of speeds at a given altitude to determine the static pressure error as a function of airspeed for that altitude. In this form, the pacer method is limited to the altitude and speed capabilities of the reference aircraft.



The Pacer Method

It is possible to make calibrations at speeds greater than the speed capabilities of the reference aircraft by the use of a variation called the aircraft fly-by method. Here, the test aircraft flies past the pacer aircraft at the same altitude. With the pressure altitude known from the pacer calibration, the static pressure error may be obtained at any speed for that altitude. It has the advantage over the basic method in that it is faster as it is not necessary to stabilize the airspeed. In the aircraft fly-by method, it is necessary that both aircraft be at the same altitude. Any deviation may be estimated or the test aircraft may be photographed as it passes by. It is helpful if the pacer aircraft can lay a trail, for example a contrait, as a reference. This allows the test aircraft to accelerate up the reference trail with data being taken as it closes on the pacer gircraft and then either pass the pacer or decelerate back down the reference trail. The acceleration-deceleration technique has the advantage that with data taken both ways the effect of lag can be averaged out or shown to be negligible. This technique is very useful for obtaining data in the transonic region. It is possible to get data up to Mach 1.2 or so with a subsonic pacer. Use of a contrail provides very accurate data since the altitude of a contrail does not usually vary more than 20 or 30 feet within 2 miles of the source. One disadvantage is that persistent contrails are sometimes difficult to obtain. In this case the pacer should be equipped with a smoke generator capable of leaving a well defined trail. Use of a smoke generator is presently limited to non-afterburing operation, however, since the smoke from existing smoke generators is nearly dissapated by the jet exhaust in afterburning.

The calibration of the test aircraft is, of course, only as good as the pacer calibration. For this reason aircraft must be kept exclusively as pacers. They should be calibrated in flight and have their instruments recalibrated at least once a month to insure the accuracy of their calibrations.
The primary advantages of the pacer method over other altitude nethods are the simplicity of scheduling, testing and data reduction, the speed and accuracy with which results can be obtained and the fact that the pacer is not restricted to one geographical area. In short, the pacer method is more convenient.

The practicality of the pacer method as compared to other methods depends on how often calibrations are required. Unless calibrations are required relatively frequently the cost of maintaining aircraft solely as pacers is prohibitive. However, when frequent calibrations are required, the pacer method becomes very practical. In general, the cost of keeping the pacer in the air is offset by the reduction in flying time necessary to establish a calibration.

### 5.6.4 Altitude Pressure Comparison Methods Requiring Pressure Survey (See Data Reduction Outline 7.8):

In this method the position of the aircraft in flight is fixed in space by the use of a radar-theodolite system or a phototheodolite complex such as an Askania range. The static pressure error is determined by comparing the aircraft indicated altitude to the pressure altitude which is determined from the tapeline altitude by means of a pressure survey.

The pressure survey can be conducted in one of several ways:

1. The test aircraft can be tracked by the radar or phototheodolite equipment as it climbs through the required altitude range at a low speed for which the static pressure error is known. It is then flown through the surveyed region at higher speeds for which a calibration is desired. It is possible to use another aircraft which has previously been calibrated to make this pressure survey. In either case, it should be noted that a survey made using this technique can be no better than the original calibration.

2. A radiosonde ballon transmitting pressure measurements can be tracked to determine pressure as a function of tapeline allitude. Better accuracy can be obtained by the use of a modified pressure capsule which is more accurate.

3. Data from a radiosonde balloon transmitting temperature and pressure can be used to find temperature as a function of pressure and the relation between altitude and pressure deduced by integration.

$$H = -\frac{R}{G} \sum_{P_{aSL}}^{P_{a}} \left(\frac{T_{a}}{P_{a}}\right) (\Delta P_{a})$$
 1.17

This integration is discussed in "Mach Number Measurements and Calibrations During Flight at High Speeds and at High Altitudes Including Data for the D-558-II Research Airplane," NACA RM H55J18, 1956 (Confidential) by Brunn and Stillwell. With the use of the same equipment, this method gives much better results than does technique 2.

The results of the pressure survey are plotted as pressure or pressure altitude versus tapeline altitude. The test aircraft is then flown in the surveyed region, recording airspeed and altitude as the radar or phototheodolite records height. The true pressure altitude for each test point is determined from the radar height and pressure survey curve. (See Figure 5.12).



#### SURVEY

Radar Records: Height Calibrated Aircraft Pilot Records: Indicated Airspeed, Vi Indicated Altitude, Hi Then:  $H_c = H_i + \Delta H_{ic} + \Delta H_{pc}$ Results of Survey Plotted As Radar Height Pressure Altitude CALIBRATION Radar Records: Height Test Aircraft Pilot Records: Indicated Airspeed, V<sub>i</sub> Indicated Altitude, Hi Then:

 $H_c$  = Pressure Altitude at Radar Height on Survey Plot.

### Figure 5.12

Altitude Pressure Comparison Method Using Radar

When a phototheodolite system is used the aircraft is tracked from a series of stations with cameras to determine the position in space. The tapeline altitude is determined by triangulation. A minimum of two stations is required for a fix. It is desirable to obtain data from more stations to give additional fixes, which reduces the uncertainty of the measurement. The accuracy with which tapeline altitude can be obtained is very good. However, the overall accuracy of the pressure error determination is limited by that of the pressure survey. Because of the complexity of the tracking system, the data must be processed on a digital computer. Hence, the data reduction time is apt to be quite large. In addition, the process is quite expensive as it requires costly equipment, large crews to maintain and operate the equipment, and machine data reduction.

The tapeline altitude can be calculated from the elevation and slant range given by a radar-theodolite assembly. The data reduction for this type of installation is much less time consuming than that required by the above installation as one station gives all the necessary information. The radar unit will not give quite as accurate results as those which can be obtained with the photo-theodolite range but its accuracy can be at least as good as that of the pressure survey.

At the Air Force Flight Test Center, radar tracking has been found to be very satisfactory, reliable and relatively economical. Provided that the target carries a beacon, it can be tracked out to a slant range of nearly 100 miles. More refined information, obtained by using a bore-sight camera to correct for radar hunt in azimuth and elevation, can be obtained out to about 20 miles, Sepending on contrast and so on. The accuracy of the height data is comparable to that with which the associated pressure can be measured or computed, but good velocity data cannot be obtained unless the target is flying in a steady manner. This is illustrated in Figure 5.13 in which a time history from a typical radar calibration is given.  $A_{i}$ relatively low frequency hunt is apparent which prevents use of the data to deduce velocity unless quite a long record can be averaged. However, when the aim is to calibrate the static pressure source, the radar method is used because of its simplicity and economy. When a more precise trajectory is required, capable of yielding ground speeds directly, the Askania range is used. This gives substantial improvement in precision but cost and complication are much greater than those of a radar calibration. A minimum of four cameras is considered essential and six or more are used if the target is to be tracked over a distance of the order of that covered by the bore-sight radar.





These techniques are costly and tedious but they can be used in many situations where some of the less complicated methods fail. They permit calibrations in high speed dives and maneuvers, as well as in level flight, and allow calibration of rocket powered aircraft and missiles as long as they stay within the range of the tracking equipment.

5.6.5 The All-Altitude Speed Course:

The principle of the ground speed course can be used at altitude to determine the error in airspeed measurement in the aircraft pitot-static system. However, the establishment of an altitude speed course which will give comparable accuracy is difficult. It is necessary to use an elaborate timing device and electronic or optical means to establish the course length. The accurate measurement of temperature at altitude presents a problem. One must rely either on a previously calibrated free air temperature probe or the weather service and a pressure survey. Also the higher winds which usually exist at altitude cause considerable scatter of data. Therefore, the speed course is not recommended for altitude calibrations.

A certified speed course has been established at the AFFTC for the purpose of obtaining internationally accepted speed records. It is possible to use this course as a speed course to obtain position error. It is an optical course approximately 10 miles long at 35,000 feet. The overall accuracy of the speeds obtained are to the order of 0.15 percent. Therefore, there is little error in the measurement of ground speed by this method. However, the problems of conversion to true speed and calibrated airspeed and the determination of pressure altitude make the use of the course for this purpose quite impractical.

5.6.6 The Trailing Bomb Method:

In this method a static pressure source is built into a "bomb" which is suspended on a long cable and allowed to trail well below and aft of the aircraft so as to be out of the aircraft pressure field and thus to record free stream static pressure. This is compared to the indicated static pressure (or altitude) to give the static pressure error. The pressures from the two sources may be connected by means of a sensitive differential pressure gage to give the pressure error directly. Hence the accuracy can be very good as long as the trailing bomb is out of the aircraft pressure field.

At low speeds the weight of the bomb is enough to keep it below the test aircraft. At higher speeds, say above 200 knots indicated, the bomb must be fitted with small wings set at a negative angle of attack to keep it out of the slipstream of the aircraft. This, however, introduces the instability problems of a towed glider.

This method is good at stalling speed as long as the downwash at high angles of attack does not cause instability. The upper limit in speed is the speed at which the system encounters high speed instability. It is believed that this high speed instability is due to cable oscillations which originate near the aircraft and are amplified by aerodynamic forces as they travel down the cable.

Years ago when aircraft were not capable of high speeds this was a very popular method. In the case of modern aircraft, this method has lost its popularity because of the high speed instability problem. It is still sometimes used, however, for the calibration of low speed aircraft such as transports.

# SECTION 6 CALIBRATION OF THE FREE AIR TEMPERATURE INSTRUMENTATION

### **6.1 INSTRUMENT ERROR**

WARRAN SIDE MORE THANKS TO AND A REAL OF

The major errors in the temperature indicating system are a result of variation of the resistance temperature coefficient in the sensing element and electrical defects in the bridge circuit and ammeter. Errors caused by the sensing elements are minimized by the selection of good quality elements from the manufacturer's lots. The standard tolerance is  $\pm 2$  degrees C; however, units having a maximum error of  $\pm 0.5$  degrees C are selected for flight test work. A laboratory calibration is conducted to determine errors in the bridge circuit and indicator. The instrument is calibrated every 2 degrees C over the temperature range of anticipated use. A typical calibration plot is shown in Figure 6.1.



### Figure 6.1



The indicated temperature corrected for instrument error  $(T_{ic})$  is obtained from this curve, as

$$T_{ic} = T_i + \Delta T_{ic} \qquad 6.1$$

where

 $T_i$  = indicated temperature  $\Delta T_{ic}$  = free air temperature instrument correction corresponding to  $T_i$ 

#### 6.2 DETERMINATION OF THE TEMPERATURE PROBE RECOVERY FACTOR

The equation for the free air temperature probe was derived in Section 2.4 as

$$\frac{T_{ic}}{T_{a}} = 1 + \frac{KM^{2}}{5}$$
 2.30

where

Ta	=	free air temperature, *K
T <sub>ic</sub>		indicated temperature corrected for instrument error, *K
М		free stream Mach number
К	*	temperature probe recovery factor

Values of K should be determined in flight as they depend on the installation. Usually at subsonic speeds variation in K with Mach number and altitude is not significant. At supersonic speeds, however, where temperature rises are much larger, variations in K may exist which must be well established in order that ambient temperatures may be calculated accurately. It is considered advisable to determine values of K throughout the speed range at a high and a low altitude to investigate possible variations in K. This is quite frequently done in conjunction with one or more of the airspeed calibration methods described in Section 5. Several techniques are discussed in the following paragraphs by which temperature probe recovery factors can be determined. The use of radiosonde temperatures in lieu of data derived from free air temperature instruments is also considered.

1. The aircraft is flown at a series of Mach numbers and the data is plotted as K versus M where

$$K = \left(\frac{T_{ic}}{T_{a}} - 1\right) \frac{5}{M^{2}}$$
 6.2

This data is readily obtained in conjunction with airspeed calibrations when a pacer aircraft is used, since ambient temperatures can be obtained with a calibrated probe. The results of a typical calibration are given in Figure 6.2.





Indicated temperatures, recorded while conducting speed-power tests, together with radiosonde temperatures may be used conveniently to make a similar presentation. Care must be taken in this case to avoid systematic errors in ambient temperature measurements.

2. The aircraft is flown at a series of speeds at a constant pressure altitude. It is therefore necessary to prepare tables in advance showing altimeter reading  $(H_i)$  at which the aircraft is to be flown for each airspeed.

$$H_i = H_c - \Delta H_{ic} - \Delta H_{pc} \qquad 6.3$$

where

. . . . . . . . . . . . . . . . . . .

 $\Delta H_{ic}$  = altimeter instrument correction corresponding to  $H_i$ 

$$\Delta H_{pc}$$
 = altimeter position error correction corresponding to  $H_{ic}$ 

The results are plotted as  $l/T_{ic}$  versus  $M^2/T_{ic}$ . When this is done, the slope of a line faired through the data is equal to (-K/5) as

$$\frac{1}{T_{ic}} = \frac{1}{T_a} - \frac{K}{5} \frac{M^2}{T_{ic}}$$

$$6.4$$

The intercept on the  $1/T_{ic}$  axis is  $1/T_a$ . Repeat tests made at different air temperatures will give a series of parallel straight lines if K is a constant for the installation. Figure 6.3 shows a plot where runs have been made at two altitudes.



Plot of  $\frac{1}{T_{ic}}$  versus  $M^2/T_{ic}$  Used to Determine the Temperature Probe Recovery Factor, K

This method has the advantage that K can be determined independently of  $T_a$ , although it is essential that it remains constant. If  $T_a$  is known, say from radiosonde data, it can be used to help establish the slope of the line.

3. Recovery factors can also be determined in conjunction with airspeed calibrations made with the tower fly-by method. In this case a very nearly constant pressure altitude is maintained during each pass by the tower. By recording temperature in the tower and in the aircraft for each pass, the value of K can be established using either of the presentations described in the preceeding paragraphs. This method assumes that temperatures recorded in the tower are the same as the ambient temperatures at the probe located on the aircraft. Errors may be incurred if the aircraft is flown higher than the tower, which it usually is, and a pronounced temperature gradient exists. Tower fly-bys are best made during the early morning, however, when the air is most stable near the ground and temperature gradients are small.

4. The speed course method of obtaining airspeed calibrations also yields data from which values of K can be computed. From Section 2.52

$$M = \frac{V_{tt}}{38.967 \sqrt{T_{at}}}$$
 2.36

Substituting this expression into equation 2,30

$$T_{ic} = T_a + KV_t^2$$
 6.5

where  $V_t$  is in knots and  $T_{ic}$  and  $T_a$  are in <sup>6</sup>K. The results are plotted as  $T_{ic}$  versus  $V_t$ . Then, the slope of a line faired through the data is equal to (+K/7592). The intercept on the  $T_{ic}$  axis is  $T_a$ . (See Figure 6.4)



Plot of  $t_{ic}$  vs  $V_t^2$  Used to Determine the Temperature Probe Recovery Factor, K

 $T_a$  can be used with this method also as an aid in establishing the slope of the line through the test points. It is necessary, however, to have low wind conditions or a considerable error in  $V_t^2$  and hence K may result.

When the value of K is established, free air temperature is most easily determined from equation 2.30. Chart 8.2 has been included in Section 8 to facilitate this operation.

Temperature probe recovery factors for supersonic flight may be determined from the methods described in paragraphs 1 and 2 above. Supersonic pacer aircraft with well established probe calibrations for the flight conditions obtained with high speed test aircraft are not generally available, however, the method described in paragraph 2, where the test aircraft is flown at constant pressure altitude, may be used but additional flight time will probably be required to define K values satisfactorily. Consequently, the use of radiosonde data is likely to be best at supersonic speeds. Recommended temperature accuracies of radiosondes listed in Air Weather Service TR105-133 are

 $\frac{1}{2} + 1.5^{\circ}C \text{ from } + 40^{\circ}C \text{ to } - 50^{\circ}C \text{ to } + 2.0^{\circ}C \text{ from } - 50^{\circ}C \text{ to } - 70^{\circ}C \text{ to } + 3.0^{\circ}C \text{ from } - 70^{\circ}C \text{ to } - 90^{\circ}C \text{ to } - 90^{\circ}C \text{ to } + 3.0^{\circ}C \text{ from } - 70^{\circ}C \text{ to } - 90^{\circ}C \text{ to } + 3.0^{\circ}C \text{ from } - 70^{\circ}C \text{ to } - 90^{\circ}C \text{ to } - 90^{\circ}C \text{ to } + 3.0^{\circ}C \text{ from } - 70^{\circ}C \text{ to } - 90^{\circ}C \text{ to } -$ 

These values were recommended as representing reasonable accuracies to be expected of the temperature data obtained from radiosondes used by the various United States meteorological services. For most accurate results, ambient temperatures from radiosonde data should be based on three or more soundings obtained from stations surrounding the area in which the test aircraft is flown. These soundings should be made within 2 or 3 hours of the time test data is taken. Also, it is best to examine the most recent weather charts prior to flight so that possible frontal passages with significant temperature differences may be avoided.

## SECTION 7

### DATA REDUCTION OUTLINES

# 7.1 CORRECTION OF AIRSPEED INDICATOR AND ALTIMETER FOR PRESSURE LAG DURING CONSTANT CLIMB, CONSTANT DESCENT, AND/OR ACCELERATION (See Section 4.2)

.

-- . •. .

1	λ <sub>sSL</sub>	sec	Altimeter Lag Constant at Standard Sea Level, from previous calibration
2	λ <sub>tSL</sub>	sec	Total Pressure Lag Constant at Standard Sea Level, from previous calibration
3	V <sub>ic</sub>	knots	Indicated Airspeed Corrected for Instrument Error
4	H <sub>ic</sub>	feet	Indicated Altitude Corrected for Instrument Error
5	dH <sub>ic</sub> /dt	ft/min	Apparent Rate of Climb (+) or Descent(-), from time history of (4) or from R/C indicator
6	dV <sub>ic</sub> /dt	kt/sec	Apparent Acceleration (+) or Deceleration (-), from time history of (3)
7	ta	°C	True Atmospheric Temperature, from t <sub>ic</sub> and K and M and Chart 8.2 or from weather service.
8	$\lambda/\lambda H_{ic}$		Lag Constant Temperature Correction, from (4) and (7) and Chart 8.62
9	$\lambda_{sH_{ic}}/\lambda_{sSL}$		Static Pressure Lag Constant Ratio, from $(4)$ and Chart 8.61 for $V_{ic}$ = STATIC
10	$^{\lambda}$ tH <sub>ic</sub> / $^{\lambda}$ tSL		Total Pressure Lag Constant Ratio, from (4) and (3) and Chart 8.61
11	λ <sub>s</sub>	sec	Altimeter Lag Constant, (1) x (8) x (9)
12	λ <sub>t</sub>	вес	Total Pressure Lag Constant, $(2) \times (8) \times (10)$
13	ΔH <sub>icl</sub>	feet	Altimeter Lag Correction, (11) x $(5)$ + 60

14	$F_1(H_{ic}, V_{ic})$		Airspeed Indicator Lag Factor, from (3) and (4) and Chart 8.63
15	ΔV <sub>icl</sub>	knots	Airspeed Indicator Lag Correction, (12) x (6) + $(11) - (12)$ x (14) x (5) $\div 60$
16	H <sub>ic!</sub>	feet	Indicated Pressure Altitude Corrected for Instrument and Lag Error, $(4) + (13)$
17	V <sub>icl</sub>	knots	Indicated Airspeed Corrected for In- strument and Lag Error, (3) + (15)

# 7.2 LABORATORY CALIBRATION FOR THE STATIC PRESSURE LAG CONSTANT (See Section 4.3)

1			Counter Number
2	t	sec	Time
3	T <sub>a</sub>	•K	Room Temperature, °C + 273.16
4	P <sub>sl</sub>	" Hg	Probe Enclosure Static Pressure Gage Reading
5	Ps	"Hg	Aircraft Static Pressure Gage Reading
6	the time diffe	erence betv	(2) on one graph. (At any pressure coordinate, ween (4) and (5) is the static pressure lag are. This lag will decrease as pressure
7	λε	sec	Static Pressure Lag Constant for any Pressure, from (6)
7 8	$\lambda_{g}$ $P_{s}/P_{aSL}$	sec	

Case a: When pressure gages are used

Gase of strengtherer o are about	Cäse	b:	When	altimeters	are used
----------------------------------	------	----	------	------------	----------

1			Counter Number
2	t	sec	Time
3	Τ <sub>a</sub>	°K	Room Temperature, °C + 273.16
4	H <sub>il</sub>	feet	Probe Enclosure Altimeter Reading
5	ΔH <sub>ic</sub>	fect	Probe Enclosu, Altimeter Instrument Correction Corresponding to (4)
6	H <sub>icl</sub>	feet	Probe Enclosure Simulated Pressure Altitude, Corrected for Instrument Error, (4) + (5)
7	H <sub>i</sub>	feet	Aircrast Altimeter Reading
8	ΔH <sub>ic</sub>	feet	Aircraft Altimeter Instrument Correction Corresponding to (7)
9	H <sub>ic</sub>	feet	Aircraft Indicated Pressure Altitude Corrected for Instrument Error, (7) + (8)
10	the time dif	ference betw	(2) on one graph. (At any altitude coordinate, ween (6) and (9) is the static pressure lag e. This lag will increase as altitude increases.)
11	λ <sub>s</sub>	sec	Static Pressure Lag Constant for any Altitude, from (10)
12	P <sub>s</sub> /P <sub>aSL</sub>		Pressure Ratio at the Pressure used to Determine (11)
13	λsSL	sec	Sea Level Static Pressure Lag Constant, (12) $\times$ (11) $\times$ 288.16 (3)

## 7.3 LABORATORY CALIBRATION FOR THE TOTAL PRESSURE LAG CONSTANT (See Section 4.3)

Case a: When pressure gages are used

1			Counter Number		
2	t	sec	Time		
3	Ta	۴K	Room Temperature, °C + 273.16		
4	P <sub>t'1</sub>	" Hg	Probe Enclosure Total Pressure Gage Reading		
5	Pť	" Hg	Aircraft Total Pressure Gage Reading		
6	Plot (4) and (5) versus (2) on one graph. (At any pressure coordinate, the time difference between (4) and (5) is the total pressure lag constant for that pressure. This lag will decrease as pressure increases.)				
7	λ <sub>t</sub>	8ec	Total Pressure Lag Constant for any Pressure, from (6)		
8	Pt'/PaSL		Pressure Ratio at the Pressure Used to Determine (7), (5)/29.92		
9	λ <sub>tSL</sub>	sec	Sea Level Total Pressure Lag Constant,		

# Case b: When airspeed indicators are used

----

1			Counter Number
2	t	sec	Time
3	T <sub>a</sub>	•K	Room Temperature, °C + 273,16
4	Pa	" Hg	Room Ambient Pressure
5	v <sub>il</sub>	knots	Probe Enclosure Airspeed Indicator Reading
6	ΔV <sub>ic</sub>	knots	Probe Enclosure Airspeed Indicator Instru- ment Correction Corresponding to (5)
7	v <sub>icl</sub>	knots	Probe Enclosure Indicated Airspeed Corrected for Instrument Error, (5) + (6)

8	v,	knots	Aircraft Airspeed Indicator Reading			
9	$\Delta V_{ic}$	knots	Aircraft Airspeed Indicator Instrument Correction Corresponding to (5)			
10	V <sub>ic</sub>	knots	Aircraft Indicated Airspeed Corrected for Instrument Error, (8) + (9)			
11	Plot (7) and (10) versus (2) on one graph. (At any airspeed coordinate, the time difference between (7) and (10) is the total pressure lag constant for that airspeed. This lag will decrease as airspeed increases.)					
12	λ <sub>t</sub>	sec	Total Pressure Lag Constant for any Airspeed, from (11)			
13	9 <sub>cic</sub>	"Hg	Differential Pressure Corresponding to the Airspeed Used to Determine (12), from Table 9.6			
14	P <sub>t</sub> '	"Hg	Total Pressure Corresponding to the Airspeed Used to Determine (12), (4) + (13)			
15	P <sub>t</sub> '/P <sub>aSL</sub>		Pressure Ratio at the Airspeed Used to Determine $(12)$ , $(14)/29.92$			
16	<sup>λ</sup> tSL	sec	Sea Level Total Pressure Lag Constant, (12) x (15) x $288.16$ (3)			

## 7.4 PRESENTATION OF RESULTS OF POSITION ERROR CALIBRATIONS AND EXTRAPOLATION PROCEDURES (See Section 5.4)

A position error calibration is usually conducted at a series of speeds for a given altitude. From this data it is possible to determine  $\Delta V_{pc}$ (and/or  $\Delta P_p/q_{cic}$ ) for a series of  $V_{ic}$  (or  $M_{ic}$ ) at a given  $H_{ic}$  as shown in Data Reduction Outlines 7.5, 7.6, 7.7 and 7.8. This information should be plotted in accordance with the following outline:



For  $M_{ic} > 0.6$ , plot  $\Delta P_p / q_{cic}$  versus  $M_{ic}$  for  $H_{ic}$ 



The plot of  $\Delta V_{pc}$  versus  $V_{ic}$  is good for all altitudes for which there are no Mach number effects. (Mach number effects will appear as altitude breakoffs at the high speed end of the curve.) The plot of  $\Delta P_p / q_{cic}$  versus  $M_{ic}$  is good for all altitudes for which there are no  $C_{Lic}$  (nW/ $\delta_{ic}$ ) effects. (Such effects will appear as nW/ $\delta_{ic}$  breakoffs, usually at the low speed end of the curve.) A check at a second altitude should be made to see if there are any such altitude breakoffs. The fellowing typical result may be obtained.



In the final report, the position error is usually plotted as  $\Delta H_{pc}$  and  $\Delta V_{pc}$ versus  $V_{ic}$  for constant  $H_{ic}$ . For each  $H_{ic}$  for which such plots are desired:

1	H <sub>ic</sub>	feet	Indicated Altitude Corrected for Instrument Error for which plot is desired.
2	V <sub>ic</sub>	knots	Arbitrary Indicated Airspeed Corrected for Instrument Error.
3	M <sub>ic</sub>		Indicated Mach Number Corrected for Instrument Error, from (1) and (2) and Chart 8.5.
4	$\frac{\Delta P_{p}}{q_{cic}}$		Position Error Pressure Coefficient, from plot of $\Delta P_p/q_{cic}$ versus $M_{ic}$ for (3) and (1).
5	Δv <sub>pc</sub>	knots	Airspeed Indicator Position Error Correction, from (2) and (4) and Chart 8. 11 or from plot of $\Delta V_{pc}$ versus $V_{ic}$ for (1) and (2).
6	ΔH <sub>pc</sub>	feet	Altimeter Position Error Correction, f. (5) and (2) and (1) and Chart 8.13. (For small errors, say $\Delta V_{pc} \leq 10$ knots, the approximate Chart 8.12 may be used.)
7	Plot 2	V and	$\Delta H_{pc}$ versus $V_{ic}$ for $H_{ic}$ . Repeat for other $H_{ic}$ .*

\* In the case of low speed aircraft in which there are no  $M_{ic}$  effects or high speed aircraft in which there are no  $n W / \delta_{ic}$  effects, the curve of  $\Delta H_{pc}$  versus  $V_{ic}$  for one altitude can be extrapolated to other altitudes by the following procedure:

Case a: Low Speed Aircraft (No M<sub>ic</sub> effect)

1	H <sub>icl</sub>	feet	Indicated Altitude Corrected for Instru- ment Error Corresponding to an Arbitrary V <sub>ic</sub>
2	۵H <sub>pcl</sub>	feet	Altimeter Position Error Correction Corresponding to (1)
3	σ		Density Ratio Corresponding to (1), from Table 9.2
4	H <sub>ic2</sub>	feet	Arbitrary Indicated Altitude Corrected for Instrument Error
5	σ <sub>2</sub>		Density Ratio Corresponding to (4), from Table 9.2
6	ΔH <sub>pc2</sub>	feet	Altimeter Position Error Correction Corresponding to (4) for same $V_{ic}$ as (1), (2) x (3) $\div$ (5)

Case b: High Speed Aircraft (No  $nW/\delta_{ic}$  effect)

का राज्य जाता.

 $\pm 12$ 

1	H <sub>icl</sub>	feet	Indicated Altitude Corrected for Instru- ment Error Corresponding to an Arbitrary Vic.
2	∆H <sub>pc1</sub>	feet	Altimeter Position Error Correction Corresponding to (1)
3	Tasl	•к	Air Temperature Corresponding to (1), from Table 9.2
4	H <sub>ic2</sub>	feet	Arbitrary Indicated Altitude Corrected for Instrument Error
5	Tao2	•к	Air Temperature Corresponding to (4), from Table 9.2
6	ΔH <sub>pc2</sub>	feet	Altimeter Position Error Correction Corresponding to (4) for same $V_{ic}$ as (1), (2) x (5) $\div$ (3)

1			Pass Number
2			Time of Day
3			Theodolite Reading
4	Δh	feet	Aircraft Height Above (+) or Below (-) Theodolite Reference Altitude, from (3)
5	Pa	" Hg	Pressure at Reference Altitude, from weather service or theodolite altimeter set at 29.92 and Corrected for Instrument Error. (If altimeter is used (6) below is obtained directly.)
6		feet	Theodolite Reference Pressure Altitude, from (5) and standard atmosphere (Table 9.2)
7	H <sub>c</sub>	feet	Pressure Altitude of Aircraft, (6) + (4)
8	H <sub>i</sub>	feet	Aircraft Indicated Altitude
9	ΔH <sub>ic</sub>	feet	Aircraft Altimeter Instrument Correction Corresponding to (8)
10	H <sub>ic</sub>	feet	Indicated Pressure Altitude Corrected for Instrument Error, (8) + (9)
11	∆н <sub>рс</sub>	feet	Aircraft Altimeter Position Error Correction (7) - (10)
12	v <sub>i</sub>	knots	Aircraft Indicated Airspeed
13	ΔVic	knots	Airspeed Indicator Instrument Correction Corresponding to (12)
14	V <sub>ic</sub>	knots	Indicated Airspeed Corrected for Instru- ment Error, (12) + (13)
15	M <sub>ic</sub>		Indicated Mach Number Corrected for Instrument Error, from (10) and (14) and Chart 8, 5

# 7.5 THE TOWER FLY-BY METHOD (See Section 5.6.1)

8 128 12 1 14

72 C E 7

16	∆v <sub>pc</sub>	knots	Airspeed Indicator Position Error Correction, from (10) and (11) and (14) and Chart 8.13. (For small errors, say $\Delta H_{pc} < 1000$ feet, the approximate Chart 8.12 can be used)
17	ΔP qcic		Position Error Pressure Coefficient, from (14) and (16) and Chart 8.11. (This must be determined only for $M_{ic} > 0.6$ or so.)

Note: For presentation of results and exchapolation, see Data Reduction Outline 7.4.

7.6	THE GROUND	SPEED	COURSE	METHOD	(See	Section	5.	6. i	2)
-----	------------	-------	--------	--------	------	---------	----	------	----

1			Pass Number
2		feet	Course Length
3	Δt <sub>1</sub>	sec	Time Across
4	∆t <sub>2</sub>	sec	Time Back
5	v <sub>g1</sub>	ft/sec	Ground Speed Across, (2) + (3)
6	v <sub>g2</sub>	ft/sec	Ground Speed Back, (2) ÷ (4)
7	v <sub>t</sub>	ft/sec	True Speed or Average Ground Speed, (5) + (6) $\frac{4}{3}$ 2
8	v <sub>t</sub>	knots	True Speed, 0.5921 x (7)
9	Δh	feet	Estimated Height of the Aircraft Above the Ground
10	ta	•C	Atmospheric Temperature at Aircraft Height, Ground Temperature - 0.00198 (9)
11	М		Mach Number, from (8) and (10) and Chart 8,4
12	v <sub>i</sub>	knots	Indicated Airspeed

13	ΔV <sub>ic</sub>	knots	Airspeed Indicator Instrument Correction Corresponding to (12)
14	V <sub>ic</sub>	knots	Indicated Airspeed Corrected for Instru- ment Error, (12) + (13)
15	H <sub>i</sub>	feet	Indicated Altitude
16	ΔH <sub>ic</sub>	feet	Altimeter Instrument Correction Corresponding to (15)
17	H <sub>ic</sub>	feet	Indicated Altitude Corrected for Instru- ment Error, (15) + (16)
18	M <sub>ic</sub>		Indicated Mach Number Corrected for Instrument Error, from (14) and (17) and Chart 8.5
19	ΔM <sub>pc</sub>		Machmeter Position Error Correction, (11) - (18)
20	$\frac{\Delta P_p}{q_{cic}}$		Position Error Pressure Coefficient, from (19) and (18) and Chart 8.18. (For small errors, say $\Delta M_{pc} \leq 0.04$ , the approximate Chart 8.17 may be used.)
21	۵۷ <sub>pc</sub>	knots	Airspeed Indicator Position Error Correction from (14) and (20) and Chart 8.11

Note: For presentation of results and extrapolation, see Data Reduction Outline 7.4

# 7.7 THE PACER METHOD (See Section 5.6.3)

Case a: The Stabilized Flight Method

1			Pass Number
2	н <sub>с</sub>	feet	Pressure Altitude, pacer $H_i + \Delta H_{ic} + \Delta H_{pc}$
3	v <sub>c</sub>	knots	Calibrated Airspeed, pacer $V_i + \Delta V_{ic} + \Delta V_{pc}$

4	Hi	feet	Test Aircraft Indicated Altitude
5	ΔH <sub>ic</sub>	feet	Test Aircraft Altimeter Instrument Correction Corresponding to (4)
6	H <sub>ic</sub>	feet	Test Aircraft Indicated Altitude Corrected for Instrument Error, $(4) + (5)$
7	дн <sub>рс</sub>	feet	Test Aircraft Altimeter Position Error Correction (2) $_{-}$ (6)
8	v <sub>i</sub>	knots	Test Aircraft Indicated Airspeed
9	ΔV <sub>ic</sub>	knots	Test Aircraft Airspeed Indicator Instrument Correction Corresponding to (8)
10	Vic	knots	Test Aircraft Indicated Airspeed Corrected for Instrument Error, (8) + (9)
11	ΔV <sub>pc</sub>	knot s	Test Aircraft Airspeed Indicator Position Error Correction, (3) $\_$ (10) or from (7) and (6) and (10) and Chart 8.13. (For small errors, say $\Delta H_{pc} \leq 1000$ feet, the approxi- mate Chart 8.12 may be used.)

Note: With the altimeter and airspeed indicator systems both using the same static source,  $\Delta H_{pc}$  and  $\Delta V_{pc}$  are related according to Chart 8, 12 or 8.13. In the pacer calibration  $\Delta H_{pc}$  and  $\Delta V_{pc}$  are both determined independently; hence one of the values is redundant. The altimeter is a much more reliable instrument than the airspeed indicator regarding such things as repeatability and hysteresis so, in general, it is best to rely on the calibrated  $\Delta H_{pc}$  and calculate  $\Delta V_{pc}$ . It is possible, however, for low airspeeds and low altitudes that  $\Delta V_{pc}$  may give better results.

12	M <sub>ic</sub>	Indicated Mach Number Corrected for Instrument Error, from (6) and (10) and Chart 8.5
13	$\frac{\Delta P_p}{q_{cic}}$	Position Error Pressure, Coefficient, from (10) and (11) and Chart 8.11. (This step is necessary only for $M_{ic} > 0.6$ or so.)

# Note: For presentation of results and extrapolation, see Data Reduction Outline 7.4

1	·		Pass Number
2	Н <sub>с</sub>	feet	Pressure Altitude, pacer $H_i + \Delta H_{ic} + \Delta H_{pc}$
3	H <sub>i</sub>	feet	Test Aircraft Indicated Altitude
4	ΔH <sub>ic</sub>	feet	Test Aircraft Altimeter Instrument Correction Corresponding to (3)
5	H <sub>ic</sub>	feet	Test Aircraft Indicated Altitude Corrected for Instrument Error, $(3) + (4)$
6	∆H <sub>pc</sub>	feet	Test Aircraft Altimeter Position Error Correction, $(2)_{-}(5)$
7	vi	knots	Test Aircraft Indicated Airspeed
8	$\Delta V_{ic}$	knots	Test Aircraft Airspeed Indicator Instru- ment Correction Corresponding to (7)
9	Vic	knots	Test Aircraft Indicated Airspeed Corrected for Instrument Error, (7) + (8)
10	ΔV <sub>pc</sub>	knots	Test Aircraft Airspeed Indicator Position Error Correction, from (6) and (5) and (9) and Chart 8.13. (For small errors, say $\Delta H_{pc} \angle 1000$ feet, the approximate Chart 8.12 may be used.)
11	M <sub>ic</sub>		Indicated Mach Number Corrected for Instrument Error, from (5) and (9) and Chart 8.5
12	$\frac{\Delta \mathbf{P}_{\mathbf{p}}}{\mathbf{q}_{\mathrm{cic}}}$		Position Error Pressure Coefficient, from (9) and (10) and Chart 8.11. (This step is necessary only for $M_{ic} > 0.6$ or so.)

Case b: The Aircraft Fly-By Method

. . . . . . .

# Note: For presentation of results and extrapolation see Data Reduction Outline 7.4



7.8 ALTITUDE PRESSURE COMPARISON METHODS REQUIRING PRESSURE

12	ΔV <sub>pc</sub>	knots	Airspeed Indicator Position Error Correction, from (7) and (6) and (10) and Chart 8.13. (For small errors, say $\Delta H_{pc} < 1000$ feet, the approximate Chart 8.12 may be used.)
13	$\frac{\Delta P_{p}}{q_{cic}}$		Position Error Pressure Coefficient, from (10) and (12) and Chart 3.11. (This step is necessary only for M <sub>ic</sub> >0.6 or so.)

- Note: For presentation of results and extrapolation see Data Reduction Outline 7.4.
- 7.9 CALIBRATION FOR TEMPERATURE PROBE RECOVERY FACTOR (See Section 6.2)

1			Pass Number
2	H <sub>i</sub>	feet	Altimeter Reading
3	ΔH <sub>ic</sub>	feet	Altimeter Instrument Correction Corres- ponding to (2)
4	H <sub>ic</sub>	feet	Indicated Pressure Altitude Corrected for Instrument Error, $(2) + (3)$
5	∆н <sub>рс</sub>	feet	Altimeter Position Error Correction Corresponding to (4)
6	H <sub>c</sub>	feet	True Pressure Altitude, (4) + (5)
7	v <sub>i</sub>	knots	Airspeed Indicator Reading
8	ΔV <sub>ic</sub>	knots	Airspeed Indicator Instrument Correction Corresponding to (7)
9	V <sub>ic</sub>	knots	Indicated Airspeed Corrected for Instrument Error, (7) + (8)
10	ΔV <sub>pc</sub>	knots	Airspeed Indicator Position Error Correc- tion Corresponding to (9)

Case a: K determined from plot of K versus M

11	v <sub>c</sub>	knots	Calibrated Airspeed, (9) + (10)
12	м		Free Stream Mach Number, from (6) and (11) and Chart 8.5
13	ti	•c	Temperature Probe Reading
14	∆t <sub>ic</sub>	°C	Temperature Probe Instrument Correc- tion Corresponding to (13)
15	t <sub>ic</sub>	°C	Indicated Temperature Corrected for Instrument Error, (13) + (14)
16	<sup>T</sup> ic	۰ĸ	Indicated Temperature Corrected for Instrument Error, (15) + 273.16
17	Ta	°K	Ambient Temperature, from previously calibrated probe or from weather service
18	$T_{ic}/T_{a}$		$(16) \div (17)$
19	ĸ		Temperature Probe Recovery Factor, $5 [(18) - 1] + (12)^2$
20		versus (12) and K. Soe Figure	d fair a line through the points giving an average e 6.2
21	К		Temperature Probe Recovery Factor, from plot of (20)

Case b: K determined from plot of  $1/T_{ic}$  versus  $M^2/T_{ic}$ 

1 to 16 as in case "a" above

17	1/T <sub>ic</sub>	1/*K	1/(16)
	M <sup>2</sup>		(12) <sup>2</sup>
19	M <sup>2</sup> /T <sub>ic</sub>	1/•К	(18) x (17)

20 Plot (17) versus (19) and fair a straight line through the points. (If  $T_a$  is known it may be plotted as  $1/T_a = 1/T_{ic}$  at  $M^2 = 0$  and be used to fair in the line.) See Figure 6.3

21	m	Slope of the straight line of (20). For any two points on the line, 1 and 2, the slope is equal to:
		$m = \frac{\left(\frac{1}{T_{ic}}\right)_2 - \left(\frac{1}{T_{ic}}\right)_1}{\left(\frac{M^2}{T_{ic}}\right)_2 - \left(\frac{M^2}{T_{ic}}\right)_1}$
22	К	Temperature Probe Recovery Factor, _ 5 x (21)

an a seren fra de la seconda de seconda

Case c: K determined from plot of  $T_{ic}$  versus  $V_t^2$  (Speed Course Method)

1		¦	Pass Number
2		feet	Course Length
3	Δt <sub>1</sub>	sec	Time Across
4	Δt <sub>2</sub>	sec	Time Back
5	v <sub>t1</sub>	ft/sec	Ground Speed Across, (2) ÷ (3)
6	v <sub>t2</sub>	ft/sec	Ground Speed Back, (2) ÷ (4)
7	v <sub>t</sub>	ft/sec	True Airspeed, (assumed to be Average Ground Speed,) $[(5) + (6)] \div 2$
8	v <sub>t2</sub>	knots	True Airspeed, 0.5921 x (7)
9	v <sub>t</sub>	knots <sup>2</sup>	(8) <sup>2</sup>
10	ti	•C	Temperature Probe Reading
11	∆t <sub>ic</sub>	°C	Temperature Probe Instrument Correction Corresponding to (10)
12	t <sub>ic</sub>	•c	Indicated Temperature Corrected for Instrument Error, (10) + (11)
13	T <sub>ic</sub>	•К	Indicated Temperature Corrected for Instru- ment Error (12) + 273.16

14	Plot (13) T <sub>a</sub> is kn to fair in	versus (9) and fair a straight line through the data. (If own it may be plotted as $T_a = T_{ic} \text{ at } V_t^2 = 0$ and be used in the line.) See Figure 6.4.
15	m	Slope of the straight line of (14). For any two points on the line, 1 and 2, the slope is equal to: $m = \frac{(T_{ic})_2 - (T_{ic})_1}{(V_t^2)_2 - (V_t^2)_1}$
16	ĸ	Temperature Probe Recovery Factor, 7592 x (15)

.

## CHART 8.1

## (See paragraph 1.1.3)

GEOPOTENTIAL ALTITUDE,  $H(G/g_{SL})$ , (Thousands of Feet) versus ALTITUDE CORRECTION FACTOR,  $h - H(G/g_{SL})$ , (Feet)

h - H 
$$\left(\frac{G}{g_{SL}}\right)$$
 =  $\frac{H^2 \left(\frac{G}{g_{SL}}\right)^2}{r - H \left(\frac{G}{g_{SL}}\right)}$ 

where:

H = geopotential altitude, geopotential feet

h = tapeline altitude, feet

$$r = 20,930,000 \text{ feet}$$

 $G/g_{SL} = 1 \text{ foot/geopotential foot}$ 

Example:

Given:	H = 76,500 geopotential feet
Required:	Tapeline altitude, h in feet
Solution:	H(G/g <sub>SL</sub> ) = 76,500 feet
	From Chart 8.1, h - $H(G/g_{SL}) = 283$ feet
	$h = \left[H(G/g_{SL})\right] + \left[h - H(G/g_{SL})\right] = 76,783 \text{ feet}$



CHART 8.1

# CHART 8.2

### (See paragraph 2.4)

MACH NUMBER, M vs ATMOSPHERE TEMPERATURE,  $T_a$ (<sup>o</sup>K) or  $t_a$  (<sup>o</sup>C) for INDICATED TEMPERATURE,  $t_{ic}$  (<sup>o</sup>C) = CONSTANT and TEMPERATURE PROBE RECOVERY FACTOR, K = CONSTANT

### A LSO

MACH NUMBER, M vs RATIO OF INDICATED TO ATMOSPHERIC TEMPERATURE,  $T_{ic}/T_{a}$  (°K/°K) for TEMPERATURE PROBE RECOVERY FACTOR, K = CONSTANT

$$\frac{T_{ic}}{T_a} = 1 + K \frac{M^2}{5}$$

Example:

Given: M = 0.785; K = 0.80;  $t_{ic} = 15^{\circ}C$ Required:  $\frac{T_{ic}}{T_{a}}$  and  $t_{a}(^{\circ}C)$ Solution: Use Page 1 of Chart 8.2. For the given conditions,  $\frac{T_{ic}}{T_{a}} = 1.0985$ ;  $t_{a} = -11.0^{\circ}C$ 



121

Σ






154 - -

MACH NUMBER, M



### COMPRESSIBILITY CORRECTION TO CALIBRATED AIRSFEED

 $V_{t}\sqrt{\sigma} = V_{e} = V_{c} + \Delta V_{c}$ 



Calibrated Airspeed, V<sub>c</sub> - Knots



....

TE 2

. 1

∀<sub>c</sub>, Calibrated Airspeed – Knots

# CHART 8.3 (See paragraph 2.5.1)

PRESSURE ALTITUDE,  $H_c$  (Thousands of Feet) versus EQUIVALENT SPEED - MACH NUMBER RATIO,  $V_e/M$  (Knots)

$$\frac{V_e}{M} = a \sqrt{\sigma}$$

where a and  $\sigma$  correspond to  $H_c$ 

Example:

-

Given: 
$$M = 0.39$$
;  $H_c = 18,000$  Feet  
Required:  $V_e$  in Knots  
Solution:  $\frac{V_e}{M} = 466$  Knots  
 $V_e = (\frac{V_e}{M})$   $M = 139.8$  Knots



## CHART 8.4 (See paragraph 2.5.2)

TEST DAY TRUE SPEED,  $V_{tt}$  versus TEST DAY ATMOSPHERIC TEMPERATURE,  $t_{at}$  (°C) for MACH NUMBER, M = CONSTANT

$$V_{tt} = 38.967 M \sqrt{t_{at}} (^{\circ}C) + 273.16$$
, knots

#### ALSO

TRUE SPEED FOR STANDARD DAY,  $V_{ts}$  versus STANDARD DAY ATMOSPHERIC TEMPERATURE,  $t_{as}$  (°C) for MACH NUMBER, M = CONSTANT

$$V_{ts} = 38.967 \text{ M} \sqrt{t_{as}} (^{\circ}\text{C}) + 273.16$$
, knots

where  $t_{ab}$  corresponds to  $H_c$ .

Example:

Given: M = 2.15;  $t_{at} = -60^{\circ}C$ Required:  $V_{tt}$  in knots Solution: Use Page 4 of Chart 8.4. For the given conditions  $V_{tt} = 1223$ , knots











#### CHART 8.5

#### (See paragraph 2.5.2)

MACH NUMBER, M versus CALIBRATED AIRSPEED,  $V_c$  for PRESSURE ALTITUDE,  $H_c$  = CONSTANT

and

MACH NUMBER, M versus CALIBRATED AIRSPEED, V for STANDARD DAY TRUE SPEED, V ts = CONSTANT

#### ALSO

INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  versus INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  = CONSTANT

Given:  $H_c$  and  $V_{ts}$ 1 (a)  $\frac{T_{as}}{T_{aSL}} = 1 - 6.87535 \times 10^{-6} H_c$   $H_c \leq 36,089.24$  feet (b)  $\frac{T_{as}}{T_{aSL}} = 0.751874$   $H_c \geq 36,089.24$  feet  $\frac{2}{T_{aSL}} = \frac{T_{as}}{T_{aSL}} = \frac{1}{2} \frac{1}{2}$ 

$$a_{SL} = (\overline{T_{aSL}})$$

THE REPORT OF A DESCRIPTION OF A DESCRIP

3 M = 
$$\frac{V_{ts}}{a_s} = \frac{V_{ts}}{a_{SL}} / \frac{a_s}{a_{SL}}$$

4 (a) 
$$\frac{q_c}{P_a} = \left[ (1 + 0.2M^2)^{3.5} -1 \right]$$
  $M \le 1.00$ 

(b) 
$$\frac{q_c}{P_a} = \left[\frac{(166.921 M^7)}{(7 M^2 - 1)^{2.5}} - 1\right]$$
  $M \ge 1.00$ 

$$5 \frac{q_c}{P_{aSL}} = \frac{q_c}{P_a} \frac{P_a}{P_{aSL}}$$

6 (a) 
$$\frac{P_a}{P_{aSL}} = (\frac{T_{as}}{T_{aSL}})^{5.2561}$$
  $H_c \leq 36,089.24$  feet

(b) 
$$\frac{P_a}{P_aSL} = 0.223358e$$
 -4.80634 x 10<sup>-5</sup>(H<sub>c</sub> - 36,089.24)  
H<sub>c</sub>  $\ge$  36,089.24 feet

7 (a) 
$$\frac{V_c}{a_{SL}} = 2.23607 \int \left(\frac{q_c}{P_{aSL}} + 1\right)^{-1} \frac{q_c}{P_{aSL}} \le 0.89293$$

(b) 
$$\frac{V_c}{a_{SL}} = 0.881284 \sqrt{\left(\frac{q_c}{P_{aSL}} + 1\right)} \left[ \frac{1 - \frac{1}{7\left(\frac{V_c}{a_{SL}}\right)^2}}{7\left(\frac{V_c}{a_{SL}}\right)^2} \right] \frac{q_c}{P_{aSL}} \approx 0.89293$$

8 
$$V_c = \left(\frac{V_c}{a_{SL}}\right) a_{SL}$$

where a<sub>SL</sub> = 661.48 knots

Example:

-

1. Given: 
$$M = 1.60$$
;  $V_c = 400$  knots  
Required:  $H_c$  and  $V_{ts}$   
Solution: Use Page 21 of Chart 8.5.  $H_c = 52,850$  feet;  $V_{ts} = 917.2$  knots

<b>2</b> .	Given:	$M = 1.20; H_c = 50,000$
	Required:	$V_{c}$ and $V_{ts}$
	Solution:	Use Page 12 of Chart 8.5. V <sub>c</sub> = 308.7 knots; V <sub>ts</sub> = 688.1 knots
3.	Given:	$H_{c} = 35,000 \text{ feet}; V_{c} = 200 \text{ knots}$
	Required:	M and V <sub>ts</sub>
	Solution:	Use Page 5 of Chart 8.5. $M = 0.6023$ ; $V_{ts} = 347.1$ knots
4.	Given:	$M_{ic} = 1.60; V_{ic} = 400 \text{ knots}$
	Required:	H <sub>ic</sub>
	Solution:	Use Page 21 of Chart 8.5. $H_{ic} = 52,850$ feet
5.	Given:	$M_{ic} = 1.20; H_{ic} = 50,000$ feet
	Required:	V <sub>ic</sub>
	Solution:	Use Page 12 of Chart 8.5. $V_{ic} = 308.7$ knots
6.	Given:	$H_{ic} = 35,000 \text{ feet}, V_{ic} = 200 \text{ knots}$
	Required:	M <sub>ic</sub>
	Solution:	Use Page 5 of Chart 8.5. $M_{ic} = 0.6023$

















CHART 8.5


































































•















185

CHART 8.5









je Je
















# 8.6 THE CORRECTION OF ALTIMETER AND/OR AIRSPEED INDICATOR READINGS FOR PRESSURE LAG ERROR

# CHART 8.61

(See paragraph 4.2.1)

LAG CONSTANT RATIO,  ${}^{\lambda}H_{ic}/{}_{SL}$  versus INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Thousands of Feet) for INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) = CONSTANT

$$\frac{{}^{\lambda}H_{ic}}{{}^{\lambda}SL} = \frac{{}^{\mu}H_{ic}}{{}^{\mu}SL} = \frac{P_{aSL}}{P_{s} + q_{cic}}$$

### CHART 8.62

(See paragraph 4.2.1)

LAG CONSTANT TEMPERATURE CORRECTION FACTOR,  $\lambda / {}^{\lambda}H_{ic}$  versus INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Thousands of Feet) for TEST DAY ATMOSPHERIC TEMPERATURE,  $t_{at}$  ( ${}^{\circ}C$ ) = CONSTANT

$$\frac{\lambda}{\lambda_{H_{ic}}} = \frac{T_{at}}{T_{ab}}$$

where T<sub>as</sub> corresponds to H<sub>ic</sub>.

# CHART 8.63 (See paragraph 4.2.2)

AIRSPEED INDICATOR LAG FACTOR,  $F_1$  ( $H_{ic}$ ,  $V_{ic}$ ) versus INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT, and  $dH_{ic}$ /dt in feet per minute.

$$F_{1}(H_{ic}, V_{ic}) = \frac{G\sigma_{s}}{2.84869V_{ic}\left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5}} \quad V_{ic} \le a_{SL}$$

$$\mathbf{F}_{1}(\mathbf{H}_{ic}, \mathbf{V}_{ic}) = G\rho_{g} \frac{\left[7 \left(\frac{\mathbf{V}_{ic}}{\mathbf{a}SL}\right)^{2} - 1\right]^{3.5}}{3738.11\left(\frac{\mathbf{V}_{ic}}{\mathbf{a}SL}\right)^{6}\left[2\left(\frac{\mathbf{V}_{ic}}{\mathbf{a}SL}\right)^{2} - 1\right]} \quad \mathbf{V}_{ic} \ge \mathbf{a}_{SL}$$

where  $\sigma_s$  and  $\rho_s$  correspond to  $H_{ic}$ 

# ALTIMETER

$$\Delta H_{icl} = \frac{\lambda_s}{60} \frac{dH_{ic}}{dt} \qquad 4.24$$

where

$$\lambda_{s} = \lambda_{sSL} \frac{\lambda_{sH_{ic}}}{\lambda_{sSL}} \frac{\lambda_{s}}{\lambda_{sH_{ic}}} 4.21$$

and 
$$dH_{ic}/dt = feet/minute$$
  
 $\lambda_{sSL} = seconds$   
 $\lambda_{s}H_{ic}/\lambda_{sSL}$  from Chart 8.61 for  $H_{ic}$ ,  $V_{ic} = STATIC$ 

$$\lambda_{s}/\lambda_{sH_{ic}}$$
 from Chart 8.62 for  $H_{ic}$ ,  $t_{at}$  (°C)

# AIRSPEED INDICATOR

$$\Delta V_{icl} = \lambda_t \frac{dV_{ic}}{dt} + \frac{(\lambda_s - \lambda_t)}{60} F_1(H_{ic}, V_{ic}) \frac{dH_{ic}}{dt} = 4.43$$

where

$$\lambda_{g} = \lambda_{gSL} \frac{\lambda_{g}H_{ic}}{\lambda_{gSL}} \frac{\lambda_{s}}{\lambda_{g}H_{ic}} 4.21$$

$$\lambda_{t} = \lambda_{tSL} \frac{\lambda_{t}H_{ic}}{\lambda_{tSL}} \frac{\lambda_{t}}{\lambda_{t}H_{ic}} 4.40$$

and 
$$dV_{ic}/dt = knots/second$$
  
 $dH_{ic}/dt = feet/minute$   
 $\lambda_{iSL}$  and  $\lambda_{sSL} = seconds$   
 $\lambda_{sH_{ic}}/\lambda_{sSL}$  from Chart 8.61 for  $H_{ic}$ ,  $V_{ic} = STATIC$   
 $\lambda_{t}H_{ic}/\lambda_{tSL}$  from Chart 8.61 for  $H_{ic}$ ,  $V_{ic}$   
 $\lambda/\lambda H_{ic}$  from Chart 8.62 for  $H_{ic}$ ,  $t_{at}$  (°C)  
 $F_{1}$  ( $H_{ic}$ ,  $V_{ic}$ ) from Chart 8.63 for  $H_{ic}$ ,  $V_{ic}$   
Example:  
Given:  $\lambda_{sSL} = 0.60$ ;  $\lambda_{tSL} = 0.10$   
 $V_{ic} = 800$  knots;  $H_{ic} = 30,000$  feet;  $t_{at} = -30$  °C

$$\frac{dv_{ic}}{dt} = 3 \text{ knots/second;} \quad \frac{dH_{ic}}{dt} = 10,000 \text{ feet/minute}$$

**Required:**  $\Delta H_{icl}$  and  $\Delta V_{icl}$ 

14 A. A. ANTO T. TARA A MELA.

Solution: From Chart 8.61 for  $H_{ic} = 30,000$  feet,  $V_{ic} = STATIC$ 

$$\frac{\lambda_{\rm sH_{ic}}}{\lambda_{\rm sSL}} = 2.80$$

From Chart 8.61 for  $H_{ic} = 30,000$  feet,  $V_{ic} = 800$  knots

$$\frac{\lambda_{tH_{ic}}}{\lambda_{tSL}} = 0.50$$

From Chart 8.62 for  $H_{ic} = 30,000$  feet;  $t_{at} = -30$  °C

$$\frac{\lambda}{\lambda_{H_{ic}}} = 1.063$$

From equation 4.21

$$\lambda_{g} = 0.60 (2.80) (1.063) = 1.786$$

From equation 4.40

$$\lambda_t = 0.10 (0.50) (1.063) = 0.053$$

From equation 4.24 for  $dH_{ic}/dt = 10,000$  feet/minute

$$\Delta H_{icl} = \frac{1.786}{60} (10,000) = 298 \text{ feet}$$

From Chart 8.63 for  $H_{ic} = 30,000$  feet,  $V_{ic} = 800$  knots

$$F_{1}(H_{ic}, V_{ic}) = 0.0030$$

From equation 4.43 for  $dV_{ic}/dt = 3 \text{ knots/second}$ ,  $dH_{ic}/dt = 10,000 \text{ feet/minute}$  $\Delta V_{icl} = 0.053 (3) + \frac{(1.786 - 0.053)}{60} (.0030) 10,000$ 

= 0.159 + 0.866 = 1.025 knots







(See paragraph 5.3.1)

INDICATED PRESSURE ALTITUDE CORRECTED for INSTRUMENT ERROR,  $H_{ic}$  (Thousands of Feet) versus  $\Delta P_p / \Delta H_{pc}$ , ("Hg/Feet) for  $\Delta H_{pc}$  (Feet) = CONSTANT

$$\frac{\Delta P_p}{\Delta H_{pc}} = 0.0010813 \,\sigma_{\tilde{s}} , \, "Hg/feet$$

where  $\sigma_{\overline{s}}$  is measured at  $(H_{ic} + \frac{\Delta H_{pc}}{2})$ 

Example:

Given:  $H_{ic} = 35,000$  feet;  $\Delta H_{pc} = +2000$  feet Required:  $\Delta P_p$  in "Hg

Solution: Use Page 2 of Chart 8.7. For the given conditions,

$$\frac{\Delta P_p}{\Delta H_{pc}} = 0.000322 \text{ "Hg/feet}$$
$$\Delta P_p = \frac{\Delta P_p}{\Delta H_{pc}} \Delta H_{pc} = +0.644 \text{ "Hg}$$







CHART 8.7

Ċ,

## (See paragraph 5.3.1)

INDICATED PRESSURE ALTITUDE CORRECTED for INSTRUMENT ERROR,  $H_{ic}$  (Thousands of Feet) versus  $\Delta P_p / \Delta H_{pc}$ , ("Hg/Feet) for  $\Delta P_p$ , ("Hg) = CONSTANT

$$\frac{\Delta P_p}{\Delta H_{pc}} = 0.0010813 \, \sigma_s \, , \, "Hg/feet$$

where  $\sigma_{c}$  is measured at  $(H_{ic} + \frac{\Delta H_{pc}}{2})$ 

Example:

Given:  $H_{ic} = 52,000 \text{ feet}; \Delta P_p = -0.50 "Hg$ Required:  $\Delta H_{pc}$  in feet

Solution:

Use Page 2 of Chart 8.8. For the given conditions,

$$\frac{\Delta P}{\Delta H_{pc}} = 0.000162$$

$$\Delta H_{pc} = \frac{\Delta P_{p}}{\Delta P_{p}} = 3090 \text{ feet}$$

$$(\frac{\Delta P}{\Delta H_{pc}})$$





# (See paragraph 5.3.2)

INDICATED AIRSPEED CORRECTED for INSTRUMENT ERROR,  $V_{ic}$  (Knots) versus  $\Delta P_p / \Delta V_{pc}$  ("Hg/Knot) for  $\Delta V_{pc}$  (Knots) = CONSTANT

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = \frac{1.4 P_{aSL}}{a_{SL}} \left(\frac{V_{ic}}{a_{SL}}\right) \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5} V_{ic} \leq a_{SL}$$

$$+ \frac{0.7 P_{aSL}}{a_{SL}} \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{1.5} \left[1 + 1.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] \frac{\Delta V_{pc}}{a_{SL}}$$

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = 52.854 \left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}} V_{ic} \geq a_{SL}$$

$$\int V_{ic} \leq a_{SL}$$

52.854 
$$\left(\frac{V_{ic}}{a_{SL}}\right)^{5} = \frac{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{4} - 4.5\left(\frac{V_{ic}}{a_{SL}}\right)^{2} + 3\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{4.5}} = \frac{\Delta V_{pc}}{a_{SL}}$$

where P = 29.92126 "Hg,  $a_{SL} = 661.48$  knots

Example:

Given: 
$$V_{ic} = 300 \text{ knots}; \Delta V_{pc} = -20 \text{ knots}$$
  
Required:  $\Delta P_{p}$  in "Hg

Solution: For the given conditions,

$$\frac{\Delta \mathbf{P}_{\mathbf{p}}}{\Delta \mathbf{V}_{\mathbf{pc}}} = 0.0305 \text{ "Hg/knot}$$
$$\Delta \mathbf{P}_{\mathbf{p}} = \frac{\Delta \mathbf{P}_{\mathbf{p}}}{\Delta \mathbf{V}_{\mathbf{pc}}} \Delta \mathbf{V}_{\mathbf{pc}} = -0.610 \text{ "Hg}$$



# INDICATED AIRSPEED, V<sub>ic</sub> (Knots)

(See paragraph 5.3.2)

INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) versus  $\Delta P_p / \Delta V_{pc}$  ("Hg/Knot) for  $\Delta P_p$  ("Hg) = CONSTANT

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = \frac{1.4 P_{a} S_{L}}{a S_{L}} \left(\frac{V_{ic}}{a S_{L}}\right) \left[1 + 0.2 \left(\frac{V_{ic}}{a S_{L}}\right)^{2}\right]^{2.5} V_{ic} \leq a_{SL}$$

$$+ \frac{0.7 P_{a} S_{L}}{s_{L}} \left[1 + 0.2 \left(\frac{V_{ic}}{a S_{L}}\right)^{2}\right]^{1.5} \left[1 + 1.2 \left(\frac{V_{ic}}{a S_{L}}\right)^{2}\right] \frac{\Delta V_{pc}}{a S_{L}}$$

$$\frac{\Delta P_{p}}{\Delta V_{pc}} = 52.854 \left(\frac{V_{ic}}{a S_{L}}\right)^{6} \frac{\left[2\left(\frac{V_{ic}}{a S_{L}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a S_{L}}\right)^{2} - 1\right]} 3.5 V_{ic} \geq a_{SL}$$

$$+ 52.854 \left(\frac{V_{ic}}{a S_{L}}\right)^{5} \frac{\left[7\left(\frac{V_{ic}}{a S_{L}}\right)^{4} - 4.5\left(\frac{V_{ic}}{a S_{L}}\right)^{2} + 3\right]}{\left[7\left(\frac{V_{ic}}{a S_{L}}\right)^{2} - 1\right]^{4.5}} \frac{\Delta V_{pc}}{a S_{L}}$$

where  $P_{aSL} = 29.92126$  "Hg;  $a_{SL} = 661.48$  knots

Example:

したい おんぞう 小田田 小田田田 ひかんかい かくりかい

Given:  $V_{ic} = 550 \text{ knots}; \Delta P_p = +2.0 "Hg$ 

Required:  $\Delta V_{pc}$  in knots

Solution: For the given conditions,

$$\Delta P_{p} = 0.0763$$

$$\Delta V_{pc} = \left(\frac{\Delta P_{p} / \Delta P}{\Delta V_{pc}}\right) = 26.2 \text{ knots}$$



# CHART 9.11

# (See paragraph 5.3.3)

AIRSPEED POSITION ERROR CORRECTION,  $\Delta V_{pc}$  (Knots) versus INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) for POSITION ERROR PRESSURE COEFFICIENT,  $\Delta P_p / q_{cic}$ 

For 
$$V_{ic} \leq a_{SL}$$
,  

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{\left\{1 \cdot 4\left(\frac{V_{ic}}{a_{SL}}\right)\left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] - \frac{\Delta V_{pc}}{a_{SL}}\right\} + \left\{0 \cdot 7\left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] - \left[1 + 1 \cdot 2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] - \left(\frac{\Delta V_{pc}}{a_{SL}}\right)^{2}\right\}}{\left[1 + 0.2\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] - 1}$$

For 
$$V_{ic} \ge a_{SL}$$
,  

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{7(\frac{V_{ic}}{a_{SL}})}{\frac{P}{q_{cic}}} \frac{\left[\frac{2(\frac{V_{ic}}{a_{SL}})^{2} - 1}{(\frac{V_{ic}}{a_{SL}})^{2} - 1}\right]}{(\frac{V_{ic}}{a_{SL}})^{2} - 1} \frac{\Delta V_{pc}}{a_{SL}} + 7 \frac{\left[\frac{7(\frac{V_{ic}}{a_{SL}})^{4} - 4 \cdot 5(\frac{V_{ic}}{a_{SL}})^{2} + 3}{(\frac{V_{ic}}{a_{SL}})^{2} + 3}\right]}{\left[\frac{V_{ic}}{a_{SL}}\right]^{2}} - \frac{(\frac{\Delta V_{pc}}{a_{SL}})^{2}}{(\frac{V_{ic}}{a_{SL}})^{2} - 1} - \frac{(\frac{V_{ic}}{a_{SL}})^{2}}{(\frac{V_{ic}}{a_{SL}})^{2}} - 1}$$

where  $a_{SL} = 661.48$  knots

Example:

Given:  $V_{ic} = 700 \text{ knots}; \Delta V_{pc} = -20 \text{ knots}$ Required:  $\Delta P_p / q_{cic}$ Solution: Use Page 2 of Chart 8.11. For the given conditions,  $\Delta P_p / q_{cic} = -0.070$ 







# CHART 8.12 (See paragraph 5.3.4)

RATIO OF ALTIMETER TO AIRSPEED INDICATOR POSITION ERROR CORRECTIONS,  $\Delta H_{pc} / \Delta V_{pc}$  (Feet/Knots) versus INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$ (Feet) = CONSTANT

$$\frac{\Delta H_{pc}}{\Delta V_{pc}} = \frac{58.566}{\sigma_{5}} \left(\frac{V_{ic}}{a_{SL}}\right) \left[1 + 0.L\left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5} \quad V_{ic} \leq a_{SL}$$

$$\frac{\Delta H_{pc}}{\Delta V_{pc}} = \frac{48,880}{\sigma_{5}} \left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2\left(V_{ic}/a_{SL}\right)^{2} - 1\right]}{\left[7\left(V_{ic}/a_{SL}\right)^{2} - 1\right]}^{3.5} \quad V_{ic} \geq a_{SL}$$

where  $\sigma_{5}$  is measured at  $H_{ic}$  and  $a_{SL} = 661.48$  knots

Note: This curve is valid for small errors only, (say  $\Delta H_{pc} \leq 1000$  feet or  $\Delta V_{pc} \leq 10$  knots). Chart 8.13 should be used for larger errors.

Example:

Given:  $H_{ic} = 20,000 \text{ feet}; V_{ic} = 600 \text{ knots}; \Delta H_{pc} = 2000 \text{ feet}$ Required:  $\Delta V_{pc}$  in knots Solution: Use Page 1 of Chart 8.12. For the given conditions,  $\Delta H_{pc} / \Delta V_{pc} = 147 \text{ feet} / \text{ knots}$  $\Delta V_{pc} = \frac{\Delta H_{pc}}{\Delta H_{pc} / \Delta V_{pc}} = 13.6 \text{ knots}$ 

Note: The exact solution is found from Chart 8.13 to be 13.0 knots.





INDICATED AIRSPEED, V<sub>ic</sub> (Knots)



(See paragraph 5, 3, 4)

AIRSPEED POSITION ERROR CORRECTION,  $\Delta V_{pc}$  (Knots) versus STATIC PRESSURE ERROR,  $\Delta P_p$  ("Hg) for INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) = CONSTANT

### and

ALTIMETER POSITION ERROR CORRECTION,  $\Delta H_{pc}$  (Feet) versus STATIC PRESSURE ERROR,  $\Delta P_{p}$  ("Hg) for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT

### ALSO

AIRSPEED POSITION ERROR CORRECTION,  $\Delta V_{pc}$  (Knots) versus ALTIMETER POSITION ERROR CORRECTION,  $\Delta H_{pc}$  (Feet) for INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots) = CONSTANT and INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT

$$\Delta P_{p} = 0.0010813 \sigma_{s} \Delta H_{pc}$$
  
for  $V_{ic} \leq a_{SL}$   
=  $1.4P_{a_{SL}}(\frac{V_{ic}}{a_{SL}}) \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{2.5} \frac{\Delta V_{pc}}{a_{SL}}$   
+  $0.7 P_{a_{SL}} \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{1.5} \left[1 + 1.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right] \left(\frac{\Delta V_{pc}}{a_{SL}}\right)^{2}$ 

for 
$$V_{ic} \ge a_{SL}$$
  
=  $7KP_{a_{SL}} \left(\frac{V_{ic}}{a_{SL}}\right)^{6} \frac{\left[2\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}} \frac{\Delta V_{pc}}{a_{SL}}$   
+  $7KP_{a_{SL}} \left(\frac{V_{ic}}{a_{SL}}\right)^{5} \frac{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{3.5}}{\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} + 3\right]\left(\frac{\Delta V_{pc}}{a_{SL}}\right)^{2}}$   
 $\left[7\left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{4.5}$ 

where K = 166.921;  $P_{aSL} = 29.92126$  "Hg;  $a_{SL} = 661.48$  knots and  $\sigma_{\rm s}$  is measured at  $H_{\rm ic} + \frac{\Delta H_{\rm pc}}{2}$ 

### Example:

 $H_{ic} = 35,000 \text{ feet ; } \Delta H_{pc} = +2000 \text{ feet}$ Given: Required:  $\Delta P_p$  in "Hg

Solution: Use Page 3 of Chart 8.13 for positive errors. For the given conditions,

$$\Delta \mathbf{P}_{\mathbf{p}} = 0.645 \, ^{\prime\prime} \mathrm{Hg}$$

D

Example 2:

Given: 
$$V_{ic} = 300 \text{ knots}; \Delta V_{pc} = -20 \text{ knots}$$
  
Required:  $\Delta P_p$  in "Hg  
Solution: Use Page 4 of Chart 8, 13 for negative errors  
For the given conditions  
 $\Delta P_n = + 0.610$  "Hg

Example 3:

Given:	$V_{ic} = 400 \text{ knots}; H_{ic} = 30,000 \text{ feet}; \Delta V_{pc} = +20 \text{ knots}$
Required:	ΔH in feet
Solution:	Use Page 3 of Chart 8.13 for positive errors.
	For the given conditions,
	$\Delta H_{pc} = + 2440 \text{ feet}$
Note: The approximate solution is found from Chart 8.12 to be	

Note: The approximate solution is found from Chart 8.12 to be  $\Delta H_{pc} = 2260$  feet.



CHART 8.13






.





233

CHART 8.13

## CHART 8.14 (See paragraph 5.3.5)

INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  versus RATIO OF MACH METER TO ALTIMETER POSITION ERROR CORRECTIONS,  $\Delta M_{pc} / \Delta H_{pc}$  (<sup>1</sup>/Feet) for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT

 $\frac{\Delta M_{pc}}{\Delta H_{pc}} = 0.007438 \frac{(1 + 0.2 M_{ic}^2)}{T_{as} M_{ic}} \qquad M_{ic} \leq 1.00$  $\frac{\Delta M_{pc}}{\Delta H_{pc}} = 0.001488 \frac{M_{ic}}{T_{as}} \frac{(7 M_{ic}^2 - 1)}{(2 M_{ic}^2 - 1)} \qquad M_{ic} \geq 1.00$ 

where T<sub>as</sub> is measured at H<sub>ic</sub>.

Note: This curve is valid for small errors only, (say  $\Delta H_{pc} \leq 1000$  feet or  $\Delta M_{nc} \leq 0.04$ ). Chart 8.15 should be used for larger errors.

Example:

ple: Given:  $M_{ic} = 2.30$ ;  $H_{ic} = 46,000$  feet;  $\Delta H_{pc} = -800$  feet Required:  $\Delta M_{pc}$ Solution: Use Page 3 of Chart 8.14. For the given conditions,  $\frac{\Delta M_{pc}}{\Delta H_{pc}} = 5.94 \times 10^{-5} \frac{1}{\text{Feet}}$  $\Delta M_{pc} = \frac{\Delta M_{pc}}{\Delta H_{pc}} \Delta H_{pc} = 0.0475$ 

Note: The exact solution is found from Chart 8.15 to be

$$\Delta M_{\rm pc} = -0.0470$$





INDICATED MACH NUMBER, Mic



CHAR**T** 8,14

(See paragraph 5, 3, 5)

MACH METER POSITION ERROR CORRECTION,  $\Delta M_{pc}$  versus RATIO OF STATIC PRESSURE ERROR TO INDICATED STATIC PRESSURE,  $\Delta P_p / P_s$ for INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  = CONSTANT

## and

ALTIMETER POSITION ERROR CORRECTION,  $\Delta H_{pc}$  (Feet) versus RATIO OF STATIC PRESSURE ERROR TO INDICATED STATIC PRESSURE,  $\Delta P_p / P_g$  for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  = CONSTANT

## ALSO

MACH METER POSITION ERROR CORRECTION,  $\Delta M_{pc}$  versus ALTIMETER POSITION ERROR CORRECTION,  $\Delta H_{pc}$  (Feet) for INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  = CONSTANT and INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT

$$\frac{\Delta P_{p}}{P_{s}} = 0.0010813 \quad \sigma_{s} / P_{s} \quad \Delta H_{pc}$$

$$= \frac{1.4 M_{ic} \Delta M_{pc}}{(1+0.2 M_{ic}^{2})} + \frac{0.7(1-1.6 M_{ic}^{2}) \Delta M_{pc}}{(1+0.2 M_{ic}^{2})^{2}} M_{ic} \leq 1.00$$

$$= \frac{7(2 M_{ic}^{2}-1) \Delta M_{pc}}{M_{ic} (7 M_{ic}^{2}-1)} - \frac{7(21 M_{ic}^{4}-23.5 M_{ic}^{2}+4) \Delta M_{pc}}{M_{ic}^{2} (7 M_{ic}^{2}-1)^{2}}$$

$$M_{ic} \geq 1.00$$

where  $P_{g}$  is measured at  $H_{ic}$  and  $\sigma_{g}$  is measured at  $H_{ic} + \frac{\Delta H_{pc}}{2}$ 

Example:

Given:  $M_{ic} = 1.00$ ,  $H_{ic} = 72,000$  feet;  $\Delta H_{pc} = 2400$  feet Required:  $\Delta M_{pc}$ Solution: Use Page 5 of Chart 8.15 for positive errors,  $\Delta M_{pc} = 0.0968$ 

Note: The approximate solution is found from Chart 8.14 to be  $\Delta M_{pc} = 0.0986$ 



CHART 8.15















## (See paragraph 5.3.6)

RATIO OF MACH METER TO AIRSPEED INDICATOR POSITION ERROR CORRECTIONS,  $\Delta M_{pc} / \Delta V_{pc}$  ( $\frac{1}{Knots}$ ) versus INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  for INDICATED PRESSURE ALTITUDE CORRECTED FOR INSTRUMENT ERROR,  $H_{ic}$  (Feet) = CONSTANT

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{P_{a}SL}{sL} \frac{1}{P_{a}} \frac{V_{ic}}{sL} \left[ 1 + 0.2 \left( \frac{V_{ic}}{a} \right)^{2} \right]^{2.5} \frac{(1 + 0.2 M_{ic}^{2})}{M_{ic}} V_{ic} \leq a_{SL} M_{ic} \leq 1.00$$

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{P_{a}SL}{5 a_{SL}} \frac{1}{P_{g}} \frac{V_{ic}}{a_{SL}} \left[ 1 + 0.2 \left( \frac{V_{ic}}{a} \right)^{2} \right]^{2.5} \frac{M_{ic} (7 M_{ic}^{2} - 1)}{(2 M_{ic}^{2} - 1)} V_{ic} \leq a_{SL} M_{ic} \leq 1.00$$

$$\frac{\Delta M_{pc}}{\Delta V_{pc}} = \frac{166.921 F_{aSL}}{aSL} \frac{1}{P_{g}} \frac{V_{ic}}{SL} \left[ 2 \left( \frac{V_{ic}}{aSL} \right)^{2} - 1 \right]}{\left[ 7 \left( \frac{V_{ic}}{aSL} \right)^{2} - 1 \right]} \frac{M_{ic} \left( 7 M_{ic}^{2} - 1 \right)}{\left( 2 M_{ic}^{2} - 1 \right)} \frac{V_{ic} \ge a_{SL}}{M_{ic} \ge a_{SL}}$$

where  $P_{aSL} = 29.92126$  "Hg;  $a_{SL} = 661.48$  knots and  $P_{s}$  is measured at  $H_{ic}$ .

Note: This curve is valid for small errors only, (say  $\Delta V_{pc} < 10$  knots or  $\Delta M_{pc} < 0.04$ ) and should not be used when the position error is larger.

Example:

Given: 
$$M_{ic} = 2.40$$
;  $H_{ic} = 60,000$  feet;  $\Delta V_{pc} = 2.0$  knots  
Required:  $\Delta M_{pc}$   
Solution: Use Page 3 of Chart 8.16. For the given conditions,  
 $\frac{\Delta M_{pc}}{\Delta V_{pc}} = 0.03945 \frac{l}{Knots}$   
 $\Delta M_{pc} = (\frac{\Delta M_{pc}}{\Delta V_{pc}}) \Delta V_{pc} = 0.0789$ 

Preceding page blank .

The method to be used in case of larger errors is illustrated by the following example.

Given:	$M_{ic} = 1.00; H_{ic} = 35,000 \text{ feet }; \Delta V_{pc} = +20 \text{ knots}$
Required:	ΔM <sub>pc</sub>
Solution:	1. $V_{ic} = 350 \text{ knots for } M_{ic}$ , $H_{ic}$ and Chart 8.5 $\Delta P_p/q_{cic} = +0.127 \text{ for } V_{ic}$ , $\Delta V_{pc}$ and Chart 8.11 $\Delta M_{pc} = +0.1000 \text{ for } M_{ic}$ , $\Delta P_p/q_{cic}$ and Chart 8.18
	2. $V_{ic} = 350$ knots for $M_{ic}$ , $H_{ic}$ and Chart 8.5 $\Delta H_{pc} = +2510$ feet for $\Delta V_{pc}$ , $V_{ic}$ , $H_{ic}$ and Chart 8.13 $\Delta M_{pc} = +0.1000$ for $\Delta H_{pc}$ , $H_{ic}$ , $M_{ic}$ and Chart 8.15

Note: Use of the approximate Chart 8.16 gives  $\Delta M_{pc} = +0.094$ 

. ....







253

CHART 8.16



## **CHART 8.17** (See paragraph 5.3.7)

INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR; M<sub>ic</sub> versus RATIO OF MACH METER POSITION ERROR CORRECTION TO POSITION ERROR PRESSURE COEFFICIENT,  $\Delta M_{pc} / (\Delta P_p / q_{cic})$ 

$$\frac{\Delta M_{pc}}{(\Delta P_{p}/q_{cic})} = \frac{(1+0.2 M_{ic}^{2})}{1.4 M_{ic}} \left[ (1+0.2 M_{ic}^{2})^{3.5} - 1 \right] M_{ic} \leq 1.00$$

$$\frac{\Delta M_{pc}}{(\Delta P_{p}/q_{cic})} = \frac{M_{ic} \left[ 166.921 M_{ic}^{7} - (7 M_{ic}^{2} - 1)^{2.5} \right]}{7(7 M_{ic}^{2} - 1)} M_{ic} \geq 1.00$$

Note: This curve is valid for small errors only, (say  $\Delta M_{pc} < 0.04$  or  $\Delta P_p/q_{cic} < 0.04$ ). Chart 8.18 should be used for larger errors.

Example:

$$M_{ic} = 0.85; \Delta P_{p}/q_{cic} = +0.10$$

Required:

Solution:

Given:

ΔM<sub>DC</sub> Use Page 1 of Chart 8.17. For the given conditions,  $\Delta M_{pc} / (\Delta P_p / q_{cic}) = 0.58$  $\Delta M_{pc} = \frac{\Delta M_{pc}}{(\Delta P_p/q_{cic})} (\Delta P_p/q_{cic}) = +0.058$ 

Note: The exact solution is found from Chart 8.18 to be  $\Delta M_{pc} = +0.0588$ 







INDICATED MACH NUMBER, Mic



## CHART 8.18 (See paragraph 5.3.7)

POSITION ERROR PRESSURE COEFFICIENT,  $\Delta P_p/q_{cic}$  versus MACH METER POSITION ERROR CORRECTION,  $\Delta M_{pc}$  for INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  = CONSTANT

$$\frac{\Delta P}{q_{cic}} = \frac{\frac{1.4 M_{ic} \Delta M_{pc}}{(1+0.2 M_{ic}^2)} + \frac{0.7 (1-1.6 M_{ic}^2) \Delta M_{pc}^2}{(1+0.2 M_{ic}^2)^2}}{(1+0.2 M_{ic}^2)^{-1}} \qquad M_{ic} \le 1.00$$

$$\frac{\Delta P_{p}}{q_{cic}} = \frac{\frac{7 (2 M_{ic}^{2} - 1)}{M_{ic} (7M_{ic}^{2} - 1)} \Delta M_{pc} - \frac{7 (21 M_{ic}^{4} - 23.5 M_{ic}^{2} + 4) \Delta M_{pc}^{2}}{M_{ic}^{2} (7 M_{ic}^{2} - 1)^{2}} M_{ic} \ge 1.00}{\left[\frac{166.921 M_{ic}^{7}}{(7 M_{ic}^{2} - 1)^{2.5} - 1}\right]}$$

Example:

Given:  $M_{ic} = 1.00; \Delta P_p / q_{cic} = +0.14$ 

Required:  $\Delta M_{pc}$ 

Solution: Use Page 1 of Chart 8.18 for positive errors. For the given conditions,

$$\Delta M_{\rm pc} = +0.111$$

Note: The approximate solution is found from Chart 8.17 to be  $\Delta M_{pc} = +0.107$ 



ΔM<sub>pc</sub>

CHART 8.18



TABLE 9.1

# UNITED STATES STANDARD ATMOSPHERE - SUMMARY OF BASIC DATA

## From NACA Report 1235

		Ň	Values	
Element	Symbol	In c.g.s. System	In m.k.s. System	In English System
Standard Sea Level Temperature	tSL	15•C	15°C	59 <b>•F</b>
Standard Sea Level Temperature	TSL	288. 16•K	298. 16•K	518.688•R
Standard Sea Level Pressure	PSL	1.013250 × 10 <sup>6</sup> dynes cm <sup>-2</sup> 76cmHg(std)	-2 1013, 250mb 760mm Hg (std) 10332, 27 kg m <sup>-2</sup>	1013.250mb 29.92126 in.Hg(std) 2116.216 ib.6.2
Standard Sea Level Density	PSL	$1.2250 \times 10^{-3} \text{ gm cm}^{-3}$	0.12492 kg sec <sup>2</sup> m <sup>-4</sup> 1.2250 kg (mass)m <sup>-3</sup>	0.076475 1h (mass) ft -3
Geopotential of Tropopause	*H	$1.0787315 \times 10^{9} \text{ cm}^{2} \text{ sec}^{-2}$	11,000 m'	36, 089. 24 ft'
Standard Temperature at the Tropopause	+ +	-56. 50°C	-56.5 <b>0°</b> C	-69.7*F
Standard Temperature at the Tropopause	T*	216.66•K	216.66•K	3 <b>89.968°</b> R
Pressure at the Tropopause	* ቢ	2.2632 x 10 <sup>5</sup> dynes cm <sup>-2</sup> 16.975 cm Hg(std)	226.32 mb 169.75 mm Hg (std) 2307 8 kg m -2	226.32 mb 6.683 in.Hg (std) 472 68 ib.6-2
Density at Geopotential H*	*d	3.6392 x 10 <sup>-4</sup> gm cm <sup>-3</sup>	0. 037109 kg sec <sup>2</sup> m <sup>-4</sup> 0. 36392 kg(mass)m <sup>-3</sup>	0.00070612 lb sec <sup>2</sup> ft <sup>-4</sup> 0.022719 lb(mass)ft -3
Temperature Lapse Rate in Troposphere		6.628155 x 10 <sup>-8</sup> C cm <sup>-2</sup> sc <sup>2</sup> 0.0065 C m' <sup>-1</sup>	0.0065°C m' <sup>-1</sup>	0. 00356616•F ft'-1
Standard Sea Level Speed of Sound	τs	34030 cm sec <sup>-1</sup>	340.30 m sec <sup>-1</sup>	1]16.45 sec <sup>-1</sup> ft 661.48 knote
Standard Sea Level Gravitational Acceleration	gSL	980. 665 cm sec <sup>-2</sup>	9.80665 m sec <sup>-2</sup>	32.17405 ft sec <sup>-2</sup>

			0.0764751b ft <sup>- J</sup> 2.46051b(mass)ff <sup>2</sup> sec <sup>2</sup>	32,17405ft <sup>c</sup> seč <sup>-</sup> ft' <sup>-1</sup>	$\frac{17.165 \times 10^{2} \text{ft}^{2} \text{sec}^{-2} \text{eR}^{-1}}{2 \text{sec}^{-2} \text{eR}^{-1}}$		
Values	In m. k. s. System	1.8286×10-5kg sec m <sup>-2</sup> 1.7932×10-5kgmass)m <sup>-1</sup> sec <sup>-1</sup>	1.2250 kg m <sup>-3</sup> 12.013kgmass)m <sup>-2</sup> sec <sup>-2</sup>	9.80665m <sup>6</sup> sec <sup>-6</sup> m' <sup>-1</sup>	$28704 \times 10^{2} \text{m}^{2} \text{sec}^{-2} \text{K}^{-1}$		
	Symbol In c.g.s. System	1.7932x10 <sup>-4</sup> gm an <sup>-1</sup> sec <sup>-1</sup>	1.20131gm cm <sup>-2</sup> sec <sup>-2</sup>	1	28704x10 <sup>6</sup> cm <sup>2</sup> se <sup>2</sup> <sup>-</sup> K <sup>-1</sup>	28.966gm mol <sup>-1</sup>	
	Symbol	<b>≁</b> SL	BL8SL	U	R	X	
	Flement	Coefficient of Viscosity of Air ASL at Temperature TSL	Standard Sea Level Specific Weight	Dimensional constart, the arnount of which determines the magnitude of the unit of H in terms of fundamental	Gas Constant for Dry Air	Apparent Molecular Weight of Dry Air	

Note: Prime quantities represent geopotential units

Example: 1m' = 1 geopotential meter

Conversion Factors

1 foot = 0.3048 meter
1 pound = 0.4535923 kilogram
1 nautical mile = 6076.1033 feet
•K = 273.16 + •C

## TABLES 9.2 AND 9.3

## THE UNITED STATES STANDARD ATMOSPHERE

For pressure altitude,  $H_c < 36,089.24$  feet:

$$P_{a} = P_{aSL} (1 - 6.87535 \times 10^{-6} H_{c})^{5.2561}$$

$$T_{a} = T_{aSL} (1 - 6.87535 \times 10^{-6} H_{c})$$

$$\rho = \rho_{SL} (1 - 6.87535 \times 10^{-6} H_{c})^{4.2561}$$

$$a = a_{SL} (\frac{T_{a}}{T_{aSL}})^{0.5}$$

For pressure altitude,  $H_c > 36,089.24$  feet:

$$P_{a} = 6.6832 e^{-4.80634 \times 10^{-5} (H_{c} - 36, 089.24)}$$
$$T_{a} = 216.66 \ ^{\kappa}K$$
$$\rho = 0.00070612 e^{-4.80634 \times 10^{-5} (H_{c} - 36, 089.24)}$$

$$a = 573.58$$
 knots

Preceding page blank

## TABLE 9.2

## THE UNITED STATES STANDARD ATMOSPHERE

Нc	Pa	8	1/5	Ta	$\int T_a$	θ	νθ
(Feet)	(''Hg)	Pa/PaSL		(• <b>ĸ</b> )	V 4	$T_a/T_{aSL}$	¥ ·
-1000	31.018	1.0366	. 9645	290.15	17.034	1.0069	1.0034
-900	30.907	1.0329	. 9680	289.95	17.028	1.0062	1,0031
-800	30.796	1.0292	. 9715	289.74	17.022	1.0055	1.0027
- 700	30.686	1.0255	.9750	289.54	17.016	1.0048	1.0024
_ 600	30.575	1.0218	. 9785	289.34	17.010	1.0041	1.0021
- 500	30.465	1.0182	.9821	289.14	17.004	1.0034	1.0017
-400	30.356	1.0145	. 9856	288.97	16.999	1.0028	1.0014
-300	30.247	1.0108	. 9892	288.77	16.993	1.0021	1.0010
-200 -100	30.138 30.029	1.0072	.9928 .9963	288.56 288.36	16.987	1.0014 1.0007	1.0007
-100					16.981		1.0003
0	29.921	1.0000	1,0000	288.16	16.975	1.0000	1.0000
100	29.813	. 9963	1.0036	287.96	16.969	. 9993	. 9997
200	29.705	. 9927	1.0072	287.76	16.963	. 9986	.9993
300	29.598	. 9892	1.0109	287.55	16.957	. 9979	.9990
400	29.491	. 9856	1.0145	287.35	16.951	. 9972	.9986
500 600	29.384 29.278	. 9820	1.0182	287.18	16.946	. 9966	.9983
700	29.278	.9785 .9749	1.0219 1.0255	286.98 286,78	16.940 16.934	.9959 .9952	.9979 .9976
800	29.066	.9714	1.0294	286.58	16.929	. 9945	.9972
900	28.960	.9679	1.0331	286.37	16.923	. 9938	.9969
1000	28.855	.9643	1.0369	286.17	16.917	. 9931	.9966
<u>    1100                              </u>	<u>28.750</u> 28.646	. 9608	1.0407	<u>285.97</u> 285.77	16,911	. 9924	.9962
1300	28.542	.9573 .9539	1.0445 1.0483	285.60	16.905 16.900	.9917 .9911	.9959 .9955
1400	28.438	. 9504	1.0521	285.39	16.894	. 9904	.9955
1500	28.334	.9469	1.0559	285.19	16.888	. 9897	.9948
1600	28.231	. 9435	1.0598	284.99	16.882	. 9890	. 9945
1700	28.128	.9400	1.0637	284.79	16.876	. 9883	, 9941
1800	28.025	. 9366	1.0676	284.59	16.870	. 9876	. 9938
1900	27.923	.9332	1.0715	284.39	16.864	.9869	.9934
2000	27.821	. 9298	1.0754	284.18	16.858	. 9862	.9931
2100	27.719	. 9264	1.0794	284.01	16.853	. 9856	. 9928
2200	27.617	. 9230	1.0834	283.81	16.847	. 9849	. 9924
2300	27.516	.9196	1.0873	283.61	16.841	.9842	.9921
2400	27.415	. 9162	1.0913	283.41	16.835	. 9835	. 9917
2500	27.315	.9129	1.0954	283.20	16.829	. 9828	. 9914
2600	27.214	. 9095	1.0994	283.00	16.823	. 9821	.9910
2700	27.114	. 9062	1.1034	282.80	16.817		.9907
2800	27.015 26.915	.9028	1.1075	282.60	16.811	. 9807	. 9903
2900	40.913	. 8995	1,1116	282.43	16.806	.9801	.9900

TABLE 9.2

## TABLE 9.3

## 1/5/0 50/8 1/50 H<sub>c</sub> σ σ а (Feet) p/pSL (Knots) -10001.0296 1.0147 .9855 .9679 663.73 .9613 .9870 9652 - 900 1,0266 1.0132 .9711 663.52 .9884 -800 1.0236 1.0117 ,9742 .9689 663.27 -700 1.0103 1.0206 .9898 .9774 .9727 663.07 - 600 .9913 1.0176 1.0088 .9807 9767 662.97 1.0073 ~ 500 1.0147 .9928 . 9338 ,9805 662.60 - 400 1.0117 1.0059 .9941 .9870 .9843 662.41 - 300 1.0088 1.0044 9956 .9902 9882 662.14 . 9922 -200 1.0058 1.0029 . 9971 .9935 661.94 -100 1.0029 1.0015 .9985 .9967 . 9961 661.68 1.0000 0 1.0000 1.0000 1.0000 1.0000 661.48 .9970 1.0040 100 .9985 1.0015 1.0033 661.28 . 9941 .9971 200 1.0029 1.0066 1.0080 661.02 .9912 300 . 9956 1.0044 1.0099 1.0121 660.82 . 9883 400 9942 1.0058 1.0132 1.0161 660.55 500 .9354 . 9927 1.0074 1.0165 1.0200 660.36 600 . 9825 .9912 1.0039 1.0198 1.024] 660.09 .9796 9898 700 1,0103 1.0232 1.0282 659.89 .9768 . 9883 1.0115 800 1.0265 1.0323 659.63 900 .9739 .9869 1.0138 1.0364 1.0300 659.43 1000 .9710 .9854 1.0148 1.0334 1.0406 659.23 1100 .9682 9840 1.0163 1.0368 1.0448 658.97 1200 .9653 .9825 1.0178 1.0402 1.0490 658.77 .9625 .9811 1300 1.)193 1.0436 1.0530 658.50 9796 1400 9596 1.0208 1.0573 658.31 i.0471 1500 .9568 .9782 1.0223 1.0505 1.0615 658.04 1600 .9540 .9767 1.0239 1.0540 1.0658 657.84 1700 .9511 . 9753 1.0253 1.0575 1.0700 657.58 1800 . 9739 .9483 1.0268 1.0510 657.38 1.0743 1900 .9455 .9724 1.0284 1.0645 1.0787 657.12 ,9710 2000 .9427 1.0299 1.0831 656.92 1.0681 2100 <u>, 9399</u> <u>9695</u> <u>1,0315</u> <u>1.0717</u> 1.0874 656,72 2200 .9371 . 9681 1.0330 1.0752 1.0918 656.45 2300 .9344 , **96**36 1.0344 1.0788 1.0962 656,26 2400 . 9316 9652 1,0361 1.0823 1.1006 655.99 2500 .9288 .9638 1.0376 655.79 1.0860 1.1050 2600 .9261 ; 9623 1.0392 1.0895 1.1094 655,53 2700 <u>. 9233</u> . 9<u>609</u> 1.0407 655.33 1.0932 1.1140 .9595 2800 . 9205 1.0422 1.0968 1.1104 655.07 2900 .9178 .9581 1.0437 1.1005 1.1230 654.87

## THE UNITED STATES STANDARD ATMOSPHERE

TABLE 9.3
Hc	$\mathbf{P}_{\mathbf{a}}$	8	1/(	Ta		θ	J <del></del> θ
(Feet)	(''Hg)	$P_a/P_{aSL}$	1/2	(°K)	√Ta	$T_a/T_{aSL}$	10
					·		
3000	26,816	. 8962	1.1157	282.22	16.800	.9794	.9896
3100	26,717	.8929	1.1198	282.02	16.794	.9787	<u>. 9893</u>
3200	26.619	.8896	1.1240	281,82	16.788	. 9780	.9889
3300	26.521	.8863	1.1282	281.62	16.782	.9773	.9886
<u>3400</u> 3500	<u>26.423</u> 26.325	.8798	<u>1.1323</u> 1.1365	281.42	16.775 16.769	.9766	.9882
3600	26.228	.8798 .8765	1.1305	281.22		.9759	.9879
3700	26.131	.8733	1.1407 1.1450	281.01	16.763 15.758	.9752	.9875
3800	25.034	.8701	1.1492	280.64	$\frac{13,756}{16,752}$	<u>.9745</u> .9739	<u>.9872</u> .9869
3900	25,938	.8668	1.1535	280.44	16,746	.9732	.9865
4000	25.841	. 8636	1.1578	280.24	16.740	.9725	,9862
4100	25.746	. 8604	1.1321	280.03	16.734	. 9718	. 9858
4200	25.650	.8572	1.1655	279.83	16.728	.9711	.9855
4300	25,555	.8540	1.1708	279.53	16.722	.9704	.9851
4400	25.460	.8509	1.1752	279.43	16.716	. 9697	.9848
4500	25.365	.8477	1.1795	279.26	16.711	. 9691	.9844
4600	25.270	.8445	1.1840	279.05	16.705	.9684	.9841
4700	<u>25.17</u> ć		1.1884	278.85	16.699	.9677	.9837
4800	25.082	.8382	1.1929	278.55	16.693	.9670	.9834
4900	24.989	. 8351	1.1973	278.45	16.687	. 9663	.9830
5000	24.895	.8320	1.2018	278,25	16.681	.9656	.9827
	24.802	. 8289	1.2063	278,05	16.675	.9649	. 9823
5200	24.710	. 8258	1.2108	277,84	16.669	.9642	. 9820
5300	24.617	.8227	1.2154	277.67	16.663	.9636	.9816
5400	24.525	. 8196	1.2200	277.47	16.657	.9629	.9813
5500	24.433	. 8166	1.2245	277.27	16.651	.9622	. 9809
5600	24.342	. 8135	1.2292	277.07	16.645	.9615	. 9805
5700	24.250	. 8104	1.2338	275.86	16.639	. 9608	. 9802
5800	24.159	.8074	1.2384	275.65	16.633	. 9601	.9799
5900	24.068	.8044	1.2431	276.46	16.627	.9594	.9795
6000	23.978	. 8013	1.2478	276.26	16.621	.9587	.9792
6100	23.888	. 7983	1.2525	276.09	16.616	. 9581	. 9788
6200	23.798	. 7953	1,2572	275.88	16.610	.9574	. 9785
6300	23.708	. 7923	1.2620	275.68	16.604	. 9567	.9781
6400	23.618	. 7893	1.2668	275.48	16.598	.9560	. 9778
6500	23.529	. 7863	1.2716	275.28	16.592	, 9553	.9774
6600	23.440	. 7834	1.2764	275.08	16.585	,9546	.9770
6700	23.352	. 7804	1.2812	274.88	16.579	.9539	.9767
6800	23.264	.7775	1.2861	274.57	13.573	. 9532	.9763
6900	23.175	.7745	1.2910	274.50	16.568	.9526	.9760

:

TABLE 9.2

H <sub>c</sub> (Feet)	σ p/psl	<u> σ</u>	1/5σ	1018	1/8/0	a (Knots)
3000	.9151	.9566	1.0454	1.1042	1.1275	654.59
3100	. 9123	. 9552	1.0469	1,1079	1.1321	654.39
3200	. 9096	. 9538	1.0484	1.1116	1.1367	654.13
<b>3300</b>	. 9069	. 9523	1.0501	1.1153	1.1412	653.93
	. 9042	. 9509	<u>1.0516</u>	1,1190	1.1458	653.67
3500	. 9015	. 9495	1.0532	1.1228	1.1506	653.47
3600	. 8988	. 9481	1,0547	1.1265	1.1552	653.20
	<u>, 8961</u>	. 9466	1.0564	1.1304	1,1599	653.01
3800	. 8934	.9452	1.0580	1.1342	1.1647	652.81
3900	.8907	.9438	1.0595	1.1380	1.1694	652.54
4000	.8880	.9424	1.0611	1.1419	1.1743	652.34
<u>4100</u>	. 8854	. 9410	1,0627	1,1457	1.1789	652,08
4200	.8827	.9395	1.0644	1.1496	1.1838	651.88
4300	.8801	. 9381	1.0660	1.1534	1.1886	651.62
4400	. 8774	. 9367	1.0676	1.1574	1.1936	651,42
4500	.8748	.9353	1.0692	1,1612	1.1983	651.15
4600	. 8721	. 9339	1.0708	1.1652	1,2033	650.96
4700	, 8695	<u>,9325</u>	1.0724	1.1691	1,2082	<u>     650 . 69  </u>
4800	, 8669	.9311	1.0740	1.1731	1,2132	650.49
4900	.8643	.9297	1.0756	1.1770	1.2180	650,23
5000	, 8616	, 9283	1.0772	1,1811	1,2232	650,03
5100	. 8590	.9269	1.0789	1.1850	1,2281	649.76
5200	, 8564	, 9255	1.0805	1.1891	1.2333	649.57
5300	, 8538	,9241	1.0821	1.1931	1,2382	649.30
5400	, 8512	. 9226	1.0839	1.1972	1,2434	649.10
5500	.8487	.9212	1,0855	1.2012	1.2485	648.84
5600	,8461	.9198	1,0872	1.2054	1.2538	648.64
5700	. 8435	. 9184	1.0889	2.2094	1,2587	<u>648.38</u>
5800	.8409	, 9171	1.0904	1.2136	1.2640	648.18
5900	.8384	.9156	1.0922	1.2177	1.2692	647.91
6000	.8358	. 9143	1.0937	1.2219	1.2746	647.71
6100	. 8333	. 9129	1.0954	1.2260	1,2797	647.45
6200	.8307	,9115	1,0971	1.2303	1,2851	647.25
6300	.8282	. 9101	1.0988	1.2344	1.2904	<b>646</b> ,99
6400	.8257	, 9087	1.1005	1,2387	1,2958	646.79
6500	. 8231	.9073	1.1022	1.2429	1.3010	646.52
6600	,8206	.9059	1.1039	1.2471	1.3064	646.26
6700	, 8181	. 9045	1,1056	1.2514	1.3119	646.06
6800	.8156	.9031	1.1073	1.2557	1.3174	645.80
6900	, 8131	.9017	1.1090	1,2601	1.3229	645.60

## 269

	_	8	/	_		_	
H <sub>c</sub>	Pa	-	1/8	Ta	$\sqrt{T_a}$	θ	<b>√</b> θ
(Feet)	<u>(''Hg)</u>	Pa/PaSL		<u>(•K)</u>		$T_a/T_{aSL}$	<u> </u>
7000	23.088	.7716	1.2959	274.30	16.562	.9519	.9756
7100	23.000	. 7687	1.3008	274.10	16,556	, 9512	.9753
7200	22.913	. 7657	1.3058	273.90	16.550	. 9505	.9749
7300	22.826	. 7628	1.3108	273.69	16.544	. 9498	.9746
7400	22.739	. 7599	1.3158	273.49	<u>16.538</u>	.9491	.9742
7500	22.653	. 7570	1.3208	273.29	16.532	. 9484	. 9739
<b>760</b> 0	22.567	. 7542	1.3258	273.09	16.525	.9477	.9735
7700	22,481	. 7513	1.3309	272.92	16.520	.9471	.9732
7800	22.395	. 7484	1.3360	272.71	16.514	.9464 .	. 9728
7900	22.310	. 7456	1.3411	272.51	16.508	، 9457	.9725
8000	22.225	. 7427	1.3462	272.31	16.502	. 9450	. 9721
8100	22.140	. 7399	1.3514	272.11	16.496	. 9443	.9718
8200	22.055	. 7371	1.3566	271.91	16.490	. 9436	.9714
8300	21.971	. 7343	1.3618	271.71	16.484	. 9429	.9710
8400	21.887	.7314	1.3670	271.50	<u>16.477</u>	. 9422	<u>, 9707</u>
8500	21.803	. 7286	1.3723	271.33	16.472	. 9416	. 9703
8600	21.719	. 7259	1,3776	271.13	16.466	. 9409	. 9700
8700	21.636	.7231	1.3829	270.93	<u>16.460</u>	. 9402	. 9696
8800	21.553	.7203	1.3882	270,73	16.454	. 9395	. 9693
8900	21.470	.7175	1.3935	270.52	16.448	. 9388'	. 9689
9000	21.388	. 7148	1.3989	270,32	16.441	. 9381	. 9686
9100	21.305	. 7120	1.4043	270.12	16.435	. 9374	. 9682
9200	21.223	. 7093	1.4098	269.92	16.429	. 9367	. 9679
<b>930</b> 0	21.142	. 7065	1.4152	269.75	16.424	. 9361	.9675
9400	21.060	. 7038	1.4207	269.54	<u>16.418</u>	. 9354	.9671
9500	20.979	. 7011	1.4262	269,34	16.412	. 9347	. 9668
9600	20,393	. 6984	1,4317	269.14	16.406	. 9340	. 9664
9700	20.817	. 6957	1.4372	268.94	16.399	. 9333	. 9661
980C	20.737	. 6930	1.4428	268.74	16.393	. 9326	. 9657
9900	20.656	. 6903	1.4484	268.54	16.387	. 9319	. 9654
10000	20.577	. 6877	1.4541	268,33	16.381	. 9312	. 9650
10100	20.497	. 6850	1.4597	268.16	16.376	. 9306	. 9647
10200	20.417	. 6823	1.4654	267.96	16.369	. 9299	. 9643
10300	20.338	. 6797	1.4711	267.76	16.363	. 9292	.9639
10400	20.259	.6771	1.4768	267.56	16.357	. 9285	. 9636
10500	20.180	. 6744	1.4826	267.35	16.351	.9278	. 9632
10600	20.102	. 6718	1.4884	267.15	16.345	. 9271	. 9629
10700	20.024	. 6692	1, 4942	266,95	16.339	. 9264	.9625
10800	19.946	. 6666	1.5001	266.75	16.332	.9257	.9622
10900	19.868	. 6640	1.5059	266.58	16.327	. 9251	.9618

				-1		
, H <sub>c</sub>	σ	σ	1/ √σ	5018	1/8/8	a
(Feet)	<u>p/psl</u>	- <b></b>	<del></del>			<u>(Knots)</u>
7000	.8106	. 9004	1.1106	1.2643	1.3283	645.34
7100	.8081	, 8990	1.1123	1.2688	1.3340	645.15
7200	.8056	. 8976	1.1141	1.2731	1.3395	644.88
7300	.8032	. 8962	1.1158	1.2775	1.3451	644.68
_7400	. 8007	. 8948	1.1176	1.2819	1.3506	<u>     644. 42</u>
7500	. 7982	. 8935	1.1192	1.2864	1.3564	644.22
7600	. 7958	. 8921	1.1210	1.2908	1.3621	643.96
7700	. 7933	. 8907	1,1227	1.2953	1,3678	643.76
7800	. 7909	. 8893	1,1245	1.2997	1.3734	643.59
7900	.7884	.8880	1.1261	1.3043	1.3793	643.29
8000	,7860	. 8866	1.1279	1.3087	1.3850	643.03
8100	7836	. 8852	1,1297	1,3133	1,3909	642,85
8200	. 7811	.8838	1.1315	1.3178	1.3966	642.57
8300	. 7787	. 8825	1,1331	1.3223	1.402	642.30
8400	. 7763	. 8811	1.1349	1.3270	1.408	<u>642,10</u>
8500	. 7739	. 8797	1.1368	1.3316	1.414	641.84
8600	.7715	.8784	1.1384	1.3363	1.420	641.64
8700			1,1403	1.3409	1.426	<u>     641.38 </u>
8800	. 7667	. 8756	1.1421	1.3456	1.432	641.18
8900	. 7643	.8743	1.1438	1.3503	1.438	640.91
9000	. 7619	. 8729	1.1456	1.3550	1.444	640.72
9100	. 7595	. 8715	1.1474	1.3597	1,450	<u>640.45</u>
9200	.7572	. 8702	1.1492	1.3645	1.456	640.25
9300	.7548	.8688	1.1510	1.3693	1.462	639.99
9400	. 7525	. 8675	1.1527	1,3740	1,469	<u>639.72</u>
9500	. 7501	. 8661	1.1546	1.3789	1.475	639.52
9600	. 7478	. 8648	1.1563	1.3836	1.481	639.26
9700	,7454	<u>, 8634</u>	1.1582	<u>1.3886</u>	1.487	<u>639.06</u>
9800	. 7431	.8621	1,1600	1.3934	1.494	638.79
9900	.7408	.8607	1.1618	1.3984	1.500	638.59
10000	. 7384	. 8593	1.1637	1.4032	1.506	638.31
10100	. 7361	, 8580	1,1655	1.4082	1,513	638.13
10200	. 7338	. 8566	1.1674	1.4131	1,519	637.86
10300	. 7315	. 8553	1.1692	1.4180	1.526	637.60
10400	.7292		1.1710	1.4231	1.532	637,40
10500	.7269	.8526	1.1729	1.4281	1.539	637.13
10600	.7246	.8513	1.1747	1.4332	1.545	636.94
10700	. 7223	. 8499	1.1766	1.4382	1,552	636,67
10800	. 7200	. 8486	1.1784	1.4434	1.559	636.47
10900	.7178	.8472	1.1804	1.4485	1.566	636.21

	_	5					
H <sub>c</sub>	Pa	0	1/8	Ta	$\sqrt{T_a}$	θ	<b>√</b> θ
(Feet)	(''Hg)	Pa/PaSL		(•K)		$T_{a}/T_{aSL}$	
11000.	19.790	.6614	1.5118	266.38	16.321	. 9244	.9614
11100	19.713	. 6588	1.5177	266.17	16.315	. 9237	.9611
11200	19.636	. 6562	1.5237	265.97	16.309	. 9230	.9607
11300	19.559	.6537	1.5297	265.77	16.302	. 9223	.9604
11400	19.483	.6511	1.5357	265.57	16,296	. 9216	. 9600
11500	19.407	. 6486	1.5417	265.37	16.290	.9209	.9597
11600	19.331	. 6460	1.5478	265.16	16.284	. 9202	.9593
11700	19.255	.6435	1.5539	264.99	<u>    16.279 </u>	. 9196	<u>. 9589</u>
11800	19.179	.6410	1.5600	264.79	16.272	. 9189	.9586
11900	19.104	.6384	1.5661	264.59	16.266	.9182	.9582
12000	19.029	.6359	1.5723	264.39	16.260	.9175	.9579
12100	18.954	6334	1,5785	264.19	16.254	. 9168	.9575
12200	18.879	. 6309	1.5848	263.98	16.248	. 9161	.9571
12300	18.805	. 6285	1.5910	263.78	16.241	.9154	. 9568
12400	18.731	. 6260	1.5973	263.58	16.235	. 9147	. 9564
12500	18.657	. 6235	1.6036	263.41	16.230	.9141	.9561
12600	18.583	. 6210	1.6100	263.21	16.224	.9134	.9557
12700	18.510	.6186	1,6164	263.00	<u>    16.217</u>	.9127	<u>.9553</u>
12800	18.437	. 6161	1.6228	262.80	16.211	.9120	.9550
12900	18.364	.6137	1.6293	262.60	16.205	.9113	.9546
13000	18.291	.6113	1.6357	262.40	16.199	.9106	.9543
<u>13100</u>	<u>18.219</u>	. 6089	1.6422	262.20	16.192	. 9099	.9539
13200	18.147	. 6064	1.6488	262.00	16.186	.9092	. 9535
13300	18.075	. 60 40	1.6554	261.82	16.181	.9086	.9532
<u>13400</u>	18,003	<u>6016</u>	1.6619	261,62	16.175	. 9079	<u>9528</u>
13500	17.931	. 5992	1.6686	261.42	16.168	.9072	.9525
13600	17.860	. 5969	1.6752	261.22	16.162	.9065	.9521
13700_	17.789	. 5945	<u>    1.6820    </u>	261.02	16.156	.9058	. 9517
13800	17.718	. 5921	1.6887	260.81	16.150	.9051	.9514
13900	17.647	. 5898	1.6954	260.61	16.143	.9044	.9510
14000	17.577	. 5874	1.7022	260.41	16.137	.9037	.9507
14100	17.507	. 5851	1.7090	260.24	16.132	. 9031	. 9503
14200	17.437	. 5827	1.7159	260.04	16.126	. 9024	. 9499
14300	17.367	. 5804	1.7228	259.83	16.119	.9017	.9496
14400	17.298	. 5781	1.7297	259.63	<u>16.113</u>	. 9010	.9492
14500	17.228	. 5758	1.7367	259.43	16.107	. 9003	.9488
14600	17.159	. 5735	1.7436	259.23	16.101	. 8996	.9485
14700	17.090	, 5712	1.7507	259.03	16.094	. 8989	<u>.9481</u>
14800	17.022	. 5689	1.7577	258.83	16.088	. 8982	.9478
14900	16.953	. 5666	1.7648	258.65	16.083	.8976	.9474

Hc	σ	Jo	$1/\sqrt{\sigma}$	$\sqrt{\theta}/\delta$	1/2/0	a
(Feet)	P/PSL	¥ -	-/ 4 -	• -• • • •		(Knots)
Contraction of the local division of the loc		0.450	1 1033	1 4525	1 6 7 2	
11000	. 7155	. 8459	1,1822	1.4535	1.572	635.94
11100	.7132	. 8446	1,1840	1.4588	1.579	635.75
11200	.7110	.8432	1.1860	1.4639	1.586	635.48
11300	. 7087	. 8419	1,1878	1.4692	1.593	635.28
11400	. 7065	.8406	1.1896	1.4743	1.599	635.03
11500	. 7043	.8392	1.1916	1.4796	1.6067	634.82
11600 11700	. 7020	.8379 .8366	1.1935 1, <u>1953</u>	1.4848 1.4901	1.6136 1.6206	634.56 <u>634.29</u>
11800	. 6976	. 8352	1.1973	1. 4955	1.6276	634.09
11900	. 6953	. 8339	1, 1992	1,5007	1.6345	633.83
11900	. 0955	. 0337	1, 1774	1, 5007	1.0545	055.05
12000	. 6931	.8326	1.2011	1,5062	1.6417	633.63
12100	. 6909	. 8312	1.2031	1.5115	1.6488	<u>633.36</u>
12200	. 6387	. 8299	1.2050	1.5168	1.6557	633.10
12300	. 6865	. 8286	1.2069	1.5224	1.6631	632.90
12400	. 6843	. 8273	1.2088	<u> </u>	1.6702	<u>632.64</u>
12500	. 6821	.8259	1,2108	1,5333	1.6776	632,44
12600	. 6000	. 8246	1.2127	1.5387	1.6346	632.17
12700	.6778	<u>. 8233</u>	1,2146	1.5442	1.6920	<u>631,91</u>
12800	. 6756	. 8220	1,2165	1,5498	1.6994	631.71
12900	. 6735	. 8207	1.2185	1.5553	1.7068	631.45
13000	. 6713	.8194	1.2204	1.5610	1.7142	631.25
13100	. 6691	<u>, 8180</u>	1.2225	1.5666	1.7217	<u>630.98</u>
13200	. 6670	. 8167	1.2244	1.5722	1.7292	630.72
13300	. 6648	. 8154	1.2264	1.5779	1.7368	630.52
<u>13400</u>	.6627	. 8141	1.2284	1,5835	<u>1.7442</u>	630.26
13500	. 6606	. 8128	1.2303	1.5894	1.7521	630.06
13600	.6584	.8115	1.2323	1.5950	1.7596	629.79
13700	. 6563	. 8102	1.2343	1.6008	1.7674	629.53
13800	. 6542	. 8089	1.2362	1.6066	1.7750	629.33
13900	. 6521	.8076	1.2382	1.6124	1.7828	629.07
14000	. 6500	. 8062	1.2404	1.6184	1.7908	628.87
14100	. 6479	. 8049	1.242.4	1,6241	1,7985	<u>628,60</u>
14200	. 6458	. 8036	1.2444	1.6300	1.8064	628.34
14300	. 6437	. 8023	1.2464	1.6360	1.8144	628.14
14400	.6416	. 8010	1.2484	1.6419	1.8224	<u>627.88</u>
14500	. 6395	. 7997	1.2505	1.6478	1.8304	627.61
14600	. 6375	. 7984	1.2525	1.6539	1.8386	627.41
14700	. 6354	. 7971	1.2545	1.6598	1.8464	627.15
14800	. 6333	. 7958	1.2566	1.6660	1.8548	626.95
14900	. 6313	. 7945	1.2587	1.6720	1.8629	626.69

.

-

TABLE 9.3

Н <sub>с</sub>	Pa	8	1/(	Ta	Ta	θ	τ <del>θ</del>
(Feet)	(''Hg)	Pa/PaSL	1/5	( • K)	1 ^ a	$T_a/T_a$	•••
			1 2210		16 0 26		0.430
15000 15100	16.885 16.817	.5643 .5620	1.7719 1.7791	258.45 258.25	16.076 16.070	.8969 .8962	.9470 .9467
15200	$-\frac{10.817}{16.750}$	. 5598	1,7863	258.05	16.064	. 8955	.9463
15300	16.682	. 5575	1.7935	257.85	16.058	.8940	.9405
15400	16.615	. 5553	1.8008	257.64	16.051	. 8941	.9456
15500	16.548	. 5530	1.8081	257.44	16.045	. 8934	.9452
15600	16.481	. 5508	1.8154	257,24	16.039	.8927	.9449
15700	16.414	, 5486	1.8228	257.07	16,033	. 8921	.9445
15800	16.348	. 5463	1.8302	256.87	16.027	. 8914	.9441
15900	16.282	5441	1.8376	256.66	16.021	. 8907	. 9438
16000	16.216	. 5419	1.8451	256.46	16.014	. 8900	.9434
16100	16.150	.5397	1.8526	256,26	16,008	. 8893	<u>9430</u>
16200	16.085	. 5375	1.8601	256.06	16.002	.8886	. 9427
16300	16.019	, 5354	1.8677	255.86	15.996	.8875	. 9423
16400	15.954	. 5332	1,8753	255.66	<u>15.989</u>	. 887?	. 9419
16500	15.889	. 5310	1.8830	255.48	15.984	. 8866	.9416
16600	15.825	. 5288	1.8907	255.28	15.978	.8859	. 9412
16700	15.760	, 5267	1.8984	255.08	15.971	.8852	.9408
16800	15.696	. 5245	1.9062	254,88	15.965	.8845	.9405
16900	15.632	. 5224	1.9140	254, 68	15.959	.8838	.9401
17000	15.568	, 5203	1.9218	254.47	15.952	. 8831	.9397
17100	15.505	. 5182	<u>    1.9297    </u>	254.27	15.946	. 8824	<u>9394</u>
17200	15.441	. 5160	1.9376	254.07	15,940	.8817	. 9390
17300	15.378	. 5139	1.9456	253.90	15,934	.8811	.9386
17400	15.315	. 5118	1.9536	253.70	15.928	. 8804	. 9383
17500	15.252	. 5097	1.9617	253.49	15.922	. 8797	.9379
17600	15.190	. 5076	1.9697	253.29	15.915	.8790	.9375
17700	15.127	. 5055	1.9778	253.09	15.909	.8783	. 9372
17800	15.065	. 5035	1.9860	252.89	15.902	.8776	. 9368
17900	15.003	. 5014	1.9942	252.69	15.896	.8769	.9364
18000	14,942	. 4993	2.0024	252,49	15.890	.8762	. 9361
<u>18100</u>	<u>14.880</u>	. 4973	2.0107	252, 31	15.884	. 8756	. 9357
18200	14.819	. 4952	2.0191	252,11	15.878	. 8749	. 9353
18300	14.758	. 4932	2.0274	251.91	15.872	.8742	. 9350
18400	14,697	. 4912	2.0358	251,71	15.865	. 8735	. 9346
18500	14.636	. 4891	2.0442	251.51	15.859	. 8728	.9342
18600	14.576	. 4871	2.0527	251,30	15.853	. 8721	.9339
18700	14.515	. 4851	2,0613	251,10	<u>15.846</u>	<u>, 8714</u>	<u>, 9335</u>
18800	14.455	. 4831	2,0698	250.90	15.840	. 8707	. 9331
18900	14.395	. 4811	2.0784	250.73	15.834	. 8701	.9328

Н <sub>с</sub>	σ	<b>√</b> σ	1/5	5018	1/50	a
(Feet)	P/PSL		-/ ( 0	1078	. 1010	(Knots)
	. 6292	7022	1 2607	1 6701	1 0713	
15000 _15100	. 6271	. 7932	1.2607 1.2626	1.6781 1.6843	1.8712 1.8794	626,42 626,22
15200	. 6251	. 7907	1.2647	1.6904	1.8877	625.96
15300	. 6231	. 7894	1.2663	1.6965	1,3960	625.69
15400	. 6210	. 7881	1.2689	1.7029	1.9047	625.50
15500	. 6190	. 7868	1.2710	1.7090	1.9129	625.23
15600	. 6170	. 7855	1.2731	1.7154	1.9216	625.03
15700	. 6149	. 7842	1.2752	1.7217	1.9301	624.77
15800	. 6129	. 7829	1.2773	1.7279	1.9386	624.50
15900	. 6109	. 7816	1.2794	1.7344	1.9473	624.30
16000	. 6089	. 7804	1.2814	1.7407	1.9559	624.04
16100	. 6069	. 7791	1.2835	1.7470	1.9645	623.78
16200	. 6049	. 7778	1.2857	1.7536	1.9735	623.58
16300	. 6029	. 7765	1.2878	1.7600	1.9822	623.30
16400	. 6009	. 7752	1.2900	1.7664	1.9909	623.04
16500	. 5990	. 7740	1.2920	1.7731	2.0001	622.84
16600	. 5970	. 7727	1,2942	1.7796	2.0090	622.58
16700	. 5950	. 7714	1.2963	1.7861	2.0178	622.31
16800	, 5931	. 7701	1.2985	1.7928	2.0270	622.11
16900	. 5911	. 7689	1.3006	1.7994	2.0361	621.85
17000	. 5891	. 7676	1.3028	1.8060	2.0451	621.58
17100	. 5872	. 7663	1.3050	1.8128	2,0543	621.39
17200	. 5853	. 7651	1.3070	1.8195	2.0636	621.12
17300	. 5833	. 7638	1.3092	1.8262	2.0728	620.86
17400		. 7625	1.3115	1,8331	2.0823	620.66
17500	. 5794	. 7612	1.3137	1.8399	2.0916	620.39
17600	. 5775 575Å	. 7600	1.3158	1.8467	2.1010	620.13
<u>17700</u> 17800	<u>.5756</u> .5737	<u>.7587</u> .7575	1.3180	1.8537	2.1106	619.93
17900	. 5718	. 7562	1.3201 1.3224	1.8605 1.8674	2.1201 2.1295	619.67
						619.40
18000	. 5699	. 7549	1.3247	1.8745	2.1393	619.21
18100	5680	. 7537	1.3268	1.8815	2.1490	618.95
18200	. 5661	. 7524	1.3291	1.8885	2.1587	618.68
18300	. 5642	. 7511	1.3314	1.8957	2.1686	618.49
18400	. 5623	7499	1.3335	1.9027	2.1783	618.22
18500	. 5604	. 7486	1.3358	1.9098	2.1882	617.96
18600	. 5585	. 7474	1.3380	1.9171	2.1983	617.76
<u>18700</u> 18800	5567	.7461	1.3403	1.9242	2,2081	617.49
18900	.5548 .5529	.7449 .7436	1.3425	1.9314	2.2182	617.23
10700	. 3364	. 1720	1.3448	1.9388	2.2284	617.03

Hc (Feet)	₽ <sub>a</sub> (''Eg)	$\int P_a/P_{aSL}$	1 /5	Та (°К)	$\sqrt{T_a}$	$\theta T_a/T_aSL$	√ <del>Θ</del>
19000	14.336	,4791	2.0871	250.53	15.828	. 8694	. 9324
19100	14.276	4771	2.0958	<u>250.32</u>	15.822	. 8687	. 9320
19200	14.217	, 4751	2.1045	250.12	15.815	. 8680	.9317
19300	14.158	. 4731	2,1133	249.92	15.809	. 8673	. 9313
<u>19400</u>	14.099	. 4 <u>712</u>	2.1221	249.72	15.803	. 8666	<u>. 9309</u>
19500	14,040	. 4692	2.1310	249.52	15.796	. 8659	. 9306
19600	13.982	. 4673	2.1399	249.32	15,790	. 8652	.9302
19700	13.523	. 4653	2.1489	249.14	15.784	. 8646	. 9298
19800	13.865	. 4634	2.1579	248.94	15.778	. 8639	. 9294
19900	13.807	. 4514	2.1659	248.74	15.771	. 8632	. 9291
20000	13.750	. 4595	2.1760	248.54	15.765	. 8625	. 9287
20100	13.692	. 4576	2.1852	248.34	15.759	. 8618	. 9283
20200	13.635	. 4557	2.1944	248.13	15.752	. 8611	. 9280
20300	13.578	. 4537	2.2036	247.93	15.746	.8604	.9276
20400	13.521	. 4518	2.2129	247.73	15.739	. 8597	.9272
20500	13.464	. 4500	2.2222	247.56	15.734	. 8591	. 9269
20600	13.407	. 4481	2.2315	247.36	15.728	.8584	.9265
20700	<u>13.351</u>	. 4462	2.2410	247.15	15.721	. 8577	<u>. 9261</u>
20800	13.295	. 4443	2.2504	246.95	15.715	. 8570	. 9257
20900	13.239	. 4424	2.2600	246.75	15.708	.8563	.9254
21000	13.183	. 4406	2,2695	246.55	15,702	. 8556	. 9250
21100	13.128	.4387	2.2792	246.35	15,695	. 8549	,9246
21200	13.072	. 4369	2.2888	246.15	15,689	. 8542	.9243
21300	13.017	. 4350	2.2985	245.97	15.684	. 8536	,9239
21400	12.962	. 4332	2.3082	245.77	15.677	. 8529	.9235
21500	12.907	. 4313	2.3180	245.57	15.671	.8522	,9231
21600	12.052	. 4295	2.3279	245,37	15,664	.8515	.9228
21700	12.798	.4277	2.3378	245.17	15.658	. 8508	.9224
21800	12.744	. 4259	2.3478	244.96	15, 351	.8501	.9220
21900	12.090	,4241	2.3578	244.76	15,645	. 8494	.9216
22000	12.636	. 4223	2.3678	244.56	15,638	.8487	.9213
22100	12.582	. 4205	2.378	244.39	15.633	. 8481	.9209
22200	12.529	.4187	2.388	244.19	15.626	. 8474	. 9205
22300	12.475	.4169	2.398	243.99	15,620	.8467	.9202
22400	12,422	. 4151	2.408	243.78	15.614		9198
22500	12.369	.4134	2.418	243,58	15.607	.8453	.9194
22600	12.316	. 4116	2.429	243.38	15.601	.8446	.9190
22700	12.264	. 4098	2.439	243.18	15.594	.8439	.9187
22800	12,211	. 4081	2.450	242.98	15.588	. 8432	.9183
22900	12.159	. 4063	2.460	242,80	15,582	.8426	.9179

17	-	ν	1/ Jo	5018	1/510	a
H <sub>c</sub> (Feet)	σ	40	1/ 40	4010	17010	(Knots)
	p/pSL					
19000	. 5511	.7424	1,3470	1.9460	2.2385	616,77
<u>19100</u>	. 5492	. 7411	1.3493	<u>1,9533</u>	2.2486	616,50
19200	, 5474	. 7399	1.3515	1.9608	2.259	616.30
19300	. 5455	. 7386	1.3539	1.9682	2.269	616.04
19400	. 5437	. 7374	1.3561	1,9756	2.279	615.77
19500	. 5419	. 7361	1.3585	1.9832	2.290	615.58
19600	. 5400	. 7349	1,3607	1.9906	2.300	615.31
<u>19700</u>	. 5382	. 7337	1.3630	1.9981	2.311	615.05
19800	. 5364	. 7324	1.3654	2.0056	2.321	614.73
19900	. 5346	. 7312	1.3676	2,0133	2.332	614.58
20000	. 5328	. 7299	1.3701	2.0209	2.343	614.32
20100	. 5310	. 7287	1,3723	2.0285	2.353	614.05
20200	. 5292	. 7275	1.3746	2,0364	2.364	613.86
20300	. 5274	. 7262	1.3770	2.0441	2.375	613.59
20400	. 5256	. 7250	<u>1.3793</u>	2.0518	2.386	<u>613,33</u>
20500	. 5238	. 7238	1.3816	2.0598	2.397	613,13
20600	. 5220	. 7225	1,3841	2.0676	2.408	612.86
20700	. 5202	. 7213	1.3864	2.0754	2.419	612.60
20800	. 5185	. 7201	1.3887	2,0833	2.431	612.34
20900	. 5167	.7188	1.3912	2.0914	2.442	612.14
21000	. 5149	. 7176	1.3935	2.0994	2.453	611.87
21100	. 5132	. 7164	1.3959	2.1074	2.465	611.61
21200	. 5114	. 7152	1.3982	2,1156	2.476	611.41
21300	. 5097	. 7139	1.4008	2.1236	2.488	611.15
21400	. 5079	. 7127	1.4031	2.1317	2.499	610.88
21500	. 5062	7115	1.4055	2.1398	2.511	610.62
21600	. 5044	, 7103	1.4079	2,1482	2.522	610.42
21700		. 7090	1.4104	2.1565	2.534	610.15
21800	.5010	. 7078	1.4128	2.1647	2,546	609.89
21900	. 4993	. 7066	1.4152	2.1730	2.558	609.62
22000	. 4975	7054	1.4176	2,1815	2.570	609.43
22100	. 4958	. 7042	1.4201	2.1899	2.582	609.16
22200	. 4941	, 7030	1.4225	2,1933	2.594	60 <b>8</b> .90
22300	. 4924	. 7017	1.4251	2,2070	<b>Z</b> . 606	60 <b>8</b> , 70
22400	. 4907	. 7005	1.4276	2.2155	2.618	. 30 <b>8.43</b>
22500	. 4890	. 6993	1 4300	2.2240	2.631	608,17
22600	. 4873	. 6981	1.4325	2,2325	2.643	607.90
22700	. 4856	. 6969	1,4349	2,2414	2.656	607.71
22800	. 4840	. 6957	1.4374	2,2500	2.668	607.43
22900	. 4823	. 6945	1.4399	2.2587	2.680	607.17

÷

TABLE 9.3

۰.

Hc	Pa	8	1/8	Ta	$\sqrt{T_a}$	θ	νθ
(Feet)	<u>("Hg)</u>	$P_a/P_{aSL}$		( ° Ř)		$T_a/T_{aSL}$	•
23000	12.107	. 4046	2.471	242.60	15.576	. 8419	.9175
23100	12.055	. 4029	2,481	242.40	15,569_	.8412	.9172
23200	12.003	. 4011	2.492	242.20	15.563	. 8405	.9168
23300	11.952	. 3994	2.503	242.00	15.556	. 8398	.9164
23400	11.901	. 3977	2,514	241.80	15,550	, 8391	,9160
23500	11.849	. 3960	2.525	241.59	15.543	.8384	.9157
23600	11.798	. 3943	2.535	241.39	15.537	.8377	.9153
23700	11.748	. 3926	2.546	241,22	15.531	. 8371	,9149
23800	11.697	. 3909	2.557	241.02	15,525	. 8364	.9145
23900	11.646	. 3892	2.569	240.82	15.518	.8357	.9142
<b>24</b> 0 <b>00</b>	11.596	. 3875	2.580	240.61	15.512	. 8350	.9138
24100	11,546	. 3859	2.591	240.41	15,505	. 8343	.9134
24200	11.496	. 3842	2.602	240.21	15.499	. 8336	.9130
24300	11.446	.3825	2.613	240.01	15.492	. 8329	.9126
24400	11.397	, 3809	2,625	239,81	<u>15.486</u>	. 8322	.9123
24500	11.347	. 3792	2.636	239.63	15,480	. 8316	.9119
24600	11.298	. 3776	2.648	239.43	15.474	.8309	.9115
24700	11.249	.3759	2,659	239.23	15.467	.8302	.9111
24800	11.200	.3743 .3727	2,671	239.03	15,461 15,454	.8295	.9108
24900	11.152	. 5727	2,683	238.83	15,454	.8288	.9104
25000	11.103	. 3710	2.694	238.63	15.448	. 8281	,9100
25100	11.055	3694	2,706	238,42	<u>15.441</u>	,8274	.9096
25200	11.006	. 3678	2.718	238.22	15.434	.8267	. 9093
25300	10.958	. 3662	2.730	238.05	15.429	.8261	.9089
25400	10.911	. 3646	2.742	237.85	15.422	.8254	. 9085
25500	10.863	. 3630	2.754	237.65	15.416	. 8247	. 9081
25600 25700	10.815 10.768	.3614 .3598	2.766	237.44 237.24	15.409	.8240 .8233	.9077
25800	10.721	.3583	<u>2.778</u> 2.790	237.04	<u>15,403</u> 15,396	.8235	<u>.9074</u> .9070
25900	10.674	. 3567	2.803	236.84	15.390	.8219	.9066
-					- ,		
26000	10.627	. 3551	2.815	236.64	15,383	.8212	. 9062
26100	10.580	3536	2,827	236.46	15.377	. 8206	.9058
26200	10.534	.3520	2.840	236.26	15.371	.8199	.9055
26300	10.487	.3505	2.852	236.06	15.364 15.358	.8192	.9051
<u>26400</u> 26500	<u>10,441</u> 10,395	.3489	2.865	<u>235,86</u> 235,66	15.351	<u>, 8185</u>	.9047
26500	10.349	.3474 .3459	2.878 2.891	235,00	15.351	.8178 .8171	.9043 .9039
26700	10.344	. 3459	2.903	235, 25	15,345	.8164	.9039
26800	10.258	, 3428	2.916	235.05	15.331	.8157	.9032
26900	10.213	. 3413	2.929	234,88	15.326	.8151	.9028
	••••		, . /				. /

Hc	σ	<u>σ</u>	1/ 5	5018	1/810	a
(Feet)	p/psl		· •			(Knots)
23000	, 4806	. 6933	1.4424	2.2674	2.693	606.90
23100	. 4789	. 6921	1.4449	2,2764	2,706	606,70
23200	. 4773	. 6909	1.4474	2.2853	2.719	606.44
23300	. 4756	. 6897	1.4499	2,2941	2.731	606.18
23400	. 4740	. 6885	1.4524	2.3030	2.744	605.91
23500	. 4723	. 6873	1.4550	2.3122	2.757	605,71
23600	. 4707	. 6861	1.4575	2.3212	2.770	605.45
23700	4690	. 6849	1,4601	2,3302	2,783	605,18
23800	. 4674	. 6837	1.4626	2.3392	2.797	604.92
23900	.4657	.6825	1.4652	2.3486	2.810	604, 72
24000	. 4641	. 6813	1.4678	2.3578	2.823	604.46
24100	. 4625	. 6801	1.4704	<u> </u>	2.837	604.19
24200	. 4609	. 6789	1.4730	2.3762	2.850	603.93
24300	. 4593	. 6777	1.4756	2.3854	2.864	603.66
24400	. 4577	. 6765	1.4782	2.3951	2.877	603.46
24500	. 4560	. 6753	1.4808	2.404	2.891	603.20
24600	. 4544	. 6742	1.4832	2.413	2.905	602.93
24700	. 4528	. 6730	1.4859	2.423	2.919	602,67
24800	. 4512	. 6718	1.4885	2.433	2.933	602.47
24900	. 4497	. 6706	1.4912	2.442	2.947	602.21
25000	. 4481	.6694	1.4939	2.452	2.961	601.94
25100	. 4465	, 6682	1.4966	2.461	2.975	601.68
25200	. 4449	. 6671	1.4990	2.471	2.990	601.48
25300	. 4433	. 6659	1.5017	2.481	3.004	601.22
25400	. 4418	6647	1.5044	2,491	3.018	<u>600.95</u>
25500	. 4402	. 6635	1.5072	2.501	3.033	600.69
25600	. 4387	. 6623	1.5099	2.511	3.047	600.42
<u>25700</u>	. 4371	6612	1.5124	2.521	3,062	600.22
25800	. 4355	. 6600	1.5152	2,531	3,077	599.96
25900	. 4340	. 6588	1.5179	2.541	3.092	599.69
26000	. 4324	. 6576	1.5207	2.551	3.106	599.43
26100	. 4309	. 6565	1.5232	2.561	3,121	<u>_599.17</u>
26200	. 4294	. 6553	1.5260	2.572	3.137	598.97
26300	. 4278	. 6541	1.5288	2.582	3.152	598.70
26400	. 4263	6530	1.5314	2.592	3.167	<u> </u>
26500	. 4248	. 6518	1.5342	2.602	3.182	598.17
26600	. 4233	. 6506	1.5370	2.613	3.198	597.91
<u>26700</u>	. 4218	. 6495	1.5396	2.623	3.214	597.71
26800	. 4203	. 6483	1.5425	2.634	3.229	597.45
26900	. 4188	.6472	1,5451	2.644	3,245	597.18

, н <sub>с</sub>	Pa	8	1/5	Ta	$\sqrt{T_a}$	θ	Vθ
(Feet)	<u>(''Hg)</u>	Pa/PaSL		(°K)		$T_a/T_{aSL}$	
27000	10.166	. 3398	2.942	234.68	15.319	. 8144	. 9024
27100	10,123	3383	2,955	234,48	15.313	.8137	,9020
27200	10.078	. 3368	2.968	234.27	15.306	. 8130	. 9017
27300	10.033	. 3353	2.982	234.07	15.299	.8123	.9013
27400	<u> </u>	. 3338	2.995	233.87	15.293	. 8116	. 9009
27500	9.944	. 3323	3.008	233.67	15.286	.8109	.9005
27600	9.900	.3308	3.022	233.47	15.280	.8102	.9001
27700	9.856	.3294	3.035	233, 29	<u>15.274</u> 15.267	. 8096	.8998
27800 27900	9.812 9.768	.3279 .3264	3.049 3.063	233.09 232.89	15.261	.8089 .8082	.8994 .8990
28000	9.724	. 3250	3.076	232.69	15.254	.8075	. 8986
28100	9.681	. 3235	3.090	232.49	15.248	. 8068	.8982
28200	9.638	. 3221	3.104	232.29	15.241	. 8061	.8978
28300	9.595	.3206 .3192	3.118 3.132	232.08 231.88	15.234	.8054	.8975
<u>28400</u> 28500	<u>9,552</u> 9,509	.3178	3.132	231.71	15,222	. 8041	.8967
28500	9.309	.3163	3.140	231.51	15.215	.8034	.8963
28700	9,424	, 3149	3,175	231, 31	15.209	.8027	.8959
28800	9.381	. 3135	3.189	231,10	15,202	. 8020	. 8955
28900	9.339	. 3121	3.203	230.90	15.195	.8013	.8952
-	0 207	.3107	3,218	230.70	15,189	. 8006	. 8948
29000 29100	9.297 9.255	. 3093	3.232	230.50	15,182	. 7999	.8944
29200	9.213	. 3079	3.247	230.30	15.176	. 7992	. 8940
29300	9.172	. 3065	3.262	230.12	15.170	. 7986	. 8936
29400	9.130	. 3051	3.276	229.92	15.163	. 7979	.8932
29500	9.089	. 3037	3.291	229.72	15,157	. 7972	. 8928
29600	9.048	. 3024	3.306	229,52	15.150	. 7965	. 8925
29700	9.007	. 3010	3.321	229.32	15,143	. 7958	. 8921
29800	8.966	. 2996	3.337	229.12	15.137	. 7951	.8917
29900	8.925	. 2983	3.352	228.91	15,130	.7944	.8913
30000	8.885	. 2969	3.367	228.71	15.123	.7937	.8909
30100	8.845	. 2956	3.382	228.54	15.118	. 7931	. 8905
30200	8.804	. 2942	3.398	228.34	15.111	. 7924	. 8901
30300	8.764	. 2929	3.413	228.14	15.104	. 7917	.8898
<u>30400</u>	8,724	. 2915	3.429	227.93	15,098	. 7910	.8894
30500	8.685	,2902 ,2889	3.445	227.73	15.091	.7903	.8890
30600	8.645 8.605	. 2889 . 2876	3.460 3.476	227.53 227.33	15.084 15.077	.7896 .7889	. 5886
<u>30700</u> 30800	8.566	. 2863	3.470	227.13	15.071	. 7882	.8882
30900	8.527	. 2849	3.508	226.95	15,065	. 7875	.8874
30,00	0.001	• • /		,_			10011

Н <sub>с</sub>	σ	<u>∫</u> π	1/ <i>(</i> <del>,</del>	1018	1/810	a
(Feet)	P/PSL	4 u	1/ 40	10/0	1 /610	(Knets)
		·				
27000	. 4173	. 6460	1.5480	2.655	3.260	596.92
27100	4157	. 6448	1.5509	2,666	3.276	<u> </u>
27200	. 4143	. 6437	1.5535	2.677	3.292	596.45
27300	. 4128	.6425	1.5564	2.637	3.308	596.19
27400	4113		1.5593	2.698	3.325	<u> </u>
27500	. 4098	. 6402	1.5620	2.709	3.341	595,66
27600	. 4083	. 6390	1.5649	2.720	3.357	595.40
27700	. 4068	. 6379	1.5676	2.731	3.374	595.20
27800	. 4054	. 6367	1.5706	2.742	3.390	594.93
27900	. 4039	.6356	1.5733	2.753	3.407	594,67
28000	. 4025	. 6344	1.5763	2,764	3,423	594.40
<u>28100</u>	. 4010	. 6333	1.5790	2.776	3,440	<u>594.14</u>
28200	. 3996	. 6321	1.5820	2.787	3.457	593.88
28300	. 3981	.6310	1.5848	2.798	3.474	593.68
28400	. 3967	. 6298	1.5878	2.810	3.491	593.41
28500	. 3952	. 6287	1.5906	2,821	3.509	593.15
28600	. 393.8	. 6276	1.5934	2.833	3,526	592.88
28700	. 3923	. 6264	<u>1.5964</u>	2,844	3.543	592.62
28500	. 3909	, 6253	1.5992	2.856	3,561	592,35
28900	. 3895	. 6241	1.6023	2,867	3.579	592.16
29000	. 3881	. 6230	1.6051	2.879	3,596	591.89
29100	. 3867	, 6219	1.6080	2.891	3,614	591.62
29200	. 3853	, 6207	1.6111	2.903	3.632	591.35
29300	, 38 <b>38</b>	,6196	1.6139	2,915	3,650	591.09
29400	. 3824	. 6184	1.6171	2,927	3.668	<u>590.82</u>
29500	. 3810	.6173	1.6200	2.939	3,686	590,56
29600	. 3796	.6162	1.6228	2,951	3. <b>7</b> 05	590.36
<u>29700</u>	. 3782	. 61 50	1.6260	2,963	3.723	590.10
29800	. 3769	. 6139	1.6289	2,975	3.742	589.83
29900	. 3755	, 6128	1.6319	2.987	3.761	589.57
30000	. 3741	.6117	1.5348	3.000	3.779	589.30
30100	. 3727	. 6105	1.6380	3.012	3.798	589.05
30200	, 3713	. 6094	1.6410	3.024	3,817	588,78
30300	. 3700	. 6083	1.6439	3,037	3.837	588.58
<u>30400</u>	<u>. 3686</u>	. 6072	1.6469	3.050	3.856	588.32
30500	, 3672	. 60 60	1.6502	3.062	3,875	588.06
30600	.3659	. 60 4 9	1.6532	3.075	3.894	587,79
<u>30700</u>	. 3645	. 6038	1.6562	3.088	3.914	587.53
30800	. 3632	,6027	1.6592	3.100	3.934	587,26
30900	.3618	.6016	1.6622	3,113	3.953	587.00

.

H <sub>c</sub>	Pa	8	1/5	Ta		θ	νθ
(Feet)	("Hg)	$P_a/P_{aSL}$	- 10	(*K)	1-2	$T_a/T_{aSL}$	••
-							
31000	8.488	. 2836	3.524	226.75	15.058	. 7869	,8871
31100	8.449	.2823	3.541	226.55	15.052	, 7862	.8867
31200	8.410	,2810	3.557	226.35	15.045	. 7855	. 8863
31300	8.371	. 2798	3.573	226.15	15.038	. 7848	. 8859
31400	8.333	.2785	3.590	225.95	15.032	. 7841	.8855
31500	8.295	. 2772	3.607	225.74	15.025	. 7834	. 8851
31600	8.256	. 2759	3.623	225.54	15.018	. 7827	.8847
<u>31700</u>	8,218	.2746	3.640	225.37	15.012	. 7821	.8843
31800	8.181	.2734	3.657	225.17	15.006	. 7814	.8839
31900	8.143	. 2721	3.674	224.97	14.999	.7807	. 8836
32000	8.105	. 2709	3.691	224.76	14.992	. 7800	.8832
32100	_8,068	, 2696	3,708	224.56	14,985	, 7793	. 8828
32200	8.030	. 2684	3,725	224.36	14.979	. 7786	. 8824
32300	7.993	, 2671	3.743	224.16	14.972	.7779	.8820
<u>32400</u>	7,956	, 2659	3,760	223.96	14,965	,7772	, 8816
32500	7.919	. 2646	3,778	223.79	14.959	. 7766	.8812
32600	7.882	, 2634	3.795	223.58	14.953	.7759	.8808
<u>32700</u>	7,846	, 2622	3.813	223.38	14.946	. 7752	.8804
32800	7.809	. 2610	3,831	223.18	14.939	.7745	. 8801
32900	7.773	. 2597	3.849	222.98	14,932	. 7738	.8797
33000	7.737	. 2585	3.867	222.78	14.926	.7731	.8793
33100	7,700	2573	3,885	222,57	14.919	.7724	<u>.8789</u>
33200	7.665	. 2561	3,903	222.37	14.912	. 7717	.8785
33300	7.629	. 2549	3.922	222.20	14.906	.7711	.8781
<u>33400</u>	7,593	. 2537	3.940	222,00	14,900	.7704	.8777
33500	7.557	. 2525	3.958	221.80	14,893	. 7697	.8773
33600	7.522	.2514	3.977	221.60	14,886	. 690	. 8769
<u>33700</u>	7.487	, 2502	3.996	221,39	14,879	. 7683	<u>, 8765</u>
33800	7.452	. 2490	4,015	221.19	14.873	. 7676	.8761
33900	7,417	. 2478	4.034	220.99	14.866	. 7669	.8757
34000	7,382	. 2467	4,053	220.79	14.859	. 7662	.8754
<u>34100</u>	7,347	, 2455	4.072	220,62	<u>14,853</u>	. 7656	8750
34200	7.312	, 2444	4.091	220.41	14.846	.7649	. 8746
34300	7.278	. 2432	4.110	<b>2</b> 20.21	14.840	. 7642	. 8742
<u>34400</u>	7,244	, 2421	4.130	220.01	14,833	. 7635_	. 8738
34500	7.209	. 2409	4.150	219.81	14.826	. 7628	.8734
34600	7.175	, 2398	4.169	219.61	14.819	. 7621	. 8730
<u>34700</u>	7,141	. 2386	4.189	219.41	14.812	. 7614	. 8726
34800	7.107	. 2375	4.209	219.20	14.806	. 7607	. 872.2
34900	7.074	. 2364	4.229	219.03	14.800	.7601	.8718

	_	( <del></del>		5015	1/6/0	
H <sub>c</sub>	σ	σ	1/√σ	1010	1/910	a (Knote)
(Feet)	P/PSL	<u> </u>			<u> </u>	(Knots)
31000	. 3605	. 6004	1,6656	3,127	3.974	586.80
<u>31100</u>	. 3592	. 5993	1.6686	3.140	3.994	<u>586,53</u>
31200	. 3578	. 5982	1.6717	3.153	4.014	586.27
31300	, 3565	. 5971	1.6748	3.166	4.034	586.01
<u>31400</u>	.3551	. 5960	1.6779	3.179	4.054	585.74
31500	. 3538	. 5949	1.6810	3.192	4.075	585,48
31600	, 3525	. 5938	1.6841	3.205	4,095	585.21
31700	. 3512	. 5926	1.6875	3.219	4.116	<u>584.95</u>
31800	. 3499	. 5916	1.6903	3.232	4.137	584.68
31900	. 3486	. 5904	1.6938	3.246	4.158	584.48
32000	. 3473	. 5893	1.6969	3.260	4.180	584.22
32100	. 3460	. 5882	1,7001	3.274	4.201	<u>583.96</u>
32200	. 3447	. 5871	1,7033	3.287	4.222	583,69
32300	. 3434	. 5860	1.7065	3.301	4.244	583.43
32400	. 3421	<u>. 5849</u>	<u>1,7097</u>	3.315	4.265	583,16
32500	. 3408	. 5838	1.7129	3.329	4.287	582.90
32600	. 3395	. 5827	1.7161	3.343	4.309	582.63.
<u>32700</u>	. 3382	. 5816	1,7194	3.357	4.331	582.37
32800	. 3370	. 5805	1.7227	3.371	4.353	582,17
32900	. 3357	. 5794	1.7259	3,386	4.376	581.91
33000	. 3344	. 5783	1.7292	3.400	4.398	581.64
33100	3332	5772	1.7325	3.414	4. 421	581.38
33200	. 33/19	. 5762	1.7355	3.429	4.443	581.11
3330ι	. 3306	. 5751	1.7388	3.443	4.466	580.85
<u>33400</u>	, 3294	. 5740	1.7422	3,458	4.489	580.58
33500	. 3281	. 5729	1.7455	3,473	4.512	580.32
33600	. 3269	. 5718	1.7489	3,487	4.535	580.05
<u>33700</u>	. 3256	. 5707	<u>1,7522</u>	<u> </u>	4.559	<u>579.79</u>
33800	. 3244	. 5696	1.7556	3.517	4.582	579.52
33900	. 3232	. 5685	1.7590	3, 532	4.606	579.26
34000	. 3219	. 5674	1.7624	3.548	4.630	579.06
34100	. 3207	<u> </u>	1.7655	3,563	4,654	<u>578.80</u>
34200	. 3195	. 5653	1.7690	3.578	4.678	578.53
34300	. 3183	. 5642	1.7724	3,593	4.703	578,27
34400	. 3171	<u>, 5631</u>	1.7759	3.609	4.727	578.00
34500	. 3158	. 5620	1.7794	3,624	4.751	577,74
34600	. 3146	.5610	1.7825	3,640	4.776	577.47
34700	. 3134	<u> </u>	1.7860	3,655	4.801	577.21
34800	. 3122	. 5588	1.7895	3,671	4.826	576.95
34900	. 3110	. 5577	1.7931	3.687	4.851	576.68

•

•

TABLE 9.3

Hc	Pa	8	1/8	Тa	$\sqrt{T_a}$	θ	<del>√ 0</del>
(Feet)	("Hg)	$P_a/P_{aSL}$	0	( • K)	· - 4	$T_a/T_{aSL}$	
35000	7.040	, 2353	4.249	218.83	14.793	. 7594	.8714
35100	7.007	.2341	4,210	218.63	14,786	,7587	.8710
35200	6.973	. 2330	4.290	218,43	14.779	. 7580	.8706
35300	6.940	. 2319	4.311	218.22	14.772	.7573	.8702
35400	6.907	. 2308	4.331	218.02	14.766	,7566	, 8698
35500	6.874	. 2297	4.352	217.82	14.759	. 7559	.8694
35600	6,841	. 2286	4.373	217.62	14.752	. 7552	.8690
<u>35700</u>	<u>6.809</u>	. 2275	4.394	217.45	14,746	,7546	. 8686
35800	6.776	. 2264	4.415	217.24	14.739	. 7539	. 8683
35900	6.744	. 2254	4.436	217.04	14.732	.7532	.8679
36000	6.711	. 2243	4.457	216,84	14.726	. 7525	.8675
36100	6.679	. 2232	4.479	216,66	14.719	.7519	
36200	6.647	. 2221	4.501	216.66	14.719	.7519	.8671
36300	6.615	. 2211	4.522	216.66	14.719	. 7519	.8671
36400	6.584	. 2200	4.544	216.66	<u>14.719</u>	. 7519	.8671
36500	6.552	. 2189	4.566	216.66	14.719	.7519	.8671
36600	6.521	.2179	4.588	216.66	14,719	.7519	.8671
36700	6.489	. 2168	4.610	216.66	14,719	.7519	.8671
36800	6.458	. 2158	4.632	216.66	14.719	.7519	,8671
36900	6.427	. 2148	4.655	216.66	14.719	. 7519	,8671
37000	6.396	. 2137	4.677	216.66	14.719	.7519	.8671
37100	6.366	, 2127	4,700	216,66	<u>14,719</u>	<u>.,7519</u>	<u>. 8671</u>
37200	6.335	, 2117	4.722	216.66	14.719	.7519	.8671
37300	6.305	. 2107	4.745	216.66	14.719	.7519	.8671
37400	6.274	. 2097	4.768	216.66	14,719	. 7519	.8671
37500	6.244	. 2087	4.791	216.66	14.719	. 7519	.8671
37600	6.215	. 2077	4.814	216.66	14.719	.7519	.8671
37700	6.185	. 2067	4.837	216.66	14.719	.7519	.8671
37800 37900	6.155 6.125	. 2057	4.861	216.65	14.719	.7519	.8671
-		. 2047	4.884	216,66	14.719	.7519	.8671
38000	6.096	. 2037	4.908	216.66	14.719	.7519	.8671
38100	6.067	. 2027	4.931	216.66	<u>14.719</u>	. 7519	. 8671
38200	6.038	. 20180	4.955	216.66	14.719	. 7519	.8671
38300	6.009	. 20083	4.979	216.66	14.719	. 7519	. 8671
38400	5.980	. 19987	5.003	216.66	<u>14.719</u>	<u>. 7519</u>	. 8671
38500	5,951	. 19892	5.027	216.66	14.719	. 7519	,8671
38600	5.923	.19797	5.051	216.66	14.719	.7519	.8671
<u>38700</u>	5.894	. 1970)	5.075	216.66	14.719	. 7519	.8671
38800	5.866	. 19607	5.100	216.66	14.719	. 7519	.8671
38900	5.838	.19513	5.124	216.66	14.719	.7519	.8671

Н <sub>с</sub>	σ	্ব	$1/\sqrt{\sigma}$	5018	1/8/0	а
(Feet)	-	10	17 4 0	1010	1/010	(Knots)
	p/psl		·			
35000	. 3098	. 5567	1.7963	3.703	4.877	576.42
35100	. 3086	. 5556	1.7999	3.719	4.902	576.15
35200	. 3075	. 5545	1.8034	3.735	4.928	575.89
35300	. 3063	. 5534	1.8070	3.751	4.953	575.62
35400	. 3051	. 5524	1.8103	3.767	4.979	575.36
35500	. 3039	. 5513	1,8139	3.783	5.005	575.09
35600	. 3027	. 5502	1.8175	3.800	5.032	574.83
35700	. 3016	. 5492	1.8208	3,816	5.058	574.57
35800	.3004	. 5481	1.8245	3.833	5.085	574.37
35900	. 2992	. 5471	1.8278	3.850	5.112	574.10
36000	. 2981	. 5460	1.8315	3.867	5.139	573.84
36100	. 2969	. 5449	1.8352	3,884	5,166	<u>573,58</u>
36200	. 2954	. 5436	1.8396	3.902	5.190	573.58
36300	. 2940	. 5423	1.8441	3.921	5.215	573,58
<u>36400</u>	. 2926	. 5410	1.8485	3.940	5.240	<u> </u>
36500	. 2912	. 5397	1.8530	3.959	5.266	573.58
36600	. 2898	. 5384	1.8574	3.978	5.291	573.58
36700	. 2884	. 5371	<u>1.8619</u>	3.997	5.317	573,58
36800	. 2870	. 5358	1.8664	4.017	5.342	573,58
36900	. 2857	. 5345	1.8709	4.036	5,368	573.58
37000	. 2843	. 5332	1.8753	4.055	5.394	573.58
37100	<u>, 2829</u>	. 5319	1.8799	4.075	5.420	<u>573.58</u>
37200	. 2816	. 5307	1.8844	4.095	5.446	573.58
37300	. 2802	. 5294	1.8889	4.114	5.472	573.58
37400	. 2789	. 5281	1.8935	4.134	<u> </u>	<u>573.58</u>
37500	. 2775	. 5269	1.8980	4.154	5.525	573.58
37600	. 2762	. 5256	1.9026	4.174	5.552	573.58
37700	. 2749	. 5243	1.9072	4.194	<u>5.579</u>	<u> </u>
37800	. 2736	. 5231	1.9117	4.215	5.605	573.58
37900	. 2723	. 5218	1.9164	4.235	5.632	573.58
38000	. 2709	. 5206	1.9210	4.255	5.660	573.58
38100	. 2697	<u>. 5195</u>	1.9256	4.276	5.687	573.58
38200	. 2684	. 5181	1.9302	4.296	5.714	573.58
38300	. 2671	. 5168	1.9349	4.317	5.742	573.58
<u>38400</u>	, 2658	<u>, 5156</u>	1,9395	4,338	5.770	573,58
38500	. 2645	. 5144	1.9442	4.359	5.797	573.58
38600	. 2633	. 5131	1.9488	4.380	5.825	573.58
38700	. 2620	5119	1,9535	4,401	<u>5,853</u>	<u>573, 58</u>
38800	. 2607	. 5107	1.9583	4.422	5,881	573.58
38900	. 2595	. 5094	1.9630	4.443	5.910	573.58

1.1.1.1.1.1.1.1.1.1.1.1.1

	_	8	(	_	<b></b>		
H <sub>c</sub>	Pa		1/8		$\sqrt{T_a}$	θ	V O
<u>(Feet)</u>	(''Hg)	$P_a/P_{aSL}$		<u>(°</u> K)		$T_a/T_{aSL}$	
39000	5.810	.19419	5.149	216.66	14.719	. 7519	.8671
<u>39100</u>	5.782	. 19326	5.174	216.66	14.719	. 7519	.8671
39200	5.754	. 19233	5.199	216.66	14.719	, 7519	.8671
39300	5.727	. 19142	5,224	216.66	14.719	, 7519	, 8671
<u>39400</u>	5.699	. 19049	5,249	216.66	14.719	. 7519	. 8671
39500	5.672	.18958	5.274	216.66	14.719	. 7519	.8671
39600	5,645	. 18868	5.300	216,66	14.719	. 7519	. 8671
<u>39700</u>	5.618	. 18776	5.325	216.66	14.719	. 7519	.8671
39800	5.591	.18687	5.351	216.66	14.719	. 7519	. 8671
39900	5.564	.18597	5.377	216.66	14.719	. 7519	.8671
40000	5,537	. 18508	5.403	216,66	14.719	. 7519	.8671
40100	5,511	.18419	5.429	216.66	14.719	. 7519	. 8671
40200	5.484	. 18331	5.455	216.66	14.719	. 7519	. 8671
40300	5.458	.18243	5,481	216.66	14.719	. 7519	.8671
40400	5.432	. 18155	5.508	216,66	14.719	. 7519	. 8671
40500	5.406	. 18068	5.534	216.66	14.719	. 7519	. 8671
<b>4060</b> 0	5.380	.17982	5.561	216,66	14.719	. 7519	.8671
40700	5,354	. 17896	5.587	216.66	14.719	. 7519	.8671
40800	5,328	. 17810	5.614	216,66	14.719	, 7519	.8671
<b>40900</b>	5,303	.17724	5,642	216.66	14.719	. 7519	.8671
41000	5.278	.17640	5,668	216.66	14.719	. 7519	.8671
41100	5,252	.17555	5.696	216.66	14.719	, 7519	. 8671
41200	5.227	.17471	5.723	216.66	14.719	. 7519	.8671
41300	5,202	.17387	5.751	216,66	14.719	.7519	.8671
<u>41400</u>	5,177	.17303	<u>5,779</u>	216,66	14.719	. 7519	.8671
41500	5,152	. 17221	5.806	216.66	14.719	. 7519	. 8671
<b>4160</b> 0	5.127	.17138	5.835	216.66	14.719	.7519	. 8671
<u>41700</u>	5,103	17056	5.863	216.66	14,719	.7519	.8671
41800	5.079	. 16974	5.891	216.66	14.719	. 7519	. 8671
41900	5.054	.16893	5.919	216.66	14.719	. 7519	. 8671
42000	5.030	,16812	5,948	216.66	14.719	. 7519	.8671
42100	5.006	. 16731	5.976	216.66	14.719	. 7519	. 8671
42200	4,982	, 16651	6.005	216.66	14,719	. 7519	,8671
42300	4.958	. 16571	6.034	216.66	14.719	.7519	. 8671
42400	4,934	, 16492	6,063	216,66	14.719	. 7519	. 8671
42500	4,910	. 16412	6.093	216.66	14.719	, 7519	, 8671
42600	4.887	. 16334	6.122	216.66	14.719	. 7519	. 8671
42700	4,863	, 16255	6.152	216.66	14.719	. 7519	<u>.8671</u>
42800	4.840	. 16178	6.181	216.66	14.719	. 7519	.8671
42900	4.817	.16100	6.211	216.66	14.719	.7519	. 8671

a à straiter

H <sub>c</sub> (Feet)	σ <u>ρ/psl</u>	10	1/ <del>\</del>	1018	1/6(8	a (Knots)
a a a a a a a a a a a a a a a a a a a						
39000	. 2582	. 5082	1.9677	4.465	5.938	573.58
<u>39100</u>	. 2570	, 5070	1.9724	4,486	5.967	<u> </u>
39200	. 2558	. 5058	1.9772	4.508	5.996	573,58
39300	. 2545	. 5046	1.9819	4.529	6.024	573,58
<u>39400</u>	. 2533	. 5033	1,9867	4.551	6.054	573.58
39500	. 2521	. 5021	1.9915	4.573	6.083	573.58
39600	. 2509	. 5009	1.9962	4.595	6.112	573.58
<u>39700</u>	. 2497	4997	2.0011	4.618	6.142	<u>573.58</u>
39800	. 2485	. 4985	2.0059	4.640	6.171	573.58
39900	. 2473	. 4973	2.0107	4.662	6.201	573.58
40000	. 2461	. 4961	2.0155	4.685	6.231	573.58
40100	. 2449	. 4949	2.0204	4,707	6.261	<u>573.58</u>
40200	. 2438	. 4938	2.0253	4.730	6.291	573.58
40300	. 2426	. 4926	2.0301	4.753	6.321	573.58
40400	. 2414	. 4914	2.0350	4.776	6.352	<u>573.58</u>
40500	. 2403	. 4902	2.0399	4.799	6.382	573.58
40600	. 2391	. 4890	2.0448	4,822	6.413	573.58
40700	. 2380	, 4879	2.0497	4,845	6.444	<u>573,58</u>
40800	, 2368	. 4867	2.0547	4.868	6.475	573.58
40900	. 2357	. 4855	2.0596	4.892	6.506	573.58
41000	. 2346	. 4844	2.0646	4.915	6.537	573,58
41100	. 2334	. 4832	2.0695	4,939	6,569	<u>573,58</u>
41200	. 2323	. 4820	2,0745	4,963	6.600	573.58
41300	. 2312	. 4809	2.0795	4.987	6.632	573,58
41400	. 2301	. <u>479</u> 7	2.0845	5.011	6.664	<u>573.58</u>
41500	. 2290	. 4786	2.0895	5.035	6.696	573.58
41600	. 2279	. 4774	2.0946	5.059	6.729	573 <b>.5</b> 8
41700	. 2268	. 4763	2.0996	5,083	6,761	<u>573,58</u>
41800	. 2257	. 4751	2,1046	5,108	6, 794	573,58
41900	. 2246	. 4740	2,1097	5.132	6.826	573.58
42000	. 2236	. 4729	2.1148	5.157	6.859	573.58
42100	, 2225	4717	2,1199	5,182	6,892	573.58
42200	. 2214	. 4706	2.1250	5.207	6.926	573.58
42300	. 2204	. 4695	2,1301	5.232	6.959	573.58
42400	. 2193	. 4683	2,1352	<u>5, 257</u>	6.992	573.58
42500	, 2182	. 4672	2,1494	5.283	7,026	573.58
42600	. 2172	. 4661	2,1455	5.308	7.060	573.58
42700	. 2162	. 4650	2,1507	5.334	7.094	573.58
42800	. 2151	. 4639	2,1558	5.359	7.128	573.58
42900	. 2141	. 4627	2.1610	5.385	7.163	573.58

		1					
H <sub>c</sub>	Pa	8	1/6	Ta	$\sqrt{T_a}$	θ	√ө
(Feet)	<u>(''Hg)</u>	$P_a/P_{aSL}$		<u>(°K)</u>		$T_a/T_{aSL}$	
43000	4.794	.16023	6.241	216.66	14.719	.7519	, 8671
43100	4.771	.15946	6.271	216.66	14.719	.7519	. 8671
43200	4.748	.15870	6.301	216,66	14.719	. 7519	. 8671
43300	4.725	.15794	6.331	216,66	14.719	, 7519	. 8671
<u>43400</u>	4.702	.15718	<u>6.362</u>	216,66	14.719	7519	
43500	4.680	.15642	6.393	216.66	14.719	. 7519	. 8671
43600	4.657	.15567	6.423	216.66	14.719	. 7519	. 8671
<u>43700</u>	4.635	. 15492	6,454	216.66	14.719	7519	
43800	4,613	.15418	6.485	216.66	14.719	.7519	. 8671
43900	4.591	.15345	6.516	216.66	14.719	.7519	.8671
44000	4.569	. 15271	6.548	216.66	14.719	,7519	.8671
44100	4.547	.15198_	6.579	216,66	14.719	, 7519	.8671
44200	4.525	. 15125	6.611	216.66	14.719	. 7519	.8671
44300	4.503	.15053	6.643	216.66	14.719	. 7519	.8671
<u>44400</u>	4.482	<u>.14980</u>	6.675	216.66	14.719	. 7519	. 8671
44500	4.460	.14908	6.707	216,66	14.719	. 7519	. 8671
44600	4.439	.14837	6.739	216,66	14.719	. 7519	.8671
44700	4.418	.14766	6.772	216.66	14.719	. 7519	. 8671
44800	4.397	.14695	6.805	216.66	14.719	. 7519	.8671
44900	4.375	.14624	6,838	216.66	14.719	. 7519	,8671
45000	4,354	.14554	6.871	216.66	14.719	.7519	,8671
<u>45100</u>	4.334	14485	6,903	216,66	14.719	. 7519	. 8671
45200	4.313	. 14415	6.937	216.66	14.719	.7519	,8671
45300	4.292	.14346	6.970	216.66	14.719	.7519	.8671
<u>45400</u>	4,271	.14277	7.004	216.66	14,719	. 7519	,8671
45500	4.251	.14208	7.038	216.66	14.719	. 7519	.8671
45600	4, 231	. 14141	7.071	216.66	14.719	.7519	,8671
45700	4.210	.14073	7.105	216,66	14,719	. 7519	.8671
45800	4.190	,14005	7.140	216.66	14.719	. 7519	. 8671
45900	4.170	.13938	7.174	216.66	14.719	.7519	.8671
46000	4.150	.13871	7.209	216.66	14.719	.7519	. 8671
<u>46100</u>	<u>4,130</u>	,13805	7,243	216,66	14,719	. 7519	.8671
46200	4.110	.13739	7.278	216.66	14,719	. 7519	.8671
46300	4.091	.13672	7,314	216,66	14.719	.7519	.8671
46400	4.071	.13607	7.349	216.66	14.719	. 7519	.8671
46500	4.051	.13542	7.384	216.66	14.719	. 7519	.8671
46600	4.032	.13477	7.420	216.66	14.719	. 7519	.8671
46700	4,013	.13412	7.456	216.66	14.719	. 7519	.8671
46800	3.993	.13348	7.491	216.66	14.719	, 7519	.8671
46900	3.974	.13284	7,527	216.66	14.719	.7519	.8671

	_	<u> </u>		5018	1/810	-
H <sub>c</sub>	σ	√σ	1/ \σ	1910	1/010	a (Kasta)
(Feet)	p/psl					(Knots)
43000	. 2131	. 4616	2.1662	5.411	7.197	573.58
43100	. 2120	. 4605	2.1715	5.437	7.232	573.58
43200	. 2110	. 4594	2.1767	5.463	7.266	573,58
43300	. 2100	. 4583	2.1819	5.490	7.301	573,58
<u>43400</u>	. 2090	. 4572	2.1872	5.516	7.337	<u>573.58</u>
43500	. 2080	. 4561	2.1924	5.543	7.372	573.58
43600	. 2070	.4550	2.1977	5.570	7.408	573.58
<u>43700</u>	. 2060	. 4539	2.2030	5.597	7.444	573,58
43800	. 2050	. 4528	2.2083	5.623	7.479	573.58
43900	. 2040	. 4518	2.2136	5.650	7.515	573.58
<b>4400</b> 0	. 20310	. 4507	2.2189	5.678	7.551	573.58
<u>44100</u>	. 20213	. 4496	2.2242	5.705	7.588	<u>573, 58</u>
44200	. 20117	. 4485	2.2296	5.732	7.624	573.58
44300	. 20020	. 4474	2.2349	5.760	7.661	573.58
<u>44400</u>	. 19924	. 4464	2,2403	<u>5,788</u>	7.698	<u>573, 58</u>
44500	. 19828	. 4453	2.2457	5.816	7.735	573.58
44600	. 19733	. 4442	2.2511	5.844	7.772	573.58
44700	. 19639	. 4432	2.2566	5,872	7.810	573.58
44800	. 19545	. 4421	2.2620	5.900	7.847	573.58
<b>44</b> 900	. 19451	. 4410	2,2674	5.929	7.886	573.58
45000	. 19358	. 4400	2.2729	5.957	7.923	573,58
<u>45100</u>	. 19265	. 4389	2.2783	5.986	7.961	573.58
<b>452</b> 00	. 19173	. 4379	2.2838	6.015	8.000	573.58
<b>4530</b> 0	. 19080	. 4368	2.2893	6.044	8.038	573.58
<u>45400</u>	18989	43.58	2.2948	6.073	8.077	<u>573,58</u>
45500	.18897	. 4347	2.3004	6.102	8.116	573.58
45600	.18807	. 4337	2.3059	6.131	8.155	573.58
45700	. 18717	. 4326	2.3114	6,161	8,194	<u>573.58</u>
45800	. 18627	. 4316	2.3170	6.191	8.234	573,58
45900	.18538	. 4306	2.3226	6.221	8.274	573.58
46000	. 18449	. 4295	2.3282	6.251	8.314	573,58
46100	. 18360	. 4285	2.3338	6.281	8.353	<u>573.58</u>
46200	. 18273	. 4275	2.3394	6.311	8.394	573.58
46300	. 18184	. 4264	2.3450	6.342	8.435	573.58
46400	. 18098	. 4254	2.3507	6.372	8.475	573.58
46500	. 18011	. 4244	2.3563	6.403	8.516	573.58
46600	. 17924	. 4234	2.3620	6.433	8.557	573,58
<u>46700</u>		4224	2,3677	6.465	8,598	<u> </u>
46800	.17753	. 4213	2.3734	6.496	8.639	573.58
<b>469</b> 00	.17668	. 4203	2.3791	6.527	8.681	573,58

н <sub>с</sub>	Pa	8	1/8	Ta	$\sqrt{T_a}$	θ	√Ð
(Feet)	("Hg)	$P_a/P_{aSL}$	.0	(•K)		$T_a/T_{aSL}$	
47000	3.955	. 13221	7.563	216.66	14.719	. 7519	.8671
47100	3.936	.13157	7.600	216.66	14.719	. 7519	.8671
47200	3.917	.13094	7.637	216.66	14.719	. 7519	,8671
47300	3.899	.13031	7.674	216.66	14.719	. 7519	.8671
<u>47400</u> 47500	<u>3.880</u> 3.861	.12969	<u>7.710</u> 7.747	<u>216.66</u> 216.66	<u>14.719</u> 14.719	. 7519	.8671
47600	3.843	.12845	7.785	216.66	14.719	.7519 .7519	.8671 .8671
47700	3.824	.12783	7.822	216.66	14.719	. 7519	.8671
47800	3.806	. 12722	7.860	216.66	14.719	. 7519	. 8671
47900	3.783	. 12660	7.898	216.66	14.719	. 7519	.8671
48000	3.770	. 12600	7.936	216.66	14.719	. 7519	.8671
48100	3.752	.12540	7.974	216.66	14.719	. 7519	.8671
48200 48300	3.734 3.716	.12480	8.012 8.052	216.66 216.66	14,719	. 7519	.8671
48400	3.698	.12360	8,090	216.66	14.719	.7519 .7519	.8671 . <u>8671</u>
48500	3.680	. 12301	8.129	216.66	14.719	. 7519	.8671
48600	3.662	.13242	8.168	216.66	14.719	. 7519	,8671
<u>48700</u>	3.645	.12183	8,208	216.66	14.719	. 7519	.8671
48800	3,627	.12125	8.247	216.66	14.719	. 7519	.8671
48900	3.610	.12067	8.287	216.66	14.719	, 7519	.8671
49000	3,593	.12009	8,327	216.66	14,719	. 7519	.8671
49100	3.576	. 11951	8.367	216.66	14.719	. 7519	.8671
49200	3,558	.11894	8.407	216.66	14,719	. 7519	.8671
49300 49400	3.541 3.524	.11837 .11780	8.448	216,66	14.719	. 7519	.8671
49500	3.507	.11724	<u>8.489</u> 8.529	<u>216.66</u> 216.66	$\frac{14.719}{14.719}$	. <u>7519</u> . 7519	<u>.8671</u> .8671
49600	3,491	. 11668	8,570	216.66	14.719	. 7519	.8671
49700	3,474	. 1 1 6 1 1	8.612	216.66	14,719	. 7519	.8671
49800	3,457	. 11556	8.653	216,66	14.719	, 7519	.8671
<b>49</b> 900	3.441	.11500	8,695	216.66	14.719	. 7519	,8671
50000	3,424	, 11445	8,737	216,66	14.719	. 7519	.8671
50100	3.408	. 11390	8.779	216.66	14.719	, 7519	. 8671
50200	3.391	. 11336	8.821	216.66	14.719	.7519	.8671
50300	3.375	.11281	8,864	216.66	14.719	. 7519	.8671
<u>50400</u>	3.359	.11227	8.907	216.66	14.719	. 7519	<u>.8671</u>
50500 50600	3.343 3.327	.11173	8.950	216.66	14,719	.7519	.8671
50700	3.327 3.311	.11120 .11067	8,992	216.66 216.66	14,719 14,719	. 7519	.8671
50800	3,295	.11014	<u>9.035</u> 9.079	216.66	$\frac{14.719}{14.719}$	<u>.7519</u> .7519	<u>.8671</u> .8671
50900	3.279	.10961	9.123 ·	216.66	14.719	.7519	.8671

ĩ

	σ	$\sqrt{\sigma}$	$1/\sqrt{\sigma}$	5018	1 /8 (0	a
(Feet)	p/psl	•				(Knots)
47000	.17584	. 4193	2.3848	6.558	8.722	573.58
47100	. 17499	. 4183	2.3905	6.590	8.765	573.58
47200	. 17415	. 4173	2.3963	6.622	8.807	573.58
47300	. 17331	. 4163	2.4021	6,654	8.850	573.58
47400	. 17249	. 4153	2.4078	6.685	8.892	573.58
47500	, 17166	. 4143	2,4136	6.718	8.935	573.58
47600	.17084	. 4133	2.4194	6.750	8.978	573,58
47700	.17002	. 4123	2.4252	6.783	9.021	573, 58
47800	. 16920	. 4113	2.4311	6.815	9.065	573.58
47900	. 16839	. 4103	2.4370	6.849	9.109	573.58
48000	.16758	. 4094	2.4428	6.881	9.152	573.58
48100	. 16678	. 4084	2,4487	<u>6.914</u>	9.196	<u>573.58</u>
48200	, 16598	. 4074	2.4546	6.947	9.240	573,58
48300	.16518	. 4064	2.4605	6.982	9.286	573.58
48400	. 16439	<u>4054</u>	2,4664	7.015	9.330	<u> </u>
48500	. 16360	. 4045	2.4723	7.049	9.375	573,58
48600	.16282	. 4035	2,4783	7.083	9.420	573.58
48700	. 16204	. 4025	2.4842	7,117	9.466	<u>573.58</u>
48800	. 16126	. 4016	2.4902	7.151	9.511	573.58
48900	.16049	. 4006	2.4962	7.185	9.557	573.58
49000	. 15972	. 3996	2.5022	7.220	9.603	573.58
49100	.15895	. 3987	2.5082	7.255	9.649	<u>573.58</u>
49200	. 15819	. 3977	2.5143	7.290	9.696	573,58
49300	. 15743	. 3968	2.5203	7.325	9.742	573.58
49400	. 15668	. 3958	2.5264	7.360	9.789	<u> </u>
49500	. 15593	. 3949	2,5325	7.395	9.836	573, 58
49600	.15518	. 3939	2,5385	7.431	9.883	573.58
49700	.15443	. 3930	2.5447	7.467	9.932	<u>573.58</u>
49800	. 15369	. 3920	2.5508	7.503	9.979	573.58
49900	.15295	. 3911	2.5570	7.540	10.028	573,58
50000	. 15222	. 3902	2.5631	7.576	10.076	573.58
50100	. 15149	<u>. 3892</u>	2.5692	7.612	10.125	573.58
50200	.15077	. 3883	2.5754	7.649	10,173	573.58
50300	,15004	. 3874	2.5816	7.686	10.222	573.58
50400	.14932	. 3864	2.5879	7.723	10.272	<u> </u>
50500	. 14861	. 3855	2.5941	7.760	10.321	573.58
50600	.14789	, 3846	2.6003	7.797	10.371	573,58
50700	.14719	. 3837	2.6065	7,835	10.420	573.58
50800	. 14648	. 3827	2.6128	7.872	10.470	573.58
<b>50</b> 900	.14578	. 3818	2.6191	7.910	10.521	573.58

_	_	8			<u> </u>		
H <sub>c</sub>	Pa		1/8	Ta	√ T <sub>a</sub>	θ 	٩V
(Feet)	<u>("Hg)</u>	$P_a/P_{aSL}$		<u>(°Ř)</u>	·····	$T_a/T_{aSL}$	
51000	3.263	.10908	9.167	216.66	14.719	.7519	.8671
51100	3,248	.10856	9.211	216.66	14,719	. 7519	. 8671
51200	3.232	.10804	9.255	216.66	14.719	. 7519	.8671
51300	3.217	.10752	9.300	216,66	14.719	. 7519	. 8671
<u>51400</u>	3.201	.10700	9.345	216.66	14,719	. 7519	, 8671
51500	3.186	.10649	9.390	216.66	14.719	. 7519	. 8671
51600	3.171	,10598	9.435	216.66	14.719	. 7519	.8671
<u>51700</u>	3.155	.10547	9.481	216.66	14.719	. 7519	.8671
51800	3.140	.10497	9.526	216.66	14.719	. 7519	.8671
51900	3.125	.10446	9.573	216.66	14.719	. 7519	. 8671
52000	3.110	.10396	9.619	216.66	14.719	. 7519	. 8671
<u>52100</u>	3.095	.10347	<u>9.664</u>	216.66	<u>14.719</u>	. 7519	.8671
52200	3.080	.10297	9.711	216.66	14.719	. 7519	.8671
52300	3.066	.10247	9.759	216.66	14.719	. 7519	. 8671
<u>52400</u>	3.051	. 10198	9.805	216,66	14.719	. 7519	
52500	3.036	.10149	9.853	216.66	14.719	.7519	.8671
52600	3.022	.10101	9.900	216.66	14.719	. 7519	.8671
<u>52700</u>	3.007	. 10052	9.948	216,66	14,719	. 7519	.8671
52800	2,993	.10004	9.996	216.66	14.719	. 7519	. 8671
52900	2.979	.09956	10.044	216.66	14.719	. 7519	. 8671
53000	2.964	.09908	10.092	216.66	14.719	. 7519	. 8671
53100	2.950	.09861	10.141	216.66	14.719	. 7519	. 8671
53200	2,936	.09813	10.190	216.66	14.719	. 7519	. 8671
53300	2.922	.09766	10.239	216.66	14.719	.7519	.8671
53400	2.908	.09719	10.289	216.66	14.719	. 7519	.8671
53500	2.894	.09673	10.338	216.66	14.719	. 7519	.8671
53600	2.880	.09627	10.387	216.66	14.719	. 7519	.8671
53700	2.866	. 09581	10.437	216.66	14.719	. 7519	.8671
53800	2,852	.09535	10.487	216.66	14.719	. 7519	.8671
53900	2.839	.09489	10.538	216.66	14.719	. 7519	.8671
54000	2.825	.09443	10.589	216.66	14.719	. 7519	.8671
<u>54100</u>	<u>    2,812  </u>	09398_	10,640	216.66	<u>14.719</u>		
54200	2.798	.09353	10.691	216.66	14.719	. 7519	.8671
54300	2,785	.09308	10.743	216.66	14.719	.7519	.8671
54400	2.771	.09264	10.794	216.66	14.719	. 7519	.8671
54500	2.758	.09219	10.847	216.66	14, 719	. 7519	. 8671
54600	2.745	.09175	10.899	216.66	14.719	,7519	.8671
54700	2,732	.09131	10.951	216.66	14.719	. 7519	. 8671
54800	2.719	. 09087	11.004	216,66	14.719	. 7519	. 8671
54900	2.706	.09044	11.057	216.66	14.719	. 7519	. 8671

Hc	σ	νσ	1/ 50	50/8	1/610	а
(Feet)	p/psl					(Knots)
51000	. 14508	. 3809	2.6254	7.949	10.572	573.58
51100	.14439	. 3800	2.6317	7.987	10.623	<u>573, 58</u>
51200	.14369	. 3791	2.6381	8.025	10.674	573.58
51300	.14300	. 3782	2.6444	8.064	10.725	573,58
51400	. 14231	. 3772	2.6508	8.103	10.773	<u>573.58</u>
51500	.14164	. 3763	2,6571	8.142	10.829	573.58
51600	.14096	.3754	2.6635	8.181	10.831	573.58
51700	.14023	. 3745	2.6700	8.221	10.934	<u> </u>
51800	.13961	. 3735	2,6764	8.260	10.986	573.58
51900	.13894	. 3727	2.6828	8.300	11,040	573.58
52000	.13827	.3719	2.6893	8.340	11.093	573.58
52100	. 13761	. 3710	2.6957	8.380	11.145	<u>573.58</u>
52200	,13695	. 3701	2.7022	8.420	11.199	573.58
52300	. 13629	. 3692	2.7088	8.462	11,254	573.58
52400	.13564	. 3683	2.7152	8.502	11.308	573.58
52500	.13499	. 3674	2.7218	8.543	11.363	573.58
52600	.13434	. 3665	2.7283	8.584	11,417	573.58
52700	,13370	. 3656	2.7349	8.626	11.472	573.58
52800	.13306	. 3648	2.7415	8.667	11.527	573.58
52900	.13242	. 3639	2.7481	8.709	11,583	573.58
53000	.13178	. 3630	2.7547	8.751	11.639	573.58
53100	.13115	. 3621	2.7613	8.793	11.695	573.58
53200	, 13052	. 3613	2.7680	8.836	11.752	573.58
53300	.12989	. 3604	2.7746	8.878	11.808	573.58
53400	. 12927	. 3595	2.7813	8.921	11.865	573.58
53500	. 12865	. 3587	2.7880	8.964	11.922	573.58
53600	.12803	. 3578	2.7947	9.007	11,979	573.58
53700	.12743	. 3570	2,8014	9,050	12.036	573.58
53800	.12681	. 3561	2.8082	9.093	12.094	573.58
53900	. 12620	. 3552	2.8149	9.137	12,153	573.58
54000	.12560	. 3544	2.8217	9.182	12.212	573.58
54100	. 12500	. 3536	2.8285	9.226	12.271	<u> </u>
54200	. 12440	. 3527	2,8353	9.270	12.330	573.58
54300	. 12380	.3518	2.8421	9.315	12.389	573.58
54400	. 12321	. 3510	2.8489	9.359	12.448	573.58
54500	, 12261	. 3502	2.8558	9.405	12.509	573.58
54600	.12203	. 3493	2.8627	9.450	12.569	573.58
<u>54700</u>	.12144	. 3485	2.8695	9.496	12.630	573.58
54800	. 12086	.3477	2.8765	9.542	12.691	573.58
54900	. 12028	. 3468	2.8834	9.587	12,751	573.58



Hc	Pa	8	1/8	Ta	$\sqrt{T_a}$	θ	Vθ
(Feet)	(''Hg)	$P_a/P_{aSL}$	- 10	( • K)	1-a	T <sub>a</sub> /T <sub>aSL</sub>	
55000	2.693	.09000	11.111	216.66	14,719	. 7519	. 8671
55100	2.680	.08957	11.164	216.66	14.719	. 7519	.8671
55200	2.667	. 03914	11.218	216.66	14.719	. 7519	.8671
55300	2.654	.08871	11.272	216.66	14.719	. 7519	. 8671
55400	2.641	.08829	11.326	216.66	14.719	. 7519	.8671
55500	2.629	.08787	11.380	216.66	14.719	. 7519	. 8671
55600	2.616	.08745	11.435	216.66	14.719	.7519	. 8671
55700	2.603	. 08702	11.491	216.66	<u>14.719</u>	. 7519	. 8671
55800	2.591	. 08661	11.546	216.66	14.719	. 7519	. 8671
55900	2.578	.08619	11.602	216.66	14.719	. 7519	. 8671
56000	2.566	,08578	11.657	216.66	14.719	. 7519	. 8671
56100	2.554	.08537	11.713	216.66	14.719	. 7519	. 8671
56200	2.542	. 08496	11.770	216,66	14.719	. 7519	. 8671
56300	2.529	, 08455	11.827	216.66	14.719	. 7519	. 8671
<u>56400</u>	2.517	. 08415	11.883	216.60	14.719	. 7519	. 8671
56500	2.505	,08374	11.941	216.66	14.719	. 7519	. 8671
56600	2.493	. 08334	11.999	216.66	14.719	. 7519	. 8671
56700	2.481	. 08294	12.056	216.66	14.719	. 7519	.8671
56800	2.469	. 08254	12.115	216.66	14.719	. 7519	. 8671
56900	2.458	.08215	12.172	216.66	14.719	. 7519	. 8671
57000	2.446	. 08175	12.232	216.66	14.719	. 7519	. 8671
57100	2.434	. 08136	12.291	216.66	14.719	. 7519	. 8671
57200	2.422	. 08097	12.350	216.66	14.719	. 7519	.8671
57300	2.411	.08058	12.410	216.66	14.719	. 7519	. 8671
57400	2.399	. 08020	12.468	216.66	14.719	. 7519	. 8671
57500	2.388	. 07981	12.529	216.66	14.719	. 7519	.8671
57600	2.376	. 07943	12.589	216.66	14.719	. 7519	.8671
<u>57700</u>	2.365	. 07905	12.650	216,66	14.719	7519	. 3671
57800	2.353	.07867 .07829	12.711 12.773	216,66 216,66	14.719 14.719	. 7519 . 7519	. 8671
57900	2.342						
58000	2.331	.07792	12.833	216.66	14,719	, 7519	, 8671
<u>58100 _</u>	2,320	, 07754	12.896	216.66	14.719	. 7519	. 8671
58200	2.309	. 07717	12.958	216,66	14.719	. 7519	.8671
58300	2.298	.07680	13.020	216.66	14.719	, 7519	.8671
58400	2.287	.07643	13.083	216.66	14.719	. 7519	. 8671
58500	2.276	.07607	13.145	216.66	14.719	. 7519	.8671
58600	2.265	.07570	13.210	216.66	14.719	. 7519	.8671
58700	2.254	. 07534	13.273	216.66	14.719	. 7519	. 8671
50800	2.243	. 07498	13.336	216.66	14.719	. 7519	. 8671
58900	2.232	. 07462	13.401	216.66	14.719	. 7519	. 8671

				- (	.(	
H <sub>c</sub>	σ	<b>√σ</b>	l/ {σ	1018	1/2/6	a
<u>(Feet)</u>	P/PSL	<del>_</del>				<u>(Knots</u> )
55000	. 11971	. 3460	2,8903	9.634	12.813	573,58
55100	.11913	. 3452	2.8973	9.680	12.875	<u>573, 5</u> 8
55200	. 11856	. 3443	2.9042	9.727	12,937	573.58
55300	. 11799	. 3435	2.9112	9.774	13,000	573.58
55400	. 11743	. 3427	2.9182	9,821	13.062	<u>573.58</u>
55500	. 11686	. 3419	2.9252	9.867	13,124	573.58
55600	. 11630	. 3410	2.9323	9.915	13.187	573.58
55700	. 11574	. 3402	2.9394	<u>9,964</u>	13, 252	<u>573,58</u>
55800	. 11519	. 3394	2.9464	10.011	13,315	573.58
55900	. 11463	, 3386	2.9535	10.060	13,380	573.58
56000	.11409	. 3378	2.9606	10.108	13.444	573.58
56100	11354	. 3370	2.9678	10.156	13,508	<u>573.58</u>
56200	.11299	. 3361	2,9749	10.205	13,573	573.58
56300	. 11245	, 3353	2.9820	10.255	13,639	573.58
5 <u>6400</u>	. 11192	. 3345	2.9892	10.304	13,704	<u>573, 58</u>
56500	.11138	. 3337	2.9964	10.354	13.771	573.58
56600	.11085	. 3329	3.0036	10.404	13.837	573.58
56700	. 11031	. 3321	3.0108	10.454	13.904	573.58
56800	. 10978	, 3313	3.0181	10.505	13.972	573.58
56900	. 10926	. 3305	3.0253	10.555	14.038	573.58
57000	. 10873	. 3297	3.0327	10.606	14.107	573,58
57100	. 10821	. 3290	3.0399	10.657	14.174	<u> </u>
57200	. 10769	. 3282	3.0472	10.708	14.242	573.58
57300	.10719	. 3274	3.0546	10.760	14.311	573,5 <b>8</b>
57400	. 10666	. 3266	3.0619	10.811	14.379	<u> </u>
57500	. 10615	. 3258	3.0693	10.864	14.450	573.58
57600	,10564	. 3250	3,0767	10.916	14.519	573.58
<u>57700</u>	. 10514	. 3242	3.0841	<b>109</b> 69	<u>14.588</u>	<u>573,58</u>
57800	.10463	. 3235	3.0915	11.022	14.659	573.58
57900	.10413	. 3227	3.0989	11.075	14.730	573.58
58000	.10363	. 3219	3.1064	11,128	14.800	573.58
58100	. 10313	. 3211	3.1139	11.182	14.872	573.58
58200	. 10264	. 3204	3.1214	11.236	14.944	573,58
58300	. 10215	. 3196	3.1289	11.290	15,016	573.58
58400	. 10166	. 3188	3.1. 54	11.345	15.088	573.58
58500	. 10117	. 3181	3,1,40	11.398	15.160	573.58
58600	. 10068	. 3173	3.1515	11.454	15.234	573, 58
<u>58700</u>	. 10020	. 3165	3.1591	11.509	15.307	<u>573,58</u>
58800	. 09972	. 3158	3.1667	11.564	15.380	573.58
58900	.09924	. 3150	3.1743	11.620	15.454	573.58



H <sub>c</sub>	$\mathbf{P}_{\mathbf{a}}$	8	1/8	Ta	√T <sub>e</sub>	θ	<b>√</b> €
(Feet)	(''Hg)	Pa/PaSL	- 10	( ° K)	1 - 2	$T_a/T_{aSL}$	
		. 07426	12 466		14 710		0671
59000	2.221	.07428	13,466 13,531	216.66 216.66	14.719 14,719	,7519 ,7519	.8671
<u>59100</u> 59200	<u>2,211</u> 2,200	.07355	13.596	216.66	$\frac{14,719}{14,719}$	. 7519	<u>.8671</u> .8671
59200	2.200	.07320	13.596	216.66	14.719	. 7519	.8671
59300 59400	2.190	. 07320	13,726	216.66	14.719	. 7519	.8671
59500	2.169	.07250	13,793	216.66	$\frac{14.719}{14.719}$	. 7519	.8671
59600	2.159	.07215	13,860	216.66	14.719	. 7519	.8671
59700	2.138	.07215	13,800	216 66	14.719	,7519	.8671
59000	2,133	. 07146	13.993	216.66	14.719	. 7519	.8671
59900	2.127	.07140	14,060	216,66	14,719	. 7519	.8671
37900	6.161	.07112	14.000	210,00	14, 117	. ( ) 1 7	
60000	2.117	.07077	14,130	216.66	14.719	.7519	.8671
60100	2.137	.07044	14.196	216.66	14.719	. 7519	. 8671
60200	2.097	.07010	14.265	216.66	14.719	. 7519	.8671
60300	2.087	. 06976	14.334	216.66	14.719	. 7519	.8671
60400	2.077	. 06943	14.403	216.66	14.719	. 7519	.8671
60500	2,067	.06910	14,471	216.66	14.719	, 7519	.8671
60600	2,057	.06376	14.543	216.66	14.719	. 7519	. 8671
60700	2.047	. 06843	14.613	216.66	<u>14, 719</u>	. 7519	<u>, 8671</u>
60800	2.037	. 06310	14,684	216.66	14,719	. 7519	. 8671
60900	2.0281	. 06778	14,753	216.66	14.719	. 7519	. 8671
61000	2.0183	. 06745	14.825	216.66	14.719	,7519	. 8671
61100	2.0087	. 06713	<u>14.896</u>	216.66	14.719	. 7519	. 8671
61200	1.9990	.06681	14,967	216,66	14.719	, 7519	.8671
61300	1,9894		15,039	216.66	14.719	.7519	. 8671
<u>61400</u>	<u>1.9799</u>	06617	15,112	216.66	<u>14,719</u>	. 7519	. 8671
61500	1.9704	. 06585	15.186	216.66	14.719	,7519	. 8671
61600	1.9610	. 06554	15.257	216,66	14.719	.7519	.8671
<u>61700</u>	1.9515	. 06522	15.332	216.66	<u>14.719</u>	. 7519	. 8671
61800	1.9422		15.405	216.66	14.719	. 7519	.8671
61900	1,9329	.06460	15.479	216.66	14.719	. 7519	. 8671
62000	1.9236	. 06429	15,554	216.66	14.719	. 7519	.8671
62100	1.9144		<u>15.629</u>	216.66	<u>14.719</u>	. 7519	.8671
62200	1.9052		15,706	216.66	14.719	.7519	. 8671
62300	1.8961	.06337	15.780	216.66	14.719	. 7519	.0671
62400	1.8870	and the second se	15.857	216.66	<u>14.719</u>	. 7519	. 8671
62500	1.8779		15.933	216.66	14.719	.7519	.8671
62600	1.8689	. 06246	<b>16.01</b> 0	216.66	14.719	.7519	. 8671
62700	1.8600	. 06216	16.087	216,66	14.719	. 7519	.8671
62300	1.8510	. 06186	16.165	216.66	14.719	. 7519	. 8671
62900	1.8421	.06157	16.241	216.66	14,719	.7519	. 8671

				- 1	,	
Hc	σ	νσ	1/ √ <del>σ</del>	<del>ر</del> ه، م	1 /6 🕞	a
<u>(Feet)</u>	<u>p/psl</u>					(Knots)
59000	.09877	. 3143	3,1820	11.676	15.529	573.58
59100	. 09829	. 3135	3,1896	11.733	15,605	573,58
59200	. 09782	. 3128	3.1973	11.789	15.679	573.58
59300	. 09735	. 3120	3.2050	11.845	15.754	573.58
<u>59400</u>	<u>. 09689</u>	. 3113	3.2127	11,902	15.830	<u>573.58</u>
59500	. 09642	. 3105	3.2204	11.960	15.906	573.58
59600	. 09596	. 3098	3.2281	12.018	15.984	573,58
<u>59700</u>	. 09550	. 3090	3.2359	12.076	16.061	<u>573.58</u>
59800	.09504	. 3083	3.2438	12.134	16.138	573.58
59900	. 09459	. 3075	3.2515	12.192	16.215	573.58
60000	.09413	. 3068	3.2594	12.252	16.295	573.58
<u>60100</u>	. 09368	. 3061	3.2672	12.309	16.372	573 <u>.58</u>
60200	. 09323	. 3053	3.2750	12.369	16.451	573.58
60300	. 09279	. 3046	3.2829	12.429	16.531	573.58
60400	.09234	. 3039	3.2908	12.488	16.610	<u>573.58</u>
60500	. 09190	. 3031	3.2987	12.548	16.689	573.58
60600	. 09146	. 3024	3.3067	12.610	16.772	573.58
60700	. 09102	. 3017	3.3147	12.671	16,853	<u> </u>
60800	. 09058	. 3010	3.3226	12,732	16,934	573.58
60900	.09015	. 3002	3.3306	12.792	17.014	573.58
61000	.08972	. 2995	3.3386	12.855	17.097	573.58
<u>61100</u>	. 08929	. 2988	3.3466	12.916	<u>17.179</u>	<u>573.58</u>
61200	. 08386	. 2981	3.3547	12.978	17,261	573.58
61300	. 08543	. 2974	3.3528	13,041	17.344	573.58
<u>61400</u>	. 08801	. 2967	3.3709	13.104	<u>17.428</u>	<u> </u>
61500	. 08758	. 2959	3.3790	13.167	17.513	573.58
61600	.08717	. 2952	3.3871	13.230	17.596	573.58
<u>61700</u>	. 08675	. 2945	3.3953	13.295	17.682	<u>573,58</u>
61800	. 08633	. 2938	3.4035	13.358	17.766	573.58
61900	.08592	. 2931	3.4116	13.422	17.852	573.58
62000	.08550	. 2924	3.4199	13.487	17.938	573.58
<u>62100</u>	.08510	. 2917	3.4280	<u>13,552</u>	18.025	<u> </u>
62200	. 08469	. 2910	3.4363	13.618	18,113	573.58
62300	.08428	. 2903	3.4445	13.683	18.198	573.58
62400	. 08388	. 2896	3.4529	13.750	18.288	<u>573.58</u>
62500	.08347	. 2889	3.4612	13.816	18.375	573.58
62600	.08307	. 2882	3.4695	13.882	18,463	573,58
62700	. 08268	. 2875	3.4778	13.949	18.552	<u> </u>
62800	. 08228	. 2868	3.4862	14.017	18.642	573.58
62900	.08188	. 2862	3.4946	14.083	18,730	573.58

Hc	Pa	8	1/8	Ta	Ta	θ	<b>√</b> <del>0</del>
(Feet)	(''Ĥg)	Pa/PaSL		( • Ř)	( - <b>G</b>	$T_a/T_{aSL}$	• -
63000	1.8333	. 06127	16.321	216.66	14.719	. 7519	. 8671
63100	1.8246	.06098	16.398	216.66	14.719	. 7519	.8671
63200	1.8158	.06069	16.477	216.66	14.719	. 7519	.8671
63300	1.8071	.06039	16.559	216.66	14.719	. 7519	. 8671
63400	1.7984	,06010	16.638	216.66	14,719	. 7519	<u>. 8671</u>
63500	1.7898	.05982	16.716	216.66	14.719	. 7519	.8671
63600	1.7812	.05953	16.798	216.66	14.719	. 7519	.8671
<u>63700</u>	1.7727	.05924	16.880	216.66	14.719	. 7519	.8671
63800	1.7642	.05896	16.960	216.66	14.719	. 7519	. 8671
63900	1.7558	,05868	17.041	216.66	14.719	. 7519	.8671
64000	1.7473	.05840	17.123	216.66	14.719	. 7519	. 8671
64100	1.7389	.05812	17.205	216,66	14.719	. 7519	<u>, 8671</u>
64200	1.7306	.05784	17.289	216.66	14.719	. 7519	.8671
64300	1.7223	.05756	17.373	216.66	14.719	. 7519	.8671
64400	1.7140	.05728	17.458	216.66	14.719	. 7519	.8671
64500	1.7058	.05701	17.540	216.66	14.719	. 7519	.8671
64600	1.6977	.05674	17.624	216.66	14.719 14.719	. 7519	.8671
<u>64700</u> 64800	<u>1.6895</u> 1.6814	.05647	<u>17.708</u> 17.793	<u>216.66</u> 216.66	$\frac{14.719}{14.719}$	<u>. 7519</u> . 7519	<u>. 8671</u> .8671
64900	1.6733	.05592	17.882	216.66	14.719	. 7519	.8671
·					· · ·		
65000	1.6653	.05566	17.966	216.66	14.719	. 7519	.8671
65100	1.6573	. 05539	18.053	216.66	14.719	. 7519	.8671
65200	1.6494	.05513	18.138	216.66	14.719	. 7519	. 867).
65300	1.6415	.05486	18,228	216.66	14.719	. 7519	.8671
<u>65400</u> 65500	$\frac{1.6336}{1.6258}$	.05460	<u>18,315</u> 18,406	<u>     216.66    </u> 216.66	<u>14.719</u> 14.719	<u>. 7519</u> . 7519	<u>.8671</u> .8671
65600	1.6179	.05407	18.494	216.66	14.719	. 7519	.8671
65700	1,6102	05382	18,580	216,66	_14,719	. 7519	.8671
65800	1.6025	. 05356	18.670	216.66	14.719	. 7519	.8671
65900	1.5948	.05330	18.761	216,66	14.719	. 7519	. 8671
66000	1.5872	.05305	18.850	216.66			
66100	1.5795	.05279	18,850	216.66	14.719 14.719	. 7519 . 7519	.8671 .8671
66200	1.5720	.05254	19.033	216.66	14.719	. 7519	.8671
66300	1.5644	.05229	19.124	216.66	14.719	. 7519	.8671
66400	1.5569	.05203	19.219	216.66	14.719	. 7519	.8671
66500	1.5495	.05179	19.308	216.66	14.719	. 7519	.8671
66600	1.5421	.05154	19.402	216.66	14.719	. 7519	.8671
66700	1.5347	.05129	19.497	216.66	14.719	. 7519	. 8671
66800	1.5273	. 05104	19.592	216,66	14.719	. 7519	.8671
66900	1.5200	.05080	19.685	216.66	14.719	. 7519	.8671

				- (	A	
H <sub>c</sub>	σ	<u>رم</u>	1/10	1018	1/810	a
(Feet)	p/psl					(Knots)
63000	.08149	. 2855	3.5030	14.152	18.822	573.58
63100	.08110	. 2848	3.5114	14.219	18.911	573.58
63200	.08071	. 2841	3.5199	14, 287	19,002	573.58
63300	.08032	. 2834	3.5284	14.358	19.096	573.58
63400	,07994	. 2827	3.5369	14.427	19.188	573,58
63500	.07956	. 2821	3,5453	14.495	19.278	573.58
63600	.07918	. 2814	3.5539	14,565	19.372	573.58
63700	.07880	, 2807	3.5624	14,637	19.467	<u> </u>
63800	.07842	, 2800	3.5710	14.706	19.559	573.58
63900	.07804	. 2794	3.5796	14.776	19.653	573.58
64000	,07767	. 2787	3.5882	14,847	19.747	573.58
<u>64100</u>	.07730	. 2780	<u>3.5968</u>	14,919	19.842	<u>573.58</u>
64200	.07693	, 2774	3.6055	14,991	19.938	573.58
64300	.07656	. 2767	3.6141	15.064	20.035	573.58
<u>64400</u>	.07619	. 2760	3.6229	15.137	20.133	<u>573, 58</u>
64500	.07582	, 2754	3.6316	15.209	20.228	573.58
64600	.07546	, 2747	3.6403	15.282	20,325	573.58
<u>64700</u>	.07510	. 2740	<u>3.6490</u>	<u>15.355</u>	20.422	<u> </u>
64800	,07474	. 2734	3.6578	15.428	20,520	573.58
64900	.07438	. 2727	3.6667	15.506	20.623	573. <b>5</b> 8
65000	.07402	. 2721	3.6755	15.578	20.719	573.58
<u>65100</u>	,07367	. 2714	3.6843	<u>15.654</u>	20,820	<u> </u>
65200	.07332	. 2708	3.6932	15.728	20.918	573.58
65300	.07296	. 2701	3.7021	15. <b>8</b> 05	21.02	573.58
<u>65400</u>	.07261	2695	3.7110	15.880	21.12	<u>573.58</u>
65500	.07227		3.7199	15.959	21.22	573.58
65600	.07192	. 2682	3.7289	16.036	21.32	573.58
<u>65700</u>	,07158	. 2675	3.7378	<u>16.111</u>	21.42	<u> </u>
65800	.07123	. 2669	3.7468	16.189	21,53	573.58
65900	. 07089	. 2663	3,7558	16.268	21.63	573.58
66000	.07055	. 2656	3.7648	16.344	21.73	573.58
66100	.07021	. 2650	3.7740	<u>16, 425</u>	21.84	<u>573.58</u>
66200	.06987	. 2643	3.7830	16.503	21.94	573.58
66300	.06954	. 2637	3.7921	16.582	22.05	573.58
<u>66400</u>	. 06921	. 2631	3.8013	16.665	22.16	<u>573.58</u>
66500	,06888	, 2624	3,8104	16.742	22.26	573.58
66600	. 06855	. 2618	3.8195	16.823	22.37	573.58
66700	. 06822	. 2612	3.8287	16.905	22.48	<u>573.58</u>
66800	.06789	. 2606	3,8380	16.988	22.59	573,58
66900	.06756	. 2599	3.8472	17.068	22.70	573,58

아이는 아이는 것이 이 아이는 것이 아이는

TABL: 9.3

H <sub>c</sub>	Pa	8	1/5	Ta	Ta	θ	TO
(Feet)	(''Hg)	Pa/PaSL	0	( • ĸ) _	a	$T_a/T_{aSL}$	
67000	1.5127	. 05056	19.778	216.66	14.719	. 7519	. 8671
67100	1.5054	.05031	19.876	216,66	14.719	. 7519	.8671
67200	1.4982	.05007	19.972	216.66	14.719	. 7519	.8671
67300	1.4911	.04983	20.068	216.66	14.719	. 7519	.8671
67400	1.4839	.04959	20.165	216.66	14,719	. 7519	.8671
67500	1.4768	.04936	20.259	216.66	14.719	. 7519	. 8671
67600	1.4697	.04912	20.358	216.66	14.719	. 7519	.8671
67700	1.4626	.04888	20.458	216.66	14.719	. 7519	.8671
67800	1.4556	.04865	20.555	216.66	14.719	. 7519	.8671
67900	1.4487	.04842	20.65	216.66	14.719	. 7519	.8671
68000	1.4417	.04818	20.75	216.66	14.719	. 7519	. 8671
<u>68100</u>	1.4348	.04795	20.85	216.66	<u>14.719</u>	. 7519	.8671
68200	1.4279	. 04772	20.95	216.66	14.719	7519	. 8671
68300	1.4210	. 04749	21.05	216.66	14.719	. 7519	. 8671
<u>68400</u>	1.4143	.04727	21.15	216.66	14.719	. 7519	. 8671
68500	1.4075	. 04704	21.25	216.66	14.719	, 7519	. 8671
68600	1.4007	. 04681	21.36	216.66	14.719	. 7519	. 8671
68700	1.3940	. 04659	21.46	216.66	14.719	. 7519	. 8671
68800	1.3873	.04637	21.56	216.66	14.719	. 7519	. 8671
68900	1.3807	.04614	21.67	21 <u>,</u> 6.66	14.719	. 7519	.8
69000	1.3741	. 04592	21.77	216.66	14.719	. 7519	. 8i
<u>69100</u>	1.3675	.04570	21.88	216.66	14.719	. 7519	. 8671
69200	1.3609	.04548	21.98	216.66	14.719	. 7519	. 8671
69 <b>30</b> 0	1.3544	.04527	22.08	216.66	14.719	. 7519	. 8671
<u>69400</u>	1.3479	. 04505	22.19	216.66	14.719	. 7519	. 8671
69500	1.3414	, 04483	22,30	216.66	14.719	. 7519	. 8671
69600	1.3350	. 04462	22.41	216.66	14.719	. 7519	. 8671
69700	1.3286	.04440	22.52	216.66	14.719	. 7519	. 8671
69800	1.3222	.04419	22.62	216.66	14.719	. 7519	.8671
69900	1.3159	.04398	22.73	216.66	14.719	. 7519	. 8671
70000	1.3096	,04377	22.84	216.66	14.719	. 7519	.8671
<u>70100</u>	1.3033	.04356	<u>22.95</u>	216.66	<u>14.719</u>	<u> </u>	<u>. 8671</u>
70200	1.2971	.04335	23.06	216.66	14.719	. 7519	. 8671
70300	1.2908	,04314	23.18	216.66	14.719	. 7519	. 8671
7_4^^_	1.2846	,04293	23.29	216.66	14,719	. 7519	. 8671
70560	1.2785	.04273	23.40	216.66	14.719	. 7519	. 8671
ንሶ 'ኃ00	1.2723	.04252	23.51	216.66	14.719	. 7519	. 8671
70700	1.2662	.04232	23.62	216.66	14.719	. 7519	. 8671
70800	1.2602	. 04212	23.74	216.66	14.719	. 7519	. 8671
70900	1.2541	. 04191	23.86	216.66	14.719	. 7519	. 8671

Hc	σ ρ/psl	σ	1/ 10	1018	1/5/0	a
(Feet)			<del></del>			(Knots)
67000	.06724	. 2593	3.8564	17.149	22.80	573.58
67100	. 06692	. 2587	3.8657	17.235	22.92	573.58
67200	. 06660	. 2581	3.8751	17,317	23.03	573.58
67300	. 06628	. 2574	3.8843	17.401	23.14	573,58
<u>67400</u>	06596	<u>. 256</u> 8	3.8937	17.485	23.25	573.58
67500	. 06564	. 2562	3.9031	17.566	23.36	573.58
67600	. 06533	. 2556	3.9124	17.652	23.47	573.58
67700	. 06501	, 2550	3.9219	17.739	23.59	573.58
67800	. 06470	. 2544	3.9313	17.823	23.70	573.58
67900	. 06439	. 2538	3.9408	17.907	23.81	573.58
68000	. 06408	. 2531	3.9503	17.997	23.93	573.58
<u>68100</u>	, 06378	. 2525	3.9597	18.083	24.05	573.58
68200	. 06347	. 2519	3.9693	18.170	24.16	573.58
68300	. 06317	. 2513	3.9789	18.258	24.28	573.58
<u>68400</u>	. 06286	. 2507	3.9884	18.343	24.39	<u>573.58</u>
68500	. 06256	. 2501	3.9980	18.433	24.51	573.58
68600	.06226	. 2495	4.0076	18.523	24.63	573.58
<u>68700</u>	. 06196	2489	4.0173	18.611	24,75	<u> </u>
68800	. 06167	. 2483	4.0269	18.699	24.87	573.58
68900	. 06137	. 2477	4.0366	18.792	24.99	573.58
<b>6900</b> 0	.06108	. 2471	4.0463	18.882	25.11	573.58
<u>69100</u>	. 06079	. 2465	4.0560	18.973	25.23	573.58
69200	. 06049	. 2460	4.0658	19.065	25.35	573.58
69300	. 06020	. 2454	4.0756	19.154	25.47	573.58
<u>69400</u>	. 05992	. 2448	4.0854	19.247	25.59	<u>573.58</u>
69500	. 05963	. 2442	4,0953	19.342	25.72	573.58
69600	. 05934	. 2436	4.1051	19.433	25.84	573.58
69700	. 05906	. 2430	4.1150	<u>    19.529    </u>	25.97	<u>573.58</u>
69800	. 05877	. 2424	4.1249	19.622	26.09	573.58
69900	. 05849	. 2418	4.1348	19.715	26.22	573.58
<b>7000</b> 0	.05821	. 2413	4.1447	19.810	26.34	573.58
70100	. 05793	. 2407	4.1547	19.905	26.47	573.58
70200	. 05765	. 2401	4.1647	20.002	26.60	73 28
70300	.05738	. 2395	4.1747	20.099	26.73	573.58
<u>70400</u>	. 05710	. 2390	4.1848	20.198	26.86	<u> </u>
70500	. 05683	. 2384	4.1948	20.292	26.98	573.58
70600	. 05656	. 2378	4.2,05	20.392	27.12	573.58
70700	. 05628	. 2372	4,215	20.489	27.25	<u> </u>
70800	. 05602	. 2367	4,225	20.586	27.38	573.58
70900	. 05575	. 2361	4.235	20.689	27.51	573.58



		,					
Н <sub>с</sub>	Pa	8	1/8	Ta	Ta	θ	10
(Feet)	(''Hg)	Pa/PaSL		<u>(•K</u> )		$T_{a}/T_{aSL}$	
71000	1.2481	.04171	23.97	216.66	14.719	. 7519	. 8671
71100	1,2422	.04151	24,09	216.66	14.719	. 7519	.8671
71200	1.2362	. 04131	24.20	216.66	14.719	. 7519	.8671
71300	1.2303	.04112	24.31	216.66	14.719	. 7519	. 8671
71400	1.2243	.04092	24.43	216.66	14,719	. 7519	.8671
71500	1.2185	.04072	24.55	216.66	14.719	. 7519	.8671
71600	1.2127	.04053	24.67	216.66	14.719	.7519	.8671
71700	1.2068	.04033	24.79	216.66	14.719	. 7519	. 8671
71800	1.2010	.04014	24.91	216.66	14.719	. 7519	.8671
71900	1.1953	.03995	25.03	216.66	14.719	. 7519	.8671
72000	1.1896	.03976	25.15	216.66	14.719	.7519	.8671
72100	1,1838	.03957	25,27	216.66	<u>14.719</u>	. 7519	. 8671
72200	1.1782	. 03938	25.39	216.66	14.719	. 7519	. 8671
72300	1.1725	.03919	25.51	216.66	14.719	. 7519	. 8671
72400	1.1669	.03900	25.64	216.66	14,719	. 7519	. 8671
72500	1.1613	,03881	25.76	216.66	14.719	. 7519	.8671
72600	1.1557	.03863	25.88	216.66	14.719	. 7519	. 8671
72700	1.1502	.03844	26.01	216.66	14.719	. 7519	. 8671
72800	1.1447	.03826	26.13	216.66	14.719	. 7519	. 8671
72900	1.1392	.03807	26.26	216.66	14.719	. 7519	. 8671
73000	1.1337	.03789	2639	216.66	14.719	. 7519	. 8671
73100	1,1283	03771	26,51	216,66	14.719	. 7519	<u>. 8671</u>
73200	1.1229	.03753	26.64	216.66	14.719	. 7519	. 8671
73300	1.1175	.03735	26.77	216.66	14.719	. 7519	. 8671
73400	1.1121	.03717	26.90	216.66	14.719	. 7519	<u>. 8671</u>
73500	1.1068	. 03699	27.03	216.66	14.719	. 7519	.8671
73600	1.1015	. 03681	27.16	216.66	14.719	.7519	. 8671
<u>73700</u>	1.0962	03664	<u>    27. 29   </u>	216,66	<u>14.719</u>	7519	. 8671
73800	1.0910	.03646	27.42	216.66	14.719	. 7519	.8671
73900	1.0857	. 03629	27.55	216.66	14.719	. 7519	.8671
74000	1.0805	.03611	27.69	216.66	14.719	. 7519	. 8671
74100	1.0753	. 03594	27.82	216.66	14.719	. 7519	<u>. 8671</u>
74200	1.0702	. 03577	27.95	216.66	14.719	. 7519	. 8671
74300	1.0650	. 03559	28.09	216.66	14.719	. 7519	.8671
74400	1.0600	. 03542	28,23	216,66	14,719	. 7519	. 8671
74500	1.0549	. 03526	28.36	216,66	14.719	. 7519	.8671
74600	1.0498	. 03 <b>509</b>	28.49	216.66	14.719	. 7519	. 8671
74700	1,0448	.03492	28.63	216,66	<u>14, 719</u>	. 7519	. 8671
74800	1.0398	. 03475	28,77	216,66	14.719	. 7519	.8671
74900	1.0348	.03458	28.91	216.66	14.719	. 7519	. 8671

Hc	σ	TT	1/ <del>(</del> <del>-</del>	JO18	1/6/0	a	
(Feet)	P/P SL	ų -		1		(Knots)	
	.05548	2255	A 245	20 700	27 64		
71000 71100	.05521	. 2355 . 2350	4,245	20,788 20,889	27.64 27.78	573.58 573.58	
71200	.05495	.2344	4.266	20.990	27.91	573.58	
71300	.05469	.2338	4.276	21.087	28.04	573.58	
71400	. 05442	.2333	4.286	21.190	28.18	573.58	
71500	. 05416	.2327	4.296	21.294	28.32	573.58	
71600	. 05390	.2322	4.307	21.394	28.45	573.58	
71700	.05354	. 2316	4.317	21.500	28.59	573.58	
71800	. 05339	.2311	4.328	21.601	28.73	573.58	
71900	. 05313	2305	4.338	21.704	28.86	573.58	
72000	.05288	. 2299	4.348	21.808	2 <b>9.</b> 00	573.58	
72100	. 05262	.2294	4.359	21.913	29.14	573.58	
72200	.05237	. 2288	4.369	22.018	29.28	573.58	
72300	. 05212	. 2283	4.380	22.125	29.42	573.58	
72400	. 05187	. 2277	4.390	22.233	29.57	573,58	
72500	.05162	. 2272	4.401	22.342	29.71	573.58	
72600	. 05137	.2267	4.412	22.446	29.85	573.58	
72700	. 05113	. 2261	4,422	22,55	30.00	573.58	
72800	. 05088	.2256	4.433	22.66	30.14	573.58	
72900	.05064	.2250	4.443	22.77	30.29	573.58	
73000	.05039	.2245	4.454	22.88	30.43	573.58	
<u>73100</u>	.05015	.2239	4.465	22.99	30.58	573.58	
73200	. 04991	.2234	4.476	23.10	30.72	573.58	
73300	. 04967	. 2229	4.486	23.21	30.87	573.58	
73400	. 04943	. 2223	<u>    4.497  </u>	23.32	31.02	<u> </u>	
73500	,04920	. 2218	4.508	23.44	31.17	573.58	
73600	. 04896	. 2213	4.519	23.55	31.32	573.58	
<u>73700</u>	.04873	.2207	4.530	23.66	31.47	573.58	
73800	.04849	. 2202	4.541	23.78	31.63	573,58	
73900	.04826	.2197	4.552	23.89	31.77	573.58	
74000	.04803	. 2192	4,562	24.01	31.93	573.58	
74100	. 04780	. 2186	4.573	24.12	32.08	<u> </u>	
74200	.04757	. 2181	4.584	24.24	32.24	573.58	
74300	.04734	.2176	4.596	24.36	32.40	573.58	
74400	. 04712	. 2171	4,607	24.48	32.55	<u> </u>	
74500	. 04689	.2165	4.618	24,59	32.70	573.58	
74600	. 04666	. 2160	4.629	24.71	32.86	573.58	
74700	. 04644	, 2155	4,640	24.83	33.02	573.58	
74800	. 04622	. 2150	4.651	24.95	33,18	573.58	
74900	. 04600	,2145	4.662	25.07	33.35	573.58	
Н <sub>с</sub>	Pa	8	115	Ta	$\sqrt{T_a}$	θ	Vθ
----------------	---------------	---------------	-------	-------------	---------------	---------------	---------------
(Feet)	<u>(''Hg)</u>	$P_a/P_{aSL}$		<u>(•K)</u>		$T_a/T_{aSI}$	
75000	1.0298	.03442	29.05	216.66	14.719	. 7519	. 8671
75100	1.0249	. 03425	29,19	216,66	14,719	, 7519	,8671
75200	1.0199	.03409	29.33	216.66	14.719	. 7519	.8671
75300	1.0151	. 03393	29.47	216.66	14.719	. 7519	.8671
<u>75400</u>	1.0102	.03376	29.62	216.66	<u>14.719</u>	<u>7519</u>	. 8671
75500	1.0054	. 03360	29.76	216.66	14.719	. 7519	.8671
75600	1.0006	.03344	29.90	216.66	14.719	. 7519	.8671
75700	. 9957	.03328	30.04	216,66	<u>14.719</u>	. 7519	. 8671
75800	. 9910	.03312	30.19	216.66	14.719	. 7519	.8671
75900	. 9862	.03296	30.33	216.66	14.719	. 7519	. 8671
76000	. 9815	. 03280	30.48	216.66	14.719	. 7519	. 8671
<u>76100</u>	<u>. 9768</u>	. 03265	30.62	216,66	14.719	. 7519	. 8671
76200	. 5721	. 03249	30.77	216.66	14.719	. 7519	.8671
76300	. 9674	. 03233	30.93	216.66	14.719	. 7519	. 8671
76400	. 9628	. 03218	31.07	216,66	14.719	. 7519	. 8671
76500	. 9582	. 03202	31.23	216.66	14.719	. 7519	.8671
76600	. 9536	.03187	31.37	216.66	14.719	. 7519	.8671
<u>76700</u>		. 03172	31.52	216.66	14.719	. 7519	.8671
76800	.9445	.03156	31.68	216.66	14.719	. 7519	. 8671
76900	. 9400	.03141	31.83	216.66	14.719	. 7519	. 8671
77000	.9354	.03126	31.98	216.66	14.719	. 7519	. 8671
77100	. 9309_	.03111	32.14	216.66	14.719	. 7519	<u>. 8671</u>
77200	. 9265	. 03096	32.29	216.66	14.719	. 7519	.8671
77300	. 9221	.03082	32.44	216.66	14.719	.7519	.8671
<u>77400</u>	. 9176	.03067	32.60	216.66	14.719	. 7519	<u>. 8671</u>
77500	. 9132	. 03052	32.76	216.66	14.719	. 7519	. 8671
77600	. 9088	.03037	32.92	216.66	14.719	. 7519	.8671
77700	. 9045	.03023	33.07	216.66	14.719	. 7519	.8671
77800	. 9002	.03008	33.24	216.66	14.719	. 7519	.8671
77900	. 8958	. 02994	33.40	216.66	14.719	. 7519	.8671
78000	. 8915	. 02980	33.55	216.66	14.719	. 7519	. 8671
78100	. 8873	. 02965	33.72	216.66	14.719	. 7519	. 8671
78200	. 8830	. 02951	33.88	216.66	14.719	. 7519	.8671
78300	. 8788	. 02937	34.04	216.66	14.719	.7519	. 8671
<u>78400</u>	.8745	.02923	34.21	216,66	14.719	. 7519	. 8671
78500	.8704	. 02909	34.37	216.66	14.719	. 7519	.8671
78600	. 8662	. 02895	34.54	216.66	14.719	. 7519	.8671
78700	. 8620	.02881	34.71	216.66	14.719	. 7519	.8671
78800	. 8579	.02867	34.87	216.66	14.719	. 7519	.8671
78900	. 8538	. 02853	35.05	216.66	14.719	. 7519	. 8671

•

Н <sub>с</sub>	¢	TO	1/ (5	5018	1// (0	a
(Feet)	P/PSL					(Knots)
75000	. 04578	. 2140	4.673	25.19	33.50	573.58
75100	. 04556	, 2134	4,685	25,31	33.67	<u>573.58</u>
75200	. 04534	. 2129	4.696	25,43	33.82	573.58
75300	.04512	. 2124	4.707	25.55	33.98	573.58
75400	. 04490	. 2119	4.719	25.68	34.16	573.58
75500	. 04469	. 2114	4.730	25.80	34.32	573.58
75600	. 04448	. 2109	4.741	25.93	34.48	573.58
75700	. 04426	2104	4.753	26.05	34.65	573.58
75800	. 04405	. 2099	4.764	26.18	34.82	573.58
75900	.04384	. 2094	4.776	26.30	34.98	573.58
76000	.04363	. 2089	4.787	26.43	35.16	573,58
76100	. 04342	. 2084	4.799	26,55	35.32	<u> </u>
76200	. 04321	. 2079	4.810	26.68	35.49	573.58
76300	.04300	. 2074	4.822	26.82	35.67	573.58
<u>76400</u>	.04280	. 2069	4.833	26.94	35.83	573.58
76500	. 04259	. 2064	4.845	27.08	36.01	573,58
76600	. 04239	. 2059	4.857	27,20	36.18	573,58
<u>76700</u>	.04218	, 2054	4.868	27.33	<u>36.35</u>	<u> </u>
76800	. 04198	. 2049	4.880	27.47	<b>3</b> 6.54	573,58
76900	. 04178	. 2044	4.892	27.60	36.71	573,58
77000	. 04158	, 2039	4.904	27.73	36.89	573.58
<u>77100</u>	. 04138	. 2034	4.915	27.87	37.07	<u>573.58</u>
77200	. 04118	. 2029	4.927	28.00	37.24	573.58
77300	. 04099	. 2024	<b>4.93</b> 9	28.13	37.41	573.58
77400	. 04079	. 2020	4.951	28.27	37.60	<u>573,58</u>
77500	. 04059	2015	4.963	28.41	37.78	573.58
77600	. 04040	. 2010	4.975	28.55	37.97	573,58
77700	. 04020	. 2005	4.987	28.68	38.14	<u> </u>
77800	.04001	. 2000	4.999	28.82	38.33	573,58
77900	. 03982	. 1996	5.011	28,96	38.51	573,58
78000	. 03963	. 1991	5.023	29.09	38.69	573.58
78100	. 03944	. 1986	5.035	29.24	<u>38,89</u>	<u>573, 58</u>
78200	. 03925	. 1981	5.047	29.38	39.07	573,58
78300	. 03906	. 1976	5.059	29.52	39.26	573.58
78400	. 03887	. 1972	5.071	29.66	39.45	<u>573.58</u>
78500	. 03869	. 1967	5.084	29.80	39.64	573.58
78600	. 03850	. 1962	5.096	29.95	39.83	573,58
<u>78700</u>	. 03832	. 1958	5.108	30.09	40,02	<u>573, 58</u>
78800	. 03813	. 1953	5.120	30.24	40.22	573.58
78900	. 03795	. 1948	5.133	30,39	40.42	573,58

TABLE 9.3

0

.

H <sub>c</sub> (Feet)	Pa (''Hg)	$\delta P_a/P_{aSL}$	1/8	Та (°К)	$\overline{T_a}$	$\theta T_{a}/T_{aSL}$	VÐ
79000	.8497	.02840	35.21	216,66	14.719	. 7519	. 8671
79100	. 8456	.02826	35.38	216,66	14,719	. 7519	.8671
79200	.8416	. 02813	35.54	216.66	14.719	. 7519	. 8671
79300	.8375	. 02799	35.72	216.66	14.719	. 7519	.8671
79400	. 8335	. 02786	35.89	216.66	14,719	. 7519	. 8671
79500	. 8295	. 02772	36.07	216.66	14.719	. 7519	. 8671
79600	. 8256	. 02759	36.24	216.66	14.719	. 7519	. 8671
79700	. 8216	.02746	36.41	216.66	14,719	. 7519	. 8671
79800	. 8176	. 02733	36.58	216.66	14.719	. 7519	.8671
79900	. 8137	. 02720	36.76	216.65	14.719	. 7519	. 8671
80000	. 8098	.02706	36,95	216.66	14.719	. 7519	. 8671

H <sub>c</sub> (Feet)	σ <u>p/psi</u>	<b>σ</b>	1/ √ज	1018	1/810	a (Knots)
79000	.03777	.1943	5.145	30.53	40.60	573.58
79100	.03759	. 1939	5,157	30.68	40.80	573.58
79200	.03741	. 1934	5,170	30.82	40.99	573.58
79300	.03723	. 1929	5.182	30.97	41.20	573.58
79400	,03705	. 1925	5.195	31.12	41.39	573.58
79500	.03687	. 1920	5.207	31.28	41.60	573.58
79600	.03670	.1916	5.220	31.42	41.79	573.58
79700	. 03652	. 1911	5.232	31,57	41.99	573.58
79800	,03634	. 1906	5.245	31,72	42,19	573.58
79900	.03617	.1902	5.258	31,87	42,39	573.58
80000	.03600	.1897	5,270	32.04	42.61	573.58

# MACH NUMBER, M FOR VARIOUS VALUES OF $q_c/P_a$

For 
$$q_c/P_a \le 0.893 \ (M \le 1.00)$$
  
 $\frac{q_c}{P_a} = (1 + 0.2M^2)^{-3.5} - 1$ 

Note:

 $q_c = P_t^{t} - P_a$ 

Where  $P_t'$  = free stream total pressure  $(P_t)$  for subsonic flight.

#### ALSO

INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  FOR VARIOUS VALUES OF  $q_{cic}/P_{s}$ 

For 
$$q_{cic}/P_s \le 0.893$$
 ( $M_{ic} \le 1.00$ )  
 $\frac{q_{cic}}{P_s} = (1 + 0.2 M_{ic}^2)^{-3.5} - 1$ 

Note:

 $q_{cic} = P_t^{i} - P_g$ 

Where  $P_t'$  = free stream total pressure  $(P_t)$  for subsonic flight.

Preceding page blank

MACH NUMBER, M FOR VARIOUS VALUES OF  $\mathfrak{q}_c/P_a$ 

M < 1.00 (Subsonic)

0 10 15099 10 10 10 10 10 10 10 10 10 10	
	2
0.1001 15501 25570 25570 25570 25570 25570 25570 25570 25555 25555 25705	621
0 0 0 0 0 0 0 0 0 0 0 0 0 0	20
0 0 0 0 0 0 0 0 0 0 0 0 0 0	61
0. 4. 0.1.7.7.7.7.7.7.7.7.7.7.7.7.7.7.7.7.7.7.	618
0.00 1300 1804 2159	17
0.0536 13075 22126 22126 22126 22126 22126 22126 22126 22126 22126 22126 22126 22126 22126 22126 22223 22223 22223 22223 22225 22225 22225 22225 22225 22225 22225 22225 22225 22225 22225 225 25	616
1 1 1 1 1 1 1 1 1 1 2 2 2 2 2 2 2 2 2 2 2 2 2	615
0.000 1194 1684 2059 2059 2059 2000	614
9 0 0 0 0 0 0 0 0 0 0 0 0 0	. 29

0.	.6324	641	650	659	668	676	85	693	01	709	717	725	32	740	47	754	762	769	776	. 7829	789	. 7964	803	809	816	822	878	834	841	847
8	63	640	649	653	667	675	84	692	700	708	716	724	32	739	746	754	761	768	75	~	789	.7958	802	808	812 812	821	828	834	840	846
7	. 6305	639	48	657	666	675	83	691	669	20	15	23	731	73	46	753	760	67	774	781	788	.7951	801	808	814	821	827	833	39	845
9	62	638	647	656	665	674	82	069	669	07	15	722	30	38	735	52	759	767	74	.7809	787	. 7944	801	807	814	820	826	833	839	845
ŝ	62	637	647	656	664	673	.6818	690	698	06	714	22	729	737	44	52	759	66	73	80	787	7938	800	07	813	819	826	832	38	844
4	627	637	646	655	663	672	3	689	697	. 7055	713	21	729	36	4	51	28	765	772	. 7795	786	7931	662	806	12	819	825	831	38	844
Э	626	636	645	4	663	671	680	683	969	. 7047	712	720	282	735	) 4 ) 4	750	757	- 49	1	. 7788	785	7924	799	805	12	818	24	31	37	43
2	22	35	44	5	5	670	679	6.87	 695	. 7039	211	710	100	1 2 5	4	14	757	764	5	.7782	2	2162	80	805	1	818	824	830	36	80
1	624	634	1 4 9 1 4 9 1 6 4 9		561	1009	0 0 0	686	202	. 7031	711	710	775	2 C 7 C	5	121	 	3 2 6	220	. 7775	0		:5	804	52	5 I 3	- C 00	220	836	842
0	623	633	543	1 ~ 7 v	222		007	207		, 7023	ī		0 1 0	3 6	<u>ה</u> ה	- 0			101	. 7768	102	n c b d	207	. U 8	20	816	2 0	2 C C C C C C C C C C C C C C C C C C C	835	.8417
Р	-c. a					# u	cc. 75			90.		. 40	• 4 • 1 • •	. 40	• •	. 44. 1 1	C# •	• 40	- 7. •	• • •						י ע י	ישר י	י ע י	ישר י	E 9

TABLE 9,4

6	53	59	64	870	76	. 8821	87	93	898	9	Ö	914	20	Ñ.	30	ŝ	40	4	50	955	٠Q	965	202	975	79	984	600	9995 9995	5	
80	85	85	86	87	87	.8816	88	89	89	96	Ō	16	-	~	. 9301	<b>n</b>	প	-	ŝ	ഹ	Q.	5	969	74	79	84	.9886	61	76	
7	851	57	863	869	75		886	892	897	ŝ	0	913	919	24	929	934	39	944	949	54	59	64	969	974	78	983	.9881	992	67	
9	851	857	63	869	874	88	886	891	897	$\sim$	908	E ]	918	23	9	4	939	944	949	.9542	59	964	968	73	978	983	.9877	992	966	
5	50	56	862	689	874	29	885	16	896	.9020	07	912	918	923	92	33	938	943	948	953	58	963	968	973	17	982	.9872	166	966	
ষ	850	856	862	867	873	29	885	890	896	• 9015	906	12	917	922	928	933	38	943	948	.9533	958	63	967	972	977	982		166	966	
3	849	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	861	867	873	878	884	890	895	. 9009	90		512	922	927	32	937	942	947	.9528	957	62	967	972	76	981	986	066	. 9955	000
2	4	854	503	866	872	878	5 8 8 5 8 3	889	894	.9004	905	11	16	921	, <u>-</u> 6 9 2 6	932	937	942	947	. 9523	957	52	986	126	926	981	985	000	. 9950	666
			ነፈ	200			.α	$\infty$		<u>ነ</u>	9052	2	5	0		6	5	) 🗨	0	השיי	u o	0	0.0	6	• •	a	$\mathbf{x}$	0	66	$\sim -$
0	<b>P.47</b>	- 40	ים מיו	200	200	110	- 0	) 0 1 0	000 000	. 8993	004	5 9	2.5		0 2 2 2 5	""	420		946	. 6	420		299	5	0 <b>7</b> 5	080	8 4 8 4 4	080	5 <b>7</b>	. 9987
<u>д</u>		<b>2</b> 0.	10.	20. 27.	60. Vy	н ч У Ч	2 4	.00		69 ·	-			 	47	•			- ۲	. 79	q			a		a	c a	. c	0 0	. 89
		Τ£	B	L	E '	9.	4								312	)														

MACH NUMBER, M FOR VARIOUS VALUES OF  $q_c/P_a$ FOR M  $\geq$  1.00 (Supersonic)

$$\frac{q_c}{P_a} = \frac{166.921 M'}{(7 M^2 - 1)^{2.5}} - 1$$

Note:

$$^{q}c = P_{t}' - P_{a}$$

Where  $P_t^{\dagger}$  = total pressure behind the shock for supersonic flight.

#### ALSO

 $q_{cic}/P_{8}$  FOR VARIOUS VALUES OF INDICATED MACH NUMBER CORRECTED FOR INSTRUMENT ERROR,  $M_{ic}$  FOR  $M \ge 1.00$ (Supersonic)

$$\frac{q_{cic}}{P_{g}} = \frac{166.921 M_{ic}}{(7 M_{ic}^{2} - 1)^{2.5}} - 1$$

Note:

$$q_{cic} = P_t^{\dagger} - P_s$$

Where  $P_t^{\dagger}$  = total pressure behind the shock for supersonic flight.

ココフタ パチキション

MACH NUMBER, M FOR VARIOUS VALUES OF  $q_c/P_a$ 

M≥1.00 (Supersonic) (q<sub>c</sub> Takes Account of Loss Through Normal Shock)

6	91	93	6	98	1.006	03	05	80	10	13	. 15	18	. 20	. 23	. 26	. 29	31	.34	.37	.40	4	1,464	. 49	. 52	. 55	, 58	. 61	.64	. 67	. 71
œ	91	. 93	95	98	1.004	02	05	07	10	12	. 15	. 18	1.207	. 23	. 26	. 28	. 31	. 34	.37	.40	4	1.461	. 49	. 52	. 55	. 58	. 61	. 64	. 67	. 70
1	90	93	95	97	1.001	. 02	05	07	10	12	-	Γ.	1.204	2.	2.	2.	3	<b>"</b>	۳.	ŝ	•	1.458	4,	<b>.</b> ک	ŝ	ŝ	• و	•	• •	~ .
6	0	92	95	57	666.	. 02	. 04	07	. 09	. 12	14	.17		. 22	. 25	. 28	. 31	. 33	. 36	•	.42	1.455	.48	. 51	. 54	. 57	. 60	. 63	. 66	0
5	06	92	94	97	.997	02	04	07	60	12	.14	.17	S.	. 22	. 25	. 28	. 30	. 33	. 36	. 39	4.	1.452	.48	. 51	54	. 57	60	63	<u>ę</u>	69
4	90	.92	94	97	. 994	0	40	90	60	11	14	. 16	1.196	. 22	. 25	. 27	. 30	. 33	. 36	• 39		1.449	47	50	53	56	60	63	66	6
ę	90	92	94	96	.992	0	64	90	60	]]	4	16	1.193	22	24	27	30	33	35	co	- 44	1.446	47	50	53	56	56	62	66	5
2	6	92	94	96	. 989	5	03	. 06	08	-	ŝ	. 16	1.190	T	24	. 27	0	. 32	. 35	. 38		1.443	47	50	ŝ	56	65	. 62	65	. 68
l	6	16	4	9	6		03	90	80	11	1.136	9	18		24	26	29	$\sim$	ŝ	80	4.		~	9	. 52	56	. 59	$\sim$	65	63
0	89	_	93	Ŷ	98	0	03	05	03			.15	13	.21	23	9	29	2	35	37	1.408	43	46	9	. 52	. 55	58	~	65	co
<u>ک</u> ۲ F	1.0	1.0	0	0	1.04	°.	Ó	0	0		1.10							•=	-		1.20	2	2	2		2				

6	.74	. 77	i.808	. 84	. 87	. 90	<b>*</b> 6 .	.97	.01	.04	08	11	15	18	22	26	$\sim$	33	37	2.410	44	48	2,524	. 56	. 60	. 64	. 68	. 72	. 76	. 80
80	. 74	. 77	1.805	. 83	. 87	. 90	. 93	.97	.00	. 04	. 07	. 11	. 14	.13	. 22	. 25	2.294	. 33	. 36	.40	. 44	.48	2.520	. 55	. 59	. 63	. 67	. 71	. 75	. 79
2		~	1.801	æ.	°,	6	<u>ۍ</u>	6	2	٩.	.07	.10	. 14	.18	. 21	. 25	2.290	. 32	.36	4	. 44	.47	2.517	. 55	. 59	. 63	. 67	.71	. 75	. 79
9	. 73	. 76	1.798	. 83	. 86	. 89	. 93	. 96	.00	. 03	. 07	<u>.</u>	.14	.17	. 21	. 25	2.287	. 32	. 36	. 39	.43	.47	2.513	. 55	. 59	. 63	. 66	. 70	74	78
Ś	. 73	. 76	1.795	. 82	.86	. 89	.92	.96	.99	. 03	. 06	. 10	.13	. 17	. 21	. 24	2.283	. 32	. 35	. 39	43	47	2.509	45	500	62	66	70	74	78
Ŧ	. 72	. 75	1.792	. 82	. 85	. 89	.92	.96	66.	. 02	. 06	. 09	. 13	.17	. 20	. 24	2.279	.31	. 35	. 39	42	.46	2.505	5.4	8 <u>6</u> .	.62	. 66	30	74	. 78
ŝ	. 72	. 75	1.788	. 82	. 85	. 88	.92	.95	.99	. 02	.06	<del>.</del> 99	<u>е</u> г.	. 16	. 20	. 23	2.276	. 3]	. 34	. 38	.42	.46	2.501	. 54	. 57	. 61	. 65	. 69	. 73	. 77
2	1.720	1.753	1.785	. 81	. 85	88	.91	. 95	.98	. 02	. 05	60.	. 12	. 16	. 19	. 23	2	. 30	. 34	œ	.42	.45	2.497	• 53	. 57	.61	. 65	. 69	. 73	. 77
-	. 71	. 74	1.782	. 81	. 84	. 88	.91	.94	.98	0	ŝ	.08	.12	. 15	. 19	. 23	. 26	. 30	.34	•	.41	•45	2.494	. 53	. 57	. 61	. 64	. 68	. 72	. 76
0	.71	. 74	1.779	. 81	. 84	. 87	. 91	.94	.98	6.	2.050	. 08	. 12	.15	. 19	. 22	. 26	. 30	. 33			. 45	. 49	. 52	. 56	. 60	. 64	. 68	. 72	. 76
W	÷.	÷.	1.32	ς. Έ	<b>۳</b> .	1.35	1.36	<b>۴</b>	ŝ	m.	<u>ч</u>	Υ.	4	4	Τ.	4	1.46	4	4	1.49	ŝ	ŝ	1.52	പ്	ĥ	۳	1.5	<b>1.</b> 5	1.5	5

315

1	σ.	. 84	. 88	.92	.96	00.	, <b>04</b>	. 09	. 13		. 22	. 26	. 30	. 35	. 39	.43	.48	3.529	. 57	. 62	. 66	. 71	. 75	. 80	. 85	. 89	. 94	.99	.04	4.089	. 13
	œ	. 83	.87	.92	. 96	00.	. 04	. 08	. 13	3.173	. 21	. 25	. 30	. 34	. 39	.43	.48	3. 524	. 56	. 61	. 66	٢.	. 75	. 79	. 84	. 89	.94	. 98	. 03	4.084	. 13
	7	. 83	.87	.91	.95	.99	.04	.08	.12	3.168	.21	. 25	. 29	. 34	.38	.43	.47	3, 520	. 56	.61	• 65	5	.74	. 79	.84	.88	.93	.98	. 03	07	.12
	6	. 83	. 87	.91	. 95	. 99	. 03	. 07	.12	3.164	. 20	. 25	. 29	. 33	. 38	.42	.47	3,515	. 56	, <b>6</b> 0	. 65	. 69	. 74	. 79	. 83	. 88	.93	- 52	. 02	4.075	. 12
	2	8	.86	.90	. 94	.99	.03	.07	. 11	9	. 20	. 24	. 28	.33	.37	.42	.46	.51	. 55	. 60	. 64	3.693	. 73	. 78	. 83	.87	.92	.97	. 02	. 07	
	4	. 82	.86	.90	.94	.98	. 02	.07	. 11	3,156	. 19	. 24	. 28	. 32	.37	.4]	.46	50	. 55	. 59	ف.	. 68	. 73	. 78	. 82	. 87	.92	. 96	0.	4.065	
	e	. 81	. 85	. 89	- 94	.98	. 02	. 06	. 10	3.151	. 19	. 23	. 28	.32	.36	.41	.45	3.502	• 54	. 59	. 63	. 68	. 73	.77	. 82	. 87	.91	.96	ō.	4.060	. 10
	2	~	8	8	6.	6.	°.	°.	~	-	-	•	2	<b>.</b>	<u>е</u> ,	.40	.45	4	. 54	. 58	3	. 67	٢.	. 77	. 81	. 86	.91	. 95	8.	4.055	. 10
	7	. 81	. 85	. 89	. 93	.97	5.0	.05	.10	4	. 18	. 22	. 27	.31	. 35	. 40	.44	6	° 5 9	. 58	. 62		. 72	. 76	. 81	. 86	6.	. 95	8	4.051	ě6
	0	. 80	. 84	. 88	.92	- 97	.01	. 05	. 09	ŝ	.18	. 22	. 26	.31	. 35	. 39	<b>1</b>	3.488	. 53	. 57	. 62	<u>، ہ</u>	.71	. 76	. 80	. 85	.90	. 95	. 99	4.046	60 •
	M	1.60	9	9	9	I.64	Ŷ	9	1.67	1.68	1.69	1.70					~	~		~	1.79	1.80	00	8.	°.	¢,	80	œ.	00	1.88	άÕ

6	. 18	. 23	, 2 8	. 33	. 38	.43	. 48	4.534	. 58	. 63		. 73	. 79	<b>8</b> 4	. 89	94	ō.	.05	10	ē	2.	. 26	. 32	.37	.43	.48	.54	. 59	5.655	.71
80	. 18	. 23	. 27	. 32	. 37	.42	.47		. 58	. 63		. 73	. 78	. 83	. 88	. 94	.99	.04	. 10	. 15		. 26	.31	. 37	.42	.48	. 53	. 59	. 64	. 70
7	.17	. 22	. 27	. 32	.37	.42	.47	4.524	.57	. 62	4.677	.72	. 78	83	.88	.93	.98	.04	.09	.14	. 20	. 25	.31	.36	.42	.47	. 53	.58	<b>64</b>	. 70
9	. 17	. 22	. 26	.31	.36	.41	.46	-	. 56	. 62	. 6	. 72	. 77	2	. 87	. 93	. 98	. 03	.09	. 14	, 19	. 25	. 30	. 36	.41	.47	. 52	. 58	s and a second s	69.
2	.16	.21	. 26	.31	.36	.41	.46	4.514	. 56	. 61	_	. 71	. 76	. 82	.87	.92	.97	. 03	.08	. 13	5.193	. 24	. 30	.35	. 41	.46	.52	.57	, 63	. 68
4	. 16	.21	. 26	. 30	. 35	.40	.45		. 55	. 61	۰ و	. 71	. 76	. 81	. 86	.92	. 97	. 02	.08	. 13	00	. 24	. 29	. 35	.40	.46	.51	. 57	. 62	. 68
Ś	. 15	. 20	. 25	.30	. 35	.40	.45	0	. 55	. 60	9.	. 70	. 75	4.811	. 86	.91	, 96	. 02	.07	. 12	. 18	. 23	. 29	. 34	. 39	.45	.51	. 56	5.621	. 67
2	. 15	. 20	. 25	. 29	. 34	. 39	44	49	54	. 60		. 70	. 75	. 80	. 85	.91	.96	.01	. 06	. 12	~	. 23	. 28	. 33	. 39	. 44	. 50	. 56	. 61	. 67
I	. 14	. 19	. 24	. 29	34	. 39	44	4.493	54	. 59		. 69	.74	.80	. 85	.90	. 95	0.	. 06	Ξ.	17	. 22	. 27	. 33	.38	. 44	.49	. 55	. 61	. 66
0	4	. 19	24	28	33	38	43	8	53	. 59		. 69	. 74	. 79	84	.90	. 95	00.	. 05	. 11	16	. 22	. 27	.32	.38	.43	.49	.54	. 60	. 66
M	•	•	•	•	•	•	•		•	•	2.00			•	•				•	•										

6		. 82	. 88	.93	.99	. 05	. 11	.17	. 22	. 28	4	.40	.46	.52	. 58	. 64	6.708	. 76	. 83	. 89	.95	0.	.07	. 14	20	. 26	. 32	. 39	7.456	52
80		8.	. 87	.93	. 99	.04	. 10	. 16	. 22	. 28	4	40	46	52	58	64	6.702	76	82	88	Ġ,	80.	. 07	.13	. 19	. 25	. 32	.38	7.450	51
~	5.756	. 81	.87	.92	.98	.04	.10	. 15	. 21	. 27	.33	. 39	.45	5.	. 57	. 63	6.695	. 75	. 81	. 87	. 94	00.	. 06	.12	. 19	. 25	.31	.38	7.443	. 50
9	5.751	. 80	. 86	.92	.97	. 03	. 09	. 15	. 21	. 27	۴.	. 38	. 44	. 50	. 56	. 62	6.689	. 75	. 81	. 87	. 93	. 99	. 05	.12	. 18	. 24	. 30	.37	7.437	. 50
		. 80	. 85	16.	.97	.03	.08	. 14	. 20	. 26	.32	. 38	44	. 50	. 56	. 62	6.683	- 74	. 80	.86	.92	.99	. 05	.11	.17	. 24	. 30	.36	7.431	46
4	. 74	. 79	. 85	6.	. 96	. 02	. 08	. 14	6.200	. 25	.31	.37	.43	.49	. 55	. 61	6.677	. 73	. 79	. 86	. 92	. 98	. 04	. 10	. 17	. 23	. 29	. 36	7.424	.48
٣	. 73	. 79	. 84	.90	.96	5.01	.07	.13	ó.194	. 25	.31	.37	.43	.49	. 55	.61	6.671	. 73	. 79	• 85	.91	.97	. 03	.10	. 16	. 22	. 29	ů.	7.418	<b>. 4</b> 8
2	. 72	. 78	. 84	. 89	02	0.	. 07	.13	6.188	. 24	<u>ج</u>	. 36	.42	.48	.54	. 60	6, 665	.72	. 78	. 84	.91	.97	. 03	<b>60</b> .	.15	. 22	- 28	. 34	7.411	.47
I	.72	. 77	. 83	89	.95	00	. 06	.12	•	4	30	. 35	.41	.47	. 53	. 59	•	.72	. 78	. 84	Ó	.96	. 02	• 09	. 15	. 21	. 27	. 34	7.405	• 46
0	11.	. 77	. 83	88	94	8	. 06	11	.17		- 29	35	41	47	. 53	. 59	6.653	.71	. 77	. 83	· 80	.95	. 02	. 08	. 14	. 20	. 27	. 33	7.399	.46
∑ BLF	2.2	2.2	2.2	2		5			2	2.29	ŝ	ς Γ	ŝ	Ĵ,	<i>с</i> .	<del>ر</del>	2, 36	<del>ب</del>	<b>.</b>	ς.	2.40	4	4	4	4	4	4	4	4	4

6	7.584	م	~	~	8	6	9	0	-	٦.			06.	8.376	44.	5.	.58	64		7	. 78	. 85	6	ð		?	-	2	2	n.	9.419	4	ŝ
8	7.578	2	70	77	83	90	96	03	10	16	ĉ	1	30	8.370	43	50	57	2	5 i		78		6	99	) U (	Ŝ	12	6	27	34	9.412	4	55
7	7.571	63	70	76	83	89	96	02	60	-	. (	V	2	ŝ	4	-4	ŝ	1	0 (		~	8.842	5	. a	) u \ (	20	12	5	26	33	9.405	47	54
ę	. 56	. 63	69	22	. 82	89	. 95	02	08	8, 155		77	28	35	42	49	5	20	20	69	76	8,835	0 0	0		5 4	Ξ	8	25	32	9.398	46	54
ŝ	. 55	. 62	68	75	6	00	4	ξē		8,148	• . • •	. 21	. 28	. 34	41	48	) ע ע יע	) ( ) \	0	. 69	.75	8.828	80		) (   	50.	.10	.17	. 24	.31	9.390	.46	. 53
4	്റ	9	9	. ~	. ~	) œ	ο.		, c	8 142	•	. 20	. 27	34	41	47		5.	5	. 68	. 75	8.821	00	20.		. 03	2.	.17	. 24	31	9.383	.45	. 52
ß	5.4	61	5		. a		50	• C	2	0.000 8 135	1 4 9	20	. 26	60 133	40		- 7  - 1  -		. 60	. 67	. 74	8.814	a	• •	<b>.</b>	. 02	. 09	. 16	. 23	. 30	9.376	.44	2
2	ŝ	ý ý		, 4 , 7		- J - J - J	200		77.	? -		. 19	. 26	32	100		2 C 2 U 2 U		. 60	. 66	73	8, 807	0	•	ざろう	5	. 08	15	. 22	29	. 36	44	
1	53		• • •	2.5	• - • -		6.0	2 C	0 u 7 (	0° U 3 3	• 1 0	. 18	25		) ( ) (	- u 		2	ŝ	. 66	5	8,800	10			8.0	. 07	14	. 22	29	.36	43	9.505
0	52		- u - u - u		 	0 u ~ 0	6.6	12.	0 7 0	8. U48 0 115		18	44	• •	• 0 • 0	0 u 7 1		7	• 58	. 65		8, 793		0 0 0	5.6	8.00	07	4	21	28	30	42	9.498
X					•								•								•	5° 40						•		i ~	i ~	2	6 <b>2.</b> 79

6	9.635	•		9.853					•	٠	•	0	0	0	0	0		ò		11.054	
80		-	-	9.845		-	-					0	0	0	0	0		0	Ċ	11.046	
7				9.838							0.3	0.4	0.5	0.5	0.6	0.7	00	رە رە	σ	11.038	
6				9.831																11.031	
Ś				9.823	•					•	<b>.</b>	ô	•	0	•	•	10.793	0		11.023	
4	9.599	9.671	9.743	9.816	9.889	9.962	10.035	10.110	10.184	10.258							10.785				
ŝ	-			9.809				-			32	40	47	ŝ	62	70	10.778	85	93	00	
2	58	65	72	80	87	94	02	6	. 16		10.317	. 39	.46	54	.61	69		84	92	8	
l											10.310	•	•	•	<b>。</b>	•		0	•	ċ	
0											10.302							-	-		11.061
W											2.90										-

DIFFERENTIAL PRESSURE, q<sub>c</sub> ("Hg) FOR VARIOUS VALUES OF CALIBRATED AIRSPEED, V<sub>c</sub> (Knots)

$$\frac{q_{c}}{P_{aSL}} = \left[1 + 0.2 \left( V_{c} / a_{SL} \right)^{2} \right]^{3.5} - 1 \qquad V_{c} \leq a_{SL}$$

$$\frac{q_{c}}{P_{aSL}} = \frac{\frac{166.921 (V_{c}/a_{SL})'}{[7 (V_{c}/a_{SL})^{2} - 1]^{2.5}} - 1 \qquad V_{c} \ge a_{SL}$$

where P<sub>aSL</sub> = 29,92126 "Hg

$$a_{SL} = 661.48$$
 Knots

Note:

w

$$q_c = P_t' - P_a$$

where  $P_t'$  = free stream total pressure ( $P_t$ ) for subsonic flight  $P_t'$  = total pressure behind the shock in supersonic flight

#### ALSO

INDICATED DIFFERENTIAL PRESSURE,  $q_{cic}$  ("Hg) FOR VARIOUS VALUES OF INDICATED AIRSPEED CORRECTED FOR INSTRUMENT ERROR,  $V_{ic}$  (Knots)

$$\frac{q_{cic}}{P_{aSL}} = \left[1 + 0.2 \left(\frac{V_{ic}}{a_{SL}}\right)^{2}\right]^{3.5} - 1 \qquad V_{ic} \leq a_{SL}$$

$$\frac{q_{cic}}{P_{aSL}} = \frac{166.921 \left(\frac{V_{ic}}{a_{SL}}\right)^{7}}{\left[7 \left(\frac{V_{ic}}{a_{SL}}\right)^{2} - 1\right]^{2.5}} - 1 \qquad V_{ic} \geq a_{SL}$$
here  $P_{aSL} = 29.92126$  "Hg  
 $a_{SL} = 661.48$  Knots

Note:

$$q_{cic} = P_t' - P_s$$

where  $P'_t$  = free stream total pressure ( $P_t$ ) for subsonic flight  $P'_t$  = total pressure behind the shock for supersonic flight

								V	
V <sub>c</sub> (Knots)	9c ("Hg)	V <sub>c</sub> (Knots)	9c ("Hg)		V <sub>c</sub> (Knots)	<sup>q</sup> c ( <u>" Hg</u> )		V <sub>c</sub> ( <u>Knots</u> )	<sup>9</sup> c ( <u>"Hg</u> )
0.0	0,0000	20.0	0.0192		40.0	0.0767		60.0	0.1727
0.5	0.0000	20.5	0.0201		40.5	0,0786		60.5	0.1756
1.0	0.0001	21.0	0.0211		41 0	0.0805		61.0	0.1785
1,5	0,0001	21.5	0.0221		41.5	0.0825		61.5	0.1814
2.0	0.0002	22.0	0,0232		42.0	0.0845		62.0	0.1844
2.5	0.0003	22.5	0.0242		42.5	0.0866		62.5	0.1874
3.0	0.0004	23.0	0.0253		43.0	0,0886		63.0	0.1904
3.5	0.0006	23.5	0.0264		43.5	0.0907		63.5	0.1935
4.0	0.0008	24.0	0.0276		44.0	0.0928		64.0	0,1965
4,5	0.0010	24.5	0.0287		44.5	0.0949		64.5	0.1996
			+						
5.0	0,0012	25.0	0.0299		45.0	0.0970		65.0	0.2027
5.5	0.0014	25.5	0.0311		45.5	0.0992		65.5	0.2059
6.0	0.0017	26.0	0.0324		46.0	0.1014		66.0	0.2090
6.5	0.0020	26.5	0.0336		46.5	0,1036		66.5	0.2122
7.0	0.0023	27.0	0.0349		47.0	0.1059		67.5	0.2134
7.5	0.0027	27.5	0.0362		47.5	0.1081			0.2219
8.0	0.0031	28.0	0.0375		48.0	0,1104		68.0	
8.5	0,0035	28.5	0.0389		48.5	0.1127		68.5	0.2252
9.0	0.0039	29.0	0.0403		49.0	0.1151		69.0	0.2285
9.5	0.0043	29.5	0.0417		49.5	0.1175		69.5	0.2319
10.0	0.0048	30.0	0.0431		50.0	0,1198		70.0	0.2352
10.5	0.0053	30.5	0.0446	1	50,5	0.1223		70.5	0.2386
11.0	0.0058	31.0	0.0460		51.0	0.1247		71,0	0.2420
11.5	0.0063	31.5	0.0475		51.5	0.1272		71,5	0.2454
12.0	0,0069	32.0	0.0490		52.0	0.1296		72.0	0.2489
12.5	0.0075	32.5	0.0506		52.5	0.1321		72,5	0.2524
13.0	0.0081	33.0	0.0522		53.0	0.1347		73.0	0.2559
13,5	0.0087	33,5	0.0538	1	53.5	0.1372		73.5	0.2594
14.0	0.0094	34.0	0.0554		54.0	0.1398		74.0	0.2629
14.5	0.0101	34.5	0.0570	ì	54.5	0.1424		74.5	0.2665
15.0	0.0108	35.0	0.0587		55.0	0.1451		75.0	0.2701
15.0	0.0108	35.5	0.0604	1	55.5	0,1477		75,5	0.2737
16.0	0.0113	36.0	0,0621	1	56,0	0,1504		76.0	0.2774
16.5	0.0120	36.5	0.0638		56.5	0,1531		76.5	0.2811
17.0	0.0138	37.0	0.0656	}	57.0	0.1558		77.0	0.2848
17.5	0.0138	37.5	0.0674	Į	57.5	0,1586	]	77,5	0.2885
18.0	0.0155	38.0	0.0692	1	58.0	0.1613	1	78.0	0.2922
18.5	0.0155	38.5	0.0710		58.5	0.1641	1	78.5	0.2960
19.0	0.0173	39.0	0.0729		59.0	0.1670	}	79.0	0.2998
19.5	0.0182	39.5	0.0748		59.5	0.1698		79.5	0.3036
× 7 • J	10.0102	57.5	0.0140	ł –		1	1		

## DIFFERENTIAL PRESSURE, q<sub>c</sub> (" Hg) FOR VARIOUS VALUES OF CALIBRATED AIRSPEED, V<sub>c</sub> (Knots)

V <sub>c</sub> (Knots)	q <sub>c</sub> ("Hg)	V <sub>c</sub> (Knots)	9 <sub>C</sub> (" Hg)	V <sub>C</sub> (Kriots)	<sup>q</sup> c ( <sup>π</sup> Hg)	V <sub>c</sub> (Krots)	<sup>q</sup> c (" Hg)
80.0 80.5	0.3075	100.0 100.5	0.4814 0.4863	120.0	0.6950	140.0 140.5	0.9488
81.0 81.5 82.0	0.3152 0.3192 0.3231	101.0 101.5 102.0	0.4912 0.4961 0.5010	121.0 121.5 122.0	0.7067 0.7126 0.7185	141.0 141.5 142.0	0.9625 0.9694 0.9764
82.5 83.0 83.5	0.3271 0.3311 0.3351	102.5 103.0 103.5	0.5059 0.5109 0.5159	122.5 123.0 123.5	0.7245 0.7305 0.7365	142.5 143.0 143.5	0.9833 0.9903 0.9974
84.0 84.5	0.3391 0.3432	104.0 104.5	0.5209 0.5260	124.0 124.5	0.7425	144.0 144.5	1.004
85.0 85.5 86.0 86.5	0.3473 0.3514 0.3555 0.3597	105.0 105.5 106.0 106.5	0,5311 0,5362 0,5413 0,5465	125.0 125.5 126.0 126.5	0.7546 0.7607 0.7669 0.7730	145.0 145.5 146.0 146.5	1.019 1.026 1.033 1.040
87.0 27.5 88.0 88.5	0.3639 0.3651 0.3723 0.3766	107.0 107.5 108.0 108.5	0.5516 0.5568 0.5621 0.5673	127.0 127.5 128.0 128.5	0.7792 0.7854 0.7916 0.7979	147.0 147.5 148.0 148.5	1.047 1.054 1.062 1.069
39.0 89.5	0.3809 0.3852	109.0 109.5	0.5726	129.0	0,8042 0.8105	149.0 149.5	1.076
90.0 90.5 91.0 91.5	0.3895 0.3939 0.3983 0.4027	110.0 110.5 111.0 111.5	0.58 <b>32</b> 0.5886 0.5939 0.5993	130.0 130.5 131.0 131.5	0.8168 0.8232 0.8295 0.8360	150.0 150.5 151.0 151.5	1.090 1.098 1.106 1.113
92.0 92.5 93.0 93.5	0.4071 0.4116 0.4161 0.4206	112.0 112.5 113.0 113.5	0.6048 0.6102 0.6157 0.6212	132.0 132.5 133.0 133.5	0.8424 0.8488 0.8553 0.8618	152.0 152.5 153.0 153.5	1.120 1.128 1.136 1.143
94.0 94.5	0.4251 0.4297	114.0 114.5	0.6267 0.6323	134.0	0.8684 0.8749	154.0 154.5	1.151 1.158
95.0 95.5 96.0 96.5	0.4342 0.4388 0.4435 0.4431	115.0 115.5 116.0 116.5	0.6379 0.6435 0.6491 0.6547	135.0 135.5 136.0 136.5	0.8815 0.8881 0.8948 0.9014	155.0 155.5 156.0 156.5	1.166 1.173 1.181 1.189
97.0 97.5 98.0	0.4528 0.4575 0.4623	117.0 117.5 118.0	0.6604 0.6661 0.6718	137.0 137.5 138.0	0.9014 0.9081 0.9148 0.9216	157.0 157.5 158.0	1.197 1.204 1.212
98.5 99.0 99.5	0.4670 9.4718 0.4766	118.5 119.0 119.5	0.6776 0,6834 0.6892	138.5 139.0 139.5	0.9283 0.9351 0.9419	158,5 159.0 159.5	1,220 1,228 1,236

·	······	· · · · · · · · · · · · · · · · · · ·		F			- <del>y</del>
V <sub>c</sub>	9c	V <sub>c</sub>	9 <sub>c</sub>	V <sub>c</sub>	Sc	V <sub>c</sub>	9 <sub>C</sub>
(Knots)	(" Hg)	(Knots)	( <b>*</b> Hg)	(Knots)	( <u>" Hg)</u>	(Knots)	(* Hg)
160.0	1,243	180.0	1.580	200.0	1.959	220.0	2.382
160.5	1.251	180.5	1.589	200.5	1.969	220.5	2.393
161.0	1.259	181.0	1.598	201.0	1.979	221.0	2.404
161.5	1.267	181.5	1.607	201.5	1.989	221.5	2.415
162.0	1.275	182.0	1,616	202.0	1.999	222.0	2.426
162.5	1.283	182.5	1.625	202.5	2.009	222.5	2,439
163.0	1.291	183.0	1.634	203.0	2.019	223.0	2.449
163.5	1.299	183.5	1.643	203.5	2.030	223.5	2.460
		184.0	1.652	204.0	2.040	224.0	2.471
164.0	1.307	184.5	1.661	204.5	2.050	224.5	2.483
164.5	1.315	104. 5	1.001	204.5	2.050		+
165.0	1.324	185.0	1.671	205.0	2.060	225.0	2.494
165.5	1,332	185.5	1.680	205.5	2. <b>07</b> 0	225.5	2,506
166.0	1.340	186.0	1.689	20 <b>6.</b> 0	2.081	226.0	2.517
166.5	1.348	186.5	1. <b>6</b> 98	206.5	2.091	226.5	2.529
167.0	1.356	187.0	1,708	207.0	2.102	227.0	2.540
167.5	1.365	187,5	1.717	207.5	2.112	227.5	2.552
168.0	1.373	188.0	1.726	208.0	2,123	228.0	2.563
168.5	1.381	188.5	1.736	208.5	2,133	228,5	2.575
169.0	1.390	189.0	1.745	209.0	2,144	229.0	2.586
169.5	1.398	189.5	1.754	209.5	2.154	229.5	2.598
170.0	1.406	190.0	1.764	210.0	2,165	230.0	2.610
170.5	1.400	190.0	1.773	210,5	2.175	230.5	2.621
171.0	1.423	191.0	1.783	211.0	2,186	231.0	2.633
171.5	1.423	191.0	1.792	211.5	2,196	231.5	2.645
172.0	1.432	192.0	1,802	212.0	2,207	232.0	2.657
172.5	1.449	192.5	1,802	212.5	2.218	232.5	2.668
173.0	1,457	193.0	1,812	213.0	2, 22.9		2.680
173.5	1.466	193.5	1.831	213.5	2.239		2.692
174.0	1.474	194.0	1.841	214.0	2,250	234.0	2.704
174.5	1.483	194.5	1.850	214.5	2,261	234.5	2.716
113.5	<b>+</b>			}	<u> </u>	!	4
175.0	1.492	195.0	1.860		2.272	235.0	2.728
175.5	1.500	195.5	1,870		2,283		2,740
176.0	1.510	196.0	1.880		2,293	1 1	2.752
176.5	1.518	196.5	1,889		2, 304	236.5	2.764
177.0	1.527	197.0	1,899		2.315	237.0	2.776
177.5	1.536	197,5	1.909		2, 326	237.5	2,788
178.0	1.544	198.0	1.919		2,337	238.0	2.800
178.5	1.553	198.5	1.929		2. 348	238.5	2.812
179.0	1.562	199.0	1.939		2.359	239.0	2.825
179.5	1.571	199.5	1,949	219.5	2, 370	239.5	2,837

C

V <sub>c</sub> (Knots)	<sup>9</sup> c (" Hg)	V <sub>c</sub> (Knots)	9c (" Hg)	V <sub>c</sub> (Knots)	<sup>9</sup> с (" Нg)	V <sub>c</sub> (Knots)	9c ("Hg)
240.0 240.5 241.0	2.849 2,861 2,874	260.0 260.5 261.0	3.363 3.376 3.390	280.0 280.5 281.0	3.924 3.939 3.953	300.0 300.5 301.0	4.534 4.550 4.566
241.5 242.0	2.886 2.898	261.5 262.0	3.403 3.417	281.5 282.0	3.968 3.983 3.997	301.5 302.0	4,582 4,598 4,614
242.5 243.0 243.5	2.911 2.923 2.936	262.5 263.0 263.5	3.430 3.444 3.458	282.5 283.0 283.5	4.012 4.027	302.5 303.0 303.5	4.630 4.646
244.0 244.5	2.948 2.961	264.0 264.5	3.471 3.485	284.0 284.5	4.042 4.057	304.0 304.5	4.662 4.678
245.0 245.5 246.0	2.973 2.986 2.998	265.0 265.5 266.0	3.499 3.512 3.526	285.0 285.5 286.0	4.072 4.087 4.102	305.0 305.5 306.0	4,695 4,711 4,727
246.5 247.0 247.5	3.011 3.024 3.036	266.5 267.0 267.5	3.540 3.554 3.568	286.5 287.0 287.5	4.117 4.132 4.147	306.5 307.0 307.5	4.743 4.760 4.776
242.0 248.5	3.049 3.062	268.0 268.5	3.581 3.595	288.0 288.5	<b>4.1</b> 62 <b>4.1</b> 77	308.0 308.5 309.0	4.792 4.809 4.825
249.0 249.5	3.074 3_087	269.0 269.5	3.609	289.5	4.192 4.208	309.5	4.842
250.0 250.5 251.0	3.100 3.113 3.126	270.0 270.5 271.0	3.637 3.651 3.665	290.5 291.0	4.223 4.238 4.253		4.858 4.875 4.891
251.5 252.0 252.5	3.139 3.152 3.165	271.5 272.0 272.5	3.680 3.694 3.708	292.0	4.269 4.234 4.299	312.0	4.908 4.925 4.941
253.0 253.5 254.0	3.178 3.191 3.204	273.0 273.5 274.0	3.722 3.736 3.750	293.0 293.5	4.315 4.330 4.346	313.0 313.5	4.958 4.975 4.991
254.5	3.217	274.5	3.765	294.5	4.361	314.5	5.008
255.0 255.5 256.0	3.230 3.243 3.256	275.0 275.5 276.0	3.779 3.793 3.808	295.5	4.377 4.393 4.408	315.5	5.025 5.042 5.059
256.5 257.0 257.5	3,269 3,283 3,296	276.5 277.0 277.5	3.822 3.837 3.851	296.5 297.0	4.424 4.439 4.455	317.0	5.076 5.093 5.110
253.0 258.5	3.309 3.323	278.0 278.5	3.866 3.880	298.0 298.5	4.471	318.0 318.5	5.127 5.144
259.0 259.5	3.336 3.349	279.0 279.5	3.895 3.909		4.502 4.518		5.161 5.178

•

V <sub>c</sub>	9 <sub>c</sub>	V <sub>c</sub>	9 <sub>c</sub>	Ì	V <sub>c</sub>	9 <sub>c</sub>	V <sub>c</sub>	9c
(Knots)	(" <u>Hg</u> )	(Knots)	( <u>" H̃g)</u>		(Knots)	<u>(* H̃g)</u>	(Knots)	(" Hg)
320.0	5.195	340.0	5.909		360.0	6.677	380.0	7.501
320.5	5.212	340.5 341.0	5.927 5.946		360.5	6.697 6.717	380.5	7.523
321.5	5.247	341.5	5.964		361.5	6.737	381.0 381.5	7.544
322.0	5.264	342.0	5.983		362.0	6.757	382.0	7.587
322.5	5.281	342.5	6.002		362.5	6.777	382.5	7.608
323.0	5.299	343.0	6.020	1	363.0	6.797	383.0	7.630
323.5 324.0	5,316 5,334	343.5 344.0	6.039 6.058		363.5	6.817	383.5	7.652
324.5	5.351	344.5	6.077		364.5	6.837 6.857	384.0 384.5	7.673
325.0	5.369	345.0	6.095		365.0	6.877	385.0	7.717
325.5	5.386	345.5	6.114		365.5	6.898	385.5	7.739
326.0	5.404	346.0	6.133		366.0	6.918	386.0	7.760
326.5	5.421	346.5	6.152		366.5	6.938	386.5	7.782
327.0	5.439 5.456	347.0 347.5	6.171 6.190		367.0 367.5	6.959 6.979	387.0 387.5	7.804
328.0	5.474	348.0	6.209		368.0	7.000	388.0	7,826 7,848
328.5	5.492	348.5	6.228		368.5	7.020	388.5	7.869
329.0	5.510	349.0	6.247		369.0	7.041	389.0	7.891
329.5	5 527	349.5	6.267		369.5	7.061	389.5	7.913
330.0	5.545	350.0	6.286		370.0	7.082	390.0	7.936
330.5	5,563	350.5	6.305		370.5	7.102	390.5	7.958
331.5	5.599	351.0	6.324 6.344		371.0 371.5	7.123	391.0 391.5	7.980 8.002
332.0	5.617	352.0	6.363		372.0	7.165	392.0	8.024
332.5	5,635	352.5	6.382		372.5	7.185	392.5	8.046
333.0	5.653	353.0	6.402		373.0	7.206	393.0	8.069
333.5	5.671 5.689	353.5 354.0	6.421		373.5	7.227	393.5	8.091
334.5	5.707	354.5	6.440 6.460		374.0 374.5	7,248 7,269	394.0 394.5	8.113 8.136
335.0	5.725	355.0	6.479		375.0	7.290	395.0	8.158
335.5	5.743	355.5	6.499		375.5	7.311	395.5	8.181
336.0	5.762	356.0	6.519		376.0	7.332	396.0	8.203
336.5	5.780	356.5	6.538		376.5	7.353	396.5	8.226
337.0 337.5	5.798 5.817	357.0 357.5	6.558 6.578		377.0 377.5	7.374	397.0 307.5	8.248
338.0	5.835	358.0	6.597		378.0	7.416	397.5 398.0	8,271 8,294
338.5	5.853	358.5	6.617		378.5	7.437	398.5	8.316
339.0	5.872	359.0	6.637		379.0	7.459	399.0	8.339
339.5	5.890	359.5	6.657		379.5	7.480	399.5	8.362

V <sub>c</sub> (Knots)	q <sub>c</sub> (" Hg)	V <sub>c</sub> (Knots)	<sup>q</sup> c (■ Hg)		V <sub>c</sub> Knots)	q <sub>c</sub> ("Hg)	Ī	V <sub>c</sub> (Knots)	9 <sub>c</sub> (■ Hg)
400.0	8.385	42.0.0	9.330		40.0	10.34		460.0 460.5	11.41 11.44
400.5	8.408 8.431	420.5 421.0	9.378	4	41.0	10.39		461.0	11.47
401.5	8.453 8.476	421.5 422.0	9.403 9.427		441.5 442.0	10.42		461.5 462.0	11.50 11.52
402.0	8.499	422.5	9.452	4	<b>4</b> 42.5	10.47		462.5	1.1.55
403.0	8.523	423.0 423.5	9.477		<b>443.0</b> 443.5	10.50 10.52		463.0 463.5	11.58 11.61
403.5	8.546	424.0	9.526		444.0	10.55		464.0	11.64
404.5	8.592	424.5	9.551	- H	444.5	10.57		464.5	1.1,66
405.0	8.615	425.0 425.5	9.576 9.600		445.0 445.5	10.60		465.0 465.5	11.69
405.5	8.638	426.0	9.625		446.0	10.65		466.0	11.75
406.5	8.685	426.5	9.650		446.5	10.68		466.5	11.78 11.81
407.0	8.708	427.0 427.5	9.675 9.700		447.0 447.5	10.71		467.5	11.81
407.5	8.755	428.0	9.725		448.0	10.76		468.0	11.86
408.5	8.779	428.5	9.750		448.5	10.79		468.5	11.89
<b>40</b> 9.0 <b>409.</b> 5	8.802	429.0 429.5	9.775 9.800		449.0 449.5	10.81		469.0	11.92 11.95
410.0	8.849	430.0	9.826		450.0	10.87		470.0	11.98
410.5	8.873	430.5	9.851		450.5	10.90		470.5	12.01
411.0	8.897	431.0	9.876		451,0 451,5	10.92		471.0	12.03
<b>411.5</b> <b>412.0</b>	8.920 8.944	431.5 432.0	9.901	, ,	452.0	10.93		472.0	12.09
412.5	8.968	432.5	9.952		452,5	11.00		472.5	12.12
413.0	8.992	433.0	9.978		453.0	11.03		473.0	12.15
413.5	9.016 9.040	433.5 434.0	10,00		453.5 454.0	11.06		474.0	12.21
414.0	9.040	434.5	10.05		454.5	11.11		474.5	12.24
415.0	9.087	435.0	10.08		455.0	11.14		475.0	12.27
415.5	9.112	435.5	10.11		455.5 456.0	11.17		475.5	12.29
416.0 416.5	9,136		10.13		456.5	11.22	•	476.5	12.35
417.0	9.184	437.0	10.18		457.0	11.25	]	477.0	12.38
417.5	9.208		10.21		457.5 458.0	11.28		477.5	12.41
418.0	9.232		10.23	11	458.5	11.30		478.5	12.47
419.0	9.281	439.0	10.29		459.0	11.36	Í	479.0	12.50
419.5	9.305		10.31	JI	459.5	11.39	J	479.5	12,53

. . . . .

V <sub>c</sub>	q <sub>c</sub>		q <sub>c</sub>	V <sub>c</sub>	q <sub>c</sub>	V <sub>c</sub>	q <sub>c</sub>
(Knots)	( <u>" Hg</u> )	(Knots)	( <u>" Hg</u> )	$(\underline{Knots})$	( <u>" Hg</u> )	( <u>Knots</u> )	(" Hg)
480.0	12.56	500.0	13.78	520.0	15.07	540.0	16.44
480.5	12.59	500.5	13.81	520,5	15.10	540.5	16.48
481.0	12.62	501.0	13.84	521.0	15.14	541.0	16.51
481.5	12.65	501.5	13.87	521.5	15.17	541.5	16.55
482.0	12.68	502.0	13.90	522.0	15.20	542.0	16.58
482.5	12.71	502.5	13.93	522.5	15.24	542.5	16.62
483.0	12.74	503.0 503.5	13,96	523.0 523.5	15.27 15.30	543.0	16.65
483,5 484.0	12.77	503.5	14.00 14.03	524.0	15.30	544.0	16.69 16.73
484.5	12.80	504.5	14.05	524.5	15.34	544.5	16.76
						· · · · · · · · · · · · · · · · · · ·	
485.0	12.86	505.0	14.10	525.0	15.40	545,0	16.80
485.5	12.89	505.5	14.12	525.5	15.44	545.5	16.83
486.0	12.92	506.0	14.16	526.0	15.47	546.0	16.87
486.5	12.95	506.5	14.19	526.5	15,51	546.5	16.90
487.0	12.98	507.0	14.22	527.0	15.54	547.0	16.94
487.5	13.01	507.5	14.25	527.5	15.57	547.5	16.98
488.0	13.04	508.0	14.28	528.0	15.61	548.0	17.01
488.5	13.07	508.5	14.32	528.5	15.64	548.5	17.05
489.0	13.10	509.0	14.35	529.0 529.5	15.68	549.0	17.09
489.5	13.13	509.5	14.38		15.71	549.5	17.12
490.0	13.16	510.0	14.41	530.0	15.74	550.0	17.16
490.5	13.19	510.5	14.44	530.5	15.78	550.5	17.19
491.0	13.22	511.0	14.48	531.0	15.81	551.0	17.23
491.5	13.25	511.5	14.51	531.5	15,85	551.5	17.27
492.0	13.28	512.0	14.54	532.0	15.88	552.0	17.30
492.5	13.31	512.5	14.57	532.5	15.92	552.5	17.34
493.0	13.34	513.0	14.61	533.0	15.95	553.0	17.38
493.5 494.0	13.37	513.5 514.0	14.64 14.67	533.5 534.0	15.99 16.02	553.5 554.0	17.41
494.0	13.40	514.0	14.07	534.5	16.02	554.0	17.45
495.0	13.46	515.0	14.74	535.0	16.09	555.0	17.52
495.5	13.49	515.5	14.77	535.5	16.13		17.56
496.0	13.53	516.0	14.80	536.0	16.16		17.60
496.5	13.56	516.5	14.84	536.5	16.19	556.5	17.64
497.0	13.59	517.0	14.87	537.0	16.23	557.0	17.67
497.5	13.62	517.5	14.90	537.5	16.26	557.5	17.71
498.0	13.65	518.0	14.94	538.0	16,30	558,0	17.75
498,5	13.68	518,5	14.97	538.5 539.0	16,33 16,37	558.5 559.0	17.78
499.0	13.71	<b>5</b> 19,0 519,5	15.00		16, 37	559.0	17.82
377.5	12.14	517.5	15.04	557.5	10,41	557.5	11.00

an an chuir an ta

			·			V	
Vc	9c	V <sub>c</sub> (Knots)	9c ("Hg)	V <sub>C</sub> (Knots)	(" Hg)	V <sub>C</sub> (Knots)	(" Hg)
(Knots)	( <b>*</b> Hg)					620.0	22.80
560.0	17.90	580.0	19.44	600.0 600.5	21.07 21.12	620.5	22.85
560.5	17.93	580,5 581.0	19.48 19.52	601.0	21.16	621.0	22.89
561.0	17.97 18.01	581.5	19.56	601.5	21.20	621.5	22.94
561.5	18.01	582.0	19.60	602.0	21.24	622.0	22.98
562.5	18.09	582.5	19.64	602.5	21.29	622.5	23.03
563.0	18.12	583.0	19.68	603.0	21.33	623.0	23.07
563.5	18.16	583.5	19.72	603.5	21.37	623.5 62 <b>4</b> .0	23.16
564.0	18.20	584.0	19.76	604.0 604.5	21.41	624.5	23.21
564.5	18.24	584.5	19.80				23.25
565.0	18.27	585.0	19.84	605.0	21.50	625.0	23.30
565.5	18.31	585.5	19.88	605.5	21.54	626.0	23.34
566.0	18.35	586.0	19.92 19.96	606.0 606.5	21.63	626.5	23.39
566.5	18.39	586.5 587.0	20.00	607.0	21.67	627.0	23.43
567.0	18.43	587.5	20.04	607.5	21.71	627.5	23.48
563.0	18.50	588.0	20.08	608.0	21,75	628.0	23.52
563.5	18.54	588.5	20.12	608.5	21,80	628.5	23.57
569.0	18.58	589.0	20.16	609.0	21.84	629.0	23.61
569.5	18.62	589.5	20.20	609.5	21.88		4
570.0	18.66	590.0	20, 25	610.0	21.93	630.0	23.71
570.5	18.70	590.5	20.29	610.5	21.97	630.5 631.0	23.75
571.0	18.73	591.0	20.33	611.0	22.01 22.06	631.5	23.84
571.5	18,77	591.5	20.37	611.5 612.0	22.10	632.0	23.89
572.0	18.81	592.0	20.41	612.5	22.14	632.5	23,94
572.5	18.85 18.89	593.0	20.49	613.0	22.19	633.0	23.98
573.0	18.93	593.5	20.53	613.5	22.23	633.5	24.03
574.0	18.97	594.0	20.57	614.0	22.27	634.0	24.07
574.5	19.01	594.5	20.62	614.5	22.32	634.5	24.12
575.0	19.05	595.0	20.66	615.0	22,36	635.0	24,17
575.5	19.09	595.5	20.70	615.5	22.41	635.5	24.21
576.0	19.12	596.0	20.74	1 1 4 4 4 4	22.45	636.0	24.26 24.31
576.5	19.16	596.5	20.78		22.49		24.35
577.0	19.20	597.0			22.54	637.5	24.40
577.5	19.24	597.5	1	618,0	22.63	638.0	24.45
578.0	19.28	598.5	· -		22.67	638.5	24.49
578.5	19.36	599.0		619.0	22.71	639.0	24.54
579.5	19.40	599.5			22.76	639.5	24.59
579.5	19.40	599.5	1 21.05			·	

Vc	č.c	V <sub>c</sub>	9c	V <sub>c</sub>	9 <sub>c</sub>	V <sub>c</sub>	<sup>q</sup> c
(Knots)	<u>(" Hg</u> )	(Knots)	( <u>" Hg</u> )	(Knots)	( <u>" Hg</u> )	(Knots)	(" Ĥg)
640.0	24.63	660.0	26.57	680.0	28.61	700.0	30,76
640.5	24,68	660.5	26.62		28.67	700.5	30.82
641.0	24.73	661.0	26.67		28.72	701.0	30.87
641.5	24.78	661.5	26.72		28.77	701.5	30.93
642.0	24.82	662.0	26.77		28,82	702.0	30.98
642.5	24.87	662.5	26.82	682.5	28,88	702.5	31.04
643.0	24.92	663.0	26.87		28.93	703.0	31.09
643.5	24.97	663.5	26.92	683.5	28,98	703,5	31,15
644.0	25.01	664.0	26.97		29.04	704.0	31.20
644.5	25.06	664.5	27.02	684,5	29.09	704.5	31.26
645.0	25.11	665.0	27.07	685.0	29.14	705.0	31.32
645.5	25.16	665.5	27.12	685.5	29.19	705.5	31.37
646.0	25.20	666.0	27.17	686.0	29,25	706.0	31,43
646.5	25.25	666.5	27.22		29.30	706.5	31.48
647.0	25.30	667.0	27.27		29.35	707.0	31.54
647.5	25.35	667.5	27.32		29.41	707.5	31.59
648.0	25.40	668.0	27.37		29.46	708.0	31,65
648.5	25.44	668.5	27.43		29.51	708.5	31.71
649.0	25.49	669.0	27.48		29.57	709.0	31.76
649.5	25.54	669.5	27.53	689.5	29.62	709.5	31.82
650.0	25.59	670.0	27.58	690.0	29.68	710.0	31.88
650.5	25.64	670.5	27.63	690.5	29.73	710.5	31,93
651.0	25.69	671.0	27.68	691.0	29.78	711.0	31.99
651.5	25.73	671.5	27.73	691.5	29.84	711.5	32.04
652.0	25.78	672.0	27.78	692.0	29.89	712.0	32.10
652.5	25.83	672.5	27.83	692.5	29.95	712.5	32.16
653.0	25.88	673.0	27.89	693.0	30.00	713.0	32.21
653.5	25.93	673.5	27.94		30,05	713.5	32.27
654.0 654.5	25.98	674.0 674.5	27.99	694.0 694.5	30.11	714.0	32.33
	26.03		28.04		30.16	714.5	32.38
655.0	26.08	675.0	28.09		30.22	715.0	32.44
655.5	26,12	675.5	28,14		30.27	715.5	32.50
656.0	26.17	676.0	28.20		30.33	716.0	32.55
656.5	26.22	676.5	28.25	696.5	30.38	716.5	32.61
657.0	26.27	677.0	28.30	697.0	30.43	717.0	32.67
657.5	26.32	677.5	28.35	697,5	30.49	717,5	32.73
658.0	26.37	678.0	28.40	698.0	30.54	718.0	32.78
658.5	26.42	678.5	28.46	698.5	30.60	718.5	32.84
659.0 659.5	26.52	679.0 679.5	28.51	699.0	30.65	719.0	32.90
0.57.5	20, 52	017.5	28.56	699.5	30.71	719.5	32,95

Vc	0-	V <sub>c</sub>	٩c	V <sub>c</sub>	9c	V <sub>c</sub>	9c
(Knots)	<sup>Q</sup> c (™ Hg)	(Knots)	( <u>* H̃g</u> )	(Knots)	( <u>* Hg</u> )	(Knots)	(* Hg)
720.0	33.01	740.0	35.36	760.0	37.80	780.0	40.32
720.5	33.07	740.5	35.42	760.5	37.86	781.0	40.45
721.0	33.13 33.19	741.0 741.5	35.48 35.54	761.5	37.98	781.5	40.52
721.5	33.24	742.0	35.60	762.0	38.04	782.0	40.58
722.5	33.30	742.5	35.66	762.5	38.11 38.17	782.5	40.64
723.0	33.36	743.0	35.72 35.78	763.0	38.23	783.5	40.77
723.5	33.42 33.47	743.5 744.0	35.84	764.0	38.29	784.0	40.84
724.0	33.53	744.5	35.90	764.5	38.36	784.5	40.90
725.0	33.59	745.0	35.96	765.0	38.42	785.0	40.97
725.5	33.65	745.5	36.02	765.5	38.48 38.54	785.5 786.0	41.03
726.0	33.71	746.0	36.08 36.14	766.0 766.5	38,61	786.5	41.16
726.5	33.76 33.82	746.5 747.0	36.20	767.0	38,67	787.0	41.23
727.0	33.88	747.5	36.26	767.5	38.73	787.5	41.29
728.0	33.94	748.0	36.32	768.0	38.80 38.86	788.0 788.5	41.36 41.42
728.5	34.00	748.5	36.38 36.44	768.5	38.92	789.0	41.49
729.0	34.06	749.0	36.50	769.5	38.98	789.5	41.55
729.5	+	750.0	36.56	770.0	39.05	790.0	41.62
730.0	34.17	750.5	36.63	770.5	39.11	790.5	41.68
731.0	34.29	751.0	36.69	771.0	39.17	791.0	41.75 41.81
731.5	34.35	751.5	36.75	771.5	39.24 39.30	792.0	41.88
732.0	34.41 34.47	752.0	36.81		39.36	792.5	41.95
732.5	34.53	753.0	36.93	773.0	39.43	793.0	42.01
733.5	34.59	753.5	36.99		39.49 39.55	793.5 794.0	42.08
734.0	34.64	754.0	37.05		39.62	794.5	42.21
734.5	34.70	754.5			39.68	795.0	42.27
735.0	34.76	755.0	37.18	1 1	39.75	795.5	42.34
735.5	34.82 34.88	756.0		776.0	39.81	796.0	42.41
736.5	34.94	756.5	37.36	776.5	39.87	796.5	42.47
737.0	35.00	757.0	37.42		39.94	797.5	42.60
737.5	35.06	757.5	37.49		40.07	798.0	42.67
738.0	35.12	758.5	37.61	778.5	40.13	798.5	42.74
739.0	35.24	759.0	37.67		40.19 40.26	799.0	42.80
739.5	35.30	759.5	37.73	779.5	1 10.20		

V <sub>c</sub>	9c	V <sub>c</sub>	9c	V <sub>c</sub>	9 <sub>c</sub>	V <sub>c</sub>	9c
(Knots)	(Hg)	(Knots)	( <u>Hg</u> )	(Knots)	(" Hg)	(Knots)	(* Hg)
800.0	42.94	820.0	45.64	840.0	48.42	860.0	51.28
800.5	43.00	820.5	45.70	840.5	48.49	860.5	51.36
801.0	43.07	821.0 821.5	45.77 45.84	841.0 841.5	48.56 48.63	861.0 861.5	51,43 51,50
801.5	<b>43.14</b> <b>43.20</b>	822.0	45.91	842.0	48.70	862.0	51.57
802.5	43.27	822.5	45.98	842.5	48.77	862.5	51.65
803.0	43.34	823.0	46.05	843.0	48.84	863.0	51.72
803.5	43.40	823.5	46.12	843.5	48.91	863.5	51.79
<b>804.</b> 0 804,5	43.47 43.54	824.0 824.5	46.19	844.0 844.5	<b>48.99</b> <b>49.06</b>	864.0 864.5	51.87 51.94
805.0	<b>43.60</b> <b>43.67</b>	825.0 825.5	<b>4</b> 6.32 <b>4</b> 6.39	845.0 845.5	49.13 49.20	865.0 865.5	52.01 52.09
805.5 806.0	43.74	826.0	46.46	846.0	49.27	866.0	52.16
806.5	43.80	826.5	46,53	846.5	49.34	866.5	52.23
807.0	43.87	827.0	46.60	847.0	49.41	867.0	52.30
807.5	43.94	827.5	46.67	847.5 848.0	49.48 49.55	867.5 868.0	52, 38 52, 45
808.0 808.5	<b>44.01</b> <b>44.07</b>	828.0 828.5	46.74 46.81	848.5	49.63	868.5	52.53
809.0	44.14	829.0	46.88	849.0	49.70	869.0	52.60
809.5	44.21	829.5	46.95	849.5	49.77	869.5	52.67
810.0	44.28	830.0	47.02	850.0	49.84	870.0	52,75
810.5	44.34	830.5	47.09	850.5	49.91	870.5	52.82
811.0	44.41	831.0	47.16	851.0	49.98	871.0	52.89
<b>811.5</b> 812.0	<b>44.4</b> 8 <b>44.</b> 55	831.5 832.0	47.23 47.30	851.5 852.0	50.06	871.5 872.0	52.97 53.04
812.5	44.61	832.5	47.37	852.5	50.20	872.5	53.11
813.0	44.68	833.0	47.44	853.0	50.27	873.0	53,19
813.5	44.75	833.5	47.51	853.5	50.34	873.5	53.26
814.0	44.82	834.0	47.58	854.0	50.42	874.0 874.5	53.34 53.41
814.5	44.89	834.5	47.65	854.5	50.49		
815,0	44.95	835.0	47.72	855.0	50.56	875.0 875.5	53,48 53,56
815.5 816.0	45.02 45.09	835.5 836.0	47.79 47.86	855.5 856.0	50.63 50.70	876.0	53.63
816.5	45.16	836.5	47.93	856.5	50.78	876.5	53.71
817.0	45.23	837.0	48,00	857.0	50.85	877.0	53.78
817,5	45.29	837.5	48.07	857.5	50.92	877.5	53.86
818.0	45.36	838.0	48.14 48.21	858.0 858.5	50.99 51.07	878.0 878.5	53.93 54.01
818.5 819.0	45.43	838.5 839.0	48.28	859.0	51.14	879.0	54.08
819.5	45.57	839.5	48.35	859.5	51.21	879.5	54.15

Vc	Чс	Vc	9c	ſ	V <sub>c</sub>	9 <sub>c</sub>		Vc	<sup>q</sup> c
(Knots)	( <u>" Hg</u> )	(Knots)	( <u>" Hg</u> )	Ľ	( <u>Knots</u> )	( <u>" Hg</u> )		(Knots)	( <u>Hg</u> )
880.0	54.23	900.0	57.25	1	920.0	60.36		940.0	63.54
880.5	54.30		57.33 57.41		920.5 921.0	60.44 60.51		940.5 941.0	63,62 63,70
831.0 831.5	54.38 54.45		57.41	1	921.5	60.59		941.5	63.78
832.0	54.53		57.56		922.0	60.67		942.0	63,86
882.5	54,60		57.64		922.5	60.75		942.5	63.94
883.0	54.68		57.71		923.0	60,83		943.0	64.02
883.5	54.75		57.79		923.5	60.91 60.99		943.5	$64.10 \\ 64.18$
884.0 884.5	54.83 54.90	1 1	57.87 57.95		924.0 924.5	61.07		9 <b>44.</b> 0 944.5	64.27
						<b>├───</b> ─		945.0	64,35
885.0 885.5	54.98 55.05	905.0 905.5	58.02 58.10		925.0 925.5	61.15 61.22		945.5	64.43
886.0	55.13		58,18		926.0	61.30		946.0	64,51
886.5	55,20		58,25		926.5	61.38		946.5	64.59
887.0	55.28		58,33		927.0	61,46		947.0	64.67
837.5	55.35		58.41 58.49		927.5 928.0	61.54 61.62		947.5 948.0	64.75 64.83
858.0 888.5	55.43 55.50	908.0 908.5	58.56		928.5	61.70		948,5	64,91
889.0	55.58		58.64		929.0	61.78		949.0	65,00
889.5	55.66	( (	58.72		929.5	61.86		949.5	65.08
890.0	55,73	910.0	58.80		930,0	61.94		950.0	65.16
890.5	55.81		58.87		930.5	62.02		950.5	65.24
891.0	55.88		58.95		931.0	62.10		951.0	65.32 65.40
891.5 892.0	55.96		59.03 59.11		931.5 932.0	62.18 62.26	ł	951.5 952.0	65.48
892.5	56.11		59,18	{ {	932.5	62.34	ł	952.5	65.57
893.0	56.19	1 1 .	59.26		933.0	62.42	]	953.0	65.65
893.5	56,26		59.34		933.5	62,50		953.5	65.73
894.0	56.34		59.42		934.0	62.58	{	95 <b>4.</b> 0 954.5	65.81 65.89
894.5	56.41	914.5	59.50		934.5	62.66	ł		
895.0	56.49	1 1	59.57		935.0 935.5	62,74 62,82		955.0 955 <b>.</b> 5	65.98 66.06
895.5	56.57		59,65		936.0	62.90	Į –	956.0	66,14
896.5	56.72	1 1	59.81		936.5	62.98		956.5	66.22
897.0	56.79	917.0	59.89		937.0	63.06		957.0	66.30
897.5	56.87		59.97		937.5	63.14		957.5	66.39
898.0	56.95		60.04		938.0	63.22 63.30		958.0 958.5	66.47 66.55
890.5	57.10	1 1	60.20		939.0	63.38		959.0	66.63
899.5	57.18		60.23		939.5	63.46		959.5	66.72

V <sub>c</sub>	9c
(Knots)	(* Hg)
960.0 960.5 961.0 961.5 962.0 962.5 963.0 963.5 964.0	66.80 66.96 67.04 67.13 67.21 67.30 67.38 67.46
964.5	67.54
965.0	67.62
965.5	67.71
966.0	67.79
966.3	67.87
967.0	67.96
967.5	68.04
968.0	68.12
968.5	68.21
968.0	68.29
968.5	68.37

V <sub>c</sub>	<sup>q</sup> c
( <u>Knots)</u>	(■ Hg)
970.0	68.46
970.5	68.54
971.0	68.62
971.5	68.71
972.0	68.79
972.5	68.87
973.0	68.96
973.5	69.04
974.0	69.12
974.5	69.21
975.0 975.5 976.0 976.5 977.0 977.5 977.0 977.5 978.0 978.5 979.0 979.5	69.29 69.38 69.46 69.54 69.63 69.71 69.80 69.88 69.96 70.05

	-
V <sub>c</sub> q <sub>c</sub> (Knots) ("Hg	2
980.0 70.13 980.5 70.22	
981.0 70.30 981.5 70.39	
982.0 70.47 982.5 70.55	7
983.0 70.64 983.5 70.72	۱Ì.
984.0 70.81 984.5 70.81	
985.0 70.98	3
985.5 71.0 986.0 71.1	5
986.5 71.2 987.0 71.3	2
987.5 71.4 988.0 71.4	9
	71
988.5 71.5 989.0 71.6 989.5 71.7	6

v <sub>c</sub>	9 <sub>C</sub>
(Knots)	(* Hg)
990.0	71.83
990.5	71.91
991.0	72.00
991.5	72.08
992.0	72.17
992.5	72.26
993.0	72.34
993.5 99 <b>4</b> .0	72.43
<b>994.5</b>	72.60
995.0	72.68
995.5	72.77
996.0	72.86
997.0	73.03
997.5	73.11
998.0	73.20
998.5	73.29
999.0	73.37
999.5	73.46
1000.0	73,04

Conversion Formulae - Fahrenheit, Centigrade and Rankine

Fahrenheit to Centigrade

$$c = \frac{5}{9}$$
 (F-32)

Fahrenheit to Rankine

$$R = F + 459.7$$

Cenuigrade to Fahrenheit

 $F = \frac{9}{5}C + 32$ 

Centigrade to Rankine

$$R = \frac{9}{5}C + 491.7$$

Rankine to Fahrenheit

$$F = R - 459.7$$

Rankine to Centigrade

$$C = \frac{5}{9}$$
 (R-491.7)

CONVERSION TABLE

•C		۰F	°C !		۰F	•c	[	۰F
-169.4	-273	-459.4	-138.9	-218	-360.4	-108.3	-163	-261.4
-168.9	-272	-457.6	-138.3	-217	-358.6	-107.8	-162	-259.6
-168.3	-271	-455.8	-137.8	-216	-356.8	-107.2	-161	-257.8
-167.8	-270	-454.0	-137.2	-215	-355.0	-106.7	-160	-256.0
-167.2	<b>-26</b> 9	-452.2	-136.7	-214	-353.2	-106.1	-159	-254.2
-166.7	-268	-450-1	-136.1	-213	-351.4	-105.6	-158	-252.4
-166.1	-267	- <u>44</u> 8.6	-135.6	-212	-349.6	-105.0	-157	-250,6
-165.6	-266	-446.8	-135.0	-211	-347.8	-104+4	-156	-248.8
-165.0	-265	-445.0	-134-4	-210	-346.0	-103.9	-155	-247.0
-164.4	-264	-443.2	-133.9	-209	-344.2	-103.3	-154	-24512
-163.9	-263	-441-4	-133.3	-208	-342-4	-102.8	-153	-243.4
-163.3	-262	-439.6	-132.8	-207	-340.6	-102.2	-152	-241.5
-162.8	-261	-437.8	-132.2	-206	-338.8	-101.7	-151	-239.8
-162.2 -161.7	-260	-436.0	-131.7	-205	-337.0	-101.1	-150	-238,0
-161.1	-259 -258 <sup>-</sup>	-434•2 -432•4	-131.1 -130.6	-204 -203	-335.2	-100.6	-149	-236.2
-160.6	-257	-432.4	-130.0	-202	-333-4	-100.0	-148	-234.4
-160.0	-256	-430.8	-129-4	-202	-331.6 -329.8	-99-4	-147	-232.6
-159-4	-255	-427.0	-128.9	-200	-328.0	-98.9	-146	-230.8
-158.9	-254	-425.2	-128.3	-199	-326.2	-98.3 -97.8	-=145 -144	-229.0 -227.2
-158.3	-253	-423.4	-127.8	-198	-324.4	-97.0	-143	-225.ù
-157.8	-252	-421.6	-127.2	-197	-322.6	-96.7	-142	-223.6
-157.2	-251	-419.8	-126.7	-196	-320.8	-96.1	-141	-221.8
-156.7	-250	-418.0	-126.1	-195	-319.0	-95.6	-140	-220.0
-156.1	-249	-416.2	-125.6	-194	-317.2	-95.0	-139	-21.8.2
-155.6	-248	-414.4	-125.0	-193	-315-4	-94.4	-138	-216.4
-155.0	-247	-412.0	-124-4	-192	-313.6	-93.9	-137	-214.6
-154-4	-246	-410.8	-123.9	-191	-311.8	-93.3	-136	-212.8
-153.9	-245	-409.0	-123.3	-190	-310.0	-92.8	-135	-211.0
-153.3	-244	-407.02	-122.8	-189	-308.2	-92.2	-134	-209.2
-152.8	-243	-405.4	-122.2	-188	-306.4	-91.7	-133	-207.4
-152.2	-242	-403.6	-121.7	-187	-304.6	-91.1	-132	-205.6
-151.7 -151.1	-241 -240	-401.8	-121.1 -120.6	-186 -185	-302.8	-90.6	-131	-203.8
-150.6	-239	-400.0 -398.2	-120.0	-184	-301.0	-90.0	-130	-202.0
-150.0	-238	-396.4	-119.4	-183	-299•2 -297•4	-89.4	-129	-200.2
-149-4	-237	-394.6	-118.9	-182	-295.6	-88.9 -88.3	-128	-198.4
-148.9	-236	-392.8	-118.3	-181	-293.8	-87.8	-127 -126	-196.6 -194.8
-148.3	-235	-391.0	-117.8	-180	-292.0	-87.2	-125	-193.0
-147.8	-234	-389.2	-117.2	-179	-290.2	-86.7	-124	-191.2
-147-2	-233	-387•4	-116.7	-178	-288-4	-86,1	-123	-189.4
-146.7	-232	-385.6	-116.1	-177	-286.6	-85.6	-122	-187.6
-146.1	-231	-383.8	-115.6	-176	-284.8	-85.0	-121	-185.8
-145.6	-230	-382.0	-115.0	-175	-283.0	-84.4		-184.0
-145.0	-229	-380.2	-114.4	-174	-281.2	-83.9		-182.2
-144.4	-228	-378-4	-113.9	-173	-279-4	-83.3		-180.4
-143.9	-227	-376.6	-113.3	-172	-277.6	-82.8		-178.6
-143.3 -142.8	-226	-374.8	-112.8	-171	-275.8	-82.2	1	-176.8
-142.0	-225 -224	-373.0	-112.2	-170	-274.0	-81.7		-175.0
-142.2	-223	-371.2 -369.4	-111.7 -111.1	<b>-169</b> -168	-272.2	-81.1		-173.2
-141.1	-222	-367.6	-111.1	-167	-270.4	-80.6		-1.71.4
-140.6	-222	-365.8	-110.0	<b>-16</b> 6	-268.6 -266.8	-80.0		-169.6
-140.0	-220	-364.0	-109.4	-165	-265.0	-79.4		-167.8
-139.4	-219	-362,2	-108.9	-164	-263.2	-78.9		-166.0
L		20040	-10007		-20)16	-78.3	L	-164.2

TABLE 9.7

CONVERSION TABLE

$\circ$ C $\circ$ F $\circ$ C $\circ$ F $\circ$ C $-77.8$ $-108$ $-162.4$ $-47.2$ $-53$ $-63.4$ $-16.7$ $-77.2$ $-107$ $-160.6$ $-46.7$ $-52$ $-61.6$ $-16.1$ $-76.7$ $-106$ $-158.8$ $16.1$ $-51$ $-59.8$ $-15.6$ $-76.1$ $-105$ $-157.0$ $-45.6$ $-50$ $-58.0$ $-15.0$ $-75.6$ $-104$ $-155.2$ $-45.0$ $-49$ $-56.2$ $-14.4$ $-77.0$ $-103$ $-153.4$ $-418$ $-54.4$ $-13.9$ $-77.4$ $-102$ $-151.6$ $-43.9$ $-47$ $-52.6$ $-13.3$	2 3 4 5 6 7 8 9	•F 35.6 37.4 39.2 41.0 42.8 44.6
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	3456 78	37•Ц 39•2 Ц1•0 Ц2•8 ЦЦ•6
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		39-2 41-0 42-8 44-6
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		0. 10 لو2 لو1 لو1
-75.6 $-104$ $-155.2$ $-45.0$ $-49$ $-56.2$ $-14.4-75.0$ $-103$ $-153.4$ $-48$ $-54.4$ $-13.9$		<u>іц.</u> 6
+75.0 $-103$ $-153.4$ $-141.4$ $-48$ $-54.4$ $-13.9-77.4$ $-102$ $-151.6$ $-13.9$ $-47$ $-52.6$ $-13.3$		بلبه ا
$  -77 \cdot \mu   -102   -151 \cdot 6     -43 \cdot 9   -47   -52 \cdot 6     -13 \cdot 3  $		
	91	46.4
-73.9 $-101$ $-149.8$ $-43.3$ $-46$ $-50.8$ $-12.8$	- <b>1</b>	48.2
-73.3 $-100$ $-148.0$ $-42.8$ $-45$ $-49.0$ $-12.2-72.8$ $-99$ $-146.2$ $-42.2$ $-44$ $-47.2$ $-11.7$	10 11	50.0
	12	51.8 53.6
	13	55.4
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	14	57.2
-70.0 -95 -139.0 -40.0 -40 -40.0 -9.4	15	59.0
-70.0 -94 -137.2 -39.4 -39 -38.2 -8.9	16	60.8
-69.1 $-93$ $-135.1$ $-38.9$ $-38$ $-36.1$ $-8.3$	17	62.6
-68.9 -92 -133.6 -38.3 -37 -34.6 -7.8	18	64.4
-68.3 -91 -131.8 -37.8 -36 -32.8 -7.2	19	66.2
-67.8 -90 -130.0 -37.2 -35 -31.0 -6.7	20	68.0
-67.2 -89 -128.2 -36.7 -34 -29.2 -6.1	ย	69.8
-66.7 -88 -126.4 -36.1 -33 -27.4 -5.6	22	71.6
-66.1 $-87$ $-124.6$ $-35.6$ $-32$ $-25.6$ $-5.0$	23 21	73•4
-65.6 $-86$ $-122.8$ $-35.0$ $-31$ $-23.8$ $-4.4-65.0$ $-85$ $-121.0$ $-34.4$ $-30$ $-22.0$ $-3.9$	24 25	75.2
	26	77.₀0 78.₀8
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	27	80.6
-63.3 $-82$ $-115.6$ $-32.8$ $-27$ $-16.6$ $-2.2$	28	82.4
-62.8 -81 -113.8 -32.2 -26 -14.8 -1.7	29	84.2
-62.2 -80 -112.0 -31.7 -25 -13.0 -1.1	30	86.0
-61.7 -79 .110.2 -31.1 -24 -11.2 -0.6	31	87.8
-61.1 -78 -108.4 -30.6 -23 -9.4 0	32	89.6
-60.6 -77 -106.6 -30.0 -22 -7.6 0.6	33	91.4
-60.0 -76 -104.8 -29.4 -21 -5.8 1.1	34	93.2
-59.4 -75 -103.0 -28.9 -20 -4.0 1.7	35	95.0
-58.9 $-74$ $-101.2$ $-28.3$ $-19$ $-2.2$ $2.2$	36	96.8
-58.3 -73 -99.4 -27.8 -18 -0.4 2.8	37	98.6
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	0ر مد	100.4 102.2
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	38	104.0
-56.1 $-69$ $-92.2$ $-25.6$ $-14$ $6.8$ $5.0$	1.1	105.8
-55.6 -68 -90.4 -25.0 -13 8.6 5.6	12	107.6
-1 - 55 - 0 + -67 + -88 - 6 + 1 - 2 + 1 + -12 + 10 - 1 + 5 -	37 38 39 29 29 29 29 29 29 20 20 20 20 20 20 20 20 20 20 20 20 20	109.4
-54.44 $-500$ $-23.94$ $-11$ $12.62$ $6.7$	Ĩ.Ĩ.	111.2
	45	113.0
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	46	114.8
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	47	116.6
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	48	118.4
	49	120.2
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	50	122.0
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	51	123.8
	72 52	125.6
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	22 21	127.4
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	50 51 52 53 54 55	131.0
-47.8 -54 -65.2 -17.2 1 33.8 13.3	56	132.8

T. BLE 9.7

CONVERSION TABLE

.

.

.

•C		۰F		•C		۰F		•c		•F
13.9	57	134.6		43.9	111	231.8		73.9	165	329.0
14.4	58 59 60	136.4		44.4	112	233.6		74.4	166	330.8
15.0	59	138.2		45.0	113	235.4		75.0	167	332.6
15.6	60	140.0		45.6	114	237.2		75.6	168	334.4
16.1	61	141.8		46.1	115	239.0		76.1	169	336.2
16.7	62	143.6		46.7	116	240.8		76.7	170	338.0
17.2	63 64	145.4		47.2	זינ 7	242.6		77.2	171	339.8
17.8	64	147.2		47.8	118	244.4		77.8	172	341.6
18.3 18.9	65 66	149.0		48.3	119	246.2		78.3	173	343.4
19.4	67	150 <b>.8</b> 152 <b>.</b> 6		48.9 49.4	120	248.0		7849	174	345.2
20.0	68	154.4		50 <b>.</b> 0	121 122	249.8		79.4	175	347.0
20.6	69	156.2		50.6	123	251.6 253.4		80.0 80.6	176	348 <b>.8</b> 350 <b>.</b> 6
21.1	70	158.0		51.1	124	255.2		81.1	177 178	352.4
21.6	71	159.8		51.7	125	257.0		81.7	179	354.2
22.2	72	161.6		52.2	126	258.8		82.2	180	356.0
22,8	73	163.4		52.8	127	260.6		82.8	181	357.8
23.3	74	165.2		53.3	128	262.4		83.3	182	359.6
23.9	75	167.0		53.9	129	264.2		83.9	183	361.4
24.4	76	168.8		54.4	130	266.0		84.4	184	363.2
25.0	77	170.6		55.0	131	267.8		85.0	185	365.0
25.6	78	172.4		55.6	132	269.6		85.6	186	366.8
26.1	79	174.2		56.1	133	271.4		86.1	187	368.6
26.7	80	176.0		56.7	134	273.2		86.7	188	370.4
27.2	81 82	177.8		57.2	135	275.0		87.2	1.89	372.2
27.8 28.3	83	179.6		57.8	136	276.8		87.8	190	374-0
28,9	84	181.4		58 <b>.3</b> 58 <b>.</b> 9	137 138	278.6		88.3	191	375.8
29.4	85	183.2 185.0		59.4	139	280.4		88.9	192	377.6
30.0	86	186.8		ó0.0	140	282 <b>.</b> 2 284.0		89.4	193	379.4
30.6	87	188.6		60.6	14	285.8		90 <b>.</b> 0	194	381.2
31.1	88	190.4		61.1	142	287.6		90.6 91.1	195 196	383.0 384.8
31.7	89	192.2		61.7	143	289.4		91.•1 91.•7	190	386.6
32.2	90	194.0		62.2	144	291.2		92.2	198	388.4
32.8	91	195.8		62.8	145	293.0		92.8	199	390.2
33•3	92	197.6		63.3	146	294.0		93.3	200	392.0
33.9	93	199.4		63.9	147	296.6		93.9	201	393.8
34.4	94	201.2		64.4	148	298.4	Í,	94.4	202	395.6
35.0	95	203.0		65.0	149	300-2		95.0	203	397•4
35.6	96	204.8		65.6	150	302.0		95.6	204	399.2
36.1	97 98	206.5		66.1	151	303.8		96.1	205	401.0
36.7 37.2	90	208.4 210.2		<b>66.</b> 7 67 <b>.</b> 2	152	305.6		96.7	206	402.8
37.8	100	210.2	í	67.8	15 <b>3</b> 154	307.4		97.2	207	404.6
38.3	101	213.8		68.3	155	309.2		97.8	208	406.4
38.9	101	215.6		68,9	156	311.0 312.8		98.3	209	408.2
39.4	103	217.4		69.4	157	314.6		98 <b>.9</b>	210	110.0 111.8
40.0	104	21.9.2		70.0	158	316.4		99•4 100•0	211 2 <b>1</b> 2	щ1.0
40.6	105	221.0		70.6	Ĵ.59	318.2		100.0	212	415.4
山.1	106	222,8		71.1	160	320.0		101.1	214	417.2
41.7	107	224.6		71.7	161	321.8		101.7	215	119.0
42.2	108	226.4		72.2	162	323.6		102.2	216	420.8
42.8	109	228.2		72.8	163	325.4		102.8	217	422.6
43.3	110	230.0		73•3	164	327.2		103.3	218	424.4

2

CONVERSION TABLE

•с		•F.		•c		●F		•C		۰F
103.9	219	426.2		133.9	273	523.4		163.9	327	620.6
104.4	220	428.0		134.4	274	525.2		164.4	328	622.4
105.0	221	429.8		135.0	275	527.0		165.0	329	624.2
105.6	222	431.6		135.6	276	528.8		165-6	330	626.0
106.1	223	433.4		136.1	277	530.6		166.1	331	627.8
106.7	224	435.2		136.7	278	532+4		166.7	332	629.6
107.2	225	437.0		137.2	279	534.2	ľ	167.2	333	631.4
107.8	226	438.8		137.8	280	536.0		167.8	334	633.2
108.3	227	山0.6		138.3	281	537.8		168.3	335	635.0
108.9	228	442.4		138.9	282	539.6		168 <b>.9</b> 169 <b>.</b> 4	336	636.8
109.4	229	ևևև.2		139.4	283	541.4		170.0	337 338	638.6
110.0	230	146.0		140.0	284	543.2 545.0		170.6	339	640.4 642.2
110,6	231	447.8		140.6 141.1	285 286	545.0		171.1	340	644.0
111.1 111.7	232	1419.6 1451.14		141.7	287	548.6		171.7	34	645.8
112.2	233	451 •4		142.2	288	550.4		172.2	342	647.6
112.8	234 235	455.0		142.8	289	552.2		172.8	343	649.4
113.3	236	456.8		143.3	290	554.0		173.3	344	651.2
113.9	237	458.6		143.9	291	555.8		173.9	345	653.0
114.4	238	460.4		1,44.4	292	557.6		174.4	346	654.8
115.0	239	462.2		145.0	293	559.4		175.0	347	656.6
115.6	240	464.0		145.6	294	561.2		175.6	348	658.4
116.1	211	465.8		146.1	295	563.0		176.1	349	660.2
116.7	242	467.6		146.7	296	564.8		176.7	350	662.0
117.2	243	469-4		147.2	297	566.6		177.2	351	663.8
117.8	214	471.2		147.8	298	568.4		177.8	352	665.6
118.3	245	473.0		148.3	299	570.2		178.9	353 354	667•4 669•2
118.9	246	474.8 476.6		148.9	300	572.0		179-4	355	671.0
119.4	247 248	478.4		110.4 150.0	301 302	573.8		180.0	356	672.8
120.0	249	480.2		150.6	303	575.6 577.4		180.6	357	674.6
121.1	250	482.0		151.1	304	579.2		181.1	358	-676-4
121.7	251	483.8	1	151.7	305	581.0		181.7	359	678.2
122.2	252	485.6	[	152.2	306	582.8		182.2	360	680.0
122.8	253	487-4	[	152.8	307	584.6	I	182.8	361	681.8
123.3	254	489.2	ł	153.3	308	586.4	I	183.3	362	683.6
123.9	255	491.0	ł	153.9	309	588.2	Į	183.9	363	685.4
124.4	256	492.8	ł	154.4	310	590.0	ţ.	184.4	364	687.2
125.0	257	494.6	Į	155.0	311	591.8		185.0 185.6	365 366	689.0
125.6	258	496.4		155.6	312	593.6		186.1	367	690.8
126.1	259	498.2	1	156.1	313	595.4	ł	186.7	368	692.6
126.7	260	500.0	1	156.7	314	597-2	1	187.2	369	696.2
127.2	261 262	501.8	l	157.2 157.8	315	599.0	l	187.8	370	698.0
127.8	263	505-4	1	158.3	316 317	600.8	1	188.3	371	699.8
1.28.9	264	507.2	ł	158.9	31.8	602 <b>.</b> 6 604 <b>.</b> 4	1	188.9	372	701.6
129.4	265	509.0	ł	159-4	319	606.2	Į	189.4	373	703.4
130.0	266	510.8	1	160.0	320	608.0		190.0	374	705-2
130.6	267	512.6		160.6	321	609.8	1	190.6	375	707.0
131.1	268	514.4		161.1	322	611.6		191.1	376	708.8
131.7	269	516.2	ł	161.7	323	613.4		191.7	377	710.8
132.2	270	518.0	1	162.2	324	615.2		192.2	378	71.2.4
132,8	271	519.8	1	162.8	325	617.0		192.8 193.3	379 380	714.2
133.3	272	521.6	1	163.3	326	618.8	J	17303		716.0
CONVERSION TABLE

-C		•F	ſ	°C		F	ſ	•C		
										=
193.9	381	717.8		223.9	435	815.0		253.9	-489	912.2
194.4	382	719.6		224.4	436	816.8		254.4	490	914.0
195.0 195.6	383	721.4		225.0	437	818.6	1	255.0	491	915.8
195.0	384	723.2		225.6	438	820.4		255.6	492	917.6
196.7	385	725.0		226.1	439	822.2		256.1	493	919.4 921.2
197.2	386 387	726.8 728.6		226•7 227•2	цію ціл	824.0 825.8		256.7 257.2	494 495	923.0
197.8	388	730.4		227.8	1442	827.6		257.8	495 496	724.8
198.3	389	732.2		228.3	443	829.4		258.3	497	926.6
198.9	390	734.0		228.9		831.2		258.9	498	928.4
199.4	391	735.8		229.4	445	833.0		259.4	1,99	930.2
200.0	392	737.6		230.0	446	834.8		260.0	500	932.0
200.6	393	739-4		230.6	447	836.6		260.6	501	933.8
201.1	394	741.2		231.1	1118	838.4		261.1	502	935.6
201.7	395	743.0		231.7	449	840.2		261.7	503	937.4
202.2	396	744.8		232.2	450	842.0		262.2	504	939.2
202.8	397	746.6		232.8	451	843.8		262.8	505	941.0
203.3	398	748.4		233.3	452	845.6		263.3	506	942.8
203.9	399	750.2		233.9	453	847.4		263.9	507	944.6
204.4	400	752.0		234.4	454	849.2		264.4	508	946.4
205.0	101	753.8		235.0	455	851.0		265.0	509	948.2
205.6	402	755.6		235.6	456	852.8		265.6	510	950.0 951.8
206.1 206.7	403	757.4		236.1	457	854.6 856.4		266.1	511 512	953.6
207.2	404	759 <b>.</b> 2 761.0		236.7	458 459	858.2		266.7	513	955-4
207.8	405 406	762.8		237.8	455	860.0		267.2 267.8	514	957.2
208.3	407	764.6		238.3	461	861.8		268.3	515	959.0
208.9	408	766.4		238.9	462	863.6		268.9	516	960.8
209.4	409	768.2	ł	239.4	463	865.4		269.4	517	962.6
210.0	110	770.0		240.0	464	867.2		270.0	518	964.4
210.6	411	771.8	Į –	240.6	465	869.0		270.6	519	966.2
211.1	412	773.6	Į	241.1	466	870.8		271.1	520	968.0
211.7	413	775.4		241.7	467	872.6	l	271.7	521	969.8
212.2	414	777.2	ł	242.2	468	874.4		272.2	522	971.6
212.8	山5	779.0		242.8	469	876.2		272.8	523	973.4
213.3	<u>ш6</u>	780.8		243.3	470	878.0 879.8		273.3	524 525	975.2 977.0
213.9 214.4	17	782.6 784.4		243.9	471 472	881.6		273.9	526	978.8
215.0	山8	786.2		244.4 245.0	473	883.4		274.4	527	980.6
215.6	420	788.0		245.6	474	885.2		275.6	528	982.4
216.1	421	789.8		246.1	475	887.0	Į.	276.1	529	984.2
216.7	422	791.6		246.7	476	888.8	1	276.7	530	986.0
217.2	423	793.4		247.2	477	890.6		277.2	531	987.8
217.8	424	795.2		247.8	478	892.4		277.8	532	989.6
218.3	425	797.0		248.3	479	894.2		278.3	533	991.4
218.9	426	798.8		248.9	480	896.0		278.9	534	993.2
219.4	427	800.6	1	249.4	2,81	897.8		279.4	535	995.0
220.0	428	802.4		250.0	482	899.6	ł	280.0	536	996.8
220.6	429	804.2		250.6	483	901.4	1	280.6	537	998.6
221.1	430	806.0		251.1	484	903.2		281.1	538	1000.4
221.7 222.2	431	807 <b>.</b> 8 809 <b>.</b> 6		251.7	485	905.0		281.7	539 540	1002.2
222.8	432	811.4		252.2	486 487	908.6	I	282.2	540	1004.0
223,3	433 434	813.2		252.8	488	910.4	Į	282.8 283.3	541	1003.6
	404	2.2	1	دوري	400	710.4	J	203.3	<u><u> </u></u>	100100

TABLE 9.7

CONVERSION TABLE

•c		• <sub>F</sub>	[	°c		op		•0		۰F
283.9	543	1009.4		313.9	597	1106.6		343.9	651	1203.8
284.4	545	1011.2		314.4	598	1108.4		344.4	652	1205.6
285.0	545	1013.0		315.0	599	1110.2		345.0	653	1207.4
285.6	546	1014.8		315.6	600	1112.0		345.6	654	1209.2
286.1	547	1016.6		316.1	601	1113.8		346.1	655	1211.0
286.7	548	1018.4	j	316.7	602	1115.6		346.7	656	1212.8
287.2	549	1020.2		317.2	603	1117•4		347.2	657	1214-6
287.8 288.3	550	1022.0		317.8	604	1119.2		347.8 348.3	658	1216.4
288.9	551 552	1023.8 1025.6		318.3	605	1121.0		348.9	659	1218.2
289.4	553	1027.4		318.9 319.4	606	1122.8 1124.6		349.4	660 661	1220.0 1221.8
290.0	553 554	1029.2		320.0	60? 608	1124.0		350.0	662	1223.6
290.6	555	1031.0		320.6	609	1128.2		350.6	663	1225.4
,291.1	556	1032.8		321.1	610	1130.0		351.1	664	1227.2
291.7	557	1034.6		321.7	611	1131.8		351.7	665	1229.0
292.2	558	1036.4		322.2	612	1133.6		352.2	666	1230.8
292.8	559	1038.2		322.8	613	1135.4	i i	352.8	667	1232.6
293•3 293•9	560 561	1040.0 1041.8		323.3	614	1137.2		353.3	668	1234.4
294.4	562	1043.6		323.9	615	1139.0		353.9	669	1236.2
295.0	563	1045.4		324.4	616	1140.8		354.4	670	1238.0
295.6	504	1047.2		325.0	617	1142.6		355•0 355•6	671	1239.8
296.1	565	1049.0		325.6 326.1	618	1144.04 1146.2		356.1	672 673	1241.6 1243.4
296.7	566	1050.8		326.7	619 .620	1148.0		356.7	674	1245.2
297.2	567	1052.6		327.2	621	1149.8		357.2	675	1247.0
297.8	565	1054.4		327.8	622	1151.6		357.8	676	1243.8
298.3	559	1056.2		328.3	623	1153.4		358.3	677	1250.6
298.9	570	1058.0		328.9	624	1155.2		358.9	678	1252.4
.299•4 300•0	571	1059.8		329 • 4	625	1157.0		359-4	679	1.254.2
300.6	572 573	1061.6 1063.4		330.0	626	1158.8		360.0	680	1256.0
301.1	574	1065.2		330.6	627	1160.6		360.6	681	1257.8
301.7	575	1067.0		331.1	628	1162.4		361.1 361.7	682	1259.6
302.2	576	1068.8		331.7 332.2	629	1164.2		362.2	683	1261.4
302.8	577	1070.6		332.8	630 631	1166.0		362.8	684 685	1263.2 1265.0
303.3	578	1072.4		333.3	632	1169.6	ĺ	363-3	686	1266.8
303.9	579	1074.2		333.9	633	1171.4		363.9	687	1268.6
304.4	580	1076.0		334.4	634	1173.2	Į	364.4	688	1270.4
305.0	581	1077.8		335.0	635	1175.0		365.0	689	1272.2
305.6 306.1	582	1079.6		335.6	636	1176.8		365.6	690	1274.0
306.7	583 584	1081.4 1083.2		336.1	637	1178.6	l	366.1	691	1275.8
307.2	585	1085.0		336.7	638	1180.4	ļ	366.7	692	1277.6
307.8	586	1086.8		337.2	639	1182.2	i	378.2	693	1279.4
308.3	587	1088.6		337.8 338.3	640	1184.0		367.8 368.3	694	1281.2
308.9	588	1090.4		338.9	641 642	1185 <b>.8</b> 1187 <b>.6</b>		368.9	6 <del>9</del> 5 696	1283.0 1284.8
309-4	589	1092.2		339.4	643	1189.4		369.4	697	1286.6
310.0	590	1094.0		340.0	644	1191.2		370.0	698	1288.4
310.6	591	1095.8		340.6	645	1193.0		370.6	699	1290.2
311.1	592	1097.6		341.1	646	1194.8		371.1	700	1292.0
311.7 312.2	593 594	1099.4		341.7	647	1196.6		371.7	701	1293.8
312.8	594	1101.2 1103.0		342.2	648	1198.4		372.2	702	1295.6
313.3	596	1104.8		342.8	649 :	1200.2		372.8	703	1297.4
	L	1 1104.0		343.3	650	1202.0	J	373 • 3	704	1299.2

TABLE 9.7

# CONVERSION TABLE

.

•C ·		°F		•C		۰F		•C		•
373.9	705	1301.0		403.3	758	1396.4		432.8	811	1491.8
374.4	706	1302.8		403.9	759	1398.2		433.3	812	1493.6
375.0	707	1304.6		404 <b>.</b> 4	760	1400.0		433.9	813	1495.4
375.6	708	1306.4		405.0	761	1401.8		434.4	814	1497.2
376.1	709	1308.2		405.6	762	1403.6		435.0	815	1499.0
376.7	710	1310.0		405.1	763	1405.4		435.6	816	1500.8
377.2	711	1311.8		406.7	764	1407.2		436.1	817	1502.6
377.8	712	1313.6		407.2	765	1409.0		436.7	818	1504.4
378.3	713	1315.4		407.8	766	1410.8		437.2	819	1506.2
378.9	$n_{4}$	1317.2		408.3	767	1412.6		437.8	820	1508.0
379.4	715	1319.0		408.9	768	1414.4		438.3	821	1509.8
380.0	716	1320.8		409.4	769	1416.2		438.9	822	1511.6
380.6	717	1322.6		110.0	770	1418.0		439.4	823	1513.4
381.1	718	1324.4		山0.6	771	1419.8		山0.0	824	1515.2
381.7	719	1326.2		41.1	772	1421.6		Щ0.6	825	1517.0
382.2	720	1328.0		ш1.7	773	1423.4		441.1	826	1518.8
382.8	721	1329.8		山2.2	774	1425.2		441.7	827	1520.6
383.3	722	1331.6		412.8	775	1427.0		442.2	828	1522.4
383.9	723	1333.4		413.3	776	1428.8		442.8	829	1524.2
384.4	724	1335-2		413.9	777	1430.6		443.3	830	1526.0
385.0	725	1337.0		414.4	778	1432.4		443.9	831	1527.8
385.6	726	1338.8		415.0	779	1434.2		444.4	832	1529.6
386.1	727	1340.6		415.6	780	1436.0		445.0	833	1531.4
386.7	728	1342.4		416.1	781	1437.8		445.6	834	1533.2
387.2	729	1344.2		416.7	782	1439.6		446.1	835	1535.0
387.8	730	1346.0		117.2	783	1441.4		446.7	836	1536.8
388.3	731	1347.8		417.8	784	1443.2		447.2	837	1538.6
388.9	732	1349.6		418.3	785	1445.0		44.7.8	838	1540.4
389.4	733	1351.4		418.9	786	1446.8		LU18-3	839	1542.2
390.0	734	1353.2		419.4	787	1448.6		448.9	840	15440
390.6	735	1355.0		420.0	788	1450.4		449.4	841	1545.8
391.1	736	1356.8		420.6	789	1452.2		450.0	842	1547.6
391.7	737	1358.6		421.1	790	1454.0		450.6	843	1549-4
392.2	738	1360-4		421.7	791	1455.8		451.1	844	1551.2
392.8	739	1362.2		422.2	792	1457.6		451.7	845	1553.0
393.3	740	1364.0		422.8	793	1459.4		452.2	846	1554.8
393.9	744	1365.8		423.3	794	1461.2	l	452.8	847	1556.6
394+4	742	1367-6		423.9	795	1463.0	1	453.3	848	1558.4
395.0	743	1369.4		424.4	796	1464.8	1	453.9	849	1560.2
395.6	744	1371.2		425.0	797	1466,6	1	454.4	850	1562.0
396.1	745	1373.0	I	425.6	798	1468.4	1	455.0	851	1563.8
396.7	746	1374.8		426.1	799	1470.2	ļ	455-6	852	1565.0
397.2	747	1376.6	1	426.7	800	1472.0	1	456.1	853	1567.4
397.8	748	1378.4		427.2	801	1473.8	l	456.7	854	1569.2
398.3	749	1380-2		427.8	802	1475.6	1	457.2	855	1571.0
398.9	750	1382.0		428.3	803	1477-4	1	457.8	856	1572.8
399-4	751	1383.8	I	428.9	804	1479.2	1	458.3	857	1574.6
400.0	752	1385.6	I	429-4	805	1481.0	1	458.9	858	1576.4
400.6	753	1387.4		430.0	806	1482.8	1	459.4	δ59	1578.2
401.1	754	1389.2	Ĩ	430.6	807	1484.6	1	460.0	860	1580.0
401.7	755	1391.0	l	431.1	808	1486.4	1	460.6	861	1581.8
402.2	756	1392.8	1	431.7	809	1488.2	ļ	461.1	862	1583.6
402.8	757	1394.6	L	432.2	810	1490.0	J .	461.7	863	1585.4

TABLE 9.7

# CONVERSION TABLE

•C		•F		•C		۰F		•C		
462.2	. 864	1587.2		491.7	917	1682.6		521.1	970 ·	1778.0
462.8	865	1589.0		492.2	918	1684.4		521.7	971	1779.8
463.3	866	1590.8		492.8	919	1686.2		522.2	972	1781.6
463.9	867	1592.6		493.3	920	1688.0		522.8	973	1783.4
464.4	868	1594.4		493.9	921	1689.8		523.3	974	1785.2
465.0	869	1596.2		494.4	922	1691.6		523.9	975	1787.0
465.6	870	1598.0		495.0	923	1693.4		524.4	976	1788.8
466.1	871	1599.8		495.6	924	1695.2		525.U 525.6	977 978	1790.6 1792.4
466.7	872	1601.6		496.1	925	1697.0		526.1	979	1794.2
167.2	873	1603.4		496 <b>.</b> 7 497 <b>.</b> 2	926	1698.8		526.7	980	1796.0
467.8 468.3	874	1605-2		497.8	927	1700.6		527.2	981	1797.8
468.9	875 876	1607.0 1608.8		498.3	928 929	1702.4		527.8	982	1799.6
469.4	877	1610.6		498.9	930	1704.2 1706.0		528.3	983	1801.4
470.0	878	1612.4		499.4	931	1707.8		528.9	984	1803.2
470.6	879	1614.2		500.0	932	1709.6		529.4	985	1805.0
471.1	880	1616.0		500.6	933	1711.4		530.0	986	1806.8
471.7	881	1617.8		501.1	934	1713.2		530.6	987	1808.6
472.2	882	1619.6		501.7	935	1715.0		531.1	988	1810.4
472.8	883	1621.4		502.2	936	1716.8		531.7	989	1812.2
473+3	884	1623.2		502.8	937	1718.6		532.2	990	1814.0
473-9	885	1625.0		503.3	938	1720.4		532.8	991	1815.8 1817.6
474.4	886	1626.8		503.9	939	1722.2		533•3 533•9	992 993	1819.4
475.0	887	1628.6		504•4 505•0	940	1724.0		534.4	994	1821.2
475.6 476.1	888	1630.4		505.6	941	1725.8		535.0	995	1823.0
476.7	889 890	1632.2 1634.0		506.1	942 943	1727.6		535.6	996	1824.8
477.2	891	1635.8		506.7	945	1729.4		536.1	997	1826.6
477.8	892	1637.6		507.2	945	1733.0		536.7	998	1828.4
478.3	893	1639.4		507.8	946	1734.8		537 -2	999	1830.2
478.9	894	1641.2		508.3	947	1736.6		537.8	1000	1832.0
479+4	895	1643.0		508.9	948	1738.4	ł	538.4	1001	1833.8
480.0	896	1644.8		509.4	949	1740.2		538.9	1002	1835.6
480.6	897	1646.6		510.0	950	1742.0		539.5	1003	1837.4
481.1	898	1648.4		510.6	951	1743.8		540.0 540.6	1004	1839.2 1841.0
481.7	899	1650,2		511.1	952	1745.6		540.0	1005	1842.8
182.2 182.8	900	1652.0		511.7 512.2	953	1747.4		54.7	1007	1844.6
483.3	901	1653.8		512.8	954	1749.2		542.3	1008	1846.4
403.5	902 903	1655.6 1657.4		513.3	955	1751.0	[	542.8	1009	1848.2
484.4	903	1659.2	ł	513.9	956 957	1752.8		543.4	1010	1850.0
185.0	905	1661.0		514.4	958	1756.4		543.9	1011	1851.8
485.6	906	1662.8	ł	515.0	959	1758.2		544.5	1012	1853.6
486.1	907	1664.6		515.6	960	1760.0	1	545.0	1013	1855.4
486.7	908	1666.4		516.1	961	1761.8		545.6	1014	1857.2
487.2	909	1668.2		516.7	962	1763.6	l	546.2	1015	1859.0
487.8	910	1670.0		517.2	963	1765.4	l	546.7	1016	1860.8
488.3	911	1671.8	1	517.8	964	1767.2		547.3	1017	1862.6
488.9	912	1673.6		518.3	965	1769.0	1	547.8 548.4	1018	1864.4 1866.2
489.4	913	1675-4		518.9 519.4	966	1770.8	1	548.9	1019	1868.0
490.0 490.6	914	1677.2		520.0	967 968	1772.6	1	549.5	1020	1869.8
491.1	915 916	1679.0 1 <b>680.8</b>		520.6	969	1774.4	1	550.0	1022	1871.6
47691	1 710	1000.0	i i		1,01	1 11000				

TABLE 9.7

# CONVERSION TABLE

<b></b>			T		<u> </u>	·	•F ]
	°∙C		•F				
5	50.6	1023	1873.4		572.3	1062	1943.6
	51.2	1024	1875.2		572.8	1063	1945-4
5	51.7	1025	1877.0		573-4	1064	1947.2
5	52.3	1026	1878.8		573.9	1065	1949-0
	52.8	1027	1880.6		574.5	<b>106</b> 6	1950-8
	53.4	1028	1882.4		575.0	1067	1952.6
	<b>53.</b> 9	1029	1884.2		575.6	1068	1954.4
	54.5	1030	1886.0		576.2	1069	1956-2
	55.0	1031	1887.8	1	576.7	1070	1958.0
	55.6	1032	1889.6		577.3	1071	1959.8
	56.2	1033	1891.4		577.8	1072	1961.6
	556.7	1034	1893.2		578.4	1073	1963.4
	57.3	1035	1895.0		578.9	1074	1965.2
	.57.8	1036	1896.8		579.5	1075	1967.0
	58.4	1037	1898.6		580.0	1076	1968.8
	558.9	1038	1900.4		580.6	1077	1970.6
	559.5	1039	1902.2		581.2 581.7	1078	1972.4
	560.0	1040	1904.0		582.3	1079	1974-2
	560.6	1041	1905.8		582.8	1080 1081	1976.0
	561.2 561.7	1042 1043	1907.6		583.4	1082	1977.8 1979.6
	562 <b>•</b> 3	1045	1909.4 1911.2		583.9	1082	1981.4
	562.8	1044	1911.2		584.5	1084	1983.2
	563.4	1049	1913.0		585.0	1084	1985.0
	563.9	1040	1914.0		585.6	1086	1986.8
	564+5	1048	1918.4		586.2	1087	1988.6
	565.0	1049	1920.2		586.7	1088	1990.4
	565.6	1050	1922.0		587.3	1089	1992.2
	566.2	1051	1923.8		587.8	1090	1994.0
	566.7	1052	1925.6	ļ	588.4	1091	1995.8
	567.3	1053	1927.4		588.9	1092	1997.6
	567.8	1054	1929.2	Í	589.5	1093	1999.4
	568.4	1055	1931.0		590.0	1094	2001.2
	568.9	1056	1932.8	1	590.6	1095	2003.0
	569.5	1057	1934.6	1	591.2	1096	2004.8
	570.0	1058	1936.4		591.7	1097	2006.6
	570.6	1059	1938.2	l	592.3	1098	2008-4
	571.2	1060	1940.0	1	592.8	1099	2010.2
Ĩ	571.7	1061	1941.8		593.4	1100	2012.0

## CHAPTER TWO

## RECIPROCATING ENGINE PERFORMANCE

#### SECTION 2.1

## Horsepower Determination for Test Conditions

In aircraft performance testing, engine performance is the evaluation of the engine as installed in the aircraft. Since aircraft induction and exhaust systems affect operation, engine manufacturers tests will not indicate the exact installed performance. Tests must be run to determine power available, critical altitude, fuel consumption, and cooling data at standard day operating temperatures. These values not only determine overall airplane performance when applied to the aerodynamic characteristics of the airframe, but also show the quality of the engine installation when compared to the performance of the isolated engine.

Considering the propeller driven aircraft, power is the first characteristic to be studied. The total useful power produced is of primary interest. This is the power with which the aircraft manufacturer can work. It may all be used to drive the propeller or part may be extracted to drive auxiliary equipment. In making guarantees, the total useful power is specified, and the assumption is made that the manufacturer may use it as he wishes. If it is used to run cooling fans, less power will be delivered to the propeller, but less cooling drag may be achieved; if it is used for cabin pressurization or electric generators, weight is saved by the elimination of auxiliary motors. As a result, in performance testing, total power available is considered to be a more useful criteria than the thrust power commonly used by aerodynamicists.

In all propeller engine combinations power is best determined by use of a torquemeter attached to the shaft to measure useful torque output. The horse-power is given by the equation:

BHT<sub>+</sub> = Torquemeter reading  $x K x N_{ic}$ 

where:

K = a constant determined by dynamometer tests

N<sub>ic</sub> = engine rpm corrected for instrument error

BHP, = brake horsepower on the test day

When torquemeters are not available power charts are used. These charts solve for brake horsepower when manifold pressure, rpm, carburetor air temperature, and pressure altitude are known. They are made partly from dynamometer tests and partly from theory. They presume that all additional factors effecting power, such as oil pressure, oil temperature, and cylinder head temperature, remain constant or vary in a predetermined manner. They also assume that the fuel-air mixture is exactly as specified and ignition is perfect. These assumptions are not valid in all installations or operations; so the charts are not exact. They do, however, represent a reasonable approximation in the absence of a better measuring system.

A typical chart consists of three parts, a sea level power graph, an altitude correction graph, and a chart carburstor air temperature graph. To determine the horsepower delivered at any given manifold pressure, rpm, altitude, and carburetor air temperature the following procedure is used on a typical power chart as shown on Figure 2.11.

(a) Find the point corresponding to the given manifold pressure and rpm on the altitude correction graph, (Point A).

(b) Find the power which would be delivered if the engine were operating at sea level with the given rpm and M.P., (Point B).

(c) Transpose Point B to the zero altitude axis of the altitude correction graph, (Point C).

(d) Connect Point A and Point C by a straight line. This line will repre-

(e) Find the horsepower (Point D) at the intersection of the given pressure altitude axis and the line AC. This horsepower is called chart horsepower; it is the horsepower which would be delivered at the given manifold pressure, rpm, pressure altitude, and chart carburetor air temperature.

(f) Correct chart horsepower to test horsepower by the empirical equation:

(2.102)

BHPt = BHPc	$\left(\frac{\mathbf{\hat{T}_{3C}}}{\mathbf{\hat{T}_{ct}}}\right)^n$		
-------------	--	--	--

wbere:

BHP<sub>t</sub> = test brake horsepower

BHPa = chart brake horsepower

- $T_{eo}$  chart carburetor air temperature for the pressure altitude. This is generally the standard temperature for the specified altitude, as the manufacturer tests the engine with no cowl or ducting.
- T<sub>ct</sub> = test carburetor air temperature
  - n = exponent specified by power curve, usually (0.5)

If the exponent is (0.5), the chart horsepower correction may be given as a 1% decrease for each 6°C the test carburetor temperature exceeds the chart carburetor air temperature when the temperatures are near 300°K. Manifold mixture temperature is often used in place of carburetor air temperature, as it also is a measure of the temperature of the charge entering the clinder. If this temperature is used instead of carburetor air temperature, the exponent may change but will still be specified by the power curve. For example, using Figure 2.11, find the test power of an R 2000-11 engine at 30° manifold pressure 2000 rpm, at 18,000 feet, carburetor air temperature -10°C.

AFTR 6273

Chart Horsepower = 728 Carb air chart = 253°K Carb air test = 263°K Test horsepower = 728 x  $\sqrt{\frac{253}{263}}$  = 716

In addition to determining power delivered under test conditions, power delivered on a standard day at certain arbitrary settings such as oruise power, maximum continuous power, and military power are required. To determine these powers, the aircraft is flown as near as possible to the desired setting on a test day, and the power obtained is corrected to the power which would be obtained on a standard day at the same pressure altitude. Most power corrections may be resolved into three cases: if the required setting, manifold pressure and rpm does not require full throttle; if the setting does require full throttle; if the setting requires full throttle but manifold pressure may be increased or decreased by a turbosupercharger.



AFT: 6273

1

#### SECTION 2.2

# Power Correction for Temperature Variation at Constant Manifold Pressure

This case is used for such problems as determining the cruising power of any engine or the military power of a supercharged engine at altitudes where wide open throttle would exceed the manufacturer's operating limits. In this case the assumption is made that, if the airplane were flown on a standard day, the throttle position would be slightly different from its position on a test day but manifold pressure would be constant.

With manifold pressure and pressure altitude the same on the test day and the standard day, the change in carburetor air temperature will be the only variable effecting power, and the correction can be made by the same relation used in obtaining test horsepower from chart horsepower. Tests have shown that carburetor air temperature will change from test day to standard day by exactly the same amount as free air temperature changes from test day to standard day. Equation 2.201 is used in this case.

$$BHP_{g} = BHP_{t} \left(\frac{T_{ct}}{T_{cg}}\right)^{n}$$
(2.201)

where:

The following table is an example of test and standard day conditions.

TEST DAY		STANDARD DAY					
Pressure	Alt 18,000'	Pressure Alt 18,000'					
T <sub>a</sub> MP	-10 <sup>•</sup> C	Тд ~20°С МР 30"Ща					
MP	<b>30"Н</b> 8	MP 30"Eg					
rpa	2000	rpm 2000					
CAT	0 <b>°C</b>	CAT -10°C					
$BHP_t$	703	BHP <sub>6</sub> 716					
Thrott	le partly open	Throttle slightly retarded from test day position					

Another form of equation 2.201 is sometimes used.  $\Delta BHP (\text{for } \Delta CAT) = BHP_t \left[ \left( \frac{T_{ot}}{T_{os}} \right)^n - 1 \right] \qquad (2.202)$ 

where:

 $\Delta BHP$  (for  $\Delta CAT$ ) = (BHP<sub>8</sub> - BHP<sub>t</sub>), brake horsepower correction for carburstor air temperature change

# SECTION 2.3

# Power Correction for Manifold Pressure Variation Resulting from Temperature Variation and Flight Mach Number Variation

This case is used for problems such as maximum power or cruising power above the critical altitude for the desired power setting. In this case the throttle will be full open in an effort to obtain the desired manifold pressure and rpm; therefore, if the airplane were flown on a standard day the throttle would remain open and manifold pressure as well as carburetor air temperature would change due to the change in free air temperature. The correction for the two effects may be stated in this form:

 $BHP_{a} \bullet BHP_{t} + \Delta BHP (for \Delta MP) \qquad (2.301)$ 

The first correction term is obtained from the carburator temperature relation as used in equation 2.202. The manifold pressure correction requires consideration of two effects: the change in pressure ratio of the supercharger due to change in inlet temperature and the change in air inlet ram pressure ratio because of any change in Mech number of the aircraft caused by power changes.

The following table is an example of test and standard day conditions for the case where a turbosupercharger is not involved and manifold pressure and Mach number vary from test to standard day.

test day		STANDARD DAY	
Test power setti	ng 30" MP	Pressure altitude	20,000
Pressure altitud	e 20,000	rpm	2000
rpm	2000	T	~?4°C
T	-14°C	T MP	27.7 "Hg
T <sub>a</sub> MP	27 "Hg	CAT	-14°C
CAT	-4°C	BHP	646
BHP,	<b>63</b> 0	Throttle wide open	
Throttle wide or	en -	-	

#### TEMPERATURE EFFECTS ON MANIFOLD PRESSURE

The exact correction of manifold pressure for change in free air temperature would result in work and instrumentation beyond the scope of flight test, because of the varied thermodynamic processes involved in induction and carburetion. An approximate method has been devised and is presented in Flight Test Coction Memorandum Report TSCEP5E-2T, 6/29/45, "<u>A Simplified Manifold Pressure</u> Correction." The basic equation used in this method is:

 $MP_{\theta} = (MP_{t}) (1 + CAT)$  (2.302)

where:

MP<sub>a</sub> = manifold pressure standard "Hg (MP<sub>t</sub>) = manifold pressure test "Hg At = (T<sub>at</sub> - T<sub>ag</sub>) different between test day carburetor air temperature and standard day carburetor air temperature (T<sub>ct</sub> - T<sub>cg</sub>) C = a constant depending upon the type process employed

The correction constant C is dependent on the pressure ratio of the process, the initial temperature of the process, and whether fuel is vaporized during the process. CHARTS 2.31 and 2.32 at the end of this chapter have been made from equation 2.302 for the value of C in the normal working range. It should be noted that, if the ratio of test manifold pressure to ambient pressure is less than 1.5, this correction is negligible for temperature variations of  $5^{\circ}$ C or less.

By use of the two graphs, any combination of induction processes for air only or a fuel air mixture may be evaluated. Manifold pressure data reduction methods for typical induction systems are included at the end of this section. It should be noticed that, once the manifold pressure corrections have been established for a typical engine installation, they can be used for all other duplicate installations by use of a small chart showing the corrections at various altitudes for various manifold pressure rpm combinations.

#### MACH NUMBER EFFECTS ON MANIFOLD PRESSURE

The determination of the standard day manifold pressure can be obtained by multiplying the temperature corrected manifold pressure by the ratio of the standard and test day ram pressure ratios. The average carburetor inlet has a ram efficiency of about (0.70 to 0.75). Using CHART 2.33 at the end of this chapter, with flight Mach number and ram efficiency, the ratio  $P_t/P_a$ may be determined. Obtaining this ratio for both test and standard Mach number, and assuming the test day pressure altitude is held,

$$\frac{P_{tg}/P_{a}}{P_{tt}/P_{a}} = \frac{P_{tg}}{P_{tt}} = \frac{MP_{g}}{MP_{\Delta t}}$$

and

1

1

$$\mathbf{P}_{\mathbf{S}} = \mathbf{MP}_{\Delta t} \times \frac{\mathbf{P}_{tg}}{\mathbf{P}_{t+1}}$$

where:

M

 $\begin{array}{rcl} MP_{g} &=& \texttt{standard manifold pressure} \\ MP_{\Delta t} &=& \texttt{manifold pressure corrected for temperature variation} \\ P_{\Delta t} &=& \texttt{standard total inlet pressure at standard flight Mach number} \\ P_{tt} &=& \texttt{test total inlet pressure at test Mach number} \end{array}$ 

(2.303)

This correction must be made by successive approximations because standard speed cannot be exactly determined from the polar until power is known. The sorrection is not normally made unless the airplane Mach number is above .6 and in overall change in speed because of change in power is over 3 knots.

## POWER CORRECTION FOR MANIFOLD PRESSURE VARIATION

Having determined standard day manifold pressure, its effect on power must be evaluated. For rough work the change in power is directly proportional to the change in manifold pressure.

$$BHP_{s} = BHP_{t} \qquad \frac{MP_{s}}{MP_{t}} \qquad (2.304)$$

or

$$\Delta \text{EHP} (\text{for } \Delta \text{MP}) = \text{EHP}_{t} \left( \frac{\text{MP}_{g}}{\text{MP}_{t}} - 1 \right)$$
(2.305)

where:

 $MP_s$  = temperature and Mach number corrected manifold pressure  $MP_t$  = test manifold pressure

Better accuracy can be obtained by making plots of  $BEP_t$  vs MP<sub>t</sub> at various altitudes and rpms in order to find the rate of change of power with MP as shown in Figure 2.31. With the slopes from such a plot  $\Delta EHP$  may be calculated. (In the pressure altitude method of data reduction this correction is not required — see Section 4.4 — since only test EHP is plotted; in this case equation 2.304 may be used to approximate standard day MP for the standard day EHP determined from the pressure altitude plot.)

$$\Delta BHP (for \Delta MP) = \Delta MP_t \times \frac{d(BHP)}{d(MP)}$$
(2.305)

With the temperature corrections of Section 2.2 and the standard manifold pressure an equation for standard day power may be written:

$$BHP_{g} = BHP_{t} \left( \left( \frac{T_{ot}}{T_{og}} \right)^{n} + \frac{MP_{g}}{MP_{t}} - 1 \right)$$
(2.307)

or

$$BHP_{g} = BHP_{t} \left(\frac{T_{ot}}{T_{os}}\right)^{n} + \Delta MP \quad \frac{d(BHP)}{d(MP)}$$
(2.308)

The final standard day power curves are presented in a form similar to that shown in Figure 2.32.



Figure 2.31 Power Variation for Manifold Pressure Variation at Constant RPM and Altitude (Test Data)

AFTR 6273



Figure 2.32 Standard Day Engine Data (Internal Gear Drive Supercharger)

AFTR 6273

#### DATA REDUCTION OUTLINE (2.31)

For Correcting Manifold Pressure for Temperature Variation (Wide Open Throttle)

#### Example 1

Normal installations where the inlet temperature is measured before the fuel is added at the carburetor and before the charge is compressed.

- Test point number
- (1) (2) (3) H<sub>c</sub>, pressure altitude
- $P_a$ , inlet (atmospheric) pressure corresponding to (2)
- (4) (5) (6) tat, test atmospheric temperature
- tas, standard atmospheric temperature
- Δt, (4) (5)
- (7) tot, test carburetor inlet temperature
- (8) MPt, test, instrument-corrected manifold pressure
- (9)  $P_2/P_1$ , manifold pressure ratio, (8) + (3)
- (10) C, from CHART 2.32 and (7) and (9)
- (11) $\Delta MP$ , manifold pressure correction (10) x (8) x (6)
- (12) $MP_{\alpha}$ , standard manifold pressure, (8) + (11)

## Example II

Fuel injection engine or any supercharger where air only is compressed; also for fuel air mixtures when the inlet temperature is measured after the fuel is vaporized.

NOTE: For this case the data reduction is identical to that for Example 1, except that (4) is substituted for (7) and CHART 2.31 is used to determine C.

### Example III

For installations where part of the induction pressure rise is with air only, and the remaining part is with a fuel air mixture. (Turbogupercharger installations and auxiliary blower installations when operating at constant RPM.)

(1) Test point number

- H<sub>o</sub>, true pressure altitude
- (2) (3)  $P_{A}$ , atmospheric pressure corresponding to (2)
- (4) tat, test atmospheric temperature
- (6) (6)  $t_{as}$ , standard atmospheric temperature corresponding to (2)
- $\Delta t$ , (4) (5)
- (7) tat, carburetor air temperature before fuel is added
- (8) P<sub>d</sub>, test carburetor deck pressure
- (9) MPt, test, instrument-corrected manifold pressure, air only stage
- $(P_2/P_1)_a$ , carburetor-deck-ambient pressure ratio, (8) + (3) (10)

- C, from CHART 2.31 and (10) and (4) (11)
- $\Delta MP_a$ , manifold pressure correction for air only stage (11) x (9) x (6) (12)

# Fuel Air Stage

- (13)
- (14)
- $(P_2/P_1)_f$ , manifold pressure deck pressure ratio (9) 4 (8) C, from CHART 2.32 and (13) and (7)  $\triangle MP_f$ , manifold pressure correction for fuel air mixture [(9) + (12)]x(15) (14)<sup>x</sup> (6)
- MP<sub>g</sub>, standard manifold pressure (9) + (12) + (15) (16)

# SECTION 2.4

# Power Correction for Turbosupercharger RPM and Back Pressure Variation at Constant Manifold Pressure

This power correction is used when the throttle is wide open but manifold pressure can be varied by changing turbosupercharger speed. This means that, in going from a test day to a standard day, manifold pressure and rpm will be constant while carburetor air temperature and turbo rpm will change giving a change in power. An example of test and standard day readings for such conditions is presented in the following table:

TEST DAY		STANDARD DAY				
Pressure Altitude	20,000	Pressure Altitude	20,000			
T <sub>at</sub> MPt	-14 °C	Tes MPe	-24 °C			
MĒt	42 <b>"Hg</b>	MP.	42 <b>"</b> Hg			
rpa	<b>225</b> 0	rpm	2250			
tot	27 °C	top	15°C			
<sup>t</sup> ot test turbo rpm	7200	sťď turbo rpm	7000			
BHP	2100	BHP	2150			
EBP <sub>t</sub>	28	ESP.	27			

The factors affecting power in this case are three: the change in carburetor temperature because of change of free air temperature; the change of carburetor air temperature because of change in turbo speed; and the change in exhaust back pressure. The equation for standard horsepower under these conditions is:

 $BHP_{e} = BHP_{t} + \Delta BHP (for \Delta CAT) + \Delta BHP (for \Delta EBP) \qquad (2.401)$ 

where:

 $\Delta EBP = \text{ohange in exhaust back pressure } (EBP_t - EBP_e)$ 

The corrections in this equation are usually made empirically. First the assumptions are made that the turbo does not change speed and that a change in manifold pressure would result from inlet temperature variation in going from a test day to a standard day. On this basis a standard day manifold pressure is computed as described in Section 2.3. This manifold pressure correction is defined as,

 $\Delta MP = (MP_t - MP_s)$  at test turbo rpm

This AMP represents the change in manifold pressure which must be made by a change in turbo rpm. To establish this change in turbo rpm and the other related factors, plots of performance data are made as shown in Figure 2.41. These curves, although not corrected, will show closely the interrelated effect of changing any one of the variables. By entering such a graph at the test turbo rpm and moving an amount equal to  $\Delta MP$ , a  $\Delta turbo rpm (RHM_2-RPM_1)$ a  $\Delta EBP (EBP_2 - EBP_1)$  and a  $\Delta CAT (T_{02} - T_{01})$  are established. Standard day values for turbo rpm and exhaust back pressure can then be fixed by direct reading on the graph or by applying the increments for the  $\Delta MP$  values to the test values.

AFTR 6273

The total change in carburator air temperature is the sum of the change because of change in free air temperature and change in turbo rpm.

$$\Delta CAT = T_{AB} - T_{At} + \Delta T$$
 (for  $\Delta turbo rpm)$ 

The power change because of carburetor air temperature variation is the same as that discussed in Section 2.2. The change in power because of change in exhaust back pressure has been empirically established as 1% increase in power for each 2 "Hg decrease. With this the standard day power equation is,

$$BHP_{s} = BHP_{t} \left[ \left( \frac{T_{ot}}{T_{os}} \right)^{n} + .005 (EBP_{t} - EBP_{s}) \right]$$
(2.402)

POWER SETTING FRRORS IN FLICHT

In addition to the correction of power for temperature change it must sometimes be corrected for errors caused by test day manifold pressure or rpm not being set according to schedule. This results from instrument error or human error of the pilot. Rpm errors must be minimized by careful adjustments and anticipation of instrument error, because correction for its variation is not practical. Manifold pressure errors are more easily corrected. Usually the correction is made at the same time that power is corrected for temperature. For a case in which throttle is wide open no correction is required. For a gear supercharged or unsupercharged engine at part throttle, correction is made by manifold pressure -power relationships as described in Section 2.3. For a turbo supercharged engine, correction is made as part of the temperature correction. For example:

```
60"
Desired MP
Actual flown MP
                        59"
                         2"
Rise in MP to standard
 day computed with as-
 sumed constant turbo
 apeed
                        61"
MP obtained from test
 MP and assumed rise
Reduction in MP to be
                       -1"
 accomplished by re-
 duoing turbo
```





APTR 6275

## DATA REDUCTION OUTLINE (2.41)

# For Determining Standard Turbo RPM and Back Pressure and CAT for Correcting BHP for Change in Turbo RFM and Back Pressure

- NOTE: Assumed  $\Delta MP$  for  $\Delta t_a$ ,  $(t_{at} - t_{as})$ , inlet is first determined with stope of Example III Data Reduction Outline 2.31.
- (17)  $\Delta$  MP assumed for  $\Delta t_A$
- (18) Indicated turbo rpm
- (19) Test turbo rpm, (18) + instrument correction
- (20) EBP1, indicated exhaust back pressure
- (21) EPPt, test exhaust back pressure (20) + instrument correction (22) EHPtot, test EHP at test carburetor temperature
- (23)  $\Delta t_{at}$ , carburetor temperature increment for assumed  $\Delta MP$  and a data plot similar to that of Figure 2.41.
- (24)  $\Delta$  turbo RPM, for assumed  $\Delta$  MP
- (25) (26)  $\Delta$  KBP, for assumed  $\Delta$  MP
- Turbo RPM (standard day) (19) + (24)
- (27) EBP (standard day) (21) + (25)
- (28)
- $t_{08}$ , standard day carburstor temperature, (7) (6) + (23) BHP<sub>8</sub>, standard day BHP, (22)  $\times \sqrt{(7) + (28)} + (22) \times (25) \times (.005)$ (20)

#### SECTION 2.5

## Critical Altitude

Critical altitude is that altitude where engine performance begins to drop because of the lowered atmospheric pressure. It is defined in two different ways: (1) the altitude at which a specific manifold pressure can no longer be maintained; (2) the altitude at which a specific horsepower can no longer be maintained. In flight test the first definition is generally used, because when an engine is installed in an aircraft, the operating limits such as oruising power, maximum continuous power, and military power are given in terms of manifold pressure and rpm rather than horsepower.

When a complete power available survey is made, critical altitude can be selected from the graph of manifold pressure vs altitude. When only critical altitude is desired, it can be estimated by the pilot; then several full throttle points are flown definitely above critical altitude and are corrected to standard day manifold pressure. Since the drop in manifold pressure is directly proportional to altitude, the point where the line of standard day full throttle manifold pressure intersects the desired manifold pressure ordinate is the critical altitude, as shown in Figure 2.51.

Notice should be taken that the critical altitude of an airplane is dependent on speed because of the ram pressure effect. Critical altitude is usually taken in level stabilized flight, but it is sometimes needed at climbing speeds.



Figure 2.51 Determination of Critical Altitude

AFTR 6273

# SECTION 2.6

# Engine Data Plotting, Prop Load, BMEP Data, Supercharger Operation

Sufficient power required data must be obtained so that power required curves covering the full speed range of the airplane may be plotted for at least two altitudes which are considered the most typical cruising altitudes for the test airplane. Each power required curve includes one point at each of the rated power settings of the engines such as war emergency, military, normal rated and maximum cruising powers. The throttles are wide open if the desired manifold pressure cannot be obtained. Sufficient points in the cruising power range are obtained (with the mixture in automatic lean) to complete the speed range with the lowest point at approximately the best climbing speed of the airplane. Cruising power points are obtained at rpm and manifold pressure combinations selected so that the engines are operating between a propeller load curve and the maximum allowable bmep. At some point near the power required for maximum range cruising, three points are run at the same power with one point at the rpm which lies on the propeller load curve, another point at the maximum allowable bmep intermediate rpm.

Additional power required data is obtained so that power required curves may be drawn at various altitudes; the lowest is obtained at sea level or the minimum practical altitude at the time of the test and the highest is obtained at the highest altitude reached in the check climbs. One high speed part of a power required curve is obtained at approximately each critical altitude. Other high power points are taken so that the maximum altitude increment between the high speed power required curves will not be more than a few thousand feet for all tests above the critical altitude of the airplane. The high speed power required curves consist of at least one point at each of the rated power settings such as war emergency, military, and normal rated for the high and low altitude power required curves and for the power curve near the critical altitude. Other power required curves may be drawn through single points using the more complete curves on either side of it as a guide in determining the proper slope of the curve drawn through the point.

At the critical altitude of the airplane, at least four points are run at the rated rpm of the engines with the first point at the rated manifold pressure and with each succeeding point at about two inches of mercury less manifold pressure. If the slope of the curve through these points does not fit in with the other ourves, it is an indication of power curve inaccuracy or rapid change in propeller efficiency. In either case it is necessary to run at least three points at each rpm for all of the power calibrations. From these additional points, the proper slope of the power required curve may be obtained.

A plot of all speed power data run at or corrected to the same take-off gross weight less the weight of fuel used to climb to the test altitude is plotted on one Power vs Speed chart as shown in Figure 2.61. Each power required or partial power required curve is clearly labeled to show the corresponding density altitude and gross weight.



# Figure 2.61 Power Required Curves

All power required data at rated manifold pressure or at wide open throttle is corrected to give the power available on a standard day at the rated manifold pressure or at wide open throttle so that curves of BHP available and the dorresponding manifold pressure may be plotted against altitude as shown in Figure 2.62. Curves of Speed vs Altitude for the desired power ratings are drawn as shown using points obtained from the power required curves in Figure 2.61. These points are read at the speeds corresponding to the power available at the altitude of the power required curve. The speeds for best climb and the ceiling of the airplane will be of aid in determining the shape of the upper part of the speed curve.

For each gross weight and altitude condition, a power required curve is obtained with the first point at rated power (auto-rich), the second at maximum cruising (auto rich and auto lean), with the rest of the points part of a complete survey of the cruising power range made between the propeller load curve and the curve for limiting bmep as shown in Figure 2.63. The higher cruising powers are run for both automatic rich and automatic lean mixture settings but only automatic lean mixture setting need be used for the lower powers. Select four or five rpm values covering the cruising range. At each rpm, select manifold pressure values varying from the maximum for limiting basep to the minimum corresponding to the power from a propeller load curve or to the minimum manifold pressure at which level flight can be maintained for the given rpm. At least three manifold pressure increments should be used for each rpm and the increments should not be greater than three inches of mercury.

The propeller load curve is drawn through the rated power and rated rpm point and determined by the following expression:

$$\left(\frac{N}{N_{R}}\right)^{3} = \frac{bhp}{bhp_{R}}$$
(2.501)

where:

N<sub>R</sub> = rated rpm (normal) bhp<sub>R</sub> = rated horsepower (normal)

The limiting cruising bmep is specified by the engine manufacturer and is usually about 140 bmep for larger engines. The bmep is determined from the following expression for any given engine rpm and bhp:

$$bmep = \frac{bhp \times 792000}{rpm \times piston displacement}$$
(2.502)

The minimum recommended oruising rpm is determined by speed limitations of the engine or accessories and is specified by the engine manufacturer; it will usually be between 1200 and 1500 rpm. The minimum power for level flight is determined in flight.

In all cruising operation the use of any but the lowest amount of supercharging should be avoided if possible. Airplanes equipped with turbosuperohargers should always be flown with the throttle butterfly wide open when using any turbo (except for what throttle is required for formation flying). Airplanes with gear driven supercharger should always be flown with the superchargers in low speed ratio up to the altitude at which the optimum indicated air speed may be obtained without exceeding the allowable rpm and manifold pressure for automatic lean mixture.

AFTR 6273

LEVEL FLIGHT DATA

Intercooler Flape 2 & Approximately 10,000' Intercooler Flape 5 & Approximately 20,000' Intercooler Flape 5 & Approximately 30,000' Military Power 2700 RPM 60 "Hg M R P 2550 RPM 50.5 "Hg



AFTE 6273





AFTR 6273

### SECTION 2.7

## Fuel Consumption

Fuel consumption is best measured as pounds per brake horsepower hour, specific fuel flow. From engine theory this is dependent upon mean effective pressure in the cylinder. For a simple engine, with all conditions ideal, power is a function of mean effective pressure, volume of the cylinder, and rpm so that for a given engine specific fuel consumption could be plotted against brake horsepower and rpm, as shown in Figure 2.71.



Figure 2.71 Typical Specific Fuel Consumption Curves for RFM and Horsepower

In practice the specific fuel consumption may also vary with altitude because of carburction, superchargers, ignition, and accessories. To run tests and make plots as in Figure 2.71 for many altitudes would be tedicus and is not required for flight test. The engine manufacturers and government agencies will run such extensive tests and determine a power schedule of mixtures, manifold pressures, rpm, and supercharger ratios for best engine operations. For flight test the only fuel consumption data generally required is that for powers obtained by operation on the normal schedules as shown in Figure 2.72. These points are run at the same time as speed power points so that later air miles per gallon and range can be computed.





As part of the general engine performance presentation a graph of standard gallons per hour at corresponding settings vs altitude is sometimes required from the specific fuel flow information and is presented as shown in Figure 2.73.





AFTR 6273

#### SECTION 2.8

## Engine Cooling

The hot strength of metals, and the temperature limits imposed by lubricants and detonation make it necessary to limit the operating temperature of engines and accessories. Army and Navy Aeronautical Specification, "Test <u>Procedures for Aircraft Power Plant Installation (AN-T-62</u>)," has set up a standardized instrumentation and test procedure. This specification should be studied when planning cooling tests.

There are two primary concerns in the cooling problem. These are whether the temperature limits can be maintained for all specified engine operating conditions and the effect on range of the various engine cooling configurations required to operate the engines within prescribed limits.

For determining the cooling requirements of a four-engine aircraft one engine is completely instrumented as specified in (AN-T-62). The remainder have less extensive instrumentation which is used as a cross check to correlate their operation with that of the completely instrumented engine. All temperatures except free air temporature and carburetor air temperature will be corrected to Army summer day conditions which are 25°C higher than NACA standard temperatures. The correction to summer day conditions is made to the indicated ambient temperature corrected for instrument error. This is done by adding the product of the correction factor, given in (AN-T-62), and the difference between the ambient temperature and the Army summer day standard temperature. In recording airflow pressure data each pressure difference may be determined in two ways; by direct measurement, or by subtracting the two values which were measured with reference to the air speed static prossure.

Temperatures and pressures thus determined allow computation of the engine cooling air flow which in turn makes possible a quantitative analysis of the cooling data for use in eliminating engine hot spots.

Figures 2.81, 2.82, and 2.83 are typical examples of how the basic data is presented graphically for the three cases usually considered; ground operation, climb and level flight.

Cooling data may be corrected to standard temperatures by solving equation 2.801

 $\Delta T = K (BHP)^{X} (\Delta P)^{a} (\sigma)^{b} (BSFC)^{c}$ 

(2.801)

where:

ΔT = engine temperature - ambient temperature
 K = constant to be determined
 BHP = brake horsepower
 ΔP = pressure drop through baffles
 σ = density ratio
 BSFC = brake specific fuel consumption
 x,a,b,o = exponents to be determined

APTR 6273

In solving for the exponent "c", level flight runs are made at about 10,000 feet pressure altitude in normal and rich mixture settings to determine the effect of mixture on cylinder head temperature. The rpm, bhp, air speed, cowl flap settings, oil cooler flap-settings, and altitude are held constant for both mixture settings. The data obtained at various rpms (1800, 2000, 2200, 2400, and 2550 rpm) from the 10,000 feet speed-power tests in normal and rich mixture are then used to determine the "c" exponent. A complete cylinder head temperature pattern is obtained on the Brown recorder on each run after the temperatures have become stabilized. The exponent "c" is the slope of the plot of log  $\Delta T$ versus log BSFC. The sign of "c" may change abruptly between manual lean and higher mixture settings.

A pressure survey is made at about 10,000 feet pressure altitude in normal mixture to determine the exponent "a" for the change in baffle pressure. This is accomplished by varying the cowl flap openings on the engine instrumented for cooling and <u>holding the horsepower</u>, rpm, air speed, oil cooler flaps, and <u>altitude constant</u>. Runs are made with the cowl flaps set in increments of 1-inch from full open to the setting resulting in the limit engine temperatures being obtained. On the runs at each 1-inch increments of cowl flap travel, the engine temperatures are stabilized and recorded on the Brown recorder. The power on the engines not being tested is varied to hold a constant air speed for all cowl flap positions at each power setting on the test engine. The exponent "a" is the slope of the plot of  $log\Delta T$  versus  $log\Delta P$ .

An altitude survey is obtained in conjunction with the speed-power data at approximately 5000, 10,000 and 15,000 foot altitudes to determine the effect of altitude on engine cooling. The stabilized temperatures obtained during the speed-power test in normal mixture at various rpm are then used to determine the exponent "b". The cowl flaps and oil cooler flaps are held constant at a setting that will give adequate cooling for all altitudes. The exponent "b" is obtained from the slope of the plot of  $\log \Delta T$  versus  $\log O$ . For determining "b" the same BHP's are used at each altitude.

In determining the exponent "x", the level flight data from other stabilized runs are used. Holding  $\Delta P$ ,  $\sigma$ , and BSFC constant, the slope of a plot of log  $\Delta T$  versus log BHP gives the value of the exponent "x". Test values of  $\Delta P$ ,  $\sigma$ , BSFC,  $\Delta T$ , and BHP are substituted in the basic equation and K determined. With this equation the engine cooling data can be corrected to a standard day temperature or to a hot day temperature; also, the amount of cowl flaps necessary for any engine power can be determined.

A more simplified method of cooling analysis for rough work can be made by plotting  $\Delta T$  for various engine components vs cowl flap setting. A  $\Delta V_1$  is also plotted on the same graph for a cowl flap drag increase reference. A typical plot of this type is shown in Figure 2.84. Plots of this type made at two speeds and altitudes are useful in determining the  $\Delta T$  ranges for various cowl flap or oil cooler control positions.

AFTR 6273



Figure 2.84 Cooling and Speed Loss Data for Various Macelle Flap Settings

AFTR 6273

B-000 USAF No. 00000 GROUND COOLING CHECK MIXTURE AR COME PLACES B" 64P INTERCOOLER D" 64P OIL COOLER 13.6" 64P



Ground Cooling Check

AFTR-6273



ግ ግ

to Altitude

APTR 6273

ί



AFTR 6273

2-30

# Power Calibration and Cooling Run




and and a reference

CHART 2,32





CHART 2.33

# SYMBOLS USED IN CHAPTER THREE

Symbol	Meaning
A	Area
Ac	Inlet capture area
A <sub>r</sub>	Ramp or compression surface area
A <sub>s</sub>	Area of ejector at primary nozzle exit
Aw	Projected ejector area
В	Mach number parameter, $1 + \frac{\gamma - 1}{2} M^2$
C <sub>da</sub>	Additive drag coefficient
Cdp	Cowl pressure drag coefficient
Cſ	Skin friction coefficient, or nozzle thrust coefficient
d	Boundary layer diverter height, or duct diameter
<b>T</b>	Hydraulic diameter
Da	Additive drag
Fe	Ram drag
Fg	Gross thrust
Fint	Intrinsic thrust
Fn	Net thrust
F <sub>post</sub>	Post-exit thrust
Fpre	Pre-entry thrust
g	Acceleration of gravity
к <sub>в</sub>	Loss coefficient in duct bend
1	Subsonic diffuser length
m	Mass flow
m1/m <sub>o</sub>	Inlet mass flow ratio = $A_0/A_c$
М	Mach number
N	Engine rotational speed
Pa	Atmospheric pressure
P.	Static pressure
Pt	Total pressure

3- i

Pw	Ejector wall static pressure
q	Dynamic pressure, 1/2 p V <sup>2</sup>
T	Average radius of curvature for duct bend
R	Universal gas constant (96.031 feet/°K)
Re	Reynold's number
Ta	Atmospheric temperature
T	Static temperature
Τ <sub>t</sub>	Total temperature
v	Velocity
V.	Secondary flow speed
vt	Airplane true speed
Wa	Engine airflow
<sup>w</sup> BL	Airflow through boundary layer bleed system
wg	Gas flow
w	Secondary airflow
-	1
x	Nossle pressure ratio parameter $\frac{P_{18}}{P_a}$ -1
¥	Angle of attack
., β. ¥	Exponents used in dimensional analysis
Ŷ	Ratio of specific heats
Y 8	Ratio of specific heats Boundary layer thickness or corrected pressure, Pa/PSL
•	Boundary layer thickness or corrected pressure,
δ	Boundary layer thickness or corrected pressure, Pa/PSL
5 7	Boundary layer thickness or corrected pressure, Pa/PSL Ram efficiency
6 ๆ ย	Boundary layer thickness or corrected pressure, P <sub>a</sub> /PSL Ram efficiency Corrected temperature
6 ๆ ย	Boundary layer thickness or corrected pressure, P <sub>a</sub> /P <sub>SL</sub> Ram efficiency Corrected temperature Flow angle relative to free stream direction
δ η θ λ π	Boundary layer thickness or corrected pressure, P <sub>a</sub> /P <sub>SL</sub> Ram efficiency Corrected temperature Flow angle relative to free stream direction Non-dimensional parameter

Subscripts:

0, 1, 2, etc.	Engine station designations
BL	Boundary layer
•	Exit
SL	Sea level
th	Throat

Superscripts:

(

•	Sonic flow conditions
T	Conditions downstream of normal shock wave

3**- iii** 

#### CHAPTER THREE

### SECTION 1

### INTRODUCTION TO THRUST MEASUREMENT

#### 1.1 PRELIMINARY COMMENTS

The turbojet engine performs a function similar to that of the reciprocating engine with a propeller. With either system thrust is produced in the same manner; that is, by accelerating a mass of air. The difference in the operation of the two systems lies in the volume and velocity of the air or gases affected. The propeller moves a comparatively large volume of air rearward at a relatively low velocity, while the turbojet engine takes in a smaller volume of air, expands it with burning fuel, and expels it to the rear at a high velocity.

The static thrust of a turbojet engine can be determined readily by direct mechanical measurement on the ground. This measurement may be made with strain gages, spring balance, etc., either with the engine installed in an airplane mounted on a thrust atand or with the bare engine located in a test ceil. It has been found that even with seemingly identical production turbojet engines, there is an appreciable difference in thrust output. Also, there is a gradual loss of thrust as operating hours are accumulated on an engine. For this reason the static thrust of engines in aircraft undergoing performance testing should be measured periodically.

Thrust measurement in flight becomes considerably more difficult than under static conditions. No satisfactory means of determining thrust by mechanical means, such as strain gages installed at the engine mounts, has been found. Approximate thrust data can be obtained from the engine manufacturer's estimated performance curves. This method is not satisfactory for flight test purposes, however, These curves are based on estimates, or an average engine and do not yield sufficiently accurate thrust data because of variations in output between engines. A useful application of these curves is in making corrections to standard conditions as described in Section 4. In this

case a high degree of accuracy is not required since the amount of the corrections is usually small compared to the total values.

### 1.2 GENERAL ANALYSIS OF IN-FLIGHT THRUST MEASUREMENT

It is convenient to consider first a simple ducted body in order to define the thrust developed by a turbojet engine. For simplification, the axis of the body is made parallel to the flight path and no mixing of internal and external flow downstream of the nozzle exit is considered (reference Figure 1.1). It is indicated from the momentum theorem that the thrust developed is equal to the rate of change of total momentum (pressure plus momentum flux) of the internal fluid contained within the stream tubes ahead of and behind the body as well as that within the body.



Figure 1.1 Flow Through a Simple Ducted Body

Assuming a uniform velocity distribution, the thrust at any arbitrary plane (P) perpendicular to the flight path can be expressed as

$$\int V \cos \lambda \, dm + \int (P_{e_p} - P_e) \, dA_p \qquad 1.1$$

where

V

= fluid velocity

- $\lambda$  = inclination of streamline to free-stream direction
- m = mass flow
- $P_{s_n}$  = static pressure at arbitrary plane (P)
- Pa = ambient pressure
- $A_p$  = area containing the internal flow at arbitrary plane (P)

### **1.3 DEVELOPMENT OF THRUST DEFINITIONS**

### 1.3.1 Net Thrust:

The fundamental definition of the net thrust of a turbojet engine is considered equal to the change of total momentum of the internal fluid between station 0 upstream and station W downstream. Station 0 is sufficiently far upstream that the boundaries of the preentry stream tube are parallel to the direction of undisturbed flow and the static pressure in the stream tube is the same as ambient pressure. Similarly, station W is located downstream where pressure disturbances resulting from the passage of the body through the air have disappeared and the static pressure is again ambient. From continuity, m = p VA and equation 1.1 may be re-stated as

$$\mathbf{F}_{n} = \int \mathbf{\rho}_{w} \, \mathbf{V}_{w}^{2} \, \mathrm{d}\mathbf{A}_{w} - \int \mathbf{\rho}_{o} \, \mathbf{V}_{o}^{2} \, \mathrm{d}\mathbf{A}_{o} \qquad 1.2$$

 $F_n$  is the net thrust in the upstream direction created by the internal flow within the stream tubes extending both upstream and downstream of the body.

1.3.2 Intrinsic Thrust:

To define the thrust produced within the body, reference stations are taken at the entry (station 1) and the exit (station 2). Consideration of the momentum theorem permits equating the rate of change of momentum to the sum of the pressure and friction forces acting on the fluid at the boundaries. Referencing pressures to ambient,

$$\int \left[ (P_{sint} - P_{a}) \sin \lambda - F \cos \lambda \right] ds = \int \rho_2 V_2^2 \cos^2 \lambda_2 dA_2 + \int (P_{s2} - P_{a}) dA_2$$
$$-\int \rho_1 V_1^2 \cos^2 \lambda_1 dA_1 - \int (P_{s1} - P_{a}) dA_1 \qquad 1.3$$

where

Psint= internal static pressureF= local friction force per unit areads= element of area of internal surface

The left-hand side of the equation represents the force in the free-stream direction exerted on the fluid by the internal surface of the duct. This force is equal to the rate of  $\operatorname{chan}_{\operatorname{SC}}$  of total momentum appearing on the right-hand side of the equation which is equivalent to a force in the free-stream direction exerted by the fluid on the internal surface of the duct. This latter quantity is defined as the intrinsic thrust.

$$F_{int} = \int (\rho_2 V_2^2 \cos^2 \lambda_2 + P_{s_2} - P_a) dA_2 - \int (\rho_1 V_1^2 \cos^2 \lambda_1 + P_{s_1} - P_a) dA_1$$
1.4

### 1.3.3 Pre-Entry Thrust and Ram Drag:

A similar analysis can be made of the pre-entry stream tube. To add physical meaning, the diverging portion of the stream tube can be considered as replaced by a thin frictionless membrane. Since the flow field is unchanged, the thrust will not be affected. With reference stations 0 and 1, the force exerted by the fluid due to pressure acting on the interior of the stream tube becomes

$$\int (\mathbf{P}_{\mathbf{s}_{ext}} - \mathbf{P}_{\mathbf{a}}) \sin \lambda \, \mathrm{d}\mathbf{s} \qquad 1.5$$

which is commonly known as additive drag.

where

P<sub>saxt</sub> \* external static pressure

As before this force may be set equal to

$$\int (\rho_1 V_1^2 \cos^2 \lambda_1 + P_{s_1} - P_s) dA_1 - \int \rho_0 V_0^2 dA_0 = 1.6$$

which is defined as pre-entry thrust. Since  $V_0$  is uniform,  $m_1 = \rho_0 V_0 A_0$  and  $\cos \lambda_0 = 1$ , the pre-entry thrust is

$$F_{pre} = (\rho_1 V_1^2 cos^2 \lambda_1 + P_{s_1} - P_s) dA_1 - m_1 V_0 = 1.7$$

The term  $m_1 V_0$  in the preceding equation is the ram drag,  $(F_0)$ .

$$\mathbf{F}_{\mathbf{0}} = \mathbf{m}_{1} \mathbf{V}_{\mathbf{0}}$$
 1.8





### 1.3.4 Post-Exit Thrust:

Similarly, reference stations 2 and W may be chosen and a thrust derived which is defined as the post-exit thrust.

$$\mathbf{F}_{\text{post}} = \int P_{w} V_{w}^{2} dA_{w} - \int (\rho_{2} V_{2}^{2} \cos^{2} \lambda_{2} + P_{s2} - P_{s}) dA_{2} \qquad 1.9$$

## 1.3.5 Standard Net Thrust:

Since net thrust is defined considering flow between stations 0 and W, the net thrust is equal to the sum of the pre-entry thrust, intrinsic thrust and post-exit thrust.

$$F_n = F_{pre} + F_{int} + F_{post}$$
 1.10

While the fundamental definition of thrust has been based on the flow between stations 0 and W, the calculation of post-exit thrust cannot be made precisely because of mixing of internal and external flows downstream of the exit. If the pressure surrounding the post-exit stream tube is assumed equal to ambient, the post-exit contribution to thrust becomes zero. This assumption is made to define standard net thrust which is presented in engine specifications by the engine manufacturers.

$$F_{n_{st'd}} = F_{pre} + F_{int} = \int (\rho_2 V_2^2 \cos^2 \lambda_2 + P_{s2} - P_s) dA_2 - m_1 V_0$$
 1.11

Practical application of the definitions which have been derived in this section is treated in detail in Section 6.

#### SECTION 2

#### TURBOJET ENGINE PERFORMANCE PARAMETERS

#### 2.1 INTRODUCTION

The number of variables which affect the performance of a turbojet engine is quite large. Fortunately its operation is such that the turbojet engine may be submitted to an extensive analytical treatment. Performance characteristics are put in a conveniently usable form by grouping these many dimensional variables into non-dimensional similarity parameters. Advantages of using these non-dimensional parameters are:

> 1. Better control of these parameters is achieved than can be obtained with the original variables.

2. There are fewer parameters than there were variables so that they can be presented and understood more readily.

3. Fewer test points are required to present the complete performance capabilities of an engine throughout its operating range.

Dimensionless parameters can be determined by the dimensional analysis methods outlined in the following paragraphs. It is emphasized that there are many sets of independent dimensionless parameters which can be formed from a given set of independent variables, and judgment must be exercised in selecting parameters which have the proper significance.

#### 2.2 APPLICATION OF DIMENSIONAL ANALYSIS

The traditional method of applying dimensional analysis is by means of the Buckingham  $\pi$  Theorem. From this theorem it is learned that if in a given problem there are n independent variables (dimensional) and k basic dimensions, (e.g., length, time and mass), then there only (n - k) truly independent non-dimensional (similarity) parameters associated with the problem. Buckingham gives these nondimensional parameters the symbolic notation  $\pi_1$ ,  $\pi_2$ ...  $\pi_{n-k}$ . A method for determining the form of these parameters is described below.

Each of the parameters determined by dimensional analysis will be composed of the product of variables raised to some power. The determination of these exponents is the central problem of dimensional analysis. One variable in each parameter can be chosen arbitrarily to be raised to the first power. As a matter of convenience these first power terms are made those of primary interest (e.g., drag, lift, fuel flow, etc.). Symbolically, the pi-parameters are written

$$\pi_{1} = (V_{1}^{\alpha_{1}}, V_{2}^{\alpha_{2}}, \dots, V_{k}^{\alpha_{k}}) V_{k} + 1$$
  
$$\pi_{2} = (V_{1}^{\beta_{1}}, V_{2}^{\beta_{2}}, \dots, V_{k}^{\beta_{k}}) V_{k} + 2$$
  
$$\pi_{n} = (V_{1}^{\beta_{1}}, V_{2}^{\beta_{2}}, \dots, V_{3}^{\beta_{3}}) V_{k} + n$$

where

 $\pi_1 \dots \pi_n$  are the dimensionless parameters  $V_{k+1} \dots V_{k+n}$  are the variables of primary interest  $a_{in} \beta_i \dots \xi_i$  are the exponents to be decormined

The exponents are obtained by replacing the variables with their fundamental dimensions of mass, M, length, L, and time, T.

$$\begin{bmatrix} * \\ 1 \end{bmatrix} = \begin{bmatrix} M^{a_1} L^{b_1} T^{c_1} \end{bmatrix}^{a_1} \begin{bmatrix} M^{a_2} L^{b_2} T^{c_2} \end{bmatrix}^{a_2} \begin{bmatrix} M^{a_3} L^{b_3} T^{c_3} \end{bmatrix}^{a_3} \begin{bmatrix} M^{a_4} L^{b_4} T^{c_4} \end{bmatrix}$$

a, bi, ci are known numbers, for example,

$$\begin{bmatrix} Velocity \end{bmatrix} = \begin{bmatrix} M^{\circ} L^{1} T^{-1} \end{bmatrix}$$

The variables are combined into terms having the dimensions  $\begin{bmatrix} M^{O} L^{O} T^{O} \end{bmatrix} \text{ so we may write}$   $\begin{bmatrix} x \end{bmatrix} = M^{O}L^{O}T^{O} = 1 = M^{a} a_{1} + a_{2}a_{2} + a_{3}a_{3} + a_{4} L^{b}a_{1} + b_{2}a_{2} + b_{3}a_{3} + b_{4}$   $T^{c}a_{3} + c_{2}a_{3} + c_{3}a_{3} + c_{4}$ 

Now equating exponents of like terms

for M  $a_1a_1 + a_2a_2 + a_3a_3 + a_4 = 0$ for L  $b_1a_1 + b_2a_2 + b_3a_3 + b_4 = 0$ for T  $c_1a_1 + c_2a_2 + b_3a_3 + c_4 = 0$ 

These are three simultaneous equations which can be used to determine the three unknowns,  $a_1$ ,  $a_2$ , and  $a_3$ .

If the factors affecting the performance of a turbojet engine are divided into dependent and independent variables, we may list them as in the following table:

Dependent Variable	Units	Dimensions
airflow, wa	lb(mase)/sec	M T <sup>-1</sup>
fuel flow, wf	ft lb/sec	$ML^2T^{-3}$
exhaust gas temperature, T <sub>t5</sub>	•K	L <sup>2</sup> T <sup>-2</sup>
thrust, Fg or Fn	lb	MLT -2
Independent Variable	Units	Dimensions
inlet total pressure, P <sub>t2</sub>	ib/ft <sup>2</sup>	$ML^{-1}T^{-2}$
inlet total temperature, Tt2	•K	$L^2T^{-2}$
engine speed, N	rad/sec	<b>T</b> <sup>-1</sup>
free-stream static pressure, Pa	1b/ft <sup>2</sup>	ML <sup>-1</sup> T <sup>-2</sup>
nozzle area, Ag	ft <sup>2</sup>	L <sup>2</sup>
flight speed, V	ft/sec	LT <sup>-1</sup>





Figure 2.1

Turbojet Engine Station Designations

## The above station designations are generally used as subscripts:

0	free stream
1	inlet duct
2	compressor inlet
3	compressor outlet
4	turbine inlet
5	turbine outlet

-

6	tailpipe inlet
7	tailpipe outlet
8	jet nozzle outlet

As an example of the application of dimensional analysis, consider thrust as a function of the independent variables taken from the preceding table.

 $F_{g} = f(P_{t2}, T_{t2}, N, P_{a}, A_{8})$ 

Since there are six variables and three fundamental units, we can determine three pi-parameters, which can be expressed in an equation made up of three dimensionless numbers.

$$\pi_{1} = P_{t_{2}}, T_{t_{2}}, A_{8}, F_{g}$$

$$\pi_{2} = P_{t_{2}}, T_{t_{2}}, A_{8}, N$$

$$\pi_{3} = P_{t_{2}}, T_{t_{2}}, A_{8}, P_{a}$$

Substituting their dimensions in place of the variables,

$$\begin{bmatrix} \pi_{1} \end{bmatrix} = \begin{bmatrix} ML^{-1}T^{-2} \end{bmatrix}^{\alpha_{1}} \begin{bmatrix} L^{2}T^{-2} \end{bmatrix}^{\alpha_{2}} \begin{bmatrix} L^{2} \end{bmatrix}^{\alpha_{3}} \begin{bmatrix} MLT^{-2} \end{bmatrix}$$
$$\begin{bmatrix} \pi_{2} \end{bmatrix} = \begin{bmatrix} ML^{-1}T^{-2} \end{bmatrix}^{\alpha_{1}} \begin{bmatrix} L^{2}T^{-2} \end{bmatrix}^{\alpha_{2}} \begin{bmatrix} L^{2} \end{bmatrix}^{\alpha_{3}} \begin{bmatrix} T^{-1} \end{bmatrix}$$
$$\begin{bmatrix} \pi_{3} \end{bmatrix} = \begin{bmatrix} ML^{-1}T^{-2} \end{bmatrix}^{\alpha_{1}} \begin{bmatrix} L^{2}T^{-2} \end{bmatrix}^{\alpha_{2}} \begin{bmatrix} L^{2} \end{bmatrix}^{\alpha_{3}} \begin{bmatrix} ML^{-1}T^{-2} \end{bmatrix}$$

After solving for exponents we have,

$$\pi_1 = \frac{F_g}{A_8 P_{t_2}}$$

$$\pi_2 = \frac{N t \overline{A}_R}{T_{t_2}}$$
$$\pi_3 = \frac{P_n}{P_{t_2}}$$

Therefore,

$$\frac{\mathbf{F}_{g}}{\mathbf{A}_{g}\mathbf{P}_{t_{2}}} = i\left(\frac{\mathbf{N}\sqrt{\mathbf{A}_{g}}}{\sqrt{\mathbf{T}_{t_{2}}}}, \frac{\mathbf{P}_{a}}{\mathbf{P}_{t_{2}}}\right)$$

Eliminating the area (Ag) for an engine of constant size and inverting  $\pi_3$  to form ram pressure ratio,

$$\frac{\mathbf{F}_{g}}{\mathbf{P}_{t_{2}}} = f\left(\frac{\mathbf{N}}{\mathbf{F}_{t_{2}}}, \frac{\mathbf{P}_{t_{2}}}{\mathbf{P}_{a}}\right)$$

It is conventional to refer temperatures and pressures to standard sea level conditions by making the following substitutions,

. .

$$\delta_{t_2} \left( \frac{P_{t_2}}{P_{aSL}} \right)$$
 for  $P_{t_2}$ 

and

$$\theta_{t2} \left( \frac{T_{t2}}{T_{agL}} \right)$$
 for  $T_{t2}$ 

we have,

$$\frac{\mathbf{r}_{\mathbf{t}}}{\mathbf{s}_{\mathbf{t}_2}} = f\left(\frac{\mathbf{N}}{\mathbf{f}\mathbf{s}_{\mathbf{t}_2}}, \frac{\mathbf{P}_{\mathbf{t}_2}}{\mathbf{P}_{\mathbf{a}}}\right)$$

By similar analysis the following relationships can be developed.

$$\frac{w_{a}\left(\overline{\theta}_{t2}\right)}{\delta_{t2}} = f\left(\frac{N}{\sqrt{\theta}t_{2}}, \frac{P_{t2}}{P_{a}}\right)$$

$$\frac{w_{f}}{\delta_{t2}\sqrt{\theta}t_{2}} = f\left(\frac{N}{\sqrt{\theta}t_{2}}, \frac{P_{t2}}{P_{a}}\right)$$

$$\frac{T_{tk}}{\theta_{t2}} = f\left(\frac{N}{\sqrt{\theta}t_{2}}, \frac{P_{t2}}{P_{a}}\right)$$

These equations remain valid with the addition or deletion of constants although the parameters are no longer dimensionless.

Thus far, we have considered only the independent variables which have a primary effect on performance. Other factors, such as viscous effects, combustion efficiency, and the ratio of specific heats have secondary effects on performance particularly at high altitudes and high Mach numbers. It is pointed out that engine manufacturers frequently publish correction curves to be used in conjunction with non-dimensional performance plots which account for these secondary effects.

# SECTION 3 AIR INDUCTION SYSTEM PERFORMANCE

### 3.1 GENERAL COMMENTS

Performance curves for an engine installed in an aircraft are usually presented in terms of conditions existing at the compressor face, i.e.,  $T_{t_2}/T_{t_0}$  and  $P_{t_2}/P_{t_0}$ . Adiabatic flow is assumed so that  $T_{t_2} = T_{t_0}$  which can be calculated from the free-stream Mach number and ambient temperatures. In order to determine  $P_{t_2}$  it is necessary to evaluate the total pressure losses between the free-stream, station "O", and the compressor face, station "2". When these losses are evaluated, the engine performance curves may be obtained in terms of aircraft speed, altitude and free-air temperature.

Air-inlet efficiency is generally expressed in terms of the pressure recovery,  $P_{t2}/P_{to}$ , because it can be shown by a simplified analysis of the turbojet engine that this parameter is directly related to both the net thrust and fuel consumption. For example,

$$\frac{F_{n_{i}} - F_{n_{a}}}{F_{n_{i}}} = L \left(1 - \frac{P_{t_{2}}}{P_{t_{0}}}\right) \qquad 3.1$$

where

 $\mathbf{F}_{n_i}$  = ideal net thrust,  $(\mathbf{P}_{t_2}/\mathbf{P}_{t_0} = 1.0)$ 

 $F_{\text{Ra}}$  = actual net thrust

Pto = free-stream total pressure

Ptz = total pressure at compressor face

L = a function of jet efficiency, nozzle exit conditions L is always greater than 1.0 and is determined by engine design and flight conditions. If the thrust loss is determined for an engine, values of the parameter  $(\Delta F_n/F_{nj})/(\Delta P_{t2}/P_{t_0})$  are usually between 1.2 and 1.8. Hence, a 1 percent loss in pressure recovery may result in a 1.8 percent loss in net thrust.

The parameter ram efficiency,  $\eta$ , is sometimes used to indicate duct losses.

$$\eta = \frac{P_{t2} - P_a}{P_{t_0} - P_a} \qquad 3.2$$

where  $P_a$  is free-stream static pressure.

Experience has proven that  $\eta$  is a useful parameter for expressing the inlet losses in subsonic flow. A conversion between ram efficiency,  $\eta$ , and total pressure recovery,  $P_{t2}/P_{t_0}$ , can be expressed as

$$\frac{P_{12}}{P_{10}} = \frac{1 + \eta \left[ \left(1 + \frac{\gamma - 1}{2} - M_0^2\right)^{\frac{1}{\gamma - 1}} - 1 \right]}{\left(1 + \frac{\gamma - 1}{2} - M_0^2\right)^{\frac{\gamma}{\gamma - 1}}} \qquad 3.3$$

If Y = 1.4 and  $B = 1 + \frac{Y-1}{2} M^2$  this becomes

$$\frac{P_{t2}}{P_{t_0}} = \frac{1 + \eta (B_0^{3.5} - 1)}{B_0^{3.5}} \qquad 3.4$$

This expression is plotted in Chart 9.1.

In the discussion which follows, some of the factors which affect inlet efficiency and approximate methods for estimating the inlet total pressure losses (providing experimental data are not available) are discussed.

### 3.2 SUBSONIC FLIGHT

Inlet total pressure losses which occur during take-off (or static run-up) and at subsonic flight speeds may be conveniently considered as (a) inlet-entry losses and (b) subsonic-diffuser losses.

## 3.2.1 Inlet Entry Losses:

Entry losses occur mainly from flow separation at the inlet lips or from the ingestion of boundary-layer flow in the inlet. Inlet total-pressure losses caused by flow interference from aircraft components other than the air induction system are small at subsonic speeds and are usually neglected. Lip losses are important for those conditions where the local stagnation-point streamline occurs outside of the inlet lip (i.e., angle of attack operation with sharp lipped inlets, or for mass flow ratios greater than 1.0 as illustrated in Figure 3.1c).



#### Figure 3.1

Schematic Representation of Mass Flow Ratio

For low speeds and large mass flow ratios, lip separation occurs and a vena contracta is formed within the inlet giving rise to relatively high total pressure losses. Since aircraft designed for supersonic flight generally have sharp inlet lips, the separation condition is aggravated and these aircraft suffer large total pressure losses during static run-up and take-off. The curves in Chart 9.2 give the average pressure recovery,  $(P_{t1}/P_{t0})_{L,IP}$ , for a number of model and full scale inlets at zero and low forward speeds. In order to increase these poor low-speed pressure recoveries and provide an adequate air supply for the engine, some aircraft designed for high-speed flight have auxiliary inlets or "blow in" doors. These auxiliary inlets reduce the operating mass flow of the main inlet and improve the overall pressure recovery.

Boundary-layer flow may enter the inlet for a side inlet installation and give rise to a loss in pressure recovery over a portion of the inlet area because of the local velocity profile. If experimental data is not available on losses due to the boundary-layer effects, an estimate of these losses may be made from the following considerations.

For most installations the entering boundary layer is considered to be turbulent. Boundary-layer thickness,  $\delta$ , may be estimated for  $\alpha = 0$  degrees by use of Chart 9.3\*. Angle of attack effects on boundary-layer thickness may be estimated from the following equation for small angles of attack:

> $\delta_{q} = \delta_{0} \left( \frac{\delta_{q} - \delta_{0}}{\delta_{0}} \right) + \delta_{0} \qquad 3.5$  $\delta_{0} = \text{boundary-layer thickness at ex} = 0$  $\delta_{r} = \text{boundary-layer thickness at angle of attack of ext}$

The inlet is located on the nose portion of the fuselage (i.e., in a flow field more conical in nature than two-dimensional) these boundary-layer thicknesses should be modified by the Mangler transformation factor  $\sqrt[1]{\sqrt{3}}$ . For example,

$$\delta$$
 conical =  $\frac{1}{\sqrt{3}}$   $\delta$  flat plate

where

and

It should be noted that equation 3.5 is applicable only for underslung inlet locations and should be used with caution for other circumferential locations because considerable error may result. (This is especially true at supersonic speeds as will be pointed out later.)

Most side-inlet installations have a boundary-layer diverter, or scoop, such as that schematically shown below.



### Figure 3.2

Boundary Layer in Side-inlet Installation

The effects of boundary-layer profile on the inlet losses may be calculated in the absence of shock wave boundary-layer interaction for this type of installation with the aid of Chart 9.4\* and the following equation for the average value of pressure recovery.

\*Note that for most cases a turbulent boundary layer is assumed and it is sufficiently accurate to use a 1/7 power velocity profile approximation.

$$\left(\frac{P_{t}}{P_{t_{0}}}\right)_{avg} = \frac{P_{t_{1}}}{P_{t_{0}}} \int_{d/\delta}^{1.0} \frac{A_{BL}}{A_{0}} + \frac{P_{t_{1}}}{P_{t_{0}}} \frac{A_{0} - A_{BL}}{A_{0}} 3.6$$
where  $P_{t_{1}}/P_{t_{0}} \int_{d/\delta}^{1.0}$  denotes the average pressure recovery of that

part of the boundary layer which is ingested into the duct. (Note that if  $d/\delta \ge 1.0$  no correction is required and  $A_{BL} = 0.$ ) In many cases  $P_{t1}/P_{t_0}$  is approximately 1.0; however, this term must be evaluated at the same station at which the reference area  $A_0$  is taken. (Reference Figure 3.2.)

#### 3. 2. 2 Internal Boundary-Layer Removal Systems:

An aircraft inlet designed for high-speed flight may have an internal boundary-layer removal system. Losses are incurred in the use of such a system, whether a ram scoop or suction through slots or perforations. The boundary-layer removal system losses are usually so large that a complete loss in free-stream momentum is usually assumed for the mass flow through the removal system. For example,

$$D_{BL} = m_{BL} V_o$$

3.7

where

D<sub>BL</sub> = boundary-layer removal system drag m<sub>BL</sub> = boundary-layer removal system mass flow

If the manufacturer does not specify the mass flow, the loss with a ram scoop may be estimated as follows with the aid of Chart 9.5. (Note that a  $\frac{1}{7}$  power velocity profile assumption is adequate for this estimate.) Since the drag, D<sub>BL</sub>, is the change in momentum of the airflow in the direction of flow then,

$$D_{BL} = \phi_0 - \phi_1$$
$$= (1 - \frac{\phi_1}{\phi_0}) \qquad 3.8$$

where the momentum ratio,  $\phi_1/\phi_0$ , may be obtained from Chart 9.5 and  $\phi_0$  from the following relation.

$$P_0 = Y P_a M_o^2 A_{BL} \qquad 3.9$$

#### 3.2.3 Subsonic Diffuser Losses:

Subsonic diffuser losses are concerned with those losses within the inlet between stations 1 and 2. Factors which contribute to these losses are skin friction, duct expansion and duct bends or offsets. Total pressure losses resulting from friction and duct expansion can be calculated if one-dimensional compressible flow and no change in skin friction coefficient with length are assumed. The results of such a calculation are shown in Chart 9.6 where the skin friction coefficient  $C_f$  is usually estimated with sufficient accuracy from Von Karman's approximate formula for turbulent flow.

$$C_{f} = \frac{.074}{(R_{e})^{1/5}}$$
 3.10

R<sub>e</sub> = Reynolds number based on average flow properties in the duct and the duct length.

Values of the parameter  $(\frac{1}{d_2} \times \frac{Cf}{.003})$  between those given in Chart 9.6 can be obtained with sufficient accuracy by linear interpolation. Here  $d_2$  is the hydraulic diameter of the duct at the compressor-face station. Total pressure losses due to compound duct bends may be estimated with sufficient accuracy by use of Chart 9.7. In this figure the loss coefficient, KB, is related to the duct total pressure loss through the following relation.

$$\frac{P_{t_2}}{P_{t_1}} = 1 - K_B \left(1 - \frac{1}{B_1^{3.5}}\right) \qquad 3.11$$

where

$$B_1 = 1 + \frac{Y_{-1}}{2} M_1^2$$

Even though Chart 9.7 is plotted for 90-degree bends the loss coefficients for bends other than 90 degrees may be obtained by simple interpolation, (e.g., for 45-degree bends the loss coefficient is reduced by about  $\frac{1}{2}$ ). All bends must be generously radiused, however, so that boundary-layer separation does not occur.

The total pressure losses for the entire inlet in subsonic flow may now be expressed as follows:

$$\frac{P_{t2}}{P_{t_0}} = \frac{P_{t1}}{P_{t_0}} \times \frac{P_{t2}}{P_{t_1}}$$
 3.12

where

$$\frac{P_{t_1}}{P_{t_0}} = 1 - \left[ \left( 1 - \frac{P_{t_1}}{P_{t_0}} \right)_{BL} + \left( 1 - \frac{P_{t_1}}{P_{t_0}} \right)_{LIP} \right] \qquad 3.13$$

$$\frac{P_{t2}}{P_{t1}} = 1 - \left[ \left(1 - \frac{P_{t2}}{P_{t1}}\right)_{\text{friction}} + \left(1 - \frac{P_{t2}}{P_{t1}}\right)_{\text{bends}} \right] 3.14$$

### 3.3 SUPERSONIC FLIGHT

Inlet total pressure losses which occur during supersonic flight may be calculated from a consideration of the pressure losses associated with the following:

1. Supersonic compression.

2. Entering stream-tube flow non-uniformities (i.e., boundarylayer flow, aircraft attitude effects, interferences effects, etc)

and unsteadiness.

3. Subsonic diffuser design.

The overall performance of the inlet may then be calculated in exactly the same manner as that given in equations 3.12 through 3.14 provided that all total pressure loss factors are included.

### 3.3.1 Supersonic Compression:

Total pressure losses die to supersonic compression may be calculated from the geometry of the inlet and free-stream Mach number. Theoretical pressure recoveries of normal shock, twoshock and three-shock external compression inlets are illustrated in Figure 3.3.





Maximum Pressure Recovery, All External Compression

From this figure it may be seen that even the relatively simple two-shock external compression inlet exhibits considerable pressure advantage over the normal shock inlet at Mach numbers over about 1.5. At Mach 2.0, for example, the two-shock inlet operates with a 20 percent improvement in pressure recovery. This may result in approximately 40 percent more thrust for a two-shock inlet than for one using a normal shock inlet. At Mach numbers much above 2.0 even the two-shock inlet yields excessive pressure losses and the more complicated flow geometries appear advantageous. For open nose inlets the total pressure ratio,  $P_{t1}/P_{t_0}$ , may be obtained directly from the free-stream Mach number and the normal shock pressure ratio given on Chart 9.8. Pressure ratios obtained with other types of inlets are shown in Charts 9.9 through 9.11.

An additional loss arises from the momentum change in the inlet stream-tube between the free-stream and the inlet face when no aircraft components other than those of the air induction system interfere with the stream-tube. This momentum loss occurs at mass flow ratios less than one and is commonly called "pre-entry thrust" or "additive drag". As defined in Section 1, pre-entry thrust is the axial component of the pressure force on the diverging portion of the entering stream-tube between station 0 and station 1.

Expressions for the calculations of additive drag have been developed in coefficient form,  $(C_{da} = D_a/q_0A_c)$ . For open nose inlets,

$$C_{d_{a}} = \frac{2}{M_{o}}^{2} \left[ \frac{B_{o}}{B_{1}^{3.5}} + \frac{P_{t_{1}}}{P_{t_{o}}} (Y M_{1}^{2} + 1) - 1 - \frac{A_{o}}{A_{1}} Y M_{o}^{2} \right] 3.15$$

For external compression inlets,

$$C_{da} = \frac{2}{\gamma M_0^2} \left[ \frac{A_1}{A_c} \frac{B_0^{3.5}}{B_1^{3.5}} \frac{P_{11}}{P_{t_0}} (\gamma M_1^2 + 1) \cos \lambda + \frac{A_r}{A_c} \frac{P_r}{P_{s_0}} - 1 - \frac{A_0}{A_c} \gamma M_0^2 \right] + C_f \frac{A_r}{A_c} 3.16$$

where

>

- $C_{da}$  = additive drag coefficient
- qo = free-stream dynamic pressure
- A<sub>c</sub> = inlet capture area
- $A_r = ramp$  or other compression surface area
- Pr = effective static pressure on compression surface forward of station 1

The results obtained from these equations are plotted in Charts 9.14 and 9.15. Also shown in Chart 9.15 is the variation of cowl position parameter (angle between axis of inlet and straight line connecting tip of center body with lip of cowl),  $\sigma_g$ , with mass flow. This parameter is useful for the determination of the maximum massflow ratio obtainable through a given conical inlet for a particular test condition.

Variations in inlet drag resulting from changes in mass flow through the inlet will cause changes in the cowl-lip suction force as well as additive drag. At subsonic speeds these two forces cancel each other and no calculation for either is necessary. However, at speeds just above sonic both forces must be calculated for an accurate determination of inlet net drag. At Mach numbers over about 2.0 additive drag becomes the dominating factor and lip suction forces are small (usually negligible) for slender, sharp-lipped inlets<sup>‡</sup>. For

<sup>\*</sup>Slender sharp-lipped inlets are defined as inlets with cowl angles less than 5 degrees and by thickness with the ratio lip thickness/inlet radius less than about 0.07.

nose inlet installations Chart 9.16 may be used for estimating lip suction effects at mass ratios greater than about 0.8. Included in this figure is the corresponding increase in additive drag coefficient to illustrate the relative magnitude of the two forces. For blunt-lipped installations and large cowl angles, experimental results are required to determine lip suction effects.

### 3.3.2 Flow Non-Uniformities and Unsteadiness Effects:

The major total pressure losses resulting from flow nonuniformity of the inlet face are caused by an entering boundary layer. These total pressure losses may be treated in the same manner discussed previously for subsonic flight; however, the correction for angle of attack effects on boundary-layer thickness must be obtained from experimental data for inlet locations other than the underslung type. This is necessary because experimental results have shown as much as 15 to 20 percent low in  $P_{tZ}/P_{t_0}$  with variations in circumferential position of the inlet and angles of attack as low as 4 to 6 degrees.

Excessive airflow distortion at the compressor face may result from operation at "off-design" conditions. Inlet performance may be degraded from Mach number and altitude effects as well as from subcritical or supercritical operation. If shock wave boundary-layer separation occurs this condition may be considerably aggravated. With supercritical operation as shown in Figure 3.4C the pressure drop across the normal shock is increased and a lower pressure recovery results. Subcritical flow as in Figure 3.4A is accompanied by a reduction in mass-flow ratio with a consequent increase in additive drag. If the reduction in mass flow ratio is too great, "inlet buzz", discussed in the following paragraph, will occur.



Figure 3.4 Operational Modes of Supersonic External Compression Inlets

Large variations in flow uniformity will give rise to thrust loss, excessive fuel consumption, loss of acceleration margin, hot spots, local blade stalling (rotating compressor stall) and engine vibration with possible structural failure.

Inlet operation at subcritical mass-flow ratios may result in an unsteady flow condition commonly called "inlet buzz". This operation is characterized by rapid changes in the inlet flow pattern which results in rapid fluctuations in in \_\_\_\_\_\_ drag as well as total-pressure

\_\_\_\_\_

4

ratio. Severely reduced engine performance results and for some buzz conditions compressor stall, flame-out, or structural failure may occur.

Flow non-uniformity may also arise from interference effects of other aircraft components on the inlet stream-tube and these effects must be determined from experimental data because no simple means for estimating the magnitudes of these effects exists for all the diverse combinations of flight attitudes and aircraft geometries.

### 3.3.3 Subsonic Diffuser Losses:

Subsonic diffuser total pressure losses are calculated in a manner similar to that described in paragraph 3.2.3. The entering flow conditions are taken as those just down-stream of the terminal shock and the length of the subsonic diffuser measured from this point.

#### **SECTION 4**

## STANDARDIZATION OF TEST DATA WITH ENGINE PARAMETERS

#### 4.1 INTRODUCTION

In the process of standardizing test data obtained under off-standard conditions, it is necessary that corrections be made based on the engine parameters developed in Section 2. For example, rate of climb determined during climb tests may be corrected to standard engine speed and standard temperature through the use of correction curves plotted as  $F_n/\delta_{t2}$  versus N/ $\sqrt[4]{\theta_a}$ . These curves are computed from the engine manufacturer's estimated minimum performance curves, as described in Data Reduction Outline 8.1. (Typical estimated minimum performance curves for an engine with a fixed nozzle are illustrated in Figure 4.1.) This correction and others using non-dimensional parameters are made quite readily for aircraft with simple jet engines but are not generally applicable to more advanced engines. Use of the fuel flow parameter and the exhaust gas temperature parameter for establishing corrections for simple jet engines is described in paragraphs 4.2 and 4.3. Similar corrections for more advanced engines are discussed in paragraph 4.4.

#### 4.2 FUEL FLOW PARAMETER

The engine manufacturer's estimated fuel flow curves are seldom used in flight testing, since flow rates are measured with test instrumentation. The same parameters are used,  $(w_f/\sqrt{\Theta_{t_2}} \delta_{t_2})$ versus  $N/\sqrt{\Theta_{t_2}}$ , in plotting test fuel flow data, however, as are found in estimated curves furnished by the engine manufacturers. Test fuel flows are corrected to standard conditions using test data plotted in this form.







# Typical Turbojet Engine Characteristics

The effects of changes in specific heat ratio, combustion efficiency, Reynolds number, etc., were neglected, (Reference Section 2), when the dimensionless parameters were developed. Consequently, an exact correlation of test data obtained over a wide range of flight speeds and altitudes cannot be expected. At altitudes up to about 35,000 feet, (depending on the engine design), quite good correlation can be expected. At higher altitudes, however, the neglect of the change in Reynolds number, in particular, becomes increasingly important, and separation from the basic polar occurs as illustrated in Figure 4.2.



Figure 4.2 Typical Fuel Flow Characteristics
At normal in-flight operating conditions there is no apparent effect from changes in  $P_{t2}/P_a$ . At low altitudes with low power settings when flow at the nozzle exit is subcritical, lines of constant  $P_{t2}/P_a$  diverge from the basic polar as shown in Figure 4.2.

## 4.3 EXHAUST GAS TEMPERATURE PARAMETER

Plots of corrected exhaust gas temperature versus corrected rpm from flight test data can be used to apply corrections to exhaust gas temperature and other performance variables. The corrections are necessary when an engine is operated at other than standard exhaust gas temperature. For example, corrections to exhaust gas temperature, thrust and rpm may be made as described in Data Reduction Outline 8.2.

### 4.4 ADVANCED ENGINES

Many different configurations in nossies, ejectors, control systems, etc., are in use or will be installed in future aircraft. Because of the variety of configurations which exist and are planned, it is not practical nor possible to describe methods for correcting engine data to standard conditions which are suitable for each type. It frequently is not immediately evident as to when non-dimensional methods are applicable. The characteristics of each of the more complex engines should be studied so that methods may be modified as required for the individual case, and the best means chosen for making corrections to standard conditions.

#### SECTION 5

### AIRFLOW MEASUREMENT

## 5.1 INTRODUCTION

As was seen in Section 3.1, the thrust of a turbojet engine is dependent on ram drag which requires a knowledge of mass flow through the engine. Three methods of measuring engine airflow in flight have been generally used and are discussed in this Section.

#### 5.2 ENGINE COMPRESSOR AIRFLOW CURVES

Of the three methods for determining airflow, use of engine compressor airflow curves furnished by the engine manufacturers is most common. These curves are generally plotted as

$$\frac{\mathbf{w}_{a} \cdot \mathbf{\theta}_{t2}}{\mathbf{\theta}_{t2}} = \mathbf{v}_{a} \cdot \frac{\mathbf{N}}{\mathbf{v}_{t2}}$$

where  $\theta_{t_2} = T_{t_2}/T_{SL}$  and  $\delta_{t_2} = P_{t_2}/P_{SL}$ 

Compressor inlet total temperature,  $T_{t2}$ , is computed from ambient temperature and free stream Mach number assuming adiabatic flow. Compressor inlet pressure,  $P_{t2}$ , is obtained preferably by measurement, but if inlet instrumentation is not installed it can be computed from total free stream pressure and estimated total pressure recovery.

#### 5.3 INLET DUCT METHOD

Airflow can be measured from total temperature and surveys of total and static pressure forward of the compressor face. From continuity

$$m = \frac{w_{A}}{g} = \rho V A$$
 5.1

where

m = mass flow, slugs/second w<sub>a</sub> = weight flow, lb/second  $\rho$  = density, slugs/ft<sup>3</sup> V = velocity, ft/second A = annular area, ft<sup>2</sup>

 $\rho$ , V and A represent values at the station where pressure measurements are taken.

Assuming a perfect gas

$$\rho = \frac{P_s}{g R T_s}$$
 5.2

where

P<sub>8</sub> = static pressure, lb/ft<sup>2</sup> R = gas constant, ft-lb/lb \*K T<sub>8</sub> = static temperature, \*K

$$V = M \sqrt{g \gamma RT} 5.3$$

where

or

M = Mach number

Y = ratio of specific heats

Substituting equations 5.2 and 5.3 in equation 5.1

$$w_{a} = \frac{P_{a}A}{RT_{a}} M \sqrt{g YRT_{a}}$$

$$w_{a} = P_{a}AM \sqrt{\frac{YB}{RT_{a}}}$$
5.4

From the insentropic relation

$$M = \sqrt{\frac{2}{\gamma - 1} \left[ \frac{\mathbf{p}_{t}}{\mathbf{p}_{s}} \frac{\gamma - 1}{\gamma} - 1 \right]} \qquad 5.5$$

and

$$T_t = T_s (1 + \frac{\gamma - 1}{2} M^2)$$
 5.6

Substituting equations 5.5 and 5.6 in equation 5.4

$$w_a = P_s A \sqrt{\frac{2 Y}{Y-1}} \frac{g}{RT_t} \left[ \left(\frac{P_t}{P_s}\right)^{\frac{Y-1}{Y}} - 1 \right] \left(\frac{P_t}{P_s}\right)^{\frac{Y-1}{Y}} 5.7$$

Total pressure surveys are commonly made by dividing the duct into equal annular areas with pressure probes located to measure the pressure in each of these areas. Probes should also be located near the wall of the duct to account for boundary-layer effects. Care should be taken to locate the total pressure probes in a straight portion of the duct, and that no struts or other obstructions which would cause pressure gradients are immediately upstream of the probes. Static pressures are measured from either pick-ups located on the total pressure rakes or from wall static taps or a combination of both.





# 5.4 TAILPIPE TEMPERATURE METHOD

Gas flow at the nozzle (which includes both air and fuel) can be calculated using the same basic equations that were used to compute airflow from inlet pressure measurements. Gas flow is frequently expressed as:

$$\frac{w_{g_8}\sqrt{T_{t_8}}}{P_{s_8}A_8} = M_8 \sqrt{\frac{Y_8}{R} (1 + \frac{Y^{-1}}{2} M_8^2)}$$
 5.8

In an ideal converging nozzle the static discharge pressure  $(F_{88})$ remains constant and equal to the ambient pressure  $(P_a)$  until, as the total discharge pressure  $(P_{18})$  is increased, the maximum obtainable Mach number of one is reached. As  $P_{18}$  is increased further the Mach number remains at one and  $P_{88}$  rises above  $P_a$  but the ratio of  $P_{18}/P_{88}$  remains constant. An exit Mach number less than one is called subcritical and a Mach number of one is called supercritical.

SubcriticalSupercritical $P_{tg}/P_{sg} < 1.85$  $P_{tg}/P_{sg} = 1.85$  $P_{sg} = P_a$  $P_{sg} > P_a$  $M_g < 1.0$  $M_g = 1.0$ 

Mach number at the exit is:

$$M_8 = \sqrt{\frac{2}{\gamma - 1}} \left[ \left( \frac{P_{t_8}}{P_{s_8}} \right)^{\frac{\gamma - 1}{\gamma}} - 1 \right]$$
 5.9

Substituting equation 5.9 in equation 5.8 and applying standard units, the subcritical flow equation becomes:

$$\frac{W_{g_0}}{P_a} \sqrt{T_{19}} = 116.23 \sqrt{X(1+X)} 5.10$$

where

 $g = 32.174 \text{ ft/sec}^{2}$  Y = 1.33 (non-afterburning) R = 96.031 ft/ \*K  $Ag = \text{ft}^{2}$   $w_{gg} = \text{lb/sec}$   $P_{tg}, P_{eg}, P_{a} = \text{"Hg}$   $X = \left(\frac{P_{tg}}{P_{eg}}\right) \frac{Y-1}{Y} - 1$ 

For supercritical flow with Mg = 1.0, equations 5.9 may be written:

$$\mathbf{P_{ag}} = \mathbf{P_{tg}} \left(\frac{2}{\gamma+1}\right) \frac{\gamma}{\gamma-1} \qquad 5.11$$

Substituting equation 5.11 in equation 5.8 with M = 1.0,

$$\frac{w_{gg}\sqrt{T}_{tg}}{P_a A_g} = \left(\frac{2}{\gamma+1}\right)^{\frac{1}{\gamma+1}}\sqrt{\frac{2g}{R}}\left(1+\frac{\gamma+1}{2}\right) \frac{P_{tg}}{P_a} 5.12$$

Assuming Y = 1.33 for engine operation,

$$\frac{w_{g8} \sqrt{T_{t8}}}{P_a A_8} = 27.54 \frac{P_{t8}}{P_a} 5.13$$

Assuming  $\gamma = 1.28$  for afterburner operation,

$$\frac{w_{g8}\sqrt{T_{t8}}}{P_a A_8} = 27.15 \frac{P_{t8}}{P_a} \qquad 5.14$$

The ideal gas flow parameter,  $w_{g8} \sqrt{T_{t8}}/P_a A_8$ , is plotted as a function of  $P_{t8}/P_a$  in Charts 9.17 and 9.18.

With a converging-diverging nozzle the gas flow near the nozzle exit where the tailpipe instrumentation is located (as with a swinging rake) becomes supersonic. Since in this case a detached shock stands ahead of the total pressure probe it is necessary to compute Mach number from the Rayleigh supersonic pitot formula.

$$\frac{\mathbf{Pt}'}{\mathbf{P_B}} = \frac{\left(\frac{\gamma+1}{2} M^2\right)}{\left(\frac{2\gamma}{\gamma+1} M^2 - \frac{\gamma-1}{\gamma+1}\right)^{\frac{1}{\gamma-1}}} 5.15$$

Use of this equation demands the measurement of static pressure which is quite sensitive to flow alignment. Yaw angles encountered at the nozzle exit are not large enough to cause significant errors in total pressures, but may produce sizeable errors in static pressure measurement (referen e paragraph 6.6.1).

Determination of airflow from tailpipe instrumentation is further complicated by difficulties in measuring exhaust gas temperatures, particularly in afterburning, and is probably the least accurate of the three methods described.

#### SECTION 6

#### IN FLIGHT THPUST MEASUREMENT

## 6.1 INTROPUCTION

Practical applications of measuring the momentum change of the internal flow to obtain in flight thrust of turbojet engines in aircraft will be presented in this section. The thrust produced by a simple jet engine with fixed exhaust nozzle is considered in detail in the following paragraph. More advanced engines with afterburners, ejectors and variable area exhaust nozzles are then considered in succeeding paragraphs.

## 6.2 FIXED EXHAUST NOZZLE

With subcritical flow it is assumed that the static pressure at the nozzle expands to ambient pressure and gross thrust is defined by the equation.

$$F_g = F_v = \frac{w_g g}{g} V_g \qquad 6.$$

1

where

wg8 = weight flow, lb/sec V8 = exit velocity, ft/sec F<sub>v</sub> = velocity thrust, lb

Substituting equations 5.1, 5.2, and 5.3 in equation 6.1 we have,

$$\mathbf{F_g} = \mathbf{P_{s8}A_8 Y M8}^2 \qquad 6.2$$

Substituting equation 5.5

$$\mathbf{F}_{g} = \mathbf{P}_{g} \mathbf{8} \mathbf{A}_{g} \frac{2\mathbf{Y}}{\mathbf{Y}-1} \left[ \left( \frac{\mathbf{P}_{tg}}{\mathbf{P}_{gg}} \right)^{\frac{\mathbf{Y}-1}{\mathbf{Y}}} - 1 \right] \qquad 6.3$$

Setting 
$$\left(\frac{P_{t\theta}}{P_a}\right) = Y = 1 = X$$
 and  $P_{a\theta} = P_a$ , we have

 $\frac{F_{u}}{P_{a}A_{8}} = 570$ 

where

 $P_a = ambient pressure, "Hg$  $As = nozzle area, <math>it^2$ Y = 1.33

Subcritical flow can only be achieved at low power settings at low altitudes. The flow is supercritical for nearly all in-flight conditions where thrust measurement is desired. (i.e., the flow in an ideal converging nozzle is choked and the Mach number at the nozzle exit is unity). In this case  $P_{88}$  rises above  $P_{8}$  and the gross thrust is made up of velocity thrust (F<sub>v</sub>) and pressure thrust, (F<sub>p</sub>).

$$F_{p} = A_{8} (P_{s8} - P_{s})$$
 6.5

6 1

or

$$F_{p} = A_{g} \left[ P_{t_{g}} \left( \frac{2}{\gamma+1} \right)^{\frac{\gamma}{\gamma-1}} - P_{a} \right] \qquad 6.6$$

Adding velocity thrust and pressure thrust the following equation may be written

$$\frac{F_{g}}{P_{a}A_{8}} = \frac{P_{tB}}{P_{a}} \left(\frac{2}{\gamma+1}\right)^{\frac{V}{Y-1}} (Y+1) - 1 \qquad 6.7$$

substituting constants,

$$\frac{F_{g}}{P_{a}A_{8}} = 70.727 \ (1.259 \ \frac{P_{t8}}{P_{a}} - 1) \tag{6.8}$$

The assumption that Y = 1.33 is made with a suitable degree of accuracy since the variation in Y with exhaust gas temperature may range from about 1.32 to 1.36, but this variation causes a change in  $(2/\gamma+1)^{\gamma/\gamma-1}(\gamma+1)$  of less than 0.5 percent.

The ideal gross thrust parameter,  $F_g/P_a$  Ag, from equations 6.4 and 6.8 has been plotted as a function of  $P_{tg}/P_a$  in Charts 9.19 and 9.20.

The preceding derivations are based on an ideal nozzle assuming isentropic one dimensional flow. This condition is not completely realized, however. A correction factor must be applied to account for deviations from an ideal nozzle as well as losses from wall friction and errors in pressure measurement.

The correction factor for computing thrust (nozzle thrust coefficient) is defined as:

$$C_{f} = \frac{\left(\frac{F_{g}}{A_{g}P_{a}}\right)}{\left(\frac{F_{g}}{A_{g}P_{a}}\right)}$$
 theoretical

where

 $F_{gactual}$  is the mechanically measured thrust  $F_g/AgP_a$  is the theoretical gross thrust parameter A is nozzle area, ft<sup>2</sup>  $P_a$  is atmospheric pressure, "Hg

Differences in thrust coefficient exist between engine-tailpipe combinations of the same model. Consequently, for most accurate inflight thrust measurements, a ground static calibration of each installation should be made and repeated if the engine or tailpipe or both are changed. When this is done the nozzle total pressure ratio may be measured satisfactorily with a single probe. Static thrust calibrations may be obtained with the aircraft mounted on a thrust stand or with a

bare engine to which a beilmouth inlet is attached. When installed thrust is determined on a thrust stand, pressure gradients are set up from air entering the inlet duct which may result in pressure forces and give erroneous gross thrust readings. This effect is likely to be most pronounced on high speed aircraft whose inlet total pressure recoveries are quite low at zero forward speed. In a bellmouth inlet pressure drops and consequently pressure forces are minimized.

The maximum pressure ratios obtained during static thrust calibrations are less than those encountered in flight, except with low power settings at low altitudes. Pressure ratios of 2 to 2.5, depending on ambient temperature, are generally found statically while values of 3 to 3.5 are typical for cruise conditions and may be as high as 10 or more at high speeds and altitudes. Consequently, thrust coefficient data obtained during static conditions must be extrapolated to higher nozzle pressure matios in order to compute thrust in flight. For a simple conical nozzle the value of thrust coefficient is constant at about 0.98 for nozzle pressure ratios greater than about 1.9 as shown in Figure 6.1.





Gross thrust can then be calculated for in-flight values of  $P_{tg}/P_a$  using the following equations.

Subcritical

$$F_g = 570.13 P_a A_8 C_f X$$
 6.9  
Supercritical

$$F_g = 70.727 P_a A_8 C_f (1.259 \frac{P_{t8}}{P_a} - 1)$$
 6.10

Ram drag has been previously defined as

$$m_1 V_0 = \frac{w_a}{g} V_{aircraft}$$

where

wa = inlet airflow, lb/sec

Substituting units the following equation may be written

$$F_{e} = .0525 w_{a} V_{tt}$$
 6.11

#### where

## $V_{tt}$ = airplane test day true speed, knots

Engine airflow is determined by one of the three methods described in Section 5.

If tailpipe instrumentation is used, the calculated gas flow is generally assumed equal to the inlet airflow. This assumption is normally satisfactory since the fuel added makes up only about 2 percent of the airflow and is approximately the compressor leakage. Also, a gas flow coefficient must be applied to the calculated theoretical gas flow to account for the same deviations from ideal nozzle flow as the nozzle thrust coefficient.

# 6.3 THRUST AUGMENTATION

Methods of obtaining thrust augmentation which are in general use are afterburning and, to a lesser extent, water injection. Afterburning is considered in this section and water injection in Section 7.

The turbojet engine may have its thrust increased by a substantial amount by burning additional fuel in the turbine exhaust ahead of the exit nozzle. This is possible since the quantity of air passing through the engine is about four times that required for combustion, and the remaining 75 percent is capable of supporting additional combustion if more fuel is added. An afterburner is made up of only four fundamental parts; the afterburner duct, fuel nozzles or spraybars, flame holders, and two-position or variable area exhaust nozzle.



Figure 6.2 Afterburner Components

The thrust of an engine with afterburner may be computed as for a simple jet engine. Thrust coefficient data should be obtained during static thrust calibrations with the afterburner both on and off (reference Figure 6.3).





3-46

Measurement of nozzle total pressure becomes more difficult with an afterburner-equipped engine because of much higher temperatures at the nozzle. A rake may be mounted across the diameter of the nozzle but since the rake is located in the extremely hot gas stream some method of cooling (such as with compressor bleed air) is essential. Also, such a rake is subject to deterioration from the extreme heat and will probably be rather short lived. When testing bare engines in test chambers, water-cooled rakes mounted at the exhaust nozzle are generally used but have not been found suitable for installation in aircraft.

...nother means of determining jet thrust is from turbine outlet pressure, (usually from probes installed by the engine manufacturer). The relationship shown in Figure 6.4 to determine nozzle total pressure,  $P_{18}$ , from turbine outlet total pressure,  $P_{15}$ , may be obtained from engine calibrations in a test chamber. For this relationship to be valid flow at the exhaust nozzle must be choked.  $P_{18}$  is measured with a water-cooled rake. (Losses in total pressure between stations 5 and 8 are incurred largely by friction losses across the flameholder).







To determine gross thrust from turbine outlet total pressure and curves similar to those in Figure 6.4, equation 6.10 may be modified to:

$$F_g = 70.727A_8C_f \left[ 1.259P_{t5} \left( 1 - \frac{P_{t5} - P_{t8}}{P_{t5}} \right) - P_a \right] 6.11$$

## 6.4 EJECTOR

An exhaust ejector as illustrated in Figure 6.5 may be used to pump tailpipe cooling air. A properly designed ejector provides adequate cooling with the afterburner operating but does not severely penalize performance at cruise power settings.



#### Figure 6.5

Typical Convergent Ejector Installation

Secondary flow originates at an intake in the vicinity of the engine air induction system inlet, or may stem from bleed passages located in the induction system subsonic diffuser. The secondary air then passes through the engine compartment, where it serves as a cooling medium, and is subsequently exhausted through the ejector outlet. In addition to the converging ejector shown in Figure 6.5, either cylindrical or converging-diverging ejectors may be employed. With a properly designed ejector, flow may be made to approach idealized flow through a converging-diverging nozzle, as described in paragraph 6.5.

The geometry of an ejector is critical for obtaining satisfactory ejector performance. The parameters diameter ratio,  $D_{p+s}/D_p$ , and spacing ratio,  $L/D_p$ , are used to describe ejector geometry. The spacing ratio should not be so large that expansion of the primary flow within the ejector results in impingement on the inner surface of the ejector. Also, the diameter ratio should not be large enough to cause circulation of external air over the shroud trailing edge, with a resulting reduction in secondary flow rate and an increase in base drag. Typical fixed ejector configurations are generally defined by spacing ratios of approximately 0.40 and diameter ratios of approximately 1.20.

The addition of an ejector to an engine installation further complicates the measurement of net thrust. In addition to the measurement of primary thrust, the ram drag and gross thrust of the secondary flow must also be considered.

This thrust contribution may be stated as:

 $F_{nej} = \frac{w_s}{g} V_s + (P_{sg} - P_a) A_s - \int_{sta \ g}^{sta \ g} (P_w - P_a) dA_w - \frac{w_a}{g} V_t \ 6.12$ sta 9

where

Fnej = ejector net thrust w<sub>8</sub> = secondary weight flow V<sub>8</sub> = secondary flow speed

Ps8	=	static pressure of secondary flow at station 8
Pa	=	ambient pressure
A <sub>s</sub>	=	area of ejector at primary nozzle exit
Pw	=	ejector wall static pressure
Aw	=	projected ejector area between stations 8 and 9
v <sub>t</sub>	=	airplane true speed

The velocity thrust (first term on right side of equation 6.12) may be modified using methods similar to those in Section 6, resulting in equation 6.13:

$$F_{n_{ej}} = P_{s8}A_s \left(\frac{2\gamma_s}{\gamma_{s-1}}\right) \left[ \left(\frac{P_{t8}}{P_{s8}}\right)^{\gamma_s} - 1 \right] + \left(P_{s8} - P_a\right)A_s - \int_{sta8}^{sta8} (P_w - P_a)dA_w - \frac{w_s}{g}V_t$$
  
sta9  
6.13

The secondary passage usually contains nozzle actuators and other equipment so that a uniform velocity profile is not obtained and accurate measurement of secondary total and static pressure is difficult. A high degree of accuracy in total pressure measurement is not required, however, since the secondary velocity thrust is small relative to the primary thrust. A detailed survey of static pressures at station 8 and axially along the ejector shroud is required in some installations. Such installations include those in which over-expansion of the primary jet occurs with a resulting shock wave system within the ejector, and those in which large ejector included angles are encountered. In these instances pressure forces become quite significant. When ejector included angles are small, the projected area  $A_W$  may be small enough so that the third term in equation 6. 13 may be omitted. With cylindrical ejectors  $A_W$  is, of course, zero.

The general equation for determining net thrust for an installation with an ejector is

$$F_{n} = F_{gp} + P_{g}A_{g}(\frac{2\gamma_{g}}{\gamma_{g}-1}) \left[ (\frac{P_{tg}}{P_{gg}})^{\gamma_{g}} - 1 \right] + (P_{gg} - P_{a})A_{g} - \int_{g}^{gta} (P_{w} - P_{a})dA_{w} - \frac{w_{1}}{g}V_{o}$$
  
sta 9  
6.14

The primary gross thrust,  $F_{gp}$ , is calculated as described in the preceding paragraph, and the ratio of specific heats in the secondary stream,  $\gamma_s$ , is assumed equal to 1.4.

Instead of the above procedures entailing internal pressure measurements, it may be more desirable to gather data with a swinging rake which samples pressures along a cross section of both the primary and secondary jets. Application of the swinging rake to thrust measurement is treated separately in paragraph 6.6.

# 6.5 CONVERGING-DIVERGING NOZZLE

A gain in thrust may be realized by replacing the more conventional conical nozzle with a converging-diverging nozzle. The increased engine performance is partially offset, however, by increased weight and is obtained at the expense of added controls and mechanical complication. The diverging portion of the nozzle in operational turbojet engines is formed aerodynamically rather than by physical structure (reference Figure 6.6).



#### Figure 6.6

Schematic Diagram of Aerodynamic Converging-Diverging Nozzle

**3**– 52

For optimum performance throughout the operating range it is necessary to modulate both the primary and secondary nozzle areas and the spacing ratio,  $L/D_p$ .

In a converging-diverging nozzle, idealized flow is supersonic and fully expanded at the nozzle exit, and the gross thrust is

$$F_g = \frac{w_g}{g} \cdot V_e$$
 6.15

Restating equation 5.4 with the Mach number equal to unity at the throat

$$w_{gth} = P_{sth} A_{th} \sqrt{\frac{\gamma_g}{RT_{sth}}}$$
 6.16

From the relations

$$P_{sth} = P_{t_{th}} \left(\frac{2}{\gamma+1}\right)^{\frac{\gamma}{\gamma-1}}$$
 6.17

and

$$\frac{T_{t_{th}}}{T_{sth}} = 1 + \frac{Y - 1}{2} M^2$$
 6.18

$$w_g = P_{th} \left(\frac{2}{\gamma+1}\right)^{\frac{1}{\gamma-1}} A_{th} \sqrt{\frac{g}{RT_{th}}} \frac{(\gamma+1)}{2} \qquad 6.19$$

Velocity at the nozzle exit may be expressed as

$$V_e = M_e \sqrt{gYRT_{s_e}}$$
 6.20

From equation 6.18

$$V_e = \frac{M_e \sqrt{g \gamma R T_{fe}}}{1 + \frac{\gamma - 1}{2} M_e}$$
 6.21

Since the flow is fully expanded at the nozzle exit

$$M_{e} = \sqrt{\frac{2}{\gamma - 1} \left[ \left( \frac{P_{t_{e}}}{P_{a}} \right)^{\frac{\gamma - 1}{\gamma}} - 1 \right]}$$
 6.22

and

.

•

$$V_{e} = \sqrt{\frac{2}{\frac{Y-1}{Y-1}} \left[ \left(\frac{P_{te}}{P_{a}}\right)^{\frac{Y-1}{Y}} - 1 \right] gYRT_{te}}{1 + \left[ \left(\frac{P_{te}}{P_{a}}\right)^{\frac{Y-1}{Y}} - 1 \right]}}$$
$$= \sqrt{\frac{2Y}{Y-1} gRT_{te}} \left[ 1 - \left(\frac{P_{a}}{P_{te}}\right)^{\frac{Y-1}{Y}} \right] \qquad 6.23$$

Substituting equations 6.19 and 6.23 in equation 6.15 to find gross thrust,

$$\mathbf{F}_{g} = \frac{\mathbf{P}_{th} \left(\frac{2}{\nabla + 1}\right)^{\frac{\gamma}{\gamma - 1}}}{g} \quad \mathbf{A}_{th} \sqrt{\frac{\gamma g}{RT_{th}}} \quad (\gamma + 1) \frac{\gamma}{\gamma - 1} gRT_{te} \left[1 - \left(\frac{\mathbf{P}_{a}}{\mathbf{P}_{te}}\right)^{\frac{\gamma - 1}{\gamma}}\right] \quad 6.24$$

Since  $P_t$  and  $T_t$  are constant in adiabatic flow,  $P_{tth} = P_{te}$ , and  $T_{tth} = T_{te}$ .

$$\mathbf{F}_{g} = \mathbf{P}_{tth} \left(\frac{2}{\gamma+1}\right)^{\frac{\gamma}{\gamma-1}} \mathbf{A}_{th} \sqrt{\frac{\gamma+1}{\gamma-1}} \left[1 - \left(\frac{\mathbf{P}_{a}}{\mathbf{P}_{tth}}\right)^{\frac{\gamma-1}{\gamma}}\right] \qquad 6.25$$

Forming the ideal gross thrust parameter

$$\frac{F_g}{P_a A_{th}} = \frac{P_{th}}{P_a} \left(\frac{2}{\gamma+1}\right)^{\frac{\gamma}{\gamma-1}} \sqrt{\frac{\gamma+1}{\gamma-1}} \left[1 - \left(\frac{P_a}{P_{th}}\right)^{\frac{\gamma-1}{\gamma}}\right] \quad 6.26$$

Equation 6.26 is presented in graphical form in Chart 9.21.

It has been pointed out previously that in a converging nozzle the flow is subcritical at nozzle pressure ration less than appreximately 1.85. At higher pressure ratios the convergent exhaust nozzle is choked and operates with a supercritical pressure ratio. In this case the static pressure at the exit is larger than atmospheric pressure; that is, the gases are underexpanded. In a converging-diverging nozzle complete expansion occurs with idealized flow, resulting in a somewhat higher thrust when the flow is supercritical. This difference in thrust is shown in Figure 6.7. It can be seen from this figure that in subsonic flight (maximum pressure ratios of the order of 4) the gain in thrust is too slight to warrant installation of a converging-diverging nozzle. In supersonic flight, however, where the pressure ratios become much higher, a substantial increase in thrust is possible.



#### Figure 6.7

Theoretical Loss in Thrust Due to Underexpansion

#### 6.6 SWINGING RAKE

Stationary air-cooled probes located in the nozzle exit have been employed with adequate results. Probes of this sort are subject to damage from high temperatures, however. Swinging rakes may provide the bast means for measuring the jet thrust of more advanced engines, such as those with afterburners and ejectors. Rakes of this design are normally stowed outside the jet exhaust where they are cooled by freestream air. They are driven across the tailpipe when data is being recorded in about 4 or 5 seconds, so that prolonged exposure to the hot jet is avoided and a cooling system is not required. The thrust contribution of an ejector together with the thrust created by the basic engine may be computed from data obtained with a swinging rake. Also, better mean values of pressure are obtained with a swinging rake than with a fixed probe, although pressures are still measured along only one cross section.

The jet of high performance engines expands rapidly, particularly at high power settings, resulting in Mach numbers which are well supersonic downstream of the nozzle. Hence, it is desirable to measure both total and static pressures at a common point in a plane as near the nozzle exit as possible. Several different designs have been utilized, although none of them satisfy this condition exactly. These designs include installations which sense both total and static pressures on the same probe, and those which sense pressures on different probes but in the same plane (reference Figure 6.8).



Figure 6.8

Various Designs of Swinging Rakes

In designs similar to that shown in Figure 6.8 (a), the probes should be as close as possible without creating excessive aerodynamic interference. Another type of installation which is perhaps the best compromise for obtaining both total and static pressures at the same point makes use of a pitot-static probe with static pressure measured on a conical surface.

6.6.1 Sources of Error:

Little information is available on the flow angularities which exist at the nozzle exit. Flow angularities of approximately 15 to 20 degrees due to swirl of the primary jet should be expected based on NACA Research Memorandum E57H28, "Experimental Results of an Investigation of Two Methods of In-Flight Thrust Measurement Applicable to Afterburning Turbojet Engines with Ejectors", by Harry E. Bloomer. No significant effect on total pressure results from flow angles of this magnitude with an adequately designed pitot tube. Static pressures are subject to quite substantial errors, however, as shown by Figure 6.9 extracted from RM E57H28.





As is pointed out in this memorandum, it would seem that large errors in thrust measurement might result with supersonic flow since the static pressure is used to correct the total pressure for bow shock. Errors in static pressure are not as serious as might be anticipated, as demonstrated in Figure 6.10 which shows variations in gross thrust from errors in static pressure.



## Figure 6.10

Error in F/A vs Nozzle Pressure Ratio for Assumed Errors in Static Pressure

Exact information on yaw angles cannot be expected for flight test installations. Approximate corrections can be made, however, to bring static pressures to within say  $\pm 10$  percent of their true values and keep the error in gross thrust caused by inaccuracies in static pressure to within ±1 percent.

Constant values of  $\gamma$ , (1.33 for non-afterburning and 1.28 for afterburning), may be used for the entire swing without introducing significant errors. Errors in total pressure may be introduced by lag, particularly during the portion of the traverse where pressure gradients are large. Lag errors may be minimized by averaging pressures taken during traverses in opposite directions. The accurate determination of probe position, from which nozzle area is found, is necessary for achieving satisfactory accuracies in thrust computation. Measurement of probe position is made more difficult by possible bending of the rake body from aerodynamic forces and thermal stresses.

6.6.2 Calculation of Gross Thrust:

The following equation may be used to compute thrust with a swinging rake:

$$\mathbf{F}_{g} = \int \left\{ \mathbf{P}_{g,g} \left[ \frac{2 \mathbf{Y}}{\mathbf{Y}-\mathbf{1}} - \left(\frac{\mathbf{P}_{1,g}}{\mathbf{P}_{g,g}}\right)^{\frac{\mathbf{Y}-\mathbf{1}}{\mathbf{Y}}} - \frac{\mathbf{Y}+\mathbf{1}}{\mathbf{Y}-\mathbf{1}} \right] - \mathbf{P}_{g,g} \right\} dAg \qquad 6.27$$

Static pressures are first corrected for yaw angle. Indicated total pressures are used directly when the flow is subsonic. With supersonic flow the total pressure behind a detached shock is sensed. In this case Mach number may be computed from the Rayleigh supersonic pitot formula (reference Chart 9.22).

$$\frac{P_{t}}{P_{s}} = \frac{\left(\frac{Y+1}{-2} - M^{2}\right)}{\left(\frac{2Y}{Y+1} - M^{2} - \frac{Y-1}{Y+1}\right)^{\frac{Y}{Y-1}}} 5.15$$

The isentropic relation,

$$\frac{P_{tq}}{P_{sq}} = (1 + \frac{Y - 1}{2} M^2) \frac{Y}{Y - 1}$$

may be used to determine  $P_{tg}/P_{sg}$ .

The area included by the probe traverse is computed from a calibration of the angular displacement of the probe from the vertical centerline versus distance from the center of the nozzle. Pressures may then be plotted as illustrated in Figure 6.11.





Gross thrust may be computed from equation 6.27 using a mechanical integration procedure, by summing values of  $\Delta F_g$  calculated from average  $P_{19}$  and average  $P_{89}$  over  $\Delta A$  (reference Figure 6.11). This procedure involves rather lengthy calculations and is not adaptable to test programs where large quantities of data are processed. Here, a machine solution which makes use of curve fits of total and static pressure distributions is virtually essential.

# SECTION 7 WATER INJECTION

## 7.1 INTRODUCTION

Thrust augmentation may be obtained by injecting water or other liquids into the airstream anywhere from the compressor inlet to the rear of the burner. A mixture of water and methyl alcohol is frequently used. The alcohol prevents freezing and also provides additional heating which compensates for the heat lost through evaporation of the wateralcohol mixture. The additional heat is supplied when the alcohol burns. Water-alcohol is usually injected in the compressor inlet, in the combustion chamber or at both locations simultaneously. An increase in thrust from about 10 to 25 percent can be obtained, depending on the type of installation, amount of water injected and the flight conditions. This increase in thrust is achieved at very high total liquid flow rates and can be employed for only short periods of time. Consequently, the use of water injection is generally limited to improving take-off performance.

Compared to afterburning, water injection is less efficient and more limited in the augmentation ratios which can be obtained. Water injection does have the advantage of simplicity of installation and operation and does not entail as large an installation weight penalty as afterburning. Also, a performance penalty is not incurred during cruise as is the case with an afterburner installation. For relatively small short duration thrust increases, water injection may, therefore, be the more suitable of the two systems.

# 7.2 INJECTION IN THE COMPRESSOR INLET

Water injection in the compressor inlet has the advantage that a greater amount of thrust augmentation is produced per pound of liquid injected. Increases in thrust are produced from the three following effects:

1. The mass flow is increased. Some of this increase is due to the mass of the injected liquid, and some from a reduction in compressor inlet temperature. It is theoretically possible to cool the inlet air to the saturation temperature before it enters the compressor. The air is not cooled to that extent in practice, however, since the rate of evaporation is limited principally by spray droplet size and air turbulence. As the spray passes into the compressor, further cooling is obtained by additional evaporation during the mechanical compression process.

 Chis power required to operate the compressor at a constant pressure ratio is decreased. This is also caused by the lowered inlet temperature which decreases the required change in enthalpy necessary to perform a given amount of compression.
 A higher pressure ratio from the compressor is obtained. This increased pressure ratio is attributed to the increased density of the gases flowing through the compressor.

Further, the decrease in compressor discharge temperature tends to be reflected in a lower exhaust gas temperature. Although the lowered compressor power input requirement tends to increase exhaust gas temperature, the net effect is generally to produce a lower temperature. Hence, more fuel, with higher mass flows, is added to the combustion chamber in order to retain the same exhaust gas temperature. These effects combine to increase the thrust output.

### 7.3 COMEUSTION CHAMBER INJECTION

Thrust may be increased by injecting water or a water-alcohol mixture into the combustion chamber. The turbine inlet pressure is increased thereby, and a higher total mass flow results. The total mass flow tends to be reduced, however, due to changing the equilibrium running conditions of the compressor with the addition of

water injection. The compressor pressure ratio is increased while the compressor rotative speed remains constant, so that the airflow is reduced. Hence, the amount of augmentation is dependent on the operating characteristics of the compressor. (See Figure 7.1).





Simplified Compressor Performance. Chart

At equilibrium, the compressor flow is lower, but the turbine and nozzle exit flow is higher as is the nozzle pressure ratio and consequently furust. Compressor surge will limit the amount of liquid which can be injected into the combustion chamber. Practical increases in thrust are limited for this reason to about 15 to 20 percent.

# SECTION 8 DATA REDUCTION

## 8.1 CONVERSION OF ESTIMATED PERFORMANCE CURVES TO CORRECTION PLOT $(F_n / \delta_{t_2} \text{ versus } N / \sqrt{\theta_a})$

- 1. Mach number desired
- 2.  $P_{t2}/P_{t_0}$  from plot of  $P_{t2}/P_0$  versus M
- 3.  $P_{t_0}/P_a = (1 + .2M^2)^{3.5}$
- 4.  $P_{t_2}/P_{a_1}$  (2) x (3)
- 5. N/ $\sqrt{\Theta_a}$ , select values to cover the flight range
- 6.  $(T_{t2}/T_a)^{\frac{1}{2}} = (1 + 2M^2)^{\frac{1}{2}}$
- 7. N/ $\sqrt{\theta_{t_2}}$ , (5)/(6)
- 8.  $F_g/\delta_{t_2}$  from engine manufacturer's estimated gross thrust curves at (4) and (7)
- 9.  $w_a \sqrt{\theta_{t_2}}/\delta_{t_2}$  from engine manufacturer's estimated airflow curves at (4) and (7)
- 10.  $F_e/\delta_{t_2}$ , (9) x (1) x 34.73/(6)
- 11.  $F_n/\delta_{t_2}$ , (8) (10)
- 12. Plot (11) versus (5) for Mach numbers selected in (1)

#### 8.2 DETERMINATION OF EXHAUST GAS TEMPERATURE, RPM AND NET THRUST CORRECTIONS FOR OFF-STANDARD EXHAUST GAS TEMPERATURE

- 1. Tt5., test exhaust gas temperature
- 2. N<sub>t</sub>, test engine speed
- 3. Na. standard engine speed
- 4.  $\theta_{at}$ ,  $T_{at}/T_{asl}$
- 5.  $\theta_{as}$ ,  $T_{as}/T_{asL}$
- 6.  $T_{t5t}/\theta_{at}$ , (1)/(4)
- 7.  $N_t / \sqrt{\theta_{at}}$ , (2) /  $\sqrt{(4)}$
- 8. N<sub>a</sub> /  $\sqrt{\Theta_{as}}$ , (3) /  $\sqrt{(5)}$

3-- 66

- 9. Tt5max, maximum allowable exhaust gas temperature
- 10.  $T_{t5_{max}}/\theta_{as}$ , corrected maximum allowable exhaust gas temperature, (9)/(5)
- 11.  $T_{t5_g}/\theta_{as}$ , standard corrected exhaust gas temperature corresponding to (8) from plot of (6) and (7) at (8)

Case I: (11) less than (10)

- <sup>12.</sup>  $T_{t5_g}$ , (11) x (5)
- 13.  $\Delta F_n / \delta_{t_2}$  from plot of  $F_n / \delta_{t_2}$  versus  $N / \sqrt{\theta}_a$  at (7) and (8)

Case II: (11) greater than (10)

- 14.  $(N/\sqrt{\theta_{as}})_{max}$ , corrected rpm corresponding to (10) from plot of (6) and (7) at (10)
- 15.  $N_{max}$  standard maximum engine speed, (14) x (5)
- 16.  $\Delta F_n / \delta_{t_2}$  from plot of  $F_n / \delta_{t_2}$  versus N/  $\sqrt{\theta_{at}}$  (14) and (7)

### 8.3 DETERMINATION OF NOZZLE THRUST COEFFICIENT

- 1. F from the mechanical thrust measuring equipment
- 2. As from measurements of the exhaust nozzle
- 3. Pa, static pressure to which the nozzle is discharging, from barometer or altimeter
- 4. Ptg, instrument corrected total pressure from probe(s) located in exhaust nozzle
- 5.  $P_{tg}/P_a$ , nozzle pressure ratio, (4)/(3)
- (Fg/A8Pa)<sub>theo</sub>, theoretical gross thrust parameter from Chart 9.17 or 9.18 and (5)
- 7.  $(F_g/A_8P_a)_{actual}$  (1)/(2) (3)
- 8. Cf, nozzle thrust coefficient, (7)/(6)
- 9. Plot (8) versus (5)
# Chart 9.1 RELATION BETWEEN TOTAL PRESSURE RECOVERY AND RAM EFFICIENCY



Total Pressure Recovery, Pt<sub>2</sub>/Ps

3-68



TOTAL PRESSURE RECOVERY FOR INLETS WITH SHARP LIPS Chart 9.2

3- 69

Mass Flow Ratio, m<sub>l</sub>/m<sub>l</sub>\*

TURBULENT BOUNDARY LAYER THICKNESS FOR FLAT - PLATES AT ZERO ANGLE OF ATTACK AS A FUNCTION OF FLIGHT SPEED AND ALTITUDE Chart 9.3





Chart 9.4 (a) PRESSURE RECOVERY OF BOUNDARY LAYER AIR

Total Pressure Recovery, Pt1/Pto, of Injected Air

Chart 9.4(b) PRESSURE RECOVERY OF BOUNDARY LAYER AIR ADMITTED INTO SIDE-INLET INSTALLATION

- LAMINAR FLOW



Total Pressure Recovery, Pt/Pt<sub>0</sub>, of Injested Air



\$1/\$0

Boundary Layer Total Momentum Ratio,

Chart 9.5 TOTAL MOMENTUM RATIO FOR VARIOUS SCOOP HEIGHT TO BOUNDARY LAYER RATIOS

3-73

TOTAL PRESSURE RECOVERY FOR STRAIGHT SUBSONIC DIFFUSERS Chart 9.6 (a)



Chart 9.6 (b) TOTAL PRESSURE RECOVERY FOR STRAIGHT SUBSONIC DIFFUSERS



Chart 9.7 TOTAL PRESSURE LOSS IN COMPOUND SUBSONIC DIFFUSER BENDS













# Chart 9.10 TOTAL - PRESSURE RATIOS FOR 2 DIMENSIONAL 3 - SHOCK COMPRESSION





# Chart 9.11 TOTAL PRESSURE RATIOS FOR CONICAL 2-SHOCK COMPRESSION







Chart 9.12 MACH NUMBER CHANGE THROUGH AN OBLIQUE SHOCK FOR A TWO DIMENSIONAL WEDGE



Chart 9.13 TOTAL PRESSURE RATIO ACROSS AN OBLIQUE SHOCK FOR A TWO DIMENSIONAL WEDGE



3-82



THEORETICAL ADDITIVE - DRAG CUEFFICIENTS FOR OPEN - NOSED INLETS Chart 9.14





# Chart 9.16 CHANGE IN COWL DRAG COEFFICIENT WITH A CHANGE IN MASS FLOW RATIO AS A FUNCTION OF MACH NUMBER



3-85









NOZZLE PRESSURE RATIO, P./P.



IDEALIZED GAS FLOW WITH SUPERCRITICAL OPERATION - CONVERGING NOZZLE Chart 9.18



DEALIZED GAS FLOW WITH SUPERCRITICAL OPERATION - CONVERGING NOZZLE Chart 9.18



3-89



Chart 9.19 GROSS THRUST PARAMETER VERSUS NOZZLE PRESSURE RATIO WITH SUBCRITICAL OPERATION

# GROSS THRUST PARAMETER VERSUS NO ZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION - CONVERGING NOZZLE Chart 9.20



GROSS THRUST PARAMETER VERSUS NOZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION - CONVERGING NOZZLE Chart 9.20



GROSS THRUST PARAMETER VERSUS NOZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION - CONVERGING NOZZLE Chart 9.20



GROSS THRUST PARAMETER VERSUS NOZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION - CONVERGING-DIVERGING NOZZLE Chart 9.21



GROSS THRUST PARA METER VERSUS NOZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION - CONVERGING-DIVERGING NOZZLE 9.21 Chart





# Chart 9.21 GROSS THRUST PARAMETER VERSUS NOZZLE PRESSURE RATIO WITH SUPERCRITICAL OPERATION -CONVERGING-DIVERGING NOZZLE





# P./P. VERSUS MACH (RAYLEIGH SUPERSONIC





MACH NUMBER

Chart 9:22

P, /P VERSUS MACH (RAYLEIGH SUPERSONIC



### CHAPTER FOUR

### LEVEL FLIGHT PERFORMANCE

### SECTION 4.1

# Density Altitude and Pressure Altitude Flight Test Methods

Aircraft level flight performance analysis is the process of determining standard day level flight characteristics from data obtained during nonetandard conditions. Until the advent of high speed aircraft and the accompanying coapressibility effects most flight test data were reduced by what is referred to as the "Density Altitude" method. With jet powered aircraft came the necessity of standardizing data for what might be called constant compressibility conditions, thus avoiding compressibility corrections. This latter type of data reduction is called the "Pressure Altitude" method.

The density altitude method of flight test data reduction has been used and, in many cases, is still used in the speed range where the assumption of constant drag for constant true speed and density altitude is valid. However, where sffects of compressibility are not negligible this method will result in erroneous standard day data. The pressure altitude method is based on the concept of maintaining a constant pressure altitude and indicated air speed and correcting data only for temperature to obtain standard day performance. With these identical test and standard day indicated air speeds the test and standard day Mach numbers will be the same. This may be easily seen by examination of the Mach number equation ( 2.23) in terms of  $q_{c}$  and  $P_{a}$ .

It is shown in aerodynamic theory that total drag is a function only of Mach number if weight and altitude are fixed. (Reynolds number effects are generally ignored in flight test work.) These facts are the basis for the simplicity and effectiveness of the pressure altitude method of flight data reduction. Using it, the performance engineer need make only temperature corrections to his test day data, and compressibility effects are automatically held constant.

As a general rule the pressure altitude method is applied to both reciprooating and jet engine aircraft. For, even in the low speed range, this method simplifies data reduction procedures in all departments of aircraft performance. In the case of reciprocating engine sizeraft there is one exception to the pressure altitude method. In this exception a power parameter (PIN) is plotted against a speed parameter (VIN) to obtain a single, standard day, weight corrected, sea-level, power-required polar for all level flight data. This PIW-VIN polar vill not be valid in the compressibility speed range.

APTR 6273

### SECTION 4.2

# Aerodynamic Forces and Their Relation to Engine Power and Propulsive Thrust

The acrodynamic forces acting on an airfoil are assumed to be functions of its size, angle of attack, speed, and atmospheric conditions (temperature, pressure, and viecosity of the air through which it is flying). The two primary forces acting on the unaccelerating aircraft are called the lift force and the drag force, and these may be defined by generalized nondimensional equatione:

Letting the lift force equal the aircraft gross weight, the drag force may be redefined by use of the above equations.

By the methods of dimensional analysis equation 4.203 may be transformed to the formi

 $\frac{\mathbf{D}}{\mathbf{P}_{\mathbf{o}}} = \mathbf{f} \left( \frac{\mathbf{W}}{\mathbf{P}_{\mathbf{o}}} \, \mathbf{M}_{\mathbf{r}} \, \mathbf{R}_{\mathbf{o}} \right)$ (4.204)

(In this form the constant aircraft size factor is cmitted)

where:

D = total aircraft drag

- P<sub>A</sub> = atmospheric pressure W = aircraft gross weight
- M = Mach number
- Re = Reynolds number

In most flight test aircraft analysis the small variations of data with Reynolds number are neglected.

By lengthy analytical methods it is possible to develop a physical equation that will approximately define the drag force.

 $D = C_{D4}S$ 

(4.205)

The total drag coefficient, CD, is further defined by analytical methods and equals the sum of the profile and induced drag coefficients.

 $C_{D} = C_{DP} + C_{Di}$ 

(4.206)

APTR 6275

With this equation 4.205 becomes

$$D = C_{DP}(qS) + C_{D1}(qS)$$
 (4.207)

The induced drag coefficient, C<sub>D1</sub> is further defined by analytical methods

$$c_{\rm D1} = \frac{c_{\rm L}^2}{\pi A \epsilon}$$
 (4.208)

And the lift coefficient  $C_L$  is defined by analytical methods as:

$$C_{\rm L} = \frac{W}{QS} \tag{4.209}$$

Substituting these last two equations in 4.207, a final analytical expression for the drag force is obtained.

$$D = C_{DP}(qS) + \frac{W^2}{\pi R^2(qS)}$$
(4.210)

where:

$$q = \frac{1}{2} \rho v^{2} = 0.7 P_{a} h^{2}$$
  
W = aircraft gross weight  
At = wing aspect ratio  
B = wing area  
e = wing efficiency factor =  $\frac{C_{L}^{2}}{(C_{D}-C_{TR})7/A}$ 

Experimental wind-tunnel data show that Cpp is primarily a function of Mach number and lift coefficient, remaining constant until compressibility effects are evident at (M = 0.5). In this compressible range Cpp may also be appreciably affected by changes in  $C_{\rm L}$ . It should be noted that the validity and usefulness of all aircraft performance parameters depend on the validity of the prevailing assumptions concerning Cpp.

Assuming Cpp to remain constant for the speed range of most reciprooating engine aircraft, several functional modifications of equation 4.210 may be derived in terms of speed, Mach number, thrust horsepower, gross weight, and atmospheric conditions.

$$THP = C_{DP} \sigma V^{3} Sk_{I} + \frac{W^{2}k_{2}}{\sigma V D^{2}e}$$
(4.211)

In this form (THP) is a function of true velocity, density ratio and gross weight

$$\operatorname{THP} \mathcal{T} = k_1 c_{DP} \left( \sigma v^2 \right)^{3/2} 8 + \frac{v^2 k_2}{\sqrt{\sigma v^2} v^2}$$
(4.212)

AFTR 6273

In this form (THPVG ) is a function of  $(\sigma v^2)$  or  $v_0^2$  and gross weight.

$$\frac{TBP\sqrt{\sigma}}{\sqrt{3/2}} = C_{DP}k_1 \left(\frac{\sigma v^2}{w}\right)^{3/2} \mathbf{s}^{+} \frac{k_2}{\left(\sqrt{\frac{\sigma v^2}{w}}\right)^{5/2}}$$
(4.213)

In this form (THP  $\sqrt{\sigma}/W^{3/2}$ ) is a function only of  $(\sigma V^2/W)$ 

$$\frac{\text{THP}}{P_{a} \sqrt{T_{a}}} = C_{\text{DP}} M^{3} S k_{3} + \left(\frac{W}{P_{a}}\right)^{2} \frac{k_{4}}{Mb^{2} \bullet}$$
(4.214)

In this form  $(THP/P_a \sqrt{T_a})$  is a function of Mach number and  $(W/P_a)$ 

$$\frac{D}{P_a} = k_5 C_{DP} M^2 S + \left(\frac{W}{P_a}\right)^2 \frac{k_6}{M^2 b^2 c}$$
(4.215)

In this form  $(D/P_{a})$  is a function of Mach number and  $(W/P_{a})$ . These last two equations are both valid in the compressible speed range because M and  $(W/P_{a})$ dofine Cpp. In the incompressible range Cpp is constant for all Mach numbers and  $(W/P_{a})$ 's; in the compressible range Cpp as a function of N must be plotted for separate (W/P) parameters. It should be noted that at a constant Mach number an increasing value of  $(W/P_{a})$  corresponds to an increase in CL. It should also be noted that equation 4.204, derived by dimensional analysis, verifies 4.215.

### The following notation applies to equations 4.211 through 4.215.

THP = (DxV)/326, or BHPx  $\gamma_P$ , = thrust horsepower  $\gamma_P$  = Propeller efficiency  $O' = P/P_{SL}$ , = 9.625 P<sub>a</sub>/T<sub>a</sub>, density ratio V = True speed, knots S = Wing area, ft<sup>2</sup> W = lbs, gross weight b<sup>2</sup> = ft<sup>2</sup>, ArxS, wing span • Airplane efficiency factor V  $\sqrt{O}$  = V<sub>e</sub> = knots, equivalent speed P<sub>a</sub> = inches Hg, atmospheric pressure T<sub>a</sub> = <sup>O</sup>Kelvin, atmospheric temperature M = Mach number, V/38.94  $\sqrt{T_a}$ , or  $\sqrt{1/.7}$  P<sub>a</sub> D = lbs, drag or propulsive thrust, F<sub>D</sub> k<sub>1</sub> = 1.0414 x 10<sup>-5</sup> k<sub>2</sub> = .28820 k<sub>3</sub> = 5.9205 k<sub>4</sub> = 7.6885 x 10<sup>-4</sup> k<sub>5</sub> = 49.5089 k<sub>6</sub> = 6.4293 x 10<sup>-4</sup>

Graphically equations 4.210 through 4.215 all take the same general parabolic form as shown in Figure 4.21. Certain useful information concerning each type of plot is noted.

AFTR 6273


ATTE 6273

#### SECTION 4.3

#### Speed Pover Curves - Reciprocating Engine Aircraft

Equation 4.214 is the basis for the pressure altitude method of level flight data reduction. If a series of points is flown at a constant pressure altitude and weight, Mach number may be plotted against THP/ $\sqrt{T_a}$ .

THP	•	(M)	weight constant	
$\sqrt{T_a}$	*	(11)	pressure constant	(4.301)

Since engine brake horsepower is the desired power criteria, the difference between brake horsepower and thrust horsepower must be considered.

THP = propeller efficiency 
$$(\gamma_p)$$
 x BHP

The actual determination of the propeller efficiency is not generally required, because performance is to be measured in terms of engine shaft power. In order to insure that the performance parameters are valid on a standard day or a test day, it is necessary to consider the variation of propeller efficiency between two points flown at the same Mach number and pressure altitude but at different temperatures. Experience has shown that this variation in propeller efficiency is usually negligible, so equation 4.301 is valid in terms of brake horsepower for a constant weight and a constant pressure altitude.

 $\frac{BHP}{\sqrt{T_{a}}} = f(M)$  (4.302)

For convenience in plotting the horsepower parameter is written,  $BHP_t \sqrt{T_{as}}/\sqrt{T_{at}}$ , where  $T_{as}$  is the standard day temperature at the pressure altitude <u>under</u> of origideration. This notation has a major value in that the radical,  $\sqrt{T_{as}/T_{at}}$ , equals unity on a standard day and the plot shows directly the standard day horsepower required to produce any given Mach number.

Another form of pressure altitude plot may be derived by considering Mach number as a function of calibrated speed  $(V_c)$  and pressure altitude  $(H_c)$  as defined in Chapter One.

BHP  $\sqrt{\frac{T_{ab}}{T_{at}}} = f(V_c)$  weight constant atmospheric pressure constant (4.303)

Typical plots of the horsepower parameter, BHP  $\sqrt{T_{as}/T_{at}}$ , va M and V<sub>c</sub> are shown in Figure 4.31.

From plots such as those in Figure 4.31 A & B the graphs of standard-day brake horsepower vs true speed may be drawn by simply converting  $V_0$  and  $H_0$ , or M and  $T_{as}$  for  $H_0$ , into standard-day true speed. In fact, this standard true speed may be computed and plotted vs BHP x  $\sqrt{T_{as}}/T_{at}$  as in Figure 4.31 without making the  $V_0$  or M plot shown. It should be remembered that the aircraft weight

AFTR 6273

has been assumed constant. Actually each level flight at a constant altitude will have to be at a different weight and will have to be corrected to a constant weight.

Cross plots of the BHP vs  $V_i$  plots and engine data at the various altitudes are made for report presentation to show standard day true speed altitude plots for normal rated power and for military power as shown in Figure 4.32.



Figure 4.31 Typical Pressure Altitude Speed-Power Plots for Level Flight

AFTR 6273

Level Flight Data

Intercooler Flaps 2° • Approximately 10,000 Ft. Intercooler Flaps 5° • Approximately 20,000 Ft. Intercooler Flaps 5° • Approximately 30,000 Ft. Military Power 2700 RPM,60 MBG



AFTR 6273

## SECTION 4.4

## <u>Weight Corrections for Speed Power Data</u> - Reciprocating Engine Alrcraft

The speed power relationships would be completely defined if all tests could be run at desired weight at constant pressure is induces; however, varied test conditions and fuel consumptions generally make this impossible. A weight correction is usually made to all data at a given pressure altitude to make it represent a fixed weight. The standard weight is usually defined as the weight the aircraft would have if it started at its normal take-off gross weight and climbed to the specified altitude at best climb power and speed settings. The correction is made by considering the isolated effect of a change of weight on BHP<sub>t</sub>  $\sqrt{T_{as}}/T_{at}$  at a constant Mach number or V<sub>c</sub> and pressure altitude.

Since the correction is to be made at a constant Mach number and will involve only small changes in angle of attack, Cpp will be assumed constant. Using equation 4.214 with Cpp held constant;

$$\Delta \left( \frac{\text{THP}}{P_{a} \sqrt{T_{a}}} \right) = \Delta \left( \frac{\mathbf{w}^2 \, \mathbf{k}_{\mu}}{P_{a}^2 \mathbf{M} \, \mathbf{b}^2 \, \mathbf{e}} \right)$$
(4.401)

Converting to brake horsepower and a constant pressure altitude, temperature, and Mach number

$$\Delta BHP \sqrt{\frac{T_{as}}{T_{at}}} = \frac{\Delta W^2}{T_{p} P_a} \frac{\sqrt{T_{ag}}}{M b^2 e}$$
(4.402)

vhere:

 $\Delta BHP = BHP_8 - BHP_t$   $\Delta W = W_8 - W_t$   $h_a = 7.6885 \times 10^{-4}$   $M_P = 0.83, average value$  $\bullet = 0.77, average value$ 

The values of propeller efficiency and airplane efficiency within normal flying speeds of propeller driven aircraft are approximately constant. At speeds less than 50% greater than stall speed, differences between test weight and standard weight should be maintained less than 20% unless detailed information regarding "N<sub>p</sub>" and "e" is available. For general work a propeller efficiency of .83 and an airplane efficiency of .77 are assumed. From equation 4.402, CHART 4.41 at the end of this chapter has been made giving  $\triangle BHP \sqrt{T_{as}/T_{at}}$  from standard weight, change in weight, wing span, test Mach number, and pressure altitude. Hotice should be taken that the  $\triangle BHP \sqrt{T_{as}/T_{at}}$  is the total change, while speed power graphs usually present horsepower/engine.

AFTR 6273

## DATA REDUCTION OUTLINE (4.41)

For Determining Weight-Corrected Standard BHP ve V t end H  $_{\rm C}$ 

			<b>.</b>
(1)	vı	knote	Indicatud air speed
(2)	Δ ν10	knota	Air speed instrument correction
(3)	W <sub>t</sub>	160	Gross weight test, take-off weight less
	-		lbs. fuel consumed
(4)	Δ٧ <sub>200</sub>	knots	Air-speed position correction corres-
	-		ponding to (1) and (3) and calibration
•			data
(5) (6)	٧ <sub>0</sub>	knote	Calibrated air speed $(1) + (2) + (4)$
	H.	feet	Indicated pressure altitude
(7)	∆ <sup>1</sup> H <sub>10</sub>	feet	Altimeter instrument correction
(8)	Δ H <sub>po</sub>	feet	Altimater position correction corres-
			ponding to (1) and (3) and (6) and
			Galibration data
(9)	H <sub>a</sub>	feet	True pressure altitude, $(6) + (7) + (8)$
(10)	M		Mach number, from (5) and (9) and CHART
			8.5
(11)	ti	<b>°C</b>	Indicated air temperature
(12)	∆ <sup>¯</sup> t <sub>ic</sub>	°C	Temperature instrument correction
(13)	tio	•C	Indicated instrument operacted air
			temperature $(11) + (12)$
(14)	tat	•C	Test free air temperature from (10) and
			(13) and CEART 8.2
(15)	tas	°C	Standard temperature for (9)
(16)	$\sqrt{T_{as}/T_{at}}$		$\sqrt{(15) + 273} / \sqrt{(14) + 273}$
(17)	BHP.		Test brake horsepower per engine from
• • • •	- U		torquemeter or power chart
(18)	$BHP_t \sqrt{T}$	AB/Tat	(16) x (17), per engine
(19)	W <sub>0</sub>	108	Standard gross weight selected
(20)	Å۷	1 <b>be</b>	$(W_{\mu} - W_{\pm})$
(21)	∧ BRP. 1/	Tas/Tat	Horpower weight correction at $W_{\pm}$ , $\Delta W_{\pm}$
(/		*80/ *8t '	b <sup>2</sup> , M and altitude, using CHARTS 4.41
			thru 4,46
(22)	$\operatorname{BHP}_{ty} \mathcal{N}$	P. /P.	Weight corrected standard BHP, (21) +(18)
(23)		mota	Standard day true speed (10) $x[(15)+273]$
(2)	V <sub>te</sub>	LIGCE	x 38.944 or CHART 8.5 and (5) and (9)
(24)	Plat (22	) ve (23) or (5) or (	(10), and (9) as shown in Figure 4.31.
(67)	1100 (26		Teall mus (2) an mount of sedara 4.21.
	NOTE: T	he standard day. vai	ght-corrected BHP (22) requires corrections
*			perature, manifold pressure, turbo rpm, and
•			If the particular speed power point is
•	•		

to carburetor air temperature, manifold pressure, turbo rpm, and exhaust back pressure. If the particular speed-power point is at full throttle the standard power (22) may not be obtainable and a speed correction will be in order. These engine condition corrections are determined by the methods of Chapter Two.

AFTR 6273

4-10 .

4

٩

## SECTION 4.5

#### Configuration Change Corrections for Speed Pover Data

The preceding sections developed a method for determining speed vs power at epscified weights, altitudes, and fixed configurations. From this data generalizations must be made to allow computations of performance at all possible weights; altitudes, and configurations. For reciprocating engine power aircraft these computations are all made on the basis of incompressible flow theory.

The first requirement is for information regarding the changes in power required for a given speed change caused by minor changes in configuration such as opening or closing covi flags, oil coolers, intercoolers, etc. To present this information from minimum flight test work, the assumption is made that, at a given speed, a minor change in configuration will not change the coefficient of induced drag (O<sub>D1</sub>).

then	Profile Drag = $C_{DP} - \frac{\rho}{2} V_t^2 S$	
	∆Profile Dreg = ∆Thrust	
and	$\Delta T = k S \Delta C_{DP} \sigma v_t^3$	(4.501)
or,	$\Delta THP 6 = k S \Delta O_{DP} (V, Vor)^3$	(4.502)

where:

- ATHP increment of THP required to balance the effects of an increment of  $\Delta C_{\rm DP}$ k = 1.0414 x 10づ
- $\Delta C_{\rm DP}$  = increment of  $C_{\rm DP}$  caused by configuration change
  - V<sub>t</sub> - true air speed, knots

If  $V_t \sqrt{\sigma}$  and  $C_{D1}$  are held constant, any change in THPV $\sigma$  will be a function of the change in Cpp. Within a small range of  $V\sqrt{\sigma}$  Propeller Efficiency,  $M_{\rho}$ , remains constant and ( $\sqrt{\theta}$  THP= $\sqrt{\sigma}$  BHP  $\eta_p$ ). Since  $C_{DP}$  is a function of size and shape, it will change with the configuration change. Therefore, by running power calibrations at one altitude and weight while changing configuration, the effects of configuration on BHP,  $V_t$ , or  $V_e$  may be determined. A typical calibration of cowl flap position effects is shown in Figure 4.51.

The values of  $\triangle \mathcal{BHP} / \mathcal{F}$  are applicable at any altitude and can be applied to any power calibration to determine the power required for the specified configuration. Notice should be taken that these changes apply to power required only. Range and top speed can, in some cases, be increased in spite of an inoreased cooling drag, because engine operating limits are raised under lowsr temperature conditions.



Bigure 4.51 Cost Flap Efforts on Power

APTR 6275

문화되었는 일을 가르는 것이다.

4+12

## SECTION 4,6

## The Generalized Power Parameter (PIW) and Speed Parameter (VIW) - Reciprocating Engine Aircraft

The problem of generalizing data for all weights and altitudes is accomplished by the speed-power polar, PIW vs VIW plot. This plot presents all speed power information with a minimum amount of data. The parameters for this plot are determined from equation 4.213. By inserting some constant gross weight in this equation, two easily calculated terms are defined which completely resolve to a single curve all flight conditions for a given configuration.

$$\frac{\text{THP}\sqrt{\sigma}}{(W_t/W_g)^{3/2}} = k_1 C_{\text{DP}} S \left(\frac{\sigma V_t^2}{(W_t/W_g)}\right)^{3/2} + \frac{k_2 W_g^2}{\left(\frac{\sigma V_t^2}{W_t/W_g}\right)^{\frac{1}{2}b^2 \cdot e}}$$
(4.601)

where:

 $k_1 = 1.0414 \times 10^{-7}$   $k_2 = .28820$   $HP = N_{P}BHP$   $\eta_{P} = propeller efficiency$ 

Assuming the propeller efficiency to be virtually constant for given ranges of  $\mathcal{O} V^2$ , equation 4.601 may be written:

$$\operatorname{BEP} \sqrt{\sigma} \frac{W_{e}}{W_{t}}^{3/2} = f\left[ \nabla_{t} \left( \sigma \frac{W_{e}}{W_{t}} \right)^{\frac{1}{2}} \right]$$
(4.602)

The left side is called "PIV." The right side is called "VIV." A t<sub>f</sub>  $\perp$  PIW-VIW plot is illustrated in Figure 4.61. Generally the standard take-off gross weight is used as  $W_{\rm m}$ .

The validity of the PIW-VIW plot can be demonstrated by dimensional analysis methods. By this means it can be shown that for a given configuration and propeller the parameters,

$$\frac{\operatorname{BEP} \sqrt{\rho}}{\sqrt{2}}, \frac{\pi \sqrt{\rho}}{\sqrt{2}}, \frac{\sqrt{\gamma}}{\sqrt{2}}$$

where: N is the engine rpm

will define speed power performance at any altitude and weight as shown in Figure 4.62. Since the three parameters can be presented on one graph the plot is most useful for estimating general performance and determining design criteria. In practice the parameters are divided by the appropriate constant standard weight to give PIW, VIW, and WIW.

AFTR 6273



Figure 4.61 Typical FIW-VIW Plot

In practical applications the value of propeller efficiency will be approximately constant over most of the VIW range. In that case only the parameters PIW and VIW will be present, and the plot is very valuable for presentation and standardization of test data as discussed previously. Since the PIW-VIW plot is the aircraft polar reduced to sea-level, standard-weight conditions, only a change in propeller efficiency could produce more than one parameter for a given consideration. At very low or high flight speeds changes in propeller efficiency may be noticeable.



Figure 4.62 PIW-VIW Plot With rpm Parameters

AFTR 6273

# DATA REDUCTION OUTLINE (4.61)

# For PIW-VIW Plot

(1)	v <sub>1</sub>	knote	Indicated air speed
(2)	۵v <sub>io</sub>	knote	Air-speed instrument correction
(3)	We.	16.	Test gross weight
(4)	ڴ٧ <sub>٥٥</sub>	knots	Air-speed position error corresponding
			to (1) and (3) and calibration data
(5) (6) (7) (8)	Y <sub>0</sub>	knote	Calibrated air speed $(1) + (2) + (4)$
(6)	H <sub>1</sub>	feet	Indicated pressure altitude
$(\underline{7})$		feet	Altimeter instrument correction
(8)		føet	Altimeter position correction corres-
	•		ponding to (1) and (3) and (6) and
	_		calibration data
(9) (10)	H <sub>o</sub> Pa N	feet	True pressure altitude, $(6) + (7) + (8)$
	Pa	"Eg	Atmospheric pressure corresponding to (9)
(11)	N		Mach number from (5) and (9) and CHART
(10)	•	••	8.5
(12)	£10 At10	°C	Indicated temperature
(13) (14)		°C	Temperature instrument correction
(14)	tio	0C	Indicated instrument corrected temper-
(15)	•	°C	ature (12) + (13)
(1)	tat	-0	Test free air temperature from (11) and
			(14) and CHART 8.2
(16)	<b>V</b> ∂F		$\sqrt{9.625(10)}/\sqrt{(15)+273}$
(17) (18)	Y. A. A	lbe	Standard groes, weight $[(5) + (17)]^2$
	$(W_t/W_B)^2$	_	$(3) + (17)^{2}$
(19)	$(W_{t}/W_{s})^{\frac{1}{2}}$ $(W_{t}/W_{s})^{\frac{1}{2}/2}$	2	$(5) + (17) ]^{3/2}$
(20)	Vtt	Inote	Test true speed 38.944 (11) x V(15) + 275
<b>.</b> .	••		or (11) and (15) and CHART 8.4
(21)	VIV	Imote	$(20) \times (16) + (18)$
(22)	BEPt		Test brake horsepower from torque meter
(0)			or power chart
(23) (24)	PTW		$(22) \times (16) + (19)$
(24)	Flot (23)	ve (21) for each	configuration.

AFTR 6273

## SECTION 4.7 Fuel Consumption - Range and Endurance - Reciprocating Engine Aircraft

FUEL CONSUMPTION AND BSFC

Data relative to fuel consumption is obtained in flight whenever possible, rather than by use of the engine manufacturer's data. Flight fuel flow data is most accurately obtained by use of timed fuel totalizer readings or directly by use of rate of flow meters. In either case volume flow must be converted to weight flow. Generally, gmsoline is considered to have an average sea level standard weight of 6.0 lbs/gal. If more accurate measurements are desired, where large quantities of fuel are involved at very low temperatures, the specific gravity should be determined before the flight and be used with a temperature correction factor to approximate in-flight specific gravity. This is only necessary where long-time high-altitude flights are involved and test gross weight may be appreciably affected. In most test work use of the beforeflight specific gravity is sufficient.

$$W_{f} = \frac{g_{a} l_{g}}{hr} \pm 8.339 \pm Sp g$$
 (4.701)

where:

In report presentation of fuel consumption or range data the test results should be corrected to a 6.0 lbs per gallon standard for gasoline.

The brake specific fuel consumption (BSFC) is determined from flow data taken during the normal power calibrations at various engine settings.

$$BSIPC = \frac{W_{f}}{BHP}$$

This data is usually plotted vs BHP, slongside the speed-power curves as shown in Figure 4.71.



\* (4.702)



AFTR 6273

## SPECIFIC RANCE

Because the range charts are used for obtaining both flight distance and determining flying technique, the range data is sometimes plotted vs both true and calibrated or indicated speed. Range is not plotted as such except for sample mission studies. The usual presentation is specific range (SRg) vs true, calibrated or indicated speed.

$$SRg = \frac{V_t}{V_f} = \text{mantical air miles/lb}$$
(4.703)



A typical plot is shown in Figure 4.72.

Figure 4.72 Specific Range Data - Constant Gross Weight

Notice should be taken that the altitudes shown on the brake specific fuel consumption graph are the various density altitudes at which the points were flown. The altitudes on the speed-power graphs are pressure altitudes. Experience has shown that, considering the small differences between test pressure altitude and test density altitude, fuel flow accuracy will not be measurably effected by assuming the fuel flow and speed-power graphs to agree at the same pressure altitude. In many cases specific fuel flow is completely independent of altitudes, but usually BEFC will increase with altitude for at least part of the altitude range.

Specific range data can be corrected for weight variations, but, because the BEFC may vary with altitude, the weight correction should not be applied across large altitude increments. Given the same density altitude, and mixture setting,

SRJ2 -	SEC1	(¥ <u>1</u> ) (¥ <u>2</u> )	(4.704)
--------	------	--------------------------------	---------

AFTR 6273

When SRg changes due to a weight change, V<sub>t</sub> must change;

$$v_{t2} = v_{t1} \sqrt{\frac{W_2}{W_1}}$$
 (4.705)

## SPECIFIC ENDURANCE

Maximum specific endurance (SEmax) can be obtained from a fuel consumption plot made alongside the BHP<sub>g</sub> vs V<sub>t</sub> or V<sub>c</sub> graphs as in Figure 4.73. Specific endurance is defined as the reciprocal of the fuel flow, W<sub>f</sub>.



Figure 4.73 Method of Presenting Fuel Consumption Data

Specific endurance data may be corrected for weight variation, but, because the BSFC may vary slightly with density altitude the correction should not be applied across large altitude increments. At the same density altitude and mixture setting,

		$/(12 ) \sqrt{3/2}$	
50°2	• SE <sub>1</sub>	$ \begin{pmatrix} (\underline{w_1}) \\ (\underline{w_2}) \end{pmatrix}^{3/2} $	(4.706)
•			

When SE changes due to a weight change  $V_t$  must change,

$$v_{t2} = v_{t1} \sqrt{\frac{w_2}{w_1}}$$
 (4.707)

AFTR 6273

#### ACTUAL RANGE AND ENDURANCE

From plots of specific range (nautical air miles per pound of fuel) vs true speed and altitude for each weight condition it will be possible to obtain the range for any desired cruising condition along with the corresponding BEP, rpm, manifold pressures and indicated speeds. For some aircraft performance reports it may be required that a test tactical mission be flown to compare expected and actual results, perhaps for a radius of action problem.

Actual range is best determined from a plot of the specific range parameter defined by equations 4.704 and 4.705 for a constant density altitude.

$$\frac{\nabla_{\mathbf{t}} \mathbf{W}}{\mathbf{W}_{\mathbf{f}}} = \mathbf{f} \left(\frac{\nabla_{\mathbf{t}}}{\sqrt{\mathbf{W}}}\right) \tag{4.708}$$

where:

 $\frac{V_{L}W}{W_{c}}$  = specific range parameter  $\frac{V_t}{\sqrt{W}}$  = speed parameter

Equation 4.708 is valid for both test and standard day conditions. For a constant value of the speed parameter the range is,

$$\mathbf{R}_{g} = \frac{\mathbf{v}_{t} \mathbf{W}}{\mathbf{W}_{f}} \int_{\mathbf{W}_{1}}^{\mathbf{W}_{2}} \frac{1}{\mathbf{W}} d\mathbf{W} \qquad (4.709)$$

where:

dW = fuel weight differential (negative)

Integrating, the range becomes,

$${}^{\mathbf{B}}_{\mathbf{g}} = \left(\frac{\mathbf{v}_{\mathbf{t}}}{\mathbf{w}_{\mathbf{f}}}\right)^{-\ln} \left(\frac{\mathbf{w}_{\mathbf{i}}}{\mathbf{w}_{\mathbf{2}}}\right)$$
(4.710)

Values of ln  $(W_1/W_2)$  may be determined from CHART 4.71 at the end of this chapter.

Actual endurance is best determined from a plot of the specific endurance from a plot of the specific endurance parameter defined by equation 4.706 and 4.707 for a constant density altitude.

$$\frac{w^{3/2}}{w_{f}} = f \left(\frac{v_{t}}{\sqrt{w}}\right) \qquad (4.711)$$
where:  

$$\frac{w^{3/2}}{w_{f}} = \text{specific endurance parameter}$$

$$6273 \qquad 4-19$$

ATTE 6273

- 1-

For a constant value of the speed parameter the endurance is,

$$\mathbf{E} = \left(\frac{\mathbf{W}^{3/2}}{\mathbf{W}_{f}}\right) \int_{\mathbf{W}_{1}}^{\mathbf{W}_{2}} \frac{1}{\mathbf{W}^{3/2}} \, \mathrm{dW}$$
(4.712)

-----

Integrating, the endurance becomes,

$$\mathbf{E} = \begin{pmatrix} \frac{3/2}{W_1} \end{pmatrix} \quad \begin{pmatrix} \frac{2}{\sqrt{W_2}} & -\frac{2}{\sqrt{W_1}} \end{pmatrix}$$
(4.713)

Values of  $2/\sqrt{W}$  may be determined from CHART 4.72 at the end of this chapter.

## DATA REDUCTION OUTLINE (4.71) For BSFC and Range Data - Reciprocating

(To be used with BHP Data Reduction Outline in Section 4.4)

(25) (26)	Vet	gals/hr	Test volumetric fuel flow
•	W <sub>ft</sub> W <sub>ft</sub>	lbs/hr	Test fuel weight-flow (25) X (conversion factor determined by fuel and temp)
(27) (28) (29)	BSTC	lbs/BHP-Hr	Brake specific fuel consumption $(26) + (17)$
(28)	Plot (27) vs	(23)	<b>-</b> 7
(29)	SRg	nautical air miles per pound	Specific range, $[(23) + (16)] + (26)$
(30)	Plot (29) Va	(23) or (5), at ea	ch altitude

ATTR 6273

#### SECTION 4.8

## Speed-Power Curves - Turbojet Aircraft

The aircraft drag parameter is a function of Mach number and the weightpressure parameter as shown in equation 4.125.

$$D/P_{a} = f(M, W/P_{a})$$
 (4.801)

In jet-powered aircraft in stabilized level flight the propulsive or net jet thrust equals the aircraft's drag. For flight performance data engine rpm is a more convenient engine criterion than thrust horsepower or drag. As previously shown in Chapter Three.

$$F_n/P_a = f(M_s N/\sqrt{T_a})$$
 (4.802)

By equating equations 4,801 and 4.802,

$$N/(T_{a} = f(M, W/P_{a})$$
 (4.803)

For convience equation 4.803 is written,

$$N/\sqrt{\theta_n} = f(M, W/\delta_n)$$
(4.894)

where

$$\theta_{a} = T_{a}/288$$

$$\delta_{a} = P_{a}/29.92$$

$$M = f(V_{c} \text{ and } P_{a})$$

$$N = rpm$$

If the value of the W/ $\delta_a$  parameter is fixed, N/ $\sqrt{\theta_a}$  versus M curves will define the speed power relationship.

Speed-power tests in turbojet aircraft are flown by setting an rpm and holding the aircraft at a specified pressure altitude until the speed is stabilized. This is done for each test point. However, at low speeds near the stall condition, jet airplanes will not stabilize well. This is because the thrust decreases as the speed decreases so that a condition may be reached where at a constant rpm the speed will slowly fall until the aircraft stalls. The aircraft can never be successfully operated in this range, but flight tests sometimes require drag evaluation at low speeds. In order to include this low speed range in the power required curves, a system has been devised to gather the necessary data from a test in which the aircraft is allowed to descend slightly to maintain its speed. From climb data reduction (Chapter Five),

or

$$W$$
  
R/D =  $\frac{101.3 V_t}{101.3 V_t}$  (D - F<sub>r</sub>

 $R/C = \frac{101.3 V_t}{T_n} (F_n - D)$ 

where:

R/D = rate of descent, ft/min

- W = weight 1bs
- $\nabla_t$  = true speed knots  $F_n$  = net thrust lbs
- $\ddot{D}$  = drag 1bs. Assumed constant for the descent and level flight condition at the same  $V_t$ .

If the aircraft is stabilized on a Mach number while making a small descent at a measured rate, the value of  $(D - F_n)$  can be computed. This value will be the difference between the thrust required for level flight at the same Mach number and the thrust being delivered. By the use of net thrust computations or engine manufacturer's curves this additional thrust required can be converted to a required rpm increase giving equivalent level flight performance. This method is used at whatever speeds the pilot finds he cannot stabilize the airplane in level flight. But rates of descent of more than 200 ft/min will not produce satisfactory results. Once a level flight speed and rpm are obtained by this method, all other level flight corrections are applied in the normal menner.

In turbojet powered aircraft compressibility phenomena, Mach number effects, are significant at all altitudes. For this reason, no generalized speedpower curve, such as the PIW-VIW curve, is applicable. Separate level flight data results must be presented as a function of rpm and speed or Mach number for each value of  $W/P_a$  flown, as in equation 4.803. The data is actually presented for each altitude, Pa, at a constant weight for that altitude. One level flight performance presentation is standard day rpm vs Vc for constant pressure altitudes, H<sub>c</sub>. A typical plot of this type is shown in Figure 4.81. Standard day rpm is defined as:

rpm std = 
$$\frac{N}{\sqrt{T_{as}}}$$
 x  $\sqrt{T_{as}}$ 

This same plot is used also to show calibrated speed reduction (drag increases) resulting from configuration changes as in Figure 4.82.

AFTR 6273



Figure 4.81 Turbojet Rpm-V<sub>c</sub> Presentation



Figure 4.82 Effect of Configuration Change on Turbojet Speed and Rym

AFTR 6273

ł

Using equation 4.804 it is possible to correct and plot all the level flight data for standard day sea level condition. This is done by plotting  $N/\sqrt{\theta_a}$  vs Mach number for constant values of  $W/\delta_a$  obtained at various altitudes. This plot, Figure 4.83, shows immediately the sea level weight limitations at maximum rpm at various Mach numbers. By interpolation the plot can be used to find speed power conditions for any weight and altitude. Figure 4.83 is presented in the final performance report to show the obvious Mach number-compressibility effects on engine performance requirements. It also must be used to obtain weight corrected data for plotting as shown in Figures 4.81 and 4.82. When external drag items such as rockets or bombs are added to the aircraft it is often necessary to present both the effects on indicated speed and the compressibility effects. Speed effects are shown on a graph similar to that of Figure 4.82. Mach number-compressibility effects on the power required for external drag items can be shown as illustrated in Figure 4.84. יי יי



Figure 4.83 Typical Sea Level Turbojet Performance Data

The type of plot used to sum up the level flight performance for a turbojet aircraft is illustrated in Figure 4.85. This graph presents the standard day true speed vs altitude data for various actual engine rpm's. Also shown are a reference max. Mach number and a recommended max. range cruise condition. This speed vs altitude plot is generally presented for each major aircraft configuration, such as with wing tip tanks installed and without. The speed or altitude curves are most easily obtained by cross plotting weight corrected test data from Figures 4.81 or 4.85.

AFTR 6273

1 :



Figure 4.84 Typical Configuration Effect on Sea Level Turbojet Performance Data

AFTE 6273



AFTR 6273



## DATA REDUCTION OUTLINE (4.81)

For Level Flight Jet Aircraft Power Calibration

Test Point Number (1) (2) H. <u>A</u>Hic Indicated pressure altitude (3) Altimeter instrument correction 4)  $\Delta H_{pc}$ Altimeter position error correction 5) 6) Ho S True pressure altitude, (2) + (3) + (4)=  $(P_{\rm sl}/P_{\rm SL})$ From (5) and standard altitude tables V. 7) 8) Indicated air speed ∆v1c Air-speed instrument correction (9) ∆ V<sub>p¢</sub> Air-speed position error correction (10) v<sub>o</sub> M Calibrated air speed (7) + (8) + (9)(11)Mach mumber from (10) and (5) and **CHART** 8.5 V<sub>te</sub> (12)Standard-day, true speed, from (10) and (5) and CHART 8.5 t<sub>1</sub> (13) Indicated air temperature ∆**\$**10 (14) Temperature instrument correction (15) t<sub>ic</sub> Instrument corrected indicated air temperature (16)Tat True test air temperature - "Kelvin, from (15) and (11) and CHART 8.2  $\sqrt{\theta_{as}} = \sqrt{T_{as}/T_{SL}}$ (17)From (5) and standard altitude tables V(16)/288°X (18)  $V \theta_{at} = \sqrt{T_{at}/T_{SL}}$ (19) Gallons fuel remaining - from fuel counter (20) Gallons fuel start Gallons fuel used, (20) - (19) (21) (22) Fuel weight used, (21) x lbs/gallon (23) Starting gross weight (24) ۷t Test weight, (23) - (22) (25) (26)  $W_t/\delta_a$ Test weight-pressure parameter (24) + (6)W\_/8 Standard weight-pressure parameter, W<sub>R</sub> + (6)  $\Delta(W/\delta_n)$ (27) Weight pressure parameter correction (26) ~ (25) (28) Ni Indicated rpm (29) Rym instrument correction (30) (31) Test rpm, average of all engines π  $\frac{\pi}{\Delta} (\pi \sqrt{\theta_{at}})_{v}$ Test rpm parameter (30) + (18)(32) Rpm parameter increment for  $\Delta(W/\delta_A)$ from weight correction graphs similar to Figures 4.91 or 4.94  $(\pi_v/\sqrt{\theta_{at}})$ (33) Weight pressure ratio corrected rpm  $R_{\rm V} \sqrt{\frac{T_{\rm as}}{T_{\rm at}}}$ parameter, (31) + (32)(34) Standard-day rpm at standard weight, (**33**) x (17) Plot (33) vs (11) for constant  $W/\delta_{a3}$ Plot (34) vs (10) for constant W and Pa (35) (36) (37) Plot (34) vs (12) and cross plot to get altitude vs  $V_{ts}$  for constant standard-day rpm's or percent rpm's

AFTR 6273

#### SECTION 4.9

## <u>Weight Change Corrections for Speed</u> <u>Power Data - Turbojet Aircraft</u>

Equation 4.210 presents the analytical expression for drag force in terms of W, q, and  $C_{\rm DP}$ . If  $C_{\rm DP}$  is constant, the equation reverts to a form in which increments of D for increments of W may be calculated. This is not the case for high-speed aircraft under the effects of compressibility. Since equation 4.210 assumes incompressible air flow conditions, it cannot be used as a means of drag-weight-speed correction at high Mach numbers. Essentially,  $C_{\rm DP}$  is an unpredictable function of  $W/P_{\rm A}$  and Mach number under these flight conditions, which means weight corrections to speed or drag must be accomplished empirically. The weight correction must be made on the basis of test data obtained from the given airplane and engine combination. Because this must be done in the high speed range the same procedure is used in the low speed range.

The tasis for the two weight-correction methods described here is the fact that the parameter  $W/S_a$  can be changed either by changing W or  $P_a$ . If a series of test runs are made at different altitudes and at the same Mach number at each altitude, a range of  $W/\overline{\Theta_a}$  and  $W/S_a$  would be covered. By making a plot of  $W/\overline{\Theta_a}$  vs  $W/S_a$ , as in Figure 4.91, with each line representing a Mach number, the change in  $W/\overline{\Theta_a}$  with respect to change in  $W/S_a$  can be found. Or, more directly, the correct  $W/\overline{\Theta_a}$  for any  $W/S_a$  and M is immediately apparent.

If  $W/\delta_B$  values are  $\pm 1\%$  of a given value, weight corrections are not required. This can be accomplished by good stabilization resulting from closely estimated speeds and rpm's required for a given  $W/\delta_B$ . A pitot connected rate of climb indicator can be used to show small amounts of acceleration in level flight and reduce time required to stabilize.





AFTR 6273



Figure 4.92 Corrected Turbojet Performance Data

In actual flight testing establishing stabilized flight on a specific Mach number is difficult, so level flight test data are plotted in the form of  $M/\frac{1}{2}$  vs M. If the test points are flown at successively lower Mach numbers at a constant pressure altitude, the  $W/\delta_a$  value diminishes for each point. This effect is shown in Figure 4.92.

### INCREMENT RATIO METHOD OF WEIGHT CORRECTION

To make necessary weight corrections an approximate method, using plots of  $\Delta(N/\sqrt{6_a})/\Delta(W/\delta_a)$ , has been devised. Normal power calibrations are flown by setting rpm and allowing speed to stabilize. A plot, Figure 4.93, is then made of  $N/\sqrt{6_a}$  vs Mach number at each altitude, disregarding the fact that the  $W/\delta_a$  for each point is different. Data at each altitude are accumulated in the same manner by starting the power calibration at high Mach numbers. Mext, at each altitude the values of the  $W/\delta_a$ 's of each point are averaged, and the entire calibration is assumed to have been flown at the average  $W/\delta_a$ . Lines of constant Mach number are drawn on the graph as in Figure 4.93. They will intersect the power calibrations at various  $N/\sqrt{6_a}$  values.

From these N/Ve values a plot of N/Ve vs V/ $\delta_a$  at constant Mach numbers can be made and the slopes,  $d(N \sqrt{\theta_a})/d(V/\delta_a)$ , determined, or  $(\Delta N/\sqrt{\theta_a})/\Delta V/\delta$ ) can be determined directly by taking increments between power calibrations at two altitudes.

Increment ratio = 
$$\frac{(\pi \sqrt{\theta_a})_2 - (\pi \sqrt{\theta_a})_1}{(\sqrt{\delta_a})_2 - (\sqrt{\delta_a})_1}$$
(4.901)

AFTR 6273



Since the assumption has already been made that each altitude calibration was flown at an average  $W/d_{\odot}$  or weight, accuracy will not be impaired by assuming that the  $W/d_{\odot}$  midway between two  $W/d_{\odot}$ 's represents the point where the increment ratio should be applied. The increment ratio is plotted against this average  $W/d_{\odot}$  altitude as in Figure 4.94.

From Figure 4.94 all test points can be corrected to a desired value of  $W/g_{a}$  which will correspond to some altitude when a weight is specified. With reciprocating engine aircraft weight correction was unde at each altitude so that the concept was established that a weight correction brought data to the flown altitude at a standard weight; however, when the  $W/g_{a}$ correction is unde both weight and altitude can be changed. By selecting a standard weight, but not requiring the standard pressure altitude to be the flown altitude, a  $(W/g_{a})_{std}$  can be found that will minimize  $\Delta W/g_{a}$  and consequently increase accuracy. For example: a series of points is flown at 20,000 ft. with weight varying from 14,000 to 12,000 lbs. giving an average  $W/g_{a}$  of 28,000. The points are to be corrected to a weight of 14,000 lbs. If an altitude of 18,100 ft. is selected  $(W/g_{a})_{std}$  will equal the average  $W/g_{a}$ . Unless a special reason was required for the 20,000 ft. calibration, the 18,100 ft. calibration gives information of equal value.

AFTR 6275





### CROSS PLOTTING AND IN-FLIGHT WEIGHT CORRECTION METRODS

Another weight correction method for jet aircraft is designed to obtain a very accurate plot of  $N/\sqrt{\theta_n}$  vs  $V/\delta_n$  for constant Mach number parameters as illustrated in Figure 4.91. This type weight correction employs four basic steps:

- Test points are flown consecutively from high Mach number to low at a constant pressure altitude. Points should be flown at small and approximately equal time intervals.
- (2) Plot N/  $\sqrt{\Theta_a}$  ve M for each altitude flown.
- (3) Plot  $W/S_{A}$  ve M for each altitude flown.
- (4) Cross plot, from (2) and (3) above, N/Ve vs W/Se for constant Mach number parameters as illustrated in Figure 4.91. This final cross plot can be used to obtain N/Ve for any N, W, and H<sub>c</sub> desired. In addition N/Ve values for max. rpm and various standard altitudes can be included to define the max. speed points.

To obviate the need for a weight correction the level flight tests for both conventional and jet powered aircraft can be flown at a constant value of  $W/\delta_a$ . The value selected for this  $W/\delta_a$  depends on the take-off gross weight, fuel used in climb, and rate of fuel used during level flight at a particular altitude.

AFTR 6273

If these factors are carefully analyzed, it is possible to make flight guides enabling the test pilot to record data at desired constant values of  $W/Q_a$ . A typical  $W/Q_a$  guide is shown in Figure 4.95.



Guide Card for Flying at Constant W/Sa

Using this guide the pilot needs only to keep a slight lead on his fuel flow counter at a particular altitude. When the fuel remaining counter value for the aircraft agrees with the fuel remaining value required by the guide card the pilot records the necessary data. The efficiency of this method for holding a constant  $W/P_g$  depends on the skill of the pilot and may require some training on his part to prevent long lapses between test points. A radio technique can also be applied this method. The ground end of the radio link doing the necessary calculations then informing the pilot of the fuel counter readings and altitudes at which to record data.

AFTR 6273

## STOTION 4.10

### Fuel Consumption Endurance and Range - Turbo, let Airchalt

Data relative to fuel consumption for jet aircraft are obtained in flight whenever possible, rather than by use of the engine manufacturer's data. Flight fuel flow data is most accurately obtained by use of timed frel total izer readings or directly by use of rate of flow meters. In either case volume flow must be converted to weight flow. Generally, jet engine fuel is of two types:

- AN-F-38 (JP-1) = 6.7 lbs/gal standard weight for sea level standard conditions
- AN-F-58 (JP-3) = 6.42 lbs/gal standard weight for sea level standard conditions

If more accurate measurements are desired, where large quantities of fuel are involved at very low temperatures, the specific gravity should be determined before the flight and be used with a temperature correction factor to approximate the in-flight specific gravity. This procedure is only necessary where long-time high altitude flights are involved and the test gross weight is appreciably affected. In most test work use of the before flight specific gravity is sufficient.

$$W_{f} = galu/hr = 8.334 \times Sp g$$
 (4.701)

Apela:

W<sub>f</sub> = 1bs/hr Sp g = Specific gravity

In report presentation of fuel consumption or range data the test results are corrected to average sea level standard pounds per gallon values for the fuel being used.

For turbojet aircraft fuel consumption is handled in the form of the fuel flow parameter developed in Chapter Three.



 $W_{f}$  = fuel flow lbs/hr or gals/hr  $P_{t2}$  = engine inlet, total pressure  $T_{t2}$  = engine inlet total temperature  $\Theta_{2}$  =  $P_{t2}/S.L.$  std pressure  $\Theta_{2}$  =  $T_{t2}/S.L.$  std temperature

AFTR 6273

As shown in Chapter Three, the fuel flow parameter is a function of N/ $\sqrt{\theta_{t2}}$ and engine inlet <sup>M</sup>ach number. On the engine manufacturer's "expected performance curves" it can be seen that through most of the in-flight N/ $\sqrt{\theta_{t2}}$ range the fuel flow parameter is a single curve at all Mach numbers. When actual flight data is plotted, N/ $\sqrt{\theta_{t2}}$  vs  $W_f/\delta_{t2}$   $\sqrt{\theta_{t2}}$ , the results form a family of curves which for all practical purposes is a single curve. Figure 4.10.1 illustrates this plot. It will be noted that at the lower corrected rpm values for each altitude the curves tend to diverge. This is a result of both decreased Mach number at the higher altitudes for the same corrected rpm values and decreased combustion efficiency at the higher altitudes.



Figure 4.10.1 Turbojet Fuel Flow Presentation

It should be noted that ambient conditions may be used in the fuel flow parameter since the affects of Mach number are negligible unless large  $W/\delta_a$ variations at low rps are possible. A plot of  $M/\sqrt{\theta_a}$  vs  $W_f/\delta_a \sqrt{\theta_a}$  appears nearly identical to Figure 4.10.1 and is as accurate for determining fuel consumption for most rpm, atmospheric conditions, and Mach numbers. Using engine inlet conditions the fuel flow parameter is applicable in both climb and level flight. With ambient conditions in the parameter it is only theoretically applicable to level flight data because of differences in the inlet pressure term resulting from differences in Mach number for the climb and level flight condition. Actually the ambient fuel flow parameter is usually applicable to climb data as the error from Mach number effects in the climb rpm range is negligible.

ATTR 6273

#### **ENDURANCE**

It can be seen from Figure 4.10.1 that maximum specific endurance,  $1/W_{f}$ , for jet aircraft will be at the lowest rpm that will maintain level flight above the stall condition. It is also seen that specific endurance increases with altitude. Endurance is best evaluated with the specific endurance parameter.

The engine fuel flow function is,

$$\frac{V_{f}}{P_{a} YT_{a}} = f\left(\frac{B}{VT_{a}}, \frac{V_{t}}{\sqrt{T_{a}}}\right)$$
(4.10.01)

For the aircraft the rpm function is,

$$\frac{\mathbf{H}}{\sqrt{\mathbf{T}_{\mathbf{a}}}} = \mathbf{f} \left( \frac{\mathbf{V}_{\mathbf{t}}}{\sqrt{\mathbf{T}_{\mathbf{a}}}} \cdot \frac{\mathbf{V}}{\mathbf{P}_{\mathbf{a}}} \right)$$
(4.10.02)

Functionally combining the above two equations in terms of B,  $\Theta$  and H,

$$\frac{\mathbf{s}_{a}}{\mathbf{v}_{f}} = \mathbf{f} \left( \mathbf{N}, \frac{\mathbf{V}}{\mathbf{s}_{a}} \right)$$
(4.10.03)

A plot of equation 4.10.03, as in Figure 4.10.2 should be made throughout the complete speed range of the aircraft, so that max. endurance and endurance at any speed can be computed.



Turbo-Jet Endurance and Max. Endurance Plote

AFTR 6273

But for this most should be obtained at constant  $W/S_{g}$ 's, but negligible errors are introduced if  $W/S_{g}$  changes are small during power calibrations at one altitude.

Rest endurance is obtained by maintaining a constant  $W/S_{a}$ . This necessitates a elight rate of climb (20-50 ft/min) to decrease  $S_{a}$  as W decreases. By plusting as in Figure 4.10.2 a  $W/S_{a}$  for max. endurance can be found. Actual en-

$$F = \left(\frac{\delta_{a}}{w_{f}}\right) \int_{w_{1}}^{w_{2}} \frac{dw}{\delta_{a}}$$
(4.10.04)

where:  $\frac{W_1}{B_{a_1}} = \frac{W_2}{B_{a_2}} = \frac{dW}{dB_{a_1}} = constant, dW = fuel weight differential (-)$ 

Substituting  $dS_a$  for dW, changing limits, integrating and simplifying

$$F = 10.512 \left(\frac{\vartheta_a}{v_f}\right) \left(\frac{\vartheta}{\vartheta_a}\right) \frac{1}{\sqrt{\vartheta_{a_1}}} \left[ \left(\frac{v_1}{v_2}\right)^{.095} - 1 \right]$$
(4.10.05A)

And

$$\mathbf{E} = 1.15 \left(\frac{\mathbf{s}_{\mathbf{a}}\sqrt{\mathbf{s}_{\mathbf{a}}}}{\mathbf{v}_{\mathbf{f}}}\right) \frac{\mathbf{v}}{\mathbf{s}_{\mathbf{a}}} = \ln\left(\frac{\mathbf{v}_{1}}{\mathbf{v}_{2}}\right)$$
(4.10.05B)

Equation 4.10.05A assumes standard temperature lapse rate and may be used below the isothermal altitude. 4.10.05B assumes constant air temperature (-55° C) and is used above the isothermal altitude.

Values of  $\ln(W_1/W_2)$  may be found in GLART 4.71 at the end of this chapter.

RANCE

The actual range of turbojet aircraft is best evaluated by using the specific range parameter derived from equation 4.10.03. By the rules of dimentional analysis this equation can be put in the form.

 $\frac{\nabla_{t} \mathbf{s}_{a}}{\nabla_{f}} = f \left( \mathbf{H}, \frac{\mathbf{V}}{\mathbf{s}_{a}} \right)$ (4.10.06)

where:

$$\frac{V_t B_a}{W_f} = SRg B_a$$
, specific range parameter, mautical miles/lb

Data for plotting this equation should be obtained at constant values of  $W/S_{\rm g}$  in level flight. However, if data are obtained at a constant altitude and small percent changes in weight are involved, the error introduced will be negligible. The data for equation 4.10.06 are plotted as in Figure 4.10.3. This figure may be cross plotted to present the data for constant Mach number parameters and to determine optimum  $W/S_{\rm g}$  for maximum range.

AFTR 6273



Figure 4.10.4 Turbojet Specific Range Parameter and Maximum Range Plots, Constant Mach No. Values

AFTR 6273

4-37

/.

From the two specific range parameter plots it is possible to determine the effects of any of the variables on the actual range. Also, if very high values of  $W/\delta_A$  can be obtained, it is possible to determine from Figure 4.10.4 a value of  $W/\delta_A$  and Mach number that gives max range independent of altitude.

It is apparent, from these figures and equation 4.10.06, that the best cruise condition for a given Mach number is at constant  $W/\delta_a$  value. The higher the altitude the greater is the range until the  $W/\delta_a$  for max range is reached. The cruise at constant  $W/\delta_a$  requires a slight continuous climb (about 20-50 ft/ min) to decrease atmospheric pressure as the weight decreases. Actual range at constant  $W/\delta_a$  may be determined by integration.

(4.10.07)

Range = 
$$\left(\frac{v_t \delta_a}{w_f}\right) \int_{w_1}^{w_2} \frac{dW}{\delta_a}$$

where:

 $\frac{W_1}{\delta_1} = \frac{W_2}{\delta_2} = \frac{dW}{d\delta_a} = \text{constant}$ dW = fuel weight differential (negative)

Substituting  $d\delta_{a}$  for dW, changing limits and integrating

$$Rg = \left(\frac{V_{t} \delta_{a}}{W_{f}}\right) \frac{W}{\delta_{a}} = \ln \left(\frac{W_{1}}{W_{2}}\right)$$
(4.10.08)

Values of ln  $(W_1/W_2)$  may be found in CHART 4.71 at the end of this chapter.

To determine the range at a constant altitude and Mach number, the plot in Figure 4.10.4 may be used and the integration accomplished in steps for intervals where the slope of the Mach number parameters is nearly constant. In this case,

 $\Delta Rg = (\frac{SRg \delta_a}{2} + (SRg \delta_a)_2) \qquad (W_1 - W_2) \qquad (4.10.09)$ 

## PERFORMANCE REPORT PRESENTATION

Ordinarily, the specific range and endurance parameters, equations 4.10.03 and 4.10.06 are not included in the aircraft performance report unless actual range and endurance data are to be included. In most cases only the specific range is presented as a function of calibrated or standard day true speed at specified altitudes and weights.

Since Mach number can be defined as a function of pressure altitude and calibrated (indicated) speed, equation 4.10.06 may be expressed at a constant altitude as,

AFTR 6273

V <sub>t</sub> Wr	= $f(v_c, W)$	(4.10.10)
V <sub>t</sub>	= f (V <sub>ta</sub> , W)	(4.10.11)

where:

Wr

 $V_{ta}$  = standard day true speed for  $V_c$  and given altitude

These equations are plotted in the final report as in Figures 4.10.5 and 4.10.6, for each major configuration. For reducing data, it should be noted that at a given Mach number, weight, and altitude the test and standard day specific range are equal. The plots of  $V_t/W_f$  vs  $V_c$  are used as an aid to pilot technique since it is simpler to fly at a constant  $\bar{V}_{c}$  than at a constant  $V_{t}$ . It should be noted that in jet powered aircraft the specific range increases with altitude; however, if high enough altitudes or heavy enough weights are flown, a value of  $W/d_{e}$  will be found that gives a max specific range.

The effects of weight variations on specific range are of much less magnitude for turbojet aircraft than for conventional aircraft of equal size. This is because the best range conditions are at about 0.5 Mach number or above, and the induced drag in this speed range is a very small percent of the total drag. Nevertheless, weight variation can have a considerable percentage effect on turbojet range because of their inherent short range characteristics for a given size. Typical weight effects on range can be seen graphically in Figure 4.10.3 where the specific range parameter is plotted vs M for constant values of  $W/\delta_a$ . In this plot the sea level standard weight effects can be seen directly, and the effects of weight variation on specific ranges at any altitude can be readily computed.

Actual range can be computed with data from plots such as Figures 4.10.5 and 4.10.6 by assuming the Mach number be held constant and the weight pressure ratio to be held constant by a slow climb. The range in this case is,

The necessary rate of climb to hold a constant weight-pressure ratio is,

$$\frac{dH_0}{dt} = \frac{1.6 W_f T_a}{W_1}$$
(4.10.13)

where:

$$\frac{1H_O}{1}$$
 = altimeter rate of climb, ft/min  
it  
 $W_f$  = fuel flow, lbs/hr  
 $T_a$  = standard temperature (°K) for pressure altitude)  
 $W_1$  = weight at start of cruise

AFTR 6273



Figure 4.10.5 Usual Report Presentation of Turbojet Specific Range Data





AFTR 6273

22.7
## DATA REDUCTION OUTLINE (4.10.1)

For Fuel Flow Parameter and Specific Range Calibrations, Turbojet Aircraft

- NOTE: This data reduction is a sequel to the Level Flight Power Calibrations Data Reduction Outline 4.81.
- (38) W<sub>ft</sub>, gals/hr Test volumetric fuel flow, from flow meter or timed increments at stabilized power conditions
   (39) W<sub>ft</sub>, lbs/hr Test fuel weight-flow (38) x lb/gal
   (40) W<sub>ft</sub> Fuel flow parameter, (39) + (6) and (18)
- (41) Plot (40) vs (31) as in Figure 4.10.1
- (42)  $\left(\frac{W_{ft}}{\delta_a}\right)_w$ (42)  $\left(\frac{W_{ft}}{\delta_a}\right)_w$ (43)  $1/\delta_a\sqrt{\theta_{as}}$ (44) rom (5) and standard altitude tables (44) SEg Hautical air miles per pound, (12) x (43) + (42)

**NOTE:** If fuel parameter weight correction is not used  $SEg = [(12) \times (18)] + [(39) \times (17)]$ (45) Plot (44) ys (10) and (12)

AFTR 6273

## SECTION 4.11

# Flight Thrust Measurement Applications to Drag and Lift Coefficient and Aircraft Efficiency Determination

Whenever thrust measuring instrumentation is installed, actual thrust delivered may be computed by the methods of Chapter Three. In lieu of tail pipe instrumentation the engine manufacturer's expected performance curves may be used to approximate flight thrust. In this case values of  $N/\sqrt{e_{t2}}$  and  $P_{t2}/P_{a}$ , ram pressure ratio, must be calculated. Since the engine net thrust equals the aircraft drag, it may be substituted wherever drag is used in plotting, or it may simply be called drag. Equations 4.210 and 4.215 define the two basic drag plots. In one plot drag is shown as a function of  $V_t \sqrt{\sigma}$ . This plot is valid only for the weight and altitudes at which it was flown. Figure 4.11.1 shows a typical  $F_n$  vs  $V_t \sqrt{\sigma}$  plot.



Figure 4.11.1 Typical Drag or Net Thrust ve Equivalent Speed Plot

Two other forms of the drag plot are shown in Figure 4.11.2, in which the thrust parameter,  $F_{\rm II}/P_{\rm g}$  or  $F_{\rm II}/S_{\rm g}$ , is plotted vs the weight pressure parameter,  $W/P_{\rm g}$  or  $W/S_{\rm g}$ , and Mach number.

AFTER 6273



Figure 4.11.2 Two Forms of Plotting the Level Flight Drag Data

The Fn/Ba vs W/8a plot with constant Mach number parameters may be derived by cross plotting as was done in Section 4.9 to obtain  $N/\sqrt{\theta_a}$  vs W/s for constant Mach number parameters. This plot is valuable in determining net thrust required for any altitude, weight, and speed.

When net thrust has been determined, CD, CDP, and "e" are determined from values of  $C_{L_{i}}$  P<sub>a</sub>, M and the aircraft dimensions. The following equalities and constants are useful in these computations.



 $C_{\rm DP} = C_{\rm D} - C_{\rm D1}$ 

With the above equalities constant C<sub>L</sub> parameters may be con-structed as shown on Figure 4.112, from simple slide rule computations.

The intersections of  $C_L$  and  $W/S_a$  parameters may be used to compute  $C_D$  values and a plot made as shown in Figure 4.11.5. AFTR 6273 4-43

where:

 $\begin{array}{rcl} F_n &= 1bs \\ V_t &= knots \\ S &= ft^2 \\ P_a &= "Hg \end{array}$ 

The  $C_L^2$  vs  $C_D$  plot will be a straight line (Figure 4.11.3) as long as the incompressible theory is valid but will show breaks where the drag curves rise due to the effect of Mach number.





Airplane efficiency is easily calculated by using the  $C_{\rm IP}$  value defined by the point  $(C_{\rm I}^2 = 0)$  on the  $C_{\rm I}^2$  vs  $C_{\rm D}$  plot. This is accomplished by drawing a straight line through the incompressible points (low speed) on the  $C_{\rm L}^2$  vs  $C_{\rm D}$ plot. The intersection of this line at  $(C_{\rm L}^2 = 0)$  defines  $C_{\rm DP}$ . In the incompressible range  $D_{\rm IP}$  is constant but it increases rapidly as supersonic flow is developed over the wing; hence, a plot of  $C_{\rm IP}$  vs Mach number as in Figure 4.11.4, will indicate the Mach numbers at which compressibility effects will be found. At higher altitudes (larger  $W/g_{\rm R}$  values)  $C_{\rm IP}$  will show an increase from compressibility at lower Mach numbers than required at lower altitudes (smaller  $W/g_{\rm R}$ values). This illustrates the fact that  $C_{\rm IP}$  in the compressible range increases with  $C_{\rm L}$  or angle of attack at constant Mach numbers.

AFTR 6273









AFTR 6273

# DATA REDUCTION OUTLINE (4.11.1)

# For Determining Drag and Lift Coefficients and Aircraft Efficiency

Note:		reduction is a by, 4.81 and 4.10	sequel to the Level Flight and Range Data ).1
(46)	Fn	lbs	Not thrust
(47)	ve	mote	Equivalent speed, from CHART 8.3 (5) and (11)
(48)	Plot (46)	) vs (47) as in I	
(49) (50) (51) (52) (53) (54)	Fn/6a S M CD CL CL	lbg ft <sup>2</sup>	Net thrust parameter, $(46) + (6)$ Wing area, from aircraft specs Aspect ratio, from aircraft specs From (11) Total drag coefficient, 0.000675 x (49) + [[50) x (52]] Lift coefficient 0.000675 x (25) + [[50) x (52]] From (54)
(55) ( <b>5</b> 6)	Diat (55	) ve (53) as in 1	
(57)	C <sub>DP</sub>		Profile drag coefficient, from extrapo- lation to $C_1^2 = 0$ of incompressible portion of (56), Figure 4.11.3
(58)	CDI		Induced drag coefficient, (53) - (57)
(59)	<b>9</b>		Aircraft efficiency factor $(55) + [(58) \times (51) \times \pi]$
(60)	Plot (57	) vs (11) as in :	Figure 4.11.4

NOTE: "e" needs to be determined at only one point on the incompressible  $C_L^2$  vs  $C_D$  plot.

AFTR 6275

h-46

#### SECTION 4.12

# STANDARDIZATION OF JET LEVEL FLIGHT DAGA FOR THEVET AUCMENTATION CONDITIONS OR PARTICULAR POWER LEVER SETTINGS OF AUTOMATICALLY CONTROLLED ENGINES

Data standardization in the high thrust range of automatically controlled engines, variable area nonzie engines, or for thrust augmentation conditions is not always amenable to the use of elementary engine operation parameters such as  $N/\sqrt{\theta_a}$ . In these cases various jet nozzle positions or particular throttle settings ("normal", "military", "maximum") will all give different flight speeds at a constant  $N/\sqrt{\theta_a}$ . For these conditions a plot such as Figure 4.83 would be meaningless.

Test requirements will usually demand standard speed, altitude, and weight data at a particular throttle position (dotent), or rpm-jet temperature combination, or augmentation setting. Standardization under these conditions may be accomplished by use of  $F_n/S_n - W/S_n - Mach No.$  plots as shown in Figure 4.11.2. The engine manufacturer's performance data will provide a means of obtaining incremental thrust correction for ambient (or inlet) temperature variations at the given power settings. Although the manufacturer's actual engine thrust data may be inaccurate the incremental values obtained from the slopes of his data will provide sufficiently accurate thrust corrections. If these net thrust corrections for ambient temperature variations (or any other nonstandard engine condition) are available, the following steps will standardize the flight speeds and Mach No's, in conjunction with Figure 4.11.2. If possible, flight thrust data can be used to obtain this figure. The manufactureer's expected thrust data can be used for this plot without seriously impairing the accuracy of standard flight speed data obtained by this method.

a. If test  $W/S_a$  values are more than one (1) percent from average values, correct data to constant  $W/S_a$  values as before.

b. Select weight and altitude desired at avarage  $W/S_g$  value. (This should not be more than  $\pm$  5000 ft. of test altitudes).

c. Determine t<sub>a</sub> for selected altitude and

$$\Delta t = t_{a_t} - t_{a_s}$$

d. Calculate  $\Delta F_n/S_a$  for  $\Delta t_a$  of (c) above and add to test value of  $F_n/S_a$  for the given thrust setting.

e. On Figure 4.11.2, using corrected  $F_n/S_n$ , establish M for standard altitude on mean  $W/S_n$  curve or, using  $\Delta F_n/S_n$  values, move test point parallel to  $W/S_n$  curve to standard  $F_n/S_n$  and M.

The final standardised plot for a given weight will appear as in Figure 4.12.1.

**AFTR 6273** 









Figure 4,12,2

AFTR 6273

POWER CORRECTION FOR WEIGHT CHANGE

ELLAR HILLE



AFTR 6273

CHART 4.41



APTR 6273



POW R CORRECTION FOR WEIGHT CHANGE

CHART 4.41



AFTR 6273





POWER CORRECTION FOR WEIGHT CHANGE

CHART 4.41



CHART 4.41

----





•

AFTR 6273

CHART 4.41

.



CHART 4.71

4-55

# UHUSS WELGHT FAUTUR FOR RANUE AND BRDUHANCE COMPUTATIONS CHART 4.72

1997 - 1975,

.020 . Uni . 016 .014 **GROSS WITCHT FACTOR, 2/VF** 10 10 10 10 10 10 11 .008 .006 .00h 40 200 60 10 20 30 夘 60 1000 300 GROSS WEIGHT, W (15 X 10-3)

CHART 4.72

4-56

.

## CHAPTER FIVE

#### CLIDE AND DESCENT PERFORMANCE

# SECTION 5.1

# Rate of Climb Parameters - Derivation

An aircraft climbs because the power being developed is greater than the power that would be required to maintain level unaccelerated flight under the same atmospheric and velocity conditions. This conversion of excess power into rate of climb may be shown graphically and analytically. Consider the forces acting on the airoraft as shown in Figure 5.11.



Figure 5.11 Vector Presentation for Climbing Aircraft

 $L = W \cos \theta$   $F_{ma} = D + W \sin \theta$   $F_{ma} V = DV + WV \sin \theta$   $\sin \theta = \frac{dh}{dt} = \frac{F_{ma}V - DV}{V}$   $\frac{dh}{dt} = \frac{35,000}{V} \left(\frac{THP_{a} - THP_{r}}{V}\right)$ 

(5.101)

(5.102)

ATTR 6275

From this figure,

Minere :

L = lift force (lbs) W = aircraft gross weight (lbs)  $\Theta$  = climb angle F<sub>DE</sub> = net thrust available (lbs) D = F<sub>RT</sub> = drag force = level flight drag (lbs) = net thrust required V = aircraft velocity on climb path (ft/min) dh/dt = tape line rate of climb (ft/min) THP<sub>E</sub> =  $\frac{F_{RE}V}{33,000}$  = available thrust horsepower THP<sub>T</sub> =  $\frac{DV}{33,000}$  = required thrust horsepower to maintain THP<sub>T</sub> =  $\frac{DV}{33,000}$  = maccelerated level flight NOTE: Equations 5.101 and 5.102 are based on the assumption that for much path and the state of 
small values of  $\theta$ ,  $0^{\circ}$ - 20, the lift force is equal to the gross weight. The case of the large climb angle will be discussed later.

Figure 5.12 displays graphically the typical variation of  $(THP_a - THP_r)$  with climbing velocity.



VELOCITY

Figure 5,12 Typical Power Available and Power Required Curves

AFTER 6273

Certain aerodynamic considerations, assuming parabolic variation of  $C_D$  and  $C_L$ , are used in developing the rate of climb, induced power, and parasite power relationships.

(5.103)

$$D = C_{Dp}qS + C_{D1}qS$$

$$D = C_{Dp}qS + \left(\frac{W}{qS}\right)^2 \frac{qS}{\pi R_0}$$
where: L = W
$$D = C_{Dp}qS + \left(\frac{W \cos \theta}{qS}\right)^2 \frac{qS}{\pi R_0}$$

where:  $L = W \cos \theta$ 

Substituting 0.7  $P_{\rm R}H^2$  for q,

$$D = C_{Dpk1}P_{a}M^{2} + \frac{W^{2}}{P_{a}M^{2}k_{2}}$$
(5.104)
where: L = W

$$D = C_{Dp} k_1 P_a H^2 + \frac{W^2 \cos^2 \theta}{P_a H^2 k_2}$$
(5.105)

where:  $L = W \cos \theta$ 

Dividing by Pa, the ambient pressure, in equation 5.104,

$$\frac{D}{F_{a}} = C_{Dp} k_{1} k^{2} + \frac{k^{2}}{P_{a}^{2} k^{2} k_{2}}$$
(5.106)

MATO:

H = Flight Mach number  $C_{DP} = f(H \text{ and } W/P_a)$   $P_a = \text{atmospheric pressure (lbs/ft<sup>2</sup>)}$   $k_1 = 0.7S (ft<sup>2</sup>)$   $k_2 = 2.2b^{2e} (ft^2)$  b = wing span (ft) e = airplane efficiency factor $\mathcal{R} = b^2/S$ 

It has been shown that,

$$\frac{p_{\rm HP}}{100} = \frac{p_{\rm HP}}{33,000} = \frac{p_{\rm HP}}{33,000}$$

where  $F_{nr}$  = thrust required to overcome Drag

APTR 6273

œ

Dividing by  $P_a \sqrt{T_a}$ ,

$$\frac{\text{THP}_{r}}{P_{a}\sqrt{T_{a}}} = \frac{D}{33,000} \frac{V}{P_{a}\sqrt{T_{a}}} = \frac{F_{DP}}{33,000} \frac{V}{P_{a}\sqrt{T_{a}}}$$
(5.107)

$$\frac{THP_{a}}{P_{a}} = \frac{P_{a}}{33,000} \frac{V}{P_{a}} \sqrt{T_{a}}$$
(5.108)

As shown in Chapter One,

$$\frac{Y}{\sqrt{T_a}}$$
 " H x constant

Using this relationship, equations 5.107 and 5.108 may be written in the form,

$$\frac{\text{TEP}_{r}}{P_{a} \sqrt{T_{a}}} = \frac{T_{m}}{P_{a}} \frac{\text{M}}{P_{a}} \qquad (5.109)$$

$$\frac{\text{TEP}_{a}}{P_{a} \sqrt{T_{a}}} = \frac{T_{m}}{P_{a}} \frac{\text{M}}{P_{a}} \qquad (5.110)$$

Substituting equation 5.109 in 5.106,

$$\frac{\mathrm{THP}_{\mathrm{r}}}{\mathrm{P}_{\mathrm{a}} \sqrt{L_{\mathrm{a}}}} = c_{\mathrm{D}\mathrm{p}} \mathrm{k}_{1} \mathrm{H}^{3} + \frac{\mathrm{W}^{2} \mathrm{k}}{\mathrm{P}_{\mathrm{a}}^{2} \mathrm{H} \mathrm{k}_{2}}$$
(5.111)

Therefore,

$$\frac{\text{THP}_{\text{F}}}{P_{\text{R}}} = f\left(H, \frac{W}{P_{\text{R}}}\right) \text{ level unaccelerated flight} (5.112)$$

From engine performance analysis,

$$\frac{\text{THP}_{a}}{P_{a}} \sim \frac{\text{HHP}_{a}}{P_{a}} \left( \begin{array}{c} \eta_{p} \end{array} \right)$$
where:  $\eta_{p} = \text{propeller efficiency}$ 

$$\frac{\text{THP}_{a}}{P_{a}} = \frac{P_{\text{IM}}}{P_{a}} \quad \text{HH} = f\left( \begin{array}{c} H \\ \sqrt{T_{a}} \end{array} \right) \quad \text{turbojet} \quad (5.113)$$

$$\frac{\text{THP}_{a}}{P_{a}} \sqrt{T_{a}} = \frac{P_{a}}{P_{a}} \quad \text{HH} = f\left( \begin{array}{c} H \\ \sqrt{T_{a}} \end{array} \right) \quad \text{turbojet} \quad (5.113)$$

AFTE 6273

54

Equation 5.112 applies to both propeller driven aircraft and turbojet aircraft, while equation 5.113 applies to turbojet aircraft only.

From the functional power and drag equations the rate of climb functional parameters are derived.

From the rate of climb equation,

$$\frac{dh}{dt} = \frac{33,000}{W} \left(\frac{THP_{a} - THP_{r}}{W}\right)$$
(5.102)

This may be divided by  $P_{\underline{a}}$  and  $\sqrt{T_{\underline{a}}}$ , then,  $P_{\underline{a}}$ 

$$\frac{dh}{dt} = \frac{33,000}{\sqrt{T_a}} \frac{P_a}{V} \left( \frac{THP_a}{P_a \sqrt{T_a}} - \frac{THP_T}{P_a \sqrt{T_a}} \right)$$
(5.114)

From equations 5.112, 5.113 and 5.114,

$$\frac{dh}{dt} = f\left(\frac{THP_{a}}{P_{a}\sqrt{T_{a}}}, N, \frac{V}{P_{a}}\right) \text{ propeller driven}$$
(5.115)

$$\frac{dh}{dt} = f \left( \frac{\pi}{\sqrt{T_a}}, H, \frac{\Psi}{P_a} \right) \qquad \text{turbojet} \qquad (5.116)$$

$$\frac{\frac{dh}{dt}}{\sqrt{T_{a}}} = f\left(\frac{Y_{ma}}{P_{a}}, M, \frac{W}{P_{a}}\right) \qquad \text{turbojet} \qquad (5.117)$$

#### SECTION 5.2

#### Temperature Variation Correction to Bate of Climb Data

As test day and standard day flights are assumed to be at the same pressure altitude, it is only necessary to make atmospheric performance corrections for temperature differences. The corrected pressure altitude at which the aircraft flies is denoted by the symbol,  $H_0$ , and the tapeline altitude by the symbol, h. The relationship between tapeline and pressure altitude increments is given by

$$\frac{dh}{dH_0} = \frac{P_B}{A_t} = \frac{T_{at}}{T_{as}} \text{ for } (P_{as} = P_{at})$$
(5.201)

where:

 $P_{s}$  \* Density for which instrument is calibrated - standard density at the pressure altitude  $P_{t}$  = Density at test pressure altitude  $T_{as}$  = Standard temperature at the pressure altitude  $T_{at}$  = Test temperature at the pressure altitude

Then,

$$\frac{dh}{dt} = \left(\frac{dH_0}{dt}\right) \left(\frac{T_{at}}{T_{as}}\right)^* \quad \text{tapeline rate of climb at test true speed} \qquad (5.202)$$

In the parameters of equations containing Ta; Tat may be substituted.

Then,

$$\frac{dh/dt}{\sqrt{T_{as}}} = R/C = tapeline rate of climb at standard day true speed.$$

And, from equation 5.202,

$$\frac{dh}{dt} \sqrt{\frac{T_{as}}{T_{at}}} = \frac{dH_0}{dt} \sqrt{\frac{T_{at}}{T_{as}}} = R/C = \frac{tapeline rate of climb at standard}{dsy true speed}$$
(5.203)

In equations 5.115, 5.116 and 5.117 it is seen that, for a constant M and  $W/P_{a}$ , the rate of climb parameter varies with the power parameters,

$$\frac{\mathbf{THP}_{\mathbf{a}}}{\mathbf{P}_{\mathbf{a}}} \operatorname{or} \frac{\mathbf{H}}{\sqrt{\mathbf{T}_{\mathbf{a}}}} \operatorname{or} \frac{\mathbf{P}_{\mathbf{b}\mathbf{a}}}{\mathbf{P}_{\mathbf{a}}}$$

As climbs are flown at constant rpm in turbojets and at constant rpm-manifold pressure schedules in propeller driven aircraft, temperature variations from test conditions will cause changes in the value of the rate of climb parameter. This rate of climb increment for temperature effects on power is called  $\Delta R/C_1$ .

AFTE 6273

$$\Delta R/C_1 = \Delta \left(\frac{dh}{dt} \sqrt{T_{at}}\right) \sqrt{T_{as}}$$
(5.204)

From equation 5.101,

$$\Delta \left(\frac{d\mathbf{h}}{dt}\right) \sqrt{\mathbf{T}_{as}} = \Delta \left[\frac{101.3 \ \mathbf{v}_{tt}(\mathbf{F}_{na}-\mathbf{D})}{\mathbf{W}_{t} \ \sqrt{\mathbf{T}_{at}}}\right] \sqrt{\mathbf{T}_{as}} = \Delta \mathbf{F}_{na} \left(\frac{101.3 \ \mathbf{v}_{tt} \ \sqrt{\mathbf{T}_{as}}}{\mathbf{W}_{t} \ \sqrt{\mathbf{T}_{at}}}\right)$$
(5.205)

where:

$$V_{tt}$$
 = test aircraft climb true speed (knots)  
 $\Delta \overline{F}_{na} = \Delta (\overline{F}_{na} - D)$ , D assumed constant  
 $W_t$  = test weight

In terms of Mach number and  $(F_{na}/\delta_a)$ ,

$$\frac{\Delta R/C_1}{\Delta(F_{ne}/\delta_e)} = \frac{151.9 \text{ N P}_e \sqrt{T_{as}}}{W_t} \quad (\text{turbojet aircraft}) \quad (5.206)$$

02\*

$$\frac{\Delta \mathbf{R}/\mathbf{C}_1}{\mathbf{M} \ \Delta(\mathbf{F}_n/\delta_n)} = \frac{3946 \ \sqrt{\mathbf{T}_{n0}}}{\mathbf{W}_t/\delta_n}$$
(5.207)

where:  $\delta_a = P_a (\text{inches Hg})/29.92$ 

From equation 5.102,

$$\Delta R/C_1 = \frac{33,000}{W_t} \left( \frac{THP_{as} - THP_{at}}{V_t} \sqrt{\frac{T_{as}}{T_{at}}} \right) \stackrel{(reciprocating (5.208))}{(reciprocating engine aircraft)}$$

From reciprocating engine power theory the effects of temperature on power available are an effect on the carburetor temperature and the effect on manifold pressure.

From Chapter Two,

$$\operatorname{BHP}_{as} \cong \operatorname{BHP}_{at} \qquad \sqrt{\frac{r_{at}}{r_{as}}} \tag{2.201}$$

Substituting equation 2.201 in equation 5.208,

$$\Delta R/C_{1} = \frac{33,000 \text{ BHP}_{as}}{W_{t}} (1 - \frac{T_{as}}{T_{at}}) n_{p}$$
(5.209)

or, in terms of test brake horsepower

AFTE 6273

$$\Delta R/C_{1} = \frac{33,000 \text{ BHP}_{at}}{W_{t}} \left[ \sqrt{\frac{T_{at}}{T_{as}}} - \sqrt{\frac{T_{as}}{T_{at}}} \right] n_{p}$$
(5.210)

And above critical altitude equation 5.208 becomes

$$\Delta R/C_{1} = \frac{33,000 \text{ BHP}_{at}}{W_{t}} \left[ \sqrt{\frac{T_{at}}{T_{as}}} + \frac{MP_{s}}{MP_{t}} - 1 - \sqrt{\frac{T_{as}}{T_{at}}} \right] \eta_{p}$$
(5.211)

where:

$$\begin{array}{l} \mathbf{MP_g} &= \texttt{standard day manifold pressure} \\ \mathbf{MP_t} &= \texttt{test day manifold pressure} \\ \mathbf{MP_f} &= \texttt{f} \left( \frac{\mathbf{T_{ag}}}{\mathbf{T_{at}}} \quad \texttt{and} \quad \frac{\mathbf{MP_t}}{\mathbf{P_g}} \right) \\ \mathbf{MP_t} &= \texttt{propellor efficiency, average value, 0.83} \\ \end{array}$$

Equation 5.211 is plotted in CHART 5.21 for use in determining  $\Delta R/C_1$  for reciprocating engine aircraft, from values of  $\text{RHP}_{at}$ ,  $T_{ag}/T_{at}$ , and  $MP_t/F_g$ . The manifold pressure correction for wir only is applied in the chart without introducing any error of magnitude in cases of mixture blowers. Values of  $\text{RHP}_{at}$ , the power developed during the climb, may be determined from torquementer readings or from engine power charts.

Equations 5.206 and 5.207, plothed in CHARTS 5.22 and 5.23, are used to determine  $\Delta R/C_1$  for jet aircraft. Values of  $(F_n/\delta_n)$  corresponding to increments of  $R/\sqrt{\theta_n}$  are found for jet aircraft from a plot of  $F_n/\delta_n$  vs  $R/\sqrt{\theta_n}$  as described in Chapter Three and illustrated in Figure 5.21 below.



Figure 5.21 Typical Corrected Net Throst-Rpm Plot for Eate of Climb Corrections

AFTER 6273

# SECTION 5.3

#### Weight Variation Correction to Rate of Climb Data

An aircraft rate of climb at some given gross weight may be corrected mathematically to give the rate of climb at some other gross weight. This mathematical rate of climb correction for weight variation assumes that all atmospheric, velocity, and power conditions are the same for both gross weights. For this reason, weight corrections are applied to standard day climb data. This method further implies that the effects of weight changes on the compressibility drag are negligible. This latter assumption is not generally valid for flight Mach numbers above 0.6. From equation 5.101,

$$\mathbf{F}_{\mathbf{DA}} = \left(\frac{d\mathbf{h}}{d\mathbf{t}}\right) \left(\frac{\mathbf{W}}{\mathbf{V}}\right) + \mathbf{D}$$
(5.301)

In terms of the drag or thrust required, equation 5.104,

$$\mathbf{F}_{\mathbf{n}\mathbf{a}} = \left(\frac{d\mathbf{h}}{d\mathbf{t}}\right) \left(\frac{\mathbf{W}}{\mathbf{V}}\right) + \mathbf{C}_{\mathbf{D}\mathbf{p}} \mathbf{k}_{1} - \frac{\mathbf{p}_{\mathbf{a}}\mathbf{M}^{2}}{\mathbf{a}} + \frac{\mathbf{W}^{2}}{\mathbf{P}_{\mathbf{a}}\mathbf{M}^{2} \mathbf{k}_{2}}$$
(5.302)

Infferentiating, with W and dh/dt the only variables,

$$d \left(\frac{dh}{dt}\right) = -\left[\left(\frac{dh}{dt}\right) \frac{dW}{W} + \frac{2V}{P_{a}W}\frac{dW}{L_{2}}\right]$$
(5.303)

where:

$$\frac{dh}{dt} = R/C + \Delta R/C_1 = \left[ \left( \frac{dH_c}{dt} \sqrt{\frac{T_{at}}{T_{as}}} \right) + \Delta R/C_1 \right], ft/min$$

$$dW = (W_a - W_t) = \Delta W, lbs$$

$$V = flight velocity, ft/min$$

$$P_a = atmospheric pressure, "Eg$$

$$k_2 = 155.6 b^2 e (ft^2)$$

Then,

$$\frac{\Delta R/C_2}{\Delta W} = -\frac{(db/dt)}{W}$$
(5.304)

$$\frac{\Delta \mathbf{R}/C_3}{\Delta \mathbf{W}} = -\frac{50.65}{P_a} \frac{\sqrt{T_{ag}}}{P_a}$$
(5.305)

Values of  $(\Delta R/C_2)/\Delta W$  are easily computed from  $W_t$  and dh/dt. Values of  $(\Delta R/C_2)/\Delta W$  may be found from CHART 5.31 as a function of pressure altitude, Mach number and wing span. The total weight correction to rate of climb is,

$$\Delta R/C_2 + \Delta R/C_3 = -\left(\frac{\Delta R/C_2}{\Delta W} + \frac{\Delta R/C_3}{\Delta W}\right) \Delta W \qquad (5.306)$$

AFTR 6273

Where the angle of climb is not negligible,  $\theta > 20^{\circ}$ , an error in  $\Delta R/C_3$  of ten percent or larger will be introduced since the normal load factor departs significantly from lg. In that case the induced drag may be set equal to

$$\frac{(nW)^2}{\pi R e(qS)}$$

and the following equation derived:

$$\Delta R/C_{3} = \frac{25.36\sqrt{T_{as}}}{P_{as}b^{2}eM} \left[ \frac{n_{t}^{2}W_{t}^{2} + \frac{\delta_{as}}{\delta_{at}} - n_{s}^{2}W_{s}^{2}}{W_{s}} \right]$$
 5.307

AFTR 6273

**5-1**Ú

## SECTION 5.4

# Vertical Wind Gradient Correction to Rate of Climb

An aircraft climbing through a vertical wind-velocity gradient will experience a horizontal acceleration if the relative wind direction is negative and a horizontal deceleration if the relative wind is positive.

$$\mathbf{a}_{\mathbf{W}} = \frac{\mathbf{d} \ \mathbf{v}_{\mathbf{W}}}{\mathbf{d}\mathbf{t}} = \left(\frac{\mathbf{d} \ \mathbf{v}_{\mathbf{W}}}{\mathbf{d}\mathbf{h}}\right) \left(\frac{\mathbf{d}\mathbf{h}}{\mathbf{d}\mathbf{t}}\right) \mathbf{a}$$
(5.401)

where:

and the second 
 $a_w = horizontal acceleration$   $\frac{d V_w}{dh} = vertical wind-velocity gradient, (+) headwind, (-) tailwind$  $<math>\left(\frac{dh}{dt}\right)_a = accelerated rate of climb$ 

The mass of the aircraft and this acceleration force is converted into a rate of climb increment if the climb path speed is considered constant.

Assuming small climb angles, the wind gradient results in a changing climb path speed relative to the earth. This results in a kinetic energy increment for the aircraft which must be balanced by a potential energy increment if the climb path speed relative to the air is maintained constant.

$$\left(\frac{\Psi}{g}\right)^{\Psi} \left(\frac{d \Psi_{\Psi}}{dt}\right) + \Psi \left(\frac{dh}{dt}\right)_{\Psi} = 0$$

From this equation,

$$\frac{\Psi}{G} = \frac{d}{dh} \left( \frac{dh}{dt} \right)_{a} = -\frac{\Psi}{\Psi} \Delta \left( \frac{dh}{dt} \right)$$
$$\Delta \left( \frac{dh}{dt} \right) = -\frac{\Psi}{G} = \frac{d}{dh} \left( \frac{dh}{dt} \right)_{a}$$
(5.402)

Then the unaccelerated rate of climb is,

$$\frac{dh}{dt} = \left(\frac{dh}{dt}\right)_{a} - \frac{v}{g} \frac{d}{dt} \frac{v_{w}}{dh} \left(\frac{dh}{dt}\right)_{a}$$

$$\frac{dh}{dt} = \left(\frac{dh}{dt}\right)_{a} \left(1 - \frac{v}{g} \frac{d}{dt} \frac{v_{w}}{dh}\right) \qquad (5.405)$$

AFTER 6273

Values of the correction factor

$$\left(1 - .000089 \, \forall \, \frac{d \, \forall_{\forall}}{dh}\right)$$

where:

$$\frac{d \nabla_{\mathbf{v}}}{db} = \frac{bnote}{1000 \text{ ft}}$$

are plotted in CHART 5.41 at the end of this chapter for values of  $V_c$ , d  $V_y$ /dh, and pressure altitude.

AFTR 6273

## SECTION 5.5

## Climb Path Acceleration Correction to Rate of Climb Data

An aircraft climbing on a standard day at a fixed calibrated air speed must accelerate bebause of atmospheric density change with altitude. This acceleration of the aircraft mass absorbs some of the thrust available, and the rate of climb during these conditions is less than the unaccelerated rate of climb.

The force absorbed is,

Fa	=	M	₫₹	=	¥	<u>d V</u>	<u>/dh</u> )
		g	dt		8	đb	$\left(\frac{dh}{dt}\right)_{a}$

and, from equation (5.101),

$$\Delta \left(\frac{dh}{dt}\right) = \frac{\nabla F_{a}}{W} = \frac{\nabla}{g} \quad \frac{d\nabla}{dh} \quad \left(\frac{dh}{dt}\right)_{a}$$

Then,

or

$$\frac{dh}{dt} = \left(\frac{dh}{dt}\right)_{a} + \frac{\Psi}{g} \quad \frac{d\Psi}{dh} \quad \left(\frac{dh}{dt}\right)_{a}$$
$$\frac{dh}{dt} = \left(\frac{dh}{dt}\right)_{a} \quad \left(1 + \frac{\Psi}{g} \quad \frac{d\Psi}{dh}\right)$$

where:

 $F_{a} = \text{the force absorbed in accelerating}$   $\frac{dh}{dt} = \text{the unaccelerated climb}$   $\binom{dh}{dt}_{a} = \text{the accelerated climb}$   $\forall = \text{the true climb path Velocity}$   $a\binom{dh}{dt} = \text{rate of climb absorbed in accelerating}$ 

In terms of Mach number the correction factor in equation 5.501 becomes,

$$1 + \frac{Y}{g} \frac{dY}{dh} = \frac{M^2 Y R}{2} \frac{dT}{dh} + M F RT \left(\frac{dM}{dh}\right) + 1$$
(5.502)

(5.501)

For a constant true climb speed the above factor has a value of one. For a constant Mach number climb (deceleration) the factor has a value of  $(1 - 0.153 \text{ H}^2)$  up to the isothermal altitude above which the factor is equal to one. For an accolerating climb condition the factor takes the form,

AFTR 6273

$$1 + \frac{v}{g} \frac{dv}{dh} = \frac{(1 + .2M^2)^{3.5} - 1}{(1 + .2M^2)^{2.5}} = 0.133 M^2 + \frac{v_c}{g\sigma} \left(\frac{1 + .2M^2 SL}{1 + .2M^2}\right)^{-1.5} \frac{dv_c}{dh} + 1 \quad (5.503)$$

Equation 5.503 is plotted in CHARTS 5.51, 5.52, 5.53 and 5.54 in terms of  $V_c$ ,  $P_a$ , M,  $V_t$ ,  $dV_t/dh$ , and  $dV_c/dh$ . Chart 5.51 contains the first two right hand terms and is used for constant  $V_c$  climbs. Chart 5.52 contains the extreme right hand term and is an additive factor for a changing  $V_c$ . Chart 5.53 presents the first two right hand terms as a function of M. Chart 5.54 presents the function in terms of  $V_t$  and  $dV_t/dh$ .

During a climb to altitude (check climb) in jet type aircraft, the indicated climb speed is decreased with increasing altitude. This means that  $dV_c/dh$  in equation 5.503 is negative and, if large enough, may belance the other terms in the equation. In reciprocating engine aircraft the indicated speed is held constant, until the critical altitude for the engine is reached, and then is decreased. Thata 5.51 and 5.52 may be used to determine the bleed-off rate required for a constant climb true speed (zero acceleration factor). During sawtooth climbs in jet or conventional aircraft the calibrated speed is held constant over an increment of altitude. In these cases it may be desirable to determine the unaccelerated rate of climb that would be obtained under the same conditions at a constant true climbing speed as might be the case in a check climb. This may be accomplished using CHAET 5.51 or 5.53 for the climb  $V_c$  and  $H_c$  or  $M_s$  ( $dV_c/dh = 0$ ). The factor is used as in equation 5.501.

Since equation 5.503 is plotted for standard day conditions it should be applied to the tapeline rate of climb at standard day true speed. This tapeline rate of climb has been defined as,

$$R/C = \frac{dH_c}{dt} \sqrt{\frac{T_{at}}{T_{as}}}$$

For an accelerating climbing aircraft requiring a power or thrust correction for temperature variation, the usual power correction is not all available to correct the R/C defined above since changing R/C changes the acceleration thrust required at shown on the preceding page.

$$(\Delta \mathbf{E}/\mathbf{C}_1)_{\mathbf{a}} = \frac{\Delta \mathbf{E}/\mathbf{C}_1}{\left(1 + \frac{\mathbf{V}}{\mathbf{g}} \cdot \frac{\mathbf{d}\mathbf{V}}{\mathbf{d}\mathbf{h}}\right)}$$

where:

 $(\Delta R/C)_{a}$  = rate of climb increment for temperature-power correction to the accelerated rate of climb,  $\left(\frac{dR_{c}}{dt} \wedge \frac{T_{a_{t}}}{T_{A_{s}}}\right)_{t}$ 

Similarly, for an accelerating climbing aircraft requiring an induced drag weight correction to rate of climb at a constant air speed and thrust or power, an increased weight (induced drag) not only absorbs more of the excess thrust and reduces rate of climb, but it also results in less thrust being required for acceleration (reduced dH/dt) and some increase in rate of climb. In this case the total effect of climb acceleration on the induced drag weight correction is,

AFTR 6273

(5,505)

(5.506)

 $(\Delta R/C_3)_{a} = \frac{\Delta R/C_3}{\left(1 + \frac{Y}{s} \frac{dY}{db}\right)}$ 

For jet powered aircraft, the best rate of climb is at approximately a constant true speed. For reciprocating engine aircraft, the best rate of climb is usually found in an accelerating climb condition.

# APPLICATIONS OF THE ACCELERATION FACTOR TO SAW-TOOTH AND CHECK CLIMBS

The acceleration factor is applied to a saw-tooth rate-of-climb ( $V_c = const.$ ) to simulate check climb data under identical conditions of standard thrust, altitude, weight, and standard speed. During check climbs at best climbing speeds some acceleration ( $\frac{1}{2}$ ) may be evident; in these cases, the acceleration factor is applied only to the thrust and induced drag corrections. These two applications of the acceleration factor in climbing will be clarified by the equation below.

Basically, for the saw-tooth climb,

$$Rc_{\pm} = Rc_{\pm a} \perp A_{\pm}$$

where

Rct = test tapeline rate-of-climb at standard day speed, no acceleration along flight path.

$$Rc_{ta} = \frac{dH_c}{dt} \sqrt{\frac{T_{a_t}}{T_{a_s}}} = \text{test tapeline rate-of-climb at standard day}$$
speed, with acceleration along climb path

A, = standard atmosphere climb accelerations factor at standard

$$\text{iny speed} \quad (1 + \frac{Y}{g} \quad \frac{dY}{dh})$$

During a saw-tooth climb at constant  $V_{c, t}$  the aircraft is accelerating and

$$Rc_{s} = Rc_{ta} + \Delta Rc_{la} + \frac{\Delta W}{W} (Rc_{ta} + \Delta Rc_{la}) + \Delta Rc_{3a} A_{g} \qquad (5.507)$$

where:

Rog = tapeline rate-of-climb at standard day speed thrust, and weight, no acceleration along flight path.

$$\Delta Rc_{1a} = \frac{\Delta Rc_{1}}{A_{f}} = \text{thrust correction to rate-of-climb during flight path} \\ \text{acceleration of standard day speed.}$$

 $\Delta Rc_{3e} = \frac{\Delta Rc_{3}}{A_{f}} =$ induced drag rate-of-climb correction for weight variation during flight path acceleration at standard day speed.

AFTR 6273

With equation 5.506 and these definitions, equation 5.507 can be put in another form.

$$Re_{\bullet} = Re_{t} + \Delta Re_{1} + \frac{\Delta V}{V} (Re_{t} + \Delta Re_{1}) + \Delta Re_{3}$$
(5.508)

A plot of  $\text{Rc}_8$  vs  $\text{Vt}_8$  defines check climb, rate-of-climb values for standard conditions except that the speeds for maximum rate-of-climb may result in some acceleration with increasing altitude. This will be immediately apparent from the plot mentioned above. From this same plot, the acceleration factor data  $(\text{dV}_{ts}/\text{dh} \text{ and V}_{ts})$  for the check climb condition may be determined in the neighborhood of maximum rate-of-climb. The best climb points may then be readjusted to more exactly simulate actual check climb data. This is done by equation 5.509.

 $Bc_{a}$  check climb =  $Rc_{a}/A_{f}$  (5.509)

Standardizing on actual check climb requires a slightly different use of the acceleration factor, because any acceleration existing is common to both the test and standard day climbs. For this condition, equations 5.507 and 5.509 are combined.

check climb 
$$\operatorname{Rc}_{s} = \operatorname{Rc}_{ta} + \Delta \operatorname{Rc}_{la} + \frac{\Delta W}{W} (\operatorname{Rc}_{ta} + \Delta \operatorname{Rc}_{la}) + \Delta \operatorname{Rc}_{3a}$$
 (5.510)

For climbs at constant  $V_c$  the acceleration factor is determined from charts 5.51 or 5.53. For plotted, zero acceleration, saw-tooth data, and for check climb data, the acceleration factor is best determined from the change of standard true speed with altitude and chart 5.54. Chart 5.52 is intended to be used only to determine the necessary decay of  $V_c$  with altitude to establish a zero acceleration. It should be noted that chart 5.51 is plotted for standard altitude data. This means that a temperature lapse rate of  $-2^\circ$  C per thousand feet is assumed below the isothermal altitude. Generally, under test conditions up to altitudes of 30,000 ft., this standard lapse rate is realized approximately. The effect on Ar of temperature lapse rate can be seen on chart 5.53. The engineer should check all saw-tooth data for temperature inversions which greatly alter the standard lapse rate, but he should be especially alert on saw-tooth data obtained above 30,000 ft. The two curves on chart 5.53 may be used to interpolate for test temperature lapse rates between  $-2^\circ$  and zero to be applied to equations 5.506 and 5.507 in place of Ar as defined.

AFTE 6273

# SECTION 5.6

# Temperature Effects on Fuel Consumption and Weight During Climb

In making a check climb from take-off to ceiling altitude, it is necessary to give consideration to fuel consumption variation between test day and standard day. This temperature variation may result in an appreciable and increasing difference in test and standard day gross weights as the climb progresses necessitating a conversion to a rate of climb increment by the methods previously described. If the temperature difference is not more than  $10^{\circ}$  C, this fuelweight rate of climb correction is negligible. In this correction, the assumption is made that the fuel consumption rate is constant at a particular time and power setting irrespective of the rate of climb variation resulting from atmospheric temperature variation from standard. This assumption is also a valid approximation for climbs conducted at power settings varying as much as five percent.

One method of determining the difference in fuel weight used because of temperature effects is by plotting and correcting for temperature the fuel rate of consumption.

- a. Plot fuel used vs time.
- b. Plot fuel rate vs altitude
- c. Correct fuel rate vs altitude plot to standard day fuel rate at standard day power conditions by use of engine manufacturer's performance data.
- d. Integrate power corrected rate of climb data and plot tentative standard day time vs altitude.
- e. Plot corrected fuel rate vs corrected time to climb to altitude.
- f. Integrate plot "e" and obtain corrected fuel used vs time.
- g. Determine fuel weight increment at each altitude from plots "a" and "f".
- h. Make  $\Delta R/C_2$  and  $\Delta R/C_3$  corrections for weight difference to  $(R/C + \Delta R/C_1)$ . This is the final corrected rate of climb.

A.-Fuel Used vs Time (Test Temp.)

B.-Altitude vs Time to Climb (Test Temp.)

C.-Altitude vs Time to Climb (Standard Temp.)



# Figure 5.61 Method of Approximating a Rate of Climb Weight Correction for Variation in Fuel Used on Test and Standard Days

This same correction may be approximated by plotting the data as shown in Figure 5.61 and by making the assumption (based on actual climb data) that the fuel used vs time plot for the test temperature and power is very nearly identical to that for the standard temperature and power.

AFTR 6273

#### SECTION 5.7

# Determination of Best Rate of Climb and Best Climbing Speed

#### SAWTOOTH METHOD

One method of obtaining climb data is the savtooth climb. This name is derived from the barograph trace resulting from a series of short time climbs through the same pressure altitude. These climbs are made at different indicated climbing speeds from a point a little below the test altitude to a point a little above it. By plotting the resulting rates of climb for the various altitudes and air speeds, as shown in Figure 5.71, the best rate of climb and best climbing velocity will be apparent for all altitudes.



Typical Sawtooth Climb Plot

Sawtooth climbs are made at the same power settings and aircraft configurations that will be used in the continuous or check climbs. The sawtooth procedure is to climb through the altitude increment for a period of about one minute at each of the various speeds or Mach numbers selected to bracket the best rate of climb. In addition, the stalling point and high speed point for level unaccelerated flight at the climb altitude are plotted to complete the curves. Each sawtooth climb point is corrected to the gross weight that would be obtained by climbing at climb power setting from a standard gross weight take-off to the test altitude. Sawtooth climbs are corrected for temperature and weight variations by the methods indicated previously. For jet aircraft the data should be corrected to zero acceleration to simulate check climb data.

AFTR 6273

#### ACCELERATION METHOD FOR RATE OF CLINB AND OPTIMUM CLIMB PATH EVALUATION

For aircraft having very large rates of climb and fuel consumption and a limited fuel supply, the sawtooth method may be difficult to fly accurately and may be excessively time consuming for complete results. In these cases the level flight acceleration method may be applied to the determination of climb data. In addition, vertical wind gradients do not affect R/C data from level accelerations.

Consider the accelerating aircraft in level flight. The total energy of the aircraft is,

E	2	¥Ъ	+	<u>¥ ¥<sup>2</sup></u> 2g	(5.701)
---	---	----	---	------------------------------	---------

 $\frac{\mathbf{E}}{\mathbf{v}} = \mathbf{h} + \frac{\mathbf{v}^2}{2\mathbf{g}} \tag{5.702}$ 

where:

E	3	total energy (ft-lbs)	7	3	true velocity (ft/sec)
¥	3	gross weight (1bs)	E	3	acceleration of gravity
ħ	-	tapeline altitude (ft)	-		$32.174 (ft/sec^2)$

Differentiating equation 5.702, (including h to allow for altimeter position error)

$$\frac{dE}{V} = \frac{V dV}{S} + dh$$
 (5.703)

Dividing by dt.

 $\frac{dE}{dt} = \frac{V}{g} \frac{dV}{dt} + \frac{dh}{dt}$ (5.704)

The force available to accelerate the aircraft is,

$$(\mathbf{F}_{n_{\mathbf{R}}} - \mathbf{D}) = \mathbf{F}_{n_{\mathbf{R}}} - \mathbf{F}_{n_{\mathbf{T}}} = \mathbf{F}_{n_{\mathbf{R}}} = \mathbf{I}_{\mathbf{X}}$$
 Excess not thrust

Then,

 $\mathbf{F}_{\mathbf{n}_{\Theta}} = \frac{\Psi}{g} \frac{\mathrm{d}\Psi}{\mathrm{d}t} + \frac{\Psi}{\Psi} \frac{\mathrm{d}\mathbf{h}}{\mathrm{d}t}$ (5.705)

From equation 5.101,

$$\mathbf{R}/\mathbf{C} = \frac{\mathbf{V} \mathbf{r}_{\mathbf{n}}}{\mathbf{V}} \tag{5.706}$$

Combining equations 5.706, 5.704, and 5.705,

$$\mathbf{E/C} = \frac{\mathbf{dE}}{\mathbf{dt}} \frac{1}{\mathbf{V}} = \frac{\mathbf{V}}{\mathbf{g}} \frac{\mathbf{dV}}{\mathbf{dt}} + \frac{\mathbf{dh}}{\mathbf{dt}}$$
(5.707)

where:

R/C = ft/sec dt = seconds

ATTR 6273

By plotting the variable,  $\nabla^2$  in equation 5.702 against time, the rate of change of the total energy may be obtained. From equation 5.706 it is seen that the point of maximum slope of this curve is at the value of  $\nabla^2$  corresponding to  $\nabla$ for maximum rate of climb. Whas been assumed constant. In knots, equation 5.702 bacomes,

 $\frac{v^2}{2g} = 0.0443 v_t^2$  knots

From a plot of 0.0443  $\nabla^2$  knots we time in minutes and  $\nabla_{ic}$  and  $M_{ic}$  we time in minutes, the indicated speed and Mach number and best rate of climb may be obtained by inspection as shown in Figure 5.72. This acceleration method of obtaining climb data is more difficult to apply than may appear from the analysis, the primery source of error being the lag in the conventional type of air-speed indicating system. This may be overcome by calibrating the system for lag, by using a no-lag system which has its instrument very near the air-speed probe, or by use of electronic pressure measuring devices to determine indicated air speed. Another approach is to determine the time, distance, velocity and acceleration by radar tracking.

A plot of  $V^2$  vs time as in Figure 5.72 is fairly adequate for determination of best climb speeds, but tests have shown it to be usually unsatisfactory for determining accurate R/C values. A better analysis of R/C and best climb speeds can be made by computation of dE/dt from level acceleration values of dV, dt, and dh. Because dV  $\ddagger$  dt is required, the accuracy with which these increments are determined sets the accuracy and data scatter of the results. Minimum increment values should be ten (10) knots dV or 15 seconds dt. Equation 5.709 is suitable for machine computation of dE/dt. Values of M should also be computed.

$$\left(\frac{dE}{dt}\right)_{a} = \Psi_{2}\left[\frac{19362 \, \theta_{a_{a}} \, (M_{2}^{2} - M_{1}^{2}) + (E_{c_{2}} - E_{c_{1}})}{t_{2} - t_{1}}\right] \sqrt{\frac{T_{at}}{T_{as}}} \qquad (5.709)$$

Two level accelerations should be made at each altitude reciprocal headings. R/C is determined from dE/dt and desired weight;  $\Delta R/C_1$  and  $\Delta R/C_3$  corrections can also be applied. The dE/dt plot appears as in Figure 5.73.

Since dE/d: is a function of both h and  $\nabla^2/2g$  a maximizing function of dE/dt involves both and takes the form,

$$\frac{d\mathbf{E}}{dt} = \max. \text{ when } \frac{\partial(d\mathbf{E}/dt)}{\partial(\mathbf{V}^2/2g)} - \frac{\partial(d\mathbf{E}/dt)}{\partial\mathbf{h}} = \mathbf{G}$$
(5.710)

or when,

$$\frac{\Delta(dE/dt)}{\Delta V} = \frac{V}{g} \frac{\Delta(dE/dt)}{\Delta h}$$
(5.711)

This optimum dE/dt can be determined graphically and shown as on Figure 5.73. This climb schedule will then involve sooming or diving to obtain a given speed and altitude in minimum time as shown in Figure 5.74.

<u>BATE OF CHANGE OF TOTAL PRESSURE FOR DEFINING d(E/W)/dt.</u> Total pressure rate of change will approximately define d(E/W)/dt, and can be demonstrated by use of equations 2.21, 4.9, 4.10, and the equation of state. The resulting equation is applicable to a differential pressure gage.

$$\frac{dPt}{(\lambda_t F_t) = const.} = \frac{d(F/W)}{dt} - \frac{dT_a}{dt} \frac{(TRM^2)}{2} - \frac{dh}{dt} (2 + .2M^2)$$
(5.712)

 $T_a$  is the instrument temperature and is essentially constant for a given altitude. Use of rate of climb instrument for this purpose has been suggested by Lt. Col. J.L. Ridley, USAF. The lag constant used must provide measurable differential pressures and not result in excessive lag in dP<sub>t</sub> indications.

APTE 6273

(5.708)


Typical Plot of Level Flight Acceleration Data



Rate of Change of Total

Energy Plot



Figure 5.74

11

Minimum-Time Climb Schedule Plot

AFTR 6273

#### SECTION 5.8

#### Dimengionless Rate of Climb Plotting

In equations 5.115, 5.116 and 5.117 it may be seen that the parameters

$$\frac{\left(\frac{dh}{dt}\right)}{\sqrt{T}}$$
, and M

are dimensionless in terms of the physical system of dimensions. The remaining parameters,

$$\frac{THP}{P_a \sqrt{T_a}}$$
,  $\frac{N}{\sqrt{T_a}}$ ,  $\frac{F_n}{P_a}$ , and  $\frac{W}{P}$ 

may be made dimensionless by the insertion in each of a characteristic length raised to some power. As this characteristic length for a particular aircraft remains constant, these latter parameters may be considered functionally dimensionless. All the above parameters will result from a dimensional analysis of the variables effecting rate of climb.

For propeller driven aircraft, BHP from torque meter readings may be substituted for THP in equation 5.115. Then the parameters for dimensionless plotting are,



For turbojet powered aircraft equation 5.116 is applicable and the parameters are,

$$\left(\frac{dh}{dt}\right)$$
,  $\frac{\pi}{\sqrt{T_a}}$ ,  $M$ ,  $\frac{W}{P_a}$ , or  $\frac{T_n}{P_a}$  in place of  $\frac{W}{\sqrt{T_a}}$ 

Generally in this type of a plot the terms  $\Theta_{\mathbf{A}}$  and  $\delta_{\mathbf{A}}$  replace  $T_{\mathbf{A}}$  and  $P_{\mathbf{A}}$ 

where:

$$\theta_{a} = \frac{T_{a} \tilde{t}}{288}$$
$$\delta_{a} = \frac{P_{a} \tilde{t}}{29.92}$$

As there are four variables involved, it is necessary to hold one of them constant during a particular climb. It can be seen that Mach number is the only term that could reasonably be held constant during a climb. By making check climbs at constant Mach number and two or three different constant rpms it is possible to plot climb data without making weight or power corrections and to use the plotted data to determine standard day standard weight rate of climb even in the region where compressibility effects are large. Figure 5.81 illustrates a dimensionless rate of climb plot for a constant Mach number, data being obtained at three different power settings.



## Figure 5.81 Typical Dimensionless Climb Data at Constant Mach Humber and Various W/Sa Parameters

The methods used for Figure 5.81 can also be used for standardizing rate of climb data. For fighter type turbojet aircraft the best rate of climb will be at speeds that maintain approximately a constant Mach number for certain altitude ranges. Two climbing Mach numbers determined by sawtooth climbs may be enough to approximate all the climbing speeds and allow some interpolation. Such a plot would appear as in Figure 5.82 and may be used to correct climb data for all airoraft of the same model with same type of engine.

Figures 5.81 and 5.82 are easily plotted for constant values of W/d at the different constant rpms by first plotting N/ $\sqrt{\theta_{\rm A}}$  vs W/d and (R/C)/ $\frac{1}{\theta_{\rm A}}$  as shown in Figure 5.83.

AFTR 6273



Figure 5.82 Typical Dimensionless Climb Data Plot for Two Mach Numbers



Figure 5.83 Method of Correcting Climb Data to Constant  $W/\delta_a$ 

ATTR 6273

If the level flight data are included, two constant rpm climbs will provide enough points to determine the N/ $\sqrt{\theta_a} - (R/C)/\sqrt{\theta_a}$  curve for constant  $\sqrt{\theta_a}$ .

In the speed range for best climb the change in  $F_n/\delta_a$  with  $N/\sqrt{\theta_a}$  is approximately linear. This fact may be utilized to determine rate of climb fata for any condition from a few constant Mach number climbs. This is especially useful for very large aircraft having large weight variations which affect the best climb speed. Figure 5.84 illustrates a dimensionless rate of climb plot that would be derived from three constant Mach number climbs and level flight data. A linear  $F_n/\delta_a$  increase with  $N/\sqrt{\theta_a}$  is assumed for each of the constant  $V/\delta_a$  parameters. These climbs would be made in a light weight condition so as to obtain maximum altitudes and  $N/\sqrt{\theta_a}$  values.





AFTR 6273

The method of dimensionless plotting described above may be especially applicable to aircraft having very large rates of climb where sawtooths are impractical. Also, when afterburner and rocket power are used to boost the rate of climb, this method may prove valuable because of the difficulty of making power corrections in this type of installation.

These same types of non-dimensional rate-of-climb presentations may be obtained from level acceleration data made at constant  $W/S_{\rm R}$  values.

#### SECTION 5.9

#### General Climb Test Information

The absolute ceiling for an aircraft is defined as the altitude where the rate of climb is zero. The service ceiling is defined as the altitude at which the rate of climb is 100 feet per minute.

For propeller driven aircraft it will be necessary to make rate of climb corrections for variation in cooler flap position. This is accomplished by making several short climbs at a constant indicated speed and various cooler flap positions. Both the rate-of-climb increment and the engine temperature increment should be plotted vs flap position.

In the aircraft performance report it is desirable to show rate of climb and time to climb for two power settings. This is required for tactical use when a large number of aircraft are to take off during a period of time and meet in formation at a specified altitude. To accomplish this the first airoraft to take off must climb slower than those taking off at a later time.

It is usually required in performance reports that some data relative to rate of climb on a reduced number of engines be included. This is usually done by making a short climb of about 10,000 feet at a mean altitude between esa level and service ceiling. The top of this reduced -number-of engines climb should be close to the apparent service ceiling for that condition:

In heavy aircraft, bombers and transports, two olimbs at normal rated power or maximum jet engine rpm should be conducted for the extreme gross weight conditions, light and heavy, when data are plotted in the conventional manner. In this way the rate of climb variation with weight can be seen graphically and a linear extrapolation used to make weight corrections or to compare the analytical weight corrections. Typical climb performance presentations are shown in Figure 5.91 for propeller driven aircraft and in Figure 5.92 for jet powered aircraft.

ATTR 6273



Figure 5.92 Standard Climb Data Presentation, Jet Engine Aircraft

ATTR 6273

# DATA REDUCTION OUTLINE (5.91)

For Reciprocating Engine Aircraft Savtooth and Check Climbs

(1)	Test Point Number	•	
(2)	H <sub>1</sub>	ft	Pressure altitude
(3)	<b>Δ</b> <sup>-</sup> <b>H</b> <sub>1</sub>	ft	Altimeter instrument correction
(4)		ft	Altimeter position error correction
(5)	H <sup>C</sup>	ft	True pressure altitude, $(2) + (3) + (4)$
(6)	v <sub>1</sub>	knots	Indicated air speed
(7)	∆vic	knote	Air-speed instrument correction
(8)		mote	Air-speed position error correction
(9)	V PC	mote	Calibrated air speed, $(6) + (7) + (8)$
(10)	V <sub>ts</sub>	mota	Standard day true speed, from CHART 8.5 and (9) and (5)
(11)	M		Mach number, from CHART 8.5 and (9) and (5)
(12)	<b>t</b> 1	•C	Indicated air temperature
(13)	∆ <sup>1</sup> t <sub>ic</sub>	•C	Temperature instrument correction
(14)	tic	•C	Instrument corrected indicated air
• •	10		temperature, $(12) + (13)$
(15)	Tat	•K	Absolute test air temperature from CHART 8.2 (14) and (11)
(16)	$\sqrt{\theta_{at}}, \sqrt{T_{at}/T_{SL}}$		$\sqrt{(15)}$ / 288
(17)	$\sqrt{\theta_{as}}, \sqrt{T_{as}/T_{SL}}$		From standard altitude tables and (5)
(18)	Pa Pa	"Eg	Atmospheric pressure, from standard alti-
(==)	-8	-0	tude tables and (5)
(19)	$\sqrt{T_{at}/T_{as}}$		(16) + (17)
(20)	Fuel remaining,	gals	
(21)	Fuel at take off,		
(22)	Fuel used,	gals	(21) - (20)
(23)	ΔWe	168	Fuel weight used in climbing, (22) r lbs/gal
(24)	W <sub>t</sub>	1bs	Test weight, take-off gross weight - (23)
(25)	٧Å	1 <b>be</b>	Standard weight at altitude, from check
	•		climb or weight at initial sawtooth point
(26)	ΔW	1b#	Climb weight correction, (25) - (24)
(27)	<b>A W</b> per engine	1 <b>b</b> #	(26) + number of engines
(28)	W <sub>t</sub> per engine	lbø	(24) + number of engines
(29)	te	Decimal	Elapsed time to glimb to altitude
	•	Minutes	
(30)	t <sub>e</sub>	Decimal	Time at start of climb
-	T	Minutes	
(31)	Δt	Decimal	Time to climb to test altitude, (29) - (30)
		Minutes	•
(32)	Plot (5) 🕶 (31)		Time to climb curve
(33)	(dH <sub>o</sub> /dt)	ft/min	Altimeter rate of climb from elopes of (32)

AFTR 6273

(34)	(dh/dt)	ft/min	Tapeline rate of climb at standard day
			true speed, test weight, and test power,
(35)	Wind Gradient	Factor	(33) x (19) CHART 5.41, (when required)
(36)			nt (34) x (35) when required
(37)	Acceleration F		CHART 5.51 and (5) and (9) for sawtooth
			climb at constant $V_c$ . CHARTS 5.51 and 5.52 and (5) and (9) for check climb not
			flown at constant $\nabla_c$ .
(38)	Throttle setti	_	
(39)	Mixture settir	- T .	
(40)	Tachometer Rea		~
(41) (42)	Engine rpm	trument Correctio	(40) + (41)
(43)	MP1	"Eg	Indicated manifold pressure
(44) (44)	$\Delta MP_{1c}$	"Hg	Manifold pressure instrument correction
(45)	MP <sub>t</sub>	*Eg	Test manifold pressure, (43) + (44)
(46)	BHPt	•	Test BHP, from power charts or torque
	Ŭ		meters
(47)	BHP <sub>t</sub> per engin	10	(46) + number of engines
(48)	$(MP_t/P_a)$		Manifold pressure - ambient pressure
(1			ratio, $(45) + (18)$
(49)		sure ratio factor	From CHART 5.21 and (19) and (48)
(50)	AR/C1		Rate of climb correction for power at unaccelerated climb conditions, from
	BEP <sub>t</sub> per engin	20	CHART 5.21 and (19) and (48) and (28)
(51)	$(\Delta R/C_1)_{A}$	ft/min	Rate of climb correction for power,
()-()	(21,01,8	••/====	accelerated climb conditions, (50) x
			(47) + (37)
(52)	$(R/C)_{a} + (\Delta R/C_{1})_{a}$ ft/min		Standard day rate of climb at standard
		- •	power and test weight for the acceler-
			ating aircraft, $(34)$ or $(36) + (51)$
(53)	MP		$\{[(49) - one] = x (19)\} + two$
	ме <sub>t</sub>		
(54)	жР <u>а</u>	"Eg	Standard day manifold pressure, (53) x (45)
(55)	$\Delta R/C_2$	ft/min	Rate of climb weight correction, acceler-
	-	·	ated conditions, (-26) $x$ (52) + (24)
(56)	∆ R/C3		Rate of climb induced drag correction, un-
	ΔW		accelerated conditions, from CHART 5.31
()		<b>a</b> 1-1-1	and (5) and (11) and wing span
(57)	$(\Delta R/C_3)_a$	ft/min	Rate of climb induced drag correction for
			accelerated climb conditions, (56) x (26) + (37)
(58)	Standard weig	ht corrected	(52) + (55) + (57)
	rate of climb		
	conditions	,	
(59)		(5) and graphical	ly integrate to get time to climb vm (5)
	NOTE: For tu	rbo superchargers	the corrections described in Chapter Two
	should	be applied, and (	equation 5.208 used to find $\Delta R/C_1$ .
			•

•

AFTR 6273

# DATA REDUCTION OUTLINE (5.92)

For Jet Aircraft Sawtooth and Check Climbs

(1)	Test Point Number		
(2)	E	ft	Pressure altitude
(3)	A H <sub>10</sub>	ft	Altimeter instrument correction
(4)		ft	Altimeter position error correction
(5)	Ho	ft	True pressure altitude, $(2) + (3) + (4)$
(6)	v <sub>i</sub>	knote	Indicated air speed
(7)	$\Delta \nabla_{i}$	knots	Air-speed instrument correction
(8)	Δ Ϋ	knota	Air-speed position correction
(9)		knote	Calibrated air speed, $(6) \neq (7) + (8)$
(10)	V T		Mach number, from CHART 8.5 and (9)
	••		and (5)
(11)	٧.	knots	Standard day true speed, from CHART 8.5
( <del></del> /	V <sub>te</sub>	*	and (9) and (5)
(12)	ta	<b>-</b> C	and (9) and (5) Indicated air temperature
(12) (13)	t <u>i</u> Ate-	•C	Temperature instrument correction
(15) (14)		•C	Instrument corrected indicated air
~~/	tic	•	Instrument corrected indicated air temperature
(15)	T. +	•x	temperature Absolute test air temperature from CHART
(47)	Tat	-	8.2 and (14) and (10)
(16)	$\sqrt{\theta_{at}}, \sqrt{T_{at}/T_{SL}}$		$\gamma(15)/288$
(17)	$\gamma \theta_{as}, \gamma T_{as}/T_{SL}$		From standard altitude tables and (5)
(18)	$V \frac{as}{T_{at}/T_{as}}$		(16) + (17)
(19)	Fuel Remaining	gals	
(19)	Fuel at take off		
(20)	Fuel at take off Fuel used	gals	(20) - (19)
(22)		gnis Ibs	(20) - (19) Fuel weight used for climbing <sub>e</sub> (21) x lbs/gal
(22)	-		
(23) (24)	W <sub>t</sub>	lbe lbe	Test weight, take-off gross weight - (22) Standard weight at altitude from check climb.
(24)	Wa	15 <b>0</b>	Standard weight at altitude from check climb, or weight at initial savtooth point
(25)	A 4	1 <b>h</b> #	
(25) (26)	$\Delta W$	lbe lbe	Climb weight correction, (24) - (23) (25) + number of engines
(26) (27)	AW per engine	lbe lbe	(25) + number of engines
(27) (28)	W <sub>t</sub> 'per engine	lbs Decimal	(23) + number of engines
(28)	te	Decimal Minutes	Elapsed time to climb altitude
(00)	<b>↓</b> .	Minutes Decimal	Mine as asked as all-
(29)	te	Decimel Minutes	Time at start of climb
(70)	A 4	Minutes	Mana da aldah da
(30)	Δt	Decimal Minutes	Time to climb to test altitude, (29) - (28)
/7• `	Blat (6) . (64)	Minutes	
(31)	Plot (5) vs (30)	<b>A</b>	Time to climb curve
(32)	dHc/dt	ft/min	Altimeter rate of climb, from slopes of (31)
(33)	ðh/ðt	ft/min	Tapeline rate of climb at standard day true
		•	speed, test weight and test thrust, $(32)$ x
/=1. *		A	
(34)	Wind Gradient Fac	tor	CHART 5.41, (when required)

AFTR 6273

(35)	dh/dt		$(33) \times (34)$ , when required
(36)	Acceleration	Wind Gradient Factor	CHART 5.51 and (5) and (9) for climb at constant $V_c$ . CHARTS 5.51 and 5.52 and (5) and (9) for climb not flown at constant $V_c$
(37)	Ni		Indicated rpm
(38)			Rpm instrument correction
(39)	N <sub>t</sub>		Test rpm, $(37) + (38)$ , average of all
	· _		engines
(40)	N/Y Bat		Test rpm parameter, $(39) + (16)$
(41)	$N/\sqrt{\theta_{AB}}$		Standard rpm parameter, (39) + (17)
(42)	$N/\gamma \Theta_{Ae}$ $\Delta F_{D}/\delta A$		Net thrust correction from (40) and (41)
			and (10) and a graph similar to Figure 5.21
(43)	$\Delta R/C_1$		Rate of climb thrust correction, unacceler-
	$\frac{\Delta R/C_1}{\Delta F_n/\delta_s}$		ated conditions, from CHART 5.22 and (5) and (10) and (27)
(44)	$(\Delta R/C_1)_{A}$	ft/min	Rate of climb thrust correction for accele-
	(		rated climb speed, $(43) \times (42) + (36)$ . If
		N - 1	no acceleration, $(44) = \Delta R/C_1$
(45)	$(R/C)_{a} + (\Delta R/C)_{a}$	C <sub>1</sub> ) <sub>a</sub> ft/min	Standard day rate of climb at standard
			power and test weight for the accelerating
(46)	$\Delta R/C_2$	ft/min	aircraft, $(33)$ or $(35) + (44)$
(-0)	AR/ V2	10/44131	Rate of climb weight correction, accelerated conditions, $(-25) \times (45) \leftrightarrow (23)$
(47)	AR/C-		Rate of climb induced drag correction, un-
( )	$\frac{\Delta \mathbf{R}/\mathbf{C_3}}{\Delta \mathbf{W}}$		accelerated climb speed, from CHART 5.31 and
	ΔW		(5) and (10) and wing span
(48)	$(\Delta R/C_3)_{a}$	ft/min	Rate of climb induced drag correction for
/	·/-7/4		accelerated climb speed, $(47) \ge (25) + (36)$ .
			If no acceleration, $(47) = \Delta R/C_3$
(49)	Standard weig rate of climb		(45) + (46) + (48)

- (49) Standard weight corrected rate of climb, accelerated conditions
- (50) Plot (49) vs (5) and graphically integrate to get time to climb vs (5)
  - NOTE: If unaccelerated climb data is desired, correct (33) or (35) to unaccelerated conditions and make other corrections assuming no acceleration.

AFTR 6273

#### SECTION 5.10

#### Rate of Descent Data

Descent data are usually presented in the performance report only for jet powered aircraft. This is because the range of jet aircraft increases considerably more with altitude than does the range of reciprocating engine aircraft. Best range in jet aircraft is obtained by flying at a very high altitude until a minimum amount of fuel remains. This is followed by a high speed descent at minimum engine fuel pressure and resulting low fuel flow rate.

It is also important for tactical considerations to evaluate control forces; buffeting and other undesirable dive characteristics. The effects of dive brakes, tip tanks and external armament on high speed descents is extremely important in fighter and ground support aircraft.

Rate of descent data reduction is identical to rate of climb data reduction, except that no thrust correction is made for temperature variation from standard. This is because the descents are made at constant fuel pressure and the effects of temperature on thrust are very small relative to the high rates of descent. The tapeline rate of descent at standard day true speed is defined by.

$$-\frac{dh}{dt} = R/D = -\frac{dH_c}{dt} \sqrt{\frac{T_{at}}{T_{as}}}$$
(5.10.01)

The weight corrections are,

$$\frac{\Delta R/D_2}{\Delta W} = -\frac{R/D}{W_t}$$
(5.10.02)  
$$\frac{\Delta R/D_3}{\Delta W} = \frac{50.65 \sqrt{T_{as}}}{P_a M b^2 s}$$
(5.10.03)

where:

 $\Delta W = W_{\pi} - W_{t}$ 

Equation 5.10.03 is solved by CHART 5.31, but the sign of the parameter  $(\Delta R/C)_3/\Delta W$  must be changed when the chart is applied to determine  $(\Delta R/D)_3/\Delta W$ . Since the aircraft accelerates during the descent at constant Mach number,  $(\Delta R/D)_3$  from the chart should be corrected to the accelerated condition. This may be done by (5.10.04). Fuel consumption weight corrections are not required.

$$(\Delta R/D_3)_a = \frac{(\Delta R/D_3) \text{ chart}}{(1 + 0.133 \text{ m}^2)}$$
 (5.10.04)

In the aircraft performance report, rate of descent data is presented as in Figure 5.10.1 to show rate of descent, time to descend, Mach number, true speed, distance traveled, fuel pressure, and fuel used.

AFTR 6273



Figure 5.10.1 Report Presentation of Descent Data

AFTR 6273



AFTR 6273

CHART 5.21

# TURBOJET RATE OF CLIMB PORER CORRECTION



CHART 5.22

5-36



CHART 5.22



AFTR-6273



AFTR 6273

5-38



APTR 6273

CHART 5.23



AFTR 6273

**CHART 5.31** 



ATTE 6273

CHART 5.31





CHART 5.41





5-43











#### CLIMB ACCELERATION CORRECTION FACTOR CHART 5.54 PATI



AFTE 6273

#### CHAPTER SIX

#### TAKE-OFF AND LANDING PERFORMANCE

#### SECTION 6.1

## Techniques and Configurations for Take-Off Tests - JATO Operation

Take-off tests consist of a series of take-offs to determine the ground distance from the start of the run to the point where the aircraft leaves the ground, and the air distance from this point to the place where the aircraft is 50 feet above the runway. During the tests, the airplane should be operated in a manner considered to give the best take-off performance within the operational limits of the airplane and engine. Unless otherwise specified, the gross weight for take-off tests includes the maximum load likely to be used in service. All tests are run as close as possible to the desired gross weight to keep all weight corrections to a minimum. As local wind conditions affect the take-off distance and techniques used, tests should not be conducted when the wind velocities exceed 10% of the take-off speed of the aircraft.

For airplanes with wing flaps that are used for take-off, several flap positions are used to determine the optimum flap position. Cowl flaps are open for take-off, the airplane is held with brakes and maximum allowable power is attained prior to brake release. Maximum power is obtained as soon as practicable during the take-off run. As an example, the F-51 aircraft is limited to 40 inches manifold pressure without anchoring the tail, but 61 inches is used for take-off power. Therefore, the maximum allowable power prior to brake release is 40 inches and the maximum power is 61 inches.

When jet assist take-offs are made and it is desired to obtain the most advantage from a short duration rocket, the rocket should be ignited at such a point during the take-off run that it will be expended as the airplane passes over the 50 foot obstacle. This point of ignition is normally obtained by estimating the number of seconds after brake release that the rocket should be ignited, and then bracketing this point by making one or two take-offs. In most cases this information should be given to the pilot in the form of a number of seconds after brake release; however, it may sometimes be advantageous to ignite the rockets at an indicated air speed. For long burning rockets, the shortest take-off will be made by igniting the rockets at brake release.

### SECTION 6.2

#### Distance and Height Measurements and Equipment

Measurements should be taken to determine the distance from the take-off starting point to the place where the aircraft leaves the ground and to the point where the airplane reaches the altitude of 50 measurements may be made in various ways. A few of the methods a general use follow:

When camera equipment is not available either of the following systems may be used. The first consists of several theodolites (sighting bars) spaced along the runway so as to cover the distance from take-off point to the simulated 50 foot obstacle. The distance and time from take-off point to each sighting station will give an approximation of the aircraft speed and take-off distance. This method is shown schematically in Figure 6.21.



#### Figure 6.21 Basic Method for Obtaining Take-Off and Landing Time and Distance Data

It is good practice to station two or three observers at the edge of the runway in the vicinity of the take-off point to mark the exact point of take-off. The data obtained by such observers are always a good check on ground roll distance regardless of the method used for obtaining data. Using this method the height of the aircraft above the runway may be obtained by a formula determined from Figure 6.22.

AFTR 6273

where:

 $H = \frac{Dh}{\Delta H} + \Delta H$ 

- H = height of aircraft above the theodolite, ft.
- D distance from ruway centerline to theodolite eye piece, ft.
- h = height read on the odolite as aircraft passes, ft.
- d = length of theodolite, ft.
- AH = height of theodolite above runway, ft.





A more accurate field method of obtaining take-off data consists of a theodolite pivoted so it may track the aircraft during the take-off run, Figure 6.25. The theodolite is constructed in such a way that, by keeping cross-hairs on the aircraft, a pencil trace of the aircraft position is placed on a chart fastened rigidly to the theodolite supports. The swiveling theodolite is set up at a known distance from the runway and so aligned as to encompass only as much of the runway as will be necessary for the tests of the particular aircraft under consideration. This is done to obtain the greatest accuracy from the instrument.

Various standard distances from the runway may be arbitrarily determined and charts prepared in advance for use on this theodolite. A timing mechanism built into the sighting bar marks every second on the chart. Ground observers are used to mark the exact point of take off, and this information may be placed on the obart at the end of each test. A typical chart and take-off graph is illustrated in Figure 6.24.

AFTR 6273



Figure 6.23 Take-off Data Installation for Swiveling Camera or Recording Theodolite





The most accurate means of recording take-off data is with a moving picture camera which will photograph the aircraft under test, a timing device, and either a grid or azimuth scale. A portable 16 mm. camera has been devised which photographs the runway, a stop watch and the position of the airplane with respect to an azimuth scale. Knowing the azimuth and the distance of the camera from the

AFTR 6273

runway along a normal to the runway, the position of the aircraft may be determined at any time during the test. A typical frame from a test camera would appear as in Figure 6.25.





A fixed grid may also be used to photographically record the tests. In this method a grid consisting of a network of calibrated wires is placed in front of a normal camera in the manner shown in Figure 6.23 and at such a distance that it will remain in focus along with the airplane being tested. A timing device may be mounted on either the grid or camera to give a time history of the take off or landing. A typical frame taken through this type of grid is shown in Figure 6.26.



Figure 6.26 Typical Frame from Camera-Caild Film

AFTR 6273

## SECTION 6.3

#### Take -Off Data Corrections for Wind, Weight, and Density

From information obtained by any of the above methods in the previous section, the observed data may be plotted as in Figure 6.31. This figure is usually included in the final report as is the corrected take-off data.



Figure 6.31 Presentation of Actual Take-off Time and Distance Data

The indicated air speed at take-off and 50 feet as well as the true air speed should be shown, as the pilot is primarily concerned with the air speed that is read on the instrument panel, and this value is independent of pressure and temperature.

All take-off performance data are corrected to sea level standard conditions and zero wind unless otherwise specified. The average of the best two of at least four take-offs is reported as performance data. Corrected data are usually presented in the following chart form:

AFTR 6273

A - CONFIGURATION - WEIGHT - POWER SETTING

(1) Ground Roll	(2) Total Distance over 50' height	(3) Indicated Speed		(4) True Speed				
(ft)		At Take-Off	At 50'Beight	At Take-Off	At 50' Height			
B - CONFIGURATION - WEIGHT- POWER SETTING								
C - CONFIGURATION - WEIGHT - POWER SETTING								

#### TAKE-OFF DATA CORRECTION

The taks-off performance of any aircraft is highly dependent on pilot technique. Even with experienced well-qualified pilots it is difficult to make the aircraft take off at the same value of lift coefficient each time. As this is the rule rather than the exception, a rigorous mathematical treatment of reducing observed take-off data to standard conditions is not warranted; therefore, no mathematically exact solutions will be given for reducing data.

The correction of ground roll for the effect of wind may be empirically expressed as,

 $\mathbf{S}_{t} = \mathbf{S}_{tw} \quad \left(\frac{\mathbf{v}_{to} + \mathbf{v}_{w}}{\mathbf{v}_{to}}\right)^{1.85} \tag{6.501}$ 

where:

This relationship has been verified by extensive flight tes . Neglecting wind, the ground roll during take off is,

 $\mathbf{S} = \int_{0}^{V_{\text{to}}} \frac{\mathbf{Y}}{\mathbf{a}} \, \mathrm{d}\mathbf{V} \tag{6.302}$ 

V<sub>to</sub> = true speed at take-off point a a acceleration

Awauming "a" to have an effective constant value at a mean value of  $\nabla^2$  the above expression becomes,

$$S = \frac{1}{2} \frac{V_{to}^{2}}{a \oplus (.7 V_{to})}$$
 (6.303)

The effective thrust acting throughout the take-off is defined as the difference between propulsive net thrust at .7  $V_{to}$  and the aircraft resistance to forward motion at the same point. This take-off thrust equation may be written as,

$$F_{\text{eff}} = \frac{\Psi}{g} = \frac{\Psi \nabla_{to}^2}{2gS}$$
(6.304)

The basic take-off distance equation then becomes,

$$s = \frac{V V_{to}^{2}}{2g F_{off}} = \frac{V V_{to}^{2}}{2g \left[F_{n} - \mu (V - L) - \bar{D}\right] \bullet .7 V_{to}}$$
(6.305)

One method of evaluating the effect of small changes in the variables of equation 6.305 is by use of logarithmic differentiation. With this mathematical process and with the assumption of constant  $C_{\rm L}$  at take-off for all conditions of the variables,

$$\frac{dD}{D} = \frac{dL}{L} = \frac{dW}{V}$$

where,

D = aircraft drag, L = aerodynamic lift

then

$$\frac{dS}{S} = \frac{dW}{V} - \frac{d\sigma}{\sigma} + \frac{T_n}{T_{off}} \left( \frac{dW}{V} - \frac{dT_n}{T_n} \right) + \frac{d\mu}{\mu} \left( \frac{T_n - D}{T_{off}} - 1 \right) \quad (6.306)$$

or in another form

$$\frac{\mathbf{s}_2}{\mathbf{s}_1} = 1 + \frac{\mathbf{y}_2}{\mathbf{y}_1} - \frac{\mathbf{g}_2}{\mathbf{\sigma}_1} + \frac{\mathbf{y}_n}{\mathbf{y}_{off}} \left( \frac{\mathbf{y}_2}{\mathbf{y}_1} - \frac{\mathbf{y}_{n_2}}{\mathbf{y}_{n_1}} \right) + \left( \frac{\mathbf{y}_2}{\mathbf{y}_1} - 1 \right) \left( \frac{\mathbf{y}_n - \mathbf{D}}{\mathbf{y}_{off}} - 1 \right) (6.307)$$

where  $T_n = T_n \oplus .7 V_{to}$ 

M = coefficient of rolling resistance

Empirical values of Mare

- M = .02 for hard surface runway
- M = .04 for firm turf
- A = .10 for soft turf

These equations should be used only where the ratios of the variables lie between 0.9 and 1.1.

If there is a weight variation the indicated speeds must be adjusted for a constant  $O_{\rm L}$  assumption.

$$\frac{\mathbf{v}_{c_1}}{\mathbf{v}_{c_2}} = \sqrt{\frac{\mathbf{v}_1}{\mathbf{v}_2}}$$

An exact relationship between take-off distances for large changes in the variables of equation 6.305 can be found by making  $\mathcal{M}$  constant and defining. S<sub>2</sub> in terms of S<sub>1</sub>, V<sub>to2</sub> in terms of V<sub>to</sub>, and F<sub>eff2</sub> in terms of F<sub>eff1</sub>, at constant C<sub>L</sub>. This expression is,

$$\frac{\mathbf{S}_{2}}{\mathbf{S}_{1}} = \frac{\mathbf{W}_{2}}{\mathbf{W}_{1}} \left. \frac{\mathbf{g}_{1}}{\mathbf{r}_{2}} \right/ \left[ \frac{\mathbf{r}_{\mathbf{n}_{1}}}{\mathbf{r}_{\mathbf{o}ff_{1}}} \left( \frac{\mathbf{W}_{1}}{\mathbf{W}_{2}} \cdot \frac{\mathbf{r}_{\mathbf{n}_{2}}}{\mathbf{r}_{\mathbf{n}_{1}}} - 1 \right) + 1 \right]$$
(6.308)

Equations 6.306, 6.307, and 6.308 as shown are directly applicable to turbojet aircraft where net thrust is quite easily determined for take-off conditions.

### Correction for Bunway Slope and CL Variation

It is sometimes necessary to correct take-off data for ranway slope. This is a simple geometric consideration. The effective thrust is.

dividing by Feffalope and substituting equation 6.304,

$$S_{level} = \frac{S_{slope}}{1 + \frac{2\varepsilon}{v_{to}}}$$
(6.309)

To correct data to constant  $C_L$  a relationship is found by multiplying equation 6.304 by (W/W) = 1.

Then,  

$$8 = \frac{V^2}{2a^3} + \frac{V_1^2}{V}$$

and

$$\mathbf{S}_{2} = \mathbf{S}_{1} \frac{(\mathbf{V}_{10}^{2}/\mathbf{W})_{2}}{(\mathbf{V}_{10}^{2}/\mathbf{W})_{1}} \quad (\text{constant } \sigma^{2}) \quad (6.310)$$

The slope and O<sub>L</sub> corrections along with the wind corrections should be applied to test data prior to density weight and thrust corrections.
#### AIR DISTANCE DATA CORRECTION

To determine the corrected horisontal air distance from lift-off point to clear a fifty foot obstacle the correction to zero wind is expressed as the product of wind velocity and time,

(6.311)

$$S'_a = S'_a + V_v t$$

where:

S'a = wind corrected test air distance S'a = observed air distance in wind V = wind component along runway t = time from lift-off to 50 ft. point

Meglecting wind the following expressions may be written for the air distance and the aircraft energy change through it.

$$V = \frac{S_A}{t}$$
 and  $\frac{dE}{dt} = \frac{(50 + h_V) W}{t}$ 

where:

V = mean true airspeed between take-off and 50 ft. height.

 $\mathbf{E} = (50 + \mathbf{V}^2/2g)\mathbf{W}$ , total energy of the aircraft.

- V = gross weight
- $h_{v} = (v_{50}^{2} v_{to}^{2})/2g$  $\frac{dE}{dt} = v(r - D)$

7 = net thrust

D = total air drag

Combining these expressions and substituting  $\nabla(T - D)$  for dE/dt a general expression for the air distance is derived.

$$S_{a} = \frac{V(50 + h_{T})}{(T - D)_{eff}}$$
(6.312)

Logarithmic differentiation may be applied to equation 6.312 to determine the effect on air distance of small changes in the variables. This process gives for a constant  $C_L$  at  $V_{to}$  and  $V_{50}$ .

AFTE 6273

$$\frac{d\mathbf{S}_{\mathbf{A}}}{\mathbf{S}_{\mathbf{A}}} = \left(\frac{d\mathbf{V}}{\mathbf{V}} - \frac{d\sigma}{\sigma'}\right) \frac{\mathbf{h}_{\mathbf{V}}}{50 + \mathbf{h}_{\mathbf{V}}} + \left(\frac{d\mathbf{V}}{\mathbf{V}} - \frac{d\mathbf{F}_{\mathbf{n}}}{\mathbf{F}_{\mathbf{n}}}\right) \frac{\mathbf{F}_{\mathbf{n}} \cdot \mathbf{S}_{\mathbf{A}}}{\mathbf{V} \cdot (50 + \mathbf{h}_{\mathbf{V}})} \quad (6.313)$$

or in another form

$$\frac{\mathbf{s}_{a_{2}}}{\mathbf{s}_{a_{1}}} = 1 + \left(\frac{\mathbf{w}_{2}}{\mathbf{w}_{1}} - \frac{\sigma_{2}}{\sigma_{1}}\right) \frac{\mathbf{h}_{\mathbf{w}}}{50 + \mathbf{h}_{\mathbf{w}}} + \left(\frac{\mathbf{w}_{2}}{\mathbf{w}_{1}} - \frac{\mathbf{w}_{u_{2}}}{\mathbf{w}_{1}}\right) \frac{\mathbf{y}_{u_{1}} \cdot \mathbf{s}_{a_{1}}}{\mathbf{w}_{1} \cdot (50 + \mathbf{h}_{\mathbf{w}})} \quad (6.314)$$

where  $V_n =$  net thrust at a mean speed between  $V_{to}^2$  and  $V_{50}^2$ .

These equations should be used only where the ratios of the variables lie between 0.9 and 1.1.

An exact relationship between air distances for large changes in the variables of equation 6.312 can be found by defining  $S_{a,2}$  in terms of  $S_a$ .  $V_{to_2}$  and  $V_{50_2}$  in terms of  $V_{to_1}$  and  $V_{50_1}$  and  $(F - D)_2$  in terms of  $(F - D)_1$  at constant  $C_{L_{to}}$  and  $V_{L_{50}}$ . This expression is,

$$\frac{s_{a_2}}{s_{a_1}} = \frac{\frac{v_2}{v_1} \frac{\sigma_1}{\sigma_2} h_{v_1} + 50}{h_{v_1} + 50 + s_{a_1} \left(\frac{v_{a_2}}{v_2} - \frac{v_{a_1}}{v_1}\right)}$$
(6.315)

Equations 6.313, 6.314, and 6.315 as shown are directly applicable to turbojet aircraft where not thrust is quite easily determined for take-off conditions.

BMPIRICAL FORMULAS FOR CORRECTION OF GROUND ROLL AND AIR DISTANCE DATA OF BOTH JET AND PROPELLER POWERED AIRCRAFT

The following expressions for the effects of changes in the independent variables involved in equations 6.305 and 6.312 were developed and checked against experimental data by Mr. K.J. Lush of the Fright Research Branch, Air Force Flight Test Center. The complete study and analysis may be found in, "Standardisation of Take-Off Performance Measurements for Airplanes", Technical Note B-12, Air Force Flight Test Center, Edwards Air Force Base, Edwards, California.

These formulas were developed by application logarithmic differentiation to equations 6.305, and 6.312, and to applicable propeller relationships. The constants were determined primarily by graphical analysis of a large amount of take-off data from typical aircraft. Corrections obtained by these formulas will give sufficiently accurate data for changes in the variables up to  $\pm 20\%$ .

The General Case: For propeller driven and jet airplanes, the general equations (6.316) and (6.317) will be used.

ATTE 6273

$$\frac{\Delta S_{f}}{S_{f}} = \frac{\Delta W}{W_{t}} \left\{ 2 + \frac{D}{P - D} \right\} - \frac{\Delta \sigma}{\sigma_{t}} - \frac{\Delta F}{F_{t}} \left\{ 1 + \frac{D}{F - D} \right\}$$

$$\frac{\Delta S_{h}}{S_{h}} = \frac{\Delta W}{W_{t}} \left\{ 1 + \frac{D}{P - D} + \frac{h_{\psi}}{h_{\psi} + 50} \right\} - \frac{\Delta F}{F_{t}} \left\{ 1 + \frac{D}{P - D} \right\}$$

$$- \frac{\Delta \sigma}{\sigma_{t}} \left\{ \frac{h_{\psi}}{h_{\psi} + 50} \right\}$$
(6.316)
(6.317)

With the proposed constants

$$\frac{\Delta S}{S_{t}} = 2.3 \frac{\Delta V}{V_{t}} - \frac{\Delta \sigma}{\sigma_{t}} - 1.3 \frac{\Delta F}{F_{t}}$$
(6.318)

alternatively 
$$\frac{S_{\mathcal{E}_{g}}}{S_{\mathcal{E}_{t}}} = \left(\frac{\frac{W_{g}}{W_{t}}}{W_{t}}\right)^{2\cdot3} \left(\frac{\sigma_{t}}{\sigma_{g}}\right) \left(\frac{P_{g}}{P_{t}}\right)^{-1\cdot3}$$
 (6.319)

$$\frac{\Delta S_a}{S_{a_t}} = 2.3 \frac{\Delta W}{W_t} = 0.7 \frac{\Delta \sigma}{\sigma_t} = 1.6 \frac{\Delta F}{F_t}$$
(6.320)

or alternatively 
$$\frac{\mathbf{S}_{\mathbf{a}_{\mathbf{g}}}}{\mathbf{S}_{\mathbf{a}_{\mathbf{t}}}} = \left(\frac{\mathbf{W}_{\mathbf{g}}}{\mathbf{W}_{\mathbf{t}}}\right)^{2.3} \left(\frac{\underline{\sigma}_{\mathbf{g}}}{\overline{\sigma}_{\mathbf{t}}}\right)^{-0.7} \left(\frac{\mathbf{F}}{\mathbf{F}_{\mathbf{t}}}\right)^{-1.6}$$
 (6.321)

for light airplanes

07

and

$$\frac{\Delta S_a}{S_a} = 2.0 \frac{\Delta V}{V_t} - 0.4 \frac{\Delta \sigma}{\sigma_t} - 1.6 \frac{\Delta F}{F_t}$$
(6.322)

For moderate corrections either type of equation is satisfactory, but if the corrections are large (for example,  $\Delta S/S \setminus < 0.2$ ) the exponential forms will be appreciably more accurate.

# Propeller Thrust Corrections:

For fixed pitch propellers,

$$\frac{\Delta \mathbf{F}}{\mathbf{Y}_{t}} = 1.1 \frac{\Delta \mathbf{F}}{\sigma_{t}} - 0.1 \frac{\Delta \mathbf{Y}}{\mathbf{Y}_{t}}$$

(6.323)

at constant engine speed

AFTR 6273

$$\frac{\Delta \mathbf{F}}{\mathbf{F}_{t}} = 1.1 \frac{\Delta \sigma}{\sigma_{t}} + 0.4 \frac{\Delta \mathbf{T}_{a}}{\mathbf{T}_{a_{t}}} - 0.1 \frac{\Delta \mathbf{Y}}{\mathbf{W}_{t}}$$
(6.324)

at full throttle

While for constant speed propellers

$$\frac{\Delta \mathbf{I}}{\mathbf{F}_{t}} = 0.7 \frac{\Delta \mathbf{P}}{\mathbf{F}_{t}} + 0.5 \frac{\Delta \sigma}{\sigma_{t}} - 0.5 \frac{\Delta \mathbf{H}}{\mathbf{N}_{t}} - 0.2 \frac{\Delta \mathbf{W}}{\mathbf{V}_{t}}$$
(6.325)

Take-Off and Air Distance Corrections for Fixed Pitch Propellers: Corrections may be required at constant engine speed or at full throttle.

At constant engine speed

$$\frac{\Delta S}{S_{t}} = 2.4 \frac{\Delta W}{W_{t}} - 2.4 \frac{\Delta \sigma}{\sigma_{t}}$$
(6.326)

and

$$\frac{\Delta S_a}{S_{a_t}} = 2.2 \frac{\Delta W}{W_t} - 2.2 \frac{\Delta \sigma}{S_t}$$
(6.327)

If  $\Delta S/S$  is numerically large, it is again preferable to use the exponential forms

$$\frac{\mathbf{S}_{\mathbf{S}_{\mathbf{S}}}}{\mathbf{S}_{\mathbf{S}_{\mathbf{t}}}} = \left(\frac{\mathbf{W}_{\mathbf{S}}}{\mathbf{W}_{\mathbf{t}}}\right)^{2.4} \left(\frac{\sigma_{\mathbf{S}}}{\sigma_{\mathbf{t}}}\right)^{-2.4}$$
(6.328)

and

$$\frac{\mathbf{s}_{\mathbf{a}_{\mathbf{g}}}}{\mathbf{s}_{\mathbf{a}_{\mathbf{t}}}} = \left(\frac{\mathbf{v}_{\mathbf{g}}}{\mathbf{v}_{\mathbf{t}}}\right)^{2.2} \left(\frac{\sigma_{\mathbf{g}}}{\sigma_{\mathbf{t}}}\right)^{-2.2}$$
(6.329)

At full throttle there will be a correction to engine speed

.

$$\frac{\Delta S_{g}}{S_{g_{t}}} = 2.4 \frac{\Delta V}{V_{t}} - 2.4 \frac{\Delta \sigma}{\sigma_{t}} + 0.5 \frac{\Delta T_{g}}{T_{g_{t}}}$$
(6.330)

AFTR 6273

$$\frac{\Delta S_a}{S_{a_t}} = 2.2 \frac{\Delta W}{W_t} - 2.2 \frac{\Delta \sigma}{\sigma_t} + 0.6 \frac{\Delta T_a}{T_{a_t}}$$
(6.331)

with the corresponding exponential forms

$$\frac{\mathbf{S}_{g_{g}}}{\mathbf{S}_{g_{t}}} = \left(\frac{\mathbf{W}_{g}}{\mathbf{W}_{t}}\right)^{2.4} \left(\frac{\sigma_{g}}{\sigma_{t}}\right)^{-2.4} \left(\frac{\mathbf{T}_{a_{g}}}{\mathbf{T}_{a_{t}}}\right)^{0.5}$$
(6.332)

$$\frac{S_{a_{s}}}{S_{a_{t}}} = \left(\frac{V_{s}}{V_{t}}\right)^{2.2} \left(\frac{\sigma_{s}}{\sigma_{t}}\right)^{-2.2} \left(\frac{T_{a_{s}}}{T_{a_{t}}}\right)^{0.6}$$
(6.333)

<u>Constant Speed Fropellers</u>: This section applies to airplanes which are entirely, or almost entirely, propeller driven at take-off. For the ground roll,

$$\frac{\Delta S_{R}}{S_{g_{t}}} = 2.6 \frac{\Delta V}{V_{t}} - 1.7 \frac{\Delta \sigma}{\sigma_{t}} - 0.7 \frac{\Delta H}{N_{t}} - 0.9 \frac{\Delta P}{P_{t}}$$
(6.334)

and the alternative form

$$\frac{\mathbf{s}_{\mathbf{g}}}{\mathbf{s}_{\mathbf{g}_{t}}} = \left(\frac{\mathbf{w}_{\mathbf{g}}}{\mathbf{w}_{t}}\right)^{2.6} \left(\frac{\sigma_{\mathbf{g}}}{\sigma_{t}}\right)^{-1.7} \left(\frac{\mathbf{w}_{\mathbf{g}}}{\mathbf{w}_{t}}\right)^{-0.7} \left(\frac{\mathbf{p}}{\mathbf{g}_{t}}\right)^{-0.9}$$
(6.335)

For the air phase, distinguish between light and heavy airplanes. For light airplanes use,

$$\frac{\Delta S_{a}}{S_{a_{t}}} = 2.3 \frac{\Delta W}{W_{t}} - 1.2 \frac{\Delta \sigma}{\sigma_{t}} - 0.8 \frac{\Delta W}{W_{t}} - 1.1 \frac{\Delta P}{P_{t}}$$
(6.336)

or alternatively

$$\frac{S_{a_{g}}}{S_{a_{t}}} = \left(\frac{W_{g}}{W_{t}}\right)^{2.3} \left(\frac{\sigma_{g}}{\sigma_{t}}\right)^{-1.2} \left(\frac{W_{g}}{W_{t}}\right)^{-0.8} \left(\frac{P_{g}}{P_{t}}\right)^{-1.1}$$
(6.337)

AFTR 6273

6-14

and

For heavy airplanes in the air phase, use,

$$\frac{\Delta S_{t}}{S_{t}} = 2.6 \frac{\Delta W}{W} - 1.5 \frac{\Delta \sigma}{\sigma_{t}} - 0.8 \frac{\Delta N}{N_{t}} - 1.1 \frac{\Delta BHP}{BHP_{t}}$$
(6.338)

or alternatively

$$\frac{S_{a_{g}}}{S_{a_{t}}} = \left(\frac{W_{g}}{W_{t}}\right)^{2.6} \left(\frac{\sigma_{g}}{\sigma_{t}}\right)^{-1.5} \left(\frac{W_{g}}{W_{t}}\right)^{-0.8} \left(\frac{BBP_{g}}{BBP_{t}}\right)^{-1.1}$$
(6.339)

Thrust Corrections for Turbo Propellers: The general case equations 6.318 or 6.320 or their alternates, may be used for turbo propeller airplanes.

The thrust correction is given by

$$\frac{\Delta \mathbf{F}}{\mathbf{F}_{t}} = \frac{\Delta \mathbf{F}_{j}}{\mathbf{F}_{t}} + \left(\mathbf{1} - \frac{\mathbf{F}_{j}}{\mathbf{F}_{t}}\right) \frac{\Delta \mathbf{F}_{p}}{\mathbf{F}_{p_{t}}}$$
(6.340)

where:  $\mathbf{F}_{t} = \mathbf{F}_{j_{t}} + \mathbf{F}_{p_{t}}$ 

and estimate  $\triangle F_p/F_{p_t}$  by equation 6.325.

$$\frac{\Delta \mathbf{F}}{\mathbf{F}_{\mathbf{p}_{+}}} = 0.7 \frac{\Delta \text{BHP}}{\text{BHP}_{t}} + 0.5 \frac{\Delta \sigma}{\sigma_{t}} + 0.5 \frac{\Delta \mathbf{H}}{\mathbf{H}_{t}} - 0.2 \frac{\Delta \mathbf{W}}{\mathbf{H}_{t}} \quad (6.325)$$

 $P_t$ Generally for turbo propeller eircraft  $\Delta F$ ignored. Then  $F_t = \frac{\Delta F}{F_p}$   $F_t$ 

<u>Part-Time Assistance</u>: Again, equations 6.318 and 6.320 are used basically, but with an effective mean thrust. The consideration here is primarily JATO, but the method can be applied to other forms of thrust boost operated over a limited period.

The ter effective, mean thrust boost  $F_R$  in either the ground roll or air phase given by equation 6.341.

$$\mathbf{F}_{\mathbf{R}_{\mathbf{0}}} = \frac{\mathbf{S}_{\mathbf{R}}}{\mathbf{S}} \mathbf{F}_{\mathbf{R}}$$
(6.341)

where

ų

F<sub>R</sub> = JATO thrust AFTR 6273

 $S_R$  = distance covered in phase with JATO operating

# S = total length of phase

The standard effective mean thrust in the air phase is either zero (ATO to cease at take-off) or equal to the actual ATO thrust (ATO to last to 50 ft.). The standard effective mean thrus' in the ground roll, however, depends on the time,  $t_{a_{\rm e}}$ , during which the ATO is to operate in the air phase under standard conditions.

t<sub>R</sub> = 0 ATO ceasing at take-off

$$t_{R_{a_{a}}} = 2S_{a_{a}}/(v_{a_{to}} + v_{a_{50}}) \text{ for ATO coasing at 50 ft.}$$
(6.342)

Hence, if  $t_{a_t}$  was the test duration of the ATO in the air phase, the ATO duration in the ground roll must be corrected by

$$\Delta t_{R_{c}} = t_{R_{a_{t}}} - t_{R_{a_{s}}}$$
(6.343)

The correction to the air phase is then given by equation 6.320 using the total mean effective thrust. For the ground roll, however, use equation 6.344

$$\frac{\Delta S_{g}}{S_{g_{t}}} = \left\{ 2.3 - \frac{2.3}{\overline{r_{t_{t}}}} \right\} \frac{\Delta V}{V_{t}} - \left\{ 1 + \frac{0.7}{\overline{r_{t_{t}}}} \right\} \frac{\Delta \sigma}{\sigma_{t}} - \frac{1.3}{\overline{r_{t_{t}}}} - 1.3 = \frac{1.3}{\overline{r_{t_{t}}}} - \frac{1.3}{\overline{r_{t}}} - \frac{1.3}{\overline{r_$$

where t

Tt. = test total thrust

Y<sub>h</sub> = basic engine thrust

The thrust terms in all the above equations are the thrusts obtained at mean take-off or air distance speeds unless otherwise defined.

APTR 6273

#### SECTION 6.4

#### Landing Performance Tests and Corrections

Landing tests consist of a series of landings to determine the total distance required to pass over a 50 foot obstacle, touch down, and come to a complete stop. The aircraft should be operated in such a manner as to give the best landing performance within the operational limits of the airplane.

The gross weight for landing tests is usually the maximum load used in service; for heavily loaded aircraft it may be less one-half the fuel and less any dropable load. All tests will be run as close as possible to the desired gross weight, as weight corrections for landing roll have not been proved consistent. The aircraft landing configuration is normally with wing flaps full down, engine at idling rpm, and cowl flaps, when installed, full closed. Any special configuration for landing will be so stated. After the aircraft has touched down maximum braking power is applied without skidding.

The measurement of air and ground distance for landing is accomplished in the same manner as described for take-off tests. The observed landing time and distance data are plotted similarly to the observed take-off data. All landing performance data are corrected to standard conditions unless otherwise specified. The average of the best two of at least four landings is reported. As in take-offs, pilot technique is a large factor in determining landing performance. Approach technique and the use of brakes are extremely important in order to produce consistent results.

#### LANDING DATA CORRECTIONS

In converting the observed data to standard, sea level conditions, the wind corrections as used for take-off are again used. These are:

ground roll corrected for wind = obe. ground roll  $\left(\frac{V_{td} + V_{w}}{V_{td}}\right)^{1.85}$ 

air distance corrected for wind = obs. air distance + W t

During the approach from the 50 foot obstacle to touch down, the aircraft has both potential and meetic energy which must be dissipated prior to touch down. This may be expressed as,

$$\frac{W}{B} = \left(\frac{v_{50}^2}{2} - \frac{v_{14}^2}{2}\right) + 50 W = FB'$$
(6.401)

where:

F = retarding force acting over the air distance S'  $V_{50}$  = true speed at 50 foot height  $V_{td}$  = true speed at touch down

AFTR 6273

Solving for air distance,

$$s' = \frac{W}{F} = \left(\frac{v_{50}^2 - v_{td}^2}{2g} + 50\right)$$
 (6.402)

The term W/F is actually an average ratio of lift to drag during the descent. Due to ground effect and transition from glide to flare out, the value of L/P is difficult to obtain, and it must be assumed to be constant for all weight and density conditions. The difference in the velocities  $V_{50}$  and  $V_{td}$  may be said to be negligible between test and standard conditions. Therefore, it is seen that the test air distance equals the standard air distance, except for the effects of wind, and the final expression for correcting air distance during landing is given as,

$$\mathbf{S}_{jt} = \mathbf{S}_{jtw} + \mathbf{v}_{w} \mathbf{t} \tag{6.403}$$

where:

S't = landing air distance from a 50 foot obstacle, zero
wind, ft.
S'tw = test landing air distance from a 50 foot obstacle,
with wind component, ft.
Vw = wind component, ft/sec
t = time, sec

For the landing ground roll, the ground distance may be given as:

 $S = \int \frac{V}{a} dV$ 

Using an average deceleration during ground roll, this expression may be simplified by integration to the following form:

$$s = \frac{v_{td}^2}{-2a}$$
 (6.405)

(6.404)

where:

-a = deceleration

Assuming for constant test and standard day touch down true speeds that the change in  $C_{L}$  at standard and observed conditions is negligible, equation 6.405 may be standardized in the same manner us the take-off equation. The final expression for landing ground roll is,

$$S_{le} = S_{ltw} \left(\frac{V_{td} + V_{w}}{V_{td}}\right)^{1.85} \left(\frac{W_{e}}{W_{t}}\right)^{2} \left(\frac{\sigma_{t}}{\sigma_{e}}\right)$$
(6.406)

where:

 $S_{lo}$  = standard landing ground distance  $S_{lty}$  = test landing ground distance with wind component

AFIR 6273

and the total landing distance is then,

Total 
$$\mathbf{S}_{ls} = (\mathbf{s}_{ltw} + \mathbf{v}_{wt}) + \mathbf{s}_{ltw} \left(\frac{\mathbf{v}_{td} + \mathbf{v}_{w}}{\mathbf{v}_{td}}\right)^{1.85} \left(\frac{\mathbf{w}_{s}}{\mathbf{w}_{t}}\right)^{2} \left(\frac{\boldsymbol{\sigma}_{t}}{\boldsymbol{\sigma}_{s}}\right)$$
 (6.407)

From observation of data obtained during many landings, it has been found that the weight correction as shown in equation 6.407 is not reliable, and, in the event of a departure from standard weight during landings, no weight corrections have been found usable because of the many factors involved in landing technique; these factors are: approach speeds, flare-out pattern, application of brakes, etc. As a result, every effort is made to keep the test weight as close to the standard weight as possible, and the final usable equation for landing roll is,

Total 
$$S_{ls} = (S_{ltw} + V_{wt}) + S_{ltw} \left(\frac{V_{td} + V_{w}}{V_{td}}\right)^{1.85} \left(\frac{\sigma_{t}}{\sigma_{s}}\right)$$
 (6.403)

## JATA REDUCTION OUTLINE (6.41)

# For Landing Test Data

(1) (2)	Setw	ft ft/sec	Observed ground roll, with wind component Ground velocity at touch down, from slope
( - /	't <b>a</b>	10/000	of time-distance curve
(3)	Vw	ft/sec	Wind component along runway, from observed
(4)	Slt	ft	data; headwind (+), tailwind (-) Test ground roll corrected for wind, (1) x $\begin{bmatrix} (2) + (3) \\ (2) \end{bmatrix}^{1.85}$
(5)	Pa	"Hg	Test barometric pressure, from observed data
(6)	tat	•C	Test ambient temperature, from observed data
(7) (8)	$\sigma_t \sigma_e$		Test density ratio, $9.625 \times (5) + [273 + (6)]$ Standard density ratio, from standard alti- tude tables and field elevation
(9)	$\sigma_{t} / \sigma_{s}$		(7) + (8)
(10)	s <sub>le</sub>	ft	Standard ground roll for zero wind $(4) \ge (9)$
(11)	Sltv	ft	Observed air distance from 50 foot height, with wind component
(12)	t	<b>80</b> 0	Time from 50 foot height to touch down
(13)	t B <b>/s</b>	ft	Standard landing air distance from 50 foot height, for zero wind $(11) + [(3) \times (12)]$
(14)	Total S/s	ſt	Total standard landing distance from 50 foot height for zero wind, (19)+ (13)

# SECTION 6.5

# Dimensionless Parameters for Take-Off and Landing Performance Data

Application of dimensional analysis to take-off performance may sometimes assist the engineer by reducing the apparent number of independent variables and, more vitally, by associating variables such as air density which are out of his control in "non-dimensional" groups with controllable variables such as airplane gross weight and engine power or thrust.

The practical value of such an approach has not yet been assessed. It seems however, to be potentially valuable in certain cases. In particular, if the test values of the non-dimensional groups derived can be accurately controlled at the values corresponding to standard conditions the standardization process can thereby be almost eliminated. Otherwise, the main attraction of the approach would be to enable more efficient use of test data obtained under a wide range of test conditions, particularly when it is desired to predict from such tests the takeoff performance under a wide range of standard conditions.

Many non-dimensional groups can usually be made up in any one problem. Those given below are useful, but may be modified if circumstances so demand.

#### RECIPROCATING ENGINE AIRCRAFT

The functional relations between the ground roll and the independent variables for an airplane in a given configuration may be written in the following form, among others:

 $S_g = f_1(BEP, N, \rho, V, V_{t_1}, \mu, g)$ 

(6.501)

where:

S\_ = ground roll

A = some fixed area of the airplane

¥ = gross weight

N = engine speed

 $V_{t_1} = true$  speed at unstick

M = coefficient of friction (with ground)

Similarly we can write for the total distance to reach a height "h",

$$S_t = f_2(BBP, H, P, V, V_{t_1}, V_{t_2}, \mu, \varepsilon, h)$$
 (6.502)

where  $V_{t_2}$  is the true speed at the height "h".

AFTR 6273

By application of dimensional analysis to the above functional relations we may deduce that

$$\frac{\mathbf{S}_{\mathbf{E}}}{\sqrt{\mathbf{A}}} = \phi_1 \left( \frac{\mathbf{BHP}}{\mathbf{V}} \frac{\sqrt{\mathbf{AP}}}{\sqrt{\mathbf{V}}}, \frac{\mathbf{NA}^{\frac{1}{2}}}{\sqrt{\mathbf{E}}}, \frac{\mathbf{V}_{\mathbf{V}}}{\sqrt{\mathbf{V}}}, \frac{\mathbf{P}_{\mathbf{A}}}{\mathbf{V}}, \frac{\mathbf{P}_{\mathbf{A}}}{\mathbf{V}}, \frac{\mathbf{P}_{\mathbf{A}}}{\mathbf{V}}, \mathbf{M} \right)$$
(6.503)

$$\frac{\mathbf{s}_{\mathbf{t}}}{\sqrt{\mathbf{A}}} = \mathbf{p}_{2} \left( \frac{\mathbf{B} \mathbf{H} \mathbf{P} \cdot \mathbf{A} \mathbf{P}}{\mathbf{V} \cdot \mathbf{V}}, \frac{\mathbf{N} \mathbf{A}^{\frac{1}{2}}}{\mathbf{T} \mathbf{\varepsilon}}, \mathbf{V}_{\mathbf{t}_{1}} \sqrt{\frac{\mathbf{P} \mathbf{A}}{\mathbf{V}}}, \mathbf{V}_{\mathbf{t}_{2}} \sqrt{\frac{\mathbf{P} \mathbf{A}}{\mathbf{V}}}, \frac{\mathbf{\varepsilon} \mathbf{P}}{\mathbf{V}}, \mathbf{A} \mathbf{T} \mathbf{A}, \mathbf{\mu}, \frac{\mathbf{h}}{\mathbf{T} \mathbf{A}^{-}} \right)$$

$$(6.504)$$

In any particular case we may omit "A" and also, for a given system of units, "g". We may also substitute  $\sigma$  for  $\gamma$  and, for normal runway surfaces, omit  $\mu$ . We then have for the ground roll,

$$\mathbf{s}_{\mathbf{s}} = \psi_{1} \left( \begin{array}{c} \underline{\mathbf{B}} \underline{\mathbf{H}} \underline{\mathbf{P}} \, \overline{\mathbf{1}} \overline{\sigma} \\ \mathbf{v} \, \sqrt{\mathbf{v}} \end{array}, \underline{\mathbf{w}}, \quad \underline{\mathbf{v}}, \quad \underline{\mathbf{v}}, \quad \underline{\mathbf{v}} \\ \sqrt{\mathbf{v}} & \mathbf{v} \end{array} \right)$$
(6.505)

and hence

$$\mathbf{s}_{\mathbf{c}} = \mathbf{y}_{1} \left( \begin{array}{c} \underline{\mathbf{MP}} \\ \mathbf{V} \end{array}, \quad \mathbf{W}, \quad \frac{\mathbf{v}_{t_{1}}}{\sqrt{\mathbf{v}}}, \quad \frac{\sigma}{\mathbf{V}} \end{array} \right)$$
(6.506)

eince  $\frac{BHP}{V} \sqrt{V}$  is proportional to  $\frac{BHP}{V}$  at given  $\sigma/V$ 

The term in  $V_{t_1}$  is proportional to  $C_L^{\frac{1}{2}}$  and will therefore be approximately constant for a given level of piloting skill or of risk. (Changes in available  $C_L$  due to change of slipetream intensity may be ignored for moderate ranges of W). If its test range is not large the  $V_{t_1}$  product may be omitted and its effects treated as scatter. Otherwise, it may be necessary to attempt to cross plot in terms of it.

From inspection of equation (6.506) it will be readily appreciated that if the tests can be made at the values of  $\sigma'/V$ , H and BHP/V which correspond to standard conditions the ground roll will be equal to the standard ground roll for the same value of the take-off lift coefficient. No standardization of the observed ground roll is required. Alternatively, test ground rolls for a range of BHP/V and  $\sigma'/V$  might be plotted against these variables and the ground roll at any desired combinations of power, density and gross weight deduced.

A similar consideration of the total distance to height "h" shows that for the particular case of the distance to 50 ft. we may write,

$$\mathbf{s}_{\mathbf{t}} = \Psi_2 \left( \frac{2HP}{V}, \mathbf{I}, \frac{\Psi_{\mathbf{t}_1}}{\sqrt{V}}, \frac{\Psi_{\mathbf{t}_2}}{\sqrt{V}}, \frac{\Psi_{\mathbf{t}_2}}{\sqrt{V}}, \frac{\sigma}{\sqrt{V}} \right)$$
 (6.507)

ATTR 6273

In this case the variable  $\frac{V_{t_1}}{\sqrt{W}}$  will usually be relatively unimportant and may be pmitted. The expression  $V_{t_2}\sqrt{\frac{T}{W}}$  is a function only of the lift

coefficient at 50 ft, and will be assumed constant between test and standard conditions for a given level of skill and risk. As with the ground roll, the expression would be ignored unless its variation from test to test was so large as to make cross plotting desirable.

#### AIRCRAFT WITH TURBO-JET OR MIXED SYSTEMS

A similar approach may be adopted for airplanes with turbo-jet engines or any propulsive system which operates throughout the take-off. If, instead of defining the performance of the propulsive system by the variables BHP. H and  $\rho$ , we write in only the net thrust  $T_n$  we may deduce the more general relations.

$$s_{g} = \psi_{3} \left( \frac{y_{n}}{v}, \frac{v_{1}}{1} \frac{\tau \sigma}{\tau v}, \frac{\sigma}{v} \right)$$
 (6.508)

and 
$$S_t = \Psi_{\downarrow} \left( \frac{F_n}{V}, \frac{V_{t_1} \gamma \sigma}{\gamma V}, \frac{V_{t_2} \gamma \sigma}{\gamma V}, \frac{\sigma}{V} \right)$$
 (6.509)

As with the reciprocating engined airplane the expressions containing  $V_{t_1}$  and  $V_{t_2}$  will usually be ignored and their effects treated as scatter. If, then, the tests can be made at the values of  $P_n/V$  and  $\sigma'/V$  which correspond to standard conditions no further standardization is necessary. Alternatively, a plot of take-off distance against these groups would yield the take-off distances for any desired combination of  $P_n$ , W and  $\sigma'$  within the experimental range of  $P_n/V$  and  $\sigma'/V$ . For example, take-off distances at moderate weight and high altitude may be deduced from tests made at a high gross weight at low altitude (high density) with a suitable thrust.

Alternative relations which may sometimes be more convenient may be obtained by including ambient air pressure and temperature as independent variables instead of air density only. This would normally increase the resultant number of nondimensional groups, but in this case it can be shown that one group is negligible. If we start with the relations.

$$S_{p} = f_{s}(T_{n}, V, P_{n}, T_{n}, V_{t_{1}}, \mu, \epsilon, A)$$
 (6.510)

and 
$$S_t = f_6(F_h, V, P_h, T_h, V_t, V_{t_2}, \mathcal{U}, \mathcal{E}, A, h)$$
 (6.511)

it can be deduced from dimensional analysis that

$$\frac{dS_{g}}{T_{a}} = \phi_{5} \left( \frac{\gamma_{1}}{\Lambda P_{a}}, \frac{\gamma_{1}}{\Lambda P_{a}}, \nabla_{t_{1}} \sqrt{\frac{\Lambda P_{a}}{\Psi T_{a}}}, \mathcal{M}, \frac{\Lambda c^{2}}{T_{a}^{2}} \right)$$
(6.512)

**AFTR 6273** 

$$\frac{\mathbf{s}_{t}}{\mathbf{T}_{a}} = \phi_{6} \left( \frac{\mathbf{F}_{n}}{\mathbf{AP}_{a}}, \frac{\mathbf{V}}{\mathbf{AP}_{a}}, \mathbf{V}_{t_{1}} \sqrt{\frac{\mathbf{AP}_{a}}{\mathbf{VT}_{a}}}, \mathbf{V}_{t_{2}} \sqrt{\frac{\mathbf{AP}_{a}}{\mathbf{VT}_{a}}}, \mu, \frac{\mathbf{Az}^{2}}{\mathbf{T}_{a}^{2}}, \frac{\mathbf{zh}}{\mathbf{T}_{a}} \right) (6.513)$$

As before, we may now omit g, A and  $\mu$ , so deducing the relations

$$\frac{\mathbf{S}_{\mathbf{r}}}{\mathbf{T}_{\mathbf{a}}} = \psi_{5} \begin{pmatrix} \mathbf{Y}_{\mathbf{a}} & \mathbf{W} \\ \mathbf{P}_{\mathbf{a}} & \mathbf{W} \end{pmatrix}, \quad \mathbf{V}_{\mathbf{Y}_{\mathbf{a}}} \begin{pmatrix} \overline{\sigma} \\ \mathbf{W} \end{pmatrix}, \quad \mathbf{T}_{\mathbf{a}} \end{pmatrix}$$
(6.514)

$$\frac{\mathbf{S}_{t}}{\mathbf{T}_{a}} = \mathbf{\gamma}_{6} \left( \frac{\mathbf{T}_{a}}{\mathbf{P}_{a}}, \frac{\mathbf{W}}{\mathbf{P}_{a}}, \mathbf{V}_{t_{1}} \sqrt{\frac{\sigma}{\mathbf{W}}}, \mathbf{V}_{t_{2}} \sqrt{\frac{\sigma}{\mathbf{W}}}, \mathbf{T}_{a}, \frac{\mathbf{h}}{\mathbf{T}_{a}} \right)$$
(6.515)

We thus have the very inconvenient variable  $T_a$  left uncombined with any controllable variable. If a dynamical analysis is made at this point, however, it can be shown with the above choice of variables  $T_a$  may be omitted. (This happens because  $P_a$  and  $T_a$  do not in fact affect take-off entirely independently but only through their ratio  $P_a/T_a$  — i.e., the density. They were introduced as independent variables only to produce the groups  $F_n/P_a$  and  $W/P_a$ ). We thus have,

$$\frac{\mathbf{S}_{\mathbf{f}}}{\mathbf{T}_{\mathbf{a}}} = \Psi_{7} \left( \frac{\mathbf{P}_{\mathbf{n}}}{\mathbf{P}_{\mathbf{a}}}, \frac{\mathbf{V}}{\mathbf{P}_{\mathbf{a}}}, \mathbf{V}_{\mathbf{t}} \sqrt{\frac{\sigma}{\mathbf{V}}} \right)$$
(6.516)

and 
$$\frac{S_{t}}{T_{a}} = \frac{V_{8}}{P_{a}} \left( \frac{Y_{n}}{P_{a}}, \frac{W}{P_{a}}, V_{t_{1}} \sqrt{\frac{T}{W}}, V_{t_{2}} \sqrt{\frac{C}{W}} \right)$$
 (6.517)

where  $S_t$  is the distance required to reach a height equal to the value of  $h/T_a$  under standard conditions, i.e. a height  $h \ge T_{a_s}/T_{a_t}$  where suffices a and t indicate standard and test conditions respectively. These equations are sometimes more convenient as  $F_n/P_a$  is usually independent of air temperature and  $V/P_a$  is more readily controlled than  $\sigma'/V$ .

AFTR 6273

and

#### CHAPTER SEVEN

#### HELICOPTER FLICHT TEST PERFORMANCE AND ANALYSIS

#### SECTION 7.1

#### Introduction

The flight testing techniques and performance data analysis methods for helicopter aircraft are still in the research and development stage. The many peculiarities of helicopter design and flight performance characteristics present difficult analytical problems in both performance standardization and instrumentation requirements. The problems of performance weight correction and accurate low, or zero, speed determination have not been solved in any simplified manner. Although the unstable flying characteristics of the helicopter may be corrected by future development, at present they present a very difficult flying task to the test pilot who seeks to obtain high quality test data.

The ability of the helicopter to move in any direction relative to its three axes presents a major control problem when stabilized flight in a certain direction is required. Although the helicopter can hover motionless at any altitude below the hovering ceiling, the determination of the true hovering condition is not possible without elaborate and ingenious devices for determining zero air speed. It might appear that, because this type aircraft is capable of maintaining stabilized flight between zero and maximum air speed, there would be no problems of stalling as are encountered in conventional aircraft. Actually, because the individual rotor blades are airfoils, a stall condition can occur on the blades individually or as a group, and this stall can be effective in limiting performance at any speed from zero to maximum. Blade stall is an important performance oriterion, as it is accompanied by excessive power requirements and weight limitations at both high and low forward speeds. In addition, at high forward speeds the blade stall is more pronounced in the retreating portion of the rotor disk area. This may result in vibration and excessive control instability in addition to the increased power demands.

The basic aerodynamic analysis for the helicopter is essentially that of the conventional propeller or airscrew. In the helicopter, suitable mechanical devices allow part of the rotor thrust to act as a horizontal component which balances the fuselage drag and propels the aircraft. The remaining rotor thrust sustains the helicopter or provides enough thrust for vertical motion upwards. Part of the total power output of the engine is also used to overcome blade drag and gear losses and to provide directional control. Figure 7.11 illustrates the rotor thrust components.

The many aerodynamic factors involved in the performance of a helicopter rotor or system of rotors have conflicting effects on optimum performance; that which serves best for the hovering condition may detract from the high forward speed performance. A particular aircraft may be the result of many aerodynamic compromises that were made to obtain optimum overall performance. Generally these compromises are made, not in the variable factors that are of interest in flight performance analysis, but in some of the aerodynamic constants such as

AFTR 6273

the rotor configuration, the rotor area, the rotor solidity factor (blade area/ rotor area), and the blade twist or shape. Flight tests may be required to evaluate warious configurations of these items, but they are not flight variables in themselves. As with conventional aircraft, the helicopter performance presentation should include the standard speed-power, rate of climb and descent, range, hovering performance and rotor efficiency for various gross weights and configurations of the particular design being tested.



Figure 7.11 Simplified Sketch of Helicopter Thrust and Lift Components

It should be noted here that theoretical studies and flight tests indicate that the most efficient level flight and hovering performance will be obtained by varying rotor fpm or tip speed. The most efficient rpm at high flight speed is greater than the most efficient rpm at low flight speed and hovering. Actually, it appears that future helicopters may have two rotor gear ratios, one giving a high rotor rpm for high-speed forward flight and one giving a lower rotor rpm at maximum engine power for hovering and low-speed forward flight. This factor will not complicate data reduction or presentation, because for most efficient operation, each of the two rotor rpm may be restricted to separate forward speed ranges and data may be plotted for a constant rotor rpm in each of these ranges.

AFTR 6273

#### SECTION 7.2

#### Level Flight Performance

The helicopter performance parameters are restricted herein to the density altitude method of data reduction and presentation. This is a result of the requirement for a rotor speed variable. If atmospheric pressure and temperature are separated, as is done in the pressure altitude approach, the level flight performance must be represented by four parameters and cannot easily be dealt with graphically.

Extensive analytical methods using the vortex theory and the blade element theory as applied to propellers can be used to determine the many design characteristics for helicopter rotors. The necessary flight performance parameters are more simply derived by the use of dimensional analysis. For the density altitude system the nondimensional functional equation is,

$$BHP = f(W, V_t, V_b, \rho, A)$$
(7.201)

where:

r Anna a state da tatta (1835).

From equation 7.201 two basic sets of parameters are derived by dimensional analysis.

$$\frac{BHP}{A \ g \ v_b^3} = f \left[ \frac{W}{A \ g \ v_b^2}, \frac{V_t}{V_b} \right]$$
(7.202)  

$$\frac{BHP}{(W) \ 3/2} = f \left[ \frac{V_t}{V_t} \left( \frac{P_A}{W} \right)^{\frac{1}{2}}, \frac{V_b}{V_b} \left( \frac{P_A}{W} \right)^{\frac{1}{2}} \right]$$
(7.205)

These parameters are referred to as:

$$\frac{BHP}{A \ g \ V_b^{\ 5}} = C_p, \text{ power coefficient}$$

$$\frac{W}{A \ g \ V_b^{\ 2}} = C_T, \text{ thrust coefficient (a function of the average } C_L \text{ for the individual rotor blades})$$

$$\frac{HP}{W \ 3/2} = power \text{ efficiency parameter}$$

$$\frac{V_t}{V_b} = \mathcal{M}, \text{ tip speed ratio}$$
AFTER 6273 7-3

 $v_t \left(\frac{\mathbf{y}_A}{\mathbf{w}}\right)^{\frac{1}{2}} =$ 

speed parameter

$$v_b \left(\frac{g_A}{W}\right)^{\frac{1}{2}} = \frac{1}{C_T^2} = \text{rotor tip speed parameter}$$

Although BHP is defined as the horsepower delivered to the rotor, the engine BHP is used for performance work since power extraction for directional control does not vary much for given values of the various parameters.

#### COEFFICIENT TYPE PERFORMANCE DATA

A typical plot of equation 7.202 is shown in Figure 7.21. It can be seen in this plot that as the values of CT increase (increasing blade  $C_L$ ) more power is required for a constant rotor tip speed. At a constant  $C_T$  the power required first decreases with forward speed and then increases at the higher forward speeds. This results from a slightly increasing rotor blade drag power with speed, a large reduction in power required to pull air through the rotor (induced power), and a large increase in fuselage drag power required with increasing forward speed. At high forward speeds the retreating rotor blades decrease their relative speed causing the blade  $C_L$  to increase momentarily; this can cause blade tip stall and result in rapidly increasing power requirements at high forward speeds. During hovering and low forward speeds, decreasing rotor tip speeds can mean a reduction in power required at increased values of Cp and  $C_T$ ; however, the extent of rotor tip speed reduction is limited by blade stall and reduction-gear power requirements.



TIP SPEED RATIO, A

Figure 7.21 Typical Plot of the Coefficient Method of Presenting Helicopter Level Flight Performance

AFTR 6273

One of the parameters in equation 7.202 must be held constant if data reduction is to be simplified. At constant rotor speeds, constant  $C_T$  values can be maintained during level flight power calibrations. This is done by using a chart of weight or fuel load vs density altitude for constant  $C_T$  values. This can also be done by flying at a constant value of  $W/d_a$  as described in Section 4.9. Within an ambient temperature range of  $\pm 3^{\circ}$  C the error introduced in  $C_T$ at constant  $W/d_a$  will be only  $\pm 1.0$  percent.

For simplified reduction and presentation of data at a constant rotor rpm, equation 7.202 may be put in a dimensional form.

 $\frac{BEP}{\sigma} = f \left[ \frac{V}{\sigma}, V_{t} \right]$ (7.204)

A typical set of curves is shown in Figure 7.22. This plot gives we level standard performance at a glance, and a complete set of faired curves can be easily converted to  $C_p$ ,  $C_T$ , and  $\mu$  values by use of the necessary sets of constants for the particular aircraft and values of EHP/ $\sigma$ , W/ $\sigma$ , and V<sub>t</sub> from the faired curves.





APPER 6273

If desired the term  $W/\sigma$  may be used in the form  $\sigma W_g/W_t$ . This allows power data for a constant rotor rpm to be directly interpreted in terms of density at standard weight, or weight ratio at sea level density. Faired data in this form can also be converted to  $C_T$  values by use of constants for the particular aircraft.

The dimensional plot of Figure 7.22 for a particular rotor rpm may be converted to that obtainable at another rotor rpm by these equalities:

$$\begin{pmatrix} \underline{\mathbf{B}}\underline{\mathbf{H}}\underline{\mathbf{P}} \\ \overline{\boldsymbol{\sigma}} \end{pmatrix}_{2} = \begin{pmatrix} \underline{\mathbf{B}}\underline{\mathbf{H}}\underline{\mathbf{P}} \\ \overline{\boldsymbol{\sigma}} \end{pmatrix}_{1} \begin{pmatrix} \underline{\mathbf{v}}_{\underline{\mathbf{b}}2} \\ \overline{\mathbf{v}}_{\underline{\mathbf{b}}1} \end{pmatrix}^{3}$$

$$\begin{pmatrix} \frac{\mathbf{W}}{\mathbf{\sigma}} \end{pmatrix}_{2} = \begin{pmatrix} \frac{\mathbf{W}}{\mathbf{\sigma}} \end{pmatrix}_{1} \begin{pmatrix} \frac{\mathbf{v}_{\underline{\mathbf{b}}2}}{\overline{\mathbf{v}}_{\underline{\mathbf{b}}1} \end{pmatrix}^{2}$$

$$\mathbf{v}_{\underline{\mathbf{t}}2} = \mathbf{v}_{\underline{\mathbf{t}}1} \begin{pmatrix} \underline{\mathbf{v}}_{\underline{\mathbf{b}}2} \\ \overline{\mathbf{v}}_{\underline{\mathbf{b}}1} \end{pmatrix}$$

The important factor in these rotor rpm conversions is whether the reduced rotor rpm is also reduced engine rpm or if normal engine rpm and power may be maintained. Any rotor rpm extrapolations should be spot checked by actual flight tests.

The determination of test brake horsepower is easily accomplished if a torquemeter is installed on the engine. Usually this device is not available on helicopters and the engine manufacturer's power chart must be used. In many cases the power charts give very inaccurate results at altitude. If this appears to be the case, manifold pressure may be substituted for brake horsepower in data presentation like that of Figure 7.22. The ordinates would then be MP/or and V<sub>t</sub>. Obviously, manifold pressure should not be substituted in the power coefficient term.

#### POWER, FORWARD SPEED, AND ROTOR SPEED PARAMETER METHOD

For a particular helicopter, equation 7.203 may be put in a dimensional form:

BHP 
$$(\sigma')^{\frac{1}{2}} \left(\frac{W_{3}}{W_{t}}\right)^{3/2} = f \left[ V_{t} \left(\sigma'\right)^{\frac{1}{2}} \left(\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}}, V_{b} \left(\sigma'\right)^{\frac{1}{2}} \left(\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}} \right]$$
(7.205)

where:

#### $W_{g} =$ some standard gross weight

The left side term in this equation is the effective power efficiency corresponding to the inverse of the figure of merit, M, the rotor efficiency. A typical plot of equation 7.205 as in Figure 7.23 shows the relative rotor efficiencies throughout the level flight range. Minimum values of the power

AFTR 6273

term correspond to maximum values of the figure of merit. The effect of reduced rotor rpm is also apparent on this plot; however, as mentioned before, the value of actual reduction of rotor rpm depends on its effect on the power output of the engine.



Figure 7.23 Typical Presentation of Level Flight Power, Forward Speed, and Rotor Speed Data

For  $V_t$  equal to zero (hovering), equation 7.205 resolves into two parameters which are effectively the rotor efficiency, M, and  $(C_T)^2$ . It can be seen from the above plot that M is a function of  $C_T$ . The derivation of the figure of merit is discussed in the next section.

For flight at constant rotor rpm equation 7.205 may be plotted as in Figure 7.24. Here the parameter  $(\mathcal{O}^{W_p/W_t})^{\frac{1}{2}}$  may be interpreted as  $(W_g/W_t)^{\frac{1}{2}}$  for  $\mathcal{O}^{W_t}$  equal to 1.0, or as  $(\mathcal{O}^{-})^{\frac{1}{2}}$  for the standard weight, or as the percent of rated rotor rpm for standard sea level and weight conditions. If data are obtained at one rotor rpm it may be replotted for some other rotor rpm by using the equality:

$$\left(\boldsymbol{\sigma} \cdot \frac{\mathbf{w}_{g}}{\mathbf{w}_{t}}\right)_{2}^{2} = \left(\boldsymbol{\sigma} \cdot \frac{\mathbf{w}_{g}}{\mathbf{w}_{t}}\right)_{1}^{2} \quad \left(\begin{array}{c} \mathbf{v}_{b1} \\ \mathbf{\overline{v}_{b2}} \end{array}\right)$$

For the true speed and altitude ranges encountered by most helicopters, calibrated speed,  $V_0$ , may be substituted for the term  $V_t \sqrt{\sigma}$ . With the parameters of equation 7.205 a constant  $\sigma'/W$  must be held during power calibrations. This is accomplished by the use of a density altitude-weight- $C_T$  chart or a weight-pressure altitude- $W/P_a$  chart as described previously.

AFTR 6273

L



Figure 7.24 Typical Presentation of Level Flight Power, Calibrated Speed, and Density Data

AFTR 6273

# DATA REDUCTION OUTLINE (7.21)

# Speed-Power or Speed-Manifold Pressure Plots for Flight Test Data Obtained at Constant $W/C^-$ ( $C_T$ ) and Constant Rotor Rpm

(1)	Test Point Number		
(2)	H <sub>1</sub>	ft	Indicated pressure altitude
(3)	AHIA	ft	Altimeter instrument correction
(4)	V <sub>1</sub>	mote	Indicated air speed
(5)	Δ <sup>v</sup> ic	knote	Air-speed instrument correction
( 5)		knote	Air-speed position correction
(7)	V <sub>c</sub>	Imote	Calibrated air speed, $(4) + (5) + (6)$
(8)	Δ H <sub>po</sub>	ft	Altimeter position error correction
(9)	H <sub>o</sub>	ft	True pressure altitude, $(2) + (3) + (3)$
(10)		°C	Indicated ambient temperature
(11)	$t_1$ $\Delta t_{1c}$	₽C	Temperature instrument correction
(12)	t <sub>at</sub>	°C −	Test ambient temperature, (10) + (11)
(13)		-	From (9) and (12) and CHART I-1
(14)	-//0 <sup>-</sup>	knots	Test true speed, (7) x (13)
(15)	Vtt	°C	Test carburetor temperature
(16)	tot	٠Č	Standard ambient temperature, from $(9)$
(10)	tas	Ũ	and Table 9.2
().7)	V Tas/Tct		$\sqrt{(16) + 273} + (15) + 273$
(18)	Engine rpm		•
(19)	MP <sub>t</sub>	"Hg	Test manifold pressure
(20)	BEP	•	Chart brake horsepower, from (9) and
• •	J.		(18) and (19)
(21)	BHPt		Test brake horsepower, (20) x (17)
(22)	W. V	1b <i>e</i>	Test aircraft weight
(23)	Wt /o	16 <b>8</b>	$(22) \times (13)^{c}$
(24)	Plot (21) vs (14)	_	
(25)	tos	°c	Standard carburetor air temperature,
• • •			$(\underline{16}) - (\underline{12}) + (\underline{15})$
(26)	VTot/Tos		$\sqrt{(15) + 273} + (25) + 273$
(27)	BEP		Standard brake horsepower, (21)x (26)
(28)	v <sub>te</sub>		Standard day true speed, from (27)
()	· C9		and (24)
(29)	Plot (27) vs (28)	; this is the	standard day speed power curve
	NOTE: If the man	ifold pressur	e-ambient pressure ratio exceeds 1.5, a
	correction	to power and	manifold pressure should be made.
			•

If the chart power data is not reliable, plot (19) vs (14).

# DATA REDUCTION OUTLINE (7.22)

For BHP/or, Vt, W/or Plot; Constant W/or (CT) and Rotor Rpm Test Data

(This is a continuation of Data Reduction Outline 7.21)

(24) BHP/
$$\sigma$$
 (21)  $\equiv$  (13)<sup>2</sup>  
(25) Plot (24) vs (14)

Note: If the chart power data is not reliable, plot  $MP_{t}/O'$  vs  $V_{tt}$ 

DATA REDUCTION OUTLINE (7.23) For BHP  $\sqrt{\sigma}$  (W<sub>s</sub>/W<sub>t</sub>)<sup>3/2</sup>, V<sub>c</sub>  $\sqrt{W_s/W_t}$ ,  $\sqrt{\sigma}W_s/W_t$  Plot; Constant W/ $\sigma$  (C<sub>T</sub>) and Notor Rpm Test Data

(This is a continuation of Data Reduction Outline 7.21)

(24)	W <sub>s</sub> lbs	Selected standard weight
(25)	Wa/Wt	Weight ratio, $(24) + (22)$
(26)	Vws/Wt	$\sqrt{(25)}$
(27)	$(W_{s}/W_{t})^{3/2}$	(25) x (26)
(28)	$BHP \sqrt{\sigma} (W_{c}/W_{t})^{3/2}$	$(21) \times (27) + (13)$
(29)	. Vc <del>/Ws/Wt</del>	(7) x (26)
(30)	Plot (28) vs (29)	
	Note: If the chart power	data is not reliable, MPt may be

substituted for BHPt.

DATA REDUCTION OUTLINE (7.24)

For Cp, CT, & Plot; Constant W/O (CT) Flight Test Data

(This is a continuation of Data Reduction Outline 7.21)

(24)	D		Rotor disk diameter
(24) (25) (26)	A Rotor rpm		Rotor area
(27)	Ψb	ft/sec	Rotor tip speed, 0.0524 x (24) x (26)
(28)	(v <sub>b</sub> ) <sup>2</sup>		(27) <sup>2</sup>
(29)	$(v_b)^3$		(27) <sup>3</sup>
(30)	1/0		(13) <sup>2</sup>
(31)	Cp		Power coefficient, $[(21) \times (30) \times 231,300]$ + $[(25) \times (29)]$
(32)	CT		Thrust coefficient, $(23) \times 421 + (25) \times (28)$
(33)	M		Tip speed ratio, [[14] x 1.69] + (27)
(34)	Plot (31) vs (3	3) for constant	CT values.

**AFTR 6273** 

# SECTION 7.3

## Rotor Thrust, Power, and Efficiency in Hovering Flight

The rotor air flow analysis is similar to that for the conventional propeller. The power required to hover is,

$$= T V_r$$
 (7.301)

where:

Ρ

P = power to the rotor T = thrust of the rotor $V_r = air velocity through the rotor (induced velocity)$ 

The thrust of the rotor is,

$$\mathbf{T} = \mathbf{P} \mathbf{A} \, \mathbf{V}_{\mathbf{r}} \, \mathbf{V}_{\mathbf{d}} \tag{7.302}$$

where:

A = rotor area  $V_d = downstream$  velocity given the air by the rotor

Ey using the actuator disk theory in which  $V_r = \frac{1}{2} V_d$ , the thrust of the rotor may be expressed also as,

 $T = 1/2 \ g \wedge v_d^2$  (7.303)

From 7.302 and 7.303,

$$\nabla_{\mathbf{r}} = \left(\frac{\mathbf{T}}{2 \ \mathbf{g} \ \mathbf{A}}\right)^{\frac{1}{2}} \tag{7.304}$$

From 7.304 and 7.301,

$$P = \frac{T^{3/2}}{(2 \ S \ A)^2} \tag{7.305}$$

The above equation assumes ideal inflow through the rotor disk and no power losses for control or other purposes.

Equation 7.305 is used to define the rotor efficiency, (M), or "figure of merit" as it is usually called.

$$M = \frac{.707 \ T^{3/2}}{550 \ \text{BHP} \ (9 \ \text{A})^2}$$
(7.506)

where:

.707 =  $1/\sqrt{2}$ , to make  $M_{max}$  equal unity BHP = brake horsepower to the rotor T = rotor thrust in hovering flight

AFTR 6275

Or, in terms of density ratio and weight, where the thrust equals the weight supported,

$$M = \frac{0264 \text{ w}^{3/2}}{\text{BHP} (\sigma_A)^{\frac{1}{2}}}$$
(7.307)

The figure of merit is also defined by the rotor thrust coefficient,  $C_{\rm T}$ , and the rotor power coefficient,  $C_{\rm p}$ .

$$M = .707 \quad \frac{c_T^{3/2}}{c_p} \tag{7.308}$$

where:

\_- ≡ 4

· · · · · · · · ·

$$C_{T} = \frac{W}{A \ 9 \ V_{b}^{2}}$$
$$C_{p} = \frac{550 \ BHP}{A \ 9 \ V_{b}^{2}}$$

 $V_{\rm b}$  = rotor tip speed or rpm

As was shown in the previous section, the figure of merit is a function of the thrust coefficient and should be plotted as shown in Figure 7.31.



Figure 7.31 Typical Figure of Merit - Thrust Coefficient Plot

AFTR 6273

Equation 7.308 may be used to show graphically the figure of merit as in Figure 7.32.



Figure 7.32 Typical Figure of Marit Comparison in Terms of Thrust and Power Coefficients

This plot presumes a constant slope of M with Cr. This will be the case if the Cr range is limited. The same plot is shown in dimensional form for a particular helicopter and a constant rotor rpm in Figure 7.33. Data in this plot may be extrapolated to other rotor rpm by the equalities derived from equation 7.202,

$$\left(\frac{\text{BHP}}{\sigma}\right)_2 = \left(\frac{\text{BHP}}{\sigma}\right)_1 \left(\frac{v_{b2}}{v_{b1}}\right)^5, \quad \left(\frac{w_{t}/w_{g}}{\sigma}\right)_2^{3/2} = \left(\frac{w_{t}/w_{g}}{\sigma}\right)_1^{3/2} \left(\frac{v_{b2}}{v_{b1}}\right)^5$$

In Figure 7.34 is shown a method of plotting an effective figure of merit vs an effective rotor speed.

From Figure 7.34 the most efficient combinations of rotor speed and weight for a given BHP or available can be determined. In this manner the hovering omiling can also be determined for a particular set of conditions. Without gear shifting arrangements, a reduced rotor tip speed results in a reduced power. This must be included if data are extrapolated to lower rotor tip speeds. For single gear ratio helicopters an increased hovering performance (increased maximum payload) for decreased rotor speed is found to exist only at very low altitudes; and the hovering ceiling will be decreased materially at reduced rotor rpm.

AFTER 6273



Figure 7.33 Typical Effective Figure of Merit Comparison in Terms of Power, Weight, and Density



Figure 7.34 Typical Effective Figure of Merit Plut in Terms of Power, Rotor Speed, and Density and Weight

AFTER 6273

.

Where maximum power is available in hovering, even when rotor speed is reduced by gearing, the weight limitations for any altitude and the hovering ceiling may be determined by plotting the effective figure of merit vs the effective weight ratio as in Figure 7.35. In this plot the dotted lines show the original data reduced to constant weights and density altitudes.



# Figure 7.35 Typical Effective Figure of Merit Plot. in Terms of Power, Rotor Speed, Density and Weight

When the helicopter is hovering above the ground at a height equivalent to about one rotor diameter a positive thrust increment is developed by the pressure field between the ground and the rotor. This so called "ground effect" has a noticeable influence on the take-off performance and acts as a cushion during landings. The ground effect actually increases the rotor efficiency relative to the rotor efficiency obtained when the aircraft is out of the ground effect. This relative efficiency increase can be shown by either of the plots illustrated in Figure 7.36.

If the slope of the rotor efficiency curve is nearly zero with respect to thrust coefficient or effective rotor speed, the ground effect hovering may be plotted as in Figure 7.37(a); if the slope is other than nearly zero a plot such as 7.27(b) must be used.

Determination of the true hovering condition is not easily done by use of present air-speed indicators. Near the ground a good reference for the pilot is the ground itself, but the tests must be conducted during low or zero wind conditions. At altitude or in appreciable winds the hovering condition can be determined by use of a long weighted cord attached to the fuselage. When the weighted cord hangs straight down from the helicopter, the aircraft is stationary with respect to the air mass in which it is flying. The weighted cord may be indexed to indicate the true hovering height during low altitude tests.

AFTR 6273



Figure 7.36 Method of Showing Ground Effect, on Hovering Performance





AFTR 6273

# DATA REDUCTION OUTLINE (7.31)

For EHP  $\mathbf{Tor} (W_g/W_t)^{3/2}$  vs  $V_b \mathbf{Tor} W_g/W_t$  Plot, and M vs C<sub>T</sub> Plot; Constant or Variable Rotor Rpm Test Data

(1)	Test Point Number	r	
(2)	Hi	ft	Indicated pressure altitude
(3)		ft	Altimeter instrument correction
(4)		ft	Altimater position error
(5)	H <sub>o</sub> f	ft	True pressure altitude, $(2) + (3) + (4)$
	NOTE: If a tape:	line is used t	to record height above the ground for low
	level wor	k it may be u	sed with the ground pressure altitude to
	obtain tr	ne pressure al	ltitude.
(6)		•C	Indicated ambient temperature
(7)	$\Delta t_{1c}$	•C	Temperature instrument correction
(8)	tat 1/10	•C	Test ambient temperature, $(6) + (7)$
(9)			From (8) and (5) and CHART I-1
(10)	5	•	From (8) and (5) and CHART I-1
(11)	tct	●C	Test carburator temperature
(12)	<sup>t,</sup> as	•c	Standard ambient temperature, from (5)
(13)	Tas/Tot		and Table 9.2 (12) + 273 + (11) + 273
(14)	Engine rpm		
(15)		# <b>T</b> _	
(16)	MPt BHPa	"Hg	Test manifold pressure
(10)	TOT 0		Chart brake horsepower, from (5) and (14) and (15)
(17)	BHPt		Test brake horsepower, (16) x (13)
(18)	W <sub>t</sub>	lbs	Test aircraft weight
(19)	A. 3/2	200	(18) <sup>3/2</sup>
	$(W_t)^{3/2}$		
(20)	W.	lbe	Standard aircraft weight
(21)	W <sub>e</sub> /W <sub>t</sub>		$(20) \div (10)$
(22)	$1 \frac{W_{e}}{W_{t}}$		121_1
(23)	$(W_{\rm g}/W_{\rm t})^{3/2}$		$(21)^{3/2}$
(24)	Rotor rpm		• •
(25)	D	<b>#1</b> :	Rotor disk diameter
(26)	Ă	ft ft <sup>2</sup>	Rotor disk area
(27)		ft/sec	Rotor tip speed, 0.0524 x (24) x (25)
(28)	v <sub>b</sub> 2	,	(27) <sup>2</sup>
(29)	BHP, 10 (Wg/Wt)3/	/2	$(17) \times (23) + (9)$
(30)	Vhlo W./Wt		
• •		N N	$[(27) \text{ or } (24)] \times (22) + (9)$
(31) (32)	Plot (29) vs (30) M	)	
()=)	<b>F</b> 1		Figure of merit, $[0.0298 \times (9) \times (19)] + [(17) - (25)]$
(33)	ĈŢ		$[(17) \times (25)]$
1221	AL.		Thrust coefficient, [421 x (18)] + [(10) x (26) x (28)]

AFTR 6273

(34) Plot (32) vs (33)

on on the spatian ∈

. . .

NOTE: If the manifold pressure ambient pressure ratio exceeds 1.4, a correction to power and manifold pressure should be made.

If the chart power data is not reliable, substitute  $MP_t$  for  $BHP_t$  in (29) and do not calculate M and  $C_T$ .

AFTR 6273

#### SECTION 7.4

# Climbs and Descents (Autorotation)

Two types of climbs must be evaluated in helicopter performance; the vertical climb and the climb at the forward speed for best climb. Only one type of descent is usually evaluated. That is the autorotational or power-off descent.

The speed for best rate of climb and minimum rate of descent may be determined by the sawtooth climb procedures used for conventional aircraft. In making climb tests at the low rates of climb and forward speeds associated with helicopters, special care must be taken to obtain data during the best atmospheric conditions; that is, negligible wind and turbulence and no temperature inversions. Since weight corrections to climb and descent data cannot be accurately determined by mathematical derivations, it is best to take the first sawtooth points at the desired weight and at the best climbing speed from the manufacturer's data. Sawtooth climb and descent data are reduced to standard conditions by the procedures uged for conventional aircraft and are presented as shown in Figure 7.41.



Figure 7.41 Typical Savtooth Climb and Descent (Autorotation) Data

AFTR 6273

#### RATE OF CLIMB EVALUATION

For helicopters the over available in level flight is constant at a particular altitude. For this reason it is not necessary to conduct sawtooth climbs. At the level flight speed for minimum power required the maximum excess power for climb is available. This speed, corresponding to minimum  $C_p$  and BHP 10 ( $W_g/W_t$ )<sup>3/2</sup> in Figures 7.32 and 7.34 respectively, may be determined from the level flight speed power performance for any values of  $C_T$ , (W/o),  $V_b$  O  $W_t/W_s$ , or O  $W_g/W_t$ ) as shown in these figures.

To equation 7.301 may be added another variable, the vertical velocity  $(V_v)$ . Using the new equation and dimensional analysis, the fcllowing equations may be obtained:

$$\frac{BHP}{A \rho v_b^3} = f \left[ \frac{W}{A \rho v_b^2}, \frac{v_t}{v_b}, \frac{v_v}{v_b} \right]$$
(7.401)

$$\frac{\text{BHP}(A \mathcal{P})^{\frac{1}{2}}}{(W)^{3/2}} = f \left[ V_{t} \left( \mathcal{P}_{M} \right)^{\frac{1}{2}}, V_{b} \left( \mathcal{P}_{M} \right)^{\frac{1}{2}}, \nabla_{v} \left( \mathcal{P}_{M} \right)^{\frac{1}{2}} \right]$$
(7.402)

The value of the forward-speed parameter for best rate of climb in both of these equations may be determined by inspection of the level flight speed-power data. If it is assumed that, for a particular set of conditions, the rate of climb varies nearly linearly with the power available, then the above equations may be evaluated graphically for all climb conditions by using the data from two check climbs at best climb speed. Equation 7.402 is the easier of the two to work with in this respect and will be used in this discussion.

In a dimensional form for a particular aircraft the speed parameter is:

 $v_t \sqrt{\sigma' \frac{W_{\theta}}{W_t}} = v_c \sqrt{\frac{W_{\theta}}{W_t}}$ 

During a check climb to the actual altitude the take-off value of  $W_g/W_t$  will not change appreciably and the best  $V_c$ -altitude schedule from the speed-power data may be computed at a constant  $W_g/W_t$ . Two check climbs at best forward speed are now required. These climbs may be accomplished at two power settings for constant  $W_g/W_t$  or at two extreme values of  $W_g/W_t$  for a constant power setting. Data from these check climbs are not reduced to standard conditions. Instead the calibration-corrected data are plotted as in Figure 7.42.

AFTR 6273



Figure 7.42 Method of Determining Two Climb and Power Parameter Values for a Constant Effective Weight Ratio

This plot establishes two power-parameter values and two rate of climb parameter values for any value of the rotor rpm or effective weight ratio parameters. This data are then crossplotted as in Figure 7.43 to show the variation of the power and rate of climb parameters at constant rotor rpm or effective weight ratio parameters. From this plot the rate of climb can be determined for any weight, altitude, or power conditions.

The vertical climb case is identical to that described for the climb at best speed. Here the forward speed is zero, but the technique of determining rate of climb for all conditions is accomplished as described above. In vertical climbs the primary source of error is in determining and maintaining zero velocity relative to the air mass. This problem is partially solved by the use of a long weighted cord having short ribbons attached to it. Keeping the cord straight and the ribbons hangir ' down assists in approximating zero forward speed.

AFTR 6273





#### AUTOROTATIONAL DESCENT EVALUATION

Retor operation without engine power is referred to as autorotation. Under certain conditions the helicopter may descend and land safely without engine power. Since, during autorotation, the air flów through the rotor is opposite to the flow during level flight power-on conditions, there is some instability and loss of altitude during the transition to autorotation and a minimum rate of descent. The helicopter performance investigation should determine altitudes and speeds at which autorotation can be assumed to result in a safe landing. The forward speed for minimum rate of descent and the effects of weight and altitude on autorotation should also be determined.

Equation 7.402 applies to the autorotative descent. In this case the power parameter is zero and the value of  $V_{\psi}$  is negative. The dimensional parameters are:

$$v_v \left( \mathbf{\sigma} \quad \frac{W_B}{W_t} \right)^{\frac{1}{2}} = f \left[ \begin{array}{c} v_t \left( \mathbf{\sigma} \quad \frac{W_B}{W_t} \right)^{\frac{1}{2}}, \quad v_b \left( \mathbf{\sigma} \quad \frac{W_B}{W_t} \right)^{\frac{1}{2}} \right]$$
 (7.403)

The maximum rotor efficiency in level flight is represented by the minimum value of  $C_p$  or BHP  $\sqrt{O}$   $(W_g/W_{\pm})^{3/2}$  for level flight. This condition represents the least power required relative to weight, forward velocity, fuselage drag, and rotor blade drag. In autorotation the total power absorbed by the helicopter is:

AFTR 6273
$$\begin{array}{rcl} P_{absorbed} &=& -V_{v} & W \\ & & & & \\ & & & V_{v} &=& P_{induced} & + & P_{rotor drag} & + & P_{fuse lage drag} \\ P_{absorbed} &=& minimum & at & -V_{v} &= minimum \end{array}$$

In autorotation the blade pitch angle is small and the angle of attack is large relative to level flight conditions. It may be assumed from drag and lift coefficient vs angle of attack data that  $P_1$ ,  $P_{rd}$ , and  $P_{fd}$  are nearly proportional to level flight values at the same forward speeds, rotor speeds, and weight. Thus the forward speed for least power required in level flight is the speed for least power absorbed and minimum rate of descent in autorotation. In fact by using the standard rate of climb equation and assuming the same rotor efficiency:

if P<sub>avail</sub> =

$$V_v$$
 ft/min = 550 (60) BHP<sub>level flight</sub>

Equation 7.404 will give a close approximation of autorotation rate of descent at any forward speed.

(7.404)

An autorotative descent at best forward speed determined from the level flight speed power performance will establish the minimum rate of descent for all conditions of weight, rotor speed, and altitude. Since the weight during descent does not change, the best descent speed will be at a nearly constant  $\nabla_c$  for an initial weight, and will increase slightly if the descent is started at the helicopter ceiling.

 $v_t \int \overline{U}_{H_t} = v_0 \int \overline{W_{H_t}}$ 

A descent should be conducted at a  $V_0$  higher and lower than the assumed best  $V_0$  to establish the magnitude of variation of rate of descent with small variation of  $V_c$ . Figure 7.44 shows a typical plot of data for equation 7.403.

#### POWER-OFF LANDINGS

With power on, a safe landing may usually be executed vertically. In autorotation a minimum rate of descent is in the middle speed range of the aircraft and the safest landings involve some ground roll if the terrain is suitable. The power-off descent is made at minimum blade pitch to provide minimum blade drag and maximum rotor speed. Within a rotor diameter of the ground this pitch angle may be rapidly increased and the rate of descent lowered considerably for a short interval. This reduces the forward speed considerably and upon touchdown a minimum ground roll will result.

AFTR 6273



Figure 7.44 Method of Plotting Autorotation Descent Data

A safe altitude for entering into autorotation may be defined as the height above the ground at which entry into autorotation will result in a minimum rate of descent at a height of one rotor diameter above the ground. This safe altitude may be plotted as a function of the forward speed at which entry into autorotation is started as in Figure 7.45. Data should be obtained as near the ground as possible without actually making a landing; a 2000 ft altitude will give desirable results if continuous engine operation is assured.

Autorotative landings over a simulated 50 ft. obstacle should be conducted at the speed for minimum rate of descent. This data will establish the approximate air distance and ground distance required for safe power-off landings. These landing data are plotted as in Figure 7.46. Since the technique of making this type of landing is not always consistent and the distances are so short, it is not feasible to apply any standardization corrections.

AFER 6273



Figure 7.45 Nethod of Showing Safe Height for Entry into Autorotation



Figure 7.46 Method of Showing Autorotative Landing Time and Distance Data

AFTE 6275

.

#### SECTION 7.5

### Fuel Consumption, Endurance and Range

Fuel consumption, range, and endurance data for helicopters powered by reciprocating engines are handled in about the same manner as for conventional aircraft. One exception to the theory and technique of Chapter Four is the inclusion of the rotor tip speed parameter in the endurance and range equations. These equations are derived from equation 7.301, substituting specific endurance, SE, for BHP, in one case and specific range, SRg, for BHP, in the other case. The two dimensional equations that result are:

$$\operatorname{SE}\left[\overline{\sigma}\left(\frac{W_{t}}{W_{g}}\right)^{3/2} = \frac{\left(\overline{\sigma}^{*}\right)^{1/2}}{W_{f}} \left(\frac{W_{t}}{W_{g}}\right)^{3/2} = f\left[\operatorname{V}_{t}\left(\overline{\sigma}^{*}\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}}, \operatorname{V}_{b}\left(\overline{\sigma}^{*}\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}}\right] (7.501)$$

$$\operatorname{SRg}\left(\frac{W_{t}}{W_{g}}\right) = \frac{V_{t}}{W_{f}} \quad \frac{W_{t}}{W_{g}} = f\left[\operatorname{V}_{t}\left(\overline{\sigma}^{*}\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}}, \operatorname{V}_{b}\left(\overline{\sigma}^{*}\frac{W_{g}}{W_{t}}\right)^{\frac{1}{2}}\right] (7.502)$$

where:

$$W_f = fuel flow (lbs/hr)$$
  
 $V_t \overline{10} = V_c$ 

For each speed-power calibration a plot should be made of fuel flow versus brake horsepower. This plot is valid for both test and standard conditions at the approximate density altitude of the flight. Some altitude effects are usually noticeable as shown in Figure 7.51. It is essential to determine fuel consumption, range and endurance data at typical flight altitudes, since the exact effects of altitude on these variables cannot be determined by extrapolation except in a narrow range near the altitude flown.

#### ENDURANCE

The forward speed for maximum endurance is found from the speed-power calibrations at the point where  $C_p$  or BHP  $\sqrt{\sigma} (W_g/W_t)^{3/2}$  is a minimum. This speed corresponds to minimum power required for the rotor parameter conditions existing. The effects of weight, altitude, rotor speed, and engine speed on SE may be evaluated at the best forward speed as illustrated in Figure 7.52. At a middle altitude the specific endurance parameter is determined at about four values of the rotor speed parameter at two engine rpm's (if reduced rotor rpm indicates reduced power-required on speed power plots). At a high and a low altitude the specific endurance parameter is found for only two extreme values of the rotor speed parameter and a curve is faired between them corresponding to that found for the complete survey at the middle altitude.

AFTR 6273



Figure 7.51 Typical Fuel Flow-Power Presentation



Figure 7.52 Typical Plot of Maximum Endurance Inta

AFTR 6273

Although endurance data, as such, is not usually presented in the aircraft performance report, the requirement may arise for presentation of endurance at all forward speeds as well as at the speed for maximum endurance. This may be accomplished by showing the endurance parameters at three typical altitudes for high and low values of the rotor speed parameters as in Figure 7.53. Endurance during the hovering condition should be separately evaluated if such data is required.



Figure 7.53 Typical Rodurance Parameter Data for all Forward Speede

# RANCE

The maximum range for helicopters is found at relatively high forward speeds for the CT or rotor speed parameter involved. In most cases some altitude effects are apparent in the fuel flow vs BHP plot and equation 7.502 does not strictly apply. If the fuel flow-BHP plots are identical at all altitudes, equation 7.502 may be plotted as indicated in Figure 7.54. One of the rotor speed parameters should be flown at a reduced engine rpm to establish any relative improvement in specific range for this condition. In general, for a given gross weight, a reduced engine speed resulting in a reduced power and forward speed does not result in any appreciable increase in range.

Altitude is usually an important factor in the specific range equations. Data should be obtained at about three representative altitudes and plotted for constant rotor rpm as in Figure 7.55.

AFTER 6273



Figure 7.54 Typical Range Parameter Data for All Forward Speeds (Fuel Flow - BHP Plots Identical at All Altitudes)

Two possible extremes of the rotor speeds parameter are obtained at each of the altitudes to permit extrapolation for this condition. If desired a range comparison for reduced engine speed may be added to Figure 7.55 as was done in Figure 7.54. When a complete range evaluation in parameter form is not considered necessary, data may be simply presented as specific range vs true speed as shown in Figure 7.56. Data for this plot, as for other types of range plots, should be obtained at various flight altitudes.

AFTR 6275



Figure 7.55 Typical Range Parameter Data for All Forward Speeds



Figure 7.56 Typical Specific Range-True Speed Plot, One Altitude, Weight, and Botor Rpm

AFTR 6273

# SECTION 7.6

# Air Speed, Altimeter, and Temperature System Calibrations

Separate air-speed indicator and altimeter calibrations should be made with helicopters, because of possible errors in the total pressure at the air-speed probe. No conversion of dV to dH should be attempted at large forward speeds. At low speeds the total effects of air-speed error  $(dV_G = \pm 15 \text{ knots})$  if converted to altimeter error would be negligible. A pacer aircraft may be used to determine altimeter and air-speed position errors at high speeds. At low speeds a take-off time and distance recording camera can be used. These calibration tests should always be flown out of ground effect. Data are reduced as described in Chapter One. The method of plotting the air-speed position error is illustrated in Figure 7.61. Altimeter position error is plotted in the customary manner, and data are usually obtained out of ground effect.



Figure 7.61 Typical Helicopter Air-speed Position Correction Plot

Because of the low true speeds of helicopters, adiabatic temperature rise is negligible. The temperature probe should be carefully shielded from engine exhaust heat and solar radiation.

AFTR 6273

# REFERENCES

# GENERAL

- 1. Anonymous, "Ground School Notes Book B Performance", The Empire Test Pilots' School, Farnborough, Ministry of Aviation, United Kingdon.
- King, J. J. et al, "Aerodynamics Handbook for Performance Flight Testing - Volume I", USAF Experimental Flight Test Pilot School, AFFTC-TN-60-28, July 1960
- Perkins, C. D. and Dommasch, D. O., eds. "Flight Test Manual- Volume I - Performance", Advisory Group for Aeronautical Research and Development, 1959.
- 4. Polve, J. H. "Performance Flight Testing", USAF Experimental Flight Test Pilot School FTC-TR-53-5, August 1953
- 5. Spillers, L. H. et al, "Pilots Handbook for Performance Flight Testing", USAF Experimental Flight Test Pilot School, FTC-TN-59-46, September 1960

# STANDARD ATMOSPHERE

- Anonymous, "Standard Atmosphere Tables and Data for Altitudes to 65,800 Feet", International Civil Aviation Organization and Langley Aeronautical Laboratory NACA Report 1235, 1955
- Minzner, R. A. et al, "U.S. Extension to the ICAO Standard Atmosphere - Tables and Data to 300 Standard Geopotential Kilometers", United States Committee on Extension to the Standard Atmosphere, U.S. Government Printing Office, 1958
- 8. Minzner, R. A. et al, "The ARDC Model Atmosphere 1959", Air Force Cambridge Research Center TR-59-267, ASTIA Document 229482, August 1959

# RECIPROCATING ENGINE AIRCRAFT PERFORMANCE

- Bikle, P. F., "A Simplified Manifold Pressure Correction", Army Air Forces, Hq. Air Technical Services Command, Wright Field, Dayton, Ohio, Flight Section Memorandum Report Serial No. TSCEP5E-1919, 29 June 1945
- Bikle, P. F., "Performance Flight Testing Methods in Use by the Flight Section", USAF Air Materiel Command, Wright-Patterson AFB, Ohio, Army Air Forces Technical Report No. 5069, 15 January 1944

# TURBOJET ENGINE AIRCRAFT PERFORMANCE

- Shoemacher, P. E. and Schonewald, R. L., "Reduction and Presentation of Flight Test Data for the F-86 Airplane having Automatic Engine Control and a Continuously Variable Jet Nozzle Area", AFFTC Technical Note R-8, July 1952
- 12. O'Neal, R. L., "Performance Standardization for a Turbojet Engine Equipped with a Variable Area Nozzle Controlled by Engine Speed", AFFTC Technical Memorandum 56-7, March 1956

#### POWER PLANTS

그는 일구의 것은 것 한 뒤나는지, 나는

- Chapel, C. E. et al, "Aircraft Power Plants", Northrup Aeronautical Institute Series of Aviation Texts, McGraw-Hill, New York, 1955
- 14. Anonymous, "The Aircraft Engine and Its Operation", Pratt and Whitney Aircraft Operating Instruction 100, February 1955
- Gagg, R. F. and Farrar, E. V., "Altitude Performance of Aircraft Engines Equipped with Gear-Driven Superchargers", S. A. E. Journal 34:217-225, June 1934
- 16. Pierce, E. F., "Altitude and the Aircraft Engine", SAE Journal 47:421-431, October 1940
- 17. Anonymous, "Performance Correction Procedures for Turbojet and Turbofan Commercial Engines", Pratt and Whitney Aircraft Operating Instruction 204, May 1961
- Anonymous, "General Operating Instructions, Axial Compressor Nonaíterburning Turbojet and Turbofan Engines", Pratt and Whitney Aircraft Operating Instruction 190, April 1963
- <sup>1</sup>9. Anonymous, "General Operating Instructions, Axial Compressor Afterburning Turbojet and Turbofan Engines", Pratt and Whitney Aircraft Operating Instruction 191, November 1963
- Anonymous, "Aircraft Performance Engineer's Manual for B-36 Aircraft Engine Operation", Strategic Air Command Manual No. 50-35, 1952

#### LAG CORRECTION

21. Irwin, K. S., "Lag in Aircraft Altitude Measuring Systems", Air Force Flight Test Center FTC-TDR-63-26, 1963

### CLIMB PERFORMANCE

- 22. Grosso, V. A., "Analytical Investigation of the Effects of Vertical Wind Gradients on High Performance Aircraft", AFFTC Technical Report 61-11, March 1901
- 23. Shoemacher, P. E., "Use of Corrected Non-Dimensional Parameters for Standardizing Jet Aircraft Rate-of-Climb Data", AFFTC Technical Report 52-13, May 1953
- 24. Grosso, V. A., "Investigation to Determine Effects of Several Parameters on Optimum Zoom Climb Performance", AFFTC Technical Note 59-32, October 1959
- 25. Grosso V. A., "F-106A Zoom Climb Study", AFFTC Technical Information Memorandum 58-1047, October 1958

# TAKE-OFF PERFORMANCE

- 26. Lush, K. J., 'Standardization of Take-Off Performance Measurements for Airplanes'', AFFTC FTC-TN-R12
- 27. Dunlap, E. W., "Corrections to Take-Off Data", AFFTC Technical Information Memorandum 59-1078, February 1959

#### HELICOPTER PERFORMANCE

- 28. Anonymous, "A Review of Current Helicopter Technology", Vertol Division of Boeing Co., Morton, Penna., Lectures to Graduating Class of Naval Test Pilot School, Patuxent Naval Air Station, February 1963
- Swope, W. A., "Lectures on Rotary Wing Performance",
   U. S. Army Transportation. Materiel Command, Aviation Test Office, Edwards AF Base, California, Lectures to USAF Experimental Test Pilot School, Spring 1960.
- 30. Wheelock, R. H., "An Introduction to the Helicopter", Bell Helicopter Co., Fort Worth, Texas, 1962



Preceding page blank



CHART I -1

AFTR-6273

















8 - 1 3







n junger og





AFTR 6273

8-17

CHART I - 2



8-18

AFTR 6273



CHART I - 4

M vs Pt / P - ISENTROPIC FLOW



CHART I - 4

AFTR 6273



CHART 1- 4





K VE PL /PR - ISENTROPIC FLOW





CHART I - 4



CHART I- 4

AFTR 6273

<sup>8-24</sup> 



NHAT 🛨 🗕 🗄





M VE P. / P. - NORMAL SHOCK CONDITIONS -5-1.4

CHART I - 5

CHART I - 5



M VS P. / P. - NORMAL SHOCK CONDITIONS- 0=1.4

1.0  $\frac{P_{t}}{P_{t}} = \begin{bmatrix} (\delta+1)u^{2} \\ (\delta-1)u^{2} + 2 \end{bmatrix}^{\frac{1}{3-1}} \begin{bmatrix} \frac{1}{3+1} \\ \frac{1}{23u^{2}} - (\delta-1) \end{bmatrix}^{\frac{1}{3-1}}$ .9 - Conditions Downstream of the Shock  $(\mathbf{1})$ .8 .1 •7 TOTAL PRESSURE RATIO, Pt / Pt .6 ...5 .4 .3 -2 2.5 MACH NUMBER, 1.0 1.5 2.0 3.0 3.5 4.0 M





8-27

CHARTI - 5



CHART I - 5

AFTR 6273

8-28


CHART I + 3





CHART I - 5



128년만 🖫 🗕 주

W ve W' - NORMAL SHOCK CONDITIONS - 8-1.4



CHART T - 5

AFTR 6273



AFTR 6273

CHART I - 5







AFTER 6273

PSYCHROMETRIC CHART



AFTR-6273

# APPENDIX II

### NOMENCLATURE

		CHART II
A	Area	n <sup>2</sup>
<b>A</b> 8	Jet Nozzle Area	n <sup>2</sup>
R	Aspect Ratio, b <sup>2</sup> /S, b/c	
a	Velocity of Sound, VygRT	38.944 V K Knote
a <sub>o</sub>	Velocity of Sound (Sea Level Standard)	761.06 mph 660.9 knots 1116 ft/soc
BHP	Brake Horsepower	550 ft-lbs/sec
BHPas	Brake Horsepower (Available Standard)	550 ft-1be/eeo
BHPat	Brake Horsepower (Available Test)	550 ft-lbs/sec
BHPc	Brake Horsepower (Churt)	550 ft-lbe/sec
BHP	Brake Horsepower (Standard)	550 ft-1bs/sec
BHPt	Brake Horsepower (Test)	550 ft-1be/seo
BHPtos	Brake Horsepower (Test at Standard Day Carburetor Temperature)	550 ft-1bs/sec
BSPC	Brake Specific Fuel Compution	lbs/bhp-hr
C	Wing Chord	ft
•C	Degrees Centigrade	
cD	Drag Coefficient, $C_D = \frac{2D}{\rho V^2 S} = \frac{D}{.7 P_g M^2 S} = \frac{D}{.9 S}$	$= \frac{294.6D}{V_e^2(100000)8}$
с <sup>рі</sup>	Induced Drag Coefficient, $\frac{C_L^2}{\pi/R_e}$	
с <sub>рр</sub>	Profile Drag Coefficient, $C_D - C_{D1}$	-
C <sub>L</sub>	Lift Coefficient, $C_{L} = \frac{2V}{\sqrt{28}} = \frac{V}{\sqrt{7} P_{B} M^2 S} = \frac{1}{\sqrt{28}}$	$\frac{1}{18} = \frac{294.64}{V_e^2 (inote)^8}$
CAT	Carburetor Air Temperature	°C or °F
ATTR 627	· · · · · ·	

AFTS 6273

313-15 A 37 do - 51

.

CHART TI-1

÷

8-35 Preceding page blank

CHART	II-l
-------	------

• •

· · · · · · · -.

ROMENCLATURE

ם	Aerodynamic Drag	lbs
₫Bp	Altimeter Position Error, H <sub>ic</sub> - H <sub>c</sub>	ft
dBpc	Altimeter Position Error Correction, H <sub>c</sub> - H <sub>ic</sub>	ft
dм <sub>р</sub>	Mach Meter Position Error, Mic - M	
ам <sub>рс</sub>	Mach Meter Position Error Correction, M - Mic	
dPp	Static Pressure Position Error, P <sub>5</sub> - P <sub>8</sub>	"Eg
dPpc	Static Pressure Position Correction, $P_a - P_s$	"Eg
d۷p	Air-speed Meter Position Error, V <sub>ic</sub> - V <sub>c</sub>	knots
₫⊽₽٥	Air-speed Meter Position Error Correction, V <sub>c</sub> - V <sub>ic</sub>	knots
дн	Pressure Altitude Increment	ft
đh	Tapeline Altitude Increment	ft
(dH <sub>o</sub> /dt)	Rate of Climb (Altimeter)	ft/min
( <b>dH/</b> dV)	Altimater-Air-speed Position Error Correction Ratio	ft/knot
(dm/dh)	Mach Number-Altimeter Position Error Correction Ratio	1/ft
(4M/dV)	Mach Number -Air-speed Position Error Correction Ratio	l/knot
(dv/dh)	Vertical Wind Gradient	knots/1000 ft
(dv <sub>c</sub> /dh <sub>c</sub> )	Air-speed-Altitude Change Ratio During Climb	knots/1000 ft
( <b>d</b> h/dt)	Rate of Climb (Tapeline)	ft/min
( <b>d</b> h/dt) <sub>B</sub>	Rate of Climb (Tapeline, Accelerating Climb Speed)	ft/min
(dq <sub>c</sub> /dV <sub>c</sub> )	Impact Pressure-Calibrated Speed Change Ratio	"Hg/knot
	Total Energy	foot-lbs
KBP	Exhaust Back Pressure	"Hg
RBP1	Exhaust Back Pressure (Indicated)	"Eg

CHART II-1

.....

AFTR-6275

8-36

.

HONGENCLATURE

KBP.	Exhaust Back Pressure (Standard)	"Eg
EBPt	Exhaust Back Pressure (Test)	"Eg
e	Partial Pressure of Water Vepor	"Hg
•	Airplane Efficiency Factor, $\frac{C_L^2}{C_{D1}\pi R}$	
<b>T</b>	Force or Thrust	lbe
• F	Degrees Fahrenheit	
¥.	Entrance Momentum per Second	lbe
Feff	Thrust (Effective)	lbs
Feffe	Thrust (Effective Standard)	10.
Fefft	Thrust (Effective Test)	100
r <sub>g</sub>	Thrust (Gross)	100
<b>7</b> <u>n</u>	Thrust (Not)	lbe
· 7 <sub>336</sub>	Thrust (Net Available)	1 <b>0</b> 0
Y <sub>10</sub>	Thrust (Net Excess)	150
Far	Thrust (Net Required)	1 <b>b</b> s
<b>P</b> p	Thrust (Pressure)	100
T,	Thrust (Velocity)	1 <b>0</b> 5
E	Acceleration of Gravity	32,174 ft/eeo <sup>2</sup>
H <sub>c</sub>	Pressure Altitude (True), $E_{10} + \Delta E_{p0}$	ft
Hi	Pressure Altitude (Indicated)	ft
Hic	Pressure Altitude (Indicated Instrument Corrected), $H_1 + \Delta H_{10}$	ft
∆ L <sub>10</sub>	Altimater Instrument Correction for H <sub>1</sub>	n
∆ E <sub>f</sub>	Ing Correction to Altimeter	ft
	· · ·	

AFTR 6273

8-37

CHART II-1

CHART II-1	NONGNCIATURE	
۵Hpg	Position Error Correction to Altimater	ft
"B20	Inches of Water	
"Hg	Inches of Mercury	
h	Tapeline Altitude	<b>1</b>
IAS	Indicated Air Speed	knots or agk
in	Inches	
X	Total Temperature Recovery Factor	
•K	Degrees Kelvin	
r <sup>e</sup>	Jet Nozzle Gas Flow Calibration Factor	
	$\frac{K_{g}}{W_{g}} = \frac{W_{g} \text{ actual}}{W_{g} \text{ theor.}}$	
<u>Rub</u>	Kilometers per hour	
<b>L</b> t	Jet Mossle Groes Thrust Calibration Faster	
	$\frac{\mathbf{E}_{t}}{\mathbf{F}_{g}} = \frac{\mathbf{F}_{g \text{ actual}}}{\mathbf{F}_{g \text{ theor}}}$	
k	Any Constant	
L	Lift Force	1be
L	Length	ft
15	Pound	
ж	Mach Number (Free Stream), V/a	
н	Møter .	
Kalt	Mach Humber at Altitude for Some $V_{\rm G}$ , Relative to Mach Humber at Sea Level for Same $V_{\rm G}$	
м	Mach Rumber (Indicated)	
M <sub>ic</sub>	Mach Number (Indicated Instrument Corrected, $M_1 + \Delta M_2$ or Mach Number Computed from $V_{10}$ and $E_{10}$ )	10/
CEART 11-1	8-38	ATER 6275

-	-		_	_
<b>IN</b>	11.	ULA	10	15

0 <b>%</b> 10	Much Moter Instrument Correction for Mg	
∆M <sub>po</sub>	Mach Noter Position Error Correction, N - Nic	
Met	Nach Humber at Sea Level for Sume Vo	
M.A.C.	Nean Aerodynamic Chord	
Min	Minutes	
ж	Manifold Presence	"Hg
MP At	Temperature Corrected Manifold Pressure	"Hg
<b>△ )₽</b> €	Nanifold Pressure Correction for Air Only Stage	"Hg
∆ <b>)₽</b> ſ	Manifold Pressure Convection for Fuel-Air Mixture Stage	*B <b>C</b>
ю,	Manifold Pressure (Standard)	"Eg
юţ	Manifold Pressure (Test)	"Eg
X	Engine RFM	27 <b>9</b> 8
Hio	Regime RPN Corrected for Instrument Brree	2.Jan
¥_	Ingine REN Bequired	27 <b>9</b> 8.
<b>HIW</b>	∎√ <b>♂</b> /(₩ <sub>t</sub> /₩ <sub>g</sub> ) <sup>‡</sup>	r ja
H.	Weight Corrected 22M	27 <b>m</b>
P	Power	ft-lbe/time
Pa	Atmospheric Pressure at Pressure Altitude	*Xg
۵۳ <sub>۶</sub>	Static Pressure Source Position Broar	"Ig
2 <sub>4</sub>	Test Carburetor Deck Pressure	"Ng
۵ <b>۲</b> 10	Presence Instrument Correction	"Eg
7 <sub>8</sub>	Static Pressure (At a Point in a System)	. "Ig
₽ <b>g</b>	Static Pressure (At a Point in a System Following Normal Shock Wave, or Under Lag Conditions)	"Hg
P <sub>e1</sub>	Static Pressure (Indicated)	"Hg
· · · · · · · · · · · · · · · · · · ·		

AFTR-6275

-

CEART II-1	NONENCLATURE	
PEL	Atmospheric Pressure (See Level Standard)	29.9212 "Hg
Pt	Total Flow Pressure	"Eg
P <sub>t</sub>	Total Flow Pressure	"Hg
P <sub>t2</sub>	Total Flow Pressure (Jet Engine Compressor Inlet)	"Eg
Pti	Total Flow Pressure (Indicated)	"Eg
Ptj or Pt8	Total Flow Pressure (Jet Nozzle)	* <b>B</b> 6
Pé	Pa/PaL	
(P2/P1)a	Carburetor Deck-Ambient Pressure Ratio	
(P <sub>2</sub> /P <sub>1</sub> ) <sub>f</sub>	Manifold Pressure-Carburetor Deck Pressure Ratio (For Fuel-Air Mixture)	
pet	Pounds per Square Foot	
<b>Je</b> 1	Pounds per Square Inch	
PIN	$\operatorname{BEP}\sqrt{\sigma}/(\mathrm{W_{t}}/\mathrm{W_{e}})^{3/2}$	ft-lbg/time
۲	Dynamic Pressure, $q = \frac{p q^2}{2} = .7 P_{q} M^2$	IP\U <sub>5</sub>
٩	Impact or Differential Pressure, Pt - Pa	"Hg
۹,	Differential Pressure	"Ig
Lic	Differential Pressure Corresponding to Vic, Pt - Pe	"Hø
8	Gas Constant	96.04 ft/°X 53.355 ft/°R
R	Viscous Damping, 32 pa L/D <sup>2</sup> A	lbs-sec/ft <sup>5</sup>
•R	Degrees Bankine	
Bg	Rectore	mautical air miles or statute air miles
Re	Beynolds Radber, VL/2-	
R/C	Hate of Climb (Standard Day Speed, Test Power and Weight), (dHg/dt) $\sqrt{T_{\rm Rt}/T_{\rm RH}}$	ft/min

CHART II-1

.

ų!

8-40

AFTER-6275

NOMERCLATURE

(R/C)	Rate of Climb (Standard), $R/C + \Delta R/C_2 + \Delta n/U_3 + \Delta R/C1$	fi/miu
(R/C)t	Rate of Climb (Test), $(dH_c/dt) (T_{at}/T_{as})$	ft/mia
(R/D)	Rate of Degcent	£1/223.
8	Wing Area	ſt <sup>2</sup>
8	Take-off Ground Roll Distance	ft
<b>S</b> ′	Take-off Air Distance to 50' Obstacle	ft
84	Landing Ground Roll Distance	ſt
S <sub>2</sub>	Landing Air Distance from 50' Obstacle	ſt
8 <sub>/8</sub>	Landing Ground Roll Distance (Standard, Zero Wind)	ft
Sís	Landing Air Distance from 50' Obstacle (Standard, Zero Wind)	ſt
<sup>8</sup> Lt	Landing ground Roll Distance (Test, Zero Wind)	ft
s' <sub>lt</sub>	Landing Air Distance from 50' Obstacle (Test, - Zero Wind)	ft
8 <sub>Ltv</sub>	Landing Ground Roll Distance (Test, with Wind Component)	ſt
s' <sub>Ltw</sub>	Landing Air Distance from 50' Obstacle (Test, with Wind Component)	ft
8,	Take-off Ground Roll Distance (Standard, Zéro Wind)	ft ·
8	Take-off Air Distance to 50' Obstacle (Standard, Zero Wind)	ſt
8 <sub>t</sub>	Take-off Ground Roll Distance (Test, Zero Wind)	ft
8 <u>'</u>	Taks-off Air Distance to 50' Obstacle (Test, Zero Wind)	ſt
S <sub>tv</sub>	Take-off Ground Roll Distance (Test, with Wind Component)	ft
s <sub>tv</sub>	Take-off Air Distance to 50' Obstacle (Test, with Wind Component)	n
52	Specific Rudurance	hrs/1b

AFTE -6273

8-41

-

CHART 11-1

53#

CHART II-1	NOMENCLATURE	
<b>9</b> e0	Seconde	
Sp g	Specific Gravity	
SRE	Specific Range	nautical air miles/lb or statute air miles/lb
Т	Temperature	•K or •R
Ta	Atmospheric Temperature	•K or •R
Tas	Atmospheric Temperature (Standard)	•K or •R
T <sub>at</sub>	Atmospheric Temperature (Test)	•K or •R
T <sub>cc</sub>	Chart Carburetor Air Temperature for Pressure Altitude	•K or •R
T <sub>cs</sub>	Standard Day Carburstor Air Temperature, <sup>T</sup> as <sup>- T</sup> at <sup>+ T</sup> ot	•K or •R
T <sub>ct</sub>	Test Day Carburetor Air Temperature	•K or •R
Ti	Temperature (Indicated)	•K or •R
Tic	Temperature (Indicated Instrument Corrected)	•K or •R
T <sub>ij</sub> or T <sub>i8</sub>	Temperature (Indicated Jet Nozzle)	•K or •R
T <sub>s</sub>	Temperature (Static)	
T <b>o</b>	Temperature (Static, Following Normal Shock Wave)	•K or •R
<sup>T</sup> SL	Temperature (Standard Sea Level)	15°C 288°K 59°F 18.4°R
Tt	Temperature (Total Flow)	•K or •R
-t T <sub>t2</sub>	Temperature (Total Flow at Compressor Inlet)	•K or •R
-12 T <sub>t5</sub>	Temperature (Total Exhaust Gas Temperature at Turbine Outlet)	•K or •R
τ <sub>θ</sub>	$T_a/T_{SL}$	
TAS	True Air Speed	kaote ar mph

CHART II-1

AFTR-6273

d in

8-42

了<sup>我们</sup>这一次说,我们们的这个小小孩儿孩,我们们的这个话,我们们的这些,我们们没有这些,我们还有这个人的,我们们没有这个人,我们

CHART II-1

- S.T. S. J.

una programma este consi

### NOMENCLATURE

AT	Air-speed Position Error Correction	knots or mph
∆v <sub>pc</sub>	-	-
Vtd	Landing Velocity	knote or mph
V <sub>SL</sub>	Sea Level Standard $V_t$ for some $V_c$	knots or mph
▼ <sub>t</sub>	Air Speed (True)	knots or mph
ví	Flow True Speed Following Normal Shock Wave	knots or mph
Vto	Take-off Velocity	knots or mph
Vtoe	Take-off Velocity (Standard)	knots or mph
Vtot	Take-off Velocity (Test)	knots or mph
V <sub>ts</sub>	True Speed (Standard)	mote or mph
V <sub>tt</sub>	True Speed (Test)	mots or mph
V <sub>W</sub>	Wind Component Along Runway, Headwind (+), Tailwind (-)	knots or mph
MIA	Speed Parameter, $\nabla \sqrt{\sigma} / (W_t / W_s)^{\frac{1}{2}}$	knots or mph
¥	Gross Weight	lbs
	-	
Δ¥	Weight Increment, Wg - Wt	lbe
AW Wa	-	lbs lbs/sec
	Weight Increment, Wg - Wt	
Wa	Weight Increment, Wg - Wt Air Flow	lba/sec
Wa Wf	Weight Increment, W <sub>g</sub> - W <sub>t</sub> Air Flow Fuel Flow	lbs/sec lbs/hr
Wa Wf Wg	Weight Increment, W <sub>g</sub> - W <sub>t</sub> Air Flow Fuel Flow Gas Flow	lbs/sec lbs/hr lbs/sec
Wa Wf Wg WSIL	Weight Increment, W <sub>g</sub> - W <sub>t</sub> Air Flow Fuel Flow Gas Flow Specific Weight of Air (Sea Level Standard)	lbs/sec lbs/hr lbs/sec 0.0765 lbs/ft <sup>3</sup>
Wa Wf Wg WSIL Wa	Weight Increment, W <sub>g</sub> - W <sub>t</sub> Air Flow Fuel Flow Gas Flow Specific Weight of Air (Sea Level Standard) Gross Weight (Standard)	lbs/sec lbs/hr lbs/sec 0.0765 lbs/ft <sup>3</sup> lbs
Wa Wf Wg WgL Wa Wt	Weight Increment, W <sub>g</sub> - W <sub>t</sub> Air Flow Fuel Flow Gas Flow Specific Weight of Air (Sea Level Standard) Gross Weight (Standard)	lbs/sec lbs/hr lbs/sec 0.0765 lbs/ft <sup>3</sup> lbs
Wa Wf Wg WgIL Wa Wt	Weight Increment, $W_g - W_t$ Air Flow Fuel Flow Gas Flow Specific Weight of Air (Sea Level Standard) Gross Weight (Standard) Gross Weight (Test) $\left[\left(\frac{P_t}{P_a}\right)^{-2481}\right]$	lbs/sec lbs/hr lbs/sec 0.0765 lbs/ft <sup>3</sup> lbs
Wa Wf Wg WgL Wa Wt X	Weight Increment, $W_g - W_t$ Air Flow Fuel Flow Gas Flow Specific Weight of Air (Sea Level Standard) Gross Weight (Standard) Gross Weight (Test) $\left[\left(\frac{P_t}{P_a}\right)^{-2481}{-1}\right]$ Ratio of Specific Heat ( $\delta = 1.4$ for Air)	lbs/sec lbs/hr lbs/sec 0.0765 lbs/ft <sup>3</sup> lbs

CHART II-1

8-43

AFTER-6273

	NOMENCLATURE	CHART II-1
THP	Thrust Horsepower, BHP x 7p	550 ft-1bs/sec
(TEP) <sub>s</sub>	Thrust Hosepower (Standard)	550 ft-1bs/sec
(THP) <sub>t</sub>	Thrust Horsepower (Test)	550 ft-1bs/sec
(THP)as	Thrust Horsepower (Available Standard)	550 ft-lbs/sec
$(\text{THP})_{at}$	Thrust Horsepower (Available Test)	550 ft-lbs/sec
t	Time	
t	Temperature	•C or •F
ta	Atmospheric Temperature	•C or •F
tas	Atmospheric Temperature (Standard)	•C or •F
tat	Atmospheric Temperature (Test)	•C or •F
te	Time Elapsed	
ti	Temperature (Indicated)	°C or °F
tic	Temperature (Indicated Instrument Corrected)	°C or °F
Δt <sub>io</sub>	Temperature (Instrument Correction)	•C or •F
Δt	Compression Temperature Rise	°C or °F
te	Temperature (Static)	•C or •F
tg	Time of Start	. ·
tt	Temperature (Total Flow)	●C or ●F
v	Velocity usually	ft/sec
۷ <sub>0</sub>	Air Speed (Calibrated), $\nabla_{1c} + \Delta \nabla_{pc}$	knots or mph
v.	Air Speed (Equivalent)	knots or mph
Vi	Air Speed (Indicated)	knots or mph
Vic	Air Speed (Indicated Instrument Corrected), $V_1 + \Delta V_{ic}$	knots or mph
۵V <sub>ic</sub>	Instrument Correction for $V_i$	knots or mph
$\Delta V_{icl}$	Indicated Air-speed Correction for Lag Condition	knots or mph
AFTR-6273	0.44	CHART II-1

.

NOMENCLATURE

CHART II-1

$\eta_{\mathbf{r}}$	Ram Efficiency	
ז <sub>פד</sub>	Dict Ram Efficiency (Total, Through Normal Shock Wave and Duct)	:
η,	Propeller Efficiency	
0	Aircraft Climb Angle	
0 <sub>a</sub>	T <sub>s</sub> /T <sub>SL</sub>	
λ	Lag Constant	890
λ	Lag Constant (Altimeter)	88 C
$\lambda_{\rm hSL}$	Lag Constant (Altimeter, Sea Level)	80C
$\lambda_t$	Lag Constant (Total Pressure)	88 C
$\lambda_{tSL}$	Lag Constant (Total Pressure, Sea Level)	800
μ	Coefficiency of Viscosity	lb-jec/ft <sup>2</sup>
Horr MsL	Coefficient of Viscosity (Standard Sea Level)	3.725 x 10 <sup>-7</sup> 1b-mec/ft <sup>2</sup>
r	Einematic Viscosity, 74/9	ft <sup>2</sup> /seo
Y or YsL	Kinematic Viscosity (Sea Level Standard)	1.5665 x 10 <sup>-4</sup> ft <sup>2</sup> /eec
Ħ	Pi ·	3.1416
ያ	Air Density	slugs/ft <sup>3</sup>
for fsi	Air Dongity (Sea Level Standard)	0.002378 elugs/ft <sup>3</sup>
σ	Air Density Ratio, p/p <sub>sl</sub>	
7 <sub>RD</sub>	Duct Ram Efficiency	

AFTR -6273

8-45

CHART II-1

### CHART II-2

• • •

.

.

## SYSTEMS OF UNITS

	ABSOLL	ITE UNITS	GRAVITAT		
Name of Unit	Britigh FPS System			Metric	Physical MLT System
Length	1 ft.	l cm	1 ft.	1 м	L
Time	1 800	1 800	1 800	1 880	T
Force	l poundal	l dyne	1 16	1 Kg	MLT-2
Area	1 ft <sup>2</sup>	1 cm <sup>2</sup>	1 ft <sup>2</sup>	1 M <sup>2</sup>	r <sub>5</sub>
Volume	1 ft <sup>3</sup>	1 cm <sup>3</sup>	1 ft <sup>3</sup>	1 M <sup>3</sup>	L <sup>3</sup>
Velocity	l ft/sec	l cm/sec	l ft/sec	1 M/sec	LT <sup>-1</sup>
Acceleration	1 ft/sec <sup>2</sup>	1 cm/sec <sup>2</sup>	1 ft/sec <sup>2</sup>	1 M/sec <sup>2</sup>	LT <sup>-2</sup>
Work or Energy	l ft-pdl.	l erg	1 ft-1b	1 K3-M	ML <sup>2</sup> T <sup>-2</sup>
Prossure	1 pd1/ft <sup>2</sup>	1 dyme/cm <sup>2</sup>	1 1b/ft <sup>2</sup>	1 Kg/M <sup>2</sup>	ML-1T-2
Momentum	l pdl-sec	1 dyne-sec	1 lb-sec	1 Kg-sec	MLT <sup>-1</sup>
Power	l ft-pdl per sec.	l erg/sec	1 ft-lb per sec	1 Kg-M per sec	ML <sup>2</sup> T <sup>-3</sup>
Mass	1 16	1 Gan	1 slug	1 slug	M
Temperature	1 °R	1 °E	1 °R	1 °K	L <sup>2</sup> T <sup>-2</sup>

CHART II-2

AFTR 6273

h			AOT				<b>N</b> )		
(Ticva) buuoi		- 1					0.4536		
TIOLIX							7	5.20HÓ	
(.rd) anollad	267.40	0.1605	4.5460	1.2009	1				
(.C.U) anollad	230.98	0.1337	3.7853	ч	0.8327				
BIGILI	61.02 <b>4</b>	3-01 x	-	0.2642	0.2200 0.8327				
teet oldu?	1728	ı	28.317	7.4805	6.229				
sedoat otáv)	 	5.7870 x 10 <sup>-4</sup>	1.6387 x 10 2	52201 x	1-07 x				
Multiply Number of Obtain	Cubic inches		Lítors	Callons (U.S.)	Callone (Br.)		Ki logram	Pound (avoir)	

Volume  $(L^3)$  and Mass (M) CHART II-3

AFTR 6273

.

.

CHART II-3

.



\*Conversion factors between absolute and gravitational units apply only under standard acceleration due to gravity conditions.

AFTR 6273

CHART II-3

	Lir	ber	Veloc		[IT <sup>-1</sup> ]	)		CH	ART :	11-3
TITCHARGER DAL FOR	0.9113	1	<b>54.5</b> 6	1.036 1.10 <sup>-2</sup>	0.6214	1	0.5396	:	27.78	7
req sretestare) bacces	3.281 x 10 <sup>-2</sup>	0 8 8 8	1,969	3.728	2.237 x 10 <sup>2</sup> 2	*	1.945 x 10 <sup>-2</sup>		Ч	•036
S(Enote)		2.852h			8 5 6	1.3621	-	н	8	:
BJOUT	1.6889	•	101.34	1.9195 2-01 x	1.15155	;	1		51.48	1.8532
(Miles per Hour)		2.1511	5	-		J	1	00.7541		
mod red self	1.4667		88	.01667	-1	-	0.8684		402.44	96.3h 1.6095
etunta reg selik	88	8	880	٦	80	8	22.10		2692	4€.36
eturim reg feet	1.6667 z 10 <sup>-2</sup>	ł	1	1.894 x 10-4	1.136 x 10 <sup>-2</sup>		9.868 x 10 <sup>-5</sup>	-	080€.0	1.829 x 10-2
S( process record)		-1		1		0.4649		9:00:2:00		
Feet par second	7		3	1.136	0.6818		1366.0		30 <b>.</b> 48	1.0973
to to of the second sec	Thet per second	(Feet per second) <sup>2</sup>	Feet par minute	Miles per minute	Xilse per hour	(Miles per hour) <sup>2</sup>	Dote	(Ehote) <sup>2</sup>	Centimters per seo	Kilcmeters per hour

AFTR 6273

÷

.

1

CHART II-3

CHART II-3		Lins	ar A	cela	matio	n (11	2)		 
fucts per second	51.479	1.6890	1.8532	1.15155 	J				
Der second Miles per hour	44.704	1.4667	1.6093	1	0.8684				
per second	27.778	0.9115	I	0.6214	0.5396				
per second Feet per second	30.480	1	0.036 1.0973	0.6818	0.5921				·
Centimetere per broose reg proces	1	3.2808 10 <sup>-2</sup>		2.2369 x 10 2	9425				
Multiply Mumber of b to Obtain	Continaters per second per second	econd	lometers per hour	Miles per hour per second	Liots per second			-	
	Centimete Becond pe	Feet per s ner second	Kilometere	Miles per per second	Riots par				

CHART II-3

AFTR 6273

Angular Ve	10011	;y (T	-1) ar	nd Ar	gular	· Acc	lora	tion	(T_5)	)
Revolutions per minute per minute						1.745	2.778	1.6667 x 10-4	-	
Revolutions per Brobes reg etunik						0.1047	1.6667 x 10 2	1	60	
per Second Per Second						6.283	1	33	3600	
bactene per Second proced req						1	0.1592	9.549	573.0	
Revolutions per Minute	9	0.1047	1.6667 x 10 2	1						
Revolutions per Becond	360	6.2834	ı	8						
brood req ensibed	57.30	ч	0.1592	942.6						
bnotel ter seatiel	ı	1.745 x 10 <sup>-2</sup>	2.778 x 10-3	0.1667						
Multiply Multiply of b	Becond	Becond	767	1.84			per	per	per Umrte	
et to	Degrees per Second	Radiane per Second	Revolutions fer Second	Revolutions per Minute		Radiane per ter second		Hevolutions per Minute ver Second	Revolutions per Minute per Minute	
0	Å	2	28	23	2	28	4 2 3	9 2 3	23	

CHART II-3 coity  $(T^{-1})$  and Angular Acceleration  $(T^{-2})$ 

APTR 6273

ar -

1

7. AP

CHART II-3

CHART II-3	Kne	57,	work	and 1	leat	(ML <sup>2</sup>	'I-2)		·	
										Į
HAN	860.01	3.6000 x 1013	3413.0	2.6322 x 100	3.6710 x 105	0146.1	8.4709 x 107	1		
af abruo4-700%	10.067	<b>1.214</b> 0 x 10 <sup>5</sup>	5.9952 x 10 <sup>-5</sup>	5.1081 x 10 <sup>-2</sup>	4.2972 x 10 <sup>-3</sup>	1.5697 x 10-9	1	1.1805 = 10 <sup>-6</sup>		
Hours	641.30	2.6845 x 1015	2545.0	1.9800 × 10 <sup>5</sup>	2.1374 x 205	1	6.3796 z 10 <sup>7</sup>	0.7457		
eret <del>ek merg</del> o <b>il</b> i	2.3427 x 10-3	-8067 107	2972	7.2330	1	3.6530 x 10 <sup>-6</sup>	232.71	2.7235 x 10 <sup>-6</sup>		
Foot -Pound	5.2389 x 10-4	1.3558 # 107	1.2854 x 10 <sup>-3</sup>	-	0.1383	5.0505 * 10 <sup>-</sup> 7	32.174	3.7662 x 10-7		
(neer) UIE	0.2320	1.0548 x 1010	-	TTL-91	107.56	3.9292	2.5030 x 10 <sup>4</sup>	2.930 x 104		
Tren	2.3889 z 10-11	-	9.48051 11-01 I	7-5754		12-01 x	2.3739 x 10-6	2.7775 x 10-15		
etroleO-mergoliX	-	4.186 x 10 <sup>10</sup>	3.9685	3087.4		1.5993 x 10 <sup>-3</sup>	46.566	1.1626 x 10 <sup>-3</sup>		
Multiply Multiply Multiply of to Oftenin	Kilogram - Calories	Ergs or centimeter- druss.	(uea	Foot-Pound	Kilogram-Meters	Horsepossr Hour	<b>Poot</b> -Poundale	5		
	Ki log	Ergs o	PLA	noot Noot	K1106	Rorge	Foot	<b>E</b> 110		

CHART IL-3

AFTR 6273

· · · · · · · · · · · · · · · · · · ·				_		_	 	_	
Kilowatte	737.56	1.0197	5413.76	1.3410	ı				
Horsepower	055	7.5988 100	2-5455	1	0.7452				
BTU per hour	0.2161	2.9869 x 10 <sup>3</sup>	ı	3.9281 x 10 4	2.9292 z 10 4				
req retemitne) mers) bacceb	7.2330 × 10 <sup>-5</sup>	1	2.34T	1.3151 x 10 <sup>-7</sup>	9.8066 x 10 <sup>-8</sup>				
Foot Founds per Second	٦	1.3825 x 10 <sup>4</sup>	4.6274	1.8182 x 10 <sup>-3</sup>	1.3558 x 10 <sup>-3</sup>				
Multiply Mumber of	Foot Founds per Second	ent imetere ocnd	4	Horse power					
obtain Obtain	Poot P		d DTd	Horse	Kilowatte				

÷,

Power or Rate of Doing Work (ML<sup>2</sup>T<sup>-3</sup>) CHART II-3

AFTR 6273

(

### CHART II-3

8-53